registrant was required to submit and post such files). oximes Yes oximes No

Common Stock, \$1 par value—1,000 shares as of October 14, 2011.

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act).  $\square$  Yes  $\boxtimes$  No Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

	FORM	[ 10-Q
$\boxtimes$	Quarterly Report Pursuant to Section 13 or 15(d) of the S	ecurities Exchange Act of 1934
	For the Quarterly Period E	nded <u>September 30, 2011.</u>
	Transition Report Pursuant to Section 13 or 15(d) of the S	ecurities Exchange Act of 1934
	For the Transition Period Fro	· ·
	Commission file	
	American A (Exact name of registrant a	
	Delaware (State or other jurisdiction of incorporation or organization)	13-1502798 (I.R.S. Employer Identification No.)
	4333 Amon Carter Blvd.	rucinication Pro.)
	Fort Worth, Texas (Address of principal executive offices)	<b>76155</b> (Zip Code)
	Registrant's telephone number, in	
	Not App (Former name, former address and forme	licable
the p	cate by check mark whether the registrant (1) has filed all reports required to be preceding 12 months (or for such shorter period that the registrant was required past 90 days.   Yes  No	
	cate by check mark whether the registrant is a large accelerated filer, an accele ge accelerated filer" in Rule 12b-2 of the Exchange Act.	erated filer, or a non-accelerated filer. See definition of "accelerated filer" and
□ I	Large Accelerated Filer $\ \square$ Accelerated Filer $\ \boxtimes$ Non-accelerated Filer $\ \square$	☐ Smaller Reporting Company
	cate by check mark whether the registrant has submitted electronically and po- nitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this ch	

# INDEX

# AMERICAN AIRLINES, INC.

PART I:	FINANCIAL INFORMATION	
Item 1.	<u>Financial Statements</u>	1
	Consolidated Statements of Operations — Three and nine months ended September 30, 2011 and 2010	1
	Condensed Consolidated Balance Sheets — September 30, 2011 and December 31, 2010	2
	Condensed Consolidated Statements of Cash Flows — Nine months ended September 30, 2011 and 2010	3
	Notes to Condensed Consolidated Financial Statements — September 30, 2011	4
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	14
Item 3.	Quantitative and Qualitative Disclosures about Market Risk	26
Item 4.	Controls and Procedures	26
PART II:	OTHER INFORMATION	27
Item 1.	Legal Proceedings	27
Item 1A.	Risk Factors	29
Item 6.	<u>Exhibits</u>	39
SIGNATU	<u>URE</u>	40

# PART I: FINANCIAL INFORMATION

# **Item 1. Financial Statements**

# AMERICAN AIRLINES, INC CONSOLIDATED STATEMENTS OF OPERATIONS

(Unaudited) (In millions)

		Three Months Ended September 30,		ths Ended ber 30,
Revenues	2011	2010	2011	2010
Passenger	\$4,816	\$4,455	\$13,508	\$12,565
Regional Affiliates	735	618	2,023	1,716
Cargo	176	168	532	492
Other revenues	644	597	1,944	1,797
Total operating revenues	6,371	5,838	18,007	16,570
Expenses	,	, i	,	
Aircraft fuel	2,000	1,442	5,588	4,258
Wages, salaries and benefits	1,609	1,575	4,764	4,683
Regional payments to AMR Eagle	664	565	1,931	1,645
Other rentals and landing fees	327	320	968	956
Maintenance, materials and repairs	271	264	781	822
Commissions, booking fees and credit card expense	284	257	809	739
Depreciation and amortization	232	236	689	696
Aircraft rentals	167	150	492	430
Food service	137	129	390	365
Other operating expenses	670	609	1,968	1,848
Total operating expenses	6,361	5,547	18,380	16,442
Operating Income (Loss)	10	291	(373)	128
Other Income (Expense)				
Interest income	6	8	20	19
Interest expense	(171)	(162)	(502)	(495)
Interest capitalized	11	7	27	23
Related party interest—net	(3)	(4)	(11)	(9)
Miscellaneous — net	(6)	(11)	(29)	(33)
	(163)	(162)	(495)	(495)
Income (Loss) Before Income Taxes	(153)	129	(868)	(367)
Income tax		_		
Net Income (Loss)	\$ (153)	\$ 129	\$ (868)	\$ (367)

The accompanying notes are an integral part of these financial statements.

# AMERICAN AIRLINES, INC.

# CONDENSED CONSOLIDATED BALANCE SHEETS

(Unaudited) (In millions)

	September 30, 2011	December 31, 2010
Assets		
Current Assets		
Cash	\$ 301	\$ 165
Short-term investments	3,985	4,322
Restricted cash and short-term investments	474	450
Receivables, net	904	719
Inventories, net	586	542
Fuel derivative contracts	137	269
Other current assets	348	277
Total current assets	6,735	6,744
Equipment and Property		
Flight equipment, net	10,260	9,815
Other equipment and property, net	2,088	2,170
Purchase deposits for flight equipment	728	355
	13,076	12,340
Equipment and Property Under Capital Leases		
Flight equipment, net	338	194
Other equipment and property, net	62	49
	400	243
International slots and route authorities	708	708
Domestic slots and airport operating and gate lease rights, less accumulated amortization, net	193	212
Other assets	1,765	2,175
	\$ 22,877	\$ 22,422
Liabilities and Stockholder's Equity	<u> </u>	
Current Liabilities		
Accounts payable	\$ 1,077	\$ 1,073
Accrued liabilities	1,800	1,955
Air traffic liability	4,392	3,656
Payable to affiliates, net	2,946	2,955
Current maturities of long-term debt	1,118	1,468
Current obligations under capital leases	79	107
Total current liabilities	11,412	11,214
Long-term debt, less current maturities	7,624	6,095
Obligations under capital leases, less current obligations	614	497
Pension and postretirement benefits	7,875	7,876
Other liabilities, deferred gains and deferred credits	2,559	3,076
Stockholder's Equity		
Common stock		_
Additional paid-in capital	4,007	3,981
Accumulated other comprehensive loss	(2,894)	(2,865)
Accumulated deficit	(8,320)	(7,452)
	(7,207)	(6,336)
	\$ 22,877	\$ 22,422

The accompanying notes are an integral part of these financial statements.

# AMERICAN AIRLINES, INC.

# CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

(Unaudited) (In millions)

	Nine Mont Septem 2011	
Net Cash Provided by Operating Activities	\$ 235	\$ 953
Cash Flow from Investing Activities:		
Capital expenditures	(1,055)	(1,171)
Net (increase) decrease in short-term investments	337	(110)
Net (increase) decrease in restricted cash and short-term investments	(24)	13
Proceeds from sale of equipment and property	(9)	3
Cash collateral on spare parts financing	_	1
Net cash provided by (used for) investing activities	(751)	(1,264)
Cash Flow from Financing Activities:		
Payments on long-term debt and capital lease obligations	(1,233)	(645)
Proceeds from:		
Issuance of debt and sale leaseback transactions	2,060	1,116
Reimbursement from construction reserve account	_	1
Funds transferred to affiliates, net	(175)	(112)
Net cash provided by (used for) financing activities	652	360
Net increase (decrease) in cash	136	49
Cash at beginning of period	165	149
Cash at end of period	\$ 301	\$ 198

The accompanying notes are an integral part of these financial statements.

# AMERICAN AIRLINES, INC. Notes to Condensed Consolidated Financial Statements (Unaudited)

- 1. The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with United States (U.S.) generally accepted accounting principles for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements. In the opinion of management, these financial statements contain all adjustments, consisting of normal recurring accruals, necessary to present fairly the financial position, results of operations and cash flows for the periods indicated. Results of operations for the periods presented herein are not necessarily indicative of results of operations for the entire year. American Airlines, Inc. (American or the Company) is a wholly owned subsidiary of AMR Corporation (AMR). The condensed consolidated financial statements also include the accounts of variable interest entities for which the Company is the primary beneficiary. For further information, refer to the consolidated financial statements and footnotes included in the American Airlines, Inc. Annual Report on Form 10-K filed on February 16, 2011 (2010 Form 10-K).
- 2. On July 19, 2011, American entered into agreements (the Boeing Agreements) with The Boeing Company (Boeing) to acquire 100 additional Boeing 737 Next Generation aircraft (the firm NG Aircraft), which, subject to certain limitations, may consist of Boeing 737-700, 737-800, or 737-900 ER aircraft. The firm NG Aircraft include three Boeing 737-800 aircraft for which American exercised purchase rights on June 30, 2011. Twenty of the firm NG Aircraft are scheduled to be delivered in each of the years 2013-2017. Under the Boeing Agreements, American also expects to acquire 100 Boeing 737 Next Generation "re-engined" aircraft (the Boeing 737 MAX Aircraft), to be equipped with new, more fuel efficient engines. American's acquisition of Boeing 737 MAX Aircraft is subject to a number of conditions, including negotiation of definitive agreements with Boeing to acquire such aircraft. If acquired, 20 Boeing 737 MAX Aircraft would be scheduled to be delivered in each of the years 2018-2022. In addition, under the Boeing Agreements, American retained purchase rights for 40 Boeing 737 Next Generation aircraft, which, if exercised, would be delivered in the years 2015-2018, and acquired purchase rights for 60 additional Boeing 737 MAX Aircraft, which, if exercised, would be delivered in the years 2020-2025.

Boeing agreed to provide primary lease financing to American for the firm NG Aircraft. If American elects to use this lease financing on any firm NG Aircraft, then subject to certain terms and conditions, including the absence of defaults under certain other agreements, BCC Equipment Leasing Corporation (a subsidiary of Boeing) or a third party arranged by Boeing will enter into a lease for such aircraft with American for an initial term of ten years. Each lease will include customary terms and conditions, including covenants regarding maintenance, operation, registration, liens and insurance with respect to the aircraft, as well as defaults relating to payment and performance of lease obligations and certain cross-default arrangements. If American does not elect to lease any firm NG Aircraft using the lease financing provided by Boeing, American may purchase such aircraft using other financing provided by a third party and arranged directly by American.

Further, American entered into agreements (the Airbus Agreements) with Airbus S.A.S. (Airbus) on July 20, 2011. Under the Airbus Agreements, American committed to lease 130 Airbus current generation A320 family aircraft (the firm Current Generation Airbus Aircraft) which, subject to certain limitations, may consist of A319, A320 or A321 aircraft, and committed to purchase 130 Airbus A320 family "new engine option" aircraft (the firm NEO Airbus Aircraft), to be equipped with new, more fuel efficient engines. Between 20-35 of the firm Current Generation Airbus Aircraft are scheduled to be delivered in each of the years 2013-2017. Ten firm NEO Airbus Aircraft are scheduled to be delivered in 2017 and thereafter between 20-25 firm NEO Airbus Aircraft are scheduled to be delivered in each of the years 2018-2022. In addition, American acquired 70 options and 15 purchase rights for additional Airbus current generation A320 family aircraft, which, if exercised, would be delivered in years 2014-2017, and options for 280 additional Airbus A320 family "new engine option" aircraft, which, if exercised, would be delivered in the years 2017-2025. Under the Airbus Agreements, subject to American's rights to purchase firm Current Generation Airbus Aircraft in certain circumstances, and subject to certain terms and conditions, including the absence of defaults under certain other agreements, the firm Current Generation Airbus Aircraft will be financed under leases with initial terms of ten years with Airbus or one of its affiliates, or with a third party arranged by Airbus. The leases will include customary terms and conditions, including covenants regarding maintenance, operation, registration, liens and insurance with respect to the aircraft, as well as defaults relating to payment and performance of lease obligations and certain cross-default arrangements.

# AMERICAN AIRLINES, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

In connection with the aircraft agreements entered into with Boeing and Airbus during the third quarter of 2011 and the Company's acceleration of its fleet renewal and replacement plan, the Company continues to evaluate the useful lives of certain fleets including MD-80, B757 and B767 aircraft. As of September 30, 2011, the Company is still finalizing its fleet plans and, accordingly, there is no event requiring assessment of whether certain fleets may be impaired. Due to various factors, the Company continues to evaluate its capacity needs in 2012 and beyond based on general economic conditions and fuel cost. Upon finalization of the fleet plan, which is anticipated in the fourth quarter, including potential changes to the useful lives of the aircraft, an analyses of their recoverability may be required. Additionally, to the extent the fleet plan involves grounding leased aircraft, we may record exit costs associated with future lease payments when such aircraft are grounded. If such analyses indicate there is an impairment, a charge could be significant.

Reflecting the above transactions, American had total aircraft acquisition commitments as of September 30, 2011 as follows:

		Boeing		Airbus			
		737 Family <sup>1</sup>	777- 200 ER <sup>2</sup>	777- 300 ER <sup>2</sup>	A320 Family	NEO	Total
Remainder of 2011	Purchase	6					6
	Lease						_
2012	Purchase	28		2			30
	Lease						_
2013	Purchase	15		7			22
	Lease	16			20		36
2014	Purchase	5	2				7
	Lease	15			35		50
2015	Purchase		2				2
	Lease	20			30		50
2016 and beyond	Purchase		2			130	132
	Lease	40			45		85
Total	Purchase	54	6	9	0	130	199
	Lease	91	0	0	130	0	221

- As of September 30, 2011, American had elected to purchase nine of the 100 Boeing 737-800 NG aircraft using the sale-leaseback financing arranged directly by American with a third party leasing company. These aircraft are therefore reflected as purchases in the above table.
- During the third quarter of 2011, the Company exercised rights to acquire an additional Boeing 777-300ER aircraft, scheduled for delivery in 2013, and converted a Boeing 777-200ER aircraft purchase commitment to a Boeing 777-300ER aircraft purchase commitment.

As of September 30, 2011, payments for the above purchase commitments and certain engines will approximate \$277 million in the remainder of 2011, \$1.1 billion in 2012, \$1.4 billion in 2013, \$494 million in 2014, \$243 million in 2015, and \$7.6 billion for 2016 and beyond. These amounts are net of purchase deposits currently held by the manufacturers. American has granted Boeing a security interest in American's purchase deposits with Boeing. The Company's purchase deposits totaled \$728 million as of September 30, 2011.

As of September 30, 2011, total future lease payments for all leased aircraft, including aircraft not yet delivered, will approximate \$164 million in the remainder of 2011, \$912 million in 2012, \$888 million in 2013, \$989 million in 2014, \$1.1 billion in 2015, and \$14.3 billion in 2016 and beyond.

# AMERICAN AIRLINES, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

In 2008, American entered into a purchase agreement with Boeing (subject to certain reconfirmation rights) to acquire 42 Boeing 787-9 aircraft, with the right to acquire an additional 58 Boeing 787-9 aircraft. The first such Boeing 787-9 aircraft is currently scheduled to be delivered (subject to such reconfirmation rights) in 2014. American has selected GE Aviation as the exclusive provider of engines for its expected order of Boeing 787-9 aircraft.

In 2010, American and Japan Airlines (JAL) entered into a Joint Business Agreement (JBA) to enhance their scope of cooperation on routes between North America and Asia through adjustments to their respective networks, flight schedules, and other business activities. American and JAL began implementing the JBA on April 1, 2011. The JBA provides for expanded codesharing, enhanced frequent flyer program reciprocity, and cooperation in other areas. American and JAL entered into a Revenue Sharing Agreement, effective April 1, 2011, as envisaged by the JBA. Under this agreement, American and JAL share certain revenues of their operations. In addition, American provided JAL a guarantee of certain minimum incremental revenue resulting from the successful operation of the joint business for the first three years following its implementation, subject to certain terms and conditions. The amount required to be paid by the Company under the guarantee in any one of such years may not exceed \$100 million, and is reduced if capacity for one of such years is less than a defined base year period capacity. Based on current Trans-Pacific capacity, the guarantee in any one of such years may not exceed approximately \$75 million. As of September 30, 2011, based on an expected probability model, American had recorded a guarantee liability that is not material.

- Accumulated depreciation of owned equipment and property at September 30, 2011 and December 31, 2010 was \$10.7 billion and \$9.8 billion, respectively. Accumulated amortization of equipment and property under capital leases at September 30, 2011 and December 31, 2010 was \$496 million and \$579 million, respectively.
- 4. The Company provides a valuation allowance for deferred tax assets when it is more likely than not that some portion, or all, of its deferred tax assets will not be realized. The Company's deferred tax asset valuation allowance increased from \$3.7 billion as of December 31, 2010 to \$4.0 billion as of September 30, 2011, including the impact of comprehensive income for the nine months ended September 30, 2011 and changes from other adjustments.

Under current accounting rules, the Company is required to consider all items (including items recorded in other comprehensive income) in determining the amount of tax benefit that results from a loss from continuing operations and that should be allocated to continuing operations. The Company generally does not record any such tax benefit allocation in interim reporting periods as the Company concluded the potential benefit is not considered realizable because the change in the pension liability, a material component of other comprehensive income, is determined annually. Thus, any such interim tax benefit allocation may subsequently be subject to reversal.

See also "Notes to Consolidated Financial Statements: Note 10" for information on the impact of the potential AMR Eagle divesture.

# AMERICAN AIRLINES, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

#### 5. Long-term debt consisted of (in millions):

	Sep	tember 30, 2011	ember 31, 2010
Secured variable and fixed rate indebtedness due through 2021 (effective rates from 2.28%—13.00% at September 30,			
2011)	\$	4,212	\$ 3,002
Enhanced equipment trust certificates due through 2021 (rates from 5.10%—12.00% at September 30, 2011)		1,985	2,002
6.00%—8.50% Special facility revenue bonds due through 2036		1,627	1,641
AAdvantage Miles advance purchase (net of discount of \$110 million) (effective rate 8.30%)		890	890
Other		28	28
		8,742	 7,563
Less current maturities		1,118	 1,468
Long-term debt, less current maturities	\$	7,624	\$ 6,095

The Company's future long-term debt and operating lease payments have changed as its ordered aircraft are delivered and such deliveries have been financed. As of September 30, 2011, maturities of long-term debt (including sinking fund requirements) for the next five years are: remainder of 2011 – \$1.0 billion, 2012 – \$1.5 billion, 2013 – \$876 million, 2014 – \$839 million, and 2015 – \$594 million. Future minimum lease payments required under operating leases that have initial or remaining non-cancelable lease terms in excess of a year as of September 30, 2011, were: remainder of 2011 – \$306 million, 2012 – \$1.1 billion, 2013 – \$996 million, 2014 – \$855 million, 2015 – \$698 million, and 2016 and beyond – \$6.3 billion.

As of September 30, 2011, American had issued guarantees covering approximately \$848 million of AMR's unsecured debt (and interest thereon). In addition, as of September 30, 2011, AMR and American had issued guarantees covering approximately \$170 million of AMR Eagle's secured debt (and interest thereon).

On January 25, 2011, American closed on a \$657 million offering of Class A and Class B Pass Through Trust Certificates, Series 2011-1 (the 2011-1 Certificates). The equipment notes held by each pass through trust were issued for each of (a) 15 Boeing 737-823 aircraft delivered new to American from 1999 to 2001, (b) six Boeing 757-223 aircraft delivered new to American in 1999 and 2001, (c) two Boeing 767-323ER aircraft delivered new to American in 1999 and 2000. At closing, 27 of the aircraft were encumbered by either private mortgages or by liens to secure debt incurred in connection with the issuance of enhanced equipment trust certificates in 2001, all of which mature in 2011. As a result, the proceeds from the sale of the 2011-1 Certificates of each trust were initially held in escrow with a depositary, pending the financing of each aircraft under an indenture relating to the 2011-1 Certificates. Interest of 5.25% and 7.00% per annum on the issued and outstanding Series A equipment notes and Series B equipment notes, respectively, will be payable semiannually on January 31 and July 31 of each year, commencing on July 31, 2011, and principal on such equipment notes is scheduled for payment on January 31 and July 31 of certain years, commencing on July 31, 2011. The payment obligations of American under the equipment notes are fully and unconditionally guaranteed by AMR. Approximately \$598 million of the proceeds from sale of the 2011-1 Certificates had been received by American as of September 30, 2011, in exchange for equipment notes secured by (a) 15 Boeing 737-823 aircraft, (b) two Boeing 757-223 aircraft (c) two Boeing 767-323 aircraft, and (d) seven Boeing 777-223 aircraft. Approximately \$59 million of the proceeds from the sale of the 2011-1 Certificates is expected to be received in the fourth quarter of 2011.

# AMERICAN AIRLINES, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

In March 2011, American issued \$1 billion aggregate principal amount of senior secured notes due 2016 (the Senior Secured Notes) guaranteed by the Company. The Senior Secured Notes bear interest at a rate of 7.50% per annum, payable semi-annually on March 15 and September 15 of each year, beginning September 15, 2011. As is customary for financings of this nature, the indebtedness evidenced by the Senior Secured Notes may be accelerated upon the occurrence of events of default under the related indenture. The Senior Secured Notes are senior secured obligations of American and are unconditionally guaranteed on an unsecured basis by the Company. Subject to certain limitations and exceptions, the Senior Secured Notes are secured by certain route authorities, airport landing and takeoff slots, and rights to use or occupy space in airport terminals, in each case that American uses to operate non-stop services between certain airports in the United States and certain airports in Japan and China.

American, at its option, may redeem some or all of the Senior Secured Notes at any time on or after March 15, 2013, at specified redemption prices, plus accrued and unpaid interest, if any. In addition, at any time prior to March 15, 2013, American, at its option, may redeem some or all of the Senior Secured Notes at a redemption price equal to 100% of their principal amount plus a "make-whole" premium and accrued and unpaid interest, if any. In addition, at any time prior to March 15, 2014, American, at its option, may redeem (1) up to 35% of the aggregate principal amount of the Senior Secured Notes with the proceeds of certain equity offerings at a redemption price of 107.5% of their principal amount, plus accrued and unpaid interest, if any, and (2) during any 12-month period, up to 10% of the original aggregate principal amount of the Senior Secured Notes at a redemption price of 103% of their principal amount, plus accrued and unpaid interest, if any. If American sells certain assets or if a "change of control" (as defined in the indenture) occurs, American must offer to repurchase the Senior Secured Notes at prices specified in the indenture.

The indenture for the Senior Secured Notes includes covenants that, among other things, limit the ability of the Company and its subsidiaries to merge, consolidate, sell assets, incur additional indebtedness, issue preferred stock, make investments and pay dividends. In addition, if American fails to maintain a collateral ratio of 1.5 to 1.0, American must pay additional interest on the notes at the rate of 2% per annum until the collateral coverage ratio equals at least 1.5 to 1.0.

Further, in July 2011, American entered into a sale-leaseback arrangement with a leasing company to finance 35 Boeing 737-800 aircraft scheduled to be delivered in 2011 through 2014. The financing of each aircraft under this arrangement will be subject to certain terms and conditions. As of the end of the third quarter of 2011, American had financed two Boeing 737-800 aircraft under this arrangement, which are accounted for as operating leases.

Almost all of the Company's aircraft assets (including aircraft eligible for the benefits of Section 1110 of the U.S. Bankruptcy Code) are encumbered, and the Company has a very limited quantity of assets which could be used as collateral in future financing.

See also "Notes to Consolidated Financial Statements: Note 10" for information on the impact of the potential AMR Eagle divesture.

#### AMERICAN AIRLINES, INC.

#### NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

The Company utilizes the market approach to measure fair value for its financial assets and liabilities. The market approach uses prices and other relevant information generated by market transactions involving identical or comparable assets or liabilities. The Company's short-term investments classified as Level 2 primarily utilize broker quotes in a non-active market for valuation of these securities. The Company's fuel derivative contracts, which include, among other things, heating oil, jet fuel, and WTI crude option and collar contracts, are valued using energy and commodity market data which is derived by combining raw inputs with quantitative models and processes to generate forward curves and volatilities. No changes in valuation techniques or inputs occurred during the nine months ended September 30, 2011.

Assets and liabilities measured at fair value on a recurring basis are summarized below:

(in millions)		Fair Value Measurements as of September 30, 20		
	Total	Level 1	Level 2	Level 3
Short-term investments 1, 2				
Money market funds	\$ 1,069	\$ 1,069	\$ —	\$ —
Government agency investments	528	_	528	_
Repurchase investments	811	_	811	_
Corporate obligations	806	_	806	_
Bank notes / Certificates of deposit / Time deposits	771	_	771	_
	3,985	1,069	2,916	_
Restricted cash and short-term investments <sup>1</sup>	474	474	_	_
Fuel derivative contracts, net <sup>1</sup>	106	_	106	_
Total	\$ 4 565	\$ 1543	\$ 3,022	<u>s</u> —

<sup>1</sup> Unrealized gains or losses on short-term investments, restricted cash and short-term investments and derivatives qualifying for hedge accounting are recorded in Accumulated other comprehensive income (loss) (OCI) at each measurement date.

The majority of the Company's short-term investments mature in one year or less except for \$569 million of Bank notes/Certificates of deposit/Time deposits, \$529 million of U.S. Government agency investments and \$702 million of Corporate obligations which have maturity dates exceeding one year.

No significant transfers between Level 1 and Level 2 occurred during the nine months ended September 30, 2011. The Company's policy regarding the recording of transfers between levels is to reflect any such transfers at the end of the reporting period.

The fair values of the Company's long-term debt were estimated using quoted market prices where available. For long-term debt not actively traded, fair values were estimated using discounted cash flow analyses, based on the Company's current estimated incremental borrowing rates for similar types of borrowing arrangements.

### AMERICAN AIRLINES, INC.

# NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

The carrying value and estimated fair values of the Company's long-term debt, including current maturities, were (in millions):

	Septembe	er 30, 2011	Decembe	r 31, 2010
	Carrying	Fair	Carrying	Fair
	Value	Value	Value	Value
Secured variable and fixed rate indebtedness	\$4,212	\$3,675	\$3,002	\$2,907
Enhanced equipment trust certificates	1,985	1,935	2,002	2,127
6.0%—8.5% Special facility revenue bonds	1,627	1,604	1,641	1,657
AAdvantage Miles advance purchase	890	906	890	903
Other	28	28	28	28
	# O = 40	<b></b>	<b>A. T. C.</b> O	A = .000
	\$8,742	\$8,148	\$7,563	\$7,622

7. The following tables provide the components of net periodic benefit cost for the three and nine months ended September 30, 2011 and 2010 (in millions):

		Doneion	Benefits	
	Sept	Months Ended ember 30,	Nine Mor Septen	nths Ended nber 30,
Components of net periodic benefit cost		2010	2011	2010
Components of her periodic benefit cost				
Service cost	\$ 97	\$ 91	\$ 289	\$ 275
Interest cost	189	184	568	553
Expected return on assets	(165)	(148)	(493)	(445)
Amortization of:				
Prior service cost	3	3	10	10
Unrecognized net loss	39	39	115	115
Net periodic benefit cost	<u>\$ 163</u>	<u>\$ 169</u>	<u>\$ 489</u>	\$ 508
	Ī	Retiree Medical and	d Other Benefits	
	Three Mon		Nine Mon	
	Three Mon Septemb	ths Ended oer 30,	Nine Mon Septem	ber 30,
Components of net periodic benefit cost	Three Mon	ths Ended	Nine Mon	
Components of net periodic benefit cost  Service cost	Three Mon Septemb	ths Ended oer 30,	Nine Mon Septem	ber 30,
	Three Mon Septeml 2011	ths Ended oer 30, 2010	Nine Mon Septem 2011	ber 30, 2010
Service cost Interest cost	Three Mon Septeml 2011  \$ 15 44	ths Ended oer 30, 2010 \$ 15 41	Nine Mon Septem 2011 \$ 45 132	\$ 45 124
Service cost	Three Mon Septeml 2011 \$ 15	ths Ended oer 30, 2010	Nine Mon Septem 2011 \$ 45	ber 30, 2010 \$ 45
Service cost Interest cost Expected return on assets	Three Mon Septeml 2011  \$ 15 44	ths Ended oer 30, 2010 \$ 15 41	Nine Mon Septem 2011 \$ 45 132	\$ 45 124
Service cost Interest cost Expected return on assets Amortization of:	Three Mon September 2011  \$ 15 44 (5)	ths Ended per 30, 2010 \$ 15 41 (4)	Nine Mon Septem 2011 \$ 45 132 (15)	\$ 45 124 (13)

#### AMERICAN AIRLINES, INC.

#### NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

The Company is required to make minimum contributions to its defined benefit pension plans under the minimum funding requirements of the Employee Retirement Income Security Act (ERISA), the Pension Funding Equity Act of 2004, the Pension Protection Act of 2006, and the Pension Relief Act (Relief Act) of 2010. Under the Relief Act, the Company estimates its 2011 minimum required contribution to its defined benefit pension plans to be approximately \$520 million, which has been completed as of the date of this filing. The Company estimates its 2012 minimum required contribution to its defined benefit pension plans will not exceed \$560 million. This estimate is subject to change based on final plan asset values as of December 31, 2011.

8. As a result of the revenue environment, high fuel prices and the Company's restructuring activities, including its capacity reductions, the Company has recorded a number of charges during the last few years. In 2008 and 2009, the Company announced capacity reductions due to unprecedented high fuel costs at that time and the other challenges facing the industry. In connection with these capacity reductions, the Company incurred special charges related to aircraft and certain other charges.

The following table summarizes the components of the Company's special charges, the remaining accruals for these charges and the capacity reduction related charges (in millions) as of September 30, 2011:

	Aircraft	Facility Exit	
	Charges	Costs	Total
Remaining accrual at December 31, 2010	\$ 59	\$ 27	\$ 86
Capacity reduction charges	_		_
Non-cash charges	_	_	_
Adjustments	26	1	27
Payments	(35)	(3)	(38)
Remaining accrual at September 30, 2011	\$ 50	\$ 25	\$ 75

Cash outlays related to the accruals for aircraft charges and facility exit costs will occur through 2017 and 2018, respectively.

As part of the Company's risk management program, it uses a variety of financial instruments, including heating oil, jet fuel, and WTI crude option and collar contracts, as cash flow hedges to mitigate commodity price risk. The Company does not hold or issue derivative financial instruments for trading purposes. As of September 30, 2011, the Company had fuel derivative contracts outstanding covering 25 million barrels of jet fuel that will be settled over the next 15 months. A deterioration of the Company's liquidity and financial position may negatively affect the Company's ability to hedge fuel in the future.

For the three and nine months ended September 30, 2011, the Company recognized a decrease of approximately \$28 million and \$238 million, respectively, in fuel expense on the accompanying consolidated statements of operations related to its fuel hedging agreements, including the ineffective portion of the hedges. For the three and nine months ended September 30, 2010, the Company recognized an increase of approximately \$20 million and \$123 million, respectively, in fuel expense on the accompanying consolidated statements of operations related to its fuel hedging agreements, including the ineffective portion of its hedges. The net fair value of the Company's fuel hedging agreements at September 30, 2011 and December 31, 2010, representing the amount the Company would receive upon termination of the agreements (net of settled contract assets), totaled \$78 million and \$257 million, respectively.

#### AMERICAN AIRLINES, INC.

### NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

The impact of cash flow hedges on the Company's consolidated financial statements is depicted below (in millions):

Fair Value of Aircraft Fuel Derivative Instruments (all cash flow hedges)

Asset Derivatives as of				Liability Derivatives as of			
September 30, 2011		December 31, 2010		September 30, 2011		December	31, 2010
Balance Sheet	T	Balance Sheet	Fair	Balance Sheet	Fair	Balance Sheet	Fair
Location Fuel derivative	Fair Value	Location	Value	Location	Value	Location	Value
contracts	\$137	Fuel derivative contracts	\$269	Accrued liabilities	\$31	Accrued liabilities	\$—

Effect of Aircraft Fuel Derivative Instruments on Statements of Operations (all cash flow hedges)

Amount of Gain (Loss) Recognized in OCI on Derivative <sup>1</sup> for the nine months ended September 30,		Location of Gain (Loss) Reclassified from Accumulated OCI into Income <sup>1</sup>	Amount of Gain (Loss) Reclassified from Accumulated OCI into Income <sup>1</sup> for the nine months ended September 30,		Location of Gain (Loss) Recognized in Income on Derivative <sup>2</sup>	Amount of Gain (Loss) Recognized in Income on Derivative <sup>2</sup> for the nine months ended September 30,	
2011	2010		2011	2010		2011	2010
\$ 152	\$(56)	Aircraft Fuel	\$255	\$(121)	Aircraft Fuel	\$(17)	\$(2)
Amount of Gain (Loss) Recognized in OCI on Derivative <sup>1</sup> for the three months ended September 30,		Location of Gain (Loss) Reclassified from Accumulated OCI into Income <sup>1</sup>	Amount of Gain (Loss) Reclassified from Accumulated OCI into Income <sup>1</sup> for the three months ended September 30,		Location of Gain (Loss) Recognized in Income on Derivative <sup>2</sup>	Amount of Gain (Loss) Recognized in Income on Derivative <sup>2</sup> for the three months ended September 30,	
2011	2010	· ·	2011	2010	· ·	2011	2010
\$ (142)	\$67	Aircraft Fuel	\$51	\$(27)	Aircraft Fuel	\$(23)	\$7

Effective portion of gain (loss)

The Company is also exposed to credit losses in the event of non-performance by counterparties to these financial instruments, and although no assurances can be given, the Company does not expect any counterparty to fail to meet its obligations. The credit exposure related to these financial instruments is represented by the fair value of contracts with a positive fair value at the reporting date, reduced by the effects of master netting agreements. To manage credit risks, the Company selects counterparties based on credit ratings, limits its exposure to a single counterparty under defined guidelines, and monitors the market position of the program and its relative market position with each

<sup>&</sup>lt;sup>2</sup> Ineffective portion of gain (loss)

# AMERICAN AIRLINES, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

counterparty. The Company also maintains industry-standard security agreements with a number of its counterparties which may require the Company or the counterparty to post collateral if the value of selected instruments exceeds specified mark-to-market thresholds or upon certain changes in credit ratings.

As of September 30, 2011, the Company had received cash collateral of \$28 million which is included in short-term investments. The Company includes changes in the fair value of certain derivative financial instruments that qualify for hedge accounting and unrealized gains and losses on available-for-sale securities in comprehensive income. For the three month periods ended September 30, 2011 and 2010, comprehensive income (loss) was \$(324) million and 263 million, respectively. For the nine month periods ended September 30, 2011 and 2010, comprehensive income (loss) was \$(897) million and \$(183) million, respectively. The difference between net earnings (loss) and comprehensive income (loss) for the three month periods ended September 30, 2011 and 2010 is due primarily to the accounting for the Company's derivative financial instruments and the actuarial loss on the pension benefit obligation of the Company's pension plans.

10. On August 11, 2011, AMR Eagle filed a Form 10 registration statement (subsequently amended on September 26, 2011 and October 6, 2011) with the Securities and Exchange Commission in connection with a potential spin-off of AMR Eagle.

In a spin-off, AMR would distribute to its shareholders 100 percent of the outstanding shares of AMR Eagle on a pro-rata basis, and AMR would not retain any ownership interest in AMR Eagle. On a historical basis, in 2010, AMR Eagle, through its subsidiaries, generated \$2.3 billion in revenue with more than \$300 million from ground handling services. AMR Eagle would operate the third largest regional airline in the United States as it provides the vast majority of American's regional flight operations. Under a nine-year services agreement with American, AMR Eagle would initially operate 281 aircraft on behalf of American. American could withdraw from AMR Eagle a specified number of jet aircraft up to 40 per year beginning in 2014 and certain turbo prop aircraft each year. The agreement would also include a provision to re-set rates to reflect any change in market levels for regional feed after four years.

AMR Eagle also would operate one of the largest ground handling operations in the U.S. serving American and other passenger airlines at more than 100 airports in the U.S., the Bahamas, the Caribbean and Canada. Under a ground handling agreement, AMR Eagle would have an eight-year term, but provide American the right to re-bid ground handling services at a specified number of airports each year.

While AMR Eagle's wholly-owned subsidiaries, American Eagle Airlines, Inc. ("Eagle") and Executive Airlines, Inc. ("Executive") are expected to continue to operate all of the Jet Aircraft (as defined below), those aircraft and the associated indebtedness will be transferred to American pursuant to the Purchase Agreement referred to below. AMR currently guarantees the indebtedness relating to each Jet Aircraft and AMR will continue to guarantee such indebtedness following American's purchase of each Jet Aircraft subject to such indebtedness. Ownership of the Jet Aircraft by American is intended to provide American control over the regional aircraft that are pivotal to its network and to protect AMR's position as the guarantor of the related indebtedness.

As contemplated by the Form 10, on August 31, 2011, American entered into a Master Purchase Agreement (the "Purchase Agreement") with Eagle and Executive under which Eagle will sell to American 47 CRJ-700 Jet Aircraft and 216 Embraer 135, 140 and 145 Jet Aircraft, including the engines installed on each such aircraft and other related assets (each, a "Jet Aircraft"). In addition, American will purchase from Eagle and Executive certain specified fixed assets, generally consisting of equipment and leasehold improvements owned by Eagle or Executive and used in connection with the regional flight operations conducted by Eagle and Executive on American's behalf and the ground handling operations of Eagle and Executive (collectively, the "Other Assets").

Each Jet Aircraft will be purchased by American on the date of delivery of such aircraft to American, and the Other Assets will be purchased by American ten days after delivery of the last Jet Aircraft to American, subject in each case to the satisfaction of certain conditions. Delivery of the Jet Aircraft began on August 31, 2011, and the last Jet Aircraft is expected to be delivered in October 2011. Following the delivery of each Jet Aircraft, American will lease the Jet Aircraft to Eagle, and Eagle will continue to provide certain regional flight operations to American.

American will take each Jet Aircraft subject to, and Eagle will be released from, all outstanding indebtedness relating to such Jet Aircraft. The indebtedness related to the Jet Aircraft consists of individual notes for each Jet Aircraft. The notes are secured by the related Jet Aircraft and certain other assets, have either fixed or floating interest rates and mature over various periods through 2023. As of September 30, 2011, the fixed rate notes had effective interest rates ranging from 4.25% to 7.50% and the floating rate notes had effective interest rates ranging from 2.247% to 3.128%. The notes include customary terms and conditions, including customary events of default and certain cross-default provisions.

As of the end of the third quarter of 2011, Eagle had delivered 103 Embraer 135, 140 and 145 Jet Aircraft to American. As of the end of the third quarter 2011, the net book value of such transferred Jet Aircraft was \$877 million, and the aggregate outstanding indebtedness (net of discount) associated with such transferred Jet Aircraft was \$712 million. Upon completion of the transfer of the Jet Aircraft from Eagle to American, it is expected that American's aggregate outstanding indebtedness (net of discount) related to the Jet Aircraft will be approximately \$2.1 billion.

In connection with a spin-off, AMR would allocate a portion of its net operating loss ("NOL") carryover balance for U.S. Federal income tax purposes to AMR Eagle. Following distribution, approximately \$800 million of the \$6.7 billion NOL carryover balance would be allocated to AMR Eagle. Further, the fleet transfer from AMR Eagle to American would result in a \$1.1 billion reduction in the Company's NOL carryover balance, which would be effectively restored through future tax depreciation.

The spin-off of Eagle would be subject to certain conditions, including U.S. Securities and Exchange Commission (SEC) clearance, receipt of regulatory approvals, an opinion from tax counsel and a favorable ruling from the Internal Revenue Service regarding the tax-free status of the spin-off to AMR shareholders, execution of inter-company agreements and approval by AMR's board of directors. Stockholder approval of the spin-off is not required. While AMR Corporation has taken this step toward a spin-off of Eagle, the timing of any spin-off could be dependent on market conditions and other factors, and is therefore uncertain. In addition, AMR could decide to retain Eagle, or the divestiture of Eagle could take another form, such as a sale.

11. In connection with preparation of the consolidated financial statements and in accordance with U.S. GAAP, the Company evaluated subsequent events after the balance sheet date of September 30, 2011 and identified the following item.

On October 4, 2011, American closed on a \$726 million offering of Class A Pass Through Trust Certificates, Series 2011-2 (the 2011-2 Certificates). Subsequent to the closing, the equipment notes held by each pass through trust were issued for each of (a) 14 Boeing 737-823 aircraft

delivered new to American from 1999 to 2001 and 2 Boeing 737-823 aircraft delivered new to American in 2009, (b) 14 Boeing 757-223 aircraft delivered new to American in 1999 and 2001 and (c) 13 Boeing 777-223ER aircraft delivered new to American in 2001. Interest of 8.625% per annum on the issued and outstanding 2011-2 Certificates will be payable semiannually on April 15 and October 15 of each year, commencing on April 15, 2012, and principal on such equipment notes is scheduled for payment on April 15 and October 15 of certain years, commencing on April 15, 2012. The payment obligations of American under the equipment notes are fully and unconditionally guaranteed by AMR. All proceeds from the sale of the Series 2011-2 Certificates have been received by American.

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

#### **Forward-Looking Information**

Statements in this report contain various forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, which represent the Company's expectations or beliefs concerning future events. When used in this document and in documents incorporated herein by reference, the words "expects," "estimates," "plans," "anticipates," "indicates," "believes," "forecast," "guidance," "outlook," "may," "will," "should," "seeks," "targets" and similar expressions are intended to identify forward-looking statements. Similarly, statements that describe the Company's objectives, plans or goals, or actions the Company may take in the future, are forward-looking statements. Forward-looking statements include, without limitation, the Company's expectations concerning operations and financial conditions, including changes in capacity, revenues, and costs; future financing plans and needs; the amounts of its unencumbered assets and other sources of liquidity; fleet plans; overall economic and industry conditions; plans and objectives for future operations; a potential spin-off or other divestiture of AMR Eagle; regulatory approvals and actions; and the impact on the Company of its results of operations in recent years and the sufficiency of its financial resources to absorb that impact. Other forward-looking statements include statements which do not relate solely to historical facts, such as, without limitation, statements which discuss the possible future effects of current known trends or uncertainties, or which indicate that the future effects of known trends or uncertainties cannot be predicted, guaranteed or assured. All forward-looking statements in this report are based upon information available to the Company on the date of this report. The Company undertakes no obligation to publicly update or revise any forward-looking statement, whether as a result of new information, future events, or otherwise.

Guidance given in this report regarding capacity, fuel consumption, fuel prices, fuel hedging and unit costs are forward-looking statements. Forward-looking statements are subject to a number of factors that could cause the Company's actual results to differ materially from the Company's expectations. The following factors, in addition to other possible factors not listed, could cause the Company's actual results to differ materially from those expressed in forward-looking statements: the materially weakened financial condition of the Company, resulting from its significant losses in recent years; weak demand for air travel and lower investment asset returns resulting from the severe global economic downturn; the Company's need to raise substantial additional funds and its ability to do so on acceptable terms; the potential requirement for the Company to maintain reserves under its credit card processing agreements, which could materially adversely impact the Company's liquidity; the ability of the Company to generate additional revenues and reduce its costs; continued high and volatile fuel prices and further increases in the price of fuel, and the availability of fuel; the resolution of pending litigation with certain global distribution systems and business discussions with certain on-line travel agents; the Company's substantial indebtedness and other obligations; the ability of the Company to satisfy certain covenants and conditions in certain of its financing and other agreements; changes in economic and other conditions beyond the Company's control, and the volatile results of the Company's operations; the fiercely and increasingly competitive business environment faced by the Company; industry consolidation and alliance changes; competition with reorganized carriers; low fare levels by historical standards and the Company's reduced pricing power; changes in the Company's corporate or business strategy; extensive government regulation of the Company's business; conflicts overseas or terrorist attacks; uncertainties with respect to the Company's international operations; outbreaks of a disease (such as SARS, avian flu or the H1N1 virus) that affects travel behavior; labor costs that are higher than those of the Company's competitors; uncertainties with respect to the Company's relationships with unionized and other employee work groups; increased insurance costs and potential reductions of available insurance coverage; the Company's ability to retain key management personnel; potential failures or disruptions of the Company's computer, communications or other technology systems; losses and adverse publicity resulting from any accident involving the Company's aircraft; interruptions or disruptions in service at one or more of the Company's primary market airports; the heavy taxation of the airline industry; and changes in the price of the Company's common stock. Additional information regarding these factors is set forth in the Risk Factors listed in Item 1A.

#### **Recent Events**

The Company continued to face challenges in the third quarter of 2011, including an uncertain global economy and persistently high and volatile fuel prices. Further, the Company's financial performance relative to its competitors continues to lag. The Company's biggest competitive challenge is its labor costs which, based on Company estimates of its labor cost disadvantage (the amount by which labor costs exceed what such costs would be if they were determined based on the average of other network carrier labor contracts), are significantly higher than other network carriers. To improve its competitive position, the Company has taken significant steps, including:

- Network: Restructured its network to concentrate in the most important business markets and maximize its revenue-generation potential
- *Alliance and Joint Business Agreements:* Implemented joint business agreements across both the Atlantic and Pacific and optimizing joint business results through better coordination of schedules and focused sales efforts amongst the joint business airlines
- · Service and Product: Enhanced customer experience from dependability to new technology to improved cabin environment and onboard services
- *New Fleet*: Accelerated its fleet renewal plan with new Boeing and Airbus agreements to secure new and more fuel efficient aircraft that will also reduce maintenance costs

The Company is in ongoing negotiations with all of its organized labor groups to address its competitive challenges and fully capitalize on these important strategic actions. The outcome of the negotiations cannot be predicted.

#### Aircraft Agreements

American entered into agreements with Airbus and Boeing under which it plans to acquire 460 narrowbody aircraft from the Boeing 737 and Airbus A320 families during the period 2013-2022. These agreements will allow American to replace and transform its narrowbody fleet over five years and solidify its fleet plan into the next decade. These new aircraft will allow American to reduce its operating and fuel costs and deliver state-of-the-art amenities to customers, while maximizing financial flexibility for American. American also has purchase rights and options through 2025 for an additional 465 aircraft from these families. As part of these agreements, starting in 2017, American expects to become the first network U.S. airline to begin taking delivery of "next generation" Airbus and Boeing narrowbody aircraft that will further accelerate fuel-efficiency gains. These new deliveries are expected to pave the way for American to have the youngest and most fuel-efficient fleet among its U.S. airline peers in approximately five years.

These firm aircraft commitments are scheduled for delivery as follows: 2013 - 40 aircraft, 2014 - 55 aircraft, 2015 - 50 aircraft, 2016 - 45 aircraft, 2017 - 50 aircraft, 2018 and beyond -120 aircraft. The manufacturers have committed financing to American of \$13 billion through lease transactions, which covers the first 100 Boeing deliveries and first 130 Airbus deliveries.

Further, in July 2011, American entered into a sale-leaseback arrangement with a leasing company to finance up to 35 Boeing 737-800 aircraft scheduled to be delivered in 2011 through 2014. The financing of each aircraft under this arrangement will be subject to certain terms and conditions. As of the end of the third quarter of 2011, American had financed two Boeing 737-800 aircraft under this arrangement.

During the third quarter of 2011, the Company exercised rights to acquire an additional Boeing 777-300 aircraft, scheduled for delivery in 2013, and converted a Boeing 777-200ER aircraft purchase commitment to a 777-300ER aircraft purchase commitment. Including these aircraft, as of the end of the third quarter of 2011, the Company had exercised rights to acquire a total of nine Boeing 777-300ER aircraft.

In connection with the aircraft agreements entered into with Boeing and Airbus during the third quarter of 2011 and the Company's acceleration of its fleet renewal and replacement plan, the Company continues to evaluate the useful lives of certain fleets including MD-80, B757 and B767 aircraft. As of September 30, 2011, the Company is still finalizing its fleet plans and, accordingly, there is no event requiring assessment of whether certain fleets may be impaired. Due to various factors, the Company continues to evaluate its capacity needs in 2012 and beyond based on general economic conditions and fuel cost. Upon finalization of the fleet plan, which is anticipated in the fourth quarter, including potential changes to the useful lives of the aircraft, an analyses of their recoverability may be required. Additionally, to the extent the fleet plan involves grounding leased aircraft, we may record exit costs associated with future lease payments when such aircraft are grounded. If such analyses indicate there is an impairment, a charge could be significant.

#### **AMR Eagle Divestiture**

On August 11, 2011, AMR Eagle filed a Form 10 registration statement (subsequently amended on September 26, 2011 and October 6, 2011) with the Securities and Exchange Commission in connection with a potential spin-off of AMR Eagle.

In a spin-off, AMR would distribute to its shareholders 100 percent of the outstanding shares of AMR Eagle on a pro-rata basis, and AMR would not retain any ownership interest in AMR Eagle. On a historical basis, in 2010, AMR Eagle, through its subsidiaries, generated \$2.3 billion in revenue with more than \$300 million from ground handling services. AMR Eagle would operate the third largest regional airline in the United States as it provides the vast majority of American's regional flight operations. Under a nine-year services agreement with American, AMR Eagle would initially operate 281 aircraft on behalf of American. American could withdraw from AMR Eagle a specified number of jet aircraft up to 40 per year beginning in 2014 and certain turbo prop aircraft each year. The agreement would also include a provision to re-set rates to reflect any change in market levels for regional feed after four years.

AMR Eagle also would operate one of the largest ground handling operations in the U.S. serving American and other passenger airlines at more than 100 airports in the U.S., the Bahamas, the Caribbean and Canada. Under a ground handling agreement, AMR Eagle would have an eight-year term, but provide American the right to re-bid ground handling services at a specified number of airports each year.

While AMR Eagle's wholly-owned subsidiaries, American Eagle Airlines, Inc. ("Eagle") and Executive Airlines, Inc. ("Executive") are expected to continue to operate all of the Jet Aircraft (as defined below), those aircraft and the associated indebtedness will be transferred to American pursuant to the Purchase Agreement referred to below. AMR currently guarantees the indebtedness relating to each Jet Aircraft and AMR will continue to guarantee such indebtedness following American's purchase of each Jet Aircraft subject to such indebtedness. Ownership of the Jet Aircraft by American is intended to provide American control over the regional aircraft that are pivotal to its network and to protect AMR's position as the guarantor of the related indebtedness.

As contemplated by the Form 10, on August 31, 2011, American entered into a Master Purchase Agreement (the "Purchase Agreement") with Eagle and Executive under which Eagle will sell to American 47 CRJ-700 Jet Aircraft and 216 Embraer 135, 140 and 145 Jet Aircraft, including the engines installed on each such aircraft and other related assets (each, a "Jet Aircraft"). In addition, American will purchase from Eagle and Executive certain specified fixed assets, generally consisting of equipment and leasehold improvements owned by Eagle or Executive and used in connection with the regional flight operations conducted by Eagle and Executive on American's behalf and the ground handling operations of Eagle and Executive (collectively, the "Other Assets").

Each Jet Aircraft will be purchased by American on the date of delivery of such aircraft to American, and the Other Assets will be purchased by American ten days after delivery of the last Jet Aircraft to American, subject in each case to the satisfaction of certain conditions. Delivery of the Jet Aircraft began on August 31, 2011, and the last Jet Aircraft is expected to be delivered in October 2011. Following the delivery of each Jet Aircraft, American will lease the Jet Aircraft to Eagle, and Eagle will continue to provide certain regional flight operations to American.

American will take each Jet Aircraft subject to, and Eagle will be released from, all outstanding indebtedness relating to such Jet Aircraft. The indebtedness related to the Jet Aircraft consists of individual notes for each Jet Aircraft. The notes are secured by the related Jet Aircraft and certain other assets, have either fixed or floating interest rates and mature over various periods through 2023. As of September 30, 2011, the fixed rate notes had effective interest rates ranging from 4.25% to 7.50% and the floating rate notes had effective interest rates ranging from 2.247% to 3.128%. The notes include customary terms and conditions, including customary events of default and certain cross-default provisions.

As of the end of the third quarter of 2011, Eagle had delivered 103 Embraer 135, 140 and 145 Jet Aircraft to American. As of the end of the third quarter 2011, the net book value of such transferred Jet Aircraft was \$877 million, and the aggregate outstanding indebtedness (net of discount) associated with such transferred Jet Aircraft was \$712 million. Upon completion of the transfer of the Jet Aircraft from Eagle to American, it is expected that American's aggregate outstanding indebtedness (net of discount) related to the Jet Aircraft will be approximately \$2.1 billion.

In connection with a spin-off, AMR would allocate a portion of its net operating loss ("NOL") carryover balance for U.S. Federal income tax purposes to AMR Eagle. Following distribution, approximately \$800 million of the \$6.7 billion NOL carryover balance would be allocated to AMR Eagle. Further, the fleet transfer from AMR Eagle to American would result in a \$1.1 billion reduction in the Company's NOL carryover balance, which would be effectively restored through future tax depreciation.

The spin-off of Eagle would be subject to certain conditions, including U.S. Securities and Exchange Commission (SEC) clearance, receipt of regulatory approvals, an opinion from tax counsel and a favorable ruling from the Internal Revenue Service regarding the tax-free status of the spin-off to AMR shareholders, execution of inter-company agreements and approval by AMR's board of directors. Stockholder approval of the spin-off is not required. While AMR Corporation has taken this step toward a spin-off of Eagle, it could decide to retain Eagle, or the divestiture of Eagle could take another form, such as a sale.

### **GDS Discussion**

Over the past several years, American has been developing a direct connection technology, designed to distribute its fare content and bookings capability directly to travel agents in order to achieve greater efficiencies, cost savings, and technological advances in the distribution of American's services. Historically, approximately 60% of American's bookings are booked through travel agencies, which typically use one or more global distribution systems ("GDSs") to view fare content from American and other industry participants. American developed its direct connect strategy with the knowledge that certain amendments to its GDS contracts would expire in the summer of 2011. American's support for direct connect technologies, which constitute a competitive threat to the GDSs, has led to litigation between American and two of the GDS owners, Sabre, and Travelport, as well as Orbitz, an online travel agency affiliated with Travelport. American's direct connect efforts have also made negotiations for new GDS agreements more difficult. American has entered into short term extensions with Sabre and Travelport while it continues to negotiate new agreements and pursue antitrust and other claims in both state and federal courts.

As part of its efforts to introduce more direct connect technologies into the distribution of its product, on November 1, 2010, American notified Orbitz that American intended to terminate certain agreements with Orbitz. Shortly thereafter, Travelport sued American in Illinois state court alleging that American's decision to terminate its Orbitz agreements violated certain terms in Travelport's agreement with American. Following an order from an Illinois court denying a Travelport request for preliminary injunctive relief, American terminated its agreements with Orbitz on December 21, 2010. On June 1, 2011, the court reversed its decision, and American re-instated its agreements with Orbitz through September 1, 2011. American appealed that decision, and on September 22, 2011 an appeals court ruled against American and upheld the preliminary injunction. On December 3, 2010, Travelport increased the fees it charges American for some bookings. American has filed breach of contract and other claims against Travelport, and Travelport has filed additional claims against American. As part of its extension agreement with Travelport, American has agreed not to terminate its agreements with Orbitz during the term of this extension.

Sabre is the largest GDS and the largest non-direct source of American's bookings, with over \$7 billion of American's passenger revenues generated from bookings made through the Sabre GDS in 2010. On January 5, 2011, in retaliation for American's public statements in support of direct connect, Sabre "biased" its system by making it more difficult for travel agents to find American's fares in the Sabre GDS and doubled the fees it charges American for bookings through its GDS. On January 10, 2011, American filed breach of contract claims in Texas state court and obtained a court order that temporarily enjoined Sabre from biasing against American's fares, and the parties subsequently entered into a Stand Down Agreement that eliminated the biasing, retracted the price increase, and suspended litigation between the two companies. That agreement expired on June 1, 2011. American subsequently amended its complaint against Sabre by adding new tortious interference claims and new claims under the Texas antitrust laws alleging that Sabre has engaged in anticompetitive practices to preserve its monopoly power, including organizing and monitoring a boycott of American's services among travel agencies. In the same action, Sabre has filed breach of contract claims and antitrust claims against American alleging that American and other airlines have entered into an agreement to withhold information from GDSs. The Texas state court has set a trial date of June 13, 2012. The extension agreement between American and Sabre, which provides for American's continued participation in Sabre, will remain in effect until 14 days after the antitrust claims in the Texas state court are resolved.

In addition to claims pending in Texas and Illinois state courts, American has filed a federal antitrust lawsuit against Travelport, Sabre, and Orbitz. The lawsuit alleges that the defendants have engaged in anticompetitive practices to preserve their monopoly power over American's ability to distribute its products through their subscribers. The lawsuit

further alleges that these actions have prevented American from employing new competing technologies and has allowed the defendants to continue to charge American supracompetitive fees. Also on June 1, 2011, Sabre filed a request to intervene in this action and stated that it intended to file its own claims against American alleging that American violated the antitrust laws by withholding certain content from the Sabre GDS.

While American is negotiating with each GDS to reach new agreements, American cannot predict the outcome of those negotiations. Failure to negotiate extensions of these agreements on reasonable terms with one or all of these GDSs could have a material adverse impact on American. While American believes that some of the bookings through Orbitz, Travelport and Sabre might transition to other distribution channels, such as other travel agencies, metasearch sites and American's AA.com web site, it is not possible at this time to estimate what the ultimate impact would be to American's business if American is unsuccessful in resolving one or more of these matters. If as a result of these matters it becomes more difficult for American's customers to find and book flights on American, American could be put at a competitive disadvantage and this may result in fewer bookings. If American is unable to sell American inventory through any or all of these channels, American's level of bookings, business and results of operations could be materially adversely affected. American intends to vigorously pursue American's claims and defenses in the lawsuits described above, but there can be no assurance of the outcome of any such lawsuit.

#### **Contingencies**

The Company has certain contingencies resulting from litigation and claims incident to the ordinary course of business. Management believes, after considering a number of factors, including (but not limited to) the information currently available, the views of legal counsel, the nature of contingencies to which the Company is subject and prior experience, that the ultimate disposition of the litigation (except as noted in "Legal Proceedings" in Part II, Item 1) and claims will not materially affect the Company's consolidated financial position or results of operations. When appropriate, the Company accrues for these contingencies based on its assessments of the likely outcomes of the related matters. The amounts of these contingencies could increase or decrease in the near term, based on revisions to those assessments.

#### **Financial Highlights**

The Company recorded a consolidated net loss of \$153 million in the third quarter of 2011 compared to net income of \$129 million in the same period last year. The Company's consolidated net loss reflects significant year-over-year increases in fuel prices, partially offset by higher operating revenues. Consolidated passenger revenue increased by \$479 million to \$5.6 billion for the third quarter of 2011 compared to the same period last year. Cargo and other revenues increased by \$55 million to \$820 million for the third quarter of 2011 compared to the same period last year. Mainline passenger unit revenues increased 8.1 percent in the third quarter of 2011 due to a 7.0 percent increase in passenger yield year-over-year. This also reflects an increase in load factor of approximately 0.9 points compared to the third quarter of 2010. The Company's revenue results reflect approximately \$25 million in lower revenue due to Hurricane Irene, which adversely impacted system operations.

The increase in total operating revenue was offset by significantly higher year-over-year fuel prices. Fuel prices continued the trend set in the first six months of 2011, with prices remaining high and extremely volatile. The Company paid an average of \$3.15 per gallon in the third quarter of 2011 compared to an average of \$2.24 per gallon in the third quarter of 2010, including the effects of hedging. As a result, fuel expense, taking into account the impact of fuel hedging, increased \$558 million year-over-year to \$2.0 billion. Hedging gains reduced fuel expense by approximately \$238 million.

The Company's ability to become profitable and its ability to continue to fund its obligations on an ongoing basis will depend on a number of factors, many of which are largely beyond the Company's control. Certain risk factors that affect the Company's business and financial results are discussed in the Risk Factors listed in Item 1A.

In order to remain competitive and to improve its financial condition, the Company must continue to take steps to generate additional revenues and to reduce its costs. Although the Company has a number of initiatives underway to address its cost and revenue challenges, some of these initiatives involve changes to the Company's business which it may be unable to implement. It has become increasingly difficult to identify and implement significant revenue enhancement and cost savings initiatives. The adequacy and ultimate success of the Company's initiatives to generate additional revenues and reduce costs cannot be assured. Moreover, whether the Company's initiatives will be adequate or successful depends in large measure on factors beyond its control, notably the overall industry environment, including passenger demand, yield

and industry capacity growth, and fuel prices. It will be very difficult for the Company to continue to fund its obligations on an ongoing basis, and to return to profitability, if the overall industry revenue environment does not continue to improve, if high fuel prices persist, or if labor costs remain uncompetitive.

#### LIQUIDITY AND CAPITAL RESOURCES

#### **Cash, Short-Term Investments and Restricted Assets**

At September 30, 2011, the Company had \$4.3 billion in unrestricted cash and short-term investments and \$474 million in restricted cash and short-term investments, both at fair value, versus \$4.5 billion in unrestricted cash and short-term investments and \$450 million in restricted cash and short-term investments at December 31, 2010.

The Company's unrestricted short-term investment portfolio consists of a variety of what the Company believes are highly liquid, lower risk instruments including money market funds, government agency investments, repurchase investments, short-term obligations, corporate obligations, bank notes, certificates of deposit and time deposits. The Company's objectives for its investment portfolio are (1) the safety of principal, (2) liquidity maintenance, (3) yield maximization, and (4) the full investment of all available funds. The Company's risk management policy further emphasizes superior credit quality (primarily based on short-term ratings by nationally recognized statistical rating organizations) in selecting and maintaining investments in its portfolio and enforces limits on the proportion of funds invested with one issuer, one industry, or one type of instrument. The Company regularly assesses the market risks of its portfolio, and believes that its established policies and business practices adequately limit those risks. As a result, the Company does not anticipate any material adverse impact from these risks.

#### **Significant Indebtedness and Future Financing**

Indebtedness is a significant risk to the Company as discussed more fully in the Risk Factors included under Item 1A. During the last five years and through September 30, 2011, the Company raised substantial financing to fund operating losses, capital commitments (mainly for aircraft and ground properties), debt maturities, employee pension obligations and to bolster its liquidity. As of the date of this Form 10-Q, the Company believes that it should have sufficient liquidity to fund its operations, including repayment of debt and capital leases, capital expenditures and other contractual obligations; however, there can be no assurances to that effect.

On January 25, 2011, American closed on a \$657 million offering of Class A and Class B Pass Through Trust Certificates, Series 2011-1 (the 2011-1 Certificates). The equipment notes expected to be held by each pass through trust will be issued for each of (a) 15 Boeing 737-823 aircraft delivered new to American from 1999 to 2001, (b) six Boeing 757-223 aircraft delivered new to American in 1999 and 2001, (c) two Boeing 767-323ER aircraft delivered new to American from 1999 to 2000. At closing, 27 of the aircraft were encumbered by either private mortgages or by liens to secure debt incurred in connection with the issuance of enhanced equipment trust certificates in 2001, all of which mature in 2011. As a result, the proceeds from the sale of the 2011-1 Certificates of each trust were initially held in escrow with a depositary, pending the financing of each aircraft under an indenture relating to the 2011-1 Certificates. Interest of 5.25% and 7.00% per annum on the issued and outstanding Series A equipment notes and Series B equipment notes, respectively, will be payable semiannually on January 31 and July 31 of each year, commencing on July 31, 2011, and principal on such equipment notes is scheduled for payment on January 31 and July 31 of certain years, commencing on July 31, 2011. The payment obligations of American under the equipment notes will be fully and unconditionally guaranteed by AMR. Approximately \$598 million of the proceeds from sale of the 2011-1 Certificates had been received by American as of September 30, 2011, in exchange for equipment notes secured by (a) fifteen Boeing 737-823 aircraft, (b) two 757-223 aircraft, (c) two Boeing 767-323 aircraft and (d) seven Boeing 777-223 aircraft. Approximately \$59 million of the proceeds from the sale of the 2011-1 Certificates is expected to be received in the fourth quarter of 2011.

On October 4, 2011, American closed on a \$726 million offering of Class A Pass Through Trust Certificates, Series 2011-2 (the 2011-2 Certificates). Subsequent to the closing, the equipment notes held by each pass through trust were issued for each of (a) 14 Boeing 737-823 aircraft delivered new to American from 1999 to 2001 and 2 Boeing 737-823 aircraft delivered new to American in 2009, (b) 14 Boeing 757-223 aircraft delivered new to American in 1999 and 2001 and (c) 13 Boeing 777-223ER aircraft delivered new to American in 2001. Interest of 8.625% per annum on the issued and outstanding 2011-2 Certificates will be payable semiannually on April 15 and October 15 of each year, commencing on April 15, 2012, and principal on such equipment notes is scheduled for payment on April 15 and October 15 of certain years, commencing on April 15, 2012. The payment obligations of American under the equipment notes are fully and unconditionally guaranteed by AMR. All proceeds from the sale of the Series 2011-2 Certificates have been received by American.

In the remainder of 2011, the Company is contractually required to make approximately \$1.0 billion of principal payments on long-term debt and approximately \$11 million in principal payments on capital leases, and the Company expects to spend approximately \$419 million on capital expenditures, including aircraft commitments. In addition, the fragile economy, rising fuel prices, the possibility of being required to post reserves under credit card processing agreements, and the obligation to post cash collateral on fuel hedging contracts and fund pension plan contributions, among other things, may in the future negatively impact the Company's liquidity. To maintain sufficient liquidity, and because the Company has significant debt, lease and other obligations in the next several years, including commitments to purchase aircraft, as well as significant pension funding obligations, the Company will need access to substantial additional funding. An inability to obtain necessary additional funding on acceptable terms would have a material adverse impact on the Company and on its ability to sustain its operations.

The Company's substantial indebtedness and other obligations have important consequences. For example, they: (i) limit the Company's ability to obtain additional funding for working capital, capital expenditures, acquisitions, investments and general corporate purposes, as well as adversely affect the terms on which such funding could be obtained; (ii) require the Company to dedicate a substantial portion of its cash flow from operations to payments on its indebtedness and other obligations, thereby reducing the funds available for other purposes; (iii) make the Company more vulnerable to economic downturns and catastrophic external events; and (iv) limit the Company's ability to withstand competitive pressures and reduce its flexibility in responding to changing business and economic conditions.

The Company's possible financing sources include refinancing of currently encumbered aircraft as the debt against them is retired, the issuance of debt secured by, or sale leaseback transactions involving, newly acquired aircraft; the issuance of debt secured by other assets; the sale or monetization of certain assets; the issuance of unsecured debt; and the issuance of equity or equity-like securities. Almost all of the Company's aircraft assets (including aircraft eligible for the benefits of Section 1110 of the U.S. Bankruptcy Code) are encumbered, and the Company has a very limited quantity of assets which could be used as collateral in future financing; however, borrowing capacity will become available as aircraft become un-encumbered. The market value of the Company's aircraft assets has declined in recent years, and may continue to decline. Some of the Company's assets may be difficult to finance, and the availability and level of the financing sources described above cannot be assured.

The Company has financing commitments covering all of the aircraft scheduled to be delivered between 2011 and 2016, except 15 widebody aircraft that it intends to finance at a later time.

See "Notes to Consolidated Financial Statements: Note 2" for a schedule of the Company's aircraft commitments and payments.

#### **Credit Ratings**

AMR's and American's credit ratings are significantly below investment grade. Additional reductions in AMR's or American's credit ratings could further increase the Company's borrowing or other costs and further restrict the availability of future financing.

#### **Credit Card Processing and Other Reserves**

American has agreements with a number of credit card companies and processors to accept credit cards for the sale of air travel and other services. Under certain of these agreements, the related credit card processor may hold back a reserve from American's credit card receivables following the occurrence of certain events, including the failure of American to maintain certain levels of liquidity (as specified in each agreement).

Under such agreements, the amount of the reserve that may be required generally is based on the processor's exposure to the Company under the applicable agreement and, in the case a reserve is required because of American's failure to maintain a certain level of liquidity, the amount of such liquidity. As of September 30, 2011, the Company was not required to maintain any reserve under such agreements. If circumstances were to occur that would allow the credit card processor to require the Company to maintain a reserve, the Company's liquidity would be negatively impacted.

#### **Pension Funding Obligation**

The Company is required to make minimum contributions to its defined benefit pension plans under the minimum funding requirements of the Employee Retirement Income Security Act (ERISA), the Pension Funding Equity Act of 2004, the Pension Protection Act of 2006, and the Pension Relief Act of 2010 (Relief Act). Under the Relief Act, the Company estimates its 2011 minimum required contribution to its defined benefit pension plans to be approximately \$520 million, which has been completed as of the date of this filing. The Company estimates its 2012 minimum required contribution to its defined benefit pension plans will not exceed \$560 million. This estimate is subject to change based on final plan asset values as of December 31, 2011.

#### **Cash Flow Activity**

At September 30, 2011, the Company had \$4.3 billion in unrestricted cash and short-term investments, which is a decrease of \$201 million from the balance as of December 31, 2010. Net cash provided by operating activities in the nine-month period ended September 30, 2011 was \$235 million, as compared to \$953 million over the same period in 2010, and which primarily reflects an increase in the Company's consolidated net loss.

The Company made scheduled debt and capital lease payments of \$1.2 billion and invested \$1.1 billion in capital expenditures in the first nine months of 2011. Capital expenditures primarily consisted of new aircraft and certain aircraft modifications.

Under certain of the Company's derivative contracts, the related counterparties are currently required to deposit collateral with the Company due to the value of the contracts. As of September 30, 2011, the cash collateral held by the Company from such counterparties was \$28 million as compared to \$73 million held by such counterparties as of December 31, 2010. Cash held from counterparties as of September 30, 2011 is included in short-term investments. As a result of movements in fuel prices, the cash collateral amounts held by the Company or the counterparties to such contracts, as the case may be, can vary significantly.

In the past, the Company has from time to time refinanced, redeemed or repurchased its debt and taken other steps to reduce its debt or lease obligations or otherwise improve its balance sheet. Going forward, depending on market conditions, its cash positions and other considerations, the Company may continue to take such actions.

Certain of the Company's debt financing agreements contain loan to value ratio covenants and require the Company to periodically appraise the collateral. Pursuant to such agreements, if the loan to value ratio exceeds a specified threshold, the Company may be required to subject additional qualifying collateral (which in some cases may include cash collateral) or, in the alternative, to pay down such financing, in whole or in part, with premium (if any).

#### **War-Risk Insurance**

The U.S. government has agreed to provide commercial war-risk insurance for U.S. based airlines through September 30, 2012, covering losses to employees, passengers, third parties and aircraft. If the U.S. government were to cease providing such insurance in whole or in part, it is likely that the Company could obtain comparable coverage in the commercial market, but the Company would likely incur higher premiums and more restrictive terms, if such coverage is available at all. If the Company is unable to obtain adequate war-risk coverage at commercially reasonable rates, the Company would be adversely affected.

#### RESULTS OF OPERATIONS

#### For the Nine Months Ended September 30, 2011 and 2010

#### **REVENUES**

The Company's revenues increased approximately \$1.4 billion, or 8.7 percent, to \$18.0 billion in the first nine months of 2011 from the same period last year. American's passenger revenues increased by 7.5 percent, or \$943 million, on a 1.6 percent decrease in capacity (available seat mile) (ASM). American's passenger load factor decreased 0.1 points while passenger yield increased by 5.9 percent to 14.09 cents. This resulted in an increase in passenger revenue per available seat mile (RASM) of 5.8 percent to 11.54 cents. American derived approximately 60 percent of its passenger revenues from domestic operations and approximately 40 percent from international operations (flights serving international destinations). Following is additional information regarding American's domestic and international RASM and capacity:

	Nii	Nine Months Ended September 30, 2011			
	RASM (cents)	Y-O-Y Change	ASMs (billions)	Y-O-Y Change	
DOT Domestic	11.47	6.3%	69.4	(0.9)%	
International	11.66	5.1	47.6	5.5	
DOT Latin America	13.15	13.7	22.5	2.7	
DOT Atlantic	10.66	(0.9)	18.3	3.2	
DOT Pacific	9.34	(8.8)	6.7	24.0	

The Company's Regional Affiliates include two wholly owned subsidiaries, American Eagle Airlines, Inc. and Executive Airlines, Inc. (collectively, AMR Eagle), and an independent carrier with which American has a capacity purchase agreement, Chautauqua Airlines, Inc. (Chautauqua).

Regional Affiliates' passenger revenues, which are based on industry standard proration agreements for flights connecting to American flights, increased \$308 million, or 17.9 percent, to \$2.0 billion as a result of higher yield and increased traffic . Regional Affiliates' traffic increased 14.1 percent to 7.4 billion revenue passenger miles (RPMs), on a capacity increase of 12.3 percent to 10.1 billion ASMs, resulting in a 1.2 point increase in passenger load factor to 73.1 percent.

Cargo revenues increased 8.1 percent, or \$40 million, to \$532 million primarily as a result of increased freight yields.

Other revenues increased 8.2 percent, or \$147 million, to \$1.9 billion primarily due to increased revenue associated with the sale of mileage credits in the AAdvantage frequent flyer program and increases in certain passenger service charge volumes and fees.

#### **OPERATING EXPENSES**

The Company's total operating expenses increased 11.78 percent, or \$1.9 billion, to \$18.4 billion in the first nine months ended September 30, 2011 compared to the same period in 2010. The Company's operating expenses per ASM increased 9.3 percent to 13.73 cents. The increase in operating expense was largely due to a year-over-year increase in fuel prices in the first nine months of 2011 compared to the same period in 2010. Fuel expense was the Company's largest single expense category in the first nine months of 2011 and the price increase resulted in \$1.4 billion in incremental year-over-year fuel expense in the first nine months of 2011 (based on the year-over-year increase in the average price per gallon multiplied by gallons consumed, inclusive of the impact of fuel hedging). Further increases in fuel prices and/or disruptions in the supply of fuel would further materially adversely affect the Company's financial condition and results of operations. The remaining increase in operating expense was primarily due to revenue related expenses, such as credit card fees and booking fees and commissions, and increased aircraft rent related to the Company's fleet renewal plan.

(in millions) Operating Expenses	Nine Mon Ended September 2011	Change	Percentage Change
Aircraft fuel	\$ 5,5	\$1,330	31.2%(a)
Wages, salaries and benefits	4,7	64 81	1.7
Regional payments to AMR Eagle	1,9	31 286	17.4(b)
Other rentals and landing fees	g	12	1.3
Maintenance, materials and repairs	7	'81 (41)	(5.0)(c)
Commissions, booking fees and credit card expense	8	309 70	9.5(d)
Depreciation and amortization	$\epsilon$	689 (7)	(1.0)
Aircraft rentals	2	92 62	14.4(e)
Food service	3	90 25	6.8(f)
Other operating expenses	1,9	120	6.5
Total operating expenses	\$ 18,3	\$1,938	11.8%

- (a) Aircraft fuel expense increased primarily due to a 32.1 percent increase in the Company's price per gallon of fuel (net of the impact of hedging gains of \$238 million).
- (b) Regional payments to AMR Eagle expense increased in conjunction with an increase in fuel expense.
- (c) Maintenance, materials and repairs decreased primarily due to timing of repairs in 2010.
- (d) Commissions, booking fees and credit card expenses increased due to an 8.7 percent increase in operating revenues.
- (e) Aircraft rental expense increased primarily due to new aircraft deliveries in 2011 and 2010.
- (f) Food service expense increased primarily due to increased international flying.

#### **OTHER INCOME (EXPENSE)**

Other income (expense) consists of interest income and expense, interest capitalized and miscellaneous—net.

An increase in short-term investment balances caused an increase in interest income of \$0.8 million, or 4.3 percent, to \$20 million for the first nine months of 2011 compared to the same period last year. Interest expense increased \$7 million, or 1.4 percent, to \$502 million primarily as a result of increases in the Company's long-term debt balance.

#### INCOME TAX

The Company did not record a net tax provision (benefit) associated with its net loss for the nine months ended September 30, 2011 or September 30, 2010 due to the Company providing a valuation allowance, as discussed in Note 4 to the condensed consolidated financial statements.

#### **REGIONAL AFFILIATES**

The following table summarizes the combined capacity purchase activity for the American Connection carriers and AMR Eagle for the nine months ended September 30, 2011 and 2010.

	 Nine Months Ended September 30,			
	 2011	2010		
Revenues:				
Regional Affiliates	\$ 2,023	\$	1,716	
Other	125		112	
	\$ 2,148	\$	1,828	
Expenses:				
Regional payments	\$ 1,987	\$	1,692	
Other incurred expenses	 327		275	
	\$ 2,314	\$	1,967	

In addition, passengers connecting to American's flights from American Connection and AMR Eagle flights generated passenger revenues for American flights of \$1.4 billion and \$1.3 billion in the first nine months of 2011 and 2010, respectively, which are included in Revenues – Passenger in the consolidated statements of operations.

#### Outlook

The Company currently expects capacity for American's mainline jet operations to decrease by approximately 3.0% in the fourth quarter of 2011 versus the fourth quarter of 2010. American's mainline capacity for the full year 2011 is expected to increase approximately 0.4% from 2010.

The Company expects fourth quarter 2011 mainline unit costs to increase between 11.7% and 12.1%. The fourth quarter 2011 unit cost expectations reflect projected fuel prices.

The Company's results are significantly affected by the price of jet fuel. Fuel prices increased dramatically during the first nine months of 2011 and remain high and extremely volatile. Based on the Company's current forecast of full year 2011 jet fuel prices, the Company estimates that its full year 2011 jet fuel cost per gallon, taking hedging into account, will increase by approximately 30% over 2010. The Company's hedging approach has been and continues to be systematic and as of September 2011, the Company had cash flow hedges, including heating oil, jet fuel, and WTI crude option and collar contracts, covering approximately 52 percent of its estimated 2011 fuel requirements.

The Company has also implemented a number of initiatives in an effort to offset this anticipated increase in fuel prices, including fare hikes and reductions in the Company's previously announced capacity growth. However, intense competition and other factors may limit the Company's ability to increase fares. Further, the catastrophic events in Japan in the first quarter and the ongoing GDS related dispute could have an adverse affect on the Company in future periods. In addition, continued increases in fuel prices may depress overall economic activity, which in turn could impact demand for air travel. Accordingly, while the Company expects that its operating results, cash flow and liquidity for the remainder of 2011 will continue to be materially and adversely impacted by high fuel prices, the magnitude of that adverse impact is subject to considerable uncertainty, as well as to numerous factors beyond the Company's control.

#### Item 3. Quantitative and Qualitative Disclosures about Market Risk

There have been no material changes in market risk from the information provided in <u>Item 7A. Quantitative and Qualitative Disclosures About Market Risk</u> of the Company's 2010 Form 10-K. The change in market risk for aircraft fuel is discussed below for informational purposes.

The risk inherent in the Company's market risk sensitive instruments and positions is the potential loss arising from adverse changes in the price of fuel, foreign currency exchange rates and interest rates as discussed below. The sensitivity analyses presented do not consider the effects that such adverse changes may have on overall economic activity, nor do they consider additional actions management may take to mitigate the Company's exposure to such changes. Therefore, actual results may differ. The Company does not hold or issue derivative financial instruments for trading purposes.

Aircraft Fuel The Company's earnings are substantially affected by changes in the price and availability of aircraft fuel. In order to provide a measure of control over price and supply, the Company trades and ships fuel and maintains fuel storage facilities to support its flight operations. The Company also manages the price risk of fuel costs primarily by using jet fuel, heating oil, and WTI crude hedging contracts. Market risk is estimated as a hypothetical 10 percent increase in the September 30, 2011 cost per gallon of fuel. Based on projected fuel usage for the next twelve months, such an increase would result in an increase to Aircraft fuel expense of approximately \$601 million, inclusive of the impact of effective fuel hedge instruments outstanding at September 30, 2011, and assumes the Company's fuel hedging program remains effective. Such an increase would have resulted in an increase to projected Aircraft fuel expense of approximately \$445 million in the twelve months ended December 31, 2010, inclusive of the impact of fuel hedge instruments outstanding at December 31, 2009. As of September 2011, the Company had cash flow hedges, with collars and options, covering approximately 52 percent of its estimated remaining 2011 fuel requirements. Comparatively, as of September 30, 2010, the Company had hedged, with collars and options, approximately 40 percent of its estimated remaining 2010 fuel requirements. The consumption hedged for the remainder of 2011 by cash flow hedges is capped at an average price of approximately \$3.01 per gallon of jet fuel, and the Company's collars have an average floor price of approximately \$2.23 per gallon of jet fuel (both the capped and floor price exclude taxes and transportation costs). The Company's collars represent approximately 41 percent of its estimated remaining 2011 fuel requirements. A deterioration of the Company's liquidity and financial position could negatively affect the Company's ability to hedge fuel in the future.

Ineffectiveness is inherent in hedging jet fuel with derivative positions based in crude oil or other crude oil related commodities. The Company assesses, both at the inception of each hedge and on an ongoing basis, whether the derivatives that are used in its hedging transactions are highly effective in offsetting changes in cash flows of the hedged items. In doing so, the Company uses a regression model to determine the correlation of the change in prices of the commodities used to hedge jet fuel (e.g., NYMEX Heating oil) to the change in the price of jet fuel. The Company also monitors the actual dollar offset of the hedges' market values as compared to hypothetical jet fuel hedges. The fuel hedge contracts are generally deemed to be "highly effective" if the R-squared is greater than 80 percent and the dollar offset correlation is within 80 percent to 125 percent. The Company discontinues hedge accounting prospectively if it determines that a derivative is no longer expected to be highly effective as a hedge or if it decides to discontinue the hedging relationship.

#### **Item 4. Controls and Procedures**

The term "disclosure controls and procedures" is defined in Rules 13a-15(e) and 15d-15(e) of the Securities Exchange Act of 1934, or the Exchange Act. This term refers to the controls and procedures of a company that are designed to ensure that information required to be disclosed by a company in the reports that it files under the Exchange Act is recorded, processed, summarized and reported within the time periods specified by the Securities and Exchange Commission. An evaluation was performed under the supervision and with the participation of the Company's management, including the Chief Executive Officer (CEO) and Chief Financial Officer (CFO), of the effectiveness of the Company's disclosure controls and procedures as of December 31, 2010. Based on that evaluation, the Company's management, including the CEO and CFO, concluded that the Company's disclosure controls and procedures were effective as of September 30, 2011. During the quarter ending on September 30, 2011, there was no change in the Company's internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

#### PART II: OTHER INFORMATION

#### **Item 1. Legal Proceedings**

On February 14, 2006, the Antitrust Division of the United States Department of Justice (DOJ) served the Company with a grand jury subpoena as part of an ongoing investigation into possible criminal violations of the antitrust laws by certain domestic and foreign air cargo carriers. At this time, the Company does not believe it is a target of the DOJ investigation. The New Zealand Commerce Commission notified the Company on February 17, 2006 that it is investigating whether the Company and certain other cargo carriers entered into agreements relating to fuel surcharges, security surcharges, war-risk surcharges, and customs clearance surcharges. On February 22, 2006, the Company received a letter from the Swiss Competition Commission informing the Company that it is investigating whether the Company and certain other cargo carriers entered into agreements relating to fuel surcharges, security surcharges, war-risk surcharges, and customs clearance surcharges. On March 11, 2008, the Company received a request for information from the Swiss Competition Commission concerning, among other things, the scope and organization of the Company's activities in Switzerland. On June 27, 2007 and October 31, 2007, the Company received requests for information from the Australian Competition and Consumer Commission seeking information regarding fuel surcharges imposed by the Company on cargo shipments to and from Australia and regarding the structure of the Company's cargo operations. On September 1, 2008, the Company received a request from the Korea Fair Trade Commission seeking information regarding cargo rates and surcharges and the structure of the Company's activities in Korea. On January 23, 2007, the Brazilian competition authorities, as part of an ongoing investigation, conducted an unannounced search of the Company's cargo facilities in Sao Paulo, Brazil. On April 24, 2008, the Brazilian competition authorities charged the Company with violating Brazilian competition laws. On December 31, 2009, the Brazilian competition authorities made a non-binding recommendation to the Brazilian competition tribunal that it find the Company in violation of competition laws. The authorities are investigating whether the Company and certain other foreign and domestic air carriers violated Brazilian competition laws by illegally conspiring to set fuel surcharges on cargo shipments. The Company is vigorously contesting the allegations and the preliminary findings of the Brazilian competition authorities. On December 19, 2006 and June 12, 2007, the Company received requests for information from the European Commission seeking information regarding the Company's corporate structure, and revenue and pricing announcements for air cargo shipments to and from the European Union. On December 18, 2007, the European Commission issued a Statement of Objection (SO) against 26 airlines, including the Company. The SO alleges that these carriers participated in a conspiracy to set surcharges on cargo shipments in violation of EU law. On November 12, 2010, the EU Commission notified the Company that it was closing its proceedings against the Company without imposing any fine or finding any wrongdoing. The Company intends to cooperate fully with all pending investigations. In the event that any investigations uncover violations of the U.S. antitrust laws or the competition laws of some other jurisdiction, or if the Company were named and found liable in any litigation based on these allegations, such findings and related legal proceedings could have a material adverse impact on the Company. Forty-five purported class action lawsuits have been filed in the U.S. against the Company and certain foreign and domestic air carriers alleging that the defendants violated U.S. antitrust laws by illegally conspiring to set prices and surcharges on cargo shipments. These cases, along with other purported class action lawsuits in which the Company was not named, were consolidated in the United States District Court for the Eastern District of New York as In re Air Cargo Shipping Services Antitrust Litigation, 06-MD-1775 on June 20, 2006. Plaintiffs are seeking trebled money damages and injunctive relief. To facilitate a settlement on a class basis, the company agreed to be named in a separate class action complaint, which was filed on July 26, 2010. The settlement of that complaint, in which the company does not admit and denies liability, was approved by the court and final judgment was entered on April 6, 2011. Approximately 40 members of the class have elected to opt out, thereby preserving their rights to sue the Company separately. Any adverse judgment could have a material adverse impact on the Company. Also, on January 23, 2007, the Company was served with a purported class action complaint filed against the Company, American, and certain foreign and domestic air carriers in the Supreme Court of British Columbia in Canada (McKay v. Ace Aviation Holdings, et al.). The plaintiff alleges that the defendants violated Canadian competition laws by illegally conspiring to set prices and surcharges on cargo shipments. The complaint seeks compensatory and punitive damages under Canadian law. On June 22, 2007, the plaintiffs agreed to dismiss their claims against the Company. The dismissal is without prejudice and the Company could be brought back into the litigation at a future date. If litigation is recommenced against the Company in the Canadian courts, the Company will vigorously defend itself; however, any adverse judgment could have a material adverse impact on the Company.

On June 20, 2006, DOJ served the Company with a grand jury subpoena as part of an ongoing investigation into possible criminal violations of the antitrust laws by certain domestic and foreign passenger carriers. At this time, the Company does not believe it is a target of the DOJ investigation. The Company intends to cooperate fully with this investigation. On September 4, 2007, the Attorney General of the State of Florida served the Company with a Civil Investigative Demand as part of its investigation of possible violations of federal and Florida antitrust laws regarding the pricing of air passenger transportation. In the event that this or other investigations uncover violations of the U.S. antitrust laws or the competition laws of some other jurisdiction, such findings and related legal proceedings could have a material adverse impact on the Company. Approximately 52 purported class action lawsuits have been filed in the U.S. against the Company and certain foreign and domestic air carriers alleging that the defendants violated U.S. antitrust laws by illegally conspiring to set prices and surcharges for passenger transportation. On October 25, 2006, these cases, along with other purported class action lawsuits in which the Company was not named, were consolidated in the United States District Court for the Northern District of California as In re International Air Transportation Surcharge Antitrust Litigation, Civ. No. 06-1793 (the Passenger MDL). On July 9, 2007, the Company was named as a defendant in the Passenger MDL. On August 25, 2008, the plaintiffs dismissed their claims against the Company in this action. On March 13, 2008, and March 14, 2008, an additional purported class action complaint, Turner v. American Airlines, et al., Civ. No. 08-1444 (N.D. Cal.), was filed against the Company, alleging that the Company violated U.S. antitrust laws by illegally conspiring to set prices and surcharges for passenger transportation in Japan and certain European countries, respectively. The Turner plaintiffs h

On August 21, 2006, a patent infringement lawsuit was filed against American and American Beacon Advisors, Inc. (then a wholly-owned subsidiary of the Company) in the United States District Court for the Eastern District of Texas (Ronald A. Katz Technology Licensing, L.P. v. American Airlines, Inc., et al.). This case has been consolidated in the Central District of California for pre-trial purposes with numerous other cases brought by the plaintiff against other defendants. The plaintiff alleges that American infringes a number of the plaintiff's patents, each of which relates to automated telephone call processing systems. The plaintiff is seeking past and future royalties, injunctive relief, costs and attorneys' fees. On December 1, 2008, the court dismissed with prejudice all claims against American Beacon. On May 22, 2009, following its granting of summary judgment to American based on invalidity and non-infringement, the court dismissed all claims against American. Plaintiff appealed, and on February 18, 2011, the Federal Circuit Court of Appeals issued a decision affirming in part and reversing in part and remanding the case back to the District Court for further proceedings. Plaintiff's petition for a rehearing of the appeal en banc before the Federal Circuit was denied. Although the Company believes that the plaintiff's claims are without merit and is vigorously defending the lawsuit, a final adverse court decision awarding substantial money damages or placing material restrictions on existing automated telephone call system operations would have a material adverse impact on the Company.

On January 10, 2011, the Company filed a lawsuit in Tarrant County, Texas State Court against Sabre alleging, among other claims, that Sabre's actions of introducing bias against the display of American's services in its global distribution system (GDS) and substantially increasing the rates that it would charge the Company for bookings made through the Sabre GDS breached its agreement with the Company. That same day, the Company successfully obtained a temporary restraining order that prohibited Sabre from continuing to bias the display of American's services. On January 23, 2011, the Company and Sabre entered into a Stand-Down Agreement, pursuant to which American agreed to suspend the litigation against Sabre, and Sabre agreed not to reintroduce biasing against American's services in its GDS and to return to the pricing in effect on January 4, 2011. The parties further agreed to enter into good faith negotiations. The Stand-Down Agreement expired on June 1, 2011. On July 8, 2011, the Company filed new breach of contract and Texas antitrust claims in this action. On June 8, 2011 and October 7, 2011, Sabre filed counterclaims against the Company alleging that American has breached its agreement and that American violated antitrust laws. On August 29, 2011, the Company entered into an agreement with Sabre that will allow American to continue to participate in the Sabre GDS until American's antitrust claims in the Texas state court are resolved. Trial in that case is now set to begin June 13, 2012. The Company intends to vigorously pursue its claims, but there can be no assurance of the outcome, and if the Court does not further enjoin Sabre from introducing bias against American's services or allowing Sabre to remove American services from its system, actions taken by Sabre could have a material adverse effect on the Company.

On April 12, 2011, the Company filed an antitrust lawsuit against Travelport and Orbitz in Federal District Court for the Northern District of Texas. On June 1, 2011, the Company amended its lawsuit to add Sabre as a defendant. The lawsuit alleges that the defendants have engaged in anticompetitive practices to preserve their monopoly power over American's

ability to distribute its products through their subscribers. The lawsuit further alleges that these actions have prevented American from employing new competing technologies and has allowed the defendants to continue to charge American supracompetitive fees. The lawsuit seeks both injunctive relief and money damages. On June 1, 2011, Sabre filed a request to intervene in this action and stated that it intended to file its own claims against American alleging that American violated the antitrust laws by withholding certain content from the Sabre GDS. In addition, all defendants have filed motions requesting that the court dismiss American's claims.

American intends to vigorously pursue these claims, but there can be no assurance of the outcome, and if the Court does not enjoin Sabre or other defendants from taking actions against American, including removing American's services from their systems, actions taken by the defendants could have a material adverse impact on the Company.

#### **Item 1A. Risk Factors**

Our ability to become profitable and our ability to continue to fund our obligations on an ongoing basis will depend on a number of risk factors, many of which are largely beyond our control.

#### As a result of significant losses in recent years, our financial condition has been materially weakened.

We incurred significant losses in recent years, which has materially weakened our financial condition. We lost \$892 million in 2005, \$821 million in 2004, \$1.3 billion in 2003, \$3.5 billion in 2002 and \$1.6 billion in 2001. Although we earned a profit of \$356 million in 2007 and \$164 million in 2006, we lost \$2.5 billion in 2008 (which included a \$1.0 billion impairment charge), and, primarily as a result of very weak demand for air travel driven by the severe downturn in the global economy, we lost \$1.5 billion in 2009 and \$469 million in 2010. In addition, we lost a total of \$868 million in the first three quarters of 2011 (including a \$31 million non-cash charge). Because of our weakened financial condition, we are vulnerable both to the impact of unexpected events (such as terrorist attacks) and to deterioration of the operating environment (such as a significant increase in jet fuel prices or significant increased competition).

The severe global economic downturn resulted in very weak demand for air travel and lower investment asset returns, which has had and could continue to have a significant negative impact on us.

Although demand for air travel has improved as the global economy continues to recover from the recent severe downturn, demand continues to be weak by historical standards. In response to weak demand, we have implemented a number of capacity reductions since late 2008, and we have announced an additional capacity reduction to be implemented in the fourth quarter of 2011. In connection with the capacity reductions we have implemented, we have incurred special charges related to aircraft, employee reductions and certain other charges. Demand for air travel may weaken if the global economy does not continue to recover. No assurance can be given that capacity adjustments or other steps we may take in response to changes in demand will be successful. Capacity reductions or other steps might result in additional special charges in the future. Further, other carriers may make capacity adjustments which may reduce the expected benefits of any steps we may take to respond to changes in demand. Industry-wide capacity may increase to the extent the economy continues to recover from the global recession. If industry capacity increases, and if consumer demand does not continue to pace those increases, we, and the airline industry as a whole, could be negatively impacted.

The economic downturn has resulted in broadly lower investment asset returns and values. Our pension assets suffered a material decrease in value in 2008 related to broader stock market declines, which resulted in higher pension expense in 2009 and 2010 and will result in higher pension expense and higher required contributions in future years. In addition, under certain circumstances, we may be required to maintain cash reserves under our credit card processing agreements and to post cash collateral on fuel hedging contracts. These issues individually or collectively may have a material adverse impact on our liquidity. Also, disruptions in the capital markets and other sources of funding may make it impossible for us to obtain necessary additional funding or make the cost of that funding prohibitive.

We face numerous challenges as we seek to maintain sufficient liquidity, and we will need to raise substantial additional funds. We may not be able to raise those funds, or to do so on acceptable terms.

In the next several years, we have significant debt, lease and other obligations, including significant pension funding obligations. We also expect to make substantial aircraft purchase and other capital expenditures during that time. For example, as of September 30, 2011, we were required to make approximately \$2.6 billion of principal payments on long-

term debt and capital leases during the second half of 2011 and during 2012. In addition, in 2011, we are required to contribute approximately \$520 million to our pension plans, of which \$419 million has been contributed to date. Based on current assumptions and market conditions, we expect to be required to contribute up to \$560 million to our pension plans in 2012. Moreover, the global economic downturn, rising fuel prices, the potential obligation to post reserves under credit card processing agreements and the potential obligation to post cash collateral on fuel hedging contracts, among other things, have negatively impacted, and may in the future negatively impact, our liquidity. To meet our commitments and to maintain sufficient liquidity as we continue to implement our revenue enhancement and cost reduction initiatives, we will need continued access to substantial additional funding.

Our ability to obtain future financing is limited by the value of our unencumbered assets. As a result of financing activity in recent years, almost all of our aircraft assets (including aircraft eligible for the benefits of Section 1110) are encumbered, and we have a very limited quantity of assets which could be used as collateral in future financing. Also, the market value of our aircraft assets has declined in recent years, and may continue to decline. In addition, many of the other financing sources traditionally available to us may be difficult to access, and no assurance can be given as to the amount of financing available to us.

Since the terrorist attacks of September 11, 2001 (the "*Terrorist Attacks*"), our credit ratings have been lowered to significantly below investment grade. These reductions have increased our borrowing costs and otherwise adversely affected borrowing terms, and limited borrowing options. Additional reductions in our credit ratings might have other effects on us, such as further increasing borrowing, lease rental or other costs or further restricting our ability to raise funds.

A number of other factors, including our financial results in recent years, our substantial indebtedness, the difficult revenue environment we face, our reduced credit ratings, recent historically high fuel prices, and the financial difficulties experienced in the airline industry, adversely affect the availability and terms of funding for us. In addition, the global economic downturn resulted in greater volatility, less liquidity, widening of credit spreads, and substantially more limited availability of funding. As a result of these and other factors, although we believe we have or can access sufficient liquidity to fund our operations and obligations, there can be no assurances to that effect. An inability to obtain necessary additional funding on acceptable terms would have a material adverse impact on us and on our ability to sustain our operations.

#### We could be required to maintain reserves under our credit card processing agreements, which could materially adversely impact our liquidity.

American has agreements with a number of credit card companies and processors to accept credit cards for the sale of air travel and other services. Under certain of these agreements, the related credit card processor may hold back a reserve from American's credit card receivables following the occurrence of certain events, including the failure of American to maintain certain levels of liquidity (as specified in each agreement).

Under such agreements, the amount of the reserve that may be required generally is based on the processor's exposure to the Company under the applicable agreement and, in the case a reserve is required because of American's failure to maintain a certain level of liquidity, the amount of such liquidity. As of September 30, 2011, the Company was not required to maintain any reserve under such agreements. If circumstances were to occur that would allow the credit card processor to require the Company to maintain a reserve, the Company's liquidity would be negatively impacted.

#### Our initiatives to generate additional revenues and to reduce our costs may not be adequate or successful.

As we seek to improve our financial condition, we must continue to take steps to generate additional revenues and to reduce our costs. Although we have a number of initiatives underway to address our cost and revenue challenges, some of these initiatives involve changes to our business which we may be unable to implement. In addition, it has become increasingly difficult to identify and implement significant revenue enhancement and cost savings initiatives. The adequacy and ultimate success of our initiatives to generate additional revenues and reduce our costs cannot be assured. Moreover, whether our initiatives will be adequate or successful depends in large measure on factors beyond our control, notably the overall industry environment, including passenger demand, yield and industry capacity growth, and fuel prices. It will be very difficult for us to continue to fund our obligations on an ongoing basis, and to return to profitability, if the overall industry revenue environment does not continue to improve or if fuel prices were to increase and persist for an extended period at high levels.

#### We may be adversely affected by increases in fuel prices, and we would be adversely affected by disruptions in the supply of fuel.

Our results are very significantly affected by the cost, price volatility and the availability of jet fuel, which are in turn affected by a number of factors beyond our control. Due to the competitive nature of the airline industry, we may not be able to pass on increased fuel prices to customers by increasing fares. Although we had some success in raising fares and imposing fuel surcharges in reaction to high fuel prices, these fare increases and surcharges did not keep pace with the extraordinary increases in the price of fuel that occurred in 2007 and 2008. Although fuel prices have abated considerably from the record high prices recorded in July 2008, they remain high and extremely volatile by historical standards. Furthermore, reduced demand or increased fare competition, or both, and resulting lower revenues may offset any potential benefit of any reductions in fuel prices.

While we do not currently anticipate a significant reduction in fuel availability, dependence on foreign imports of crude oil, limited refining capacity and the possibility of changes in government policy on jet fuel production, transportation and marketing make it impossible to predict the future availability of jet fuel. If there are additional outbreaks of hostilities or other conflicts in oil producing areas or elsewhere, or a reduction in refining capacity (due to natural disasters or weather events, for example), or governmental limits on the production or sale of jet fuel (including as a consequence of increased environmental regulation), there could be a reduction in the supply of jet fuel and significant increases in the cost of jet fuel. Major reductions in the availability of jet fuel or significant increases in its cost would have a material adverse impact on us.

We have a large number of older aircraft in our fleet, and these aircraft are not as fuel efficient as more recent models of aircraft. We believe it is imperative that we continue to execute our fleet renewal plans. However, there will be significant delays in the deliveries of the Boeing 787-9 aircraft we have on order, and there could be delays in the deliveries of other new aircraft we have on order.

Our aviation fuel purchase contracts generally do not provide meaningful price protection. While we seek to manage the risk of fuel price increases by using derivative contracts, there can be no assurance that, at any given time, we will have derivatives in place to provide any particular level of protection against increased fuel costs. In addition, a deterioration of our financial position could negatively affect our ability to enter into derivative contracts in the future. Moreover, declines in fuel prices below the levels established in derivative contracts may require us to post material amounts of cash collateral to secure the loss positions on such contracts, and if such contracts close when fuel prices are below the applicable levels, we would be required to make payments to close such contracts; these payments would be treated as additional fuel expense.

# We could be materially adversely affected if we are unable to resolve favorably our pending litigation with certain GDSs and business discussions with certain on-line travel agents.

We are currently involved in litigation with certain GDSs and in business discussions with certain on-line travel agents. An adverse outcome in any of these matters could have a material adverse effect on our level of bookings, business and results of operations. See "Management's Discussion and Analysis of Financial Condition and Results of Operations—Recent Events—GDS Discussion." In addition, agreements with the GDSs operated by Sabre, Travelport and Amadeus have been extended beyond 2011. We could be adversely affected if we are unable to renegotiate acceptable new contractual terms for our participation in these systems.

#### Our indebtedness and other obligations are substantial and could adversely affect our business and liquidity.

We have and will continue to have significant amounts of indebtedness, obligations to make future payments on aircraft equipment and property leases, and obligations under aircraft purchase agreements, as well as a high proportion of debt to equity capital. We expect to incur substantial additional debt (including secured debt) and lease obligations in the future. We also have substantial pension funding obligations. Our substantial indebtedness and other obligations have important consequences. For example, they:

- limit our ability to obtain additional funding for working capital, capital expenditures, acquisitions, investments and general corporate purposes, and adversely affect the terms on which such funding can be obtained;
- require us to dedicate a substantial portion of our cash flow from operations to payments on our indebtedness and other obligations, thereby reducing the funds available for other purposes;
- · make us more vulnerable to economic downturns and catastrophic external events; and
- · limit our ability to withstand competitive pressures and reduce our flexibility in responding to changing business and economic conditions.

# Our business is affected by many changing economic and other conditions beyond our control, and our results of operations tend to be volatile and fluctuate due to seasonality.

Our business and our results of operations are affected by many changing economic and other conditions beyond our control, including, among others:

- actual or potential changes in international, national, regional and local economic, business and financial conditions, including recession, inflation, higher interest rates, wars, terrorist attacks or political instability;
- changes in consumer preferences, perceptions, spending patterns or demographic trends;
- · changes in the competitive environment due to industry consolidation, changes in airline alliance affiliations and other factors;
- actual or potential disruptions to the air traffic control systems;
- increases in costs of safety, security and environmental measures;
- · outbreaks of diseases that affect travel behavior; and
- · weather and natural disasters.

As a result, our results of operations tend to be volatile and subject to rapid and unexpected change. In addition, due to generally greater demand for air travel during the summer, our revenues in the second and third quarters of the year tend to be stronger than revenues in the first and fourth quarters of the year.

# The airline industry is fiercely competitive and may undergo further consolidation or changes in industry alliances, and we are subject to increasing competition.

Service over almost all of our routes is highly competitive and fares remain at low levels by historical standards. We face vigorous, and, in some cases, increasing, competition from major domestic airlines, national, regional, all-cargo and charter carriers, foreign air carriers, low-cost carriers and, particularly on shorter segments, ground and rail transportation. We also face increasing and significant competition from marketing/operational alliances formed by our competitors. Competition with foreign air carriers and with such marketing/operational alliances has been increasing in recent years in part due to the adoption of liberalized open skies aviation agreements between the United States and an increasing number of countries around the world. Moreover, the percentage of routes on which we compete with carriers having substantially lower operating costs than ours has grown significantly over time, and we now compete with low-cost carriers over a very large part of our network. Our ability to compete effectively depends in part on our ability to maintain a competitive cost structure. If we cannot do so, then our business, financial condition and operating results would be adversely affected.

Certain airline alliances have been, or may in the future be, granted immunity from antitrust regulations by governmental authorities for specific areas of cooperation, such as joint pricing decisions. To the extent alliances formed by our competitors can undertake activities that are not available to us, our ability to effectively compete may be hindered.

Pricing decisions are significantly affected by competition from other airlines. Fare discounting by competitors historically has had a negative effect on our financial results because we must generally match competitors' fares, since failing to match would result in even less revenue. We have faced increased competition from carriers with simplified fare structures, which are generally preferred by travelers. Any fare reduction or fare simplification initiative may not be offset by increases in passenger traffic, reduction in cost or changes in the mix of traffic that would improve yields. Moreover, decisions by our competitors that increase or reduce overall industry capacity, or capacity dedicated to a particular domestic or foreign region, market or route, can have a material impact on related fare levels.

There have been numerous mergers and acquisitions within the airline industry and numerous changes in industry alliances. Southwest Airlines Co. acquired AirTran Airways, Inc. in May 2011, and the recent mergers of United Air Lines, Inc. with Continental Airlines, Inc. and Delta Air Lines Inc. with Northwest Airlines Corporation have resulted in the formation of larger competitors than ourselves with more extensive networks than ours. We are seeking to address these competitive challenges with our cornerstone market and alliance strategies; however there can be no assurances as to the level of success of these strategies.

In the future, there may be additional mergers and acquisitions, and changes in airline alliances, including those in which we may participate and those that may be undertaken by others. Any airline industry consolidation or changes in airline alliances, including **one**world, could substantially alter the competitive landscape and result in changes in our corporate or business strategy. We regularly assess and explore the potential for consolidation in our industry and changes in airline alliances, our strategic position and ways to enhance our competitiveness, including the possibilities for our participation in merger activity. Consolidation involving other participants in our industry could result in the formation of one or more airlines with greater financial resources, more extensive networks, and/or lower cost structures than exist currently, which could have a material adverse effect on our competitive position and adversely affect our business and results of operations. For similar reasons, changes in airline alliances could have a similar impact on us.

We recently began implementing a joint business agreement and related marketing arrangements with British Airways and Iberia, and antitrust-immunized cooperation with British Airways, Iberia, Finnair and Royal Jordanian. In addition, American recently began implementing an antitrust-immunized joint business agreement with the Japan Airlines Group. No assurances can be given as to any arrangements that may ultimately be implemented or any benefits that we may derive from such arrangements.

#### We compete with reorganized carriers, which results in competitive disadvantages for us.

We must compete with air carriers that have reorganized under the protection of Chapter 11 of the Bankruptcy Code in recent years, including United, Delta and US Airways. It is possible that other significant competitors may seek to reorganize in or out of Chapter 11.

Successful reorganizations by other carriers present us with competitors with significantly lower operating costs and stronger financial positions derived from renegotiated labor, supply, and financing contracts. These competitive pressures may limit our ability to adequately price our services, may require us to further reduce our operating costs, and could have a material adverse impact on us.

#### Fares are at low levels and our reduced pricing power adversely affects our ability to achieve adequate pricing, especially with respect to business travel.

Our passenger yield (on an inflation-adjusted basis) remains low by historical standards. We believe that this is due in large part to a corresponding decline in our pricing power. Our reduced pricing power is the product of several factors including: greater cost sensitivity on the part of travelers (particularly business travelers); pricing transparency resulting from the use of the internet; greater competition from low-cost carriers and from carriers that have reorganized in recent years under the protection of Chapter 11; other carriers being better hedged against rising fuel costs and able to better absorb high jet fuel prices; fare simplification efforts by certain carriers; and the economy. We believe that this pricing environment could persist indefinitely.

#### Our corporate or business strategy may change.

In light of the rapid changes in the airline industry, we evaluate our assets on an ongoing basis with a view to maximizing their value to us and determining which are core to our operations. We also regularly evaluate our corporate and business strategies, and they are influenced by factors beyond our control, including changes in the competitive landscape we face. Our corporate and business strategies are, therefore, subject to change.

AMR's regional airline subsidiary, AMR Eagle, has filed a Form 10 registration statement with the SEC in connection with a potential spin-off of AMR Eagle. See "Management's Discussion and Analysis of Financial Condition and Results of Operations—Recent Events—AMR Eagle Divestiture." There can be no assurance that a spin-off or other divestiture of AMR Eagle will be consummated, and no prediction can be made as to the impact of any such transaction on stockholder value, AMR or American.

Our business is subject to extensive government regulation, which can result in increases in our costs, disruptions to our operations, limits on our operating flexibility, reductions in the demand for air travel, and competitive disadvantages. In particular, existing and possible future environmental regulations may adversely affect our business and financial results.

Airlines are subject to extensive domestic and international regulatory requirements. Many of these requirements result in significant costs. For example, the FAA from time to time issues directives and other regulations relating to the maintenance and operation of aircraft. In response to legislation which requires final agency rules by August 2013, the FAA has recently proposed regulations that would affect crewmember hiring and crewmember rest and duty requirements. The industry is seeking clarification from the FAA of certain provisions of the proposed regulations. If the proposed regulations are not amended, we believe they could have a material adverse impact on us. In addition, as a result of heightened levels of concern regarding data privacy, we are subject to an increasing number of domestic and foreign laws regarding the privacy and security of passenger and employee data.

Compliance with regulatory requirements drives significant expenditures and has in the past, and may in the future, cause disruptions to our operations. In addition, the ability of U.S. carriers to operate international routes is subject to change because the applicable arrangements between the U.S. and foreign governments may be amended from time to time (such as through the adoption of an open skies policy), or because appropriate slots or facilities are not made available. Any such change could adversely impact the value of our international route authorities and related assets.

Moreover, additional laws, regulations, taxes and airport rates and charges have been enacted from time to time that have significantly increased the costs of airline operations, reduced the demand for air travel or restricted the way we can conduct our business. For example, the Aviation and Transportation Security Act, which became law in 2001, mandated the federalization of certain airport security procedures and resulted in the imposition of additional security requirements on airlines.

The results of our operations, demand for air travel, and the manner in which we conduct our business each may be affected by changes in law and future actions taken by governmental agencies, including:

- changes in law which affect the services that can be offered by airlines in particular markets and at particular airports, or the types of fees that can be charged to passengers;
- the granting and timing of certain governmental approvals (including foreign government approvals) needed for codesharing alliances and other arrangements with other airlines;
- restrictions on competitive practices (for example court orders, or agency regulations or orders, that would curtail an airline's ability to respond to a competitor);
- the adoption of new passenger security standards or regulations that impact customer service standards (for example, a "passenger bill of rights");
- restrictions on airport operations, such as restrictions on the use of takeoff and landing slots at airports or the auction or reallocation of slot rights currently or previously held by us; or
- ullet the adoption of more restrictive locally imposed noise restrictions.

In addition, the U.S. air traffic control (ATC) system, which is operated by the FAA, is not successfully managing the growing demand for U.S. air travel. U.S. airlines carry about 750 million passengers a year and are forecast to accommodate a billion passengers annually by 2021. Air traffic controllers rely on outdated technologies that routinely overwhelm the system and compel airlines to fly inefficient, indirect routes. We support a common sense approach to ATC modernization that would allocate costs to all ATC system users in proportion to the services they consume. Long-term funding for the FAA expired in 2007. Reauthorization of legislation that funds the FAA, which includes proposals regarding upgrades to the ATC system, is under consideration in Congress. It is uncertain whether such legislation will become law. In the meantime, FAA funding continues under temporary periodic extensions. The current extension expires on January 31, 2012. To date there have been over 20 extensions.

Many aspects of our operations are subject to increasingly stringent environmental regulations. Concerns about climate change and greenhouse gas emissions, in particular, may result in the imposition of additional legislation or regulation. The EU has adopted a directive under which each EU member state is required to extend the existing EU emissions trading scheme (ETS) to aviation. This will require us to annually submit emission allowances in order to operate flights to and from EU member states in January 2012 and thereafter, including flights between the U.S. and EU member states. In December 2009, the ATA, joined by American, Continental and United, filed a legal action in the United Kingdom challenging the implementation of the EU ETS as applied to aviation. The case was subsequently referred to the Court of Justice of the European Union. The court has heard the case and the parties are awaiting a decision. We believe that non-EU governments are also likely to consider formal challenges to the EU ETS as applied to aviation. It is not clear whether the EU ETS directive will withstand legal challenges. However, unless interim relief is granted, we will be required to continue to comply with the EU ETS during the pendency of the legal challenges. Although the cost of compliance with the EU ETS is difficult to predict given the uncertainty of a number of variables, such as the number and price of emission allowances we may be required to purchase, such costs could be significant.

Other legislative or regulatory actions addressing climate change and emissions from aviation that may be taken in the future by the U.S., state or foreign governments or through international treaties may adversely affect our business and financial results. The United Nations' International Civil Aviation Organization (ICAO), for example, recently adopted a resolution identifying certain fuel efficiency goals and emission trading system principles for international aviation, which may provide a basis for such future legislative or regulatory action. Climate change legislation was previously introduced in the U.S. Congress; such legislation could be re-introduced in the future by the U.S. Congress and state legislatures, and could contain provisions affecting the aviation industry. In addition, the U.S. Environmental Protection Agency could seek to regulate greenhouse gas emissions from aircraft. It is currently unknown how climate change legislation or regulation, if enacted, would specifically apply to the aviation industry. However, the impact on us of any climate change legislation or regulation is likely to be adverse and related costs of compliance could be significant. Such legislation or regulation could result in, among other things, increased fuel costs, carbon taxes or fees, the imposition of requirements to purchase emission offsets or credits, increased aircraft and equipment costs, and restrictions on the growth of airline operations. We continue to evaluate ongoing climate change developments at the international, federal and state levels and assess the potential associated impacts on our business and operations.

#### We could be adversely affected by conflicts overseas or terrorist attacks.

Actual or threatened U.S. military involvement in overseas operations has, on occasion, had an adverse impact on our business, financial position (including access to capital markets) and results of operations, and on the airline industry in general. The continuing conflicts in Iraq and Afghanistan, or other conflicts or events in the Middle East or elsewhere, may result in similar adverse impacts.

The Terrorist Attacks had a material adverse impact on us. The occurrence of another terrorist attack (whether domestic or international and whether against us or another entity) could again have a material adverse impact on us.

Our international operations are subject to economic and political instability and could be adversely affected by numerous events, circumstances or government actions beyond our control.

Our current international activities and prospects could be adversely affected by factors such as reversals or delays in the opening of foreign markets, exchange controls, currency and political risks (including changes in exchange rates and currency devaluations), environmental regulation, increases in taxes and fees and changes in international government regulation of our operations, including the inability to obtain or retain needed route authorities and/or slots.

For example, the open skies air services agreement between the U.S. and the EU which took effect in March 2008 provides airlines from the U.S. and EU member states open access to each other's markets, with freedom of pricing and unlimited rights to fly beyond the U.S. and to any airport in the EU including London's Heathrow Airport. The agreement has resulted in American facing increased competition in these markets, including Heathrow Airport. In addition, an open skies air services agreement between the U.S. and Japan that provides airlines from the U.S. and Japan open access to each other's markets took effect in November 2010.

#### We could be adversely affected by an outbreak of a disease that affects travel behavior.

In the second quarter of 2009, there was an outbreak of the H1N1 virus which had an adverse impact throughout our network but primarily on our operations to and from Mexico. In 2003, there was an outbreak of Severe Acute Respiratory Syndrome (SARS), which had an adverse impact primarily on our Asia operations. In addition, in the past there have been concerns about outbreaks or potential outbreaks of other diseases, such as avian flu. Any outbreak of a disease (including an additional outbreak of the H1N1 virus) that affects travel behavior could have a material adverse impact on us. In addition, outbreaks of disease could result in quarantines of our personnel or an inability to access facilities or our aircraft, which could adversely affect our operations.

## Our labor costs are higher than those of our competitors.

Wages, salaries and benefits constitute a significant percentage of our total operating expenses. In 2010, they constituted approximately 28 percent of our total operating expenses. All of the major hub-and-spoke carriers with which American competes have achieved significant labor cost savings through or outside of bankruptcy proceedings. We believe American's labor costs are higher than those of its primary competitors, and it is unclear how long this labor cost disadvantage may persist. These higher labor costs adversely affect our ability to achieve and sustain profitability while competing with other airlines with lower labor costs. Additionally, we cannot predict the outcome of our ongoing negotiations with our unionized work groups. Significant increases in pay and benefits resulting from changes to our collective bargaining agreements could have a material adverse effect on us.

#### We could be adversely affected if we are unable to have satisfactory relations with any unionized or other employee work group.

Our business is labor intensive. To the extent that we are unable to have satisfactory relations with any unionized or other employee work group, our operations and our ability to execute our strategic plans could be adversely affected. In addition, any disruption by an employee work group (e.g., sick-out, slowdown, full or partial strike, or other job action) may materially adversely affect our operations and impair our financial performance.

In April 2003, American reached agreements (the Labor Agreements) with each of its three major unions, the Allied Pilots Association (APA), the Transport Workers Union of America AFL-CIO (TWU) and the Association of Professional Flight Attendants (APFA). The Labor Agreements substantially moderated the labor costs associated with the employees represented by the unions. In conjunction with the Labor Agreements, American also implemented various changes in the pay plans and benefits for non-unionized personnel. The Labor Agreements became amendable in 2008 (although the parties agreed that they could begin the negotiations process as early as 2006). American has been in negotiations with the APA since September 20 2006, the TWU since May 11, 2006 (with respect to Dispatchers), and since November 7, 2007 (with respect to the other six groups at American represented by the TWU), and with the APFA since June 2008 (expedited negotiations) and September 10, 2008 (standard negotiations), to amend their respective Labor Agreements. At this time, all such negotiations are mediated negotiations under the auspices of the National Mediation Board (NMB). NMB mediation with the APFA began on May 6, 2008, with the TWU (with respect to the Dispatchers) on October 28, 2008, with the other TWU groups on various dates in 2009, and with the APFA on January 22, 2009. These negotiations are governed by the Railway Labor Act (RLA), which prescribes no set timetable for the negotiations and mediation process. The negotiations and mediation process in the airline industry typically is slow and sometimes contentious. The RLA prohibits the parties from engaging in self-help prior to the exhaustion of the RLA's bargaining process. That process is not exhausted until the NMB has declared the parties are at a bargaining impasse, one or both parties has declined the NMB's proffer of binding arbitration, and a 30-day cooling off period has expired without the appointment of a Presidential Emergency Board. If we are unable to reach agreement w

In May, 2010, American negotiated tentative agreements with several workgroups within the TWU, including the Maintenance Control Technician group, the Material Logistics Specialists group and the Mechanic and Related group. Agreements with the TWU groups are subject to ratification by the relevant membership of TWU, and, while the Maintenance Control Technician group ratified their agreement, the Material Logistics Specialists group and the Mechanic and Related group tentative agreements were not ratified. In July 2011, American negotiated tentative agreements with two additional TWU-represented workgroups – the Simulator Technicians and the Ground School and Simulator Pilot Instructors. The Instructor group ratified their agreement; however, the Simulator Technician tentative agreement was not ratified.

Mediated negotiations with the APA, with the APFA and with the TWU with respect to groups other than the Maintenance Control Technician and Instructor groups continue. In addition, the APA has filed a number of grievances, lawsuits and complaints, most of which American believes are part of a corporate campaign related to the union's labor agreement negotiations with American. While American is vigorously defending these disputes, unfavorable outcomes in one or more of them could require American to incur additional costs, change the way it conducts some parts of its business, or otherwise adversely affect us.

#### Increases in insurance costs or reductions in coverage could have an adverse impact on us.

We carry insurance for public liability, passenger liability, property damage and all-risk coverage for damage to our aircraft. As a result of the Terrorist Attacks, aviation insurers significantly reduced the amount of insurance coverage available to commercial air carriers for liability to persons other than employees or passengers for claims resulting from acts of terrorism, war or similar events (war-risk coverage). At the same time, these insurers significantly increased the premiums for aviation insurance in general. While the price of commercial insurance has declined since the period immediately after the Terrorist Attacks, in the event commercial insurance carriers further reduce the amount of insurance coverage available to us, or significantly increase its cost, we would be adversely affected.

The U.S. government has agreed to provide commercial war-risk insurance for U.S. based airlines through September 30, 2012, covering losses to employees, passengers, third parties and aircraft. If the U.S. government were to cease providing such insurance in whole or in part, it is likely that we could obtain comparable coverage in the commercial market, but we could incur substantially higher premiums and more restrictive terms, if such coverage is available at all. If we are unable to obtain adequate war-risk coverage at commercially reasonable rates, we would be adversely affected.

#### We may be unable to retain key management personnel.

We are dependent on the experience and industry knowledge of our key management employees, and there can be no assurance that we will be able to retain them. Any inability to retain our key management employees, or attract and retain additional qualified management employees, could have a negative impact on

# We are increasingly dependent on technology and could be adversely affected by a failure or disruption of our computer, communications or other technology systems.

We are heavily and increasingly dependent on technology to operate our business, reduce our costs and enhance customer service. The computer and communications systems on which we rely could be disrupted due to various events, some of which are beyond our control, including natural disasters, power failures, terrorist attacks, equipment failures, system implementation failures, software failures and computer viruses and hackers. We have taken certain steps to help reduce the risk of some (but not all) of these potential disruptions. There can be no assurance, however, that the measures we have taken are adequate to prevent or remedy disruptions or failures of these systems. Any substantial or repeated failure of these systems could impact our operations and customer service, result in the loss of important data, loss of revenues, and increased costs, and generally harm our business. Moreover, a failure of certain of our vital systems could limit our ability to operate our flights for an extended period of time, which would have a material adverse impact on our operations and our business. In addition, we will need to continue to make significant investments in technology to pursue initiatives to reduce costs and enhance customer service. If we are unable to make these investments, our business could be negatively impacted.

#### We are at risk of losses and adverse publicity which might result from an accident involving any of our aircraft.

If one of our aircraft were to be involved in an accident, we could be exposed to significant tort liability. The insurance we carry to cover damages arising from any future accidents may be inadequate. In the event that our insurance is not adequate, we may be forced to bear substantial losses from an accident. In addition, any accident involving an aircraft operated by us could adversely affect the public's perception of us.

#### Interruptions or disruptions in service at one or more of our primary market airports could have an adverse impact on us.

Our business is heavily dependent on our operations at our primary market airports in Dallas/Fort Worth, Chicago, Miami, New York City and Los Angeles. Each of these operations includes flights that gather and distribute traffic from markets in the geographic region around the primary market to other major cities. A significant interruption or disruption in service at one or more of our primary markets could adversely impact our operations.

#### The airline industry is heavily taxed.

The airline industry is subject to extensive government fees and taxation that negatively impact our revenue. The U.S. airline industry is one of the most heavily taxed of all industries. These fees and taxes have grown significantly in the past decade for domestic flights and various U.S. fees and taxes also are assessed on international flights. In addition, the governments of foreign countries in which we operate impose on U.S. airlines, including us, various fees and taxes, and these assessments have been increasing in number and amount in recent years. Under new Department of Transportation regulations that take effect on January 24, 2012, all government taxes and fees must be included in the fares we quote or advertise to our customers. Due to the competitive revenue environment, many increases in these fees and taxes have been absorbed by the airline industry rather than being passed on to the passenger. Further increases in fees and taxes may reduce demand for air travel, and thus our revenues.

#### Item 6. Exhibits

Exhibits required to be filed by Item 601 of Regulation S-K. Where the amount of securities authorized to be issued under any of American's long-term debt agreements does not exceed 10 percent of American's assets, pursuant to paragraph (b) (4) of Item 601 of Regulation S-K, in lieu of filing such as an exhibit, American hereby agrees to furnish to the Commission upon request a copy of any agreement with respect to such long-term debt.

The following exhibits are included herein:

- 10.1 Supplemental Agreement No. 25 to Purchase Agreement No. 1980 by and between American Airlines, Inc. and The Boeing Company dated as of July 19, 2011. Portions of this Exhibit have been omitted and filed separately with the Securities and Exchange Commission pursuant to a confidential treatment request under Rule 24b-2 of the Securities and Exchange Act of 1934, as amended.
- 10.2 Supplemental Agreement No. 26 to Purchase Agreement No. 1980 by and between American Airlines, Inc. and The Boeing Company dated as of July 26, 2011. Portions of this Exhibit have been omitted and filed separately with the Securities and Exchange Commission pursuant to a confidential treatment request under Rule 24b-2 of the Securities and Exchange Act of 1934, as amended.
- 10.3 Supplemental Agreement No. 35 to Purchase Agreement No. 1977 by and between American Airlines, Inc. and The Boeing Company dated as of August 19, 2011. Portions of this Exhibit have been omitted and filed separately with the Securities and Exchange Commission pursuant to a confidential treatment request under Rule 24b-2 of the Securities and Exchange Act of 1934, as amended.
- 10.4 A320 Family Aircraft Purchase Agreement by and between American Airlines, Inc. and Airbus S.A.S. dated as of July 20, 2011. Portions of this Exhibit have been omitted and filed separately with the Securities and Exchange Commission pursuant to a confidential treatment request under Rule 24b-2 of the Securities and Exchange Act of 1934, as amended.
- 12 Computation of ratio of earnings to fixed charges for the three and nine months ended September 30, 2011 and 2010.
- 31.1 Certification of Chief Executive Officer pursuant to Rule 13a-14(a).
- 31.2 Certification of Chief Financial Officer pursuant to Rule 13a-14(a).
- Certification pursuant to Rule 13a-14(b) and section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code).
- The following materials from American's Quarterly Report on Form 10-Q for the quarter ended September 30, 2011, formatted in XBRL (Extensible Business Reporting Language): (i) the Condensed Consolidated Statements of Operations, (ii) the Condensed Consolidated Balance Sheets, (iii) the Condensed Consolidated Statements of Cash Flows, and (iv) Notes to Condensed Consolidated Financial Statements, tagged as blocks of text.\*
- \* Pursuant to Rule 406T of Regulation S-T, the Interactive Data Files on Exhibit 101 hereto are deemed not filed or part of a registration statement or prospectus for purposes of Sections 11 or 12 of the Securities Act of 1933, as amended, are deemed not filed for purposes of Section 18 of the Securities and Exchange Act of 1934, as amended, and otherwise are not subject to liability under those sections.

# Signature

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

AMERICAN AIRLINES, INC.

Date: October 19, 2011 BY: /s/ Isabella D. Goren

Isabella D. Goren

Senior Vice President and Chief Financial Officer

Supplemental Agreement No. 25

to

Purchase Agreement No. 1980

between

The Boeing Company

and

# AMERICAN AIRLINES, INC

Relating to Boeing Model 777 Aircraft

THIS SUPPLEMENTAL AGREEMENT, entered into this day of August, 2011, (*SA-25*) by and between THE BOEING COMPANY, a Delaware corporation with offices in Seattle, Washington, (*Boeing*) and American Airlines, Inc. (*Customer*);

#### **RECITALS:**

WHEREAS, Boeing and Customer entered into Purchase Agreement No. 1980 dated as of October 31, 1997, as amended and supplemented relating to Boeing Model 777 aircraft (the *Purchase Agreement*) (capitalized terms used herein without definition shall have the meanings specified therefor in such Purchase Agreement); and

WHEREAS, Customer has provided notice to exercise as a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

WHEREAS, Boeing and Customer have agreed to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

NOW THEREFORE, In consideration of the mutual covenants herein contained, the parties agree to amend the Purchase Agreement as follows:

#### Table of Contents:

The "Table of Contents" to the Purchase Agreement is deleted in its entirety and a revised "Table of Contents" attached hereto and identified with an "SA-25" legend, is substituted in lieu thereof to reflect the changes made by this SA-25.

P.A. No. 1980 i SA-25

#### 2. Table 1-7:

Table 1-7 entitled 777-323ER [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft Delivery, Description, Price and Advance Payments is deleted in its entirety and a revised Table 1-7, attached hereto, is substituted in lieu thereof to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 3. <u>Supplemental Exhibit BFE1-2</u>:

Supplemental Exhibit BFE1-2 entitled Buyer Furnished Equipment Variables relating to Boeing Model 777-323ER Aircraft is deleted in its entirety and a revised BFE1-2, attached hereto, is substituted in lieu thereof to set forth the preliminary BFE [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 4. Letter Agreement No. 6-1162-AKP-110R3:

Attachment C entitled Information Regarding QADP Rights to Letter Agreement No. 6-1162-AKP-110R3 entitled Aircraft Purchase Rights and Substitution Rights is deleted in its entirety and a revised Attachment C, attached hereto, is substituted in lieu thereof [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] via this SA-25.

Attachment B entitled Information Regarding MADP Rights to Letter Agreement No. 6-1162-AKP-110R3 entitled Aircraft Purchase Rights and Substitution Rights is deleted in its entirety and a revised Attachment B, attached hereto, is substituted in lieu thereof [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] via this SA-25.

Nothing in this SA-25 precludes Customer from hereafter exercising its Substitution Right in accordance with Paragraph 5 of Letter Agreement No. 6-1162-AKP-110R3 entitled Aircraft Purchase Rights and Substitution Rights.

#### 5. Letter Agreement No. AAL-PA-1980-LA-04205R2:

Letter Agreement No. AAL-PA-1980-LA-04205R2 entitled Aircraft Performance Guarantees – 777-323ER is deleted in its entirety and a revised Letter Agreement No. AAL-PA-1980-LA-04205R3, attached hereto, is substituted in lieu thereof to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE

P.A. No. 1980 ii SA-25

COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] which are being exercised via this SA-25, in addition to all Aircraft set forth in Table 1-7, which were exercised via Supplemental Agreement Numbers 22, 23, and 24.

#### 6. Letter Agreement No. AAL-PA-1980-1105629:

Letter Agreement No. AAL-PA-1980-1105629 entitled Advance Payments and Permitted Transactions, attached hereto, is added to the Purchase Agreement to set forth, among other things, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The Purchase Agreement will be deemed to be amended to the extent provided herein and as so amended will continue in full force and effect. In the event of any inconsistency between the above provisions and the provisions contained in the referenced exhibits to this Supplemental Agreement No. 25, the terms of the exhibits will control.

EXECUTED IN DUPLICATE as of the day and year first above written.

THE BOEING COMPANY	AMERICAN AIRLINES, INC.	
Ву:	Ву:	
Name:	Name:	
Its: Attorney-In-Fact	Its:	
P.A. No. 1980	iii	SA-25

## TABLE OF CONTENTS

ARTICLES 1. 2. 3. 4. 5.	Quantity, Model and Description Delivery Schedule Price Payment Miscellaneous	PAGE & SA NUMBER 1, SA-20 1, SA-20 1, SA-20 2, SA-20 2, SA-20
TABLE 1.	777-200ER Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	SA NUMBER Original Purchase Agreement, SA-3, SA-17 & SA-18
1-1.	777-200IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe & Engine Base Year	SA-1
P.A. No. 19	980	SA-25

Table of Contents, Page i

1-2.	777-200IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-2 & SA-6
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe & Engine Base Year	
1-3.	777-200IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-4, SA-5, SA-6, SA-7 & SA-9
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year	
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	
1-4.	777-200IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-5, SA-6 & SA- 9
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year	
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	

P.A. No. 1980 SA-25

Table of Contents, Page ii

1-5.	777-223IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-10, SA-11, SA-12 & SA-15
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year	
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	
1-6.	777-200IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-13
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year	
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	
1-7.	777-323ER [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR	SA-25

P.A. No. 1980 SA-25

Table of Contents, Page iii

CONFIDENTIAL TREATMENT] Base Year

TABLE 2.	777-223IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE	SA NUMBER SA-15 & SA-16
۷.	COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	5A-15 & 5A-16
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year	
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	
3.	777-223IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-15 & SA-16
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	
	Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	
EXHIBIT A.	Aircraft Configuration	SA NUMBER
A1.	Aircraft Configuration – 777-323ER	SA-20

A. Aircraft Configuration

A1. Aircraft Configuration – 777-323ER

B. Aircraft Delivery Requirements and Responsibilities

C. Defined Terms

SA-20

SA-20

P.A. No. 1980

Table of Contents, Page iv

SUPPLEMEN	TAL EXHIBITS	SA NUN	MBER
AE1.	Escalation Adjustment Airframe and Optional Features - 777-323ER	SA-20	
BFE1.	BFE Variables		
BFE1-2.	BFE Variables - 777-323ER	SA-25	
CS1.	Customer Support Variables		
CS1-2	Customer Support Variables - 777-323ER	SA-20	
SLP1	Service Life Policy Components		
EE1-BR1.	Engine Escalation and Engine Warranty	SA-15	
EE1-2.	Engine Escalation, Engine Warranty and Patent Indemnity – 777-323ER	SA-20	
P.A. No. 19	80		SA-25

Table of Contents, Page v

LETTER AGREEMENTS		PA or SA NUMBER
6-1162-AKP-070	Miscellaneous Commitments for Model 737, 757, 767 and 777 Aircraft	HOMBER
6-1162-AKP-071R1	Purchase Obligations	PA3219
6-1162-AKP-072R3	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-20
6-1162-AKP-073R1	Accident Claims and Litigation	PA3219
6-1162-AKP-109R3	Business Considerations	SA-20
6-1162-AKP-110R3	Aircraft Purchase Rights and Substitution Rights	SA-20
Attachment A	Description and Price for Eligible Models	SA-20
Attachment B	Information Regarding MADP Rights	SA-25
Attachment C	Information Regarding QADP Rights	SA-25
Attachment D	Forms of Purchase Agreement Supplement	SA-20
Attachment E	Letter Agreements	SA-20
Attachment F	Information regarding MADP and QADP Rights if no 787s are reconfirmed	SA-20
6-1162-AKP-111	Aircraft Performance Guarantees	
AAL-PA-1980-LA-1003346	Aircraft Performance Guarantees - 777-323ER	SA-20
AAL-PA-1980-LA-04205R3	Aircraft Performance Guarantees – 777-323ER	SA-25
6-1162-AKP-112	Spares Matters	
6-1162-AKP-113	Model 777 Miscellaneous Commitments	
6-1162-AKP-114R1	Installation of Cabin Systems Equipment	SA-22
AAL-PA-1980-LA-1003493	Installation of Cabin Systems Equipment – 777-323ER	SA-20
6-1162-AKP-115	Component and System Reliability Commitments	
6-1162-AKP-116	Price Adjustment on Rolls-Royce Engines	
6-1162-AKP-117	Delivery Schedule	
6-1162-AKP-118R2	Confidentiality	SA-20
6-1162-AKP-204	Multiple Operating Weight Program Model 777-200IGW Aircraft	SA-6
P.A. No. 1980		SA-25

Table of Contents, Page vi

P.A. No. 1980	SA-25
AAL-PA-1980-LA-1105629 Advance Payments and Permitted Transactions	SA-25
AAL-PA-1980-LA-1104563 Performance Guarantees for Rights Aircraft	SA-23
AAL-PA-1980-LA-1003344 Open Configuration Matters -777-323ER	SA-20
AAL-PA-1980-LA-1003536R1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-21
A A I DA 1000 I A 10025001 CONTRIBENTIAL PORTION OMITTER AND FILED CRAD ATTENVIAUTH THE COMMISSION	CA 21

**BOEING PROPRIETARY** 

Table of Contents, Page vii

#### Table 1-7

# 777-323ER [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft

# Aircraft Delivery, Description, Price and Advance Payments

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

AAL-PA-01980 SA-20 APR 55725 SA-22, SA-23, SA-24, SA 25 APR 56305

**Boeing Proprietary** 

SA-25

Page 1

# BUYER FURNISHED EQUIPMENT VARIABLES

between

# THE BOEING COMPANY

and

American Airlines, Inc.

Supplemental Exhibit BFE1-2 to Purchase Agreement Number 1980

P.A. No. 1980 BFE1-2

SA-25 Page 1

# **BUYER FURNISHED EQUIPMENT VARIABLES**

## relating to

# **BOEING MODEL 777-323ER AIRCRAFT**

This Supplemental Exhibit BFE1-2 contains supplier selection dates, on-dock dates and other requirements applicable to the Model 777-323ER aircraft (Aircraft).

# Supplier Selection.

Customer will:

Select and notify Boeing of the suppliers and part numbers of the following BFE items by the following dates:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	N/A
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1980 BFE1-2 SA-25 Page 2

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	***
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

\*\* [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

\*\*\* [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1980 BFE1-2 SA-25

# 2. <u>On-dock Dates and Other Information</u>.

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

<u>Item</u>	Preliminary On-Dock Dates	
	[CONFIDENTIAL	[CONFIDENTIAL
	PORTION OMITTED	PORTION OMITTED
	AND FILED	AND FILED
	SEPARATELY WITH	SEPARATELY WITH
	THE COMMISSION	THE COMMISSION
	PURSUANT TO A	PURSUANT TO A
	REQUEST FOR	REQUEST FOR
	CONFIDENTIAL	CONFIDENTIAL
	TREATMENT] 2012	TREATMENT] 2012
	<u>Aircraft</u>	<u>Aircraft</u>
Seats	[CONFIDENTIAL	[CONFIDENTIAL
Galleys/Furnishings	PORTION OMITTED	PORTION OMITTED
Galleys/Furnishings	AND FILED	AND FILED
Antennas & Mounting Equipment	SEPARATELY WITH	SEPARATELY WITH
Avionics	THE COMMISSION	THE COMMISSION
	PURSUANT TO A	PURSUANT TO A
Cabin Systems Equipment	REQUEST FOR	REQUEST FOR
Miscellaneous Emergency Equipment	CONFIDENTIAL TREATMENT]	CONFIDENTIAL TREATMENT]
Textiles/Raw Material	TREATMENT	IREATMENT
TCALICS/TAW MILLERIN		
Item	Preliminary On-Dock Dates	
	1 Tellimitary	OII-DOCK Dates
<del>_</del>	[CONFIDENTIAL	[CONFIDENTIAL
<del>_</del>	[CONFIDENTIAL PORTION OMITTED	[CONFIDENTIAL PORTION OMITTED
<del>_</del>	[CONFIDENTIAL PORTION OMITTED AND FILED	[CONFIDENTIAL PORTION OMITTED AND FILED
<u> </u>	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft
Seats	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED
Seats	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED
Seats Galleys/Furnishings Antennas & Mounting Equipment	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH
Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION
Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A
Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR
Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 2013 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

P.A. No. 1980 BFE1-2

SA-25

<u>Item</u>	Preliminary On-Dock Dates		
	[CONFIDENTIAL	[CONFIDENTIAL	
	PORTION OMITTED	PORTION OMITTED	
	AND FILED	AND FILED	
	SEPARATELY WITH	SEPARATELY WITH	
	THE COMMISSION	THE COMMISSION PURSUANT TO A	
	PURSUANT TO A REQUEST FOR	REQUEST FOR	
	CONFIDENTIAL	CONFIDENTIAL	
	TREATMENT] 2013	TREATMENT] 2013	
	<u>Aircraft</u>	<u>Aircraft</u>	
Seats	[CONFIDENTIAL	[CONFIDENTIAL	
Galleys/Furnishings	PORTION OMITTED	PORTION OMITTED	
	AND FILED	AND FILED	
Antennas & Mounting Equipment	SEPARATELY WITH	SEPARATELY WITH	
Avionics	THE COMMISSION PURSUANT TO A	THE COMMISSION PURSUANT TO A	
Cabin Systems Equipment	REQUEST FOR	REQUEST FOR	
	CONFIDENTIAL	CONFIDENTIAL	
Miscellaneous Emergency Equipment	TREATMENT]	TREATMENT]	
Textiles/Raw Material			
Tanan	n. II. 1	D 1 D .	
<u>Item</u>	Preliminary On-	[CONFIDENTIAL	
	PORTION OMITTED	PORTION OMITTED	
	AND FILED	AND FILED	
	SEPARATELY WITH	SEPARATELY WITH	
	THE COMMISSION	THE COMMISSION	
	PURSUANT TO A	PURSUANT TO A	
	REQUEST FOR	REQUEST FOR	
	CONFIDENTIAL	CONFIDENTIAL	
	TREATMENT] <u>2013</u> Aircraft	TREATMENT] <u>2013</u> <u>Aircraft</u>	
	<del></del>		
Seats	[CONFIDENTIAL	[CONFIDENTIAL	
Galleys/Furnishings	PORTION OMITTED	PORTION OMITTED	
•	AND FILED	AND FILED	
Antennas & Mounting Equipment	SEPARATELY WITH THE COMMISSION	SEPARATELY WITH THE COMMISSION	
Avionics	PURSUANT TO A	PURSUANT TO A	
Cabin Systems Equipment	REQUEST FOR	REQUEST FOR	
Miscellaneous Emergency Equipment	CONFIDENTIAL	CONFIDENTIAL	
Miscellaneous Emergency Equipment	TREATMENT]	TREATMENT]	

Textiles/Raw Material

# 3. <u>Additional Delivery Requirements - Import.</u>

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1980 BFE1-2 SA-25 Page 6

# Attachment B to Letter Agreement 6-1162-AKP-110R3 (Model 777) Information Regarding MADP Rights

[CONFIDENTIAL PORTION OMITTED AND

FILED SEPARATELY WITH THE

COMMISSION PURSUANT TO A REQUEST

FOR CONFIDENTIAL TREATMENT]

PA No. 1980 SA No. 25 Page 1 of 1

# Attachment C to Letter Agreement 6-1162-AKP-110R3 (Model 777) Information Regarding QADP Rights

[CONFIDENTIAL PORTION OMITTED AND

FILED SEPARATELY WITH THE

COMMISSION PURSUANT TO A REQUEST

FOR CONFIDENTIAL TREATMENT]

PA No. 1980 SA No. 25 Page 1 of 1





AAL-PA-1980-LA-04205R3

American Airlines, Inc. P.O. Box 619616

Dallas-Fort Worth Airport, Texas 75261-9616

Subject: Aircraft Performance Guarantees – 777-323ER

Reference: Purchase Agreement No. PA-1980 (Purchase Agreement) between The Boeing Company (Boeing) and American Airlines, Inc. (Customer)

relating to Model 777-323ER aircraft (Aircraft)

This letter agreement (**Letter Agreement**) amends and supplements the Purchase Agreement. All terms used but not defined in this Letter Agreement shall have the same meaning as in the Purchase Agreement.

For the Aircraft set forth in Table 1-7, which are exercised via Supplemental Agreements No. 22, 23, 24, and 25, Boeing agrees to provide Customer with the guarantees set forth in Attachment A hereto. These guarantees are exclusive and will expire upon delivery of the Aircraft to Customer.

1. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL

TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2. Confidential Treatment.

The information contained herein represents confidential business information and has value precisely because it is not available generally or to other parties. Customer will limit the disclosure of its contents to employees of Customer with a need to know the contents for purposes of helping Customer perform its obligations under the Purchase Agreement and who understand they are not to disclose its contents to any other person or entity without the prior written consent of Boeing.

AAL-PA-1980-LA-04205R2 Performance Guarantees – 777-323ER SA-25 Page 1



Very truly yours,

THE BOEING COMPANY

By				
Its	Attorney-In-Fact			
ACCEPTED AND AGREED TO this				
Date:				
American Airlines, Inc.				
By				
Its				

AAL-PA-1980-LA-04205R2 Performance Guarantees – 777-323ER

**BOEING PROPRIETARY** 

SA-25 Page 2





AAL-PA-1980-LA-1105629

American Airlines, Inc. P.O. Box 619616

P.A. No. 1980

Dallas-Fort Worth Airport, Texas 75261-9616

Subject: Advance Payments and Permitted Transactions

Reference: Purchase Agreement No. 1980 (the Purchase Agreement) between The Boeing Company (Boeing) and American Airlines, Inc. (Customer)

relating to Model 777 aircraft (the Aircraft)

This Letter Agreement amends and supplements the Purchase Agreement. All terms used but not defined in this Letter Agreement have the same meaning as in the Purchase Agreement.

#### 1. Advance Payments for Aircraft.

Notwithstanding Article 4.2 of the Purchase Agreement and any tables to the Purchase Agreement that set forth Boeing's standard [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Advance Payment Schedule for exercised Rights Aircraft ("the Standard Advance Payment Schedule"), Boeing and Customer agree that the Advance Payment Schedule [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT], is hereby amended as follows:

1.1 Customer will make Advance Payments to Boeing in the amount of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of each Aircraft and will do so as follows (the "Modified Advance Payment Schedule"): (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT], and (b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Advance Payments and Permitted Transactions

SA-25

Page 1



- 1.2 Notwithstanding Section 1.3 of Letter Agreement No. 6-1162-AKP-070 entitled Miscellaneous Commitments for Model 737, 757, 767 and 777 Aircraft, Customer will pay Boeing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.5 Attachment A hereto sets forth the calculation of the revised Advanced Payments due as of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2. Permitted Transactions.

Notwithstanding Article 9.2 of the AGTA, Boeing agrees that Customer may from time to time prior to delivery of any Aircraft enter into arrangements and agreements with lenders, lessors or other parties (whose business includes, but may not be limited to, the financing of aircraft) for the purpose of (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (an "Advance Payment Facility") and/or (b) consummating [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] pursuant to which the right to purchase an Aircraft, and the corresponding obligation to make Advance Payments with respect thereto, are conveyed to such a party [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and, together with an Advance Payment Facility, the "Permitted Transactions").

For the purpose of securing an Advance Payment Facility and notwithstanding the provisions of the Security Agreement between Boeing and Customer dated as of October 16, 2002, as may be subsequently amended or revised, Customer may [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] including the right to purchase an Aircraft provided that Customer and Customer's financiers accept Boeing's customary terms and conditions for consenting

P.A. No. 1980 Advance Payments and Permitted Transactions SA-25 Page 2



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] including, but not limited to, the following:

- (i) if Customer's financier gives notice that it intends to exercise its [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Boeing shall have the right to assume those rights with respect to such Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and received and retained by Boeing with respect to that Aircraft (Manufacturer's Option);
- (ii) Customer shall continue at all times to remain liable to Boeing under the Purchase Agreement to perform all duties and obligations of Customer;
- (iii) Boeing shall not be subject to any additional liability as a result of the pledge of security which Boeing would not otherwise be subject to under the Purchase Agreement;
- (iv) the pledge of security shall not modify in any respect the continued rights of Boeing under the Purchase Agreement, or require Boeing to divest itself of title to or possession of the Aircraft, or any other things, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and
- (v) In lieu of the concessions granted to Customer under the Purchase Agreement, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] In calculating the amount payable by the financier for an Aircraft, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and this irrespective of whether the [CONFIDENTIAL PORTION OMITTED

P.A. No. 1980 Advance Payments and Permitted Transactions SA-25

Page 3



#### AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

To assist Customer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Customer may execute [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Any such [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be subject to the conditions that (i) it shall not subject Boeing to any liability as a result of the assignment which Boeing would not otherwise be subject to under the Purchase Agreement and (ii) no such assignment will require Boeing to divest itself of title to or possession of the aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] the Purchase Agreement in connection with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] is contingent upon Boeing's consent, which shall not be unreasonably withheld or delayed. Boeing agrees to cooperate in good faith with Customer and to take such actions as may be reasonably requested by Customer to facilitate Permitted Transactions. Notwithstanding Paragraph 1.1 above, Advance Payments in respect of any Aircraft that is subjected to a Permitted Transaction will be made in accordance with the Standard Advance Payment Schedule. Advance Payments for all Aircraft that are not subject to Permitted Transactions will be made in accordance with Paragraph 1 above.

For avoidance of doubt, nothing within this Letter Agreement is intended to, nor shall it, derogate the rights and obligations of Boeing and Customer with regard to the financing of aircraft at or following delivery in accordance with Article 9.2 of the AGTA.

#### 3. Confidential Treatment.

Customer and Boeing understand certain commercial and financial information contained in this Letter Agreement is considered by Boeing and Customer as confidential. Customer and Boeing agree that each will treat this Letter Agreement and

P.A. No. 1980 Advance Payments and Permitted Transactions SA-25

Page 4



Very truly yours,

the information contained herein as confidential and will not, without the prior written consent of the other, disclose this Letter Agreement or any information contained herein to any other person or entity, except as required by law or government regulation.

THE BOEING COMPANY			
Зу			
ts Attorney-In-Fact			
ACCEPTED AND AGREED TO this			
Date: , 2011			
AMERICAN AIRLINES, INC.			
Зу			
ts			
P.A. No. 1980			

Advance Payments and Permitted Transactions

**BOEING PROPRIETARY** 

SA-25

Page 5

Attachment A
to
AAL-PA-1980-LA-1105629
entitled

**Advance Payments and Permitted Transactions** 

[CONFIDENTIAL PORTION OMITTED AND

FILED SEPARATELY WITH THE COMMISSION

PURSUANT TO A REQUEST FOR CONFIDENTIAL

TREATMENT]

P.A. No. 1980 Advance Payments and Permitted Transactions SA-25

Supplemental Agreement No. 26

to

Purchase Agreement No. 1980

between

The Boeing Company

and

# AMERICAN AIRLINES, INC

Relating to Boeing Model 777 Aircraft

THIS SUPPLEMENTAL AGREEMENT, entered into this day of September 2011, (*SA-26*) by and between THE BOEING COMPANY, a Delaware corporation with offices in Seattle, Washington, (*Boeing*) and American Airlines, Inc. (*Customer*);

#### **RECITALS:**

WHEREAS, Boeing and Customer entered into Purchase Agreement No. 1980 dated as of October 31, 1997, as amended and supplemented (capitalized terms used herein without definition shall have the meanings specified therefor in such Purchase Agreement) relating to Boeing Model 777 aircraft (the *Purchase Agreement*); and

WHEREAS, Customer has provided notice to substitute a 777-323ER Aircraft in lieu of the 777-223ER Aircraft scheduled for delivery in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

NOW THEREFORE, In consideration of the mutual covenants herein contained, the parties agree to amend the Purchase Agreement as follows:

#### Table of Contents:

The <u>"Table of Contents"</u> to the Purchase Agreement is deleted in its entirety and a revised "<u>Table of Contents</u>" attached hereto and identified with an "SA-26" legend, is substituted in lieu thereof to reflect the changes made by this SA-26.

## 2. <u>Table 1:</u>

Table 1 entitled Aircraft Delivery, Description, Price and Advance Payments is deleted in its entirety and a revised Table 1, attached hereto, is substituted in lieu thereof to remove the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 777-223ER Aircraft as it is being substituted to a 777-323ER Aircraft.

P.A. No. 1980 i SA-26

#### 3. Table 1-7:

Table 1-7 entitled 777-323ER Substitute Aircraft Delivery, Description, Price and Advance Payments is deleted in its entirety and a revised Table 1-7, attached hereto, is substituted in lieu thereof to set forth the 777-323ER Aircraft with delivery scheduled for [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 4. <u>Supplemental Exhibit BFE1-2</u>:

Supplemental Exhibit BFE1-2 entitled Buyer Furnished Equipment Variables relating to Boeing Model 777-323ER Aircraft is deleted in its entirety and a revised BFE1-2, attached hereto, is substituted in lieu thereof to set forth the preliminary BFE [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The Purchase Agreement will be deemed to be amended to the extent provided herein and as so amended will continue in full force and effect. In the event of any inconsistency between the above provisions and the provisions contained in the referenced exhibits to this Supplemental Agreement No. 26, the terms of the exhibits will control.

EXECUTED IN DUPLICATE as of the day and year first above written.

THE BUEING COMPANY	AMERICAN AIRLINES, INC.	
Ву:	Ву:	
Name: Lanine Lange	Name:	
Its: Attorney-In-Fact	Its:	
P.A. No. 1980	ii	SA-26

### TABLE OF CONTENTS

ARTICLES

Table of Contents, Page i

PAGE & SA NUMBER

1.	Quantity, Model and Description	1, SA-20
2.	Delivery Schedule	1, SA-20
3.	Price	1, SA-20
4.	Payment	2, SA-20
5.	Miscellaneous	2, SA-20
TABLE 1.	777-200ER Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	SA NUMBER Original Purchase Agreement, SA-3, SA-17, SA-18 & SA-26
1-1.	777-200IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe & Engine Base Year	SA-1
1-2.	777-200IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe & Engine Base Year	SA-2 & SA-6
P.A. No. 1980		

1-3.	777-200IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	SA-4, SA-5, SA-6, SA-7 & SA-9
1-4.	777-200IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	SA-5, SA-6 & SA-9
1-5.	777-223IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	SA-10, SA-11, SA-12 & SA-15
P.A. No. 19	980	SA-26

P.A. No. 1980 Table of Contents, Page ii

1-6. 777-200IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE SA-13 COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year 1-7. 777-323ER Substitute Aircraft SA-25 & SA-26 Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Base Year TABLE 2. SA NUMBER 777-223IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE SA-15 & SA-16 COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year P.A. No. 1980 SA-26

Table of Contents, Page iii BOEING PROPRIETARY

3.	777-223IGW Aircraft Information Table: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Airframe Base Year [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]  Jul. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Base Year	SA-15 & SA-16
EXHIBIT		SA NUMBER
A.	Aircraft Configuration	
A1.	Aircraft Configuration – 777-323ER	SA-20
B.	Aircraft Delivery Requirements and Responsibilities	SA-20
C.	Defined Terms	SA-20
SUPPLEMENT	AL EXHIBITS	SA NUMBER
AE1.	Escalation Adjustment Airframe and Optional Features - 777-323ER	SA-20
BFE1.	BFE Variables	
BFE1-2.	BFE Variables - 777-323ER	SA-25 & SA-26
CS1.	Customer Support Variables	
CS1-2	Customer Support Variables - 777-323ER	SA-20
SLP1	Service Life Policy Components	
EE1-BR1.	Engine Escalation and Engine Warranty	SA-15
EE1-2.	Engine Escalation, Engine Warranty and Patent Indemnity – 777-323ER	SA-20
P.A. No. 198	30	SA-26
Table of Co	ntents, Page iv BOEING PROPRIETARY	

LETTER AGREEMENTS		PA or SA NUMBER
6-1162-AKP-070	Miscellaneous Commitments for Model 737, 757, 767 and 777 Aircraft	
6-1162-AKP-071R1	Purchase Obligations	PA3219
6-1162-AKP-072R3	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION	SA-20
	PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	
6-1162-AKP-073R1	Accident Claims and Litigation	PA3219
6-1162-AKP-109R3	Business Considerations	SA-20
6-1162-AKP-110R3	Aircraft Purchase Rights and Substitution Rights	SA-20
Attachment A	Description and Price for Eligible Models	SA-20
Attachment B	Information Regarding MADP Rights	SA-25
Attachment C	Information Regarding QADP Rights	SA-25
Attachment D	Forms of Purchase Agreement Supplement	SA-20
Attachment E	Letter Agreements	SA-20
Attachment F	Information regarding MADP and QADP Rights if no 787s are reconfirmed	SA-20
6-1162-AKP-111	Aircraft Performance Guarantees	
AAL-PA-1980-LA-1003346	Aircraft Performance Guarantees - 777-323ER	SA-20
AAL-PA-1980-LA-04205R3	Aircraft Performance Guarantees – 777-323ER	SA-25
6-1162-AKP-112	Spares Matters	
6-1162-AKP-113	Model 777 Miscellaneous Commitments	
6-1162-AKP-114R1	Installation of Cabin Systems Equipment	SA-22
AAL-PA-1980-LA-1003493	Installation of Cabin Systems Equipment – 777-323ER	SA-20
6-1162-AKP-115	Component and System Reliability Commitments	
6-1162-AKP-116	Price Adjustment on Rolls-Royce Engines	
6-1162-AKP-117	Delivery Schedule	
6-1162-AKP-118R2	Confidentiality	SA-20
6-1162-AKP-204	Multiple Operating Weight Program Model 777-200IGW Aircraft	SA-6
AAL-PA-1980-LA-1003536R1	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-21
AAL-PA-1980-LA-1003344	Open Configuration Matters -777-323ER	SA-20
AAL-PA-1980-LA-1104563	Performance Guarantees for Rights Aircraft	SA-23
AAL-PA-1980-LA-1105629	Advance Payments and Permitted Transactions	SA-25
P.A. No. 1980		SA-26
Table of Contents, Page v	BOEING PROPRIETARY	

### Table 1 to Purchase Agreement No. 1980 Aircraft Delivery, Description, Price and Advance Payments

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

AAL - SA26 Boeing Proprietary Page 1

# Table 1-7 777-323ER Substitute Aircraft Aircraft Delivery, Description, Price and Advance Payments

 $[{\tt CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]}$ 

AAL-PA-01980 SA-20 APR 55725 SA-22, SA-23, SA-24, SA 25 APR 56305 SA-26 APR 58150

**Boeing Proprietary** 

SA-26 Page 1

# BUYER FURNISHED EQUIPMENT VARIABLES

between

# THE BOEING COMPANY

and

American Airlines, Inc.

Supplemental Exhibit BFE1-2 to Purchase Agreement Number 1980

P.A. No. 1980 BFE1-2

SA-26 Page 1

### BUYER FURNISHED EQUIPMENT VARIABLES

### relating to

# **BOEING MODEL 777-323ER AIRCRAFT**

This Supplemental Exhibit BFE1-2 contains supplier selection dates, on-dock dates and other requirements applicable to the Model 777-323ER aircraft (Aircraft).

### Supplier Selection.

Customer will:

Select and notify Boeing of the suppliers and part numbers of the following BFE items by the following dates:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST	
FOR CONFIDENTIAL TREATMENT]	N/A
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1980 BFE1-2 SA-26 Page 2

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH

SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR

CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

\*\*\*

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL

[CONFIDENTIAL

TREATMENT]

- \*\* [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- \*\*\* [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2. On-dock Dates and Other Information.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1980 BFE1-2

SA-26 Page 3

<u>Item</u>	Preliminary On-Dock Dates		
	[CONFIDENTIAL	[CONFIDENTIAL	
	PORTION OMITTED	PORTION OMITTED	
	AND FILED	AND FILED	
	SEPARATELY WITH THE COMMISSION	SEPARATELY WITH THE COMMISSION	
	PURSUANT TO A	PURSUANT TO A	
	REQUEST FOR	REQUEST FOR	
	CONFIDENTIAL	CONFIDENTIAL	
	TREATMENT] 2012	TREATMENT] 2012	
	<u>Aircraft</u>	<u>Aircraft</u>	
Seats	[CONFIDENTIAL	[CONFIDENTIAL	
Galleys/Furnishings	PORTION OMITTED	PORTION OMITTED	
	AND FILED	AND FILED	
Antennas & Mounting Equipment	SEPARATELY WITH THE COMMISSION	SEPARATELY WITH THE COMMISSION	
Avionics	PURSUANT TO A	PURSUANT TO A	
Cabin Systems Equipment	REQUEST FOR	REQUEST FOR	
	CONFIDENTIAL	CONFIDENTIAL	
Miscellaneous Emergency Equipment	TREATMENT]	TREATMENT]	
Textiles/Raw Material			
<u>Item</u>	Preliminary	On-Dock Dates	
	[CONFIDENTIAL	[CONFIDENTIAL	
	PORTION OMITTED	PORTION OMITTED	
	AND FILED	AND FILED	
	SEPARATELY WITH	SEPARATELY WITH	
	THE COMMISSION PURSUANT TO A	THE COMMISSION PURSUANT TO A	
	REQUEST FOR	REQUEST FOR	
	CONFIDENTIAL	CONFIDENTIAL	
	TREATMENT] 2013	TREATMENT] 2013	
	<u>Aircraft</u>	<u>Aircraft</u>	
Seats	[CONFIDENTIAL	[CONFIDENTIAL	
Galleys/Furnishings	PORTION OMITTED	PORTION OMITTED	
	AND FILED	AND FILED	
Antennas & Mounting Equipment	SEPARATELY WITH THE COMMISSION	SEPARATELY WITH THE COMMISSION	
Avionics	PURSUANT TO A	PURSUANT TO A	
Cabin Systems Equipment	REQUEST FOR	REQUEST FOR	
Miscellaneous Emergency Equipment	CONFIDENTIAL	CONFIDENTIAL	
Miscendicous Emergency Equipment	TREATMENT]	TREATMENT]	

Textiles/Raw Material

Item	Preliminary On-Dock Dates	
_ <del></del>	[CONFIDENTIAL	[CONFIDENTIAL
	PORTION OMITTED	PORTION OMITTED
	AND FILED	AND FILED
	SEPARATELY WITH	SEPARATELY WITH
	THE COMMISSION PURSUANT TO A	THE COMMISSION PURSUANT TO A
	REQUEST FOR	REQUEST FOR
	CONFIDENTIAL	CONFIDENTIAL
	TREATMENT] 2013	TREATMENT] 2013
	<u>Aircraft</u>	<u>Aircraft</u>
Seats	[CONFIDENTIAL	[CONFIDENTIAL
Galleys/Furnishings	PORTION OMITTED	PORTION OMITTED
	AND FILED	AND FILED
Antennas & Mounting Equipment	SEPARATELY WITH	SEPARATELY WITH
Avionics	THE COMMISSION PURSUANT TO A	THE COMMISSION PURSUANT TO A
Cabin Systems Equipment	REQUEST FOR	REQUEST FOR
	CONFIDENTIAL	CONFIDENTIAL
Miscellaneous Emergency Equipment	TREATMENT]	TREATMENT]
Textiles/Raw Material		
<u>Item</u>	Preliminary On-Dock Dates	
	[CONFIDENTIAL	[CONFIDENTIAL
	PORTION OMITTED  AND FILED	PORTION OMITTED  AND FILED
	SEPARATELY WITH	SEPARATELY WITH
	THE COMMISSION	THE COMMISSION
	PURSUANT TO A	PURSUANT TO A
	REQUEST FOR	REQUEST FOR
	CONFIDENTIAL	CONFIDENTIAL
	TREATMENT] 2013	TREATMENT] 2013
	<u>Aircraft</u>	<u>Aircraft</u>
Seats	[CONFIDENTIAL	[CONFIDENTIAL
Galleys/Furnishings	PORTION OMITTED	
Gancy3/1 armsmigs		PORTION OMITTED
Antennas & Mounting Equipment	AND FILED SEPARATELY WITH	AND FILED SEPARATELY WITH
Antennas & Mounting Equipment	AND FILED	AND FILED
Antennas & Mounting Equipment Avionics	AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A	AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A
Antennas & Mounting Equipment	AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR	AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR
Antennas & Mounting Equipment Avionics	AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL	AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL
Antennas & Mounting Equipment Avionics Cabin Systems Equipment	AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR	AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR

Item **Preliminary On-Dock Dates** [CONFIDENTIAL [CONFIDENTIAL PORTION OMITTED PORTION OMITTED AND FILED AND FILED SEPARATELY WITH SEPARATELY WITH THE COMMISSION THE COMMISSION PURSUANT TO A PURSUANT TO A REQUEST FOR REQUEST FOR CONFIDENTIAL **CONFIDENTIAL** TREATMENT] 2013 TREATMENT] 2013 **Aircraft Aircraft** [CONFIDENTIAL Seats PORTION OMITTED Galleys/Furnishings AND FILED Antennas & Mounting Equipment SEPARATELY WITH THE COMMISSION Avionics PURSUANT TO A Cabin Systems Equipment REQUEST FOR

### 3. <u>Additional Delivery Requirements - Import.</u>

Textiles/Raw Material

Miscellaneous Emergency Equipment

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1980 BFE1-2 SA-26

CONFIDENTIAL

Page 6

TREATMENT]

1

Supplemental Agreement No. 35

n

Purchase Agreement No. 1977

between

The Boeing Company

and

American Airlines, Inc.

Relating to Boeing Model 737-800 Aircraft

THIS SUPPLEMENTAL AGREEMENT, entered into as of , 2011, (Supplemental Agreement Number 35) by and between THE BOEING COMPANY, a Delaware corporation with offices in Seattle, Washington, (Boeing) and AMERICAN AIRLINES, INC., a Delaware corporation with offices in Fort Worth, Texas, together with its successors and permitted assigns (Customer);

WHEREAS, Boeing and Customer entered into Purchase Agreement No. 1977 dated October 31, 1997, relating to Boeing Model 737-823 aircraft, as amended and supplemented (the "Purchase Agreement") and capitalized terms used herein without definitions shall have the meanings specified therefore in such Purchase Agreement;

WHEREAS, pursuant to Letter Agreement No. 6-1162-AKP-075 titled Aircraft Purchase Rights and Substitution Rights (the "Rights Letter"), Boeing and Customer have agreed to, among other things, the treatment of aircraft Purchase Rights;

WHEREAS, pursuant to Business Offer AAL-PBO-1105157R3, Boeing and Customer, among other things, agreed to exercise one hundred (100) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Rights Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

WHEREAS, pursuant to Business Offer AAL-PBO-1105157R3, the parties agreed to enter into this supplemental agreement in order to incorporate the terms and conditions of such Business Offer into the Purchase Agreement;

WHEREAS, such 100 Aircraft referred to above, other than those that have previously been exercised, are hereby exercised and all such 100 Aircraft are now

P.A. No. 1977 SA-35 AAL

considered Firm Aircraft as defined in the Rights Letter and no further Purchase Agreement Supplements or supplemental agreements will be required to be executed in regard to these 100 Aircraft;

WHEREAS, Boeing and Customer agree that for all [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and

WHEREAS, Boeing and Customer agree that for all Aircraft[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

WHEREAS, Boeing agrees to develop and make formal offerability of a 737-900ER configuration in accordance with Letter Agreement No. AAL-PA-1977-LA-1105616 entitled Open Configuration Matters;

WHEREAS, Boeing and Customer have agreed that Customer will retain [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

NOW THEREFORE, in consideration of the mutual covenants herein contained, the parties agree to amend the Purchase Agreement as follows:

- 1. The Table of Contents is removed in its entirety and replaced with a revised Table of Contents, attached hereto, to reflect the changes made by this Supplemental Agreement Number 35.
- 2. Table 1F entitled Aircraft Delivery, Description, Price and Advance Payments Purchased Aircraft is added and is hereby made part of the Purchase Agreement.
- 3. Supplemental Exhibit BFE1 entitled Buyer Furnished Equipment Variables is deleted in its entirety and replaced with a revised Supplemental Exhibit BFE1, attached hereto, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4. Rights Letter is removed and revised Letter Agreement No. 6-1162-AKP-075R1, attached hereto, is substituted in lieu thereof to:
  - a. Revise certain matters regarding the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1977 SA-35 AAL

- b. Revise Attachment A and Attachments A-1 though A-4 to reflect, among other things, the addition of the 737-900ER aircraft pricing;
- c. Revise Attachment B to reflect [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Supplemental Agreement Number 35; and
- d. Delete Attachment C in its entirety, as there are no further [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 5. Letter Agreement No. AAL-PA-1977-LA-1105271 entitled Advance Payments and Permitted Transactions 4, attached hereto, is added to set forth, among other things, certain matters related to Advance Payments with respect to Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6. Letter Agreement No. AAL-PA-1977-LA-1105272 entitled Business Considerations 2, attached hereto, is added to set forth certain business considerations for Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7. Letter Agreement No. AAL-PA-1977-LA-1105509 entitled Aircraft Performance Guarantees 737-700 attached hereto is hereby added to set forth certain matters related to the performance guarantees applicable to a 737-723 aircraft that may be purchased by Customer.
- 8. Letter Agreement No. AAL-PA-1977-LA-1105511 entitled Aircraft Performance Guarantees 737-900ER attached hereto is hereby added to set forth certain matters related to the performance guarantees applicable to a 737-923ER aircraft that may be purchased by Customer.
- 9. Letter Agreement No. AAL-PA-1977-LA-1105863 entitled Performance Guarantees for Rights Aircraft is added to set forth certain matters related to the performance guarantees applicable to the 737-823 Rights Aircraft.
- 10. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1977 AAL SA-35

3

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- b. Revise Attachment B to reflect, among other things, the addition of the 737-723 and 737-923ER aircraft.
- 11. Letter Agreement No. AAL-PA-1977-LA-1105616 entitled Open Configuration Matters, attached hereto, is added to reflect certain matters concerning the configuration of 737-723 and 737-923ER aircraft.
- 12. As a result of Letter Agreement No. AAL-PA-1977-LA-1105271 entitled Advance Payments and Permitted Transactions 4, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

EXECUTED IN DUPLICATE as of the day and year first above written.

AAL

THE BOEING COMPANY	AMERICAN AIRLINES, INC.	
By:	By:	
Its: Attorney-In-Fact	Its:	
P.A. No. 1977	SA-35	4

### TABLE OF CONTENTS

ARTICLES		NUMBER
1.	Quantity, Model and Description	SA-21
2.	Delivery Schedule	
3.	Price	
4.	Payment	
5.	Miscellaneous	
TABLE		
1	Aircraft Delivery, Description, Price and Advance Payments	SA-29
1A	Aircraft Delivery, Description, Price and Advance Payments – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-33
1B	Aircraft Delivery, Description, Price and Advance Payments – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-33
1C	Aircraft Delivery, Description, Price and Advance Payments – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	SA-33
1D	Aircraft Delivery, Description, Price and Advance Payments – Boeing Sky Interior Aircraft	SA-33
1E	Aircraft Delivery, Description, Price and Advance Payments – SA-34 Exercised Aircraft	SA-34
1F	Aircraft Delivery, Description, Price and Advance Payments – Purchased Aircraft	SA-35
PA No. 1977	i	SA-35

EXHIBITS	<u>3</u>			
A.	Aircraft Configuration			
A1	Aircraft Configuration		SA-33	
A2	Aircraft Configuration		SA-33	
В	Aircraft Delivery Requirements and Responsibilities			
С	Defined Terms			
SUPPLEM	SUPPLEMENTAL EXHIBITS			
AE1	Escalation Adjustment Airframe and Optional Features		SA-20	
BFE1	BFE Variables		SA-35	
CS1	Customer Support Variables			
SLP1	Service Life Policy Components			
EE1	Engine Escalation, Engine Warranty and Patent Indemnity			
PA No.	1977	ii	SA-35	

LETTER AGREEMENTS		
6-1162-AKP-070	Miscellaneous Commitments for Model 737, 757, 767 and 777 Aircraft	
6-1162-AKP-072R1	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	
6-1162-AKP-073	Accident Claims and Litigation	
6-1162-AKP-074R2	Business Considerations	
6-1162-AKP-075 <b>R1</b>	Aircraft Purchase Rights and Substitution Rights - Attachment A - Attachment B - Attachment C	SA-35 SA-35 SA-35 SA-35
6-1162-AKP-076	Aircraft Performance Guarantees	
6-1162-AKP-077	Spares Matters	
6-1162-AKP-078	Model 737 Miscellaneous Commitments	
6-1162-AKP-079	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Sharing	
6-1162-AKP-080	Installation of Cabin Systems Equipment	
6-1162-AKP-081	Model 737 Maintenance Cost Commitment	
6-1162-AKP-082	Confidentiality	
6-1162-AKP-083	Model 737 Introduction Cost Commitment	
6-1162-AKP-084	Performance Retention Commitment	
6-1162-AKP-085	Component Reliability Commitments	
6-1162-AKP-117	Delivery Schedule	
6-1162-SSM-1405 <b>R1</b>	Multiple Operating Weight Program Attachment B	SA-35 SA-35
6-1162-CLO-1035	Aircraft Performance Guarantees	SA-28
PA No. 1977	iii	SA-35

AAL-PA-1977-LA-1105509	Aircraft Performance Guarantees, 737-700	SA-35
AAL-PA-1977-LA-1105511	Aircraft Performance Guarantees, 737-900ER	SA-35
6-1162-CLO-1038	Advance Payments and Permitted Transactions	SA-31
6-1162-CLO-1082	Advance Payments and Permitted Transactions 2	SA-32
AAL-PA-1977-LA-01073	Advance Payments and Permitted Transactions 3	SA-34
AAL-PA-1977-LA-1105271	Advance Payments and Permitted Transactions 4	SA-35
AAL-PA-1977-LA-1105272	Business Considerations 2	SA-35
AAL-PA-1977-LA-1105616	Open Configuration Matters	SA-35
AAL-PA-1977-LA-1105863	Performance Guarantees for Rights Aircraft	SA-35
PA No. 1977	iv	SA-35

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

**Boeing Proprietary** 

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

**Boeing Proprietary** 

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

**Boeing Proprietary** 

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

**Boeing Proprietary** 

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

**Boeing Proprietary** 

# BUYER FURNISHED EQUIPMENT VARIABLES

between

# THE BOEING COMPANY

and

# AMERICAN AIRLINES, INC.

Supplemental Exhibit BFE1 to Purchase Agreement Number 1977

#### **BOEING PROPRIETARY**

### BUYER FURNISHED EQUIPMENT VARIABLES

relating to

#### **BOEING MODEL 737 AIRCRAFT**

This Supplemental Exhibit BFE1 contains vendor selection dates, on-dock dates and other variables applicable to the Aircraft.

### Supplier Selection.

Customer will:

1.1 Select and notify Boeing of the suppliers and part numbers of the following BFE items by the following dates:

Galley SystemCompleteGalley InsertsCompleteSeats (passenger)CompleteCabin Systems EquipmentCompleteMiscellaneous Emergency EquipmentCompleteCargo Handling SystemsComplete

For a new certification, supplier requires notification [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Cargo Handling System on-dock date.

#### **BOEING PROPRIETARY**

# 2. <u>On-dock Dates</u>

On or before [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT], Boeing will provide to Customer a BFE Requirements On-Dock/Inventory Document (BFE Document) or an electronically transmitted BFE Report which may be periodically revised, setting forth the items, quantities, on-dock dates and shipping instructions relating to the in-sequence installation of BFE. For planning purposes, a preliminary BFE on-dock schedule is set forth below:

<u>item</u>	Preliminary On-Dock Dates	
	[CONFIDENTIAL	[CONFIDENTIAL
	PORTION OMITTED	PORTION OMITTED
	AND FILED	AND FILED
	SEPARATELY WITH	SEPARATELY WITH
	THE COMMISSION	THE COMMISSION
	PURSUANT TO A	PURSUANT TO A
	REQUEST FOR	REQUEST FOR
	CONFIDENTIAL	CONFIDENTIAL
	TREATMENT]	TREATMENT]
	<u>Aircraft</u>	<u>Aircraft</u>
Seats		
Galleys/Furnishings	[CONFIDENTIAL	[CONFIDENTIAL
Antennas & Mounting Equipment	PORTION OMITTED	PORTION OMITTED
Avionics	AND FILED	AND FILED
Cabin Systems Equipment	SEPARATELY WITH	SEPARATELY WITH
Miscellaneous Emergency Equipment	THE COMMISSION	THE COMMISSION
Textiles/Raw Material	PURSUANT TO A	PURSUANT TO A
Cargo Systems	REQUEST FOR	REQUEST FOR
Provision Kits	CONFIDENTIAL	CONFIDENTIAL
Winglets	TREATMENT]	TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Item

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment

Miscellaneous Emergency Equipment

Textiles/Raw Material Cargo Systems Provision Kits Winglets

[CONFIDENTIAL [CONFIDENTIAL PORTION OMITTED PORTION OMITTED AND FILED AND FILED SEPARATELY WITH SEPARATELY WITH THE COMMISSION THE COMMISSION PURSUANT TO A PURSUANT TO A REQUEST FOR REQUEST FOR CONFIDENTIAL CONFIDENTIAL TREATMENT] TREATMENT]

Item

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Galleys/Furnishings Antennas & Mounting Equipment

Avionics
Cabin Systems Equipment

Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits

Winglets

Seats

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Item

Preliminary On-Dock Dates

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

Item

Provision Kits

Winglets

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

**[CONFIDENTIAL** 

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Item

Winglets

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Item

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Item

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

NT] T

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Seats

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

Item

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**[CONFIDENTIAL** 

PORTION OMITTED

Item

Seats

Provision Kits

Winglets

Winglets

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

<u>Aircraft</u>

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH

THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL

TREATMENT]

<u>Aircraft</u>

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Item

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

NT] T

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Seats

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment

Miscellaneous Emergency Equipment

Textiles/Raw Material Cargo Systems Provision Kits Winglets

[CONFIDENTIAL [CONFIDENTIAL PORTION OMITTED PORTION OMITTED AND FILED AND FILED SEPARATELY WITH SEPARATELY WITH THE COMMISSION THE COMMISSION PURSUANT TO A PURSUANT TO A REQUEST FOR REQUEST FOR CONFIDENTIAL CONFIDENTIAL TREATMENT] TREATMENT]

Item

Winglets

Seats

Preliminary On-Dock Dates

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL

TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

Item

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material

PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Textiles/Raw Materia Cargo Systems Provision Kits Winglets

Item

Preliminary On-Dock Dates

[CONFIDENTIAL

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

<u>Aircraft</u>

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Item

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

NT] T

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Seats

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats Galleys/Furnishings

Antennas & Mounting Equipment

Avionics

Winglets

Cabin Systems Equipment

Miscellaneous Emergency Equipment

Textiles/Raw Material Cargo Systems Provision Kits PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Item

Seats

Provision Kits

Winglets

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Item

Seats

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material

PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL

TREATMENT]

[CONFIDENTIAL

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Textiles/Raw M Cargo Systems Provision Kits Winglets

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

**Aircraft** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment

Miscellaneous Emergency Equipment

Textiles/Raw Material Cargo Systems Provision Kits Winglets [CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

Item

Seats

Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats
Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits
Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

<u>Aircraft</u>

[CONFIDENTIAL

**Aircraft** 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Item

Seats

Avionics

Galleys/Furnishings

Cabin Systems Equipment

Textiles/Raw Material

Cargo Systems

Provision Kits

Winglets

Antennas & Mounting Equipment

Miscellaneous Emergency Equipment

Preliminary On-Dock Dates

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Seats

Winglets

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

<u>Aircraft</u>

[CONFIDENTIAL

**Aircraft** 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Item

Seats

Avionics

Galleys/Furnishings

Cabin Systems Equipment

Textiles/Raw Material

Cargo Systems

Provision Kits

Winglets

Antennas & Mounting Equipment

Miscellaneous Emergency Equipment

Preliminary On-Dock Dates

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Aircraft

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Seats

Winglets

Galleys/Furnishings
Antennas & Mounting Equipment
Avionics
Cabin Systems Equipment
Miscellaneous Emergency Equipment
Textiles/Raw Material
Cargo Systems
Provision Kits

#### **BOEING PROPRIETARY**

<u>Item</u> Preliminary On-Dock Dates

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]
Aircraft

Seats

Galleys/Furnishings

Antennas & Mounting Equipment

Avionics

Cabin Systems Equipment

Miscellaneous Emergency Equipment

Textiles/Raw Material Cargo Systems Provision Kits

Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 3. Additional Delivery Requirements

Customer will insure that Customer's BFE suppliers provide sufficient information to enable Boeing, when acting as Importer of Record for Customer's BFE, to comply with all applicable provisions of the U.S. Customs Service.





6-1162-AKP-075R1

American Airlines, Inc. P.O. Box 619616

Dallas/Ft. Worth Airport, Texas 75261-9616

Subject: Aircraft Purchase Rights and Substitution Rights

Reference: Purchase Agreement No. 1977 between The Boeing Company and American Airlines, Inc. relating to Model 737-823 Aircraft

This letter agreement ("<u>Letter Agreement</u>") is entered into on the date below and constitutes a part of the above-referenced Purchase Agreement, as the same may hereafter be amended, modified or supplemented and including, without limitation, as part thereof the exhibits, appendices, schedules, attachments and letter agreements thereto (the "<u>737-823 Purchase Agreement</u>").

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1
Aircraft Purchase Rights and Substitution Rights

SA-35 Page 1



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1
Aircraft Purchase Rights and Substitution Rights

SA-35 Page 2



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1
Aircraft Purchase Rights and Substitution Rights

SA-35 Page 3



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1 Aircraft Purchase Rights and Substitution Rights SA-35 Page 4



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1 Aircraft Purchase Rights and Substitution Rights SA-35 Page 5



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1 Aircraft Purchase Rights and Substitution Rights SA-35

Page 6



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

5. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1
Aircraft Purchase Rights and Substitution Rights

SA-35 Page 7



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1 Aircraft Purchase Rights and Substitution Rights SA-35 Page 8



## [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**6.** [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1
Aircraft Purchase Rights and Substitution Rights

SA-35 Page 9



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

7. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1 Aircraft Purchase Rights and Substitution Rights SA-35 Page 10



8. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1
Aircraft Purchase Rights and Substitution Rights

SA-35 Page 11



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 9. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- ${f 10.}$  [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 11. <u>Confidential Treatment</u>. Customer and Boeing understand that certain commercial and financial information contained in this Letter Agreement are considered by Boeing and Customer as confidential. Customer and Boeing agree that each will treat this Letter Agreement and the information contained herein as confidential and will not, without the prior written consent of the other, disclose this Letter Agreement or any information contained herein to any other person or entity, except as provided in this Letter Agreement or in the Applicable Purchase Agreements.

6-1162-AKP-075R1
Aircraft Purchase Rights and Substitution Rights

SA-35

Page 12



Very truly yours,

THE BOEING COMPANY

By	
Its	Attorney-In-Fact
ACCEPTED AND AGREED TO this	
Date:	, 2011
AMERICAN AIRLINES, INC.	
Ву	
Its	

Attachment A: [CONFIDENTIAL PORTION Attachment B: OMITTED AND FILED Attachment C: SEPARATELY WITH THE

Attachment D: COMMISSION PURSUANT TO A REQUEST Attachment E: FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-075R1 Aircraft Purchase Rights and Substitution Rights

BOEING PROPRIETARY

SA-35 Page 13 Attachment A. to 6-1162-AKP-075R1

 $[{\tt CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]}$ 

PA No. 1977

SA No. 35 Boeing Proprietary

# Attachment A-1 to Letter Agreement 6-1162-AKP-075R1 737-923ER Eligible Model Description and Price

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

PA No. 1977 Boeing Proprietary Page 1 of 1

# Attachment A-2 to Letter Agreement 6-1162-AKP-075R1 737-823 Eligible Model Description and Price

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

PA No. 1977 **Boeing Proprietary** Page 1 of 1 SA-35

# Attachment A-3 to Letter Agreement 6-1162-AKP-075R1 737-723 Eligible Model Description and Price

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

PA No. 1977 Boeing Proprietary Page 1 of 1

# Attachment A-4 to Letter Agreement 6-1162-AKP-075R1 737-623 Eligible Model Description and Price

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

PA No. 1977 Boeing Proprietary Page 1 of 1

### Attachment B to Letter Agreement 6-1161-AKP-075R1 MADP Rights Aircraft Delivery Months and Exercise Dates

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

SA35 Page 1 of 1



The Boeing Company P.O. Box 3707 Seattle, WA 98124-2207

SA-35

6-1162-SSM-1405R1

American Airlines, Inc. P.O. Box 619616

Dallas-Fort Worth Airport, Texas 75261-9616

Subject:

Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

References:

- a) Customer Services General Terms Agreement No. 23-1 (the CSGTA) between The Boeing Company (Boeing) and American Airlines, Inc. (Customer)
- b) Purchase Agreement No. 1977 (Purchase Agreement), dated October 31, 1997, as amended and supplemented, between Boeing and Customer
- c) Aircraft General Terms Agreement AGTA-AAL (AGTA), dated October 31, 1997, as amended and supplemented, between Boeing and Customer

This Order incorporates the terms and conditions of the CSGTA. All terms used but not defined in this Order have the same meaning as in the CSGTA. In this Order, the term "Aircraft" means the Model 737-723, -823, and -923ER aircraft that are listed in Attachment B.

#### 1. Description of Agreement

This Order represents the agreement of the parties to the terms and conditions under which Boeing will provide, for the Aircraft, a "Service" as defined in Article 2.30 of Part 1 of the CSGTA. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1977 Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft



#### 2. Multiple Operating Weights

Customer has selected [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The range of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 3. Flight Manuals

### 3.1 Delivery of Airplane Flight Manuals

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 3.2 Flight Manual Revisions

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1977 Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

**BOEING PROPRIETARY** 



#### 4. Administrative Fees

4.1. New Sub-Fleet Program Initialization

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

4.2 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Sub-Fleets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

4.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Sub-Fleet

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

4.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1977 Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

**BOEING PROPRIETARY** 



#### 5. Aircraft Included in the Program

#### 5.1 New Aircraft

The configuration of the Aircraft (set forth in Exhibit A to the Purchase Agreement) for Aircraft to be delivered by Boeing to Customer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

5.3 Other Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1977 Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

**BOEING PROPRIETARY** 



### 5.4 Aircraft on Operating Lease

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 6. Projected TOW and Actual TOW

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1977 Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

**BOEING PROPRIETARY** 



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

7. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

7.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

**BOEING PROPRIETARY** 



- 7.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 8. Price [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and Payment
- 8.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - 8.2 Boeing will provide an invoice for any payments due to Boeing and such payments will be made under the terms of this Order and the CSGTA.

### 9. Protection of Asset Value

As long as the Aircraft are in the legal possession of Customer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1977 Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

**BOEING PROPRIETARY** 



[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

10. Default

P.A. No. 1977

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

ICONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

11. Term of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

**BOEING PROPRIETARY** 



12. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 13. Notices

The ATOW reported annually and the notices related to the sale, lease or other disposition of Aircraft subject to this Program will be addressed as follows:

Boeing Commercial Airplane Group

P.O. Box 3707

Seattle, Washington 98124-2207

U.S.A.

Attention: Director

Aircraft Contracts

MC 21-43

Attention: Director

Airline Analysis Marketing

MC 21-43

P.A. No. 1977 Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

**BOEING PROPRIETARY** 



All notices to Customer required by this Order will be sent in the manner required by the CSGTA and with a copy to:

American Airlines, Inc. 4333 Amon Carter Boulevard MD 5423 HDQ Fort Worth, Texas 76155

Attention: Managing Director

**Operations Engineering** 

and

American Airlines, Inc. 4333 Amon Carter Boulevard MD 5569 HDQ Fort Worth, Texas 76155

Attention: Vice President, Corporate Development and Treasurer

#### 14. Performance Guarantees

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

15. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

16. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 16.1. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 16.2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1977 Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

BOEING PROPRIETARY



# 16.3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 17. Confidentiality

P.A. No. 1977

THE BOEING COMPANY

The terms and conditions of this Order and the reports required hereunder shall be considered to be confidential and shall not be disclosed by either party (except as reasonably necessary to its respective employees, insurers, auditors or professional advisors) without the prior written consent of the other party.

Please indicate your acceptance of this Order by returning one (1) executed copy to the attention of the undersigned.

By	
Its	Attorney-In-Fact
ACCEP	TED AND AGREED TO this
Date:	, 2011
AMERI	CAN AIRLINES, INC.
Ву	
Its	
Custom	er's Purchase Order Number:

Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

BOEING PROPRIETARY

American Airlines, Inc. 6-1162-SSM-1405 Page 1

# TABLE 1 – MODEL 737 AIRCRAFT [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

P.A. No. 1977 Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

SA-35

#### **Definitions**

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 1. The configuration of the Aircraft in Exhibit A to the Purchase Agreement;
- 2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and
- 4. The certified weight of the Aircraft at Delivery.

# <u>Data Requirements for the Calculation of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]</u>

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1977 Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

BOEING PROPRIETARY



# Data Requirements for the Calculation of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# Calculation of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Step 1:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Step 2:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### Where:

P.A. No. 1977

- $P_1 = [CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$
- P<sub>2</sub> = [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- $P_3 = [CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$
- $F_1 = [CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$
- $F_2 = [CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$
- $F_3 = [CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$

Multiple Operating Weight Program, Model 737-723/-823/-923ER Aircraft

**BOEING PROPRIETARY** 

# Attachment B1

**CSGTA Order Multiple Operating Weight (MOW)** 

 $[{\tt CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT}]$ 

Attachment B1
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Attachment B1
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Attachment B1
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Attachment B1
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Attachment B1
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Attachment B1
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Attachment B1
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Attachment B2
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Attachment B3
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Attachment B2
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Attachment B3
[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]



The Boeing Company P.O. Box 3707 Seattle, WA 98124-2207

AAL-PA-1977-LA-1105509

American Airlines, Inc. P.O. Box 619616

Dallas-Fort Worth Airport, Texas 75261-9616

Subject: Aircraft Performance Guarantees – 737-723

Reference: Purchase Agreement No. PA-1977 (Purchase Agreement) between The Boeing Company (Boeing) and American Airlines, Inc. (Customer)

relating to Model 737-823 aircraft

This letter agreement (**Letter Agreement**) amends and supplements the Purchase Agreement. All terms used but not defined in this Letter Agreement shall have the same meaning as in the Purchase Agreement.

Boeing agrees to provide Customer with the performance guarantees in the Attachment. These guarantees are exclusive and expire upon delivery of the 737-723 aircraft (Aircraft) to Customer. Notwithstanding the provision of the performance guarantees in the Attachment, Boeing and Customer will work together in good faith to communicate the performance guarantees that will be offered [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2. Confidential Treatment.

The information contained herein represents confidential business information and has value precisely because it is not available generally or to other parties. Customer will limit the disclosure of its contents to employees of Customer with a need to know the contents for purposes of helping Customer perform its obligations under the Purchase Agreement and who understand they are not to disclose its contents to any other person or entity without the prior written consent of Boeing.

AAL-PA-1977-LA-1105509 Performance Guarantees, 737-700

Page 1

Very truly yours,

THE BOEING COMPANY

By

Its Attorney-In-Fact

ACCEPTED AND AGREED TO this

Date: , 2011

American Airlines, Inc.

By

AAL-PA-1977-LA-1105509 Performance Guarantees, 737-700

Its

**BOEING PROPRIETARY** 

Page 2

# MODEL 737-700 WITH WINGLETS PERFORMANCE GUARANTEES FOR AMERICAN AIRLINES, INC.

SECTION	CONTENTS
1	AIRCRAFT MODEL APPLICABILITY
2	FLIGHT PERFORMANCE
3	MANUFACTURER'S EMPTY WEIGHT
4	SOUND LEVELS
5	AIRCRAFT CONFIGURATION
6	GUARANTEE CONDITIONS
7	GUARANTEE COMPLIANCE
8	EXCLUSIVE GUARANTEES

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364 BOEING PROPRIETARY

### 1 AIRCRAFT MODEL APPLICABILITY

The guarantees contained in this Attachment (the "Performance Guarantees") are applicable to the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2 FLIGHT PERFORMANCE

#### 2.1 Takeoff

2.1.1 The FAA approved takeoff field length at a gross weight at the start of the ground roll of 154,500 pounds, at a temperature of 86°F, at a sea level altitude, with an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord, and using maximum takeoff thrust, shall not be more than the following guarantee value:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2.1.2 The FAA approved takeoff gross weight at the start of ground roll, at a temperature of 95°F, at an altitude of 607 feet and satisfying the conditions defined below, with an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord, and using maximum takeoff thrust, shall not be less than the following guarantee value:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### Conditions:

The takeoff runway available (TORA) is 12,816 feet.

The takeoff distance available (TODA) is 12,816 feet.

The accelerate-stop distance available (ASDA) is 12,816 feet.

The runway slope is 0.03 percent downhill.

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

The following obstacle definition is based on a straight-out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

	Distance	Height
1.	2,650 feet	56 feet
2.	51,154 feet	107 feet

2.1.3 The FAA approved takeoff gross weight at the start of ground roll, at a temperature of 86°F, at an altitude of 56 feet and satisfying the conditions defined below, with an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord, and using maximum takeoff thrust, shall not be less than the following guarantee value:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### Conditions:

The takeoff runway available (TORA) is 5,701 feet.

The takeoff distance available (TODA) is 5,701 feet.

The accelerate-stop distance available (ASDA) is 5,701 feet.

The runway slope is 0.16 percent uphill.

The following obstacle definition is based on a straight-out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

	Distance	Height
1.	1,096 feet	18 feet
2.	2,141 feet	43 feet
3.	2.757 feet	63 feet

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

### 2.2 Landing

The FAA approved landing field length at a gross weight of 129,200 pounds and at a sea level altitude, shall not be more than the following guarantee value:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2.3 Mission

### 2.3.1 Mission Payload

The payload for a stage length of 2,990 nautical miles in still air (equivalent to a distance of 2,417 nautical miles with an 87 knot headwind, representative of a Boston to Los Angeles route in Winter) using the conditions and operating rules defined below, shall not be less than the following guarantee value:

NOMINAL: [CONFIDENTIAL PORTION OMITTED AND FILED TOLERANCE: SEPARATELY WITH THE COMMISSION PURSUANT TO A

GUARANTEE: REQUEST FOR CONFIDENTIAL TREATMENT]

Conditions and operating rules:

Stage Length: The stage length is defined as the sum of the distances for the climbout maneuver, climb, cruise, and descent.

Takeoff: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

The airport temperature is 41°F.

The takeoff runway available (TORA) is 9,883 feet.

The takeoff distance available (TODA) is 9,883 feet.

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

AERO-B-BBA4-M11-0668 SS11-0364

The accelerate-stop distance available (ASDA) is 9,883 feet.

The runway slope is 0.06 percent uphill.

The following obstacle definition is based on a straight out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

	Distance	Height
1.	1,904 feet	57 feet
2.	2,215 feet	65 feet
3.	28,053 feet	508 feet

Takeoff performance is based on an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord.

Maximum takeoff thrust is used for the takeoff.

The takeoff gross weight shall conform to FAA Regulations.

Climbout Maneuver: Following the takeoff to 35 feet, the Aircraft accelerates to 250 KCAS while climbing to 1,500 feet above the departure airport altitude and retracting flaps and landing gear.

Climb:

The Aircraft climbs from 1,500 feet above the departure airport altitude to 10,000 feet altitude at 250 KCAS.

The Aircraft then accelerates at a rate of climb of 500 feet per minute to the recommended climb speed for minimum block fuel.

The climb continues at the recommended climb speed for minimum block fuel until 0.78 Mach number is reached.

The climb continues at 0.78 Mach number to the initial cruise altitude.

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

The temperature is standard day during climb.

Maximum climb thrust is used during climb.

Cruise: The Aircraft cruises at the Long Range Cruise (LRC) speed.

The initial cruise altitude is 36,000 feet.

A step climb or multiple step climbs of 2,000 feet altitude may be used when beneficial to minimize fuel burn.

The temperature is standard day during cruise.

The cruise thrust is not to exceed maximum cruise thrust except during a step climb when maximum climb thrust may be

used.

Descent: The Aircraft descends from the final cruise altitude at 0.78 Mach number until 250 KCAS is reached.

The descent continues at 250 KCAS to an altitude of 1,500 feet above the destination airport altitude.

Throughout the descent, the cabin pressure will be controlled to a maximum rate of descent equivalent to 300 feet per minute

at sea level.

The temperature is standard day during descent.

Approach and Landing Maneuver:

The Aircraft decelerates to the final approach speed while extending landing gear and flaps, then descends and lands.

The destination airport altitude is 126 feet.

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

BOEING PROPRIETARY

SS11-0364

Fixed Allowances:

For the purpose of this guarantee and for the purpose of establishing compliance with this guarantee, the following shall be used as fixed quantities and allowances:

Taxi-Out:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Takeoff and Climbout Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Approach and Landing Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Taxi-In (shall be consumed from the reserve fuel):

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Usable reserve fuel remaining upon completion of the approach and landing maneuver: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2.3.2 Mission Payload

The payload for a stage length of 1,676 nautical miles in still air (equivalent to a distance of 1,636 nautical miles with an 11 knot

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

headwind, representative of a Quito to Miami route in Winter) using the conditions and operating rules defined below, shall not be less than the following guarantee value:

NOMINAL: [CONFIDENTIAL PORTION OMITTED AND FILED

TOLERANCE: SEPARATELY WITH THE COMMISSION PURSUANT TO A

GUARANTEE: REQUEST FOR CONFIDENTIAL TREATMENT]

Conditions and operating rules:

Stage Length: The stage length is defined as the sum of the distances for the climbout maneuver, climb, cruise, and descent.

Takeoff: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

The airport temperature is 63°F.

The takeoff runway available (TORA) is 10,236 feet. The takeoff distance available (TODA) is 10,236 feet.

The accelerate-stop distance available (ASDA) is 10,236 feet.

The runway slope is 0.18 percent uphill.

The following obstacle definition is based on a straight out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

Distance Height

1. 49,764 feet 620 feet

Takeoff performance is based on an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord.

Maximum takeoff thrust is used for the takeoff.

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

The takeoff gross weight shall conform to FAA Regulations.

Climbout

Following the takeoff to 35 feet, the Aircraft accelerates to 250 KCAS while climbing to 1,500 feet above the departure

Maneuver:

airport altitude and retracting flaps and landing gear.

Climb:

The Aircraft then accelerates at a rate of climb of 500 feet per minute to the recommended climb speed for minimum block

fuel.

The climb continues at the recommended climb speed for minimum block fuel until 0.78 Mach number is reached.

The climb continues at 0.78 Mach number to the initial cruise altitude.

The temperature is standard day during climb.

Maximum climb thrust is used during climb.

Cruise:

The Aircraft cruises at the Long Range Cruise (LRC) speed.

The initial cruise altitude is 36,000 feet.

A step climb or multiple step climbs of 2,000 feet altitude may be used when beneficial to minimize fuel burn.

The temperature is standard day during cruise.

The cruise thrust is not to exceed maximum cruise thrust except during a step climb when maximum climb thrust may be

used.

Descent:

The Aircraft descends from the final cruise altitude at 0.78 Mach number until 250 KCAS is reached.

The descent continues at 250 KCAS to an altitude of 1,500 feet above the destination airport altitude.

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

Throughout the descent, the cabin pressure will be controlled to a maximum rate of descent equivalent to 300 feet per minute

at sea level

The temperature is standard day during descent.

Approach and Landing Maneuver:

The Aircraft decelerates to the final approach speed while extending landing gear and flaps, then descends and lands.

The destination airport altitude is 8 feet.

Fixed Allowances: For the purpose of this guarantee and for the purpose of establishing compliance with this guarantee, the following shall be

used as fixed quantities and allowances:

Taxi-Out:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Takeoff and Climbout Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Approach and Landing Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

Taxi-In (shall be consumed from the reserve fuel):

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Usable reserve fuel remaining upon completion of the approach and landing maneuver: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2.3.3 Mission Block Fuel

The block fuel for a stage length of 1,200 nautical miles in still air with a 27,840 pound payload using the conditions and operating rules defined below, shall not be more than the following guarantee value:

NOMINAL: [CONFIDENTIAL PORTION OMITTED AND FILED TOLERANCE: SEPARATELY WITH THE COMMISSION PURSUANT TO A

GUARANTEE: REQUEST FOR CONFIDENTIAL TREATMENT]

Conditions and operating rules:

Stage Length: The stage length is defined as the sum of the distances for the climbout maneuver, climb, cruise, and descent.

Block Fuel: The block fuel is defined as the sum of the fuel used for taxi-out, takeoff and climbout maneuver, climb, cruise, descent,

approach and landing maneuver, and taxi-in.

P.A. No. PA-1977

AERO-B-BBA4-M11-0668 SS11-0364

Takeoff: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

The takeoff gross weight is not limited by the airport conditions.

Climbout Following the takeoff to 35 feet, the Aircraft accelerates to 250 KCAS while climbing to 1,500 feet above the departure

Maneuver: airport altitude and retracting flaps and landing gear.

Climb: The Aircraft climbs from 1,500 feet above the departure airport altitude to 10,000 feet altitude at 250 KCAS.

The Aircraft then accelerates at a rate of climb of 500 feet per minute to the recommended climb speed for minimum block

fuel.

The climb continues at the recommended climb speed for minimum block fuel until 0.78 Mach number is reached.

The climb continues at 0.78 Mach number to the initial cruise altitude.

The temperature is standard day during climb.

Maximum climb thrust is used during climb.

Cruise: The Aircraft cruises at the Long Range Cruise (LRC) speed.

The initial cruise altitude is 38,000 feet.

A step climb or multiple step climbs of 2,000 feet altitude may be used when beneficial to minimize fuel burn.

The temperature is standard day during cruise.

P.A. No. PA-1977

AERO-B-BBA4-M11-0668 SS11-0364

The cruise thrust is not to exceed maximum cruise thrust except during a step climb when maximum climb thrust may be

used.

Descent: The Aircraft descends from the final cruise altitude at 0.78 Mach number until 250 KCAS is reached.

The descent continues at 250 KCAS to an altitude of 1,500 feet above the destination airport altitude.

Throughout the descent, the cabin pressure is controlled to a maximum rate of descent equivalent to 300 feet per minute at

sea level.

The temperature is standard day during descent.

Approach and Landing Maneuver:

The Aircraft decelerates to the final approach speed while extending landing gear and flaps, then descends and lands.

The destination airport altitude a sea level.

Fixed Allowances: For the purpose of this guarantee and for the purpose of establishing compliance with this guarantee, the following shall be

used as fixed quantities and allowances:

Operational Empty Weight, OEW (Paragraph 2.3.5): [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY

WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Taxi-Out:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO

A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

Takeoff and Climbout Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Approach and Landing Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Taxi-In (shall be consumed from the reserve fuel):

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Usable reserve fuel remaining upon completion of the approach and landing maneuver: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2.3.4 Operational Empty Weight Basis

The Operational Empty Weight (OEW) derived in Paragraph 2.3.5 is the basis for the mission guarantees of Paragraphs 2.3.1 and 2.3.2.

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

2.3.5 737-700 Weight Summary - American Airlines

## **Standard Model Specification MEW**

Pounds

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] American Airlines

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

American Airlines Manufacturer's Empty Weight (MEW)

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Standard and Operational Items Allowance (Paragraph 2.3.6)

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

BOEING PROPRIETARY

SS11-0364

**American Airlines Operational Empty Weight (OEW)** 

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

[CONFIDENTIAL	[CONFIDENTIAL	[CONFIDENTIAL
PORTION	PORTION	PORTION
OMITTED AND	OMITTED AND	OMITTED AND
FILED	FILED	FILED
SEPARATELY	SEPARATELY	SEPARATELY
WITH THE	WITH THE	WITH THE
COMMISSION	COMMISSION	COMMISSION
PURSUANT TO	PURSUANT TO	PURSUANT TO
A REQUEST FOR	A REQUEST FOR	A REQUEST FOR
CONFIDENTIAL	CONFIDENTIAL	CONFIDENTIAL
TREATMENT]	TREATMENT]	TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

# 2.3.6 Standard and Operational Items Allowance

Qty Pounds Pounds Pounds

# Standard Items Allowance

Unusable Fuel

Oil

Oxygen Equipment

Passenger Portable

Crew Masks

Miscellaneous Equipment

Crash Axe

Megaphones

Flashlights

Smoke Hoods

Galley Structure & Fixed Inserts

# **Operational Items Allowance**

Crew and Crew Baggage

Flight Crew (incl. baggage)

Cabin Crew (incl. baggage)

Catering Allowance & Removable Inserts

First Class

Tourist Class

Passenger Service Equipment

Potable Water - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE

COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Waste Tank Disinfectant

**Emergency Equipment** 

Escape Slides - Forward

Escape Slides - Aft

Life Vests - Crew and Passengers

Life Rafts

Auto Radio Beacon (ELT)

P.A. No. PA-1977

AERO-B-BBA4-M11-0668

[CONFIDENTIAL PORTION
OMITTED AND FILED
SEPARATELY WITH THE
COMMISSION PURSUANT TO A
REQUEST FOR CONFIDENTIAL
TREATMENT]

SS11-0364

## **Total Standard and Operational Items Allowance**

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 3 MANUFACTURER'S EMPTY WEIGHT

The Manufacturer's Empty Weight (MEW) is guaranteed not to exceed the value in Section 03-60-00 of Detail Specification TBD plus one percent.

- 4 SOUND LEVELS
- 4.1 Community Sound Levels

The Aircraft shall be certified in accordance with the requirements of 14CFR Part 36, Stage 4 and ICAO Annex 16, Volume 1, Chapter 4.

- **4.2** [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] **Single Event Noise Exposure Levels**
- **4.2.1** [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

4.2.2 The guarantee of Paragraph 4.2.1 is based on the following conditions, procedures and microphone locations:

> [CONFIDENTIAL PORTION OMITTED AND

REQUEST FOR CONFIDENTIAL TREATMENT]

REQUEST FOR CONFIDENTIAL TREATMENT]

REQUEST FOR CONFIDENTIAL TREATMENT]

**FILED** 

SEPARATELY WITH THE COMMISSION

PURSUANT TO A REQUEST FOR

CONFIDENTIAL TREATMENT]

[CONFIDENTIAL **PORTION** 

OMITTED AND

**FILED** 

**SEPARATELY** 

WITH THE COMMISSION

PURSUANT TO A REQUEST FOR CONFIDENTIAL

TREATMENT]

[CONFIDENTIAL **PORTION** 

OMITTED AND

**FILED** 

**SEPARATELY** 

WITH THE

COMMISSION

PURSUANT TO A

REQUEST FOR

CONFIDENTIAL

TREATMENT]

P.A. No. PA-1977

AERO-B-BBA4-M11-0668

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

SS11-0364

## 4.3 Interior Sound Levels in Flight

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 4.4 Ramp Sound Levels

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 5 AIRCRAFT CONFIGURATION

- The guarantees contained in this Attachment are based on the Aircraft configuration as defined in the original release of Detail Specification TBD (hereinafter referred to as the Detail Specification). Appropriate adjustment shall be made for changes in such Detail Specification approved by the Customer and Boeing or otherwise allowed by the Purchase Agreement which cause changes to the flight performance, sound levels, and/or weight and balance of the Aircraft. Such adjustment shall be accounted for by Boeing in its evidence of compliance with the guarantees.
- 5.2 The guarantee payloads of Paragraph 2.3.1 and 2.3.2 will be adjusted by Boeing for the effect of the following on OEW and the Manufacturer's Empty Weight guarantee of Section 3 will be adjusted by Boeing for the following in its evidence of compliance with the guarantees:
  - (1) Changes to the Detail Specification or any other changes mutually agreed upon between the Customer and Boeing or otherwise allowed by the Purchase Agreement.
  - (2) The difference between the component weight allowances given in Appendix IV of the Detail Specification and the actual weights.

## 6 GUARANTEE CONDITIONS

**6.1** All guaranteed performance data are based on the International Standard Atmosphere (ISA) and specified variations therefrom; altitudes are pressure altitudes.

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

- The Federal Aviation Administration (FAA) regulations referred to in this Attachment are, unless otherwise specified, the 737-700 Certification Basis regulations specified in the Type Certificate Data Sheet A16WE, Revision 44, dated August 8, 2009.
- 6.3 In the event a change is made to any law, governmental regulation or requirement, or in the interpretation of any such law, governmental regulation or requirement that affects the certification basis for the Aircraft as described in Paragraphs 4.1.1 or 6.2, and as a result thereof, a change is made to the configuration and/or the performance of the Aircraft in order to obtain certification, the guarantees set forth in this Attachment shall be appropriately modified to reflect any such change.
- The takeoff and landing guarantees, and the takeoff portion of the mission guarantees are based on hard surface, level and dry runways with no wind or obstacles, no clearway or stopway, 225 mph tires, with Category F brakes and anti-skid operative, and with the Aircraft center of gravity at the most forward limit unless otherwise specified. The takeoff performance is based on no engine bleed for air conditioning or thermal anti-icing and the Auxiliary Power Unit (APU) turned off unless otherwise specified. Unbalanced field length calculations and the improved climb performance procedure will be used for takeoff as required. The landing performance is based on the use of automatic spoilers.
- 6.5 The climb, cruise and descent portions of the mission guarantees include allowances for normal power extraction and engine bleed for normal operation of the air conditioning system. Normal electrical power extraction shall be defined as not less than a 50 kilowatts total electrical load. Normal operation of the air conditioning system shall be defined as pack switches in the "Auto" position, the temperature control switches in the "Auto" position that results in a nominal cabin temperature of 75°F, and all air conditioning systems operating normally. This operation allows a maximum cabin pressure differential of 8.35 pounds per square inch at higher altitudes, with a nominal Aircraft cabin ventilation rate of 2,200 cubic feet per minute including passenger cabin recirculation (nominal recirculation is 33 percent). The APU is turned off unless otherwise specified.
- 6.6 Long Range Cruise (LRC) speed is defined to be the highest speed where cruise fuel mileage is 99 percent of the maximum cruise fuel mileage.

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

- 6.7 The climb, cruise and descent portions of the mission guarantees are based on an Aircraft center of gravity location of 22.2 percent of the mean aerodynamic chord.
- Performance, where applicable, is based on a fuel Lower Heating Value (LHV) of 18,580 BTU per pound and a fuel density of 6.7 pounds per U.S. gallon.
- 6.9 Sound pressure levels are measured in decibels (dB) referred to the standard reference pressure of 20 micro Pascals. Overall sound pressure level (OASPL) is defined to be the sound pressure level for the frequency range from 45 to 11,200 Hz. A-weighted sound level (dBA) is similar to OASPL except that an A-weighting is applied to the sound pressure levels. The weighting is defined in Section 5.4 of IEC 61672-1 for the frequency range from 45 to 11,200 Hz. Speech interference level (SIL) is defined to be the arithmetic average of sound pressure levels in the three octave bands with center frequencies at 1,000, 2,000 and 4,000 Hz.
- The guarantee for interior sound levels in flight pertains to normal operation of the Aircraft in cruise during straight and level flight at an altitude of 35,000 feet and 0.78 Mach number. The Aircraft shall have a complete interior installation including standard thermal/acoustic insulation, all lining and partition panels, a full ship set of fabric upholstered seats and floor covering in the passenger cabin. The cargo compartment shall have a full set of compartment liners and thermal/acoustic insulation. The interior configuration is defined in B376011 (Interior Arrangement). The procedures used for the measurement of sound levels shall be in accordance with ISO 5129-2001(E).
- The guarantees for ramp sound levels pertain to a parked Aircraft during in-service turn around with the APU, environmental control system, environmental control system recirculation fans, electronic equipment cooling fans operating, and with the main propulsion system not operating. The guarantee for ramp sound levels on the 20-meter perimeter pertains to sound levels measured on a rectangular perimeter 20 meters on either side of the Aircraft centerline, 20 meters forward of the nose of the fuselage and 20 meters aft of the tail of the fuselage.

#### 7 GUARANTEE COMPLIANCE

7.1 Compliance with the guarantees of Sections 2, 3 and 4 shall be based on the conditions specified in those sections, the Aircraft configuration of Section 5 and the guarantee conditions of Section 6.

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

- **7.2** Compliance with the takeoff and landing guarantees and the takeoff portion of the mission guarantee shall be based on the FAA approved Airplane Flight Manual for the Model 737-700.
- **7.3** Compliance with the takeoff guarantee and the takeoff portion of the mission guarantee shall be shown using an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord.
- 7.4 Compliance with the climb, cruise and descent portions of the mission guarantees shall be established by calculations based on flight test data obtained from an aircraft in a configuration similar to that defined by the Detail Specification.
- 7.5 The OEW used for compliance with the mission guarantees shall be the actual MEW plus the Standard and Operational Items Allowance in Paragraph 03-60-00 of the Detail Specification.
- 7.6 Compliance with the Manufacturer's Empty Weight guarantee shall be based on information in the "Weight and Balance Control and Loading Manual Aircraft Report."
- 7.7 The data derived from tests shall be adjusted as required by conventional methods of correction, interpolation or extrapolation in accordance with established engineering practices to show compliance with these guarantees.
- 7.8 Compliance with the single event noise exposure level guarantee of Paragraphs 4.2.1 will be by calculations made by Boeing using standard engineering practices with noise information measured by Boeing using the Model 737-700 with CFM56-7B (SAC) engines. The noise calculation process will be as follows:
- 7.8.1 Noise (dBA) time histories for the brake release gross weights (BRGW) guaranteed in Section 4.2 for the model 737-700W with CFM56-7B26E-B2F engines will be calculated by combining full power takeoff and cutback dBA time histories from noise data measured at the noise certification test of May 1997. Adjustments will be appropriate to simulate the takeoff procedure described in Paragraph 4.2.2. These dBA time histories will then be integrated as described in CALIFORNIA NOISE STANDARDS (California Administrative Code, Title 21, Chapter 2.5, Sub-Chapter 6, effective January 1, 1986, Division of Aeronautics (Department of

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364

- Transportation)), except that the threshold noise level will be 65 dBA as used at SNA for noise monitors M1 and M2 to calculate SENEL.
- **7.8.2** The SENEL of Paragraph 7.8.1 will then be compared to the noise limits of 101.8 dB SENEL for microphone location M1 and 101.1 dB SENEL for microphone location M2, respectively, to determine compliance with that guarantee.
- **7.8.3** A document will be prepared by Boeing to show compliance with the single event noise exposure level guarantees of Paragraph 4.2.
- **7.8.5** The following compliance conditions apply to Paragraphs 4.2.1: If the guarantee weight is not met, Boeing and American will work together to improve the weight capability (no financial penalty will be levied).
- **7.9** The data derived from tests shall be adjusted as required by conventional methods of correction, interpolation or extrapolation in accordance with established engineering practices to show compliance with these guarantees.
- Compliance with the guarantee for interior sound levels in flight shall be demonstrated by Boeing Document D047A011. Compliance with the guarantee for ramp sound levels shall be demonstrated by Boeing Document D047A001. This compliance documentation is based on sound level surveys on production 737-700 aircraft acoustically similar to the Buyer's Aircraft. The measured data may be adjusted for sound level increases resulting from Buyer Furnished Equipment, Boeing Purchased Equipment, and from changes to the Detail Specification approved by the Customer and Boeing or otherwise allowed by the Purchase Agreement.
- **7.11** Compliance shall be based on the performance of the airframe and engines in combination, and shall not be contingent on the engine meeting its manufacturer's performance specification.

#### 8 EXCLUSIVE GUARANTEES

The only performance guarantees applicable to the Aircraft are those set forth in this Attachment.

P.A. No. PA-1977 AERO-B-BBA4-M11-0668

SS11-0364



The Boeing Company P.O. Box 3707 Seattle, WA 98124-2207

AAL-PA-1977-LA-1105511

American Airlines, Inc. P.O. Box 619616

Dallas-Fort Worth Airport, Texas 75261-9616

Subject: Aircraft Performance Guarantees – 737-923ER

Reference: Purchase Agreement No. PA-1977 (Purchase Agreement) between The Boeing Company (Boeing) and American Airlines, Inc. (Customer)

relating to Model 737-823 aircraft

This letter agreement (**Letter Agreement**) amends and supplements the Purchase Agreement. All terms used but not defined in this Letter Agreement shall have the same meaning as in the Purchase Agreement.

Boeing agrees to provide Customer with the performance guarantees in the Attachments. Attachment A sets forth performance guarantees for 737-923ER duo class and Attachment B sets forth performance guarantees for the 737-923ER tri class. These guarantees are exclusive and expire upon delivery of the 737-923ER aircraft (Aircraft) to Customer. Notwithstanding the provision of the performance guarantees in the Attachment, Boeing and Customer will work together in good faith to communicate the performance guarantees that will be offered [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2. Confidential Treatment.

The information contained herein represents confidential business information and has value precisely because it is not available generally or to other parties. Customer will limit the disclosure of its contents to employees of Customer with a need to know the contents for purposes of helping Customer perform its obligations under the Purchase Agreement and who understand they are not to disclose its contents to any other person or entity without the prior written consent of Boeing.

AAL-PA-1977-LA-1105511 Performance Guarantees, 737-900ER

Page 1

Very truly yours,

Its

THE BOEING COMPANY

By
Its Attorney-In-Fact

ACCEPTED AND AGREED TO this

Date: , 2011

American Airlines, Inc.

By

AAL-PA-1977-LA-1105511 Performance Guarantees, 737-900ER

**BOEING PROPRIETARY** 

Page 2

# MODEL 737-900ER DUO CLASS WITH WINGLETS PERFORMANCE

## **GUARANTEES**

# FOR AMERICAN AIRLINES, INC.

SECTION	CONTENTS
1	AIRCRAFT MODEL APPLICABILITY
2	FLIGHT PERFORMANCE
3	MANUFACTURER'S EMPTY WEIGHT
4	SOUND LEVELS
5	AIRCRAFT CONFIGURATION
6	GUARANTEE CONDITIONS
7	GUARANTEE COMPLIANCE
8	EXCLUSIVE GUARANTEES

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

## 1 AIRCRAFT MODEL APPLICABILITY

The guarantees contained in this Attachment (the "Performance Guarantees") are applicable to the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2 FLIGHT PERFORMANCE

#### 2.1 Takeoff

2.1.1 The FAA approved takeoff field length at a gross weight at the start of the ground roll of 187,700 pounds, at a temperature of 86°F, at a sea level altitude, with an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord, and using maximum takeoff thrust, shall not be more than the following guarantee value:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2.1.2 The FAA approved takeoff gross weight at the start of ground roll, at a temperature of 95°F, at an altitude of 607 feet and satisfying the conditions defined below, with an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord, and using maximum takeoff thrust, shall not be less than the following guarantee value:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### Conditions:

The takeoff runway available (TORA) is 12,816 feet.

The takeoff distance available (TODA) is 12,816 feet.

The accelerate-stop distance available (ASDA) is 12,816 feet.

The runway slope is 0.03 percent downhill.

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

The following obstacle definition is based on a straight-out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

	Distance	Height
1.	2,650 feet	56 feet
2.	51.154 feet	107 feet

2.1.3 The FAA approved takeoff gross weight at the start of ground roll, at a temperature of 86°F, at an altitude of 56 feet and satisfying the conditions defined below, with an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord, and using maximum takeoff thrust, shall not be less than the following guarantee value:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## Conditions:

The takeoff runway available (TORA) is 5,701 feet.

The takeoff distance available (TODA) is 5,701 feet.

The accelerate-stop distance available (ASDA) is 5,701 feet.

The runway slope is 0.16 percent uphill.

The following obstacle definition is based on a straight-out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

	Distance	Height
1.	1,096 feet	18 feet
2.	2,141 feet	43 feet
3.	2.757 feet	63 feet

## 2.2 Landing

The FAA approved landing field length at a gross weight of 157,300 pounds and at a sea level altitude, shall not be more than the following guarantee value:

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2.3 Mission

## 2.3.1 Mission Payload

The payload for a stage length of 2,825 nautical miles in still air (equivalent to a distance of 2,273 nautical miles with a 90 knot headwind, representative of a Newark to Los Angeles route in Winter) using the conditions and operating rules defined below, shall not be less than the following guarantee value:

NOMINAL: [CONFIDENTIAL PORTION OMITTED AND FILED TOLERANCE: SEPARATELY WITH THE COMMISSION PURSUANT TO A

GUARANTEE: REQUEST FOR CONFIDENTIAL TREATMENT]

Conditions and operating rules:

Stage Length: The stage length is defined as the sum of the distances for the climbout maneuver, climb, cruise, and descent.

Takeoff: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

The airport temperature is 45°F.

The takeoff runway available (TORA) is 10,000 feet. The takeoff distance available (TODA) is 10,000 feet.

The accelerate-stop distance available (ASDA) is 10,000 feet.

The runway slope is 0.02 percent uphill.

P.A. No. PA-1977

AERO-B-BBA4-M11-0670 SS11-0364
BOEING PROPRIETARY

The following obstacle definition is based on a straight out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

	Distance	Height
1.	8 feet	5 feet
2.	5,656 feet	101 feet
3.	16,455 feet	237 feet
4.	80,745 feet	401 feet

Takeoff performance is based on an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord.

Maximum takeoff thrust is used for the takeoff.

The takeoff gross weight shall conform to FAA Regulations.

Climbout Maneuver: Following the takeoff to 35 feet, the Aircraft accelerates to 250 KCAS while climbing to 1,500 feet above the departure airport altitude and retracting flaps and landing gear.

Climb:

The Aircraft climbs from 1,500 feet above the departure airport altitude to 10,000 feet altitude at 250 KCAS.

The Aircraft then accelerates at a rate of climb of 500 feet per minute to the recommended climb speed for minimum block fuel.

The climb continues at the recommended climb speed for minimum block fuel to the initial cruise altitude.

The temperature is standard day during climb.

Maximum climb thrust is used during climb.

Cruise: The Aircraft cruises at the Long Range Cruise (LRC) speed.

The initial cruise altitude is 32,000 feet.

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

A step climb or multiple step climbs of 2,000 feet altitude may be used when beneficial to minimize

fuel burn.

The temperature is standard day during cruise.

The cruise thrust is not to exceed maximum cruise thrust except during a step climb when maximum climb thrust may be

used.

Descent: The Aircraft descends from the final cruise altitude at 250 KCAS to an altitude of 1,500 feet above the destination airport

altitude.

Throughout the descent, the cabin pressure will be controlled to a maximum rate of descent equivalent to 300 feet per minute

at sea level.

The temperature is standard day during descent.

Approach and Landing Maneuver:

The Aircraft decelerates to the final approach speed while extending landing gear and flaps, then descends and lands.

The destination airport altitude is 126 feet.

Fixed Allowances: For the purpose of this guarantee and for the purpose of establishing compliance with this guarantee, the following shall be

used as fixed quantities and allowances:

Taxi-Out:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

Takeoff and Climbout Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

Approach and Landing Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

Taxi-In (shall be consumed from the reserve fuel):

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

AERO-B-BBA4-M11-0670 SS11-0364

Usable reserve fuel remaining upon completion of the approach and landing maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2.3.2 **Mission Payload**

P.A. No. PA-1977

The payload for a stage length of 2,712 nautical miles in still air (equivalent to a distance of 2,313 nautical miles with a 68 knot headwind, representative of a Los Angeles to Honolulu route in Winter) using the conditions and operating rules defined below, shall not be less than the following guarantee value:

> NOMINAL: [CONFIDENTIAL PORTION OMITTED AND FILED TOLERANCE: SEPARATELY WITH THE COMMISSION PURSUANT TO A

**GUARANTEE:** REQUEST FOR CONFIDENTIAL TREATMENT]

Conditions and operating rules:

Stage Length: The stage length is defined as the sum of the distances for the climbout maneuver, climb, cruise, and descent.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A Takeoff:

REQUEST FOR CONFIDENTIAL TREATMENT]

The airport temperature is 63°F.

The takeoff runway available (TORA) is 12,091 feet. The takeoff distance available (TODA) is 12,091 feet.

AERO-B-BBA4-M11-0670

SS11-0364

The accelerate-stop distance available (ASDA) is 12,091 feet.

The runway slope is 0.14 percent uphill.

The following obstacle definition is based on a straight out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

Distance Height
1. 3,907 feet 82 feet

Takeoff performance is based on an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord.

Maximum takeoff thrust is used for the takeoff.

The takeoff gross weight shall conform to FAA Regulations.

Climbout Maneuver: Following the takeoff to 35 feet, the Aircraft accelerates to 250 KCAS while climbing to 1,500 feet above the departure

airport altitude and retracting flaps and landing gear.

Climb: The Aircraft climbs from 1,500 feet above the departure airport altitude to 10,000 feet altitude at 250 KCAS.

The Aircraft then accelerates at a rate of climb of 500 feet per minute to the recommended climb speed for minimum block fuel.

The climb continues at the recommended climb speed for minimum block fuel to the initial cruise altitude.

The temperature is standard day during climb.

Maximum climb thrust is used during climb.

Cruise: The Aircraft cruises at the Long Range Cruise (LRC) speed.

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

The initial cruise altitude is 32,000 feet.

A step climb or multiple step climbs of 2,000 feet altitude may be used when beneficial to minimize fuel burn.

The temperature is standard day during cruise.

The cruise thrust is not to exceed maximum cruise thrust except during a step climb when maximum climb thrust may be

ısed.

Descent: The Aircraft descends from the final cruise altitude at 250 KCAS to an altitude of 1,500 feet above the destination airport

altitude.

Throughout the descent, the cabin pressure will be controlled to a maximum rate of descent equivalent to 300 feet per minute

at sea level.

The temperature is standard day during descent.

Approach and Landing Maneuver:

The Aircraft decelerates to the final approach speed while extending landing gear and flaps, then descends and lands.

The destination airport altitude is 13 feet.

Fixed Allowances: For the purpose of this guarantee and for the purpose of establishing compliance with this guarantee, the following shall be

used as fixed quantities and allowances:

Taxi-Out:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

Takeoff and Climbout Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

Approach and Landing Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Taxi-In (shall be consumed from the reserve fuel):

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Usable reserve fuel remaining upon completion of the approach and landing maneuver

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2.3.3 Mission Payload

The payload for a stage length of 1,439 nautical miles in still air (equivalent to a distance of 1,300 nautical miles with a 45 knot headwind, representative of a New York LaGuardia to Dallas route in Summer) using the conditions and operating rules defined below, shall not be less than the following guarantee value:

NOMINAL: [CONFIDENTIAL PORTION OMITTED AND FILED TOLERANCE: SEPARATELY WITH THE COMMISSION PURSUANT TO A

GUARANTEE: REQUEST FOR CONFIDENTIAL TREATMENT]

Conditions and operating rules:

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

ERO-B-BBA4-M11-0670 SS11-0364

Stage Length:

The stage length is defined as the sum of the distances for the climbout maneuver, climb, cruise, and descent.

Takeoff:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The airport temperature is 82°F.

The takeoff runway available (TORA) is 7,003 feet.

The takeoff distance available (TODA) is 7,003 feet.

The accelerate-stop distance available (ASDA) is 7,003 feet.

The runway slope is 0.04 percent downhill.

The following obstacle definition is based on a straight out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

	Distance	Height
1.	393 feet	12 feet
2.	3,217 feet	90 feet
3.	7,767 feet	208 feet
4.	10,054 feet	264 feet
5.	24,010 feet	283 feet

Takeoff performance is based on an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord.

Maximum takeoff thrust is used for the takeoff.

The takeoff gross weight shall conform to FAA Regulations.

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

Climbout

Following the takeoff to 35 feet, the Aircraft accelerates to 250 KCAS while climbing to 1,500 feet above the departure

Maneuver: airport altitude and retracting flaps and landing gear.

Climb: The Aircraft climbs from 1,500 feet above the departure airport altitude to 10,000 feet altitude at 250 KCAS.

The Aircraft then accelerates at a rate of climb of 500 feet per minute to the recommended climb speed for minimum block

fuel.

The climb continues at the recommended climb speed for minimum block fuel until 0.78 Mach number is reached.

The climb continues at 0.78 Mach number to the initial cruise altitude.

The temperature is standard day during climb.

Maximum climb thrust is used during climb.

Cruise: The Aircraft cruises at the Long Range Cruise (LRC) speed.

The initial cruise altitude is 34,000 feet.

A step climb or multiple step climbs of 2,000 feet altitude may be used when beneficial to minimize fuel burn.

The temperature is standard day during cruise.

The cruise thrust is not to exceed maximum cruise thrust except during a step climb when maximum climb thrust may be

used.

Descent: The Aircraft descends from the final cruise altitude at 250 KCAS to an altitude of 1,500 feet above the destination airport

altitude.

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

D A No. DA 1077

SS11-0364

Throughout the descent, the cabin pressure will be controlled to a maximum rate of descent equivalent to 300 feet per minute

at sea level.

The temperature is standard day during descent.

Approach and Landing Maneuver:

The Aircraft decelerates to the final approach speed while extending landing gear and flaps, then descends and lands.

The destination airport altitude is 607 feet.

Fixed Allowances: For the purpose of this guarantee and for the purpose of establishing compliance with this guarantee, the following shall be

used as fixed quantities and allowances:

Taxi-Out:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

Takeoff and Climbout Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

Approach and Landing Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

Taxi-In (shall be consumed from the reserve fuel):

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

Usable reserve fuel remaining upon completion of the approach and landing maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2.3.4 Mission Block Fuel

The block fuel for a stage length of 1,200 nautical miles in still air with a 42,720 pound payload using the conditions and operating rules defined below, shall not be more than the following guarantee value:

NOMINAL: [CONFIDENTIAL PORTION OMITTED AND FILED TOLERANCE: SEPARATELY WITH THE COMMISSION PURSUANT TO A

GUARANTEE: REQUEST FOR CONFIDENTIAL TREATMENT]

Conditions and operating rules:

Stage Length: The stage length is defined as the sum of the distances for the climbout maneuver, climb, cruise, and descent.

Block Fuel: The block fuel is defined as the sum of the fuel used for taxi-out, takeoff and climbout maneuver, climb, cruise, descent,

approach and landing maneuver, and taxi-in.

Takeoff: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

The takeoff gross weight is not limited by the airport conditions.

P.A. No. PA-1977

AERO-B-BBA4-M11-0670 SS11-0364

Climbout

Following the takeoff to 35 feet, the Aircraft accelerates to 250 KCAS while climbing to 1,500 feet above the departure

Maneuver: airport altitude and retracting flaps and landing gear.

Climb: The Aircraft climbs from 1,500 feet above the departure airport altitude to 10,000 feet altitude at 250 KCAS.

The Aircraft then accelerates at a rate of climb of 500 feet per minute to the recommended climb speed for minimum block

fuel.

The climb continues at the recommended climb speed for minimum block fuel until 0.78 Mach number is reached.

The climb continues at 0.78 Mach number to the initial cruise altitude.

The temperature is standard day during climb.

Maximum climb thrust is used during climb.

Cruise: The Aircraft cruises at the Long Range Cruise (LRC) speed.

The initial cruise altitude is 34,000 feet.

A step climb or multiple step climbs of 2,000 feet altitude may be used when beneficial to minimize fuel burn.

The temperature is standard day during cruise.

The cruise thrust is not to exceed maximum cruise thrust except during a step climb when maximum climb thrust may be

SS11-0364

used.

Descent: The Aircraft descends from the final cruise altitude at 250 KCAS to an altitude of 1,500 feet above the destination airport

altitude.

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

DA N. DA 1077

Throughout the descent, the cabin pressure is controlled to a maximum rate of descent equivalent to 300 feet per minute at

sea level

The temperature is standard day during descent.

Approach and Landing Maneuver:

The Aircraft decelerates to the final approach speed while extending landing gear and flaps, then descends and lands.

The destination airport altitude is sea level.

Fixed Allowances: For the purpose of this guarantee and for the purpose of establishing compliance with this guarantee, the following shall be

used as fixed quantities and allowances:

Operational Empty Weight, OEW (Paragraph 2.3.6): [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY

WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Taxi-Out:

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO$ 

A REQUEST FOR CONFIDENTIAL TREATMENT]

Takeoff and Climbout Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO

A REQUEST FOR CONFIDENTIAL TREATMENT]

Approach and Landing Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO

A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

Taxi-In (shall be consumed from the reserve fuel):

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Usable reserve fuel remaining upon completion of the approach and landing maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2.3.5 Operational Empty Weight Basis

The Operational Empty Weight (OEW) derived in Paragraph 2.3.6 is the basis for the mission guarantees of Paragraphs 2.3.1, 2.3.2, 2.3.3 and 2.3.4.

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

2.3.6 737-900ER Weight Summary - American Airlines

**Standard Model Specification MEW** 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**Pounds** 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] American Airlines

PORTION
OMITTED AND
FILED
SEPARATELY
WITH THE
COMMISSION
PURSUANT TO
A REQUEST
FOR
CONFIDENTIAL
TREATMENT]

[CONFIDENTIAL

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977

AERO-B-BBA4-M11-0670 SS11-0364

## American Airlines Manufacturer's Empty Weight (MEW)

Standard and Operational Items Allowance (Paragraph 2.3.7)

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

American Airlines Operational Empty Weight (OEW)

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL	[CONFIDENTIAL	[CONFIDENTIAL
PORTION	PORTION	PORTION
OMITTED AND	OMITTED AND	OMITTED AND
FILED	FILED	FILED
SEPARATELY	SEPARATELY	SEPARATELY
WITH THE	WITH THE	WITH THE
COMMISSION	COMMISSION	COMMISSION
PURSUANT TO	PURSUANT TO	PURSUANT TO
A REQUEST	A REQUEST	A REQUEST
FOR	FOR	FOR
CONFIDENTIAL	CONFIDENTIAL	CONFIDENTIAL
TREATMENT]	TREATMENT]	TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

## 2.3.7 Standard and Operational Items Allowance

## **Standard Items Allowance**

Unusable Fuel

Oil

Oxygen Equipment

Passenger Portable

Crew Masks

Miscellaneous Equipment

Crash Axe

Megaphones

Flashlights

Smoke Hoods

Galley Structure & Fixed Inserts

## **Operational Items Allowance**

Crew and Crew Baggage

Flight Crew (incl. baggage)

Cabin Crew (incl. baggage)

Catering Allowance & Removable Inserts

First Class

**Tourist Class** 

Passenger Service Equipment

Potable Water - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE

COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Waste Tank Disinfectant

**Emergency Equipment** 

Escape Slides - Forward

Escape Slides - Aft

Life Vests - Crew and Passengers

Life Rafts

Auto Radio Beacon (ELT)

P.A. No. PA-1977

AERO-B-BBA4-M11-0670

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Qty Pounds Pounds Pounds

SS11-0364

**Total Standard and Operational Items Allowance** 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3 MANUFACTURER'S EMPTY WEIGHT

The Manufacturer's Empty Weight (MEW) is guaranteed not to exceed the value in Section 03-60-00 of Detail Specification TBD plus one percent.

- 4 SOUND LEVELS
- 4.1 Community Sound Levels

The Aircraft shall be certified in accordance with the requirements of 14CFR Part 36, Stage 4 and ICAO Annex 16, Volume 1, Chapter 4.

- 4.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Single Event Noise Exposure Levels
- 4.2.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

## **4.2.2** The guarantee of Paragraph 4.2.1 is based on the following conditions, procedures and microphone locations:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED

CONFIDENTIAL TREATMENT]

SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED

AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977

AERO-B-BBA4-M11-0670 SS11-0364

# 4.3 Interior Sound Levels in Flight

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 4.4 Ramp Sound Levels

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 5 AIRCRAFT CONFIGURATION

5.1 The guarantees contained in this Attachment are based on the Aircraft configuration as defined in the original release of Detail

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

Specification TBD (hereinafter referred to as the Detail Specification). Appropriate adjustment shall be made for changes in such Detail Specification approved by the Customer and Boeing or otherwise allowed by the Purchase Agreement which cause changes to the flight performance, sound levels, and/or weight and balance of the Aircraft. Such adjustment shall be accounted for by Boeing in its evidence of compliance with the guarantees.

- The guarantee payloads of Paragraph 2.3.1, 2.3.2, and 2.3.3 will be adjusted by Boeing for the effect of the following on OEW and the Manufacturer's Empty Weight guarantee of Section 3 will be adjusted by Boeing for the following in its evidence of compliance with the guarantees:
  - (1) Changes to the Detail Specification or any other changes mutually agreed upon between the Customer and Boeing or otherwise allowed by the Purchase Agreement.
  - (2) The difference between the component weight allowances given in Appendix IV of the Detail Specification and the actual weights.

### **6 GUARANTEE CONDITIONS**

- **6.1** All guaranteed performance data are based on the International Standard Atmosphere (ISA) and specified variations therefrom; altitudes are pressure altitudes.
- The Federal Aviation Administration (FAA) regulations referred to in this Attachment are, unless otherwise specified, the 737-900ER Certification Basis regulations specified in the Type Certificate Data Sheet A16WE, Revision 40, dated April 27, 2007.
- 6.3 In the event a change is made to any law, governmental regulation or requirement, or in the interpretation of any such law, governmental regulation or requirement that affects the certification basis for the Aircraft as described in Paragraphs 4.1 or 6.2, and as a result thereof, a change is made to the configuration and/or the performance of the Aircraft in order to obtain certification, the guarantees set forth in this Attachment shall be appropriately modified to reflect any such change.
- The takeoff and landing guarantees, and the takeoff portion of the mission guarantees are based on hard surface, level and dry runways with no wind or obstacles, no clearway or stopway, 225

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

mph tires, with Category H brakes and anti-skid operative, and with the Aircraft center of gravity at the most forward limit unless otherwise specified. The takeoff performance is based on no engine bleed for air conditioning or thermal anti-icing and the Auxiliary Power Unit (APU) turned off unless otherwise specified. Unbalanced field length calculations and the improved climb performance procedure will be used for takeoff as required. The landing performance is based on the use of automatic spoilers.

- 6.5 The climb, cruise and descent portions of the mission guarantees include allowances for normal power extraction and engine bleed for normal operation of the air conditioning system. Normal electrical power extraction shall be defined as not less than a 50 kilowatts total electrical load. Normal operation of the air conditioning system shall be defined as pack switches in the "Auto" position, the temperature control switches in the "Auto" position that results in a nominal cabin temperature of 75°F, and all air conditioning systems operating normally. This operation allows a maximum cabin pressure differential of 8.35 pounds per square inch at higher altitudes, with a nominal Aircraft cabin ventilation rate of 3,300 cubic feet per minute including passenger cabin recirculation (nominal recirculation is 47 percent). The APU is turned off unless otherwise specified.
- 6.6 Long Range Cruise (LRC) speed is defined to be the highest speed where cruise fuel mileage is 99 percent of the maximum cruise fuel mileage.
- 6.7 The climb, cruise and descent portions of the mission guarantees are based on an Aircraft center of gravity location of 19.7 percent of the mean aerodynamic chord.
- Performance, where applicable, is based on a fuel Lower Heating Value (LHV) of 18,580 BTU per pound and a fuel density of 6.7 pounds per U.S. gallon.
- 6.9 Sound pressure levels are measured in decibels (dB) referred to the standard reference pressure of 20 micro Pascals. Overall sound pressure level (OASPL) is defined to be the sound pressure level for the frequency range from 45 to 11,200 Hz. A-weighted sound level (dBA) is similar to OASPL except that an A-weighting is applied to the sound pressure levels. The weighting is defined in Section 5.4 of IEC 61672-1 for the frequency range from 45 to 11,200 Hz. Speech interference level (SIL) is defined to be the arithmetic average of sound pressure levels in the three octave bands with center frequencies at 1,000, 2,000 and 4,000 Hz.

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

- The guarantee for interior sound levels in flight pertains to normal operation of the Aircraft in cruise during straight and level flight at an altitude of 35,000 feet and 0.78 Mach number. The Aircraft shall have a complete interior installation including standard thermal/acoustic insulation, all lining and partition panels, a full ship set of fabric upholstered seats and floor covering in the passenger cabin. The cargo compartment shall have a full set of compartment liners and thermal/acoustic insulation. The interior configuration is defined in B377507 (Interior Arrangement). The procedures used for the measurement of sound levels shall be in accordance with ISO 5129-2001(E).
- 6.11 The guarantees for ramp sound levels pertain to a parked Aircraft during in-service turn around with the APU, environmental control system, environmental control system recirculation fans, electronic equipment cooling fans operating, and with the main propulsion system not operating. The guarantee for ramp sound levels on the 20-meter perimeter pertains to sound levels measured on a rectangular perimeter 20 meters on either side of the Aircraft centerline, 20 meters forward of the nose of the fuselage and 20 meters aft of the tail of the fuselage.

### 7 GUARANTEE COMPLIANCE

- **7.1** Compliance with the guarantees of Sections 2, 3 and 4 shall be based on the conditions specified in those sections, the Aircraft configuration of Section 5 and the guarantee conditions of Section 6.
- **7.2** Compliance with the takeoff and landing guarantees and the takeoff portion of the mission guarantee shall be based on the FAA approved Airplane Flight Manual for the Model 737-900ER.
- **7.3** Compliance with the takeoff guarantee and the takeoff portion of the mission guarantee shall be shown using an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord.
- 7.4 Compliance with the climb, cruise and descent portions of the mission guarantees shall be established by calculations based on flight test data obtained from an aircraft in a configuration similar to that defined by the Detail Specification.

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

- 7.5 The OEW used for compliance with the mission guarantees shall be the actual MEW plus the Standard and Operational Items Allowance in Paragraph 03-60-00 of the Detail Specification.
- 7.6 Compliance with the Manufacturer's Empty Weight guarantee shall be based on information in the "Weight and Balance Control and Loading Manual Aircraft Report."
- 7.7 The data derived from tests shall be adjusted as required by conventional methods of correction, interpolation or extrapolation in accordance with established engineering practices to show compliance with these guarantees.
- 7.8 Compliance with the single event noise exposure level guarantee of Paragraphs 4.2.1 will be by calculations made by Boeing using standard engineering practices with noise information measured by Boeing using the Model 737-700 with CFM56-7B (SAC) engines. The noise calculation process will be as follows:
- 7.8.1 Noise (dBA) time histories for the brake release gross weights (BRGW) guaranteed in Section 4.2 for the model 737-900ERW with CFM56-7B27E-B1F engines will be calculated by combining full power takeoff and cutback dBA time histories from noise data measured at the noise certification test of May 1997. Adjustments will be appropriate to simulate the takeoff procedure described in Paragraph 4.2.2. These dBA time histories will then be integrated as described in CALIFORNIA NOISE STANDARDS (California Administrative Code, Title 21, Chapter 2.5, Sub-Chapter 6, effective January 1, 1986, Division of Aeronautics (Department of Transportation)), except that the threshold noise level will be 65 dBA as used at SNA for noise monitors M1 and M2 to calculate SENEL.
- **7.8.2** The SENEL of Paragraph 7.8.1 will then be compared to the noise limits of 101.8 dB SENEL for microphone location M1 and 101.1 dB SENEL for microphone location M2, respectively, to determine compliance with that guarantee.
- **7.8.3** A document will be prepared by Boeing to show compliance with the single event noise exposure level guarantees of Paragraph 4.2.
- **7.8.5** The following compliance conditions apply to Paragraphs 4.2.1:
  - If the guarantee weight is not met, Boeing and American will work together to improve the weight capability (no financial penalty will be levied).

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

- **7.9** The data derived from tests shall be adjusted as required by conventional methods of correction, interpolation or extrapolation in accordance with established engineering practices to show compliance with these guarantees.
- Compliance with the guarantee for interior sound levels in flight shall be demonstrated by Boeing Document D047A014. Compliance with the guarantee for ramp sound levels shall be demonstrated by Boeing Document D537A300. This compliance documentation is based on sound level surveys on production 737-900ER aircraft acoustically similar to the Buyer's Aircraft. The measured data may be adjusted for sound level increases resulting from Buyer Furnished Equipment, Boeing Purchased Equipment, and from changes to the Detail Specification approved by the Customer and Boeing or otherwise allowed by the Purchase Agreement.
- **7.11** Compliance shall be based on the performance of the airframe and engines in combination, and shall not be contingent on the engine meeting its manufacturer's performance specification.

### 8 EXCLUSIVE GUARANTEES

The only performance guarantees applicable to the Aircraft are those set forth in this Attachment.

P.A. No. PA-1977 AERO-B-BBA4-M11-0670

SS11-0364

# MODEL 737-900ER TRI CLASS WITH WINGLETS PERFORMANCE

# **GUARANTEES**

# FOR AMERICAN AIRLINES, INC.

SECTION	CONTENTS
1	AIRCRAFT MODEL APPLICABILITY
2	FLIGHT PERFORMANCE
3	MANUFACTURER'S EMPTY WEIGHT
4	SOUND LEVELS
5	AIRCRAFT CONFIGURATION
6	<b>GUARANTEE CONDITIONS</b>
7	GUARANTEE COMPLIANCE
8	EXCLUSIVE GUARANTEES

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

### 1 AIRCRAFT MODEL APPLICABILITY

The guarantees contained in this Attachment (the "Performance Guarantees") are applicable to the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2 FLIGHT PERFORMANCE

### 2.1 Takeoff

2.1.1 The FAA approved takeoff field length at a gross weight at the start of the ground roll of 187,700 pounds, at a temperature of 86°F, at a sea level altitude, with an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord, and using maximum takeoff thrust, shall not be more than the following guarantee value:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]t

2.1.2 The FAA approved takeoff gross weight at the start of ground roll, at a temperature of 95°F, at an altitude of 607 feet and satisfying the conditions defined below, with an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord, and using maximum takeoff thrust, shall not be less than the following guarantee value:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### Conditions:

The takeoff runway available (TORA) is 12,816 feet.

The takeoff distance available (TODA) is 12,816 feet.

The accelerate-stop distance available (ASDA) is 12,816 feet.

The runway slope is 0.03 percent downhill.

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

The following obstacle definition is based on a straight-out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

	Distance	Height
1.	2,650 feet	56 feet
2.	51,154 feet	107 feet

**2.1.3** The FAA approved takeoff gross weight at the start of ground roll, at a temperature of 86°F, at an altitude of 56 feet and satisfying the conditions defined below, with an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord, and using maximum takeoff thrust, shall not be less than the following guarantee value:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### Conditions:

The takeoff runway available (TORA) is 5,701 feet.

The takeoff distance available (TODA) is 5,701 feet.

The accelerate-stop distance available (ASDA) is 5,701 feet.

The runway slope is 0.16 percent uphill.

The following obstacle definition is based on a straight-out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

	Distance	Height
1.	1,096 feet	18 feet
2.	2,141 feet	43 feet
3.	2.757 feet	63 feet

### 2.2 Landing

The FAA approved landing field length at a gross weight of 157,300 pounds and at a sea level altitude, shall not be more than the following guarantee value:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

## 2.3 Mission

# 2.3.1 Mission Payload

The payload for a stage length of 2,853 nautical miles in still air (equivalent to a distance of 2,292 nautical miles with a 91 knot headwind, representative of a New York to Los Angeles route in Winter) using the conditions and operating rules defined below, shall not be less than the following guarantee value:

NOMINAL: [CONFIDENTIAL PORTION OMITTED AND FILED TOLERANCE: SEPARATELY WITH THE COMMISSION PURSUANT TO A

GUARANTEE: REQUEST FOR CONFIDENTIAL TREATMENT]

Conditions and operating rules:

Stage Length: The stage length is defined as the sum of the distances for the climbout maneuver, climb, cruise, and descent.

Takeoff: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

The airport temperature is 44°F.

The takeoff runway available (TORA) is 13,806 feet.

The takeoff distance available (TODA) is 13,806 feet.

The accelerate-stop distance available (ASDA) is 13,806 feet.

The runway is level.

P.A. No. PA-1977

AERO-B-BBA4-M11-0669 SS11-0364

The following obstacle definition is based on a straight out departure where obstacle height and distance are specified with reference to the liftoff end of the runway:

	Distance	Height
1.	2,804 feet	59 feet
2.	20,571 feet	270 feet

Takeoff performance is based on an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord.

Maximum takeoff thrust is used for the takeoff.

The takeoff gross weight shall conform to FAA Regulations.

Climbout Maneuver:  $Following \ the \ takeoff \ to \ 35 \ feet, \ the \ Aircraft \ accelerates \ to \ 250 \ KCAS \ while \ climbing \ to \ 1,500 \ feet \ above \ the \ departure$ 

airport altitude and retracting flaps and landing gear.

Climb: The Aircraft climbs from 1,500 feet above the departure airport altitude to 10,000 feet altitude at 250 KCAS.

The Aircraft then accelerates at a rate of climb of 500 feet per minute to the recommended climb speed for minimum block

fuel.

The climb continues at the recommended climb speed for minimum block fuel to the initial cruise altitude.

The temperature is standard day during climb.

Maximum climb thrust is used during climb.

Cruise: The Aircraft cruises at the Long Range Cruise (LRC) speed.

The initial cruise altitude is 32,000 feet.

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

BOEING PROPRIETARY

SS11-0364

A step climb or multiple step climbs of 2,000 feet altitude may be used when beneficial to minimize fuel burn.

The temperature is standard day during cruise.

The cruise thrust is not to exceed maximum cruise thrust except during a step climb when maximum climb thrust may be

used.

Descent: The Aircraft descends from the final cruise altitude at 250 KCAS to an altitude of 1,500 feet above the destination airport

iltitude.

Throughout the descent, the cabin pressure will be controlled to a maximum rate of descent equivalent to 300 feet per minute

at sea level.

The temperature is standard day during descent.

Approach and Landing Maneuver:

The Aircraft decelerates to the final approach speed while extending landing gear and flaps, then descends and lands.

The destination airport altitude is 126 feet.

Fixed Allowances: For the purpose of this guarantee and for the purpose of establishing compliance with this guarantee, the following shall be

used as fixed quantities and allowances:

Post-Delivery Auxilliary Tank Installation

(Paragraph 2.3.4):

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO

A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977

AERO-B-BBA4-M11-0669 SS11-0364

### Taxi-Out:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### Takeoff and Climbout Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### Approach and Landing Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Taxi-In (shall be consumed from the reserve fuel):

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Usable reserve fuel remaining upon completion of the approach and landing maneuver [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2.3.2 Mission Block Fuel

The block fuel for a stage length of 1,200 nautical miles in still air with a 29,279 pound payload using the conditions and operating rules defined below, shall not be more than the following guarantee value:

NOMINAL: [CONFIDENTIAL PORTION OMITTED AND FILED TOLERANCE: SEPARATELY WITH THE COMMISSION PURSUANT TO A

GUARANTEE: REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

AERO-B-BBA4-M11-0669 SS11-0364
BOEING PROPRIETARY

Conditions and operating rules:

Stage Length: The stage length is defined as the sum of the distances for the climbout maneuver, climb, cruise, and descent.

Block Fuel: The block fuel is defined as the sum of the fuel used for taxi-out, takeoff and climbout maneuver, climb, cruise, descent,

approach and landing maneuver, and taxi-in.

Takeoff: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

The takeoff gross weight is not limited by the airport conditions.

Climbout Following the takeoff to 35 feet, the Aircraft accelerates to 250 KCAS while climbing to 1,500 feet above the departure

Maneuver: airport altitude and retracting flaps and landing gear.

Climb: The Aircraft climbs from 1,500 feet above the departure airport altitude to 10,000 feet altitude at 250 KCAS.

The Aircraft then accelerates at a rate of climb of 500 feet per minute to the recommended climb speed for minimum block

fuel.

The climb continues at the recommended climb speed for minimum block fuel until 0.78 Mach number is reached.

P.A. No. PA-1977

AERO-B-BBA4-M11-0669 SS11-0364

The climb continues at 0.78 Mach number to the initial cruise altitude.

The temperature is standard day during climb.

Maximum climb thrust is used during climb.

Cruise: The Aircraft cruises at the Long Range Cruise (LRC) speed.

The initial cruise altitude is 36,000 feet.

A step climb or multiple step climbs of 2,000 feet altitude may be used when beneficial to minimize fuel burn.

The temperature is standard day during cruise.

The cruise thrust is not to exceed maximum cruise thrust except during a step climb when maximum climb thrust may be

used.

Descent: The Aircraft descends from the final cruise altitude at 250 KCAS to an altitude of 1,500 feet above the destination airport

altitude

Throughout the descent, the cabin pressure is controlled to a maximum rate of descent equivalent to 300 feet per minute at

sea level.

The temperature is standard day during descent.

Approach and Landing Maneuver:

The Aircraft decelerates to the final approach speed while extending landing gear and flaps, then descends and lands.

The destination airport altitude is sea level.

Fixed Allowances: For the purpose of this guarantee and for the purpose of establishing compliance with this guarantee, the following shall be

used as fixed quantities and allowances:

Operational Empty Weight, OEW (Paragraph 2.3.4): [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY

WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977

AERO-B-BBA4-M11-0669 SS11-0364

### Taxi-Out:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### Takeoff and Climbout Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# Approach and Landing Maneuver:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Taxi-In (shall be consumed from the reserve fuel):

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Usable reserve fuel remaining upon completion of the approach and landing maneuver: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 2.3.3 Operational Empty Weight Basis

The Operational Empty Weight (OEW) derived in Paragraph 2.3.4 is the basis for the mission guarantees of Paragraphs 2.3.1 and 2.3.2.

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

# 2.3.4 737-900ER Weight Summary - American Airlines

# **Standard Model Specification MEW**

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

Pounds

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] American Airlines

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

American Airlines Manufacturer's Empty Weight (MEW)

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Standard and Operational Items Allowance

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

(Paragraph 2.3.5)

American Airlines Operational Empty Weight (OEW)

[CONFIDENTIAL
PORTION OMITTED
AND FILED
SEPARATELY WITH
THE COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

American Airlines Post-Delivery Operational Empty Weight (OEW)

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL	[CONFIDENTIAL	[CONFIDENTIAL
PORTION	PORTION	PORTION
OMITTED AND	OMITTED AND	OMITTED AND
FILED	FILED	FILED
SEPARATELY	SEPARATELY	SEPARATELY
WITH THE	WITH THE	WITH THE
COMMISSION	COMMISSION	COMMISSION
PURSUANT TO	PURSUANT TO	PURSUANT TO
A REQUEST FOR	A REQUEST FOR	A REQUEST FOR
CONFIDENTIAL	CONFIDENTIAL	CONFIDENTIAL
TREATMENT]	TREATMENT]	TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

# 2.3.5 Standard and Operational Items Allowance

Qty Pounds Pounds Pounds

### **Standard Items Allowance**

Unusable Fuel

Oil

Oxygen Equipment

Passenger Portable

Crew Masks

Miscellaneous Equipment

Crash Axe

Megaphones

Flashlights

Smoke Hoods

Galley Structure & Fixed Inserts

# **Operational Items Allowance**

Crew and Crew Baggage

Flight Crew (incl. baggage)

Cabin Crew (incl. baggage)

Catering Allowance & Removable Inserts

First Class

**Business Class** 

**Economy Class** 

Passenger Service Equipment

Potable Water - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE

COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Waste Tank Disinfectant

**Emergency Equipment** 

Escape Slides - Forward

Escape Slides - Aft

Life Vests - Crew and Passengers

Life Rafts

Auto Radio Beacon (ELT)

P.A. No. PA-1977

AERO-B-BBA4-M11-0669

[CONFIDENTIAL PORTION
OMITTED AND FILED
SEPARATELY WITH THE
COMMISSION PURSUANT TO A
REQUEST FOR CONFIDENTIAL
TREATMENT]

SS11-0364

**Total Standard and Operational Items Allowance** 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 3 MANUFACTURER'S EMPTY WEIGHT

The Manufacturer's Empty Weight (MEW) is guaranteed not to exceed the value in Section 03-60-00 of Detail Specification TBD plus one percent.

- 4 SOUND LEVELS
- 4.1 Community Sound Levels

The Aircraft shall be certified in accordance with the requirements of 14CFR Part 36, Stage 4 and ICAO Annex 16, Volume 1, Chapter 4.

- 4.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Single Event Noise Exposure Levels
- **4.2.1** The brake release gross weight for a Single Event Noise Exposure Level (SENEL) at the Class A limits of 101.8 dB SENEL at microphone location M1 and 101.1 dB SENEL at microphone location M2, whichever is limiting for the microphone locations defined below, shall not be less than the performance-limited takeoff weight of Paragraph 2.1.3.

P.A. No. PA-1977 AERO-B-BBA4-M11-0669 SS11-0364

4.2.2 The guarantee of Paragraph 4.2.1 is based on the following conditions, procedures and microphone locations:

> [CONFIDENTIAL PORTION OMITTED AND

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

REQUEST FOR CONFIDENTIAL TREATMENT]

REQUEST FOR CONFIDENTIAL TREATMENT]

**FILED** 

SEPARATELY WITH THE COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL **PORTION** OMITTED AND

**FILED** 

**SEPARATELY** 

WITH THE

COMMISSION PURSUANT TO A

REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL

**PORTION** OMITTED AND

**FILED** 

**SEPARATELY** 

WITH THE COMMISSION

PURSUANT TO A

REQUEST FOR CONFIDENTIAL

TREATMENT]

P.A. No. PA-1977

AERO-B-BBA4-M11-0669 SS11-0364

**BOEING PROPRIETARY** 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A

> [CONFIDENTIAL **PORTION** OMITTED AND

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**FILED SEPARATELY** WITH THE COMMISSION PURSUANT TO A REQUEST FOR

### TREATMENT] 4.3 **Interior Sound Levels in Flight**

CONFIDENTIAL

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT1

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 4.4 Ramp Sound Levels

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. PA-1977

AERO-B-BBA4-M11-0669 SS11-0364

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 5 AIRCRAFT CONFIGURATION

- 5.1 The guarantees contained in this Attachment are based on the Aircraft configuration as defined in the original release of Detail Specification TBD (hereinafter referred to as the Detail Specification). Appropriate adjustment shall be made for changes in such Detail Specification approved by the Customer and Boeing or otherwise allowed by the Purchase Agreement which cause changes to the flight performance, sound levels, and/or weight and balance of the Aircraft. Such adjustment shall be accounted for by Boeing in its evidence of compliance with the guarantees.
- 5.2 The guarantee payload of Paragraph 2.3.1 will be adjusted by Boeing for the effect of the following on OEW and the Manufacturer's Empty Weight guarantee of Section 3 will be adjusted by Boeing for the following in its evidence of compliance with the guarantees:
  - (1) Changes to the Detail Specification or any other changes mutually agreed upon between the Customer and Boeing or otherwise allowed by the Purchase Agreement.
  - (2) The difference between the component weight allowances given in Appendix IV of the Detail Specification and the actual weights.

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

### 5 GUARANTEE CONDITIONS

- **6.1** All guaranteed performance data are based on the International Standard Atmosphere (ISA) and specified variations therefrom; altitudes are pressure altitudes.
- The Federal Aviation Administration (FAA) regulations referred to in this Attachment are, unless otherwise specified, the 737-900ER Certification Basis regulations specified in the Type Certificate Data Sheet A16WE, Revision 40, dated April 27, 2007.
- In the event a change is made to any law, governmental regulation or requirement, or in the interpretation of any such law, governmental regulation or requirement that affects the certification basis for the Aircraft as described in Paragraphs 4.1 or 6.2, and as a result thereof, a change is made to the configuration and/or the performance of the Aircraft in order to obtain certification, the guarantees set forth in this Attachment shall be appropriately modified to reflect any such change.
- 6.4 The takeoff and landing guarantees, and the takeoff portion of the mission guarantee are based on hard surface, level and dry runways with no wind or obstacles, no clearway or stopway, 225 mph tires, with Category H brakes and anti-skid operative, and with the Aircraft center of gravity at the most forward limit unless otherwise specified. The takeoff performance is based on no engine bleed for air conditioning or thermal anti-icing and the Auxiliary Power Unit (APU) turned off unless otherwise specified. Unbalanced field length calculations and the improved climb performance procedure will be used for takeoff as required. The landing performance is based on the use of automatic spoilers.
- 6.5 The climb, cruise and descent portions of the mission guarantees include allowances for normal power extraction and engine bleed for normal operation of the air conditioning system. Normal electrical power extraction shall be defined as not less than a 50 kilowatts total electrical load. Normal operation of the air conditioning system shall be defined as pack switches in the "Auto" position, the temperature control switches in the "Auto" position that results in a nominal cabin temperature of 75°F, and all air conditioning systems operating normally. This operation allows a maximum cabin pressure differential of 8.35 pounds per square inch at higher altitudes, with a nominal Aircraft cabin ventilation rate of 3,300 cubic feet per minute including passenger cabin recirculation (nominal recirculation is 47 percent). The APU is turned off unless otherwise specified.

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

- 6.6 Long Range Cruise (LRC) speed is defined to be the highest speed where cruise fuel mileage is 99 percent of the maximum cruise fuel mileage.
- 6.7 The climb, cruise and descent portions of the mission guarantees are based on an Aircraft center of gravity location of 19.7 percent of the mean aerodynamic chord.
- Performance, where applicable, is based on a fuel Lower Heating Value (LHV) of 18,580 BTU per pound and a fuel density of 6.7 pounds per U.S. gallon.
- 6.9 Sound pressure levels are measured in decibels (dB) referred to the standard reference pressure of 20 micro Pascals. Overall sound pressure level (OASPL) is defined to be the sound pressure level for the frequency range from 45 to 11,200 Hz. A-weighted sound level (dBA) is similar to OASPL except that an A-weighting is applied to the sound pressure levels. The weighting is defined in Section 5.4 of IEC 61672-1 for the frequency range from 45 to 11,200 Hz. Speech interference level (SIL) is defined to be the arithmetic average of sound pressure levels in the three octave bands with center frequencies at 1,000, 2,000 and 4,000 Hz.
- The guarantee for interior sound levels in flight pertains to normal operation of the Aircraft in cruise during straight and level flight at an altitude of 35,000 feet and 0.78 Mach number. The Aircraft shall have a complete interior installation including standard thermal/acoustic insulation, all lining and partition panels, a full ship set of fabric upholstered seats and floor covering in the passenger cabin. The cargo compartment shall have a full set of compartment liners and thermal/acoustic insulation. The interior configuration is defined in B377661 (Interior Arrangement). The procedures used for the measurement of sound levels shall be in accordance with ISO 5129-2001(E).
- 6.11 The guarantees for ramp sound levels pertain to a parked Aircraft during in-service turn around with the APU, environmental control system, environmental control system recirculation fans, electronic equipment cooling fans operating, and with the main propulsion system not operating. The guarantee for ramp sound levels on the 20-meter perimeter pertains to sound levels measured on a rectangular perimeter 20 meters on either side of the Aircraft centerline, 20 meters forward of the nose of the fuselage and 20 meters aft of the tail of the fuselage.

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

### 7 GUARANTEE COMPLIANCE

- **7.1** Compliance with the guarantees of Sections 2, 3 and 4 shall be based on the conditions specified in those sections, the Aircraft configuration of Section 5 and the guarantee conditions of Section 6.
- **7.2** Compliance with the takeoff and landing guarantees and the takeoff portion of the mission guarantees shall be based on the FAA approved Airplane Flight Manual for the Model 737-900ER.
- 7.3 Compliance with the takeoff guarantee and the takeoff portion of the mission guarantee shall be shown using an alternate forward center of gravity limit of 15 percent of the mean aerodynamic chord.
- **7.4** Compliance with the climb, cruise and descent portions of the mission guarantee shall be established by calculations based on flight test data obtained from an aircraft in a configuration similar to that defined by the Detail Specification.
- 7.5 The OEW used for compliance with the mission guarantee shall be the actual MEW plus the Standard and Operational Items Allowance in Paragraph 03-60-00 of the Detail Specification.
- 7.6 Compliance with the Manufacturer's Empty Weight guarantee shall be based on information in the "Weight and Balance Control and Loading Manual Aircraft Report."
- 7.7 The data derived from tests shall be adjusted as required by conventional methods of correction, interpolation or extrapolation in accordance with established engineering practices to show compliance with these guarantees.
- 7.8 Compliance with the single event noise exposure level guarantee of Paragraphs 4.2.1 will be by calculations made by Boeing using standard engineering practices with noise information measured by Boeing using the Model 737-700 with CFM56-7B (SAC) engines. The noise calculation process will be as follows:
- **7.8.1** Noise (dBA) time histories for the brake release gross weights (BRGW) guaranteed in Section 4.2 for the model 737-900ERW with CFM56-7B27E-B1F engines will be calculated by combining full

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

power takeoff and cutback dBA time histories from noise data measured at the noise certification test of May 1997. Adjustments will be appropriate to simulate the takeoff procedure described in Paragraph 4.2.2. These dBA time histories will then be integrated as described in CALIFORNIA NOISE STANDARDS (California Administrative Code, Title 21, Chapter 2.5, Sub-Chapter 6, effective January 1, 1986, Division of Aeronautics (Department of Transportation)), except that the threshold noise level will be 65 dBA as used at SNA for noise monitors M1 and M2 to calculate SENEL.

- **7.8.2** The SENEL of Paragraph 7.8.1 will then be compared to the noise limits of 101.8 dB SENEL for microphone location M1 and 101.1 dB SENEL for microphone location M2, respectively, to determine compliance with that guarantee.
- **7.8.3** A document will be prepared by Boeing to show compliance with the single event noise exposure level guarantees of Paragraph 4.2.
- **7.8.5** The following compliance conditions apply to Paragraphs 4.2.1:
  - If the guarantee weight is not met, Boeing and American will work together to improve the weight capability (no financial penalty will be levied).
- **7.9** The data derived from tests shall be adjusted as required by conventional methods of correction, interpolation or extrapolation in accordance with established engineering practices to show compliance with these guarantees.
- 7.10 Compliance with the guarantee for interior sound levels in flight shall be demonstrated by Boeing Document D047A014. Compliance with the guarantee for ramp sound levels shall be demonstrated by Boeing Document D537A300. This compliance documentation is based on sound level surveys on production 737-900ER aircraft acoustically similar to the Buyer's Aircraft. The measured data may be adjusted for sound level increases resulting from Buyer Furnished Equipment, Boeing Purchased Equipment, and from changes to the Detail Specification approved by the Customer and Boeing or otherwise allowed by the Purchase Agreement.
- **7.11** Compliance shall be based on the performance of the airframe and engines in combination, and shall not be contingent on the engine meeting its manufacturer's performance specification.

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364

# 8 EXCLUSIVE GUARANTEES

The only performance guarantees applicable to the Aircraft are those set forth in this Attachment.

P.A. No. PA-1977 AERO-B-BBA4-M11-0669

SS11-0364





AAL-PA-1977-LA-1105271

American Airlines, Inc.

P.O. 619616

Dallas-Fort worth Airport, Texas 75261-9616

Subject: Advance Payments and Permitted Transactions 4

Reference: (a) Purchase Agreement No. PA-1977 (Purchase Agreement) between The Boeing Company (Boeing) and American Airlines, Inc.

(Customer) relating to Model 737-823 aircraft (Aircraft)

(b) Letter Agreement No. 6-1162-AKP-075R1, as amended, entitled Aircraft Purchase Rights and Substitution Rights

(c) Letter Agreement No. AAL-PA-1977-LA-01073, as amended, entitled Advance Payments and Permitted Transactions 3

This letter agreement (**Letter Agreement**) amends and supplements the Purchase Agreement. All terms used but not defined in this Letter Agreement shall have the same meaning as in the Purchase Agreement.

### 1. Advance Payments for Aircraft.

Notwithstanding Article 4.2, Table 1D as it relates to aircraft with deliveries scheduled in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Letter Agreement No. AAL-PA-1977-LA-01073 as it relates to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT], and Table 1F of the Purchase Agreement, which set forth Boeing's standard [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] advance payment schedule (Standard Advance Payment Schedule), Boeing and Customer agree that the Advance Payment Schedule[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] is hereby amended as follows:

1.1 Customer will make Advance Payments to Boeing in the amount of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

P.A. No. 1977 Advance Payments and Permitted Transactions 4 SA-35 Page 1

Advance Payments and Permitted Transactions 4



- 1.2 Notwithstanding Section 1.3 of Letter Agreement No. 6-1162-AKP-070 entitled Miscellaneous Commitments for Model 737, 757, 767 and 777 Aircraft or any other agreement between the parties, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 1.3 Attachment A hereto sets forth the calculation of the Advance Payments for the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft and the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2. Permitted Transactions.

Notwithstanding Article 9.2 of the AGTA, Boeing agrees that Customer may from time to time [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] or Purchased Aircraft, enter into arrangements and agreements with lenders, lessors or other parties (whose business includes, but may not be limited to, the financing of aircraft) for the purpose of (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (Advance Payment Facility) and/or (b) consummating[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] pursuant to which the right to purchase a Purchased Aircraft, and the corresponding obligation to make Advance Payments with respect thereto, are conveyed to such a party [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT) and, together with an Advance Payment Facility, Permitted Transactions).

For the purpose of securing an Advance Payment Facility and notwithstanding the provisions of the Security Agreement between Boeing and Customer dated as of October 16, 2002, as may be subsequently amended or revised, Customer may [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] including the right to purchase a Purchased Aircraft provided that Customer and Customer's financiers accept Boeing's customary terms and conditions for consenting[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] including, but not limited to, the following:

(i) if Customer's financier gives notice that it intends to exercise its [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY

P.A. No. 1977 Advance Payments and Permitted Transactions 4 SA-35 Page 2



WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Boeing shall have the right to assume those rights with respect to such[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and received and retained by Boeing with respect to CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] **Manufacturer's Option**);

- (ii) Customer shall continue at all times to remain liable to Boeing under the Purchase Agreement to perform all duties and obligations of Customer;
- (iii) Boeing shall not be subject to any additional liability as a result of the pledge of security which Boeing would not otherwise be subject to under the Purchase Agreement;
- (iv) the pledge of security shall not modify in any respect the continued rights of Boeing under the Purchase Agreement, or require Boeing to divest itself of title to or possession of the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and
- (v) In lieu of the concessions granted to Customer under the Purchase Agreement, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] In calculating the amount payable by the financier for a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and this irrespective of whether the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

To assist Customer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Customer may execute a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Any such [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be subject to the conditions that (i) it shall not subject Boeing to any liability as a

P.A. No. 1977 Advance Payments and Permitted Transactions 4 SA-35 Page 3



Very truly yours,

result of the assignment which Boeing would not otherwise be subject to under the Purchase Agreement and (ii) no such assignment will require Boeing to divest itself of title to or possession of the aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] the Purchase Agreement in connection with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] is contingent upon Boeing's consent, which shall not be unreasonably withheld or delayed. Boeing agrees to cooperate in good faith with Customer and to take such actions as may be reasonably requested by Customer to facilitate Permitted Transactions. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

For avoidance of doubt, nothing within this Letter Agreement is intended to, nor shall it, derogate the rights and obligations of Boeing and Customer with regard to the financing of aircraft at or following delivery in accordance with Article 9.2 of the AGTA.

THE BOEING COMPANY		
Ву		
Its Attorney-In-Fact		
ACCEPTED AND AGREED TO this		
Date		
American Airlines, Inc.		
Ву		
Its		
P.A. No. 1977		

Advance Payments and Permitted Transactions 4

BOEING PROPRIETARY

SA-35 Page 4

# Attachment A <u>To</u> AAL-PA-1980-LA-1105629 entitled

# **Advance Payments and Permitted Transactions**

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

P.A. No. 1980 Advance Payments and Permitted Transactions

SA-25

**Boeing Proprietary** 





AAL-PA-1977-LA-1105272

American Airlines, Inc.

P.O. 619616

Dallas-Fort worth Airport, Texas 75261-9616

Subject:

**Business Considerations 2** 

Reference:

- a) Purchase Agreement No. PA-1977 (**Purchase Agreement**) between The Boeing Company (**Boeing**) and American Airlines, Inc. (**Customer**) relating to Model 737-823 aircraft (**Aircraft**)
- b) Letter Agreement 6-1162-AKP-074R2 or as may be subsequently amended, entitled Business Considerations.
- c) Letter Agreement 6-1162-SSM-1405R1 or as may be subsequently amended, entitled Multiple Operating Weight Program Model, 737-723/-823/-923ER Aircraft

This letter agreement (**Letter Agreement**) amends and supplements the Purchase Agreement. All terms used but not defined in this Letter Agreement shall have the same meaning as in the Purchase Agreement.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 1. Model 737-923ER Basic Credit Memoranda.
- a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

AAL-PA-1977-LA-1105272 Business Considerations 2

Page 1



- c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2. Model 737-823 Basic Credit Memoranda.
- a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. Model 737-723 Basic Credit Memoranda.
- a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

AAL-PA-1977-LA-1105272 Business Considerations 2

Page 2

**BOEING PROPRIETARY** 



## 4. Assignment.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 5. Application of Credit Memoranda.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 6. Confidential Treatment.

**Business Considerations 2** 

Very truly yours,

The information contained herein represents confidential business information and has value precisely because it is not available generally or to other parties. Customer will limit the disclosure of its contents to employees of Customer with a need to know the contents for purposes of helping Customer perform its obligations under the Purchase Agreement and who understand they are not to disclose its contents to any other person or entity without the prior written consent of Boeing.

THE BOEING COMPANY				
Ву				
Its	Attorney-In-Fact			
ACCEPTED AND AGREED TO this				
Date:	, 2011			
American Airlines, Inc.				
Ву				
Its				
AAL-PA-1977-LA-1105272				

**BOEING PROPRIETARY** 

Page 3





AAL-PA-1977-LA-1105616

American Airlines, Inc. P.O. Box 619616

Dallas-Fort Worth Airport, Texas 75261-9616

Subject: Open Configuration Matters

References: (a) Purchase Agreement No. PA-1977 (**Purchase Agreement**) between The Boeing Company (**Boeing**) and American Airlines, Inc.

(Customer) relating to Model 737-823 aircraft (Aircraft)

This letter agreement (**Letter Agreement**) amends and supplements the Purchase Agreement. All terms used but not defined in this Letter Agreement shall have the same meaning as in the Purchase Agreement.

### 1. Aircraft Configuration.

- 1.1 <u>Initial Configuration</u>. The initial configuration of Customer's Model 737-700 and 737-900ER Aircraft has been defined by Configuration Specification D019A001, Revision 0, March 31, 2011 as described in Article 1 and Exhibit A of the Purchase Agreement. Final configuration of the Aircraft will be completed as described in this Letter Agreement
- 1.2 <u>Final Configuration Schedule</u>. Customer and Boeing hereby agree to complete the configuration of the Aircraft using the then-current Boeing configuration documentation (**Final Configuration**) in accordance with the following schedule:
- 1.2.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.2.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.2.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

AAL-PA-1977-LA-1105616 Open Configuration Matters

Page 1

**BOEING PROPRIETARY** 



- 2. <u>Amendment of the Purchase Agreement</u>. Within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] days following Final Configuration, Boeing and Customer will execute a written amendment to the Purchase Agreement which will reflect the following:
- 2.1 Changes applicable to the basic Model + aircraft which are developed by Boeing between the date of signing of the Purchase Agreement and date of Final Configuration;
- 2.2 Incorporation into Exhibit A of the Purchase Agreement, by written amendment, those optional features which have been agreed to by Customer and Boeing pursuant to Article 1.2 above (**Customer Configuration Changes**);
- 2.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.5 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. Other Letter Agreements.

Boeing and Customer acknowledge that as the definition of the Aircraft progresses, there may be a need to execute letter agreements addressing one or more of the following subjects:

- 3.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

AAL-PA-1977-LA-1105616 Open Configuration Matters

**BOEING PROPRIETARY** 

Page 2

( BOEING
----------

Very truly yours,

THE BOEING COMPANY

By

Its Attorney-In-Fact

ACCEPTED AND AGREED TO this

Date: , 2011

American Airlines, Inc.

AAL-PA-1977-LA-1105616 Open Configuration Matters

 ${\bf B}{\bf y}$ 

Its

**BOEING PROPRIETARY** 

Page 3



The Boeing Company P.O. Box 3707 Seattle, WA 98124-2207

AAL-PA-1977-LA-1105863

American Airlines, Inc. P.O. Box 619616

Dallas-Fort Worth Airport, Texas 75261-9616

Subject:

Performance Guarantees for Rights Aircraft

Reference:

a.) Purchase Agreement No. PA-1977 (**Purchase Agreement**) between The Boeing Company (**Boeing**) and American Airlines, Inc.

(Customer) relating to Model 737-823 aircraft (Aircraft)

b) Letter Agreement No. 6-1162-CLO-1035 entitled Aircraft Performance Guarantees, as may be subsequently amended

This letter agreement (**Letter Agreement**) amends and supplements the Purchase Agreement. All terms used but not defined in this Letter Agreement shall have the same meaning as in the Purchase Agreement.

Boeing and Customer will work together in good faith to communicate the performance guarantees that will be offered prior to Customer exercising a Rights Aircraft as a 737-823 minor model. The intent of this communication is to avoid situations in which the offered performance guarantees for an exercised Rights Aircraft differ from the performance guarantees as provided for in the reference b) letter agreement.

Very truly yours,

THE BOEING COMPANY

Ву					
Its	Attorney-In-Fact				
ACCEPTED AND AGREED TO this					
Date:	, 2011				
American Airlines, Inc.					
Ву					
Its					

AAL-PA-1977-LA-1105863

Performance Guarantees for Rights Aircraft

SA-35 Page 1

**BOEING PROPRIETARY** 

A320 FAMILY AIRCRAFT

PURCHASE AGREEMENT

**BETWEEN** 

AIRBUS S.A.S.

as Seller

<u>AND</u>

AMERICAN AIRLINES, INC.

as Buyer

# $\underline{\textbf{CONTENTS}}$

0	DEFINITI	ONS	Ţ		
1	SALE AND PURCHASE; LEASE				
2	SPECIFIC	CATION	16		
3	PRICE		20		
4	PRICE RE	EVISION	29		
5	PAYMENT	T TERMS	30		
6	MANUFACTURE PROCEDURE - INSPECTION				
7	CERTIFICATION				
8	TECHNIC	CAL ACCEPTANCE	43		
9	DELIVER	Y	43		
10	EXCUS	SABLE DELAY AND TOTAL LOSS	45		
11	INEXC	USABLE DELAY	48		
12		ANTIES AND SERVICE LIFE POLICY	50		
13		E SECRET, PATENT AND COPYRIGHT INDEMNITY	66		
14	TECHN	NICAL DATA AND SOFTWARE SERVICES	69		
15	SELLE	R REPRESENTATIVE SERVICES	76		
16		IING SUPPORT AND SERVICES	79		
17	7 EQUIPMENT SUPPLIER PRODUCT SUPPORT 9				
18		R FURNISHED EQUIPMENT	92		
19		INITIES AND INSURANCE	99		
20		INATION	102		
21		NMENTS AND TRANSFERS	107		
22	MISCE	LLANEOUS PROVISIONS	109		
EXH	<u>IIBITS</u>	TITLES_			
Exh	nibit A-1	A319 AIRCRAFT STANDARD SPECIFICATION			
		A320 AIRCRAFT STANDARD SPECIFICATION			
		A321 AIRCRAFT STANDARD SPECIFICATION			
Exh	nibit A-2	A319 AIRCRAFT SCN LISTING			
Exhibit A-3 A320 AIRCRAFT SCN LISTING		A320 AIRCRAFT SCN LISTING			

Exhibit A-4

Exhibit A-5

Exhibit A-6
Exhibit A-7

Exhibit B-1

Exhibit B-2

Exhibit C-1
Exhibit C-2

A321 AIRCRAFT SCN LISTING

A319 NEO AIRCRAFT SCN LISTING A320 NEO AIRCRAFT SCN LISTING

A321 NEO AIRCRAFT SCN LISTING

FORM OF SPECIFICATION CHANGE NOTICE

SELLER PRICE REVISION FORMULA 2011

SELLER PRICE REVISION FORMULA 2010

FORM OF MANUFACTURER'S SPECIFICATION CHANGE NOTICE

,	
Exhibit C-3	CFM INTERNATIONAL PROPULSION SYSTEM PRICE REVISION FORMULA
Exhibit C-4	INTERNATIONAL AERO ENGINES PROPULSION SYSTEM PRICE REVISION FORMULA
Exhibit C-5	PRATT & WHITNEY PROPULSION SYSTEM PRICE REVISION FORMULA
Exhibit D	FORM OF CERTIFICATE OF ACCEPTANCE
Exhibit E	FORM OF BILL OF SALE
Exhibit F	SERVICE LIFE POLICY – LIST OF ITEMS

Exhibit G TECHNICAL DATA INDEX
Exhibit H MATERIAL SUPPLY AND SERVICES

Exhibit I INDEX OF LETTER AGREEMENTS

**SCHEDULES** 

Schedule I DELIVERY SCHEDULE

## A320 FAMILY AIRCRAFT PURCHASE AGREEMENT

This A320 Family Aircraft Purchase Agreement is made July 20, 2011

between

**AIRBUS S.A.S.**, a *société par actions simplifiée*, created and existing under French law having its registered office at 1 Rond-Point Maurice Bellonte, 31707 Blagnac-Cedex, France and registered with the Toulouse *Registre du Commerce* under number RCS Toulouse 383 474 814 (the "**Seller**"),

\_... \_

**AMERICAN AIRLINES, INC.**, a Delaware corporation having its principal office at 4333 Amon Carter Boulevard, Fort Worth, Texas 76155, United States of America (the "**Buyer**").

**WHEREAS,** subject to the terms and conditions of this Agreement, the Seller desires to sell the Aircraft to the Buyer or the Lessor, and the Buyer desires to purchase or lease the Aircraft from the Seller or the Lessor, respectively.

NOW THEREFORE IT IS AGREED AS FOLLOWS:

#### 0 DEFINITIONS

For all purposes of this Agreement (defined below), except as otherwise expressly provided, the following terms will have the following meanings:

<u>A319 Aircraft</u> – any or all of the A319-100 model aircraft sold by the Seller and purchased by the Buyer or sold by the Seller to the Lessor and leased to the Buyer pursuant to this Agreement, including the A319 Airframe and all components, equipment, parts and accessories installed in or on such aircraft and the applicable A319 Propulsion System installed thereon upon Delivery.

A319 Airframe – any A319 Aircraft, excluding the A319 Propulsion System therefor.

A319 NEO Aircraft – any or all of the A319-100 model aircraft incorporating the New Engine Option sold by the Seller and purchased by the Buyer pursuant to this Agreement, including the A319 NEO Airframe and all components, equipment, parts and accessories installed in or on such aircraft and the applicable A319 NEO Propulsion System installed thereon upon Delivery.

A319 NEO Airframe – any A319 NEO Aircraft, excluding the A319 NEO Propulsion System therefor.

A319 NEO Group 1 Aircraft – any or all of the A319 NEO Aircraft for which the Buyer has selected a LEAP-X Propulsion System pursuant to Clause 2.3.

A319 NEO Group 1 Airframe – any A319 NEO Group 1 Aircraft, excluding the A319 NEO Propulsion System therefor.

A319 NEO Group 2 Aircraft - any or all of the A319 NEO Aircraft other than the A319 NEO Group 1 Aircraft.

A319 NEO Group 2 Airframe – any A319 NEO Group 2 Aircraft, excluding the A319 NEO Propulsion System therefor.

A319 NEO Propulsion System - as defined in Clause 2.4.4.

A319 Propulsion System – as defined in Clause 2.4.1.

<u>A319 Specification</u> – either (a) the A319 Standard Specification if no SCNs or MSCNs are applicable or (b) if SCNs or MSCNs are issued or deemed issued, the A319 Standard Specification as amended by all applicable SCNs and MSCNs.

<u>A319 Standard Specification</u> – the A319 standard specification document number J.000.01000, Issue 7, dated June 20, 2011, which includes a maximum take-off weight (MTOW) of 166,449 pounds, a maximum landing weight (MLW) of 137,789 pounds and a maximum zero fuel weight (MZFW) of 128,970 pounds, a copy of which is annexed hereto as Exhibit A-1.

<u>A320 Aircraft</u> – any or all of the A320-200 model aircraft sold by the Seller and purchased by the Buyer or sold by the Seller to the Lessor and leased to the Buyer pursuant to this Agreement, including the A320 Airframe and all components, equipment, parts and accessories installed in or on such aircraft and the applicable A320 Propulsion System installed thereon upon Delivery.

A320 Airframe – any A320 Aircraft, excluding the A320 Propulsion System therefor.

A320 Family Aircraft – any or all of the A319 Aircraft, the A320 Aircraft and the A321 Aircraft.

A320 Family Base Period – as defined in Clause 3.1.1.1.

A320 Family NEO Aircraft – any or all of the A319 NEO Aircraft, the A320 NEO Aircraft and the A321 NEO Aircraft.

A320 Family NEO Base Period – as defined in Clause 3.1.1.4.

<u>A320 NEO Aircraft</u> – any or all of the A320-200 model aircraft incorporating the New Engine Option sold by the Seller and purchased by the Buyer pursuant to this Agreement, including the A320 NEO Airframe and all components, equipment, parts and accessories installed in or on such aircraft and the applicable A320 NEO Propulsion System installed thereon upon Delivery.

A320 NEO Airframe – any A320 NEO Aircraft, excluding the A320 NEO Propulsion System therefor.

A320 NEO Group 1 Aircraft – any or all of the A320 NEO Aircraft for which the Buyer has selected a LEAP-X Propulsion System pursuant to Clause 2.3.

A320 NEO Group 1 Airframe – any A320 NEO Group 1 Aircraft, excluding the A320 NEO Propulsion System therefor.

A320 NEO Group 2 Aircraft - any or all of the A320 NEO Aircraft other than the A320 NEO Group 1 Aircraft.

A320 NEO Group 2 Airframe - any A320 NEO Group 2 Aircraft, excluding the A320 NEO Propulsion System therefor.

A320 NEO Propulsion System – as defined in Clause 2.4.5.

A320 Propulsion System – as defined in Clause 2.4.2.

<u>A320 Specification</u> – either (a) the A320 Standard Specification if no SCNs or MSCNs are applicable or (b) if SCNs or MSCNs are issued or deemed issued, the A320 Standard Specification as amended by all applicable SCNs and MSCNs.

<u>A320 Standard Specification</u> – the A320 standard specification document number D.000.02000, Issue 8, dated June 20, 2011, which includes a maximum take-off weight (MTOW) of 171,960 pounds, a maximum landing weight (MLW) of 145,505 pounds and a maximum zero fuel weight (MZFW) of 137,789 pounds, a copy of which is annexed hereto as Exhibit A-2.

<u>A321 Aircraft</u> – any or all of the A321-200 model aircraft sold by the Seller and purchased by the Buyer or sold by the Seller to the Lessor and leased to the Buyer pursuant to this Agreement, including the A321 Airframe and all components, equipment, parts and accessories installed in or on such aircraft and the applicable A321 Propulsion System installed thereon upon Delivery.

A321 Airframe – any A321 Aircraft, excluding the A321 Propulsion System therefor.

<u>A321 NEO Aircraft</u> – any or all of the A321-200 model aircraft incorporating the New Engine Option sold by the Seller and purchased by the Buyer pursuant to this Agreement, including the A321 NEO Airframe and all components, equipment, parts and accessories installed in or on such aircraft and the applicable A321 NEO Propulsion System installed thereon upon Delivery.

A321 NEO Airframe – any A321 NEO Aircraft, excluding the A321 NEO Propulsion System therefor.

A321 NEO Group 1 Aircraft – any or all of the A321 NEO Aircraft for which the Buyer has selected a LEAP-X Propulsion System pursuant to Clause 2.3.

A321 NEO Group 1 Aircraft, excluding the A321 NEO Propulsion System therefor.

A321 NEO Group 2 Aircraft – any or all of the A321 NEO Aircraft other than the A321 NEO Group 1 Aircraft.

A321 NEO Group 2 Airframe – any A321 NEO Group 2 Aircraft, excluding the A321 NEO Propulsion System therefor.

<u>A321 NEO Propulsion System</u> – as defined in Clause 2.4.6.

A321 Propulsion System - as defined in Clause 2.4.3.

<u>A321 Specification</u> – either (a) the A321 Standard Specification if no SCNs or MSCNs are applicable or (b) if SCNs or MSCNs are issued or deemed issued, the A321 Standard Specification as amended by all applicable SCNs and MSCNs.

<u>A321 Standard Specification</u> – the A321 standard specification document number E.000.02000, Issue 5, dated June 20, 2011, which includes a maximum take-off weight (MTOW) of 206,132 pounds, a maximum landing weight (MLW) of 171,519 pounds and a maximum zero fuel weight (MZFW) of 162,701 pounds and the installation of two (2) auxiliary center tanks, a copy of which is annexed hereto as Exhibit A-3.

<u>AACS</u> – Airbus Americas Customer Services, Inc., a corporation organized and existing under the laws of Delaware, having its registered office located at 198 Van Buren Street, Suite 300, Herndon, VA 20170, or any successor thereto.

Affiliate — with respect to any person or entity, any other person or entity directly or indirectly controlling, controlled by or under common control with such person or entity. For purposes of this definition, "control" when used with respect to any person or entity, means the power to direct the management and policies of such person or entity, directly or indirectly, whether through the ownership of voting securities, by contract or otherwise, and the terms "controlling" and "controlled" have meanings correlative to the foregoing.

<u>Agreement</u> – this A320 family aircraft purchase agreement, including all letter agreements listed on Exhibit I and all exhibits, appendices and schedules attached hereto, as the same may be amended or modified and in effect from time to time.

AirbusWorld - as defined in Clause 14.5.4.

<u>Aircraft</u> – any or all of the one hundred thirty (130) firm A320 Family Aircraft and one hundred thirty (130) firm A320 Family NEO Aircraft for which the delivery schedule is set forth in Schedule I as of the date hereof and any other A320 Family Aircraft and A320 Family NEO Aircraft to be sold by the Seller

and purchased by the Buyer (or, in the case of the A320 Family Aircraft, leased by the Buyer from the Lessor) pursuant to this Agreement, including the Airframe and all components, equipment, parts and accessories installed in or on such aircraft and the Propulsion System installed thereon upon Delivery.

<u>Aircraft Training Services</u> – all flight support services including but not limited to any and all training courses, flight training, flight assistance, line training, line assistance and more generally all flights of any kind performed by the Seller, its agents, employees or subcontractors, and maintenance support, maintenance training (including Practical Training), training support of any kind performed on aircraft and provided to the Buyer pursuant to this Agreement.

<u>Airframe</u> – any Aircraft, excluding the Propulsion System therefor.

<u>ATA Specification</u> – recommended specifications developed by the Air Transport Association of America reflecting consensus in the commercial Aviation industry on accepted means of communicating information, conducting business, performing operations and adhering to accepted practices.

<u>Aviation Authority</u> – when used with respect to any jurisdiction, the government entity that, under the laws of such jurisdiction, has control over civil aviation or the registration, airworthiness or operation of civil aircraft in such jurisdiction.

Balance of the Final Price - as defined in Clause 5.4.

Base Price – for any Aircraft, as defined in Clause 3.1.

Base Price of the A319 Airframe – as defined in Clause 3.1.1.1.

Base Price of the A320 Airframe – as defined in Clause 3.1.1.2.

Base Price of the A321 Airframe – as defined in Clause 3.1.1.3.

Base Price of the A319 NEO Group 1 Airframe – as defined in Clause 3.1.1.4.

Base Price of the A319 NEO Group 2 Airframe – as defined in Clause 3.1.1.5.

Base Price of the A320 NEO Group 1 Airframe – as defined in Clause 3.1.1.6.

Base Price of the A320 NEO Group 2 Airframe – as defined in Clause 3.1.1.7.

Base Price of the A321 NEO Group 1 Airframe – as defined in Clause 3.1.1.8.

Base Price of the A321 NEO Group 2 Airframe - as defined in Clause 3.1.1.9.

<u>Base Price of the Airframe</u> – the Base Price of the A319 Airframe, the Base Price of the A320 Airframe, the Base Price of the A321 Airframe, the Base Price of the A319 NEO Group 1 Airframe, the Base Price of the A320 NEO Group 1 Airframe, the Base Price of the A320 NEO Group 2 Airframe, the Base Price of the A321 NEO Group 2 Airframe, as applicable.

<u>Base Price</u> of the <u>Propulsion System</u> – the Engine A Base Price, the Engine B Base Price, the Engine C Base Price, the Engine D Base Price, the Engine E Base Price, the Engine F Base Price, the Engine G Base Price, the Engine H Base Price, the Engine I Base Price, the Engine J Base Price, the Engine K Base Price, or the Engine L Base Price, as applicable.

<u>Business Day</u> – with respect to any action to be taken hereunder, a day other than a Saturday, Sunday or other day on which commercial banks in the jurisdiction where such action is required to be taken are authorized or required by law to close.

Buyer Furnished Equipment or BFE – as defined in Clause 18.1.1.

<u>Certificate of Acceptance</u> – as defined in Clause 8.3.

CFM – CFM International.

Corrected Part – as defined in Clause 12.1.9.

Correction – as defined in Clause 12.1.4.1.

Custom Duty – as defined in Clause 18.1.6.

Customization Milestone Chart – as defined in Clause 2.4.1.

DAP – as defined in Clause 14.5.2.

<u>Declaration of Design and Performance</u> or <u>DDP</u> – the documentation provided by an equipment manufacturer guaranteeing that the corresponding equipment meets the requirements of the Specification, the interface documentation and all relevant certification requirements.

<u>Delivery</u> – with respect to any Aircraft, the transfer of title to such Aircraft from the Seller to the Buyer (or, with respect to any Leased Aircraft, the Lessor) in accordance with Clause 9.

<u>Delivery Date</u> – the date the Delivery occurs.

<u>Delivery Location</u> – the facilities of the Seller at the location of final assembly of the Aircraft, which will be located in Hamburg, Germany, Toulouse, France or another location mutually agreed by the Buyer and the Seller in writing.

<u>Development Changes</u> – as defined in Clause 2.2.2.

<u>Dispute Resolution Process</u> – as defined in Clause 12.1.6.1.

EASA - the European Aviation Safety Agency or any successor thereto.

End-User License Agreement for Airbus Software – as defined in Clause 14.10.4.

Engine A Base Price – as defined in Clause 3.1.2.1.

Engine A Reference Price – as defined in Clause 3.1.2.1.

Engine B Base Price – as defined in Clause 3.1.2.2.

Engine B Reference Price – as defined in Clause 3.1.2.2.

Engine C Base Price – as defined in Clause 3.1.2.3.

Engine C Reference Price – as defined in Clause 3.1.2.3.

Engine D Base Price – as defined in Clause 3.1.2.4.

Engine D Reference Price – as defined in Clause 3.1.2.4.

Engine E Base Price – as defined in Clause 3.1.2.5

<u>Engine E Reference Price</u> – as defined in Clause 3.1.2.5.

Engine F Base Price – as defined in Clause 3.1.2.6.

Engine F Reference Price – as defined in Clause 3.1.2.6.

Engine G Base Price – as defined in Clause 3.1.2.7.

Engine G Reference Price – as defined in Clause 3.1.2.7.

Engine H Base Price – as defined in Clause 3.1.2.8.

Engine H Reference Price – as defined in Clause 3.1.2.8.

Engine I Base Price – as defined in Clause 3.1.2.9.

<u>Engine I Reference Price</u> – as defined in Clause 3.1.2.9.

Engine J Base Price – as defined in Clause 3.1.2.10.

Engine J Reference Price – as defined in Clause 3.1.2.10.

Engine K Base Price – as defined in Clause 3.1.2.11.

<u>Engine K Reference Price</u> – as defined in Clause 3.1.2.11.

Engine L Base Price – as defined in Clause 3.1.2.12.

Engine L Reference Price – as defined in Clause 3.1.2.12.

Excusable Delay – as defined in Clause 10.1.

Export Certificate of Airworthiness – an export certificate of airworthiness issued by the Aviation Authority of the Delivery Location.

 $\underline{FAA}$  – the U.S. Federal Aviation Administration, or any successor thereto.

Final Price – as defined in Clause 3.2.

First Quarter or 1st Quarter – January, February and March of any given calendar year.

Fourth Quarter or 4th Quarter – October, November and December of any given calendar year.

General Terms and Conditions of Access to and Use of AirbusWorld or GTC - as defined in Clause 14.11.3.

Goods and Services – any goods, excluding Aircraft, and services that may be purchased by the Buyer from the Seller or Affiliates of the Seller controlled by the Seller

Indemnified Airframe – as defined in Clause 13.1.1.

Inexcusable Delay – as defined in Clause 11.1.

IAE - International Aero Engines.

<u>Independent Party</u> – as defined in Clause 12.1.6.1.

<u>Irrevocable SCNs</u> – the list of SCNs set forth in Exhibits A-5, A-6 and A-7 that are irrevocably part of the A320 Family NEO Aircraft, as applicable.

<u>Leased Aircraft</u> – any or all of the one hundred thirty (130) firm A320 Family Aircraft for which the delivery schedule is set forth in Schedule I as of the date hereof that are to be leased by the Buyer from the Lessor pursuant to this Agreement.

<u>Lessor</u> – an owner trust, the owner participant of which is (a) Seller (or any Affiliate of Seller); (b) a third party lessor designated by Seller; (c) a special purpose company established by either of the parties referred to in (a) or (b) for the purposes of leasing one or more Leased Aircraft to Buyer; or (d) such other entity as may be agreed by Seller and Buyer.

<u>Liable Party</u> – as defined in Clause 5.5.4.

LIBOR – the rate for deposits in US dollars for a period of one month which appears on the Reuters Screen LIBOR01 Page as of 11:00 A.M., London time, as of the date such amounts came due. If such rate does not appear on the Reuters Screen LIBOR01 Page, the rate will be determined on the basis of the rates at which deposits in US dollars are offered by the principal London offices of the Reference Banks at approximately 11:00 A.M. time, London to prime banks in the London interbank market for a period of one month commencing on the date such amounts came due and in an amount of US\$ 10,000,000 (US dollars – ten million). Seller will request the principal London office of each of the Reference Banks to provide a quotation of its rate. If at least two such quotations are provided, the rate for such date of determination will be the arithmetic mean of the quotations. If fewer than two quotations are provided as requested, the rate for such date will be the arithmetic mean of the rates quoted by major banks in New York City, selected by the Seller, at approximately 11:00 A.M., New York City time, on such date for loans in US dollars to leading European banks for a period of one month commencing on such date and in an amount of US\$ 10,000,000 (US dollars – ten million).

Manufacture Facilities – the various facilities of the Seller, its Affiliates or any subcontractor, where the Airframe or its parts are manufactured or assembled.

Manufacturer Specification Change Notice or MSCN – as defined in Clause 2.2.2.1.

<u>Master Charge Engine</u> – in respect of the A319 NEO Group 1 Aircraft, as set forth in Clause 3.1.1.4(iv); in respect of the A320 NEO Group 1 Aircraft, as set forth in Clause 3.1.1.8(iv).

New Engine Option or NEO – as defined in Clause 2.1.2.1.

P&W – Pratt & Whitney.

<u>Predelivery Payment</u> – any of the payments determined in accordance with Clause 5.3.

Predelivery Payment Reference Price – as defined in Clause 5.3.2.

<u>Propulsion System</u> – any A319 Propulsion System, A320 Propulsion System, A321 Propulsion System, A319 NEO Propulsion System, A320 NEO Propulsion System, as applicable.

<u>Propulsion System Manufacturer</u> – CFM, IAE and P&W, as applicable.

<u>Propulsion System Price Revision Formula</u> – for any Propulsion System, the applicable price revision formula as set forth in Exhibit C-3, C-4 or C-5.

<u>Propulsion System Reference Price</u> – the Engine A Reference Price, the Engine B Reference Price, the Engine C Reference Price, the Engine D Reference Price, the Engine E Reference Price, the Engine F Reference Price, the Engine H Reference Price, the Engine I Reference Price, the Engine I Reference Price, the Engine L Reference Price, as applicable.

<u>Purchase Aircraft</u> – any or all of the one hundred thirty (130) firm A320 Family NEO Aircraft for which the delivery schedule is set forth in Schedule I as of the date of this Agreement to be sold by the Seller and purchased by the Buyer pursuant to this Agreement.

Quarter – the First Quarter, Second Quarter, Third Quarter or Fourth Quarter.

<u>Ready for Delivery</u> – with respect to any Aircraft, when (i) the Technical Acceptance Process has been successfully completed for such Aircraft and (ii) such Aircraft is eligible to receive an Export Certificate of Airworthiness.

Receiving Party - as defined in Clause 5.5.4.

Reference Banks – Citibank, JP Morgan Chase Bank, Deutsche Bank, and such other or additional banking institutions as may be designated from time to time by mutual agreement of the Seller and the Buyer.

Scheduled Delivery Month – as defined in Clause 9.1.

Scheduled Delivery Quarter - as defined in Clause 9.1.

<u>Second Quarter</u> or <u>2nd Quarter</u> – April, May and June of any given calendar year.

<u>Seller Price Revision Formula 2010</u> – the price revision formula set forth in Exhibit C-2.

Seller Price Revision Formula 2011 – the price revision formula set forth in Exhibit C-1.

Seller Price Revision Formula – the Seller Price Revision Formula 2010 or the Seller Price Revision Formula 2011, as applicable.

Seller Representative – as defined in Clause 15.1.1.

Seller Service Life Policy - as defined in Clause 12.2.

<u>Sharklets</u> – a new large wingtip device, currently under development by the Seller, designed to enhance the eco-efficiency and payload range performance of the A320 Family Aircraft, and which are part of the New Engine Option and corresponding Irrevocable SCNs.

Specification – the A319 Specification, the A320 Specification or the A321 Specification, as applicable.

Specification Change Notice or SCN – as defined in Clause 2.2.1.

Standard Specification - the A319 Standard Specification, the A320 Standard Specification or the A321 Standard Specification, as applicable.

Supplier – as defined in Clause 12.3.1.1.

Supplier Part – as defined in Clause 12.3.1.2.

<u>Supplier Product Support Agreement</u> – as defined in Clause 12.3.1.3.

Taxes – as defined in Clause 5.5.5.

Technical Data - as defined in Clause 14.1.

Third Quarter or 3rd Quarter – July, August and September of any given calendar year.

Total Loss - as defined in Clause 10.4.

<u>Training Conference</u> – as defined in Clause 16.1.3.

Type Certificate – as defined in Clause 7.1.

Warranted Part – as defined in Clause 12.1.1.

<u>Warranty Beneficiary</u> – the Buyer; and, with reference herein to any Aircraft that is subject to an assignment pursuant to Clause 21.5 or Clause 21.6, the applicable purchaser or lessee.

Warranty Claim – as defined in Clause 12.1.5.

The definition of a singular in this Clause 0 will apply to the plural of the same word.

Except where otherwise indicated, references in this Agreement to an exhibit, schedule, article, section, subsection or clause refer to the appropriate exhibit or schedule to, or article, section, subsection or clause in this Agreement.

Each agreement defined in this Clause 0 will include all appendixes, exhibits and schedules thereto. If the prior written consent of any person is required hereunder for an amendment, restatement, supplement or other modification to any such agreement and the consent of each such person is obtained, references in this Agreement to such agreement will be to such agreement as so amended, restated, supplemented or modified.

References in this Agreement to any statute will be to such statute as amended or modified and in effect at the time any such reference is operative.

The term "including" when used in this Agreement means "including without limitation" except when used in the computation of time periods.

Technical and trade terms not otherwise defined herein will have the meanings assigned to them as generally accepted in the aircraft manufacturing industry.

# 1 SALE AND PURCHASE; LEASE

The Seller will sell and deliver the Purchase Aircraft to the Buyer, and the Buyer will purchase and take delivery of the Purchase Aircraft from the Seller, subject to the terms and conditions contained in this Agreement. The Seller will sell, or transfer, and deliver the Leased Aircraft to the Lessor, and, subject to the terms and conditions contained in this Agreement, the Buyer will lease from and take delivery of the Leased Aircraft from the Lessor.

### 2 SPECIFICATION

2.1 Aircraft Specification

Each Aircraft will be manufactured in accordance with the Standard Specification applicable to such Aircraft, as modified or varied prior to the date of this Agreement by the Specification Change Notices listed in Exhibits A-2 through A-7.

- 2.1.1 New Engine Option
- 2.1.1.1 The Seller is currently developing a new engine option (the "New Engine Option" or "NEO") that will be applicable to the A319-100, A320-200 and A321-200 model aircraft. The specification of the A320 Family NEO Aircraft will be derived from the relevant Standard Specification and will include (i) as applicable, the Propulsion Systems set forth in Clauses 2.3.4, 2.3.5 and 2.3.6 below, (ii) Sharklets, (iii) airframe structural adaptations, and (iv) Aircraft systems and software adaptations required to operate A320 Family NEO Aircraft. The foregoing is currently reflected in the Irrevocable SCNs, the implementation of which is hereby irrevocably accepted by the Buyer.
- 2.1.1.2 The Irrevocable SCNs include a modification to the design weights of the Standard Specifications as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

It is agreed and understood that the above design weights may be updated upon final NEO specification freeze.

2.2 Specification Amendment

The parties understand and agree that any Specification may be further amended following the execution and delivery by the parties of this Agreement in accordance with the terms of this Clause 2.

2.2.1 Specification Change Notice

Any Specification may be amended by written agreement between the parties substantially in the form set out in Exhibit B-1 (each, a "Specification Change Notice" or "SCN") and will set out the SCN's Aircraft embodiment rank and will also set forth, in detail, the particular change to be made to any Specification and the effect, if any, of such change on design, performance, weight, Delivery Date of the Aircraft affected thereby and on the text of any Specification. An SCN may result in an adjustment of the Base Price of the Aircraft, which adjustment, if any, will be specified in the SCN. In addition, at the request of the Buyer, the Seller will provide to the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Approval of SCNs by the Seller will not be unreasonably withheld, conditioned or delayed.

### 2.2.2 Development Changes

Any Specification may also be amended to incorporate changes deemed necessary by the Seller to improve the Aircraft that is the subject of such Specification, prevent delay or ensure compliance with this Agreement ("**Development Changes**"), as set forth in this Clause 2.

## 2.2.2.1 Manufacturer Specification Changes Notices

Any Specification may be amended by the Seller through a Manufacturer Specification Change Notice ("MSCN"), which will be substantially in the form set out in Exhibit B-2 hereto and will set out the MSCN's Aircraft embodiment rank as well as, in detail, the particular Development Change to be made to such Specification and the effect, if any, of such Development Change on performance, weight, Base Price of the Aircraft, Delivery Date of the Aircraft affected thereby and interchangeability or replaceability requirements under the Specification. In addition, at the request of the Buyer, the Seller will provide to the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The Seller will promptly notify the Buyer in writing of any MSCN applicable to the Aircraft, and such notice will describe the reason for making such amendment and the revised Specification. Except when the MSCN is necessitated by an Aviation Authority directive or by equipment obsolescence, in which case the MSCN will be implemented without requiring the Buyer's consent, if the MSCN adversely affects the performance, weight, Base Price of the Aircraft affected thereby, Delivery Date of the Aircraft affected thereby or the interchangeability or replaceability requirements under the applicable Specification, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

For the purposes of Clause 2.2.2.1, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2.2.2.2 If the Seller revises any Specification to incorporate Development Changes that have no adverse effect on any of the elements identified in Clause 2.2.2.1 above, such Development Change will be performed by the Seller without the Buyer's consent.
- 2.2.2.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUAN TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3 The Buyer will on a regular basis have access to the details of all SCNs, Development Changes and MSCNs related to Aircraft that are subject to this Agreement through the relevant application in AirbusWorld.

- 2.4 Propulsion System
- 2.4.1 Each A319 Airframe will be equipped with a set of two CFM International CFM56-5B6/3 engines or International Aero Engines V2524-A5 engines (such set, upon selection, an "A319 Propulsion System").
- 2.4.2 Each A320 Airframe will be equipped with a set of two CFM International CFM56-5B4/3 engines or International Aero Engines V2527-A5 engines (such set, upon selection, an "A320 Propulsion System").
- 2.4.3 Each A321 Airframe will be equipped with a set of two CFM International CFM56-5B3/3 engines or International Aero Engines V2533-A5 engines (such set, upon selection, an "A321 Propulsion System").
- 2.4.4 Each A319 NEO Airframe will be equipped with a set of two CFM International LEAP-X1A24 engines or Pratt & Whitney PW1124G engines (such set, upon selection, an "A319 NEO Propulsion System").
- 2.4.5 Each A320 NEO Airframe will be equipped with a set of two CFM International LEAP-X1A26 engines or Pratt & Whitney PW1127G engines (such set, upon selection, an "A320 NEO Propulsion System").
- 2.4.6 Each A321 NEO Airframe will be equipped with a set of two CFM International LEAP-X1A32 engines or Pratt & Whitney PW1133G engines (such set, upon selection, an "A321 NEO Propulsion System").
- 2.4.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.5 Milestones
- 2.5.1 Customization Milestones Chart

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] the Seller will provide the Buyer with a customization milestone chart (the "Customization Milestone Chart"), setting out how far in advance of the Scheduled Delivery Month or Scheduled Delivery Quarter, as applicable, of each Aircraft an SCN is required to be executed in order to integrate into the applicable Specification any items requested by the Buyer from the Seller's catalogs of Specification change options or from sources other than the Seller's catalogs of Specification change options.

2.5.2 Contractual Definition Freeze

The Customization Milestone Chart will specify the date(s) by which the contractual definition of each Aircraft is required to be finalized and all SCNs are required to have been executed by the Buyer in order to enable the incorporation of such SCNs into the manufacturing of such Aircraft and Delivery of such Aircraft in the Scheduled Delivery Month or Scheduled Delivery Quarter, as applicable.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] pursuant to the Customization Milestone Chart.

#### 3 PRICE

3.1 Base Price of the Aircraft

The "Base Price" of each Aircraft is the sum of:

- (i) the applicable Base Price of the Airframe, and
- (ii) the applicable Base Price of the Propulsion System.
- 3.1.1 Base Price of the Airframe
- 3.1.1.1 Base Price of the A319 Airframe

The "Base Price of the A319 Airframe" is the sum of the following base prices:

(i) the base price of the A319 Airframe as defined in the A319 Standard Specification (excluding Buyer Furnished Equipment), including nacelles and thrust reversers, is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

(ii) the sum of the base prices of any and all SCNs set forth in Exhibit A-2 is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1.2 Base Price of the A320 Airframe

The "Base Price of the A320 Airframe" is the sum of the following base prices:

(i) the base price of the A320 Airframe as defined in the A320 Standard Specification (excluding Buyer Furnished Equipment), including nacelles and thrust reversers, is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

(ii) the sum of the base prices of any and all SCNs set forth in Exhibit A-3 is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1.3 Base Price of the A321 Airframe

The "Base Price of the A321 Airframe" is the sum of the following base prices:

- (i) the base price of the A321 Airframe as defined in the A321 Standard Specification (excluding Buyer Furnished Equipment), including nacelles and thrust reversers, is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) the sum of the base prices of any and all SCNs set forth in Exhibit A-4 is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1.4 Base Price of the A319 NEO Group 1 Airframe

The "Base Price of the A319 NEO Group 1 Airframe" is the sum of the following base prices:

- (i) the base price of the A319 NEO Airframe as defined in the A319 Standard Specification (excluding Buyer Furnished Equipment), including nacelles and thrust reversers, is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) the sum of the base prices of any and all SCNs (other than Irrevocable SCNs to the extent included in clause (iii) below) set forth in Exhibit A-5 is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) the sum of the base prices of the Irrevocable SCNs set forth in Exhibit A-5 is the sum of:
  - (a) the base price of the New Engine Option is:

    [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (b) the base price of the Sharklets is:

  [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) the base price of the Master Charge Engine is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1.5 Base Price of the A319 NEO Group 2 Airframe

The "Base Price of the A319 NEO Group 2 Airframe" is the sum of the following base prices:

- (i) the base price of the A319 NEO Airframe as defined in the A319 Standard Specification (excluding Buyer Furnished Equipment), including nacelles and thrust reversers, is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) the sum of the base prices of any and all SCNs (other than Irrevocable SCNs to the extent included in clause (iii) below) set forth in Exhibit A-
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) the sum of the base prices of the Irrevocable SCNs set forth in Exhibit A-5 is the sum of:
  - (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1.6 Base Price of the A320 NEO Group 1 Airframe

The "Base Price of the A320 NEO Group 1 Airframe" is the sum of the following base prices:

- (i) the base price of the A320 NEO Airframe as defined in the A320 Standard Specification (excluding Buyer Furnished Equipment), including nacelles and thrust reversers, is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) the sum of the base prices of any and all SCNs (other than Irrevocable SCNs to the extent included in clause (iii) below) set forth in Exhibit A-6 is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) the sum of the base prices of the Irrevocable SCNs set forth in Exhibit A-6 is the sum of:
  - (a) the base price of the New Engine Option is:
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (b) the base price of the Sharklets is:
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) the base prices of the Master Charge Engine is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1.7 Base Price of the A320 NEO Group 2 Airframe

The "Base Price of the A320 NEO Group 2 Airframe" is the sum of the following base prices:

(i) the base price of the A320 NEO Airframe as defined in the A320 Standard Specification (excluding Buyer Furnished Equipment), including nacelles and thrust reversers, is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (ii) the sum of the base prices of any and all SCNs (other than Irrevocable SCNs to the extent included in clause (iii) below) set forth in Exhibit A-6 is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) the sum of the base prices of the Irrevocable SCNs set forth in Exhibit A-6 is the sum of:
  - (a) the base price of the New Engine Option is:

    [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (b) the base price of the Sharklets is:

    [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1.8 Base Price of the A321 NEO Group 1 Airframe

The "Base Price of the A321 NEO Group 1 Airframe" is the sum of the following base prices:

- (i) the base price of the A321 NEO Airframe as defined in the A321 Standard Specification (excluding Buyer Furnished Equipment), including nacelles and thrust reversers, is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) the sum of the base prices of any and all SCNs (other than Irrevocable SCNs to the extent included in clause (iii) below) set forth in Exhibit A-7 is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) the sum of the Irrevocable SCNs set forth in Exhibit A-7 is the sum of:
  - (a) the base price of the New Engine Option is:

    [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (b) the base price of the Sharklets is:

  [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) the base price of the Master Charge Engine at delivery conditions prevailing in January 2010 is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1.9 Base Price of the A321 NEO Group 2 Airframe

The "Base Price of the A321 NEO Group 2 Airframe" is the sum of the following base prices:

- (i) the base price of the A321 NEO Airframe as defined in the A321 Standard Specification (excluding Buyer Furnished Equipment), including nacelles and thrust reversers, is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) the sum of the base prices of any and all SCNs (other than Irrevocable SCNs the extent included in clause (iii) below) set forth in Exhibit A-7 is:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) the sum of the Irrevocable SCNs set forth in Exhibit A-7 is the sum of:
  - (a) the base price of the New Engine Option is:

    [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (b) the base price of the Sharklets is:

    [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.1.2 Base Price of Propulsion System
- 3.1.2.1 The base price (the "Engine A Base Price") of a set of two (2) CFM International CFM56-5B6/3 engines is:

Said base price has been established in accordance with the delivery conditions prevailing in January 2011 and has been calculated from the reference price (the "Engine A Reference Price") indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.2.2 The base price (the "Engine B Base Price") of a set of two (2) CFM International CFM56-5B4/3 engines is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Said base price has been established in accordance with the delivery conditions prevailing in January 2011 and has been calculated from the reference price (the "Engine B Reference Price") indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.2.3 The base price (the "Engine C Base Price") of a set of two (2) CFM International CFM56-5B3/3 engines is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Said base price has been established in accordance with the delivery conditions prevailing in January 2011 and has been calculated from the reference price (the "Engine C Reference Price") indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REOUEST FOR CONFIDENTIAL TREATMENT!

3.1.2.4 The base price (the "Engine D Base Price") of a set of two (2) International Aero Engines V2524-A5 engines is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Said base price has been established in accordance with the delivery conditions prevailing in January 2011 and has been calculated from the reference price (the "Engine D Reference Price") indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.2.5 The base price (the "Engine E Base Price") of a set of two (2) International Aero Engines V2527-A5 engines is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Said base price has been established in accordance with the delivery conditions prevailing in January 2011 and has been calculated from the reference price (the "Engine E Reference Price") indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.2.6 The base price (the "Engine F Base Price") of a set of two (2) International Aero Engines V2533-A5 engines is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Said base price has been established in accordance with the delivery conditions prevailing in January 2011 and has been calculated from the reference price (the "Engine F Reference Price") indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.2.7 The base price (the "Engine G Base Price") of a set of two (2) CFM International LEAP-X1A24 engines is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Said base price has been established in accordance with the delivery conditions prevailing in January 2010 and has been calculated from the reference price (the "Engine G Reference Price") indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.2.8 The base price (the "Engine H Base Price") of a set of two (2) CFM International LEAP-X1A26 engines:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Said base price has been established in accordance with the delivery conditions prevailing in January 2010 and has been calculated from the reference price (the "Engine H Reference Price") is indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.2.9 The base price (the "Engine I Base Price") of a set of two (2) CFM International LEAP-X1A32 engines is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Said base price has been established in accordance with the delivery conditions prevailing in January 2010 and has been calculated from the reference price (the "Engine I Reference Price") indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.2.10 The base price (the "Engine J Base Price") of a set of two (2) Pratt & Whitney PW1124G engines is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Said base price has been established in accordance with the delivery conditions prevailing in January 2010 and has been calculated from the reference price (the "Engine J Reference Price") indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.2.11 The base price (the "Engine K Base Price") of a set of two (2) Pratt & Whitney PW1127G engines is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Said base price has been established in accordance with the delivery conditions prevailing in January 2010 and has been calculated from the reference price (the "Engine K Reference Price") indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.2.12 The base price (the "Engine L Base Price") of a set of two (2) Pratt & Whitney PW1133G engines is:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Said base price has been established in accordance with the delivery conditions prevailing in January 2010 and has been calculated from the reference price (the "Engine L Reference Price") indicated by the Propulsion System Manufacturer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 3.2 Final Price of the Aircraft

The "Final Price" of each Aircraft will be the sum of:

- (i) the applicable Base Price of the Airframe, as adjusted to the applicable Delivery Date of such Aircraft in accordance with Clause 4.1;
- (ii) the aggregate of all increases or decreases to the applicable Base Price of the Airframe as agreed in any Specification Change Notice,
  Manufacturer Specification Change Notice, or part thereof applicable to such Airframe subsequent to the date of this Agreement as adjusted to
  the Delivery Date of such Aircraft in accordance with Clause 4.1;
- (iii) the Propulsion System Reference Price for the Propulsion System for such Aircraft, adjusted to the Delivery Date in accordance with Clause 4.2:
- (iv) the aggregate of all increases or decreases of such Propulsion System Reference Price as agreed in any Specification Change Notice or part thereof applicable to such Propulsion System subsequent to the date of this Agreement as adjusted to the Delivery Date in accordance with Clause 4.2;
- (v) any other amount resulting from any other provisions of this Agreement and/or any other written agreement between the Buyer and the Seller relating to such Aircraft.

### 4 PRICE REVISION

4.1 Seller Price Revision Formula

For each Airframe, the Base Price of the Airframe is subject to revision up to and including the applicable month of Delivery in accordance with the Seller Price Revision Formula.

- 4.2 Propulsion System Price Revision
- 4.2.1 The Propulsion System Reference Price applicable to any Propulsion System is subject to revision up to and including the applicable Delivery Date in accordance with the applicable Propulsion System Price Revision Formula.
- 4.2.2 The Propulsion System Reference Price for each Propulsion System, the prices of the related equipment and the Propulsion System Price Revision Formula are based on information received from the applicable Propulsion System Manufacturer and are subject to amendment by such Propulsion System Manufacturer at any time prior to Delivery. If any Propulsion System Manufacturer makes any such amendment, the amendment will be deemed to be incorporated into this Agreement and the Propulsion System Reference Price, the prices of the related equipment and the Propulsion System Price Revision Formula will be adjusted accordingly. The Seller agrees to notify the Buyer as soon as the Seller receives notice of any such amendment from any Propulsion System Manufacturer.

### 5 PAYMENT TERMS

5.1 Method and Place of Payment

The Buyer will perform its obligations to pay the Predelivery Payments, the Balance of the Final Price and any other amount due hereunder from the United States in immediately available funds in United States dollars to:

Beneficiary Name: AIRBUS

account identification: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

with

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

or to such other account as may be designated by the Seller.

- 5.2 INTENTIONALLY LEFT BLANK
- 5.3 Predelivery Payments
- 5.3.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5.3.2 The "**Predelivery Payment Reference Price**" for an Aircraft to be delivered in calendar year T is determined in accordance with the following formula:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

5.3.3 Predelivery Payments will be paid according to the following schedule.

CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

TOTAL PAYMENT PRIOR TO DELIVERY

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 5.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5.3.5 Specification Change Notice Predelivery Payments

The Seller will be entitled to require, and the Buyer will pay, Predelivery Payments for each SCN (excluding those listed in Exhibits A-2 to A-7) executed after execution and delivery of this Agreement in accordance with the following payment schedule:

- (i) For each SCN executed [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) For each SCN executed [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (iii) For each SCN executed [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5.4 Payment of Balance of the Final Price of the Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (the "Balance of the Final Price").

- 5.5 Taxes
- 5.5.1 The amounts stated in this Agreement to be payable by the Buyer are exclusive of value added tax ("VAT") chargeable under the laws of any jurisdiction and accordingly the Buyer will pay any VAT chargeable with respect to the sale of any Aircraft other than Leased Aircraft, and any component, accessory, equipment, part or service delivered or furnished under this Agreement with respect to a Purchase Aircraft. The Seller will comply with all local laws regarding VAT collection and remittance, including without limitation the issuance of timely and proper VAT invoices and, where applicable, the issuance of local invoices and accepting local payment.
- 5.5.2 The Seller will pay all other Taxes (except for Taxes based on or measured by the income of the Buyer or any Taxes levied against the Buyer for the privilege of doing business in any jurisdiction), levied, assessed, charged or collected, on or prior to Delivery of any Aircraft, for or in connection with the manufacture, assembly, sale and delivery under this Agreement of such Aircraft or any parts, instructions or data installed thereon or incorporated therein (other than Buyer Furnished Equipment).
- 5.5.3 The Buyer will pay all Taxes not assumed by the Seller under Clause 5.5.2 (except for Taxes based on or measured by the income of the Seller or any Taxes levied against the Seller for the privilege of doing business in any jurisdiction).
- 5.5.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 5.5.5 "Taxes" means any present or future tax, stamp or other levies, imposts, duties, charges, fees, deductions or withholdings arising from the transactions contemplated or effectuated under this Agreement, now or hereafter imposed, levied, collected, withheld or assessed by any governmental authority or any political subdivision or taxing authority thereof or therein, including without limitation any exactions based upon or measured by the assets, capital, property, gross receipts or payroll of the Buyer or the Seller.
- 5.6 Application of Payments

Notwithstanding any other rights the Seller may have at contract or at law, the Buyer and the Seller hereby agree that should any matured and undisputed amounts [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] become due and payable by the Buyer or its Affiliates, and not be paid in full in immediately available funds on the date due [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] then the Seller will have the right, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] debit and apply, in whole or in part, the Predelivery Payments paid to the Seller by the Buyer against such unpaid amount(s). The Seller will promptly notify the Buyer in writing after such debiting and application, and the Buyer will pay to the Seller the amount required to comply with Clause 5.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 5.7 Setoff Payments
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5.8 Overdue Payments

## 5.9 Proprietary Interest

Notwithstanding any provision of law to the contrary, the Buyer will not, by virtue of anything contained in this Agreement (including, without limitation, any Predelivery Payments hereunder, or any designation or identification by the Seller of a particular aircraft as an Aircraft to which any of the provisions of this Agreement refers) acquire any proprietary, insurable or other interest whatsoever in any Aircraft before Delivery of and payment for such Aircraft, as provided in this Agreement.

## 5.10 Payment in Full

Except as expressly provided in Clause 5.7, the Buyer's obligation to make payments to the Seller hereunder will not be affected by and will be determined without regard to any counterclaim, recoupment, defense or other right that the Buyer may have against the Seller or any other person and, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] all such payments will be made without deduction or withholding of any kind. The Buyer will ensure that the sums received by the Seller under this Agreement will be equal to the full amounts expressed to be due to the Seller hereunder, without deduction or withholding on account of and free from any and all taxes, levies, imposts, duties or charges of whatever nature, except that if the Buyer is required by law to make any such deduction or withholding the Buyer will pay an additional amount to the Seller [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 5.11 Other Charges

Unless expressly stipulated otherwise, any charges due from the Buyer under this Agreement other than those set out in Clauses 5.3 and 5.8 will be paid by the Buyer at the same time as payment of the Balance of the Final Price or, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 6 MANUFACTURE PROCEDURE – INSPECTION

#### 6.1 Manufacture Procedures

Each Airframe will be manufactured in accordance with this Agreement and the requirements of the laws of the jurisdiction of incorporation of the Seller or of its relevant Affiliate as enforced by the Aviation Authority of such jurisdiction.

# 6.2 Inspection

- 6.2.1 Subject to providing the Seller with certificates evidencing compliance with the insurance requirements set forth in Clause 19, the Buyer or its duly authorized representatives (the "Buyer's Inspector(s)") will be entitled to inspect the manufacture of each Airframe and all materials and parts obtained by the Seller from the sub-contractors for the manufacture of each Airframe, receive reasonable notice of and attend inspection presentations, have access to technical log books and quality log books and monitor production acceptance tests (the "Inspection") on the following terms and conditions:
  - (i) any Inspection will be conducted pursuant to the Seller's system of inspection and the relevant procedures of the Seller, as developed under the supervision of the relevant Aviation Authority;
  - (ii) the Buyer's Inspector(s) will have access to such relevant technical documentation as is reasonably necessary for the purpose of the Inspection;
  - (iii) any Inspection and any related discussions with the Seller by the Buyer's Inspector(s) will be at reasonable times during business hours and will take place in the presence of the relevant inspection department personnel of the Seller; and
  - (iv) the Inspections will be performed in a manner not to unduly delay or hinder the manufacture or assembly of any Aircraft or the performance of this Agreement by the Seller or any other work in progress at the Manufacture Facilities.

### 6.2.2 Location of Inspections

The Buyer's Inspector(s) will be entitled to conduct any such Inspection at the relevant Manufacture Facility of the Seller or the Affiliates and, if requested by the Buyer, at the Manufacture Facilities of the sub-contractors; provided that if access to any part of the Manufacture Facilities where the manufacture of the Airframe or its materials or parts is in progress or materials or parts are stored is restricted for security or confidentiality reasons, the Seller will be allowed reasonable time to make the relevant items available in another location reasonably convenient for the Buyer's Inspector(s).

## 6.3 Seller's Service for Buyer's Inspector(s)

For the purpose of the Inspections, and starting from a mutually agreed date [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] until the final Delivery Date, the Seller will furnish without additional charge suitable space and office equipment (including computer communication access) in or conveniently located with respect to each Delivery Location for the use of up to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the Buyer's Inspectors at each such location.

6.4 No Effect on Representations, Warranties, Indemnities and Agreements

The representations, warranties, indemnities and agreements of the Seller in this Agreement and any exceptions or limitations thereto will not be affected or deemed waived by reason of any investigation or inspection made or not made by the Buyer pursuant to this Clause 6.

## 7 CERTIFICATION

Except as set forth in this Clause 7, the Seller will not be required to obtain any certificate or approval with respect to any Aircraft.

7.1 Type Certification

The Aircraft have been type certificated under EASA procedures for joint certification in the transport category. The Seller will obtain or cause to be obtained an FAA type certificate (the "**Type Certificate**") to allow the issuance of the Export Certificate of Airworthiness.

7.2 Export Certificate of Airworthiness

Subject to the provisions of Clause 7.3, each Aircraft will be delivered to the Buyer with an Export Certificate of Airworthiness and in a condition enabling the Buyer to obtain at the time of Delivery a Standard Airworthiness Certificate issued pursuant to Part 21 of the US Federal Aviation Regulations and a Certificate of Sanitary Construction issued by the U.S. Public Health Service of the Food and Drug Administration. However, the Seller will have no obligation to make and will not be responsible for any costs of alterations or modifications to such Aircraft to enable such Aircraft to meet FAA or U.S. Department of Transportation requirements for specific operation on the Buyer's routes, whether before, at or after Delivery of any Aircraft.

If the FAA requires additional or modified data before the issuance of the Export Certificate of Airworthiness, the Seller will provide such data or implement the required modification to the data, in either case, at the Buyer's cost.

- 7.3 Specification Changes before Aircraft Ready for Delivery
- 7.3.1 If, any time before the date on which an Aircraft is Ready for Delivery, any law, rule or regulation is enacted, promulgated, becomes effective and/or an interpretation of any law, rule or regulation is issued by EASA that requires any change to the Specification for the purposes of obtaining the Export Certificate of Airworthiness (a "Change in Law"), the Seller will make the required modification and the parties hereto will sign an SCN pursuant to Clause 2.2.1.
- 7.3.2 The Seller will as far as practicable, but at its sole discretion and without prejudice to Clause 7.3.3(ii), take into account the information available to it concerning any proposed law, rule or regulation or interpretation that could become a Change in Law, in order to minimize the costs of changes to the Specification as a result of such proposed law, regulation or interpretation becoming effective before the applicable Aircraft is Ready for Delivery.
- 7.3.3 The cost of implementing the required modifications referred to in Clause 7.3.1 will be:
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3.4 Notwithstanding the provisions of Clause 7.3.3, if a Change in Law relates to an item of BFE or to the Propulsion System the costs related thereto will be borne by the Buyer, and the Seller will have no obligation with respect thereto.

# 7.4 Specification Changes after Aircraft Ready For Delivery

Nothing in Clause 7.3 will require the Seller to make any changes or modifications to, or to make any payments or take any other action with respect to, any Aircraft that is Ready for Delivery before the compliance date of any law or regulation referred to in Clause 7.3. Any such changes or modifications made to an Aircraft after it is Ready for Delivery will be at the Buyer's expense.

## 8 TECHNICAL ACCEPTANCE

- 8.1 Technical Acceptance Process
- 8.1.1 Prior to Delivery, each Aircraft will undergo a technical acceptance process developed by the Seller [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (the "Technical Acceptance Process"). Completion of the Technical Acceptance Process will demonstrate the satisfactory functioning of such Aircraft and will be deemed to demonstrate compliance with the Specification. If, during the Technical Acceptance Process any Aircraft does not comply with all of the Technical Acceptance Process requirements, the Seller will without hindrance from the Buyer carry out, at its cost, any necessary changes and, as soon as practicable thereafter, resubmit such Aircraft to such further Technical Acceptance Process as is necessary to demonstrate the elimination of the non-compliance.
- 8.1.2 The Technical Acceptance Process will:
  - (i) commence on a date notified by the Seller to the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) take place at the Delivery Location,
  - (iii) be carried out by the personnel of the Seller with the participation of the Buyer's representatives in accordance with the Technical Acceptance Process and Clause 8.2.2 if the Buyer elects to attend the Technical Acceptance Process, and
  - (iv) include [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (the "Technical Acceptance Flight").
- 8.2 Buyer's Attendance
- 8.2.1 The Buyer is entitled to attend the Technical Acceptance Process.
- 8.2.2 If the Buyer elects to attend the Technical Acceptance Process, the Buyer:
  - (i) will comply with the reasonable requirements of the Seller, with the intention of completing the Technical Acceptance Process within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) may have a maximum of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of whom will have access to the cockpit at any one time) accompany the Seller's representatives on the Technical Acceptance Flight, during which the Buyer's representatives will comply with the instructions of the Seller's representatives; and
  - (iii) will be entitled to have a pilot of the Buyer conduct routine flight maneuvers and tests during the Technical Acceptance Flight as may be reasonably required to demonstrate to

the Buyer the satisfactory functioning of the Aircraft, subject to the supervision and operational control of the Seller's flight test personnel at all times during such flight maneuvers and tests.

- 8.2.3 If the Buyer does not attend or fails to cooperate in the Technical Acceptance Process, the Seller will be entitled to complete the Technical Acceptance Process and the Buyer will be deemed to have accepted that the Technical Acceptance Process has been satisfactorily completed, in all respects.
- 8.3 Certificate of Acceptance

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] the Buyer will execute and deliver to the Seller a certificate of acceptance in respect of such Aircraft in the form of Exhibit D (the "Certificate of Acceptance").

8.4 Finality of Acceptance

The Buyer's execution and delivery of the Certificate of Acceptance for an Aircraft will constitute waiver by the Buyer of any right it may have, under the Uniform Commercial Code as adopted by the State of New York or otherwise, to revoke acceptance of each Aircraft for any reason known to the Buyer at the time of acceptance. The Buyer's execution and delivery of the Certificate of Acceptance for an Aircraft will not impair the Seller's obligations pursuant to any warranty, indemnity or performance guarantee or other support set forth in the Agreement or any remedies of the Buyer thereunder.

8.5 Aircraft Utilization

The Seller will, without payment or other liability, be entitled to use any and all Aircraft before Delivery of such Aircraft as may be necessary to obtain the certificates required under Clause 7. Such use will not limit the Buyer's obligation to accept Delivery hereunder.

## 9 DELIVERY

9.1 Delivery Schedule

Subject to Clauses 2, 7, 8, 10 and 18, the Seller will have the Aircraft Ready for Delivery at the Delivery Location within the following months (each, a "Scheduled Delivery Month") or quarters (each, a "Scheduled Delivery Quarter"), as applicable set forth on Schedule I hereto (the "Delivery Schedule").

- 9.1.1 In respect of each Aircraft for which a Scheduled Delivery Quarter is set forth in Clause 9.1 above, the Seller will communicate to the Buyer the Scheduled Delivery Month no later than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 9.1.2 The Seller will give the Buyer at least [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] written notice of the anticipated date on which each Aircraft will be Ready for Delivery. Such notice will also include the starting date and the planned schedule of the Technical Acceptance Process. Thereafter the Seller will notify the Buyer of any change to such dates.
- 9.2 Delivery Process
- 9.2.1 The Buyer will send its representatives (which, with respect to Leased Aircraft, may include representatives of the Lessor) to the Delivery Location to take Delivery of each Aircraft at the date on which such Aircraft is Ready for Delivery, and fly such Aircraft from the Delivery Location.
- 9.2.2 The Seller will deliver and transfer title to each Aircraft to the Buyer (or, with respect to a Leased Aircraft, the Lessor) free and clear of all encumbrances (except for any liens or encumbrances created by or on behalf of the Buyer) provided that the Balance of the Final Price of such Aircraft has been paid by the Buyer pursuant to Clause 5.4 (or, with respect to a Leased Aircraft, provided that Buyer has executed and delivered to Lessor a definitive lease agreement) and that the Certificate of Acceptance has been signed and delivered to the Seller pursuant to Clause 8.3. The Seller will provide the Buyer (or with respect to Leased Aircraft, the Lessor) with a bill of sale in the form of Exhibit E (the "Bill of Sale") and such other documentation confirming transfer of title and receipt of such Final Price of such Aircraft as may reasonably be requested by the Buyer (and, with respect to Leased Aircraft, to the Lessor). Title to and risk of loss of or damage to such Aircraft will pass to the Buyer (or, with respect to Leased Aircraft, the Lessor) contemporaneously with the delivery by the Seller to the Buyer of such Bill of Sale.
- 9.2.3 (i) If the Buyer fails to deliver the signed Certificate of Acceptance with respect to an Aircraft to the Seller when required pursuant to Clause 8.3, or (ii) if Buyer fails to pay the Balance of the Final Price of such Aircraft to the Seller (or with respect to a Leased Aircraft, if Buyer fails to execute and deliver to Lessor a definitive lease agreement), then the Buyer will be deemed to have rejected Delivery wrongfully when such Aircraft was duly tendered to the Buyer hereunder. If such a deemed rejection arises, then in addition to the remedies of Clause 5.8.1, the Seller will retain title to such Aircraft and the Buyer will indemnify and hold the Seller harmless against any and all costs (including but not limited to any parking, storage, and insurance costs) and consequences resulting from the Buyer's rejection (including but not limited to risk of loss of or damage to such Aircraft), it being understood that the Seller will be under no duty to the Buyer to store, park, insure or otherwise protect such Aircraft. These rights of the Seller will be in addition to the Seller's other rights and remedies in this Agreement.

0.3	Flyzwa

- 9.3.1 The Buyer and the Seller will cooperate to obtain any licenses that may be required by the Aviation Authority of the Delivery Location for the purpose of exporting the Aircraft.
- 9.3.2 All expenses of, or connected with, flying the Aircraft from the Delivery Location after Delivery will be borne by the Buyer. The Buyer will make direct arrangements with the supplying companies for the fuel and oil required for all post-Delivery flights.

## 10 EXCUSABLE DELAY AND TOTAL LOSS

10.1 Scope of Excusable Delay

Neither the Seller nor any Affiliate of the Seller, will be responsible for or be deemed to be in default on account of delays in delivery of any Aircraft or failure to deliver or otherwise in the performance of this Agreement or any part hereof due to causes beyond the Seller's, or any Affiliate's control or not occasioned by the Seller's, fault or negligence ("Excusable Delay"), including, but not limited to: (i) acts of God or the public enemy, natural disasters, fires, floods, storms beyond ordinary strength, explosions or earthquakes; epidemics or quarantine restrictions; serious accidents; any law, decision, regulation, directive or other act (whether or not having the force of law) of any government or of the Council of the European Community or the Commission of the European Community or of any national, Federal, State, municipal or other governmental department, commission, board, bureau, agency, court or instrumentality, domestic or foreign; governmental priorities, regulations or orders affecting allocation of materials, facilities or a completed Aircraft; war, civil war or warlike operations, terrorism, insurrection or riots; failure of transportation; strikes or labor troubles causing cessation, slow down or interruption of work; [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] inability after due and timely diligence to procure materials, accessories, equipment or parts; general hindrance in transportation; or failure of a subcontractor or supplier to furnish materials, components, accessories, equipment or parts; (ii) any delay caused directly or indirectly by the action or inaction of the Buyer; and (iii) delay in delivery or otherwise in the performance of this Agreement by the Seller due in whole or in part to any delay in or failure of the delivery of, or any other event or circumstance relating to, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

10.2 Consequences of Excusable Delay

If an Excusable Delay occurs:

- (i) the Seller will notify the Buyer of such Excusable Delay as soon as practicable after becoming aware of the same;
- (ii) the Seller will not be responsible for any damages arising from or in connection with such Excusable Delay suffered or incurred by the Buyer;
- (iii) the Seller will not be deemed to be in default in the performance of its obligations hereunder as a result of such Excusable Delay;
- (iv) the Seller will as soon as practicable after the removal of the cause of such delay resume performance of its obligations under this Agreement and in particular will notify the Buyer of the revised Scheduled Delivery Month.
- 10.3 Termination on Excusable Delay
- 10.3.1 If any Delivery is delayed as a result of an Excusable Delay for a period of more than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after the

last day of the Scheduled Delivery Month, then either party may terminate this Agreement with respect to the affected Aircraft, by giving written notice to the other party within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] However, the Buyer will not be entitled to terminate this Agreement pursuant to this Clause 10.3.1 if the Excusable Delay is caused directly or indirectly by the action or inaction of the Buyer.

- 10.3.2 If the Seller advises the Buyer in its notice of a revised Scheduled Delivery Month pursuant to Clause 10.2.1(iv) that there will be a delay in Delivery of an Aircraft of more than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after the last day of the Scheduled Delivery Month, then [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] may terminate this Agreement with respect to the affected Aircraft. Termination will be made by giving written notice to the other party within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after the Buyer's receipt of the notice of a revised Scheduled Delivery Month.
- 10.3.3 If this Agreement is not terminated under the terms of Clause 10.3.1 or 10.3.2, then the Seller will be entitled to reschedule Delivery. The Seller will notify the Buyer of the new Scheduled Delivery Month after the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] period referred to in Clause 10.3.1 or 10.3.2, and this new Scheduled Delivery Month will be deemed to be an amendment to the applicable Scheduled Delivery Month in Clause 9.1.
- 10.4 Total Loss, Destruction or Damage
  - If, prior to Delivery, any Aircraft is lost, destroyed or in the reasonable opinion of the Seller is damaged beyond economic repair ("Total Loss"), the Seller will notify the Buyer to this effect within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of such occurrence. The Seller will include in said notification (or as soon after the issue of the notice as such information becomes available to the Seller) the earliest date consistent with the Seller's other commitments and production capabilities that an aircraft to replace such Aircraft may be delivered to the Buyer and the Scheduled Delivery Month will be extended as specified in the Seller's notice to accommodate the delivery of the replacement aircraft; provided, however, that if the Scheduled Delivery Month is extended to a month that is later than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after the last day of the original Scheduled Delivery Month then this Agreement will terminate with respect to said Aircraft unless:
  - (i) the Buyer notifies the Seller within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the date of receipt of the Seller's notice that it desires the Seller to provide a replacement aircraft during the month quoted in the Seller's notice; and

(ii) the parties execute an amendment to this Agreement recording the change in the Scheduled Delivery Month.

Nothing herein will require the Seller to manufacture and deliver a replacement aircraft if such manufacture would require the reactivation of its production line for the model or series of aircraft that includes the applicable Aircraft.

10.5 Termination Rights Exclusive

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

10.6 Remedies

THIS CLAUSE 10 SETS FORTH THE SOLE AND EXCLUSIVE REMEDY OF THE BUYER FOR DELAYS IN DELIVERY OR FAILURE TO DELIVER, OTHER THAN SUCH DELAYS AS ARE COVERED BY CLAUSE 11, AND THE BUYER HEREBY WAIVES ALL RIGHTS TO WHICH IT WOULD OTHERWISE BE ENTITLED IN RESPECT THEREOF, INCLUDING, WITHOUT LIMITATION, ANY RIGHTS TO INCIDENTAL AND CONSEQUENTIAL DAMAGES OR SPECIFIC PERFORMANCE. THE BUYER WILL NOT BE ENTITLED TO CLAIM THE REMEDIES AND RECEIVE THE BENEFITS PROVIDED IN THIS CLAUSE 10 WHERE THE DELAY REFERRED TO IN THIS CLAUSE 10 IS CAUSED BY THE NEGLIGENCE OR FAULT OF THE BUYER OR ITS REPRESENTATIVES.

## 11 INEXCUSABLE DELAY

# 11.1 Liquidated Damages

Should an Aircraft not be Ready for Delivery within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after the last day of the Scheduled Delivery Month (as such month may be changed pursuant to Clauses 2, 7 and/or 10) (the "Delivery Period") and such delay is not as a result of an Excusable Delay or Total Loss, then such delay will be termed an "Inexcusable Delay." In the event of an Inexcusable Delay, the Buyer will have the right to claim, and the Seller will pay the Buyer liquidated damages of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

In no event will the amount of liquidated damages exceed the total of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] in respect of any one Aircraft.

The Buyer's right to liquidated damages in respect of an Aircraft is conditioned on the Buyer's submitting a written claim for liquidated damages to the Seller not later than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after the last day of the Scheduled Delivery Month.

# 11.2 Renegotiation

If, as a result of an Inexcusable Delay, the Delivery does not occur within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after the last day of the Delivery Period the Buyer will have the right, exercisable by written notice to the Seller given between [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to require from the Seller a renegotiation of the Scheduled Delivery Month for the affected Aircraft. Unless otherwise agreed between the Seller and the Buyer during such renegotiation, the said renegotiation will not prejudice the Buyer's right to receive liquidated damages in accordance with Clause 11.1.

#### 11.3 Termination

If, as a result of an Inexcusable Delay, the Delivery does not occur within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after the last day of the Delivery Period and the parties have not renegotiated the Delivery Date pursuant to Clause 11.2, then both parties will have the right exercisable by written notice to the other party, given between [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to terminate this Agreement in respect of the affected Aircraft. In the event of termination, neither party will have any claim against the other, except that the Seller will pay to the Buyer any amounts due pursuant to Clause 11.1 and will pay to the Buyer an amount equal to the Predelivery Payments received from the Buyer hereunder in respect of such affected Aircraft.

## 11.4 Remedies

THIS CLAUSE 11 SETS FORTH THE SOLE AND EXCLUSIVE REMEDY OF THE BUYER FOR DELAYS IN DELIVERY OR FAILURE TO DELIVER, OTHER THAN SUCH DELAYS AS ARE COVERED BY CLAUSE 10, AND THE BUYER HEREBY WAIVES ALL RIGHTS TO WHICH IT WOULD OTHERWISE BE ENTITLED IN RESPECT THEREOF, INCLUDING WITHOUT LIMITATION ANY RIGHTS TO INCIDENTAL AND CONSEQUENTIAL DAMAGES OR SPECIFIC PERFORMANCE. THE BUYER WILL NOT BE ENTITLED TO CLAIM THE REMEDIES AND RECEIVE THE BENEFITS PROVIDED IN THIS CLAUSE 11 WHERE THE DELAY REFERRED TO IN THIS CLAUSE 11 IS CAUSED BY THE NEGLIGENCE OR FAULT OF THE BUYER OR ITS REPRESENTATIVES.

## 12 WARRANTIES AND SERVICE LIFE POLICY

This Clause covers the terms and conditions of the warranty and service life policy.

# 12.1 Standard Warranty

#### 12.1.1 Nature of Warranty

For the purpose of this Agreement the term "Warranted Part" will mean any Seller proprietary component, equipment, system, software (to the extent not covered by the End-User License Agreement for such software), accessory, structure or part, that is installed on an Aircraft at Delivery thereof and

- (a) that is manufactured to the detailed design of the Seller or a subcontractor of the Seller and
- (b) that bears a part number of the Seller at the time of such Delivery.

Subject to the conditions and limitations as hereinafter provided for and except as provided for in Clause 12.1.2, the Seller warrants to the Warranty Beneficiary that each Aircraft and each Warranted Part will at Delivery to such Warranty Beneficiary be free from defects:

- (i) in material;
- (ii) in workmanship, including without limitation processes of manufacture;
- (iii) in design (including without limitation the selection of materials) having regard to the state of the art at the date of such design; and
- (iv) arising from failure to conform to the Specification, except to those portions of the Specification relating to performance or where it is expressly stated that they are estimates or approximations or design aims.

## 12.1.2 Exclusions

The warranties set forth in Clause 12.1.1 will not apply to Buyer Furnished Equipment, nor to the Propulsion System, nor to any component, equipment, accessory or part installed on any Aircraft at Delivery that is not a Warranted Part except that:

- (i) any defect in the Seller's workmanship in respect of the installation of such items in such Aircraft, including any failure by the Seller to conform to the installation instructions of the manufacturers of such items, that invalidates any applicable warranty from such manufacturers, will constitute a defect in workmanship for the purpose of this Clause 12.1 and be covered by the warranty set forth in Clause 12.1.1 (ii); and
- (ii) any defect inherent in the Seller's design of the installation, in consideration of the state of the art at the date of such design, that impairs the use of such items, will constitute a defect in design for the purpose of this Clause 12.1 and be covered by the warranty set forth in Clause 12.1.1 (iii).

## 12.1.3 Warranty Period

The warranties set forth in Clauses 12.1.1 and 12.1.2 will be limited to those defects that become apparent within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after Delivery of the affected Aircraft (the "Warranty Period").

# 12.1.4 Limitations of Warranty

- 12.1.4.1 The Warranty Beneficiary's remedy and the Seller's obligation and liability under Clauses 12.1.1 and 12.1.2 are, at the Seller's expense, (a) the repair, replacement or correction of any Warranted Part that is defective (or the supply of modification kits rectifying the defect) (a "Correction") or (b) the reimbursement of the cost to the Warranty Beneficiary for a replacement for the defective Warranted Part and any other materials installed, incorporated, consumed or expended by the Warranty Beneficiary in performing the Correction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 12.1.4.2 In the event of a defect covered by Clauses 12.1.1 (iii), 12.1.1 (iv) and 12.1.2 (ii) becoming apparent within the Warranty Period, the Seller will also, if so requested by the Buyer in writing, correct such defect in any Aircraft that has not yet been delivered to the Buyer, provided, however,
  - (i) that the Seller will not be responsible, nor be deemed to be in default on account of any reasonable delay in Delivery of any Aircraft or otherwise in respect of the performance of this Agreement, due to the Seller's undertaking to make such correction and provided further
  - (ii) that, rather than accept a delay in the Delivery of any such Aircraft, the Buyer and the Seller may agree to deliver such Aircraft with subsequent correction of the defect by the Buyer at the Seller's expense, or the Buyer may elect to accept Delivery and thereafter file a Warranty Claim as though the defect had become apparent immediately after Delivery of such Aircraft.

### 12.1.4.3 Cost of Inspection

In addition to the remedies set forth in Clauses 12.1.4.1 and 12.1.4.2, the Seller will reimburse the direct labor costs incurred by the Warranty Beneficiary in performing inspections of the affected Aircraft to determine whether or not a defect exists in any Warranted Part within the Warranty Period subject to the following conditions:

- (i) such inspections are recommended by a Seller Service Bulletin, service letter, or other Seller correspondence issued or to be performed within the Warranty Period;
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) the labor rate for the reimbursement will be the In-house Warranty Labor Rate; and
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 12.1.5 Warranty Claim Requirements

The Warranty Beneficiary's remedy and the Seller's obligation and liability under this Clause 12.1 with respect to any warranty claim submitted by the Buyer (each a "Warranty Claim") are subject to the following conditions:

- (i) the defect having become apparent within the Warranty Period;
- (ii) the Warranty Beneficiary having filed a Warranty Claim within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and
- (iii) the Seller having received a Warranty Claim complying with the provisions of Clause 12.1.6 below.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 12.1.6 Warranty Administration

The warranties set forth in Clause 12.1 will be administered as hereinafter provided for:

### 12.1.6.1 Claim Determination

Determination as to whether any claimed defect in any Warranted Part is a valid Warranty Claim will be made by the Seller and will be based upon the claim details, reports from the Seller's Representatives, historical data logs, inspections, tests, findings during repair, defect analysis and other relevant documents; [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 12.1.6.2 Transportation Costs

The cost of transporting a Warranted Part claimed to be defective to the facilities designated by the Seller [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 12.1.6.3 Return of an Aircraft

If the Warranty Beneficiary and the Seller mutually agree, prior to such return, that it is necessary to return an Aircraft to the Seller for consideration of a Warranty Claim, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 12.1.6.4 On Aircraft Work by the Seller

If the Seller determines that a defect subject to this Clause 12.1 justifies the dispatch by the Seller of a working team to repair or correct such defect through the embodiment of one or several Seller's Service Bulletins at the Buyer's facilities, or if the Seller accepts the return of an Aircraft to perform or have performed such repair or correction, then the labor costs for such on-Aircraft work will be borne by the Seller at the In-House Warranty Labor Rate.

The condition which has to be fulfilled for on-Aircraft work by the Seller is that, in the opinion of the Seller, the work necessitates the technical expertise of the Seller as manufacturer of the affected Aircraft.

If such condition is fulfilled and if the Seller is requested to perform the work, the Seller and the Buyer will agree on a schedule and place for the work to be performed.

# 12.1.6.5 Warranty Claim Substantiation

Each Warranty Claim filed by the Warranty Beneficiary under this Clause 12.1 will contain at least the following data [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (a) description of defect and action taken, if any,
- (b) date of incident and/or removal date,
- (c) description of Warranted Part claimed to be defective,
- (d) part number,
- (e) serial number (if applicable),
- (f) position on Aircraft,
- (g) total flying hours or calendar time, as applicable, at the date of defect appearance,
- (h) time since last shop visit at the date of defect appearance,
- (i) Manufacturer Serial Number of the affected Aircraft and/or its registration,
- (j) Aircraft total flying hours and/or number of landings at the date of defect appearance,
- (k) Warranty Claim number,
- (l) date of Warranty Claim,
- (m) Delivery Date of Aircraft or Warranted Part to the Buyer,

Warranty Claims are to be addressed as follows:

Airbus

Customer Services directorate Warranty Administration Rond Point Maurice Bellonte B.P. 33 31707 Blagnac Cedex

France

# 12.1.6.6 Replacements

Replaced components, equipment, accessories or parts will become the Seller's property.

Title to and risk of loss of any Aircraft, component, accessory, equipment or part returned by the Buyer to the Seller will at all times remain with the Buyer, except that:

(i) when the Seller has possession of a returned Aircraft, component, accessory, equipment or part to which the Buyer has title, the Seller will have such responsibility therefor as is chargeable by law to a bailee for hire, but the Seller will not be liable for loss of use, and;

(ii) title to and risk of loss of a returned component, accessory, equipment or part will pass to the Seller upon shipment by the Seller to the Buyer of any item furnished by the Seller to the Buyer as a replacement therefor.

Upon the Seller's shipment to the Buyer of any replacement component, accessory, equipment or part provided by the Seller pursuant to this Clause 12.1, title to and risk of loss of such replacement component, accessory, equipment or part will pass to the Buyer.

## 12.1.6.7 Claims Review

The Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] each Warranty Claim and provide notice to the Warranty Beneficiary of approval or rejection of a Warranty Claim [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] The Seller will provide reasonable written substantiation in case of rejection of a Warranty Claim. In the event of rejection, the Buyer will have the opportunity to resubmit the Warranty Claim if additional information not provided in the Warranty Claim is or becomes available. In the event the Seller approves any Warranty Claim and determines to offer a reimbursement, the Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] issue a warranty credit to the Buyer.

### 12.1.6.8 Inspection

The Seller will have the right to inspect the affected Aircraft, documents and other records relating thereto in the event of any Warranty Claim under this Clause 12.1.

- 12.1.7 In-house Warranty
- 12.1.7.1 Seller's Authorization

The Seller hereby authorizes the Buyer to repair Warranted Parts ("In-house Warranty") subject to the terms of this Clause 12.1.7.

#### 12.1.7.2 Conditions for Seller's Authorization

The Buyer will be entitled to repair such Warranted Parts:

- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] The Buyer's notification will include sufficient detail regarding the defect, estimated labor hours and material to allow the Seller to ascertain the reasonableness of the estimate. The Seller agrees to use all reasonable efforts to ensure a prompt response and will not unreasonably withhold, condition or delay authorization;
- provided adequate facilities and qualified personnel are available to the Buyer;
- provided repairs are performed in accordance with the Seller's Technical Data or written instructions; and

- only to the extent specified by the Seller, or, in the absence of such specification, to the extent reasonably necessary to correct the defect, in accordance with the standards set forth in Clause 12.1.10.

## 12.1.7.3 Seller's Rights

The Seller will have the right to require the return of any Warranted Part, or any part removed therefrom, which is claimed to be defective if, in the judgment of the Seller, the nature of the claimed defect requires technical investigation. Such return will be subject to the provisions of Clause 12.1.6.2. Furthermore, the Seller will have the right to have a Seller Representative present during the disassembly, inspection and testing of any Warranted Part claimed to be defective, subject to such presence being practical and not unduly delaying the repair.

### 12.1.7.4 In-house Warranty Claim Substantiation

Claims for In-house Warranty credit will be filed within the time period set forth in 12.1.5 (ii) and will contain the same information as that required for Warranty Claims under Clause 12.1.6.5 and in addition will include:

- (a) a report of technical findings with respect to the defect,
- (b) for parts required to remedy the defect:
  - part numbers,
  - serial numbers (if applicable),
  - parts description,
  - quantity of parts,
  - unit price of parts,
  - related Seller's or third party's invoices (if applicable),
  - total price of parts,
- (c) detailed number of labor hours,
- (d) In-house Warranty Labor Rate,
- (e) total claim value.

#### 12.1.7.5 Credit

The Buyer's sole remedy and the Seller's sole obligation and liability with respect to In-house Warranty Claims will be the warranty credit issued to the Buyer in an amount equal to the reasonable direct labor costs expended in performing the repair of a Warranted Part and to the direct costs of materials incorporated in such repair, determined as set forth below:

(i) to determine direct labor costs, only man-hours spent on [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] disassembly, inspection, repair, reassembly, final inspection and test of the Warranted Part [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be counted. Any man-hours required for maintenance work concurrently being carried out on the affected Aircraft or the Warranted Part will not be included. (ii) The man-hours counted as set forth above will be multiplied by an agreed labor rate of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] ("In-house Warranty Labor Rate"), which is deemed to represent the Buyer's composite labor rate meaning the average hourly rate (excluding all fringe benefits, premium time allowances, social security charges, business taxes and the like) paid to the Buyer's employees whose jobs are directly related to the performance of the repair.

The In-house Warranty Labor Rate is subject to annual adjustment by multiplication by the ratio ECIn/ECIb. For the purposes of this Clause 12.1.7.5 only, ECIn will be equal to the Labor Index, ECI336411W, defined in the Seller's Price Revision Formula 2011 set forth in Exhibit C-1 to the Agreement.

(iii) Direct material costs are determined by the prices at which the Buyer acquired such material, excluding any parts and materials used for overhaul and as may be furnished by the Seller at no charge.

### 12.1.7.6 Limitation

The Buyer will in no event be credited for repair costs (including labor and material) for any Warranted Part in excess of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the Seller's current catalogue price for a replacement of such defective Warranted Part.

## 12.1.7.7 Scrapped Material

The Buyer will retain any defective Warranted Part beyond economic repair and any defective part removed from a Warranted Part during repair for a period of either [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after the date of completion of the repair or [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after submission of a claim for In-house Warranty credit relating thereto, whichever is longer, and during such period such parts will be returned to the Seller within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of receipt of the Seller's request to that effect.

Notwithstanding the foregoing, the Buyer may scrap any such defective parts, which are beyond economic repair and not required for technical evaluation locally, with the agreement of the Seller Representative(s) at any time.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

12.1.8 Standard Warranty in case of Pooling or Leasing Arrangements

Without prejudice to Clause 21.1, the warranties provided for in this Clause 12.1 for any Warranted Part will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 12.1.9 Warranty for Corrected, Replaced or Repaired Warranted Parts

Notwithstanding the definition of "Warranty Period" set forth in Clause 12.1.2.3, whenever any Warranted Part, which contains a defect for which the Seller is liable under Clause 12.1, has been corrected, replaced or repaired pursuant to the terms of this Clause 12.1 (a "Corrected Part"), the period of the Seller's warranty with respect to such Corrected Part will be the remaining portion of the original warranty [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

If a defect is attributable to a defective repair or replacement by the Buyer, a Warranty Claim with respect to such defect will be rejected, notwithstanding any subsequent correction or repair, and will immediately terminate the remaining warranties under this Clause 12.1 in respect of the affected Warranted Part.

## 12.1.10 Accepted Industry Standard Practices Normal Wear and Tear

The Buyer's rights under this Clause 12.1 with respect to any Warranted Part are subject to such Warranted Part being maintained, overhauled, repaired and operated in accordance with accepted industry standard practices, all Technical Data and any other instructions issued by the Seller, the Suppliers and the Propulsion System Manufacturer with respect to such Warranted Part and all applicable rules, regulations and directives of the relevant Aviation Authorities with respect to such Warranted Part.

The Seller's liability under this Clause 12.1 will not extend to normal wear and tear nor to:

- (i) any Warranted Part that has been repaired, altered or modified after Delivery, except by the Seller or in accordance with the Seller's Technical Data or the Seller's written instructions;
- (ii) any Warranted Part that has been operated in a damaged state, except to the extent such operation was until the next commercially reasonable opportunity to replace such Warranted Part; and
- (iii) any Warranted Part from which the part or serial number has been removed.

# 12.2 Seller Service Life Policy

In addition to the warranties set forth in Clause 12.1, the Seller further agrees to provide the Warranty Beneficiary with a "Seller Service Life Policy" as set forth in this Clause 12.2.

12.2.1 Should a Failure occur in any Item (as these terms are defined below) that is not caused by an extrinsic force, then, subject to the general conditions and limitations set forth in Clause 12.2.4, the provisions of this Clause 12.2 will apply.

For the purposes of this Clause 12.2:

(i) "Item" means any item listed in Exhibit F;

(ii) "Failure" means a breakage or defect that can reasonably be expected to occur on a fleet wide basis and which materially impairs the utility of the Item.

### 12.2.2 Periods and Seller's Undertakings

Subject to the general conditions and limitations set forth in Clause 12.2.4, the Seller agrees that if a Failure occurs in an Item within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after the Delivery of such Aircraft, the Seller will, at its discretion and as promptly as practicable and with the Seller's financial participation as hereinafter provided, either:

- design and furnish to the Buyer a Correction or Corrected Part for such Item with a Failure and provide any parts required for such Correction (including Seller designed standard parts but excluding industry standard parts), or
- replace such Item.

# 12.2.3 Seller's Participation in the Costs

Subject to the general conditions and limitations set forth in Clause 12.2.4, any part or Item that the Seller is required to furnish to the Buyer under the Seller Service Life Policy in connection with the correction or replacement of an Item will be furnished to the Buyer at the Seller's then current sales price therefore, less the Seller's financial participation determined in accordance with the following formula:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### where:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 12.2.4 General Conditions and Limitations

12.2.4.1 The undertakings set forth in this Clause 12.2 will be valid after the period of the Seller's warranty applicable to an Item under Clause 12.1.

- 12.2.4.2 The Buyer's remedies and the Seller's obligations and liabilities under the Seller Service Life Policy are subject to the prior compliance by the Buyer with the following conditions:
  - (i) the Buyer will maintain log books and other historical records with respect to each Item, reasonably adequate to enable the Seller to determine whether the alleged Failure is covered by the Seller Service Life Policy and, if so, to define the portion of the costs to be borne by the Seller in accordance with Clause 12.2.3;
  - (ii) the Buyer will keep the Seller informed of any significant incidents relating to an Aircraft, howsoever occurring or recorded;
  - (iii) the Buyer will comply with the conditions of Clause 12.1.10;
  - (iv) the Buyer will implement specific structural inspection programs for monitoring purposes as may be established from time to time by the Seller. Such programs will be as compatible as practicable with the Buyer's operational requirements and will be carried out at the Buyer's expense. Reports relating thereto will be regularly furnished to the Seller;
  - (v) the Buyer will report any breakage or defect in an Item in writing to the Seller within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after such breakage or defect becomes apparent, whether or not such breakage or defect can reasonably be expected to occur in any other aircraft, and the Buyer will have provided to the Seller reasonably sufficient detail on the breakage or defect to enable the Seller to determine whether such breakage or defect is subject to the Seller Service Life Policy.
- 12.2.4.3 Except as otherwise provided for in this Clause 12.2, any claim under the Seller Service Life Policy will be administered as provided for in, and will be subject to the terms and conditions of. Clause 12.1.6.
- 12.2.4.4 In the event of the Seller having issued a modification applicable to an Aircraft, the purpose of which is to avoid a Failure, the Seller may elect to supply the necessary modification kit free of charge to the Buyer or under a pro rata formula established by the Seller. If such a kit is so offered to the Buyer free of charge, then, to the extent of such Failure and any Failures that could ensue therefrom, the validity of the Seller's commitment under this Clause 12.2 will be subject to the Buyer incorporating such modification in the relevant Aircraft, as promulgated by the Seller and in accordance with the Seller's instructions, within a reasonable period of time.
- 12.2.4.5 THE SELLER SERVICE LIFE POLICY IS NEITHER A WARRANTY, PERFORMANCE GUARANTEE, NOR AN AGREEMENT TO MODIFY ANY AIRCRAFT OR AIRFRAME COMPONENTS TO CONFORM TO NEW DEVELOPMENTS OCCURRING IN THE STATE OF AIRFRAME DESIGN AND MANUFACTURING ART. THE SELLER'S OBLIGATION UNDER THIS CLAUSE 12.2 IS TO MAKE ONLY THOSE CORRECTIONS TO THE ITEMS OR FURNISH REPLACEMENTS THEREFOR AS PROVIDED FOR IN THIS CLAUSE 12.2. THE BUYER'S SOLE REMEDY AND RELIEF FOR THE NON-PERFORMANCE OF ANY OBLIGATION OR LIABILITY OF THE SELLER ARISING UNDER OR BY VIRTUE OF THE SELLER SERVICE LIFE POLICY WILL BE [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] LIMITED TO THE AMOUNT THE BUYER REASONABLY EXPENDS IN PROCURING A CORRECTION OR REPLACEMENT FOR ANY ITEM THAT IS THE SUBJECT OF A FAILURE COVERED BY THE SELLER SERVICE LIFE POLICY AND TO WHICH SUCH

NON-PERFORMANCE IS RELATED, LESS THE AMOUNT THAT THE BUYER OTHERWISE WOULD HAVE BEEN REQUIRED TO PAY UNDER THIS CLAUSE 12.2 IN RESPECT OF SUCH CORRECTED OR REPLACEMENT ITEM.

12.3 Supplier Warranties and Service Life Policies

Prior to or at Delivery of the first Aircraft, the Seller will provide the Buyer, in accordance with the provisions of Clause 17, with the warranties and, where applicable, service life policies that the Seller has obtained for Supplier Parts pursuant to the Supplier Product Support Agreements. The Seller authorizes the Buyer to enforce directly against all Suppliers all such warranties and service life policies to the fullest extent permitted by the applicable Supplier Product Support Agreement.

- 12.3.1 Definitions
- 12.3.1.1 "**Supplier**" means any supplier of Supplier Parts.
- 12.3.1.2 **"Supplier Part"** means any component, equipment, accessory or part installed in an Aircraft at the time of Delivery thereof and for which there exists a Supplier Product Support Agreement. For the sake of clarity, the Propulsion System and Buyer Furnished Equipment and other equipment selected by the Buyer to be supplied by suppliers with whom the Seller has no existing enforceable warranty agreements are not Supplier Parts.
- 12.3.1.3 **"Supplier Product Support Agreements"** means agreements between the Seller and Suppliers, as described in Clause 17.1.2, containing enforceable and transferable warranties and, in the case of landing gear suppliers, service life policies for selected structural landing gear elements.
- 12.3.2 Supplier's Default
- 12.3.2.1 In the event that any Supplier, under any standard warranty obtained by the Seller in a Supplier Product Support Agreement, defaults in the performance of any obligation with respect thereto and subject to (i) the Buyer using its commercially reasonable efforts to enforce its rights under such Supplier Product Support Agreement and (ii) the Buyer submitting in reasonable time to the Seller reasonable evidence that such default has occurred, then Clause 12.1 will apply to the extent the same would have been applicable had such Supplier Part been a Warranted Part, except that the Supplier's warranty period as indicated in the Supplier Product Support Agreement will apply.
- 12.3.2.2 In the event that any Supplier, under any Supplier service life policy obtained by the Seller in a Supplier Product Support Agreement, defaults in the performance of any obligation with respect thereto and subject to (i) the Buyer using its commercially reasonable efforts to enforce its rights under such Supplier Product Support Agreement and (ii) the Buyer submitting in reasonable time to the Seller reasonable evidence that such default has occurred, then Clause 12.2 will apply as if such Supplier Part was an Item, except that the Supplier's service life policy period as indicated in the Supplier Product Support Agreement will apply.

12.3.2.3 At the Seller's request, the Buyer will assign to the Seller, and the Seller will be subrogated to, all of the Buyer's rights against the relevant Supplier with respect to and arising by reason of such default and will provide reasonable assistance to enable the Seller to enforce the rights so assigned.

#### 12.4 Interface Commitment

# 12.4.1 Interface Problem

If the Warranty Beneficiary experiences any technical problem in the operation of an Aircraft or its systems due to a malfunction, the cause of which, after due and reasonable investigation, is not readily identifiable by the Warranty Beneficiary but which the Warranty Beneficiary reasonably believes to be attributable to the design characteristics of one or more components of such Aircraft ("Interface Problem"), the Seller will, if so requested by the Warranty Beneficiary, and without additional charge to the Warranty Beneficiary, promptly conduct or have conducted an investigation and analysis of such problem to determine, if possible, the cause or causes of the problem and to recommend such corrective action as may be feasible. The Warranty Beneficiary will furnish to the Seller all data and information in the Warranty Beneficiary's possession reasonably relevant to the Interface Problem and will reasonably cooperate with the Seller in the conduct of the Seller's investigations and such tests as may be required.

At the conclusion of such investigation, the Seller will promptly advise the Warranty Beneficiary in writing of the Seller's opinion as to the cause or causes of the Interface Problem and the Seller's recommendations as to corrective action.

# 12.4.2 Seller's Responsibility

If the Seller determines that the Interface Problem is primarily attributable to the design of a Warranted Part, the Seller will, if so requested by the Warranty Beneficiary and pursuant to the terms and conditions of Clause 12.1, correct the design of such Warranted Part to the extent of the Seller's obligation as defined in Clause 12.1.

## 12.4.3 Supplier's Responsibility

If the Seller determines that the Interface Problem is primarily attributable to the design of any Supplier Part, the Seller will, if so requested by the Warranty Beneficiary, reasonably assist the Warranty Beneficiary in processing any warranty claim the Warranty Beneficiary may have against the Supplier. The Seller will also take whatever reasonable action is permitted by its contracts with such Supplier in an effort to obtain a correction of the Interface Problem acceptable to the Warranty Beneficiary. If the Supplier fails within a reasonable period of time to take appropriate action on the Seller's recommendation as to the necessary corrective action and the Warranty Beneficiary submits to the Seller, within a reasonable period of time, proof of such failure, then the Seller will take action in accordance with the provisions of Clause 12.3

## 12.4.4 Joint Responsibility

If the Seller determines that the Interface Problem is attributable partially to the design of a Warranted Part and partially to the design of any Supplier Part, the Seller will, if so requested by the Warranty Beneficiary, seek a solution to the Interface Problem through cooperative efforts of the Seller and any Supplier involved.

The Seller will promptly advise the Warranty Beneficiary of such corrective action as may be proposed by the Seller and any such Supplier. Such proposal will be consistent with any then existing obligations of the Seller hereunder and of any such Supplier towards the Warranty Beneficiary. Such corrective action, unless reasonably rejected by the Warranty Beneficiary, will constitute full satisfaction of any claim the Warranty Beneficiary may have against either the Seller or any such Supplier with respect to such Interface Problem.

- 12.4.5 General
- 12.4.5.1 All requests under this Clause 12.4 will be directed to both the Seller and the affected Supplier.
- 12.4.5.2 Except as specifically set forth in this Clause 12.4, this Clause will not be deemed to impose on the Seller any obligations not expressly set forth elsewhere in this Agreement.
- 12.4.5.3 All reports, recommendations, data and other documents furnished by the Seller to the Warranty Beneficiary pursuant to this Clause 12.4 will be deemed to be delivered under this Agreement and will be subject to the terms, covenants and conditions set forth in this Clause 12 and in Clause 22.11
- 12.5 Exclusivity of Warranties

THIS CLAUSE 12 SETS FORTH THE EXCLUSIVE WARRANTIES, EXCLUSIVE LIABILITIES AND EXCLUSIVE OBLIGATIONS OF THE SELLER, AND THE EXCLUSIVE REMEDIES AVAILABLE TO THE BUYER, WHETHER UNDER THIS AGREEMENT OR OTHERWISE, ARISING FROM ANY DEFECT OR NONCONFORMITY OR PROBLEM OF ANY KIND IN ANY AIRCRAFT, COMPONENT, EQUIPMENT, ACCESSORY, PART, SOFTWARE, DATA OR SERVICE DELIVERED BY THE SELLER UNDER THIS AGREEMENT.

THE BUYER RECOGNIZES THAT THE RIGHTS, WARRANTIES AND REMEDIES IN THIS CLAUSE 12 ARE ADEQUATE AND SUFFICIENT TO PROTECT THE BUYER FROM ANY DEFECT OR NONCONFORMITY OR PROBLEM OF ANY KIND IN THE GOODS AND SERVICES SUPPLIED UNDER THIS AGREEMENT. THE BUYER HEREBY WAIVES, RELEASES AND RENOUNCES ALL OTHER WARRANTIES, OBLIGATIONS, GUARANTEES AND LIABILITIES OF THE SELLER AND ALL OTHER RIGHTS, CLAIMS AND REMEDIES OF THE BUYER AGAINST THE SELLER, WHETHER EXPRESS OR IMPLIED BY CONTRACT, TORT, OR STATUTORY LAW OR OTHERWISE, WITH RESPECT TO ANY NONCONFORMITY OR DEFECT OR PROBLEM OF ANY KIND IN ANY AIRCRAFT, COMPONENT, EQUIPMENT, ACCESSORY, PART, SOFTWARE, DATA OR SERVICE DELIVERED BY THE SELLER UNDER THIS AGREEMENT, INCLUDING BUT NOT LIMITED TO:

- (1) ANY IMPLIED WARRANTY OF MERCHANTABILITY AND/OR FITNESS FOR ANY GENERAL OR PARTICULAR PURPOSE;
- (2) ANY IMPLIED OR EXPRESS WARRANTY ARISING FROM COURSE OF PERFORMANCE, COURSE OF DEALING OR USAGE OF TRADE:
- (3) ANY RIGHT, CLAIM OR REMEDY FOR BREACH OF CONTRACT;

- (4) ANY RIGHT, CLAIM OR REMEDY FOR TORT, UNDER ANY THEORY OF LIABILITY, HOWEVER ALLEGED, INCLUDING, BUT NOT LIMITED TO, ACTIONS AND/OR CLAIMS FOR NEGLIGENCE, GROSS NEGLIGENCE, INTENTIONAL ACTS, WILLFUL DISREGARD, IMPLIED WARRANTY, PRODUCT LIABILITY, STRICT LIABILITY OR FAILURE TO WARN;
- (5) ANY RIGHT, CLAIM OR REMEDY ARISING UNDER THE UNIFORM COMMERCIAL CODE OR ANY OTHER STATE OR FEDERAL STATUTE;
- (6) ANY RIGHT, CLAIM OR REMEDY ARISING UNDER ANY REGULATIONS OR STANDARDS IMPOSED BY ANY INTERNATIONAL, NATIONAL, STATE OR LOCAL STATUTE OR AGENCY;
- (7) ANY RIGHT, CLAIM OR REMEDY TO RECOVER OR BE COMPENSATED FOR:
  - (a) LOSS OF USE OR REPLACEMENT OF ANY AIRCRAFT, COMPONENT, EQUIPMENT, ACCESSORY OR PART PROVIDED UNDER THIS AGREEMENT;
  - (b) LOSS OF, OR DAMAGE OF ANY KIND TO, ANY AIRCRAFT, COMPONENT, EQUIPMENT, ACCESSORY OR PART PROVIDED UNDER THIS AGREEMENT;
  - (c) LOSS OF PROFITS AND/OR REVENUES;
  - (d) ANY OTHER INCIDENTAL OR CONSEQUENTIAL DAMAGE.

THE WARRANTIES AND SELLER SERVICE LIFE POLICY PROVIDED BY THIS AGREEMENT WILL NOT BE EXTENDED, ALTERED OR VARIED EXCEPT BY A WRITTEN INSTRUMENT EXECUTED AND DELIVERED BY THE SELLER AND THE BUYER. IN THE EVENT THAT ANY PROVISION OF THIS CLAUSE 12 SHOULD FOR ANY REASON BE HELD UNLAWFUL, OR OTHERWISE UNENFORCEABLE, THE REMAINDER OF THIS CLAUSE 12 WILL REMAIN IN FULL FORCE AND EFFECT.

FOR THE PURPOSES OF THIS CLAUSE 12.5, THE "SELLER" WILL BE UNDERSTOOD TO INCLUDE THE SELLER AND ANY OF ITS SUPPLIERS, SUBCONTRACTORS, AND AFFILIATES.

## 12.6 Duplicate Remedies

The remedies provided to the Buyer under Clause 12.1 and Clause 12.2 as to any defect in respect of any Aircraft or any part thereof are mutually exclusive and not cumulative. The Buyer will be entitled to the remedy that provides the maximum benefit to it, as the Buyer may elect, pursuant to the terms and conditions of this Clause 12 for any particular defect for which remedies are provided under this Clause 12; provided, however, that the Buyer will not be entitled to elect a remedy under both Clause 12.1 and Clause 12.2 for the same defect. The Buyer's rights and remedies herein for the nonperformance of any obligations or liabilities of the Seller arising under these warranties will be in monetary damages limited to the amount the Buyer expends in procuring a correction or replacement for any covered part subject to a defect or nonperformance covered by this Clause 12, and the Buyer will not have any right to require specific performance by the Seller.

## 12.7 Negotiated Agreement

The Buyer specifically recognizes that:

- (i) the Specifications have been agreed upon after careful consideration by the Buyer using its judgment as a professional operator of aircraft used in public transportation and as such is a professional within the same industry as the Seller;
- (ii) this Agreement, and in particular this Clause 12, has been the subject of discussion and negotiation and is fully understood by the Buyer; and
- (iii) the price of the Aircraft and the other mutual agreements of the Buyer set forth in this Agreement were arrived at in consideration of, inter alia, the provisions of this Clause 12, specifically including the waiver, release and renunciation by the Buyer set forth in Clause 12.5.

# 12.8 Disclosure to Third Party Entity

In the event of the Buyer intending to designate a third party entity (a "**Third Party Entity**") to administer this Clause 12, the Buyer will notify the Seller of such intention prior to any disclosure of this Clause to the selected Third Party Entity and will cause such Third Party Entity to enter into a confidentiality agreement and or any other relevant documentation with the Seller solely for the purpose of administrating this Clause 12.

# 12.9 Transferability

Unless otherwise permitted pursuant to Clause 21, the Buyer's rights under this Clause 12 may not be assigned, sold, transferred, novated or otherwise alienated by operation of law or otherwise, without the Seller's prior written consent, which will not be unreasonably withheld.

Any transfer in violation of this Clause 12.9 will, as to the particular Aircraft involved, void the rights and warranties of the Buyer under this Clause 12 and any and all other warranties that might arise under or be implied in law.

12.10 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 13 TRADE SECRET, PATENT AND COPYRIGHT INDEMNITY

- 13.1 Indemnity
- 13.1.1 Subject to the provisions of Clause 13.2.3, the Seller will indemnify, defend and hold harmless the Warranty Beneficiary from and against any damages, losses, costs, claims, suits and/or expenses including legal costs (excluding damages, losses, costs, claims, suits, expenses, loss of profits and other liabilities in respect of or resulting from loss of use of the Aircraft) resulting from (a) any trade secret misappropriation or alleged trade secret misappropriation by the Seller which interferes with the Warranty Beneficiary's use of any Airframe, any part or software installed in any Airframe at Delivery (including the Propulsion System interface and excluding Buyer Furnished Equipment, such Airframe, an "Indemnified Airframe"), any replacement part or software or Technical Data, in each case provided by the Seller for any Indemnified Airframe, any part or software installed in any Indemnified Airframe at Delivery, any replacement part or software or Technical Data, in each case provided by the Seller for any Indemnified Airframe in accordance with this Agreement of:
  - (i) any British, French, German, Spanish or U.S. patent; and
  - (ii) any patent issued under the laws of any other country in which the Warranty Beneficiary may lawfully operate such Aircraft, provided that at the time of infringement or claim of infringement:
    - (1) such country and the flag country of such Aircraft are each a party to the Chicago Convention on International Civil Aviation of December 7, 1944, and are each fully entitled to all benefits of Article 27 thereof, or in the alternative,
    - (2) such country and the flag country of such Aircraft are each a party to the International Convention for the Protection of Industrial Property of March 20, 1883 ("Paris Convention"); and
  - (iii) any copyright, provided that the Seller's obligation to indemnify, defend and hold harmless will be limited to infringements in the United States of America and in countries which, at the time of infringement or claim of infringement, are members of The Berne Union.
- 13.1.2 Clause 13.1.1 will not apply to
  - i) Buyer Furnished Equipment or Propulsion System; or
  - (ii) parts not manufactured to the Seller's detailed design.
- 13.1.3 In the event that the Warranty Beneficiary, due to circumstances contemplated in Clause 13.1.1, is prevented from using any Aircraft (whether by a valid judgment of a court of competent jurisdiction or by a settlement arrived at between claimant, the Seller and Warranty Beneficiary), the Seller will at its discretion and expense either:
  - (i) procure for the Warranty Beneficiary the right to use such Aircraft; or

- (ii) as applicable, cure the misappropriation or replace the infringing part of such Aircraft as soon as possible with a non-infringing substitute, in each case complying in all other respects with the requirements of this Agreement.
- 13.2 Administration of Patent and Copyright Indemnity Claims
- 13.2.1 If the Warranty Beneficiary receives a written claim or a suit is threatened or commenced against the Warranty Beneficiary for misappropriation or infringement referred to in Clause 13.1, the Warranty Beneficiary will:
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] notify the Seller giving particulars thereof, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) furnish to the Seller all data, papers and records within the Warranty Beneficiary's control or possession relating to such suit or claim;
  - (iii) refrain from admitting any liability or making any payment or assuming any expenses, damages, costs or royalties or otherwise acting in a manner prejudicial to the defense or denial of such suit or claim; provided that nothing in this sub-Clause (iii) will prevent the Warranty Beneficiary from paying such sums as may be required in order to obtain the release of the affected Aircraft; provided further that such payment is accompanied by a denial of liability and is made without prejudice;
  - (iv) reasonably co-operate with, and render reasonable assistance to, the Seller as may be pertinent to the defense or denial of the suit or claim;
  - (v) act in such a way as to reasonably mitigate damages, costs and expenses and / or reduce the amount of royalties which may be payable.
- 13.2.2 The Seller will be entitled upon prior written notice to the Warranty Beneficiary, either in its own name or on behalf of the Warranty Beneficiary, to conduct negotiations with the party or parties alleging misappropriation or infringement and will assume and conduct the defense or settlement of any suit or claim. Absent the prior written consent of the Warranty Beneficiary, the Seller may not enter into any settlement, unless such settlement (i) does not include any finding or admission of wrongdoing by the Warranty Beneficiary, (ii) includes an unconditional written release by the claimant or plaintiff of the Warranty Beneficiary from all liability in respect of such suit or claim, and (iii) does not impose equitable remedies or any obligation on the Warranty Beneficiary other than solely the payment of money damages for which the Warranty Beneficiary will be fully indemnified hereunder.

If the Seller fails or refuses to assume the defense of any claim or suit notified to it under this Clause 13, the Warranty Beneficiary will, upon prior written notice to the Seller, have the right, at the Seller's expense, to proceed with the defense or settlement of the claim or suit as it deems prudent; provided, that absent the prior written consent of the Seller, the Warranty Beneficiary may not enter into any settlement, unless such settlement (i) does not include any finding or admission of wrongdoing by the Seller, (ii) includes an unconditional written release by the claimant or plaintiff of the Seller from all liability in respect of such suit or claim and (iii) does not impose equitable remedies on the Seller.

13.2.3 The Seller's liability hereunder will be conditional upon the compliance by the Warranty Beneficiary with the terms of this Clause and is in lieu of any other liability to the Warranty Beneficiary express or implied which the Seller might incur at law as a result of any actual or claimed misappropriation or infringement.

THE INDEMNITY PROVIDED IN THIS CLAUSE 13 AND THE OBLIGATIONS AND LIABILITIES OF THE SELLER UNDER THIS CLAUSE 13 ARE EXCLUSIVE AND IN SUBSTITUTION FOR, AND THE WARRANTY BENEFICIARY HEREBY WAIVES, RELEASES AND RENOUNCES ALL OTHER INDEMNITIES, WARRANTIES, OBLIGATIONS, GUARANTEES AND LIABILITIES ON THE PART OF THE SELLER AND RIGHTS, CLAIMS AND REMEDIES OF THE WARRANTY BENEFICIARY AGAINST THE SELLER, EXPRESS OR IMPLIED, ARISING BY LAW OR OTHERWISE (INCLUDING WITHOUT LIMITATION ANY OBLIGATION, LIABILITY, RIGHT, CLAIM OR REMEDY ARISING FROM OR WITH RESPECT TO LOSS OF USE OR REVENUE OR CONSEQUENTIAL DAMAGES), WITH RESPECT TO ANY ACTUAL OR ALLEGED TRADE SECRET MISAPPROPRIATION OR PATENT OR COPYRIGHT INFRINGEMENT BY ANY INDEMNIFIED AIRFRAME, PART OR SOFTWARE INSTALLED IN ANY INDEMNIFIED AIRFRAME AT DELIVERY, REPLACEMENT PART OR SOFTWARE OR ANY TECHNICAL DATA IN EACH CASE PROVIDED BY THE SELLER FOR ANY AIRFRAME, PROVIDED THAT, IN THE EVENT THAT ANY OF THE AFORESAID PROVISIONS SHOULD FOR ANY REASON BE HELD UNLAWFUL OR OTHERWISE INEFFECTIVE, THE REMAINDER OF THIS CLAUSE WILL REMAIN IN FULL FORCE AND EFFECT. THIS INDEMNITY AGAINST TRADE SECRET MISAPPROPRIATIONS AND PATENT AND COPYRIGHT INFRINGEMENTS WILL NOT BE EXTENDED, ALTERED OR VARIED EXCEPT BY A WRITTEN INSTRUMENT EXECUTED AND DELIVERED BY THE SELLER AND THE WARRANTY BENEFICIARY.

### 14 TECHNICAL DATA AND SOFTWARE SERVICES

14.1 Scope

This Clause 14 covers the terms and conditions for the supply of technical data (together with any revisions thereto, and including the technical data set forth in Exhibit G hereto, the "**Technical Data**") and software services described hereunder (hereinafter, and including the software services set forth in Exhibit G hereto, "**Software Services**") to support the Aircraft operation, maintenance and training.

- 14.1.1 Except as otherwise set forth in this Clause 14, the Technical Data will be supplied in the English language using the aeronautical terminology in common use. The Technical Data will reference U.S. conventions for pounds, feet, seconds and gallons.
- 14.1.2 Range, form, type, format, quantity and delivery schedule of the Technical Data to be provided under this Agreement are outlined in Exhibit G hereto.
- 14.2 Aircraft Identification for Technical Data
- 14.2.1 For those Technical Data that are customized to the Buyer's Aircraft, the Buyer agrees to the allocation of fleet serial numbers ("**Fleet Serial Numbers**") in the form of a block of numbers selected in the range from 001 to 999.
- 14.2.2 The sequence will not be interrupted unless two (2) different Propulsion System or two (2) different models of Aircraft are selected.
- 14.2.3 The Buyer will indicate to the Seller the Fleet Serial Number allocated to each Aircraft corresponding to the delivery schedule set forth in Schedule I no later than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] months before the Scheduled Delivery Month of the first Aircraft. Neither the designation of such Fleet Serial Numbers nor the subsequent allocation of the Fleet Serial Numbers to Manufacturer Serial Numbers for the purpose of producing certain customized Technical Data will constitute any property, insurable or other interest of the Buyer in any Aircraft prior to the Delivery of such Aircraft as provided for in this Agreement.

The customized Technical Data that are affected thereby are the following:

- Aircraft Maintenance Manual,
- Illustrated Parts Catalogue,
- Trouble Shooting Manual,
- Aircraft Wiring Manual,
- Aircraft Schematics Manual, and
- Aircraft Wiring Lists.
- 14.3 Integration of Equipment Data
- 14.3.1 Supplier Equipment

Information, including revisions, relating to Supplier equipment that is installed on the Aircraft at Delivery, or through Airbus Service Bulletins thereafter, will be introduced [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 14.3.2 Buyer Furnished Equipment
- 14.3.2.1 The Seller will introduce Buyer Furnished Equipment data for Buyer Furnished Equipment that is installed on the Aircraft by the Seller (hereinafter "BFE Data") into the customized Technical Data, at no additional charge to the Buyer or the Lessor for the initial issue of the Technical Data provided at or before Delivery of the first Aircraft of each type, configuration or version, provided such BFE Data is provided in accordance with the conditions set forth in Clauses 14.3.2.2 through 14.3.2.6.
- 14.3.2.2 The Buyer will supply the BFE Data to the Seller at least [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] prior to the Scheduled Delivery Month of the first Aircraft of each type. If the Buyer does not supply such BFE Data to the Seller by such time, then the Seller will, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] incorporate such BFE Data at the first scheduled revision following [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 14.3.2.3 The Buyer will supply the BFE Data to the Seller in English and in compliance with the then applicable revision of ATA Specification 2200 (iSpec 2200), Information Standards for Aviation Maintenance.
- 14.3.2.4 The Buyer and the Seller will reasonably agree on the requirements for the provision to the Seller of BFE Data for "on-aircraft maintenance", such as but not limited to timeframe, media and format in which the BFE Data will be supplied to the Seller, in order to manage the BFE Data integration process in an efficient, expeditious and economic manner.
- 14.3.2.5 The BFE Data will be delivered in digital format (SGML) and/or in Portable Document Format (PDF), as reasonably agreed between the Buyer and the Seller.
- 14.3.2.6 All costs related to the delivery to the Seller of the applicable BFE Data will be borne by the Buyer.
- 14.4 Supply
- 14.4.1 Technical Data will be supplied on-line and/or off-line, as set forth in Exhibit G hereto.
- 14.4.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 14.5 Delivery
- 14.5.1 For Technical Data provided off-line, such Technical Data and corresponding revisions will be sent to up to two (2) addresses as indicated by the Buyer at no charge to the Buyer in accordance with the terms set forth under Clause 14.5.2.

- 14.5.2 Technical Data provided off-line will be delivered by the Seller at the Buyer's named place of destination under DAP conditions. The term Delivered At Place (DAP) is defined in the Incoterms 2010 publication issued by the International Chamber of Commerce, (the "**DAP**").
- 14.5.3 The Technical Data will be delivered according to a mutually agreed schedule to correspond with the Delivery of each Aircraft in accordance with Exhibit G. The Buyer will provide no less than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] prior notice when requesting a change to such delivery schedule.
- 14.5.4 It will be the responsibility of the Buyer to coordinate and satisfy local Aviation Authorities' requirements with respect to Technical Data. Reasonable quantities of such Technical Data will be supplied by the Seller at no charge to the Buyer at the Buyer's named place of destination.
- 14.6 Revision Service

For each firmly ordered Aircraft covered under this Agreement, revision service for the Technical Data will be provided free of charge for a period starting at Delivery of the first Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (the "Revision Service Period").

Thereafter revision service will be provided in accordance with the terms and conditions set forth in the Seller's then current Customer Services Catalog.

- 14.7 Service Bulletins (SB) Incorporation
  - During the Revision Service Period and upon the Warranty Beneficiary's request, Seller Service Bulletin information will be incorporated into the Technical Data, provided that the Buyer notifies the Seller through the relevant Airbus World on-line Service Bulletin Reporting application that it intends to accomplish such Service Bulletin. The split effectivity for the corresponding Service Bulletin will remain in the Technical Data until notification from the Buyer that embodiment has been completed on all of the Buyer's Aircraft. The foregoing is applicable for Technical Data relating to maintenance only. For operational Technical Data either the pre or post Service Bulletin status will be shown.
- 14.8 Technical Data Familiarization
  - Upon request by the Buyer, the Seller will provide up to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of Technical Data familiarization training at the Seller's or the Buyer's facilities, at the Buyer's option. The basic familiarization course is tailored for maintenance and engineering personnel.
- 14.9 Customer Originated Changes (COC)
  - If the Buyer wishes to introduce Buyer originated data (hereinafter "COC Data") into any of the customized Technical Data that are identified as eligible for such incorporation in the Seller's then current Customer Services Catalog, the Buyer will notify the Seller of such intention.

	The incorporation of any COC Data will be performed under the methods and tools for achieving such introduction and the conditions specified in the Seller's then current Customer Services Catalog.
14.10	AirN@v Family products
14.10.1	The Technical Data listed below is provided on DVD and include integrated software (hereinafter together referred to as "AirN@v Family").

- 14.10.2 The AirN@v Family covers several Technical Data domains, reflected by the following AirN@v Family products:
  - AirN@v / Maintenance,
  - AirN@v / Planning,
  - AirN@v / Repair,
  - AirN@v / Workshop,
  - AirN@v / Associated Data,
  - AirN@v / Engineering.
- 14.10.3 Further details on the Technical Data included in such products are set forth in Exhibit G.
- 14.10.4 The licensing conditions for the use of AirN@v Family integrated software will be set forth in a separate agreement (the "End-User License Agreement for Airbus Software") to be executed by the parties prior to Delivery of the first Aircraft.
- 14.10.5 The revision service and the license to use AirN@v Family products will be granted free of charge for [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] At the end of such Revision Service Period, the yearly revision service for AirN@v Family products and the associated license fee will be provided to the Buyer under the commercial conditions set forth in the Seller's then current Customer Services Catalog.
- 14.11 On-Line Technical Data
- 14.11.1 The Technical Data defined in Exhibit G as being provided on-line will be made available to the Buyer through AirbusWorld, as set forth in a separate agreement to be executed by the parties the prior to Delivery of the first Aircraft.
- 14.11.2 Such provisions will be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 14.11.3 Access to AirbusWorld will be subject to the General Terms and Conditions of Access to and Use of AirbusWorld (hereinafter the "GTC"), as set forth in a separate agreement to be executed by the parties prior to Delivery of the first Aircraft.
- 14.11.4 The list of the Technical Data provided on-line may be extended from time to time.

For any Technical Data which is or becomes available on-line, the Seller reserves the right to eliminate other formats for the concerned Technical Data.

- 14.11.5 Access to AirbusWorld will be granted free of charge for an unlimited number of the Buyer's users [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of Buyer's administrators) for the Technical Data related to the Aircraft which will be operated by the Buyer.
- 14.11.6 For the sake of clarification, Technical Data accessed through AirbusWorld which access will be covered by the terms and conditions set forth in the GTC will remain subject to the conditions of this Clause 14.

In addition, should AirbusWorld provide access to Technical Data in software format, the use of such software will be subject to the conditions of the End-User License Agreement for Airbus Software.

14.12 Waiver, Release and Renunciation

The Seller warrants that the Technical Data are prepared in accordance with the state of the art at the date of their development. Should any Technical Data prepared by the Seller contain a non-conformity or defect, the sole and exclusive liability of the Seller will be to take all reasonable and proper steps to correct such Technical Data. Irrespective of any other provisions herein, no warranties of any kind will be given for the COC Data, as set forth in Clause 14.9.

THE WARRANTIES, OBLIGATIONS AND LIABILITIES OF THE SELLER (AS DEFINED BELOW FOR THE PURPOSES OF THIS CLAUSE) AND REMEDIES OF THE WARRANTY BENEFICIARY SET FORTH IN THIS CLAUSE 14 ARE EXCLUSIVE AND IN SUBSTITUTION FOR, AND THE WARRANTY BENEFICIARY HEREBY WAIVES, RELEASES AND RENOUNCES ALL OTHER WARRANTIES, OBLIGATIONS AND LIABILITIES OF THE SELLER AND RIGHTS, CLAIMS AND REMEDIES OF THE WARRANTY BENEFICIARY AGAINST THE SELLER, EXPRESS OR IMPLIED, ARISING BY LAW, CONTRACT OR OTHERWISE, WITH RESPECT TO ANY NON-CONFORMITY OR DEFECT OF ANY KIND, IN ANY TECHNICAL DATA OR TECHNICAL SERVICES DELIVERED UNDER THIS AGREEMENT, INCLUDING BUT NOT LIMITED TO:

- A. ANY WARRANTY AGAINST HIDDEN DEFECTS;
- B. ANY IMPLIED WARRANTY OF MERCHANTABILITY OR FITNESS;
- C. ANY IMPLIED WARRANTY ARISING FROM COURSE OF PERFORMANCE, COURSE OF DEALING OR USAGE OR TRADE;
- D. ANY OBLIGATION, LIABILITY, RIGHT, CLAIM OR REMEDY, WHETHER IN CONTRACT OR IN TORT, WHETHER OR NOT ARISING FROM THE SELLER'S NEGLIGENCE, ACTUAL OR IMPUTED; AND
- E. ANY OBLIGATION, LIABILITY, RIGHT, CLAIM, OR REMEDY FOR LOSS OF OR DAMAGE TO ANY AIRCRAFT, COMPONENT, EQUIPMENT, ACCESSORY, PART, SOFTWARE, DATA OR SERVICES DELIVERED UNDER THIS AGREEMENT, FOR LOSS OF USE, REVENUE OR PROFIT, OR FOR ANY OTHER DIRECT, INCIDENTAL OR CONSEQUENTIAL DAMAGES;

- PROVIDED THAT, IN THE EVENT THAT ANY OF THE AFORESAID PROVISIONS SHOULD FOR ANY REASON BE HELD UNLAWFUL OR OTHERWISE INEFFECTIVE, THE REMAINDER OF THIS AGREEMENT WILL REMAIN IN FULL FORCE AND EFFECT.
  - FOR THE PURPOSES OF THIS CLAUSE 14, THE "SELLER" WILL BE UNDERSTOOD TO INCLUDE THE SELLER, ANY OF ITS SUBCONTRACTORS, ITS AFFILIATES AND ANY OF THEIR RESPECTIVE INSURERS.
- 14.13 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 14.14 Proprietary Rights
- 14.14.1 All proprietary rights relating to Technical Data, including but not limited to patent, design and copyrights, will remain with the Seller and/or its Affiliates, as the case may be.
  - These proprietary rights will also apply to any translation into a language or languages or media that may have been performed or caused to be performed by the Buyer.
- 14.14.2 Whenever this Agreement and/or any Technical Data provides for manufacturing by the Buyer, the consent given by the Seller will not be construed as any express or implicit endorsement or approval whatsoever of the Buyer or of the manufactured products. Subject to Article 1.7.1 of Exhibit H, the supply of the Technical Data will not be construed as any further right for the Buyer to design or manufacture any Aircraft or part thereof, including any spare part.
- 14.15 Performance Engineer's Program
- 14.15.1 In addition to the Technical Data provided under Clause 14, the Seller will provide to the Buyer Software Services, which will consist of the Performance Engineer's Programs ("PEP") for the Aircraft type covered under this Agreement. Such PEP is composed of software components and databases, and its use is subject to the license conditions set forth in the End-User License Agreement for Airbus Software.
- 14.15.2 Use of the PEP will be limited to two (2) copies to be used on the Buyer's computers for the purpose of computing performance engineering data. The PEP is intended for use on ground only and will not be placed or installed on board the Aircraft.
- 14.15.3 The license to use the PEP and the revision service will be provided [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] for the duration of the corresponding Revision Service Period as set forth in Clause 14.6.
- 14.15.4 At the end of such PEP Revision Service Period, the PEP will be provided to the Buyer at the standard commercial conditions set forth in the Seller's then current Customer Services Catalog.
- 14.16 Future Developments

The Seller continuously monitors technological developments and applies them to Technical Data, document and information systems' functionalities, production and methods of transmission.

The Seller will implement and the Buyer will accept such new developments, it being understood that the Buyer will be informed in due time by the Seller of such new developments and their application and of the date by which the same will be implemented by the Seller.

- 14.17 Confidentiality
- 14.17.1 This Clause, the Technical Data, the Software Services and their content are designated as confidential. All such Technical Data and Software Services are provided to the Buyer for the sole use of the Buyer who undertakes not to disclose the contents thereof to any third party without the prior written consent of the Seller, except as permitted therein or pursuant to any government or legal requirement imposed upon the Buyer.
- 14.17.2 If the Seller authorizes the disclosure of this Clause or of any Technical Data or Software Services to third parties either under this Agreement or by an express prior written authorization or, specifically, where the Buyer intends to designate a maintenance and repair organization or a third party to perform the maintenance of the Aircraft or to perform data processing on its behalf (each a "**Third Party**"), the Buyer will notify the Seller of such intention prior to any disclosure of this Clause and/or the Technical Data and/or the Software Services to such Third Party.

The Buyer hereby undertakes to cause such Third Party to agree to be bound by the conditions and restrictions set forth in this Clause 14 with respect to the disclosed Clause, Technical Data or Software Services and will in particular cause such Third Party to enter into a confidentiality agreement with the Seller and appropriate licensing conditions, and to commit to use the Technical Data solely for the purpose of maintaining the Buyer's Aircraft and the Software Services exclusively for processing the Buyer's data. The Seller will reasonably cooperate with the Buyer in the Buyer's efforts to fulfill its undertakings set forth in this paragraph.

### 14.18 Transferability

Unless otherwise permitted pursuant to Clause 21, the Buyer's rights under this Clause 14 may not be assigned, sold, transferred, novated or otherwise alienated by operation of law or otherwise, without the Seller's prior written consent.

Any transfer in violation of this Clause 14.17 will, as to the particular Aircraft involved, void the rights and warranties of the Buyer under this Clause 14 and any and all other warranties that might arise under or be implied in law.

### 15 SELLER REPRESENTATIVE SERVICES

The Seller will provide [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to the Buyer the services described in this Clause 15, at the Buyer's main base or at other locations to be mutually agreed by the parties, acting reasonably.

- 15.1 Customer Support Representative(s)
- 15.1.1 The Seller will provide [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to the Buyer the services of Seller customer support representative(s), as defined in Appendix A to this Clause 15 (each a "Seller Representative"), at the Buyer's main base or such other locations as the parties may agree. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 15.1.2 In providing the services as described herein, any Seller Representatives, or any Seller employee(s) providing services to the Buyer hereunder, are deemed to be acting in an advisory capacity only and at no time will they be deemed to be acting as Buyer's employees, contractors or agents, either directly or indirectly.
- 15.1.3 The Seller will provide to the Buyer an annual written accounting of the consumed man-months and any remaining man-month balance from the allowance defined in Appendix A to this Clause 15. Such accounting will be deemed final and accepted by the Buyer unless the Seller receives written objection from the Buyer within thirty (30) calendar days of receipt of such accounting.
- 15.1.4 In the event of a need for Aircraft On Ground ("AOG") technical assistance after the end of the assignment referred to in Appendix A to this Clause 15, the Buyer will have non-exclusive access to:
  - (i) AIRTAC (Airbus Technical AOG Center); and
  - (ii) the Seller Representative network closest to the Buyer's main base. A list of contacts of the Seller Representatives closest to the Buyer's main base will be provided to the Buyer.
  - As a matter of reciprocity, the Buyer agrees that Seller Representative(s) may provide services to other airlines during any assignment with the Buyer.
- 15.1.5 Should the Buyer request Seller Representative services exceeding the allocation specified in Appendix A to this Clause 15, the Seller may provide such additional services subject to terms and conditions to be mutually agreed.
- 15.1.6 The Seller will cause services similar to the services described in this Clause 15 to be provided by representatives of the Propulsion System Manufacturer and Suppliers, when reasonably requested by the Buyer.
- 15.2 Buyer's Support
- 15.2.1 From the date of arrival of the first Seller Representative and for the duration of the assignment, the Buyer will provide, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY

WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and if requested by the Seller, a suitable, lockable office, conveniently located with respect to the Buyer's maintenance facilities, with complete office furniture and equipment including telephone, internet, email and facsimile connections for the sole use of the Seller Representative(s). [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 15.2.2 Absence of an assigned Seller Representative during normal statutory vacation periods will be covered by other seller representatives on the same conditions as those described in Clause 15.1.4, and such services will be counted against the total allocation provided in Appendix A hereto.
- 15.2.3 The Seller will use commercially reasonable efforts to obtain, and the Buyer will assist the Seller in obtaining, from the civil authorities of the Buyer's country those documents that are necessary to permit the Seller Representative to live and work in the Buyer's country. If the Seller is unable to obtain such necessary documents for any Seller Representative after the exercise of commercially reasonable efforts by the Seller to do so, the Seller will be relieved of any obligation to the Buyer under the provisions of Clause 15.1 with respect to such Seller Representative until the Seller is able to obtain such necessary documents.
- 15.3 Withdrawal of the Seller Representative

The Seller will have the right to withdraw its assigned Seller Representatives as it sees fit if conditions arise, which are in the Seller's opinion dangerous to their safety or health or prevent them from fulfilling their contractual tasks.

15.4 Indemnities

INDEMNIFICATION PROVISIONS, INCLUDING THE LIMITATIONS RELATED THERETO, APPLICABLE TO THIS CLAUSE 15 ARE SET FORTH IN CLAUSE 19.

# **APPENDIX A TO CLAUSE 15**

# **SELLER REPRESENTATIVE**

- The Seller will provide to the Buyer Seller Representative services at the Buyer's main base or at other locations to be mutually agreed for [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2 For the sake of clarification, such Seller Representatives' services will include [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- The number of Seller Representatives assigned to the Buyer at any one time will be mutually agreed, but will at no time exceed [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Seller Representatives.

16	TDAINING	CLIDDODT	AND SERVICES

- 16.1 General
- 16.1.1 This Clause 16 sets forth the terms and conditions for the supply of training support and services for the Buyer's personnel to support the Aircraft operation.
- 16.1.2 The range, quantity and validity of training to be provided free of charge under this Agreement are covered in Appendix A to this Clause 16.
- 16.1.3 Scheduling of training courses covered in Appendix A will be mutually agreed during a training conference (the "Training Conference") that will be held no later than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 16.2 Training Location
- 16.2.1 The Seller will provide training at its affiliated training center in Miami, U.S.A. or such other of its training centers as agreed upon by the Seller and the Buyer (individually a "**Seller's Training Center**" and collectively the "**Seller's Training Centers**").
- 16.2.2 If the unavailability of facilities or scheduling difficulties make training by the Seller at any Seller's Training Center impractical, the Seller will ensure that the Buyer is provided with such training at another location designated by the Seller.
- 16.2.3 Upon the Buyer's request, the Seller may also provide certain training at a location other than the Seller's Training Centers, including one of the Buyer's bases, if and when practicable for the Seller, under terms and conditions to be mutually agreed upon. In such event, all additional charges listed in Clauses 16.5.2 and 16.5.3 will be borne by the Buyer.
- 16.2.4 If the Buyer requests training at a location as indicated in Clause 16.2.3 and requires such training to be an Airbus approved course, the Buyer undertakes that the training facilities will be approved by the Seller prior to the performance of such training. The Buyer will, as necessary and with adequate time prior to the performance of such training, provide access to the training facilities set forth in Clause 16.2.3 to the Seller's and the competent Aviation Authority's representatives for approval of such facilities.
- 16.3 Training Courses
- Training courses will be as described in the Seller's customer services catalog (the "**Seller's Customer Services Catalog**"). The Seller's Customer Services Catalog also sets forth the minimum and maximum number of trainees per course.

All training requests or training course changes made outside of the scope of the Training Conference will be submitted by the Buyer with a minimum of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] months prior notice.

- 16.3.2 The following terms and conditions will apply to training performed by the Seller:
  - (i) Training courses will be the Seller's standard courses as described in the Seller's Customer Services Catalog valid at the time of execution of the course. The Seller will be responsible for all training course syllabi, training aids and training equipment necessary for the organization of the training courses. For the avoidance of doubt, such training equipment does not include provision of aircraft for the purpose of performing training.
  - (ii) The training equipment and the training curricula used for the training of flight, cabin and maintenance personnel will not be fully customized but will be configured in order to obtain the relevant Aviation Authority's approval and to support the Seller's training programs.
  - (iii) Training data and documentation for trainees receiving the training at the Seller's Training Centers will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Training data and documentation will be marked "FOR TRAINING ONLY" and as such are supplied for the sole and express purpose of training; training data and documentation will not be revised.
- 16.3.3 When the Seller's training courses are provided by the Seller's instructors (individually an "**Instructor**" and collectively "**Instructors**") the Seller will deliver to each trainee a Certificate of Recognition or a Certificate of Course Completion (each a "**Certificate**") or an attestation (an "**Attestation**"), as applicable, upon successful completion of any such training course. Any such Certificate or Attestation will not represent authority or qualification by any Aviation Authority but may be presented to such Aviation Authority in order to obtain relevant formal qualification.
  - In the event of training courses being provided by a training provider selected by the Seller as set forth in Clause 16.2.2, the Seller will cause such training provider to deliver to each traine a Certificate or Attestation, which will not represent authority or qualification by any Aviation Authority, but may be presented to such Aviation Authority in order to obtain relevant formal qualification.
- 16.3.4 Should the Buyer wish to exchange any of the training courses provided under Appendix A hereto, the Buyer will place a request for exchange to this effect with the Seller. The Buyer may exchange, subject to the Seller's confirmation, the training allowances granted under Appendix A of the present Agreement as follows:
  - (i) flight operations training courses as listed under Article 1 of Appendix A may be exchanged for any flight operations training courses described in the Seller's Customer Services Catalog current at the time of the Buyer's request;
  - (ii) maintenance training courses as listed under Article 3 of Appendix A may be exchanged for any maintenance training courses described in the Seller's Customer Services Catalog current at the time of the Buyer's request;
  - (iii) should any one of the allowances granted thereunder (flight operations or maintenance) have been fully drawn upon, the Buyer will be entitled to exchange flight operations or maintenance training courses as needed against the remaining allowances.

The exchange value will be based on the Seller's Training Course Exchange Matrix applicable at the time of the request for exchange and which will be provided to the Buyer at such time.

It is understood that the above provisions will apply to the extent that training allowances granted under Appendix A remain available to the full extent necessary to perform the exchange.

All requests to exchange training courses will be submitted by the Buyer with a minimum of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] prior notice. The requested training will be subject to the Seller's then existing planning constraints.

- 16.3.5 Should the Buyer use none or only part of the training to be provided pursuant to this Clause 16, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 16.3.6 Should the Buyer decide to cancel or reschedule a training course, fully or partially, and irrespective of the location of the training, a minimum advance notification of at least [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] prior to the relevant training course start date is required.
- 16.3.7 If the notification occurs less than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] but more than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] calendar days prior to such training, a cancellation fee corresponding to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of such training will be, as applicable, either deducted from the training allowance defined in Appendix A or invoiced at the Seller's then applicable price.
- 16.3.8 If the notification occurs less than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] calendar days prior to such training, a cancellation fee corresponding to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of such training will be, as applicable, either deducted from the training allowance defined in Appendix A or invoiced at the Seller's then applicable price.
- 16.3.9 All courses exchanged under Clause 16.3.4 will remain subject to the provisions of this Clause 16.3.
- 16.4 Prerequisites and Conditions
- 16.4.1 Training will be conducted in English and all training aids used during such training will be written in English using common aeronautical terminology.
- 16.4.2 The Buyer hereby acknowledges that all training courses conducted pursuant to this Clause 16 are standard transition training courses and not a*b initio* training courses.
- 16.4.3 Trainees will have the prerequisite knowledge and experience specified for each course in the Seller's Customer Services Catalog.

- 16.4.4 The Buyer will be responsible for the selection of the trainees and for any liability with respect to the entry knowledge level of the trainees.
- 16.4.5 The Seller reserves the right to verify the trainees' proficiency and previous professional experience.
- 16.4.6 The Seller will provide to the Buyer during the Training Conference an Airbus Pre-Training Survey for completion by the Buyer for each trainee.

The Buyer will provide the Seller with an attendance list of the expected trainees for each course, with the validated qualification of each trainee, at the time of reservation of the training course and in no event any later than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] before the start of the training course. The Buyer will return concurrently thereto the completed Airbus Pre-Training Survey, detailing the trainees' associated background. If the Seller determines through the Airbus Pre-Training Survey that a trainee does not match the prerequisites set forth in the Seller's Customer Services Catalog, following consultation with the Buyer, such trainee will be withdrawn from the program or directed through a relevant entry level training program, which will be at the Buyer's expense, and the Buyer will be entitled to substitute a new trainee for the trainee that has been so withdrawn or directed.

- 16.4.7 If the Seller determines at any time during the training that a trainee lacks the required level, following consultation with the Buyer, such trainee will be withdrawn from the program or, upon the Buyer's request, the Seller may be consulted to direct the above mentioned trainee(s), if possible, to any other required additional training, which will be at the Buyer's expense.
- 16.4.8 The Seller will in no case warrant or otherwise be held liable for any trainee's performance as a result of any training provided.
- 16.5 Logistics
- 16.5.1 Trainees
- 16.5.1.1 Living and travel expenses for the Buyer's trainees will be borne by the Buyer.
- 16.5.1.2 It will be the responsibility of the Buyer to make all necessary arrangements relative to authorizations, permits and/or visas necessary for the Buyer's trainees to attend the training courses to be provided hereunder. Rescheduling or cancellation of courses due to the Buyer's failure to obtain any such authorizations, permits and/or visas will be subject to the provisions of Clauses 16.3.6 thru 16.3.8.
- 16.5.2 Training at External Location Seller's Instructors
- 16.5.2.1 In the event of training being provided at the Seller's request at any location other than the Seller's Training Centers, as provided for in Clause 16.2.2, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 16.5.2.2 In the event of training being provided by the Seller's Instructor(s) at any location other than the Seller's Training Centers at the Buyer's request, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 16.5.2.3 Living Expenses

Except as provided for in Clause 16.5.2.1 above, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 16.5.2.4 Air Travel

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 16.5.2.5 Buyer's Indemnity

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] the Seller will not be held liable to the Buyer for any delay or cancellation in the performance of any training outside of the Seller's Training Centers associated with any transportation described in this Clause 16.5.2, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 16.5.3 Training Material and Equipment Availability - Training at External Location

Training material and equipment necessary for course performance at any location other than the Seller's Training Centers or the facilities of a training provider selected by the Seller will be provided by the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] in accordance with the Seller's specifications.

Notwithstanding the foregoing, should the Buyer request the performance of a course at another location as per Clause 16.2.3, the Seller may, upon the Buyer's request, provide the training material and equipment necessary for such course's performance. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 16.6 Flight Operations Training

The Seller will provide training for the Buyer's flight operations personnel as further detailed in Appendix A to this Clause 16, including the courses described in this Clause 16.6.

### 16.6.1 Flight Crew Training Course

The Seller will perform a flight crew training course program for the Buyer's flight crews, each of which will consist of two (2) crew members, who will be either captain(s) or first officer(s).

- 16.6.2 Base Flight Training
- 16.6.2.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] performed in accordance with the related Airbus training course definition (the "Base Flight Training").
- 16.6.2.2 Should it be necessary to ferry the Buyer's delivered Aircraft to the location where the Base Flight Training will take place, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 16.6.2.3 If the Base Flight Training is performed outside of the zone where the Seller usually performs such training, the ferry flight to the location where the Base Flight Training will take place will be performed by a crew composed of the Seller's and/or the Buyer's qualified pilots, in accordance with the relevant Aviation Authority's regulations related to the place of performance of the Base Flight Training.
- 16.6.3 Flight Crew Line Initial Operating Experience

In order to assist the Buyer with initial operating experience after Delivery of the first Aircraft, the Seller will provide to the Buyer pilot Instructor(s) as set forth in Appendix A to this Clause 16.

Should the Buyer request, subject to the Seller's consent, such Seller pilot Instructor(s) to perform any other flight support during the flight crew line initial operating period, such as but not limited to line assistance, demonstration flight(s), ferry flight(s) or any flight(s) required by the Buyer during the period of entry into service of the Aircraft, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

It is hereby understood by the parties that the Seller's pilot Instructor(s) will only perform the above flight support services to the extent they bear the relevant qualifications to do so.

- 16.6.4 Type Specific Cabin Crew Training Course
  - The Seller will provide type specific training for cabin crews at the Seller's Training Centers.

If the Buyer's Aircraft is to incorporate special features, the type specific cabin crew training course will be performed no earlier than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] before the scheduled Delivery Date of the Buyer's first Aircraft.

16.6.5 Training on Aircraft

During any and all flights performed in accordance with this Clause 16.6, the Buyer will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The Buyer will assist the Seller, if necessary, in obtaining the validation of the licenses of the Seller's pilots performing Base Flight Training or initial operating experience by the Aviation Authority of the place of registration of the Aircraft.

16.7 Performance / Operations Courses

The Seller will provide performance/operations training for the Buyer's personnel as defined in Appendix A to this Clause 16.

The available courses will be listed in the Seller's Customer Services Catalog current at the time of the course.

16.8 Maintenance Training

16.8.1 The Seller will provide maintenance training for the Buyer's ground personnel at the Seller's Training Centers as further set forth in Appendix A to this Clause 16.

The available courses will be as listed in the Seller's Customer Services Catalog current at the time of the course.

The practical training provided in the frame of maintenance training will be performed on the training devices in use in the Seller's Training Centers.

16.8.2 Practical Training on Aircraft

Notwithstanding Clause 16.8.1 above, upon the Buyer's request, the Seller may provide Instructors for the performance of practical training on aircraft ("Practical Training").

Irrespective of the location at which the training takes place, the Buyer will provide at its own cost an aircraft for the performance of the Practical Training.

Should the Buyer require the Seller's Instructors to provide Practical Training at facilities selected by the Buyer, such training will be subject to prior approval of the facilities by the Seller. All costs related to such Practical Training, including but not limited to the Seller's approval of the facilities, will be borne by the Buyer.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] subject to the conditions detailed in Paragraph 4.4 thereof.

16.9 Supplier and Propulsion System Manufacturer Training

Upon the Buyer's request, the Seller will provide to the Buyer the list of the maintenance and overhaul training courses provided by major Suppliers and the applicable Propulsion System Manufacturer on their respective products.

16.10 Proprietary Rights

All proprietary rights, including but not limited to patent, design and copyrights, relating to the Seller's training data and documentation will remain with the Seller and/or its Affiliates and/or its Suppliers, as the case may be.

These proprietary rights will also apply to any translation into a language or languages or media that may have been performed or caused to be performed by the Buyer.

# 16.11 Confidentiality

The Seller's training data and documentation are designated as confidential and as such are provided to the Buyer for the sole use of the Buyer, for training of its own personnel, who undertakes not to disclose the content thereof in whole or in part, to any third party without the prior written consent of the Seller, save as permitted herein or otherwise pursuant to any government or legal requirement imposed upon the Buyer.

In the event of the Seller having authorized the disclosure of any training data and documentation to third parties either under this Agreement or by an express prior written authorization, the Buyer will cause such third party to agree to be bound by the same conditions and restrictions as the Buyer with respect to the disclosed training data and documentation and to use such training data and documentation solely for the purpose for which they are provided.

# 16.12 Transferability

Unless otherwise permitted pursuant to Clause 21, the Buyer's rights under this Clause 16 may not be assigned, sold, transferred, novated or otherwise alienated by operation of law or otherwise, without the Seller's prior written consent.

### 16.13 Indemnities and Insurance

INDEMNIFICATION PROVISIONS AND INSURANCE REQUIREMENTS APPLICABLE TO THIS CLAUSE 16 ARE AS SET FORTH IN CLAUSE 19.

THE BUYER WILL PROVIDE THE SELLER WITH AN ADEQUATE INSURANCE CERTIFICATE PRIOR TO ANY TRAINING ON AIRCRAFT.

### **APPENDIX A TO CLAUSE 16**

# TRAINING ALLOWANCE

For the avoidance of doubt, all quantities indicated below are the total quantities granted for all Aircraft firmly ordered, unless otherwise specified.

The contractual training courses defined in this Appendix A will be provided [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] under this Agreement.

Notwithstanding the above, flight operations training courses granted per firmly ordered Aircraft in this Appendix A will be provided by the Seller within a period starting [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Any deviation to such training delivery schedule will be mutually agreed between the Buyer and the Seller.

### 1 FLIGHT OPERATIONS TRAINING

# 1.1 Flight Crew Training (standard transition course)

The Seller will provide flight crew training (standard transition course) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 1.2 Low Visibility Operations Training

The Seller will provide [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Low Visibility Operations Training for [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 1.3 Flight Crew Line Initial Operating Experience

The Seller will provide to the Buyer pilot Instructor(s) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Unless otherwise agreed during the Training Conference, in order to follow the Aircraft Delivery schedule, the maximum number of pilot Instructors present at any one time will be limited to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] pilot Instructors.

### 1.4 Type Specific Cabin Crew Training Course

The Seller will provide to the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] type specific training for cabin crews for up to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 1.5 Airbus Pilot Instructor Course (APIC)

The Seller will provide to the Buyer transition Airbus Pilot Instructor Course(s) (APIC), for flight and synthetic instruction, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the Buyer's flight instructors. APIC courses will be performed in groups of two (2) trainees.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 2 PERFORMANCE / OPERATIONS COURSE(S)

The Seller will provide to the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of performance / operations training [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] for the Buyer's personnel.

### 3 MAINTENANCE TRAINING

- 3.1 The Seller will provide to the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of maintenance training [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] for the Buyer's personnel.
- 3.2 The Seller will provide to the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Engine Run-up courses [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 4 TRAINEE DAYS ACCOUNTING

Trainee days are counted as follows:

- **4.1** For instruction at the Seller's Training Centers: one (1) day of instruction for one (1) trainee equals one (1) trainee day. The number of trainees originally registered at the beginning of the course will be counted as the number of trainees to have taken the course.
- **4.2** For instruction outside of the Seller's Training Centers: one (1) day of instruction by one (1) Seller Instructor equals the actual number of trainees attending the course or a minimum of

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] trainee days, except for structure maintenance training course(s).

- **4.3** For structure maintenance training courses outside the Seller's Training Center(s), one (1) day of instruction by one (1) Seller Instructor equals the actual number of trainees attending the course or the minimum number of trainees as indicated in the Seller's Customer Services Catalog.
- For practical training, whether on training devices or on aircraft, one (1) day of instruction by one (1) Seller Instructor equals the actual number of trainees attending the course or a minimum of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 17 EQUIPMENT SUPPLIER PRODUCT SUPPORT

- 17.1 Equipment Supplier Product Support Agreements
- 17.1.1 The Seller has obtained enforceable and transferable Supplier Product Support Agreements from all Suppliers of Supplier Parts installed on the Aircraft, the benefit of which is hereby transferred to the Warranty Beneficiary. Such agreements become enforceable as soon as and for as long as an operator is identified as an Airbus aircraft operator. For the avoidance of doubt, transfer to the Warranty Beneficiary of any Supplier Product Support Agreement does not limit the right of the Warranty Beneficiary to enter into any additional agreement with such Supplier on different terms than provided in such Supplier Product Support Agreement.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 17.1.2 These agreements are based on the World Airlines Suppliers Guide, are made available to the Warranty Beneficiary through the SPSA Application, and include Supplier commitments as contained in the Supplier Product Support Agreements which include the following provisions:
  - (i) Technical data and manuals required to operate, maintain, service and overhaul the Supplier Parts will be prepared in accordance with the applicable provisions of ATA Specification including revision service and be published in the English language. The Seller will recommend that a software user guide, where applicable, be supplied in the form of an appendix to the Component Maintenance Manual. Such data will be provided in compliance with the applicable ATA Specification;
  - (ii) Warranties and guarantees, including standard warranties. In addition, landing gear Suppliers will provide service life policies for selected structural landing gear elements:
  - (iii) Training to ensure efficient operation, maintenance and overhaul of the Supplier Parts for the Buyer's instructors, shop and line service personnel;
  - (iv) Spares data in compliance with ATA Specification 2200 (iSpec 2200), initial provisioning recommendations, spare parts and logistic service including routine and expedite deliveries;
  - (v) Technical service to assist the Warranty Beneficiary with maintenance, overhaul, repair, operation and inspection of Supplier Parts as well as required tooling and spares provisioning.

### 17.2 Supplier Compliance

The Seller will monitor Suppliers' compliance with support commitments defined in the Supplier Product Support Agreements and will, if necessary, jointly take remedial action with the Warranty Beneficiary. The Seller authorizes the Warranty Beneficiary to enforce directly against all Suppliers all such support commitments to the fullest extent permitted by the applicable Supplier Product Support Agreement.

- 17.3 Nothing in this Clause 17 will be construed to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 17.4 Familiarization Training

Upon the Buyer's request, the Seller will provide the Warranty Beneficiary with Supplier Product Support Agreements familiarization training at the Seller's facilities in Blagnac, France. An on-line training module will be further available through AirbusWorld, access to which will be subject to the GTC.

### 18 BUYER FURNISHED EQUIPMENT

- 18.1 Administration
- 18.1.1 In accordance with the Specification, the Seller will install those items of equipment that are identified in the Specification as being furnished by the Buyer ("Buyer Furnished Equipment" or "BFE"), provided that the BFE and the supplier of such BFE (the "BFE Supplier") are referred to in the Airbus BFE Product Catalog valid at the time the BFE Supplier is selected.
- 18.1.2 Notwithstanding the foregoing and without prejudice to Clause 2.5.2, if the Buyer wishes to install BFE that is not referred to in the Airbus BFE Product Catalog or if the Buyer wishes to use a supplier that is not a BFE Supplier, the Buyer will so inform the Seller and the Seller will, if the Seller deems necessary, complete a feasibility study [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] In addition, it is a prerequisite to such approval that the considered supplier be qualified by the Seller's Aviation Authorities to produce equipment for installation on civil aircraft. The Buyer will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] cause any BFE supplier approved under this Clause 18.1.2 (each an "Approved BFE Supplier") to comply with the conditions set forth in this Clause 18 and specifically Clause 18.2.

Except for the specific purposes of this Clause 18.1.2, the term BFE Supplier will be deemed to include Approved BFE Suppliers.

18.1.3 The Seller will advise the Buyer of the dates by which, in the planned release of engineering for the Aircraft, the Seller requires from each BFE Supplier a written detailed engineering definition encompassing a Declaration of Design and Performance (the "BFE Engineering Definition"). The Seller will provide to the Buyer and/or the BFE Supplier(s), within an appropriate timeframe, the necessary interface documentation to enable the development of the BFE Engineering Definition.

The BFE Engineering Definition will include the description of the dimensions and weight of BFE, the information related to its certification and the information necessary for the installation and operation thereof, including when applicable 3D models compatible with the Seller's systems. The Buyer will furnish, or cause the BFE Suppliers to furnish, the BFE Engineering Definition by the dates specified.

Thereafter, the BFE Engineering Definition will not be revised, except through an SCN executed in accordance with Clause 2.

18.1.4 The Seller will also provide to the Buyer, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] a schedule of dates and the shipping addresses for delivery of the BFE and, where requested by the Seller, additional spare BFE to permit installation in the Aircraft and Delivery of the Aircraft in accordance with the Aircraft delivery schedule. The Buyer will provide, or cause the BFE Suppliers to provide, the BFE by such dates in a serviceable condition, in order to allow performance of any assembly, installation, test or acceptance process in accordance with the Seller's industrial schedule. In order to facilitate the follow-up of the timely receipt of BFE, the Buyer will, upon the Seller's request, provide to the Seller dates and references of all BFE purchase orders placed by the Buyer. The Seller will reasonably cooperate with the Buyer and the BFE Suppliers to help ensure that the BFE is provided to the Seller by the requisite dates in proper and serviceable condition.

The Buyer will also provide, when requested by the Seller, at the Airbus Operations S.A.S. facility in Toulouse, France, and/or the Airbus Operations GmbH Division Hamburger Flugzeugbau facility in Hamburg, Germany, adequate field service including support from BFE Suppliers to act in a technical advisory capacity to the Seller in the installation, calibration and possible repair of BFE.

18.1.5 Without prejudice to the Buyer's obligations hereunder, in order to facilitate the development of the BFE Engineering Definition, the Seller will organize meetings between the Buyer and BFE Suppliers. The Buyer hereby agrees to participate in such meetings and to provide adequate technical and engineering expertise to reach decisions within a timeframe specified by the Seller.

In addition, prior to Delivery of the Aircraft to the Buyer, the Buyer agrees:

- (i) to monitor the BFE Suppliers and ensure that they will enable the Buyer to fulfill its obligations, including but not limited to those set forth in the Customization Milestone Chart;
- (ii) that, should a timeframe, quality or other type of risk be identified at a given BFE Supplier, the Buyer will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] allocate resources to such BFE Supplier so as not to jeopardize the industrial schedule of the Aircraft;
- (iii) for major BFE, including, but not being limited to, seats, galleys and IFE ("**Major BFE**") to participate on a mandatory basis in the specific meetings that take place between BFE Supplier selection and BFE delivery, namely:
  - (a) Preliminary Design Review ("PDR"),
  - (b) Critical Design Review ("CDR");
- (iv) to attend the First Article Inspection ("FAI") for the first shipset of all Major BFE. Should the Buyer not attend such FAI, the Buyer will delegate the FAI to the BFE Supplier thereof and confirmation thereof will be supplied to the Seller in writing;
- (v) to attend the Source Inspection ("SI") that takes place at the BFE Supplier's premises prior to shipping, for each shipset of all Major BFE. Should the Buyer not attend such SI, the Buyer will delegate the SI to the BFE Supplier and confirmation thereof will be delivered to the Seller in writing. Should the Buyer not attend the SI, the Buyer will be deemed to have accepted the conclusions of the BFE Supplier with respect to such SI.

The Seller will attend the PDR, the CDR and the FAI. In doing so, the Seller's employees will be acting in an advisory capacity only and at no time will they be deemed to be acting as Buyer's employees or agents, either directly or indirectly.

18.1.6 The Seller will have the authority to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The BFE will be imported into France or into Germany by the Buyer under a suspensive customs system (*Régime de l'entrepôt douanier ou -régime de perfectionnement actif* or *Zollager oder Aktive Veredelung*) without application of any French or German tax or customs duty

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

BFE will be delivered to the following shipping addresses:

Airbus Operations S.A.S. 316 Route de Bayonne 31300 Toulouse France

or

Airbus Operations GmbH Kreetslag 10 21129 Hamburg Germany,

or such other location as may be specified by the Seller.

- 18.2 Applicable Requirements
- 18.2.1 The Buyer will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] ensure, at its expense, that the BFE will:
  - (i) be manufactured by either a BFE Supplier referred to in the Airbus BFE Product Catalog or an Approved BFE Supplier, and

- (ii) meet the requirements of the applicable Specification of the Aircraft, and
- (iii) be delivered with the relevant certification documentation, including but not limited to the DAP (Delivery at Place) according to the Incoterms, and
- (iv) comply with the BFE Engineering Definition, and
- (v) comply with applicable requirements incorporated by reference to the Type Certificate and listed in the Type Certificate Data Sheet, and
- (vi) be approved by the Aviation Authority issuing the Export Airworthiness Certificate and by the Buyer's Aviation Authority for installation and use on the Aircraft at the time of Delivery of the Aircraft, and
- (vii) not infringe any patent, copyright or other intellectual property right of the Seller or any third party, and
- (viii) at the time of delivery to the Seller, not be subject to any legal obligation or other encumbrance that may prevent, hinder or delay the installation of the BFE in the Aircraft and/or the Delivery of the Aircraft.

The Seller will be entitled to refuse any item of BFE that is incompatible with the Specification, the BFE Engineering Definition or the certification requirements.

- 18.2.2 Unless otherwise provided in this Clause 18, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] the Seller will reasonably cooperate with the Buyer and the BFE Suppliers with regard to the timely manufacture, shipment, delivery, certification, design and installation of the BFE and provide for storage of the BFE.
- 18.3 Buyer's Obligation and Seller's Remedies
- 18.3.1 Any delay or failure by the Buyer or the BFE Suppliers in:
  - (i) complying with the undertakings set forth in Clause 18.2.1 or in providing the BFE Engineering Definition or field service mentioned in Clause 18.1.4, or
  - (ii) furnishing the BFE in a serviceable condition at the requested delivery date, or
  - (iii) obtaining any required approval for such BFE under the above mentioned Aviation Authorities' regulations,

may delay the performance of any act to be performed by the Seller, including Delivery of the Aircraft. The Seller will not be responsible for a delay in the Delivery of the Aircraft to the extent such delay is caused by a delay or failure by the Buyer or the BFE Suppliers described in the immediately preceding sentence. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 18.3.2 In addition, in the event of any delay or failure mentioned in 18.3.1 above, the Seller may [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) select, purchase and install equipment similar to the BFE at issue if such BFE is required to obtain the Export Certificate of Airworthiness for the affected Aircraft, in which event the Final Price of the affected Aircraft will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] or
  - (ii) if the BFE is delayed by more [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 18.4 Title and Risk of Loss

Title to and risk of loss of any BFE will at all times remain with the Buyer except that risk of loss (limited to cost of replacement of such BFE) will be with the Seller for as long as such BFE is under the care, custody and control of the Seller.

- 18.5 Disposition of BFE Following Termination
- 18.5.1 If a termination of this Agreement pursuant to Clause 20 occurs with respect to an Aircraft in which all or any part of the BFE has been installed prior to the date of such termination, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 18.5.2 If a termination of this Agreement pursuant to Clause 20 occurs with respect to an Aircraft in which all or any part of the BFE has been installed prior to the date of such termination [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 18.5.3 If a termination of this Agreement pursuant to Clause 20 occurs with respect to an Aircraft, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 18.5.4 If a termination of this Agreement pursuant to Clause 20 occurs with respect to an Aircraft, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 18.5.5 If a termination of this Agreement by either party pursuant to Clause 11 occurs with respect to an Aircraft in which all or any part of the BFE has been installed prior to the date of such termination, the Seller will, at the Seller's option, either [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 18.5.6 If a termination of this Agreement by either party pursuant to Clause 11 occurs with respect to an Aircraft, for any BFE delivered to the Seller's facilities for such Aircraft but not installed prior to the date of such termination, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 18.5.7 If a termination of this Agreement pursuant to Clause 10 occurs with respect to an Aircraft in which all or any part of the BFE has been installed prior to the date of such termination, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 18.5.8 If a termination of this Agreement pursuant to Clause 10 occurs with respect to an Aircraft in which all or any part of the BFE has been installed prior to the date of such termination, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 18.5.9 If a termination of this Agreement pursuant to Clause 10 occurs with respect to an Aircraft, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

18.5.10	If a termination of this Agreement pursuant to Clause 10 occurs with respect to an Aircraft, and [CONFIDENTIAL PORTION OMITTED AND
	FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 18.5.11 The Seller will notify the Buyer as to those items of BFE [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and, at the Seller's request, the Buyer will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the date of such notice. The Buyer will have no claim against the Seller for damage, loss or destruction of any item of BFE removed from the affected Aircraft and not removed from Seller's facility [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 18.5.12 The Buyer will have no claim against the Seller for damage to or destruction of any item of BFE damaged or destroyed in the process of being removed from the affected Aircraft pursuant to this Clause 18.5, provided that the Seller will use reasonable care in such removal.
- 18.5.13 The Buyer will grant the Seller title to any BFE items that cannot be removed from the affected Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 19 INDEMNITIES AND INSURANCE

The Seller and the Buyer will each be liable for Losses (as defined below) arising from the acts or omissions of its respective directors, officers, agents or employees occurring during or incidental to such party's exercise of its rights and performance of its obligations under this Agreement, except as provided in Clauses 19.1 and 19.2.

### 19.1 Seller's Indemnities

The Seller will, except in the case of gross negligence or willful misconduct of the Buyer, its directors, officers, agents and/or employees, be solely liable for and will indemnify, defend and hold the Buyer, its Affiliates and each of their respective directors, officers, agents, employees and insurers harmless against all losses, liabilities, claims, damages, costs and expenses, including settlement costs, court costs, expert witness costs and reasonable attorneys' fees ("Losses"), arising from:

- claims for injuries to, or death of, the Seller's directors, officers, agents or employees, or loss of, or damage to, property of the Seller or its
  employees when such Losses result from either party's exercise of any right or performance of any obligation under this Agreement, and
- (ii) claims for injuries to, or death of, third parties, or loss of, or damage to, property of third parties, that result from the Technical Acceptance Flights.

# 19.2 Buyer's Indemnities

The Buyer will, except in the case of gross negligence or willful misconduct of the Seller, its directors, officers, agents and/or employees, be solely liable for and will indemnify, defend and hold the Seller, its Affiliates, its subcontractors, and each of their respective directors, officers, agents, employees and insurers, harmless against all Losses arising from:

- (i) claims for injuries to, or death of, the Buyer's directors, officers, agents or employees, or loss of, or damage to, property of the Buyer or its employees, when such Losses result from either party's exercise of any right or performance of any obligation under this Agreement, and
- (ii) claims for injuries to, or death of, third parties, or loss of, or damage to, property of third parties, that result from (i) the provision of Seller Representatives services under Clause 15 including services performed on board the aircraft or (ii) the provision of Aircraft Training Services to the Buyer.

### 19.3 Notice and Defense of Claims

If any claim is made or suit is brought against a party or entity entitled to indemnification under this Clause 19 (the "Indemnitee") for damages for which liability has been assumed by the other party under this Clause 19 (the "Indemnitor"), the Indemnitee will promptly [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] give notice of such claim or suit to the Indemnitor, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Upon receipt of such notice, the Indemnitor (unless otherwise requested by the Indemnitee) will assume

and conduct the defense, or settlement, of such claim or suit. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Notice of the claim or suit will be accompanied by all information pertinent to the matter as is reasonably available to the Indemnitee and will be followed by such cooperation by the Indemnitee as the Indemnitor or its counsel may reasonably request, at the expense of the Indemnitor.

If the Indemnitor fails or refuses to assume the defense of any claim or suit notified to it under this Clause 19, the Indemnitee will, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] have the right, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to proceed with the defense or settlement of the claim or suit as it deems prudent; [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 19.4 Buyer's Insurance
- 19.4.1 The Buyer will:
  - (i) cause the Seller, its Affiliates, its subcontractors and each of their respective directors, officers, agents and employees to be named as additional insured under the Buyer's Comprehensive Aviation Legal Liability insurance policies, including War Risk and Allied Perils coverage of the scope covered by AVN 52E (or any then-current equivalent form), and
  - (ii) with respect to the Buyer's Hull All Risks and Hull War Risks insurances and Allied Perils, cause the insurers of the Buyer's hull insurance policies to waive all rights of subrogation against the Seller, its Affiliates, its subcontractors and each of their respective directors, officers, agents, employees and insurers.
- Any deductible incurred pursuant to this Clause 19.4 will be borne by the Buyer. The Buyer will furnish to the Seller, not less than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] a certificate of insurance, in English, evidencing the limits of liability cover of no less than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and period of insurance coverage in a form reasonably acceptable to the Seller from the Buyer's insurance broker(s), certifying that such policies have been endorsed as follows:
  - (i) under the Comprehensive Aviation Legal Liability Insurances, the Buyer's policies are primary and non-contributory to any insurance maintained by the Seller,
  - (ii) such insurance can only be cancelled or materially altered by the giving of not less than thirty (30) days (but seven (7) days or such lesser period as may be customarily available in respect of War Risks and Allied Perils) prior written notice thereof to the Seller, and

- (iii) under any such cover, all rights of subrogation against the Seller, its Affiliates, its subcontractors and each of their respective directors, officers, agents, employees and insurers have been waived.
- 19.5 Seller's Insurance
- 19.5.1 The Seller will:
  - cause the Buyer, its Affiliates and each of their respective directors, officers, agents and employees to be named as additional insured under the Seller's Aviation Legal Liability insurance policies, including War Risk and Allied Perils coverage of the scope covered by AVN 52E (or any then equivalent form), and
  - (ii) with respect to the Seller's Hull All Risks and Hull War Risks insurances and Allied Perils, cause the insurers of the Seller's hull insurance policies to waive all rights of subrogation against the Buyer, its Affiliates and each of their respective directors, officers, agents, employees and insurers.
- 19.5.2 Any deductible incurred pursuant to this Clause 19.5 will be borne by the Seller. The Seller will furnish to the Buyer, not less than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] prior to each Delivery under this Agreement, a certificate of insurance, in English, evidencing the limits of liability cover of no less than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and period of insurance coverage in a form reasonably acceptable to the Buyer from the Seller's insurance broker(s), certifying that such policies have been endorsed as follows:
  - (i) under the Comprehensive Aviation Legal Liability Insurances, the Seller's policies are primary and non-contributory to any insurance maintained by the Buyer,
  - (ii) such insurance can only be cancelled or materially altered by the giving of not less than thirty (30) days (but seven (7) days or such lesser period as may be customarily available in respect of War Risks and Allied Perils) prior written notice thereof to the Buyer, and
  - (iii) under any such cover, all rights of subrogation against the Buyer, its Affiliates and each of their respective directors, officers, agents, employees and insurers have been waived.

### 20 TERMINATION

#### 20.1 Termination Events

Each of the following will constitute a "Termination Event"

- (1) The Buyer or any of its Affiliates commences in any jurisdiction any case, proceeding or other action with respect to the Buyer or any of its Affiliates or their properties relating to bankruptcy, insolvency, reorganization, winding-up, liquidation, dissolution or other relief from, or with respect to, or readjustment of, its debts or obligations.
- (2) An action is commenced in any jurisdiction seeking the appointment of a receiver, trustee, custodian or other similar official for the Buyer or any of its respective Affiliates or for all or any substantial part of their respective assets, and such action remains unstayed, undismissed or undischarged for [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] or the Buyer or any of its Affiliates makes a general assignment for the benefit of its creditors.
- (3) An action is commenced in any jurisdiction against the Buyer or any of its respective Affiliates seeking issuance of a warrant of attachment, execution, distraint or similar process against all or any substantial part of their respective assets, and such action remains unstayed, undismissed or undischarged for [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (4) The Buyer or any of its Affiliates becomes the object, in any jurisdiction, of a case, proceeding or action similar or analogous to any of the events mentioned in Clause 20.1(1), (2) or (3).
- (5) The Buyer or any of its Affiliates is generally not able, or is expected to be unable to, or will admit in writing its inability to, pay its debts as they become due.
- (6) The Buyer or any of its Affiliates commences negotiations with significant creditors, existing or potential, either with the intention of restructuring all or a substantial part of all of its outstanding obligations or in preparation for a bankruptcy filing under the U.S. Bankruptcy
- (7) The Buyer or any of its Affiliates fails to make (i) any payment required to be made under this Agreement or any other material agreement between the Buyer or any of its Affiliates and the Seller or any of its Affiliates when such payment is due, (ii) any Predelivery Payment required to be made under this Agreement when such payment is due, (iii) payment of all or part of the Final Price of any Aircraft required to be made under this Agreement; (iv) any payment to a Lessor with respect to any Leased Aircraft.
- (8) The Buyer repudiates, cancels or terminates this Agreement in whole or in part.
- (9) The Buyer defaults in its obligation to take delivery of an Aircraft as provided in Clause 9.2.

- (10) The Buyer or any of its Affiliates defaults in the observance or performance of any other covenant, undertaking or obligation contained in this Agreement or any other material agreement between the Buyer or its Affiliates, on the one hand, and the Seller or its Affiliates on the other hand, provided that, if such breach or default is capable of being cured and such breach or default is not cured within any specified cure period.
- (11) Any other event that the parties agree in writing constitutes a Termination Event.
- 20.2 Remedies in Event of Termination
- 20.2.1 If a Termination Event occurs, the Buyer will be in material breach of this Agreement, and the Seller can elect any of the following remedies under the applicable law:
  - A. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - B. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - C. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - D. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 20.2.2 In the event Seller elects a remedy under any of Clauses 20.2.1(A)(B) or (C), above:
  - A. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - B. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - C. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 20.2.3 If the Seller elects a Termination under Clause 20.2.1(D) above:
  - A. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - i. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - ii. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- iii. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iv. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- v. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- vi. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- vii. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- B. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 20.2.4 The parties to this Agreement are commercially sophisticated parties acting within the same industry, and represented by competent counsel and the parties expressly agree and declare as follows:
  - A. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - B. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - C. the liquidated damages provision of this Clause 20 has been fully negotiated by sophisticated parties represented by counsel, is a material component of the consideration granted and, in the absence of such liquidated damages provision, the consideration would have been materially different.

#### 20.3 Definitions

For purposes of this Clause 20, the terms "Affected Aircraft", "Applicable Date" and "Escalated Price" are defined as follows:

- i. "Affected Aircraft" any or all Aircraft with respect to which the Seller has cancelled or terminated this Agreement pursuant to Clause 20.2.1 D,
- ii. "Applicable Date" for any Affected Aircraft, the date the Seller issues the notice and [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] pursuant to Clause 20.2.3 B.

iii. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] **Price**" - will have the same meaning as the "Final Price" of the Aircraft as that term is defined in Clause 3.2, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 20.4 Notice of Termination Event

Within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of becoming aware of the occurrence of a Termination Event by the Buyer, the Buyer will notify the Seller of such occurrence in writing, provided, that any failure by the Buyer to notify the Seller will not prejudice the Seller's rights or remedies hereunder.

#### 20.5 Information Covenants

The Buyer hereby covenants and agrees that, from the date of this Agreement until no further Aircraft are to be delivered hereunder, the Buyer will furnish or cause to be furnished to the Seller the following:

- a. Annual Financial Statements. As soon as available and in any event no later than the date that the Buyer furnishes such annual statements to the Securities and Exchange Commission or successor thereto (the "SEC") (i) a copy of the SEC Form 10-K filed by the Buyer with the SEC for such fiscal year, or, if no such Form 10-K was filed by the Buyer for such a fiscal year, the consolidated balance sheet of the Buyer and its Subsidiaries, as at the end of such fiscal year and the related consolidated statements of operations, of common stockholders' equity (deficit) (in the case of the Buyer and its Subsidiaries) and of cash flows for such fiscal year, setting forth comparative consolidated figures as of the end of and for the preceding fiscal year, and examined by any firm of independent public accountants of recognized standing selected by the Buyer and reasonably acceptable to the Seller, whose opinion will not be qualified as to the scope of audit or as to the status of the Buyer as a going concern, and (ii) a certificate of such accounting firm stating that its audit of the business of the Buyer was conducted in accordance with generally accepted auditing standards.
- b. Quarterly Financial Statements. As soon as available and in any event no later than the date that the Buyer furnishes such quarterly statements to the Securities and Exchange Commission or successor thereto, a copy of the SEC Form 10-Q filed by the Buyer with the SEC for such quarterly period, or, if no such Form 10-Q was filed by the Buyer with respect to any such quarterly period, the consolidated balance sheet of the Buyer and its Subsidiaries, as at the end of such quarterly period and the related consolidated statements of operations for such quarterly period and for the elapsed portion of the fiscal year ended with the last day of such quarterly period and in each case setting forth comparative consolidated figures as of the end of and for the related periods in the prior fiscal year, all of which will be certified by an Authorized Officer of the Buyer, subject to changes resulting from audit and normal year-end audit adjustments.
- c. <u>Debt Rescheduling</u>. (i) Promptly upon the Buyer commencing negotiations with one or more of its significant creditors with a view to general readjustment or rescheduling of all or any material part of its indebtedness under circumstances in which a reasonable

business person, in the exercise of prudent business judgment, would conclude that the Buyer would otherwise not be able to pay such indebtedness as it falls due, notice of commencement of such negotiations, and (ii) thereafter timely advice of the progress of such negotiations until such negotiations are terminated or completed.

- d. Acceleration of other indebtedness. Immediately upon knowledge by the Buyer that the holder of any bond, debenture, promissory note or any similar evidence of indebtedness of the Buyer or Affiliate thereof ("Other Indebtedness") has demanded payment, given notice or exercised its right to a remedy having the effect of acceleration with respect to a claimed event of default under any Other Indebtedness, where the impact of the acceleration is likely to have a material adverse effect on the Buyer's ability to perform its obligations under or in connection with the transactions contemplated by this Agreement, notice of the demand made, notice given or action taken by such holder and the nature and status of the claimed event of default and what the action the Buyer is taking with respect thereto.
- e. <u>Other Information</u>. Promptly upon transmission thereof, copies of any filings and registrations with, and reports to, the SEC by the Buyer or any of its Subsidiaries, and, with reasonable promptness, such other information or documents (financial or otherwise) as the Seller may reasonably request from time to time.
  - For the purposes of this Clause 20, (x) an "**Authorized Officer**" of the Buyer will mean the Chief Executive Officer, the Chief Financial Officer or any Vice President and above who reports directly or indirectly to the Chief Financial Officer and (y) "**Subsidiaries**" will mean, as of any date of determination, those companies owned by the Buyer whose financial results the Buyer is required to include in its statements of consolidated operations and consolidated balance sheets.
- Nothing contained in this Clause 20 will be deemed to waive or limit the Seller's rights or ability to request adequate assurance under Article 2, Section 609 of the Uniform Commercial Code (the "UCC"). It is further understood that any commitment of the Seller or the Propulsion Systems manufacturer to provide financing to the Buyer will not constitute adequate assurance under Article 2, Section 609 of the UCC.

#### 21 ASSIGNMENTS AND TRANSFERS

#### 21.1 Assignments

Except as hereinafter provided, neither party may sell, assign, novate or transfer its rights or obligations under this Agreement to any person without the prior written consent of the other, except that either party may sell, assign or transfer its rights or obligations under this Agreement to any of its Affiliate without the other party's consent, provided that the assigning party will remain ultimately responsible for fulfillment of all obligations undertaken by such party in this Agreement.

#### 21.2 Assignments on Sale, Merger or Consolidation

The Buyer will be entitled to assign its rights under this Agreement at any time due to a merger, consolidation or a sale of all or substantially all of its assets without the consent of the Seller, provided that:

- (i) the surviving or acquiring entity has executed an assumption agreement, in form and substance reasonably acceptable to the Seller, agreeing to assume all of the Buyer's obligations under this Agreement;
- (ii) at the time, and immediately following the consummation, of the merger, consolidation or sale, no Buyer Termination Event exists or will have occurred and be continuing; and
- (iii) there exists with respect to the surviving or acquiring entity no basis for a Buyer Termination Event.

# 21.3 Designations by Seller

The Seller may at any time by notice to the Buyer designate facilities or personnel of the Seller or any other Affiliate of the Seller at which or by whom the services to be performed under this Agreement will be performed. Notwithstanding such designation, the Seller will remain ultimately responsible for fulfillment of all obligations undertaken by the Seller in this Agreement.

#### 21.4 Transfer of Rights and Obligations upon Reorganization

In the event that the Seller is subject to a corporate restructuring having as its object the transfer of, or succession by operation of law in, all or a substantial part of its assets and liabilities, rights and obligations, including those existing under this Agreement, to a person (the "Successor") that is an Affiliate of the Seller at the time of that restructuring, for the purpose of the Successor carrying on the business carried on by the Seller at the time of the restructuring, such restructuring will be completed without consent of the Buyer following notification by the Seller to the Buyer in writing. The Buyer recognizes that succession of the Successor to the Agreement by operation of law that is valid under the law pursuant to which that succession occurs will be binding upon the Buyer.

#### 21.5 Assignment in Connection with Aircraft Financing

Prior to Delivery of an Aircraft, the Buyer will not resell, lease, or transfer such Aircraft without the Seller's written consent, which consent will not be unreasonably withheld. The Seller will take any

requested action (including, but not limited to, the execution and delivery of a consent and agreement in substantially the form to be agreed by the Buyer and the Seller prior to the Delivery of the first Aircraft) reasonably required for the purpose of causing an Aircraft, at or following Delivery, to be subject to a sale, lease equipment trust, conditional sale, lien or other arrangement for the financing by the Buyer of the Aircraft. However, no such action will require the Seller to divest itself of title to or possession of the Aircraft until Delivery of and payment for the Aircraft.

#### 21.6 Assignment in Connection with Sale or Lease of Aircraft

With the consent of the Seller, not to be unreasonably withheld or delayed, if, following Delivery of an Aircraft, the Buyer sells or leases such Aircraft, the Buyer may assign all or any of its rights under this Agreement to the purchaser or lessee of such Aircraft if the purchaser or lessee of such Aircraft enters into an agreement in substantially the form to be agreed by the Buyer and the Seller prior to the Delivery of the first Aircraft, such agreement to contain provisions whereby the purchaser or lessee agrees to be bound by and comply with all applicable terms of this Agreement.

#### 22 MISCELLANEOUS PROVISIONS

#### 22.1 Data Retrieval

On the Seller's reasonable request, the Buyer will provide the Seller with all the necessary data, as customarily compiled by the Buyer and pertaining to the operation of the Aircraft, to assist the Seller in making an efficient and coordinated survey of all reliability, maintenance, operational and cost data with a view to monitoring the efficient and cost effective operations of the Airbus fleet worldwide.

#### 22.2 Notices

All notices, requests and other communications required or authorized hereunder will be given in writing either by personal delivery to an authorized officer of the party to whom such notice request or other communication is given or by recognized commercial courier that provides signed acknowledgment of receipt, certified air mail (return receipt requested) or facsimile at the addresses and numbers set forth below. The date on which any such notice, request or other communication is given by recognized commercial courier, certified air mail or facsimile, the date on which received, will be deemed to be the effective date of such notice, request or other communication (provided that notices, requests and other communications received on a day that is not a Business Day or after 5:00 p.m., local time at the place of receipt, on a Business Day will be deemed to have been received on the next Business Day).

The Seller will be addressed at:

Airbus S.A.S.

Attention: Senior Vice President Contracts

1. Rond Point Maurice Bellonte

31707 Blagnac Cedex,

France

Facsimile: 33 (05) 61 93 39 94 The Buyer will be addressed at:

4333 Amon Carter Boulevard Fort Worth, Texas 76155 United States of America

Facsimile: (817) 967-4313 Attention: Corporate Secretary

From time to time, the party receiving the notice, request or other communication may designate another address or another person by notice given in accordance with this Clause 22.2.

#### 22.3 Waive

The failure of either party to enforce at any time any of the provisions of this Agreement, to exercise any right herein provided or to require at any time performance by the other party of any of the provisions hereof will in no way be construed to be a present or future waiver of such

provisions nor in any way to affect the validity of this Agreement or any part hereof or the right of the other party thereafter to enforce each and every such provision. The express waiver by either party of any provision, condition or requirement of this Agreement will not constitute a waiver of any future obligation to comply with such provision, condition or requirement.

#### 22.4 International Supply Contract

The Buyer and the Seller recognize that this Agreement is an international supply contract which has been the subject of discussion and negotiation, that all its terms and conditions are fully understood by the parties, and that the Specification and price of the Aircraft and the other mutual agreements of the parties set forth herein were arrived at in consideration of, inter alia, all provisions hereof, specifically including all waivers, releases and remunerations by the Buyer set out herein.

#### 22.5 Certain Representations of the Parties

#### 22.5.1 Buyer's Representations

The Buyer represents and warrants to the Seller:

- (i) the Buyer is a corporation organized and existing in good standing under the laws of the State of Delaware and has the corporate power and authority to enter into and perform its obligations under this Agreement;
- (ii) neither the execution and delivery by the Buyer of this Agreement, nor the consummation of any of the transactions by the Buyer contemplated hereby, nor the performance by the Buyer of the obligations hereunder, constitutes a breach of any agreement to which the Buyer is a party or by which its assets are bound;
- (iii) this Agreement has been duly authorized, executed and delivered by the Buyer and constitutes the legal, valid and binding obligation of the Buyer enforceable against the Buyer in accordance with its terms, subject as to enforceability to applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium, and similar laws affecting creditors' rights and remedies generally and to general principles of equity (whether enforcement is sought in a proceeding at law or in equity).

#### 22.5.2 Seller's Representations

The Seller represents and warrants to the Buyer:

- (i) the Seller is organized and existing in good standing under the laws of the Republic of France and has the corporate power and authority to enter into and perform its obligations under the Agreement;
- (ii) neither the execution and delivery by the Seller of this Agreement, nor the consummation of any of the transactions by the Seller contemplated hereby, nor the performance by the Seller of the obligations hereunder, constitutes a breach of any agreement to which the Seller is a party or by which its assets are bound;
- (iii) this Agreement has been duly authorized, executed and delivered by the Seller and constitutes the legal, valid and binding obligation of the Seller enforceable against the

Seller in accordance with its terms, subject as to enforceability to applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium, and similar laws affecting creditors' rights and remedies generally and to general principles of equity (whether enforcement is sought in a proceeding at law or in equity).

#### 22.6 Interpretation and Law

22.6.1 THIS AGREEMENT WILL BE GOVERNED BY AND CONSTRUED AND THE PERFORMANCE THEREOF WILL BE DETERMINED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT GIVING EFFECT TO ITS CONFLICTS OF LAWS PROVISIONS THAT WOULD RESULT IN THE APPLICATION OF THE LAW OF ANY OTHER JURISDICTION.

Each of the Seller and the Buyer hereby irrevocably submits itself to the nonexclusive jurisdiction of the courts of the state of New York, Borough of Manhattan, and of the United States District Court for the Southern District of New York, for the purposes of any suit, action or other proceeding arising out of this Agreement, the subject matter hereof or any of the transactions contemplated hereby brought by any party or parties hereto. To the extent that either the Buyer or the Seller has or may hereafter acquire any right of immunity, whether characterized as sovereign immunity or otherwise, and whether under the United States Foreign Sovereign Immunities Act of 1976 (or any successor legislation) or otherwise, from any legal proceedings, whether in the United States or elsewhere, to enforce or collect upon this Agreement, including, without limitation, immunity from suit or service of process, immunity from jurisdiction or judgment of any court or tribunal or execution of a judgment, or immunity of any of its property from attachment prior to any entry of judgment, or from attachment in aid of execution upon a judgment, it hereby irrevocably and expressly waives any such immunity, and agrees not to assert any such right or claim in any such proceeding, whether in the United States or elsewhere.

THE PARTIES HEREBY ALSO AGREE THAT THE UNITED NATIONS CONVENTION ON CONTRACTS FOR THE INTERNATIONAL SALE OF GOODS WILL NOT APPLY TO THIS TRANSACTION.

22.6.2 Service of process in any suit, action or proceeding in respect of any matter as to which the Seller or the Buyer has submitted to jurisdiction under Clause 22.6 may be made (i) on the Seller by delivery of the same personally or by dispatching the same via Federal Express, UPS, or similar international air courier service prepaid, return receipt requested, to CT Corporation, New York City offices as agent for the Seller, it being agreed that service upon CT Corporation will constitute valid service upon the Seller or by any other method authorized by the laws of the State of New York, and (ii) on the Buyer by delivery of the same personally or by dispatching the same by Federal Express, UPS, or similar international air courier service prepaid, return receipt requested to its address in Clause 22.2, or by any other method authorized by the laws of the State of New York.

#### 22.7 Headings

All headings in this Agreement are for convenience of reference only and do not constitute a part of this Agreement.

#### 22.8 Waiver of Jury Trial

EACH OF THE PARTIES HERETO WAIVES ITS RIGHT TO TRIAL BY JURY IN ANY LITIGATION ARISING OUT OF OR RELATING TO THIS AGREEMENT AND FOR ANY COUNTERCLAIM OR CROSS-CLAIM THEREIN.

#### 22.9 Waiver of Consequential Damages

In no circumstances will either party claim or receive consequential, or incidental damages under this Agreement (other than any such damages actually paid to a third party that is not an indemnified party under this Agreement pursuant to a third party claim).

#### 22.10 No Representations Outside of this Agreement

The parties declare that, prior to the execution of this Agreement, they, with the advice of their respective counsel, apprised themselves of sufficient relevant data in order that they might intelligently exercise their own judgments in deciding whether to execute this Agreement and in deciding on the contents of this Agreement. Each party further declares that its decision to execute this Agreement is not predicated on or influenced by any declarations or representations by any other person, party, or any predecessors in interest, successors, assigns, officers, directors, employees, agents or attorneys of any such person or party, except as set forth in this Agreement. This Agreement resulted from negotiation involving counsel for all of the parties hereto and no term herein will be construed or interpreted against any party under the *contra proferentum* or any related doctrine.

#### 22.11 Confidentiality

Subject to any legal or governmental requirements of disclosure, the parties (which for this purpose will include their employees and legal counsel) will maintain the terms and conditions of this Agreement and any reports or other data furnished hereunder strictly confidential, including but not limited to, the Aircraft pricing and delivery schedule (the "Confidential Information"). Without limiting the generality of the foregoing, each of the Buyer and the Seller will use its commercially reasonable efforts to limit the disclosure of the contents of this Agreement to the extent legally permissible in (i) any filing required to be made with any governmental agency and will make such applications as will be necessary to implement the foregoing, and (ii) any press release concerning the whole or any part of the contents and/or subject matter hereof or of any future addendum hereto. With respect to any public disclosure or filing, each of the Buyer and the Seller agrees to submit to the other party a copy of the proposed document to be filed or disclosed and will give the other party a reasonable period of time in which to review such document. The Buyer and the Seller will consult with each other prior to the making of any public disclosure or filing, permitted hereunder, of this Agreement or the terms and conditions thereof. Subject to any legal or governmental requirements of disclosure, the parties will agree in writing on any press release or other public disclosure to announce the transactions contemplated by this Agreement.

The provisions of this Clause 22.11 will survive any termination of this Agreement.

#### 22.12 Severability

If any provision of this Agreement should for any reason be held ineffective, the remainder of this Agreement will remain in full force and effect. To the extent permitted by applicable law, each party hereto hereby waives any provision of law that renders any provision of this Agreement prohibited or unenforceable in any respect.

#### 22.13 Entire Agreement

This Agreement contains the entire agreement between the parties with respect to the subject matter hereof and supersedes any previous understanding, commitments or representations whatsoever, whether oral or written. This Agreement will not be amended or modified except by an instrument in writing of even date herewith or subsequent hereto executed by both parties or by their fully authorized representatives.

#### 22.14 Inconsistencies

In the event of any inconsistency between the terms of this Agreement and the terms contained in either (i) the Specification, or (ii) any other Exhibit, in each such case the terms of this Agreement will prevail over the terms of the Specification or any other Exhibit. For the purpose of this Clause 22.14, the term Agreement will not include the Specification or any other Exhibit hereto.

#### 22.15 Language

All correspondence, documents and any other written matters in connection with this Agreement will be in English.

#### 22.16 Counterparts

This Agreement has been executed in two (2) original copies.

Notwithstanding the foregoing, this Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute but one and the same instrument.

# 22.17 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

 $[Remainder\ of\ the\ page\ is\ intentionally\ blank]$ 

AIRBUS S.A.S.
By: Title:
AMERICAN AIRLINES, INC.
Ву:
Title:

IN WITNESS WHEREOF, this Agreement was entered into as of the day and year first above written.

# **DELIVERY SCHEDULE**

Aircraft Rank	
1 through 10	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY
11 through 15	WITH THE COMMISSION PURSUANT TO A REQUEST FOR
16 through 20	CONFIDENTIAL TREATMENT]
21 through 28	
29 through 37	
38 through 46	
47 through 55 56 through 63	
64 through 70	
71 through 78	
79 through 85	
86 through 91	
92 through 98	
99 through 104	
105 through 110	
111 through 118	
119 through 125	
126 through 130	
131 through 132	
133 through 140	
141 through 147	
148 through 153	
154 through 159	•
160 through 165	
166 through 172	
173 through 178	
179 through 184	
185 through 190	
191 through 197	
198 through 203	

Aircraft Rank	
204 through 209	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY
210 through 215	WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]'
216 through 221	
222 through 228	
229 through 234	
235 through 240	
241 through 245	
246 through 250	
251 through 255	
256 through 260	

# STANDARD SPECIFICATIONS

The A319 Aircraft Standard Specification is contained in a separate folder.

The A320 Aircraft Standard Specification is contained in a separate folder.

The A321 Aircraft Standard Specification is contained in a separate folder.

# FORM OF A SPECIFICATION CHANGE NOTICE

			EXHIBIT B-1
SPECIFICATION CE		For SCN Number Issue Dated Page	
Title:			
Description :			
Remarks / References			
Specification changed by this SCN			
This SCN requires prior or concurrent a	cceptance of the following SCN	(s):	
Price per aircraft			
US DOLLARS: AT DELIVERY CONDITIONS:			
This change will be effective on	AIRCRAFT N°	and subsequent.	
Provided approval is received by			
Buyer approval	Seller approval		
By:	By:		
Date:	Date :		

	EXHIBIT B-1		
SPECIFICATION CHANGE NOTICE (SCN)	For SCN Number Issue Dated Page		
Specification repercussion:  After contractual agreement with respect to weight, performance, delivery, etc, the indicated part of the specification wording will read as follows:			

	For
	SCN Number
	Issue
SPECIFICATION CHANGE NOTICE	Dated
(SCN)	Page
a de la grap avrapa de revolución de la contraction de la contract	
Scope of change (FOR INFORMATION ONLY)	

FORM OF A MANUFACTURER'S SPECIFICATION CHANGE NOTICE

MANUFACTURER'S SPECIFICAT CHANGE NOTICE (MSCN)	TION	For MSCN Number Issue Dated Page
Title:	<u>'</u>	
Description :		
<ul> <li>Effect on weight:</li> <li>• Manufacturer's Weight Empty change:</li> <li>• Operational Weight Empty change:</li> <li>• Allowable Payload change:</li> </ul>		
Remarks / References		
Specification changed by this MSCN		
Price per aircraft		
US DOLLARS: AT DELIVERY CONDITIONS:		
This change will be effective on	AIRCRAFT N°	and subsequent.

Seller approval

By:

Date:

Provided MSCN is not rejected by

**Buyer approval** 

By:

Date:

MANUFACTURER'S SPECIFICATION CHANGE NOTICE (MSCN)	For MSCN Number Issue Dated Page		
Specification repercussion:  After contractual agreement with respect to weight, performance, delivery, etc, the indicated part of the specification wording will read as follows:			

	For
MANUFACTURER'S SPECIFICATION CHANGE NOTICE	MSCN Number Issue
	Dated
(MSCN)	Page
Scope of change (FOR INFORMATION ONLY)	

# **SELLER PRICE REVISION FORMULA - 2011**

#### 1.1 Base Price

The Base Price as quoted in Clause 3.1 of the Agreement is subject to adjustment for [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 1.2 Base Period

The Airframe Base Price has been established in accordance with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.3 Indexes

<u>Labor Index:</u> [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Material Index: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1.4 Revision Formula

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 1.5 <u>General Roundings</u>
- 1.5.1 Roundings

The Labor Index average and the Material Index average [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1.5.2 Substitution of Indexes for [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

If:

- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] or
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.5.3 <u>Final Index Values</u>

The Index values as defined in Paragraph 1.4 above will be considered final [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.5.4 Limitation

# **SELLER PRICE REVISION FORMULA -2010**

#### 1.1 Base Price

The Base Price as quoted in Clause 3.1 of the Agreement is [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.2 Base Period

The Airframe Base Price has been [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.3 Indexes

<u>Labor Index:</u> [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Material Index: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1.4 Revision Formula

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1.5 <u>General Roundings</u>

1.5.1 Roundings

The Labor Index average and the Material Index average [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1.5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

If:

- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] or
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.5.3 <u>Final Index Values</u>

The Index values as defined in Paragraph 1.4 above will be considered final [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.5.4 Limitation

#### CFM INTERNATIONAL PROPULSION SYSTEM PRICE REVISION FORMULA

#### 1.1 Reference Price of the Propulsion System

The Reference Price for a set of two (2) CFM International CFM56-5B6/3 series Propulsion System is as quoted in Clause 3.1.2.1 of the Agreement.

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.1 of the Agreement.

The Reference Price for a set of two (2) CFM International CFM56-5B4/3 series Propulsion System is as quoted in Clause 3.1.2.2 of the Agreement.

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.2 of the Agreement.

 $The \ Reference \ Price \ for \ a \ set \ of \ two \ (2) \ CFM \ International \ CFM56-5B3/3 \ series \ Propulsion \ System \ is \ as \ quoted \ in \ Clause \ 3.1.2.3 \ of \ the \ Agreement.$ 

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.3 of the Agreement.

The Reference Price for a set of two (2) CFM International LEAP-X1A24 series Propulsion System is as quoted in Clause 3.1.2.7 of the Agreement.

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.4 of the Agreement.

The Reference Price for a set of two (2) CFM International LEAP-X1A26 series Propulsion System is as quoted in Clause 3.1.2.8 of the Agreement.

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.5 of the Agreement.

The Reference Price for a set of two (2) CFM International LEAP-X1A32 series Propulsion System is as quoted in Clause 3.1.2.9 of the Agreement.

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.6 of the Agreement.

These Reference Prices are [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT and] in accordance with the provisions of Paragraphs 1.4. and 1.5. hereof.

#### 1.2 Reference Periods

The Reference Price for a set of two (2) CFM International CFM56-5B series Propulsion System has been established in accordance with the economic conditions prevailing for a theoretical delivery in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as

defined by CFM International by the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The Reference Price for a set of two (2) CFM International LEAP-X series Propulsion System has been established in accordance with the economic conditions prevailing for a theoretical delivery in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as defined by CFM International by the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.3 <u>Indexes</u>

Labor Index: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Material Index: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 1.4 <u>Revision Formula</u>

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.5 <u>General Provisions</u>

## 1.5.1 Roundings

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1 5	5.2	⊬inal	Indov	Values

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 1.5.3 <u>Interruption of Index Publication</u>

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1.5.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.5.5 <u>Limitation</u>

#### INTERNATIONAL AERO ENGINES PROPULSION SYSTEM PRICE REVISION FORMULA

#### 1.1 <u>Propulsion System Reference Price</u>

The Reference Price for a set of two (2) International Aero Engines V2524-A5 series Propulsion System is as quoted in Clause 3.1.2.4 of the Agreement.

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.1 of the Agreement.

The Reference Price for a set of two (2) International Aero Engines V2527-A5 series Propulsion System is as quoted in Clause 3.1.2.5 of the Agreement.

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.2 of the Agreement.

The Reference Price for a set of two (2) International Aero Engines V2533-A5 series Propulsion System is as quoted in Clause 3.1.2.6 of the Agreement.

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.3 of the Agreement.

These Reference Prices are subject to adjustment for changes in economic conditions as measured by data obtained from the US Department of Labor, Bureau of Labor Statistics, and in accordance with the provisions hereof.

#### 1.2 Reference Period

The above Reference Price has been established in accordance with the averaged economic conditions prevailing in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as defined, according to International Aero Engines by the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.3 <u>Indexes</u>

Labor Index: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Material Index: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH

THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.4 Revision Formula

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.5 <u>General Provisions</u>

#### 1.5.1 Roundings

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.5.2 <u>Final Index Values</u>

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.5.3 <u>Interruption of Index Publication</u>

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1.5.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.5.5 <u>Limitation</u>

### PRATT AND WHITNEY PROPULSION SYSTEM PRICE REVISION FORMULA

### 1.1 Reference Price of the Propulsion System

The Reference Price for a set of two (2) Pratt and Whitney PW1124G Propulsion System is as quoted in Clause 3.1.2.10 of the Agreement.

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.4 of the Agreement.

The Reference Price for a set of two (2) Pratt and Whitney PW1127G Propulsion System is as quoted in Clause 3.1.2.11 of the Agreement.

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.5 of the Agreement.

The Reference Price for a set of two (2) Pratt and Whitney PW1133G Propulsion System is as quoted in Clause 3.1.2.12 of the Agreement.

This Reference Price applies to the Propulsion System type as specified in Clause 2.3.6 of the Agreement.

These Reference Prices are subject to adjustment for changes in economic conditions as measured by data obtained from the US Department of Labor, Bureau of Labor Statistics, and in accordance with the provisions hereof.

### 1.2 Base Period

The Reference Price has been established in accordance with the average economic conditions prevailing in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.3 Indexes

<u>Labor Index</u>: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Metal Index: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 1.4 Revision Formula

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 1.5 <u>General Provisions</u>

## 1.5.1 Roundings

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 1.5.2 <u>Substitution of Indexes for Price Revision Formula</u>

If:

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] or
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] or
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 1.5.3 Final Index Values

The Index values as defined in Paragraph 1.4 above will be considered final [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 1.5.4 Limitation

EXHIBIT D

## FORM OF CERTIFICATE OF ACCEPTANCE

In accordance with the terms of [clause [—]] of the purchase agreement dated [day] [month] [year] and made between [insert name of the party to the Purchase Agreement] (the "Customer") and Airbus S.A.S. as amended and supplemented from time to time (the "Purchase Agreement"), the Technical Acceptance Process (as defined in the Purchase Agreement) relating to one Airbus A3[—]-[—] aircraft, bearing manufacturer's serial number [—], and registration mark [—], [Customer Airplane Number [—], with two (2) [—] series propulsion systems installed thereon, serial numbers [—] (position #1) and [—] (position #2) (the "Aircraft") has been successfully completed in [Blagnac/Hamburg].

The Customer, [as agent of [insert the name of the lessor/SPC] (the "**Owner**") pursuant to the [purchase agreement assignment] dated [*day*] [*month*] [*year*], between the Customer and the Owner] hereby accepts the Aircraft for delivery in accordance with the provisions of the Purchase Agreement.

Such acceptance will not impair the rights that may be derived from the warranties, indemnities and performance guaranties relating to the Aircraft set forth in the Purchase Agreement.

Any right at law or otherwise to revoke this acceptance of the Aircraft is hereby irrevocably waived.

IN WITNESS WHEREOF, the Customer, [as agent of the Owner] has caused this instrument to be executed by its duly authorized representative this of [month], [year] in [Blagnac/Hamburg].

day

### **CUSTOMER** [as agent of **OWNER**]

Name: Title: Signature:

### FORM OF WARRANTY BILL OF SALE

Know all men by these presents that Airbus S.A.S., a société par actions simplifiée existing under French law and having its principal office at 1 rond-point Maurice Bellonte, 31707 Blagnac Cedex, FRANCE (the "Seller"), was this [day] [month] [year] the owner of the title to the following airframe (the "Airframe"), the [engines/propulsion systems] as specified (the "[Engines/Propulsion Systems]") and [all appliances, components, parts, instruments, appurtenances, accessories, furnishings, modules and other equipment of any nature], [excluding buyer furnished equipment ("BFE"),] incorporated therein, installed thereon or attached thereto on the date hereof (the "Parts"):

**AIRFRAME**:

[ENGINES/PROPULSION SYSTEMS]:

[Insert name of engine or propulsion system manufacturer] Model [—]

AIRBUS Model A3[—]-[—]

MANUFACTURER'S
SERIAL NUMBER: [—]

DATE OF MANUFACTURE: [—]

**ENGINE SERIAL NUMBERS**:

LH: [—] RH: [—]

REGISTRATION MARK: [—]

[and has such title to the BFE as was acquired by it from [insert name of vendor of the BFE] pursuant to a bill of sale dated

[month] [year] (the "BFE Bill

of Sale")].

The Airframe, [Engines/Propulsion Systems] and Parts are hereafter together referred to as the "Aircraft".

The Seller does this day of [month] [year], grant, convey, bargain, sell, transfer, deliver and set over all of its rights, title and interest in and to the Aircraft [and the BFE] to the following entity and to its successors and assigns forever, such Aircraft [and the BFE] to be the property thereof:

[Insert Name/Address of Buyer] (the "Buyer")

The Seller hereby warrants to the Buyer, its successors and assigns that it has [(i)] good and lawful right to sell, deliver and transfer title to the Aircraft to the Buyer and that there is hereby conveyed to the Buyer good, legal and valid title to the Aircraft, free and clear of all liens, claims, charges, encumbrances and rights of others and that the Seller will warrant and defend such title forever against all claims and demands whatsoever [and (ii) such title to the BFE as Seller has acquired from [insert name of vendor of the BFE] pursuant to the BFE Bill of Sale].

This Warranty Bill of Sale will be governed by and construed in accordance with the laws of [same governing law as the Purchase Agreement].

# EXHIBIT E

IN WITNESS WHEREOF, the undersigned has caused this instrument to be executed by its duly authorized representative this [Blagnac/Hamburg].

day of [month], [year] in

# AIRBUS S.A.S.

Name:

Title:

Signature:

### SELLER SERVICE LIFE POLICY – LIST OF ITEMS

- 1 The Items covered by the Service Life Policy pursuant to Clause 12.2 are those Seller Items of primary and auxiliary structure described hereunder.
- 2 WINGS CENTER AND OUTER WING BOX (LEFT AND RIGHT)
- 2.1 Wing Structure
- 2.1.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.1.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.1.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.2 Fittings
- 2.2.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.2.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.2.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3 Auxiliary Support Structure
- 2.3.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3.1.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3.1.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3.2.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3.2.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2.3.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3.3.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3.3.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2.4 Pylon

- 2.4.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.1.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.1.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.1.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.1.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 3 FUSELAGE

### 3.1 Fuselage structure

- 3.1.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.5 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.6 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.8	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR
	CONFIDENTIAL TREATMENT]

## 3.2 Fittings

- 3.2.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 4 STABILIZERS

- **4.1** [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.1.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.1.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.1.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.1.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.1.5 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.1.5.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.1.5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 4.2 Vertical Stabilizer Main Structural Box

- 4.2.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.2.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.2.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### **EXHIBIT F**

4.2.4	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR
	CONFIDENTIAL TREATMENT]

- 4.2.5 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.2.5.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.2.5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### TECHNICAL DATA INDEX

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# OPERATIONAL MANUALS AND DATA

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# OPERATIONAL MANUALS AND DATA

# MAINTENANCE AND ASSOCIATED MANUALS

# STRUCTURAL MANUALS

# OVERHAUL DATA

# ENGINEERING DOCUMENTS

# ENGINEERING DOCUMENTS

# ENGINEERING DOCUMENTS

# ENGINEERING DOCUMENTS

# **EXHIBIT H**

# MATERIAL

# SUPPLY AND SERVICES

### 1. GENERAL

- 1.1 Scope
- **1.1.1** This Exhibit H sets forth the terms and conditions for the support and services offered by the Seller to the Buyer with respect to Material (as defined below).
- 1.1.2 References made to Articles will be deemed to refer to articles of this Exhibit H unless otherwise specified.
- **1.1.3** For purposes of this Exhibit H:
  - (i) The term "Supplier" will mean any supplier providing any of the Material listed in Article 1.2.1 and the term "Supplier Part" will mean an individual item of Material.
  - (ii) The term "SPEC 2000" means the "E-Business Specification for Materiels Management" document published by the Air Transport Association of America.

### 1.2 Material Categories

- **1.2.1** Each of the following constitutes "**Material**" for purposes of this Exhibit H:
  - (i) Seller Parts;
  - (ii) Supplier Parts classified as Repairable Line Maintenance Parts (as defined in SPEC 2000);
  - (iii) Supplier Parts classified as Expendable Line Maintenance Parts (as defined in SPEC 2000); and
  - (iv) Seller and Supplier ground support equipment and specific-to-type tools.

where "Seller Parts" means Seller's proprietary parts bearing a part number of the Seller or for which the Seller has the exclusive sales rights.

- 1.2.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- **1.2.3** Airworthiness Certification

The Seller represents and warrants that each Material is manufactured under an approval granted or accepted by the FAA under 14 C.F.R. Part 21 of the Federal Aviation Regulations. Upon delivery to the Buyer, the Seller will provide with each Material a certification that the Material has been approved or accepted by the FAA under 14 C.F.R. Part 21.

Acceptable certification documents include the following:

- Federal Aviation Administration FAA Form 8130-3 Airworthiness Approval Tag
- Joint Aviation Authorities JAA Form One Authorized Release Certificate
- European Aviation Safety Agency EASA Form One Authorized Release Certificate
- Transport Canada TC Form 24-0078 Authorized Release Certificate

• A Certificate of Conformance or Material Certification Document that lists the parts by Manufacturer's part number and condition, and contains a signed statement certifying that such parts are aeronautical replacement parts listing the specific FAA approval or acceptance criteria applicable to such parts.

To meet the above criteria, it is permissible for the Seller to provide the Buyer with copies of certification documents originally supplied to the Seller with shipments of Material prior to the transfer of the Material to the Buyer. The original certification documents are required to meet certification requirements stated above and the original copies, if not provided to the Buyer with the shipments, will remain on file at the Seller's facility.

### 1.3 Term

During a period commencing on the date hereof and continuing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] the Seller will maintain, or cause to be maintained, a reasonable stock of Seller Parts.

### 1.4 Airbus Material Store

**1.4.1** US Spares Center

The Seller has established and will maintain or cause to be maintained, during the Term, a US store ("US Spares Center"). The US Spares Center will be operated twenty-four (24) hours per day, seven (7) days per week, for the handling of AOG and critical orders for Seller Parts

**1.4.2** Material Support Center, Germany

The Seller has established its material headquarters in Hamburg, Germany (the "Airbus Material Center") and will, during the Term, maintain, or have maintained on its behalf, a central store of Seller Parts. The Airbus Material Center will be operated twenty-four (24) hours per day, seven (7) days per week.

- **1.4.3** Other Points of Shipment
- **1.4.3.1** In addition to the AACS Spares Center and the Airbus Material Center, the Seller and its Affiliates operate a global network of regional satellite stores (The "**Regional Satellite Stores**"). A list of such stores will be provided to the Buyer upon the Buyer's request.
- **1.4.3.2** Subject to Article 1.4.1, the Seller reserves the right to effect deliveries from distribution centers other than the US Spares Center or the Airbus Material Center, which may include the Regional Satellite Stores or any other production or Supplier's facilities.
- **1.4.4** The Seller will use its commercially reasonable efforts to deliver Seller Parts to the Buyer from any Material store specified in Clause 1.4.1, 1.4.2, or 1.4.3.
- 1.5 INTENTIONALLY LEFT BLANK

### 1.6 Customer Order Desk

The Seller operates a "Customer Order Desk", the main functions of which are:

- (i) Management of order entries for all priorities, including Aircraft On Ground ("AOG");
- (ii) Management of order changes and cancellations;
- (iii) Administration of Buyer's routing instructions;
- (iv) Management of Material returns;
- (v) Clarification of delivery discrepancies; and
- (vi) Issuance of credit and debt notes.

The Buyer hereby agrees to communicate its orders for Material to the Customer Order Desk either in electronic format (SPEC 2000) or via the Internet.

### 1.7 Commitments of the Buyer

- 1.7.1 The Buyer may [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.7.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.7.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.7.2.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1.7.2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2 INITIAL PROVISIONING

### 2.1 Period

The initial provisioning period commences with the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] ("Initial Provisioning Period").

### 2.2 Pre-Provisioning Meeting

2.2.1 The Seller will organize a pre-provisioning meeting at US Spares Center or at the Airbus Material Center, or at any other agreed location, for the purpose of setting an acceptable schedule and working procedure for the preparation of the initial issue of the Provisioning Data and the Initial Provisioning Conference referred to in Articles 2.3 and 2.4 below (the "Pre-Provisioning Meeting").

During the Pre-Provisioning Meeting, the Seller will familiarize the Buyer with the provisioning processes, methods and formulae of calculation and documentation.

2.2.2 The Pre-Provisioning Meeting will take place on an agreed date that is no later than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] prior to Scheduled Delivery Month of the first Aircraft, allowing a minimum preparation time of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] for the Initial Provisioning Conference.

### 2.3 Initial Provisioning Conference

The Seller will organize an initial provisioning conference at the US Spares Center or at the Airbus Material Center (the "Initial Provisioning Conference"), the purpose of which will be to agree the material scope and working procedures to accomplish the initial provisioning of Material (the "Initial Provisioning").

The Initial Provisioning Conference will take place at the earliest [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2.4 Provisioning Data

- 2.4.1 Provisioning data generally in accordance with SPEC 2000, Chapter 1, for Material described in Articles 1.2.1 (i) through 1.2.1 (iv) ("Provisioning Data") will be supplied by the Seller to the Buyer in the English language, in a format and timeframe to be agreed during the Pre-Provisioning Meeting.
- 2.4.1.1 Unless a longer revision cycle has been agreed, the Provisioning Data will be revised [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- **2.4.1.2** The Seller will ensure that Provisioning Data is provided to the Buyer in time to permit the Buyer to perform any necessary evaluation and to place orders in a timely manner.
- 2.4.1.3 Provisioning Data generated by the Seller will comply with the configuration of the Aircraft as documented [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] before the date of issue

This provision will not cover:

- (i) Buyer modifications not known to the Seller,
- (ii) other modifications not approved by the Seller's Aviation Authorities.

### 2.4.2 Supplier-Supplied Data

Provisioning Data relating to each Supplier Part (both initial issue and revisions) will be produced by Supplier thereof and may be delivered to the Buyer either by the Seller or such Supplier. It is agreed and understood by the Buyer that the Seller will not be responsible for the substance, accuracy and quality of such data. Such Provisioning Data will be provided in either SPEC 2000 format or any other agreed format.

### 2.4.3 Supplementary Data

The Seller will provide the Buyer with data supplementary to the Provisioning Data, comprising local manufacture tables, ground support equipment, specific-to-type tools and a pool item candidate list.

### 2.5 Commercial Offer

Upon the Buyer's request, the Seller will submit a commercial offer for Initial Provisioning Material which will include a delivery date for such Initial Provisioning Material.

### 2.6 Delivery of Initial Provisioning Material

- 2.6.1 During the Initial Provisioning Period, Initial Provisioning Material will conform to the latest known configuration standard of the Aircraft for which such Material is intended as reflected in the Provisioning Data transmitted by the Seller.
- **2.6.2** The delivery of Initial Provisioning Material will take place (i) according to the conditions specified in the commercial offer mentioned in Article 2.5 and (ii) at a location designated by the Buyer.
- 2.6.3 All Initial Provisioning Material will be packaged in accordance with ATA 300 Specification.

### 2.6.4 Shortages, Overshipments, Non-Conformity in Orders

The Buyer will, within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] after delivery of Material pursuant to a purchase order, advise the Seller:

a) of any alleged shortages or over shipments, and

b) of any non-conformities of delivered Material.

In the event of the Buyer not having advised the Seller of any such alleged shortages, over shipments or non-conformity within the above-defined period, the Buyer will be deemed to have accepted the delivery.

In the event of the Buyer reporting over shipments or non-conformity to the specifications within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] the Seller will, if the Seller recognizes such over shipment or non-conformity, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - vi) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- vii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- f) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- g) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 3. OTHER MATERIAL SUPPORT

- 3.1 As of the date hereof, the Seller currently offers various types of parts support through the Customer Services Catalog on the terms and conditions set forth therein from time to time, including, but not limited to the lease of certain Seller Parts, the repair of Seller Parts and the sale or lease of ground support equipment and specific-to-type tools.
- 3.2 Unless otherwise agreed in the commercial offer subject to Clause 2.5 lead times will be in accordance with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Seller Parts as per Article 1.2.1(i) listed in the Seller's Spare Parts Price Catalog will be dispatched within the lead times [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Lead times for Seller Parts as per Article 1.2.1(i), [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Material defined in Articles 1.2.1(ii) through 1.2.1(vi) will be dispatched [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 4 WARRANTIES

#### 4.1 Seller Parts

Subject to the limitations and conditions as hereinafter provided, the Seller warrants to the Buyer that all Seller Parts, sold under this Exhibit H will at delivery to the Buyer:

- (i) be free from defects in material.
- (ii) be free from defects in workmanship, including without limitation processes of manufacture.
- (iii) be free from defects in design having regard to the state of the art of such design; and
- (iv) be free from defects arising from failure to conform to the applicable specification for such part.

### 4.1.1 Warranty Period

- 4.1.1.1 The warranty period for Seller Parts is [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] from delivery of such parts to the Buyer.
- 4.1.1.2 Whenever any Seller Part that contains a defect for which the Seller is liable under Article 4.1 has been corrected, replaced or repaired pursuant to the terms of this Article 4.1, the period of the Seller's warranty with respect to such corrected, repaired or replacement Seller Part, as the case may be, will be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 4.1.2 Buyer's Remedy and Seller's Obligation

The Buyer's remedy and Seller's obligation and liability under this Article 4.1 are, at the Seller's expense, (a) the repair, replacement or correction, at the Seller's expense, of any Seller Part that is defective (or the supply of modification kits rectifying the defect) or (b) the reimbursement of the cost to the Buyer for a replacement for the defective Seller Part and any other materials installed, incorporated, consumed or expended by the Buyer in performing such repair, replacement or correction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The provisions of Clauses 12.1.5 through 12.1.9 of the Agreement will apply to claims made pursuant to this Article 4.1.

### 4.2 Supplier Parts

With respect to Supplier Parts to be delivered to the Buyer under this Exhibit H, the Seller agrees to transfer to the Buyer the benefit of any warranties, which the Seller may have obtained from the corresponding Suppliers.

### 4.3 Waiver, Release and Renunciation

THIS ARTICLE 4 (INCLUDING ITS SUBPARTS) SETS FORTH THE EXCLUSIVE WARRANTIES, EXCLUSIVE LIABILITIES AND EXCLUSIVE OBLIGATIONS OF THE SELLER, AND THE EXCLUSIVE REMEDIES AVAILABLE TO THE BUYER, WHETHER

UNDER THIS EXHIBIT HOR OTHERWISE, ARISING FROM ANY DEFECT OR NONCONFORMITY OR PROBLEM OF ANY KIND IN ANY SELLER PART, MATERIAL, LEASED PART, OR SERVICES DELIVERED BY THE SELLER UNDER THIS EXHIBIT H.

THE BUYER RECOGNIZES THAT THE RIGHTS, WARRANTIES AND REMEDIES IN THIS ARTICLE 4 ARE ADEQUATE AND SUFFICIENT TO PROTECT THE BUYER FROM ANY DEFECT OR NONCONFORMITY OR PROBLEM OF ANY KIND IN THE SELLER PARTS, MATERIALS, LEASED PARTS, OR SERVICES SUPPLIED UNDER THIS EXHIBIT H. THE BUYER HEREBY WAIVES, RELEASES AND RENOUNCES ALL OTHER WARRANTIES, OBLIGATIONS, GUARANTEES AND LIABILITIES OF THE SELLER AND ALL OTHER RIGHTS, CLAIMS AND REMEDIES OF THE BUYER AGAINST THE SELLER AND ITS SUPPLIERS, WHETHER EXPRESS OR IMPLIED BY CONTRACT, TORT, OR STATUTORY LAW OR OTHERWISE, WITH RESPECT TO ANY NONCONFORMITY OR DEFECT OR PROBLEM OF ANY KIND IN ANY SELLER PART, MATERIAL, LEASED PART, OR SERVICES DELIVERED BY THE SELLER UNDER THIS EXHIBIT H, INCLUDING BUT NOT LIMITED TO:

- (1) ANY IMPLIED WARRANTY OF MERCHANTABILITY AND/OR FITNESS FOR ANY GENERAL OR PARTICULAR PURPOSE;
- (2) ANY IMPLIED OR EXPRESS WARRANTY ARISING FROM COURSE OF PERFORMANCE, COURSE OF DEALING OR USAGE OF TRADE;
- (3) ANY RIGHT, CLAIM OR REMEDY FOR BREACH OF CONTRACT;
- (4) ANY RIGHT, CLAIM OR REMEDY FOR TORT, UNDER ANY THEORY OF LIABILITY, HOWEVER ALLEGED, INCLUDING, BUT NOT LIMITED TO, ACTIONS AND/OR CLAIMS FOR NEGLIGENCE, GROSS NEGLIGENCE, INTENTIONAL ACTS, WILLFUL DISREGARD, IMPLIED WARRANTY, PRODUCT LIABILITY, STRICT LIABILITY OR FAILURE TO WARN;
- (5) ANY RIGHT, CLAIM OR REMEDY ARISING UNDER THE UNIFORM COMMERCIAL CODE OR ANY OTHER STATE OR FEDERAL STATUTE;
- (6) ANY RIGHT, CLAIM OR REMEDY ARISING UNDER ANY REGULATIONS OR STANDARDS IMPOSED BY ANY INTERNATIONAL, NATIONAL, STATE OR LOCAL STATUTE OR AGENCY;
- (7) ANY RIGHT, CLAIM OR REMEDY TO RECOVER OR BE COMPENSATED FOR:
  - (a) LOSS OF USE OR REPLACEMENT OF ANY AIRCRAFT, COMPONENT, EQUIPMENT, ACCESSORY OR PART PROVIDED UNDER THE AGREEMENT;
  - (b) LOSS OF, OR DAMAGE OF ANY KIND TO, ANY AIRCRAFT, COMPONENT, EQUIPMENT, ACCESSORY OR PART PROVIDED UNDER THE AGREEMENT;
  - (c) LOSS OF PROFITS AND/OR REVENUES;

### (d) ANY OTHER INCIDENTAL OR CONSEQUENTIAL DAMAGE.

THE WARRANTIES PROVIDED BY THIS EXHIBIT H WILL NOT BE EXTENDED, ALTERED OR VARIED EXCEPT BY A WRITTEN INSTRUMENT EXECUTED AND DELIVERED BY THE SELLER AND THE BUYER. IN THE EVENT THAT ANY PROVISION OF THIS ARTICLE 4 SHOULD FOR ANY REASON BE HELD UNLAWFUL, OR OTHERWISE UNENFORCEABLE, THE REMAINDER OF THIS ARTICLE 4 WILL REMAIN IN FULL FORCE AND EFFECT.

FOR THE PURPOSES OF THIS ARTICLE 4, THE "SELLER" WILL BE UNDERSTOOD TO INCLUDE THE SELLER, ANY OF ITS SUPPLIERS, SUBCONTRACTORS AND AFFILIATES.

### 4.4 Duplicate Remedies

The remedies provided to the Buyer under this Article 4 as to any part thereof are mutually exclusive and not cumulative. The Buyer will be entitled to the remedy that provides the maximum benefit to it, as the Buyer may elect, pursuant to the terms and conditions of this Article 4 for any particular defect for which remedies are provided under this Article 4; provided, however, that the Buyer will not be entitled to elect a remedy under one part of this Article 4 that constitutes a duplication of any remedy elected by it under any other part hereof for the same defect. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 5. COMMERCIAL CONDITIONS

### 5.1 Delivery Terms

All Material prices are quoted on the basis of Free Carrier (FCA) delivery terms, without regard to the place from which such Material is shipped. The term "Free Carrier (FCA)" is as defined by 2010 publication of the International Chamber of Commerce.

### 5.2 Payment Procedures and Conditions

All payments under this Exhibit H will be made in accordance with the terms and conditions set forth in the then current Customer Services e-Catalog, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5.3 Title

Title to any Material purchased under this Exhibit H will remain with the Seller until full payment of the invoices and interest thereon, if any, has been received by the Seller.

The Buyer hereby undertakes that Material title to which has not passed to the Buyer, will be kept free from any debenture or mortgage or any similar charge or claim in favour of any third party.

#### 6. EXCUSABLE DELAY

Clauses 10.1 and 10.2 of the Agreement will apply, mutatis mutandis, to all Material support and services provided under this Exhibit H.

## 7. TERMINATION OF MATERIAL PROCUREMENT COMMITMENTS

If the Agreement is terminated with respect to any Aircraft, the Seller will have the right to suspend deliveries to the Buyer under this Exhibit H and terminate the terms of this Exhibit H with respect to such Aircraft. If the Seller terminates the terms of this Exhibit H with respect to any Aircraft, then the Seller will have no further obligations and the Buyer will have no further rights under this Exhibit H with respect to such Aircraft.

### 8. INCONSISTENCY

In the event of any inconsistency between this Exhibit H and the Customer Services Catalog or any order placed by the Buyer, this Exhibit H will prevail to the extent of such inconsistency.

#### INDEX OF LETTER AGREEMENTS

- 1. Letter Agreement No. 1 Leasing Matters
- 2. Letter Agreement No. 2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. Letter Agreement No. 3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4. Letter Agreement No. 4 Option Aircraft
- 5. Letter Agreement No. 5 Flexibility
- 6. Letter Agreement No. 6 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7. Letter Agreement No. 7 *Reserved*
- 8. Letter Agreement No. 8 *Reserved*
- 9. Letter Agreement No. 9 Miscellaneous
- 10. Letter Agreement No. 10 Support Matters
- 11. Letter Agreement No. 11 Performance Guarantees
- 12. Letter Agreement No. 12 Reserved
- 13. Letter Agreement No. 13 Reserved
- 14. Letter Agreement No. 14 NEO [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft
- 15. Letter Agreement No. 15 Seller Parts and Seller Services
- 16. Letter Agreement No. 16 Specification Matters
- 17. Letter Agreement No. 17 Termination
- 18. Letter Agreement No. 18 Reserved

22.17.1

## AMERICAN AIRLINES A319 CUSTOMIZATION BUDGET PROPOSAL

Based on Standard Specification A319-100 issue 7.0 dated 20 June 2011

 $[{\tt CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]}$ 

Prepared by Airbus Americas Technical Contracts

## AMERICAN AIRLINES A319 CUSTOMIZATION BUDGET PROPOSAL

Based on Standard Specification A319-100 issue 7.0 dated 20 June 2011

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Prepared by Airbus Americas Technical Contracts

## AMERICAN AIRLINES A320 CUSTOMIZATION BUDGET PROPOSAL

Based on Standard Specification A320-200 issue 8.0 dated 20 June 2011

 $[{\tt CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]}$ 

Prepared by Airbus Americas Technical Contracts

## AMERICAN AIRLINES A320 CUSTOMIZATION BUDGET PROPOSAL

Based on Standard Specification A320-200 issue 8.0 dated 20 June 2011

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Prepared by Airbus Americas Technical Contracts

## AMERICAN AIRLINES A321 CUSTOMIZATION BUDGET PROPOSAL

Based on Standard Specification A321-200 issue 5.0 dated 20 June 2011

 $[{\tt CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]}$ 

Prepared by Airbus Americas Technical Contracts

## AMERICAN AIRLINES A321 CUSTOMIZATION BUDGET PROPOSAL

Based on Standard Specification A321-200 issue 5.0 dated 20 June 2011

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Prepared by Airbus Americas Technical Contracts

## AMERICAN AIRLINES A319NEO CUSTOMIZATION BUDGET PROPOSAL

Based on Standard Specification A319-100 issue 7.0 dated 20 June 2011

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

Prepared by Airbus Americas Technical Contracts

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Prepared by Airbus Americas Technical Contracts

## AMERICAN AIRLINES A320NEO CUSTOMIZATION BUDGET PROPOSAL

Based on Standard Specification A320-200 issue 8.0 dated 20 June 2011

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

Prepared by Airbus Americas Technical Contracts

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Prepared by Airbus Americas Technical Contracts

## AMERICAN AIRLINES A321NEO CUSTOMIZATION BUDGET PROPOSAL

Based on Standard Specification A321-200 issue 5.0 dated 20 June 2011

 $[{\tt CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT}]$ 

Prepared by Airbus Americas Technical Contracts

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Prepared by Airbus Americas Technical Contracts

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement dated as of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement"), which covers, among other matters, the sale by the Seller and the purchase by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 2 (this "Letter Agreement") certain additional terms and conditions regarding the sale or lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement will have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

#### 1 A319 AIRCRAFT

- 1.1 In respect of each A319 Aircraft that is sold by the Seller and purchased by the Buyer, the Seller will provide to the Buyer the following [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.2 The A319 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.3 The A319 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2 A320 AIRCRAFT

- 2.1 In respect of each A320 Aircraft that is sold by the Seller and purchased by the Buyer, the Seller will provide to the Buyer the following [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.2 The A320 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3 The A320 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 3 A321 AIRCRAFT

3.1 In respect of each A321 Aircraft that is sold by the Seller and purchased by the Buyer, the Seller will provide to the Buyer the following [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2 The A321 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3 The A321 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 4 A319 NEO AIRCRAFT

- 4.1 In respect of each A319 NEO Aircraft, the Seller will provide to the Buyer the following [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.2 The A319 NEO [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.3 The A319 NEO [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5 A320 NEO AIRCRAFT

- 5.1 In respect of each A320 NEO Aircraft, the Seller will provide to the Buyer the following [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5.2 The A320 NEO [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5.3 The A320 NEO [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 6 A321 NEO AIRCRAFT

- 6.1 In respect of each A321 NEO Aircraft, the Seller will provide to the Buyer the following [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.2 The A321 NEO [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.3 The A321 NEO [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

## 7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 7.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.1.1 Paragraph 4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to the Agreement is deleted and replaced with the following quoted text:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.1.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - Paragraph 1.5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to the Agreement is deleted and replaced with the following quoted text:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENTIUOTE
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.1.3 Paragraph 5.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to the Agreement is deleted and replaced with the following quoted text:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2.1 Paragraph 4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to the Agreement is deleted and replaced with the following quoted text:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - Paragraph 1.5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to the Agreement is deleted and replaced with the following quoted text:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2.3 Paragraph 5.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to the Agreement is deleted and replaced with the following quoted text:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3.1 For each Aircraft that is sold by the Seller and purchased by the Buyer and delivered to the Buyer on or before [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3.2 For each Aircraft that is sold by the Seller and purchased by the Buyer and delivered to the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - (b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - (c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - (d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3.4 For the avoidance of doubt, all references in the Agreement to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 8 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

#### 9 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

#### 10 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the Seller.	
	Very truly yours,
	AIRBUS S.A.S.
	Ву:
	Its:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By: Its:	

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an Airbus A320 Family Aircraft Purchase Agreement dated as of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement"), which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 3 (this "Letter Agreement") certain additional terms and conditions regarding the sale or lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement will have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

#### 1 DEFINITIONS

Clause 0 of the Agreement is amended to delete in its entirety the defined term "[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]"

#### 2 COMMITMENT FEE

Clause 5.2 of the Agreement is deleted in its entirety and replaced with the following quoted text:

OUOTE

The Seller acknowledges receipt from the Buyer of payment in the amount of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] which represents [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

UNQUOTE

#### 3 PREDELIVERY PAYMENTS

3.1 Clauses 5.3.1, 5.3.2, 5.3.3 and 5.3.4 of the Agreement are deleted in their entirety and replaced with the following quoted text:

QUOTE

- 5.3.1 Except as expressly set forth in the Agreement, Predelivery Payments are [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and will be paid by the Buyer to the Seller for the Aircraft.
- 5.3.2 INTENTIONALLY LEFT BLANK
- 5.3.3 The Buyer has paid or will pay Predelivery Payments to the Seller [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] in accordance with the following schedule (the "**Predelivery Payments**"):

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

5.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

UNQUOTE

3.2 Clause 5.3.5 is deleted in its entirety and replaced with the following quoted text:

QUOTE

5.3.5 Leased Aircraft Matters

The Buyer will pay [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

UNQUOTE

#### 4 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

#### 5 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

#### 6 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the Seller.	
	Very truly yours,
	AIRBUS S.A.S.
	Ву:
	Its:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By:	

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: OPTION AIRCRAFT

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 4 (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

#### 1 DEFINITIONS

Clause 0 of the Agreement is amended to either modify or add the following defined terms between the words "QUOTE" and "UNQUOTE": OUOTE

Additional Options – as defined in Paragraph 2.2.1 of Letter Agreement No. 4 hereto.

<u>Additional Option Aircraft</u> – all of the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft set forth in Schedule IB hereto that the Buyer has the option to purchase from the Seller subject to the terms and conditions hereof.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Additional Option Exercise Notice – as defined in Paragraph 2.2.3 of Letter Agreement No. 4 hereto.

Additional Option Exercise Notice Date – as defined in Paragraph 2.2.3 of Letter Agreement No. 4 hereto.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft – all of the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft set forth in Schedule IIB hereto that the Buyer has the option to purchase from the Seller subject to the terms and conditions hereof.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Option Aircraft – any or all of the CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft set forth in Schedule III that the Buyer has the option to purchase from the Seller subject to the terms and conditions hereof.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Option – as defined in Paragraph 2.1.1 of Letter Agreement No. 4 hereto.

Option Aircraft – any or all of the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft set forth in Schedule IA that the Buyer has the option to purchase from the Seller subject to the terms and conditions hereof.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

<u>Predelivery Payment</u> – any of the payments determined in accordance with Clause 5.3, Paragraphs 2.1.4.2, 2.2.4.2, 2.3.4.2, 2.4.4.2, 2.5.4.2 and 3.3.2 of Letter Agreement No. 4 hereto.

Purchase Right – as defined in Paragraph 3.1 of Letter Agreement No. 4 hereto.

<u>Purchase Right Aircraft</u> – any or all of the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft that the Buyer has the right to purchase from the Seller subject to the terms and conditions hereof.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Scheduled Delivery Month – as defined in Clause 9.1 or as otherwise provided in this Agreement.

Scheduled Delivery Quarter – as defined in Clause 9.1 or as otherwise provided in this Agreement.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Option Aircraft – any or all of the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft set forth in Schedule IIA that the Buyer has the option to purchase from the Seller subject to the terms and conditions hereof.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

UNQUOTE

#### 2 OPTION AIRCRAFT

2.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft

2.1.1 Option to Firmly Order

The Seller hereby grants the Buyer the option (the "**Option**") to purchase the Option Aircraft subject to the terms and conditions set forth in this Paragraph 2.1.

2.1.2 Delivery

The Seller will have the Option Aircraft Ready for Delivery at the Delivery Location within the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2.1.3 Option Exercise

If the Buyer wishes to exercise any Option, it will do so as follows:

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.1.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and Predelivery Payments

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2.1.4.1 Predelivery Payments

Predelivery Payments for each Option Aircraft will be paid according to the following schedule:

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Except as set forth herein, Clause 5.3 of the Agreement will apply to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2.1.5 Option Lapse
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.2 Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft

2.2.1 Option to Firmly Order Additional Option Aircraft

The Seller hereby grants the Buyer the option (the "Additional Options") to purchase the Additional Option Aircraft subject to the terms and conditions set forth in this Paragraph 2.2.1.

2.2.2 Delivery

The Seller will have the Additional Option Aircraft Ready for Delivery at the Delivery Location within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2.2.3 Additional Options Exercise

If the Buyer wishes to exercise any of the Additional Options, it will do so as follows:

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.2.4 Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Predelivery Payments
- 2.2.4.1 Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2.2.4.2 Predelivery Payments

Predelivery Payments for each Additional Option Aircraft will be paid according to the following schedule:

[CONTINUED ON NEXT PAGE]

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Except as set forth herein, Clause 5.3 of the Agreement will apply to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2.2.5 Additional Option Lapse

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft
- 2.3.1 Option to Firmly Order

The Seller hereby grants the Buyer the option (the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to purchase the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft subject to the terms and conditions set forth in this Paragraph 2.3.

2.3.2 Delivery

The Seller will have the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft Ready for Delivery at the Delivery Location within the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2.3.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Exercise

If the Buyer wishes to exercise any [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] it will do so as follows:

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]Predelivery Payments
- 2.3.4.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3.4.2 Predelivery Payments

Predelivery Payments for each [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft will be paid according to the following schedule:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Except as set forth herein, Clause 5.3 of the Agreement will apply to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2.3.5 Option Lapse
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4 Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft
- 2.4.1 Option to Firmly Order Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft

The Seller hereby grants the Buyer the option (the "Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option") to purchase the Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft subject to the terms and conditions set forth in this Paragraph 2.4.

- 2.4.2 Delivery
  - The Seller will have the Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft Ready for Delivery at the Delivery Location within the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.3 Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Exercise

If the Buyer wishes to exercise the Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option, it will do so as follows:

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2.4.4 Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Predelivery Payments
- 2.4.4.1 Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.4.2 Predelivery Payments

Predelivery Payments for each Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft will be paid according to the following schedule:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- Except as set forth herein, Clause 5.3 of the Agreement will apply to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.5 Additional [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Lapse
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.5 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft
- 2.5.1 Option to Firmly Order

The Seller hereby grants the Buyer the option (the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to purchase the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft subject to the terms and conditions set forth in this Paragraph 2.5.

- 2.5.2 Delivery
  - The Seller will have the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft Ready for Delivery at the Delivery Location within the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.5.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Exercise

If the Buyer wishes to exercise any [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option, it will do so as follows:

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.5.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and Predelivery Payments
- 2.5.4.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.5.4.2 Predelivery Payments

Predelivery Payments for each [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft will be paid according to the following schedule:

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Except as set forth herein, Clause 5.3 of the Agreement will apply to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2.5.5 Option Lapse

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 3 PURCHASE RIGHT AIRCRAFT

- 3.1 The Seller hereby grants the Buyer the right (the "**Purchase Right**") to purchase the Purchase Right Aircraft subject to the terms and conditions set forth in this Paragraph 3.
- 3.2 Purchase Right Exercise

The Buyer will exercise its Purchase Right as follows:

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 3.3 Purchase Right Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Predelivery Payments
- 3.3.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.2 Predelivery Payments for each Purchase Right Aircraft will be paid according to the following schedule:

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Except as set forth herein, Clause 5.3 of the Agreement will apply to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.3.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 4 GENERAL PROVISIONS APPLICABLE TO THIS LETTER AGREEMENT

4.1 The Base Price of the Propulsion System for any Option Aircraft, Additional Option Aircraft, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft, Additional [CONFIDENTIAL PORTION OMITTED AND FILED

SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Option Aircraft and Purchase Right Aircraft that is firmly ordered hereunder will be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

4.2 Notwithstanding any provision of the Agreement or this Letter Agreement, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 5 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

# 6 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

## 7 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the Seller.	
	Very truly yours,
	AIRBUS S.A.S.
	Ву:
	Its:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By: Its:	

C	CH	СТ	71	TT	$\mathbf{r}$	T	Λ

OPTION AIRCRAFT DELIVERY SCHEDULE

SCHEDULE IB

ADDITIONAL OPTION AIRCRAFT DELIVERY SCHEDULE

# SCHEDULE IIA

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] OPTION AIRCRAFT DELIVERY SCHEDULE

SCHEDULE IIB

ADDITIONAL [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] OPTION AIRCRAFT DELIVERY SCHEDULE

SCHEDULE III

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY	WITH THE COMMISSION PURSUANT	TO A REQUEST FOR CONFIDENTIAL
TREATMENT] OPTION AIRCRAFT DELIVERY SCHEDULE		

## LETTER AGREEMENT NO. 5

As of July 20, 2011

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: FLEXIBILITY

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 5 (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement constitutes an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement is governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

## 1 DEFINITIONS

Clause 0 to the Agreement is amended to either modify or add the following defined terms between the words "QUOTE" and "UNQUOTE": QUOTE

<u>Leased A319 Aircraft</u> – any or all of the Leased Aircraft that are A319 Aircraft.

Purchase A319 NEO Aircraft – any or all of the Purchase Aircraft that are A319 NEO Aircraft.

UNQUOTE

# 2 TYPE FLEXIBILITY

# 2.1 Conversion Rights

- 2.1.1 The Seller grants the Buyer a one-time right to (A) convert A320 Family Aircraft to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] type (with respect to each such Aircraft, a "Conversion Right") and (B) convert A320 Family NEO Aircraft to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (with respect to each such Aircraft, a "NEO Conversion Right"), subject to the following limitations:
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (vi) The NEO Conversion Right will not apply to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

(vii) In connection with the Buyer's exercise of the Conversion Right or the NEO Conversion Right with respect to any Aircraft, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 2.1.2 Conversion Right Exercise

The Buyer will exercise its Conversion Right or NEO Conversion Right subject to the following terms and conditions:

- (i) The Buyer must give written notice to the Seller of its desire to exercise any Conversion Right or NEO Conversion Right no later than: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Each such notice received from the Buyer will specify the type of Aircraft selected by the Buyer.
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 3 RESCHEDULING RIGHTS

- 3.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (vi) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Exercise

The [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will exercise its [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] subject to the following terms and conditions:

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] A320 NEO FAMILY AIRCRAFT

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

# 5 GENERAL PROVISIONS APPLICABLE TO THIS LETTER AGREEMENT

- 5.1 Nothing contained in this Letter Agreement will oblige the Seller to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5.2 Notwithstanding any provision of the Agreement or this Letter Agreement, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 6 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

# 7 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

## 8 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please ex Seller.	ecute the original and one (1) copy hereof in the space provided below and return a copy to the
	Very truly yours,
	AIRBUS S.A.S.
	Ву:
	Its:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
Ву:	
Its:	

SCHEDULE I

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an Airbus A320 Family Aircraft Purchase Agreement dated as of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement"), which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 6 (this "Letter Agreement") certain additional terms and conditions regarding the sale or lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement will have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

# 1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 1.1 Defined Terms

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]r.

"Seller Party" means the Seller or any Affiliate of the Seller.

# 1.2 Commitment to the Buyer

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2 EXCLUSIONS

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 3 ADMINISTRATION

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 4 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

## 5 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

## 6 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument.

Seller.	
	Very truly yours,
	AIRBUS S.A.S.
	By:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By:	

If the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: MISCELLANEOUS

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an Airbus A320 Family Aircraft Purchase Agreement dated as of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement"), which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 9 (this "Letter Agreement") certain additional terms and conditions regarding the sale or lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement will have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

# 1 DEFINITIONS

Clause 0 of the Agreement is amended it add the following defined term between "QUOTE" and "UNQUOTE".

OUOTE

<u>Leasing Documentation</u> – as defined in Letter Agreement No. 1.

UNQUOTE

## 2 CERTIFICATIONS

Clause 7 of the Agreement is deleted in its entirety and is replaced with Clause 7 attached hereto as Appendix 1.

# 3 DELIVERY

Clause 9 of the Agreement is deleted in its entirety and is replaced with Clause 9 attached hereto as Appendix 2.

## 4 EXCUSABLE DELAY

Clause 10 of the Agreement is deleted in its entirety and is replaced with Clause 10 attached hereto as Appendix 3.

# 5 INEXCUSABLE DELAY

Clause 11 of the Agreement is deleted in its entirety and is replaced with Clause 11 attached hereto as Appendix 4.

# 6 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

# 7 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

# 8 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the Seller.	
	Very truly yours,
	AIRBUS S.A.S.
	Ву:
	Its:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By:	
Its:	

## 7 CERTIFICATION

Except as set forth in this Clause 7, the Seller will not be required to obtain any certificate or approval with respect to any Aircraft. If the use of any certificate identified in this Agreement is discontinued during the performance of this Agreement, thereafter reference to such discontinued certificate will be deemed to be a reference to any other certificate or instrument issued by the applicable Aviation Authority which corresponds to such discontinued certificate.

7.1 Type Certification

The Aircraft have been type certificated under EASA procedures for joint certification in the transport category. The Seller will obtain or cause to be obtained an FAA type certificate (the "Type Certificate") to allow the issuance of the Export Certificate of Airworthiness for the Aircraft. The Seller will obtain or cause to be obtained any additional certificates for the Aircraft that, under the requirements of the FAA generally applicable to aircraft manufacturers, the Seller is required to obtain to permit operation of aircraft of the same type as the Aircraft.

7.2 Export Certificate of Airworthiness

The Seller will cause each Aircraft to be delivered to the Buyer with an Export Certificate of Airworthiness without restriction or limitation and in a condition enabling the Buyer to obtain at the time of Delivery a Standard Airworthiness Certificate issued pursuant to Part 21 of the US Federal Aviation Regulations and a Certificate of Sanitary Construction issued by the U.S. Public Health Service of the Food and Drug Administration. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 7.3 Specification Changes before Aircraft Ready for Delivery
- 7.3.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3.2 The Seller will, as far as is practicable, take into account the information available to it concerning any proposed Change in Law in order to minimize the costs of changes or modifications to the Specification as a result of such Change in Law that occurs before the applicable Aircraft is Ready for Delivery.
- 7.3.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.4 Specification Changes after Aircraft Ready For Delivery

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]		

## 9 DELIVERY

- 9.1 Delivery Schedule
  - Subject to Clauses 2, 7, 8, 10 and 18, the Seller will have the Aircraft Ready for Delivery at the Delivery Location within the months (each, a "Scheduled Delivery Month") or Quarters (each, a "Scheduled Delivery Quarter"), as applicable, set forth on Schedule I hereto (the "Delivery Schedule").
- 9.1.1 In respect of each Aircraft for which a Scheduled Delivery Quarter is set forth in Schedule I, the Seller will communicate to the Buyer the Scheduled Delivery Month no later than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REOUEST FOR CONFIDENTIAL TREATMENT]
- 9.1.2 The Seller will give the Buyer (i) at least [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 9.2 Delivery Process
- 9.2.1 The Buyer will send its representatives (which, with respect to Leased Aircraft, may include representatives of the Lessor) to the Delivery Location to take Delivery of each Aircraft at the date on which such Aircraft is Ready for Delivery, and fly such Aircraft from the Delivery Location.
- 9.2.2 The Seller will deliver and transfer title to each Aircraft to the Buyer (or, with respect to a Leased Aircraft, the Lessor) free and clear of all liens, claims, charges and encumbrances of every kind whatsoever (except for any liens or encumbrances created by or on behalf of the Buyer) provided that the Balance of the Final Price of such Aircraft has been paid by the Buyer pursuant to Clause 5.4 (or, with respect to a Leased Aircraft, provided that the Buyer has executed and delivered the Leasing Documentation for such Leased Aircraft to which the Buyer is party, upon which the Seller will cause the Lessor and each other counterparty to such Leasing Documentation to execute and deliver such Leasing Documentation to the Buyer) and that the Certificate of Acceptance has been executed and delivered to the Seller pursuant to Clause 8.3. The Seller will provide the Buyer (or with respect to Leased Aircraft, the Lessor) with (i) a warranty bill of sale in the form of Exhibit E (the "Warranty Bill of Sale"), (ii) FAA Form 8050-2 (the "Aircraft Bill of Sale") and (iii) such other documentation confirming transfer of title and receipt of such Final Price of such Aircraft as may reasonably be requested by the Buyer (and, with respect to Leased Aircraft, by the Lessor). Title to and risk of loss of or damage to such Aircraft will pass to the Buyer (or, with respect to Leased Aircraft, the Lessor) contemporaneously with the delivery by the Seller to the Buyer of such Warranty Bill of Sale.
- 9.2.3 (i) If the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 9.3 Flyaway
- 9.3.1 The Buyer and the Seller will cooperate to obtain any licenses that may be required by the Aviation Authority of the Delivery Location for the purpose of exporting the Aircraft.
- 9.3.2 In the event that the Buyer is unable to export the Aircraft on the date Delivery occurs due to a delay in receiving temporary FAA approval to fly the Aircraft to the United States (the "fly-wire"), the Seller will park and store such Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 9.3.3 All expenses of, or connected with, flying the Aircraft from the Delivery Location after Delivery will be borne by the Buyer, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 9.4 Delegation of Authority

With respect to each Aircraft, the Seller will present to the Buyer prior to Delivery of such Aircraft a certified copy of the powers of attorney designating and authorizing certain persons to act on its behalf in connection with the Delivery of such Aircraft, including the person executing the transfer of title documents for such Aircraft.

With respect to each Aircraft, the Buyer will present to the Seller prior to Delivery of such Aircraft a certified copy of the applicable delegation of authority of the Buyer designating and authorizing certain persons to act on its behalf in connection with the Delivery of such Aircraft, including (in the case of Purchase Aircraft) the person executing the transfer of title documents for such Aircraft.

- 9.5 International Registry
- 9.5.1 The Seller will consent to the contract of sale registrations with the International Registry of Mobile Assets ("**International Registry**") at or promptly following the Delivery of each Aircraft, consistent with the Seller's customary practices.
- 9.5.2 The Seller will assure that the Airframe is populated in the drop down boxes with the International Registry prior to the Delivery of the applicable Aircraft. Prior to the Delivery Date for each Aircraft, the Seller will provide to the Buyer's FAA counsel (which will be Daugherty, Fowler, Peregrin, Haught & Jenson or such other counsel designated by the Buyer) an original undated executed copy of the Aircraft Bill of Sale for such Aircraft.

## 10 EXCUSABLE DELAY AND TOTAL LOSS

10.1 Scope of Excusable Delay

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

10.2 Consequences of Excusable Delay

If an Excusable Delay occurs with respect to any Aircraft:

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] notify [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] in writing of such Excusable Delay as soon as practicable after becoming aware of the same;
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) the Seller will as soon as practicable after the removal of the cause of such delay resume performance of its obligations under this Agreement and in particular will notify the Buyer of the revised Scheduled Delivery Month or Scheduled Delivery Quarter, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.3 Termination on Excusable Delay
- 10.3.1 If any Delivery is delayed as a result of an Excusable Delay for a period of more than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.3.2 If any Delivery is delayed as a result of an Excusable Delay for a period of more than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.3.3 If the Seller advises the Buyer in its notice of a revised Scheduled Delivery Month or Scheduled Delivery Quarter pursuant to Clause 10.2(iv) that [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 10.3.4 If this Agreement is not terminated with respect to any Aircraft under the terms of Clause 10.3.1, 10.3.2 or 10.3.3, then the Seller will be entitled to reschedule Delivery. The Seller will notify the Buyer of the new Scheduled Delivery Month after the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] period referred to in Clause 10.3.1, 10.3.2 or 10.3.3, and this new Scheduled Delivery Month will be deemed to be an amendment to the applicable Scheduled Delivery Month or Scheduled Delivery Quarter in Schedule I.
- 10.4 Total Loss, Destruction or Damage

If, prior to Delivery, any Aircraft is lost, destroyed or in the reasonable opinion of the Seller is damaged beyond economic repair ("Total Loss"), the Seller will notify the Buyer in writing to this effect [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] The Seller will include in such notification (or as soon after the delivery of the notice as such information becomes available to the Seller) the earliest date consistent with the Seller's other commitments and production capabilities that an aircraft to replace such Aircraft may be delivered to the Buyer and the Scheduled Delivery Month or Scheduled Delivery Quarter will be extended as specified in the Seller's notice to accommodate the delivery of the replacement aircraft; [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Nothing herein will require the Seller to manufacture and deliver a replacement aircraft if such manufacture would require the reactivation of its production line for the model or series of aircraft that includes the applicable Aircraft.

10.5 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Notwithstanding the provisions of Clauses 3.2, 4.1 and 4.2, if any Excusable Delay or Total Loss occurs with respect to any Aircraft, the Buyer will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

10.6 Predelivery Payments

If an Excusable Delay or Total Loss occurs with respect to any Aircraft, the payment schedule for unpaid Predelivery Payments with respect to such Aircraft will be adjusted to reflect the revised Scheduled Delivery Month or Scheduled Delivery Quarter that the Seller notifies the Buyer.

In the event this Agreement is terminated with respect to any Aircraft under the terms of Clause 10.3 or 10.4, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 10.7 Termination Rights Exclusive

If this Agreement is terminated as provided for under the terms of Clauses 10.3 or 10.4, such termination will discharge all obligations and liabilities of the parties hereunder with respect to such affected Aircraft and undelivered material, services, data or other items applicable thereto and to be furnished under the Agreement, except as provided in this Clause 10 and Clause 18.5.

## 10.8 Remedies

EXCEPT AS PROVIDED IN LETTER AGREEMENT NO. 17 TO THE AGREEMENT, THIS CLAUSE 10 SETS FORTH THE SOLE AND EXCLUSIVE REMEDY OF THE BUYER FOR DELAYS IN DELIVERY OR FAILURE TO DELIVER, OTHER THAN SUCH DELAYS AS ARE COVERED BY CLAUSE 11, AND THE BUYER HEREBY WAIVES ALL RIGHTS TO WHICH IT WOULD OTHERWISE BE ENTITLED IN RESPECT THEREOF, INCLUDING, WITHOUT LIMITATION, ANY RIGHTS TO INCIDENTAL AND CONSEQUENTIAL DAMAGES OR SPECIFIC PERFORMANCE. THE BUYER WILL NOT BE ENTITLED TO CLAIM THE REMEDIES AND RECEIVE THE BENEFITS PROVIDED IN THIS CLAUSE 10 TO THE EXTENT THE DELAY REFERRED TO IN THIS CLAUSE 10 IS CAUSED BY THE ACTION OR INACTION OF THE BUYER OR ITS REPRESENTATIVES.

#### 11 INEXCUSABLE DELAY

11.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Should an Aircraft not be Ready for Delivery within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (A) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (B) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (C) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (D) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (E) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (F) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

If an Inexcusable Delay occurs with respect to any Aircraft:

- (i) the Seller will notify the Buyer in writing of such Inexcusable Delay as soon as practicable after becoming aware of the same; and
- (ii) the Seller will, as soon as is practicable after the removal of the cause of such delay, resume performance of its obligations under this Agreement and in particular will notify the Buyer of the revised Scheduled Delivery Month or Scheduled Delivery Quarter, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 11.2 Renegotiation

If, as a result of an Inexcusable Delay, the Delivery does not occur within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

11.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Notwithstanding the provisions of Clause [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] if an Inexcusable Delay occurs with respect to any Aircraft, the Buyer will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 11.4 Predelivery Payments

If an Inexcusable Delay occurs with respect to any Aircraft the payment schedule for unpaid Predelivery Payments with respect to such Aircraft will be adjusted to reflect the revised Scheduled Delivery Month or Scheduled Delivery Quarter notified to the Buyer.

In the event this Agreement is terminated with respect to any Aircraft under the terms of Clause 11.5.1 or 11.5.2, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 11.5 Termination
- 11.5.1 If, as a result of an Inexcusable Delay, the Delivery does not occur within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 11.5.2 If, as a result of an Inexcusable Delay, the Delivery does not occur within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 11.5.3 In the event of termination, neither party will have any claim against the other with respect to such Aircraft, except as provided in this Clause 11 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and Clause 18.5.

#### 11.6 Remedies

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] THIS CLAUSE 11 SETS FORTH THE SOLE AND EXCLUSIVE REMEDY OF THE BUYER FOR DELAYS IN DELIVERY OR FAILURE TO DELIVER, OTHER THAN SUCH DELAYS AS ARE COVERED BY CLAUSE 10, AND THE BUYER HEREBY WAIVES ALL RIGHTS TO WHICH IT WOULD OTHERWISE BE ENTITLED IN RESPECT THEREOF, INCLUDING WITHOUT LIMITATION ANY RIGHTS TO INCIDENTAL AND CONSEQUENTIAL DAMAGES OR SPECIFIC PERFORMANCE. THE BUYER WILL NOT BE ENTITLED TO CLAIM THE REMEDIES AND RECEIVE THE BENEFITS PROVIDED IN THIS CLAUSE 11 TO THE EXTENT THE DELAY REFERRED TO IN THIS CLAUSE 11 IS CAUSED BY THE NEGLIGENCE OR FAULT OF THE BUYER OR ITS REPRESENTATIVES.

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: SUPPORT MATTERS

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement dated as of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement"), which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 10 (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement will have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

# 1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] IN-SERVICE ISSUES

In the event that, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

the remedies set forth in Clauses 12.1.4.1 and 12.1.4.2 of the Agreement for such Aircraft will apply [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] In addition, in the event that [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] the Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Buyer.

# 2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] SUPPORT

The Seller will provide to the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] WARRANTY

In the case of Warranted Parts [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 4 COMPUTER BASED TRAINING [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

(v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 5 SELLER PARTS [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5.1 For each Aircraft delivered pursuant to the Agreement, the Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 5.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6 FLIGHT CREW TRAINING <u>[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]</u>
- Pursuant to Paragraph 1.1 of Appendix A to Clause 16 of the Agreement, the Seller will provide flight crew training (standard transition course)
  [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR
  CONFIDENTIAL TREATMENT]
- 6.2 If the Buyer [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.3 The Flight Crew [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] in accordance with Paragraph 7 of Letter Agreement No. 2 to the Agreement.
- 6.4 The Buyer will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 7 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

## 8 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

## 9 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execu Seller.	ute the original and one (1) copy hereof in the space provided below and return a copy to the
	Very truly yours,
	AIRBUS S.A.S.
	Ву:
	Its:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
Ву:	
Its:	

#### LETTER AGREEMENT NO. 11A

As of July 20, 2011

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: A319 AIRCRAFT PERFORMANCE GUARANTEE – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 11A (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

#### 1 AIRCRAFT CONFIGURATION

The guarantees defined in Paragraphs 2, 3, 4 and 5 below (the "Guarantees") are applicable to the A319 Aircraft as described in the A319 Standard Specification [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] for:

- 1.1 Basic Configuration (not applicable to Paragraph 3.2)
  - i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - iii) the following design weights:
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 1.2 Optional Configuration

- i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iii) the following design weights:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

hereinafter referred to as the "Specification" without taking into account any further changes thereto as provided in the Agreement.

## 2 GUARANTEED PERFORMANCE

2.1 Take-Off

#### 2.1.1 Take-Off Field Length

The FAR approved Take-Off Field Length at an A319 Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at the start of ground run at sea level pressure altitude in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2.1.2 Take-Off Weight

When the A319 Aircraft is operated in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

the FAR approved Take-Off Weight of the A319 Aircraft at the start of ground run shall be not less than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2.2 Landing Field Length

The FAR approved dry Landing Field Length at an A319 Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at sea level pressure altitude shall be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 3 MISSION GUARANTEES

3.1 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A319 Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.1.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT])
- 3.1.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.1.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A319 Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.2.1 The departure airport conditions [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] are as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.2.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.3	Mission Block Fuel
	In carrying a fixed payload of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the A319 Aircraft will be not more than
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	when the A319 Aircraft is operated as defined here below.
3.3.1	The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.3.2	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.3.3	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.3.4	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.3.5	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.3.6	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.3.7	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.3.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4 The Mission Payload and Block Fuel guarantees are based on [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (see Appendix A).

#### 4 MANUFACTURER'S WEIGHT EMPTY GUARANTEE

The Seller guarantees a Manufacturer's Weight Empty of the A319 Aircraft as defined in Paragraph 1 above of not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

This is the Manufacturer's Weight Empty of the A319 Aircraft as defined in Section 13-10 of the A319 Standard Specification as amended by the SCN's defined in Paragraph 1 above [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as defined in the Paragraph 8 below.

### 5 NOISE GUARANTEE

5.1 Noise Certification

The A319 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise

The Seller shall provide the Buyer with a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

5.3 Interior Noise in Flight

#### 5.3.1 Cockpit

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5.3.2 Cabin

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5.4 Ramp Noise

The APU noise during ground operation of the A319 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 6 GUARANTEE CONDITIONS

6.1 The certification requirements for the A319 Aircraft, except where otherwise noted, will be as stated in Section 02 of the Specification.

- 6.2 For the determination of FAR take-off and landing performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - When establishing take-off performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - Cruise performance at [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.4 Where applicable the Guarantees assume the use of an approved fuel having a density of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.5 Sound levels are to be specified in decibels (dB), at a reference pressure of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - The definition of dB(A) shall be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - The Speech Interference Level (SIL) is defined as the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 7 GUARANTEE COMPLIANCE

- 7.1 Compliance with the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2 Compliance with the take-off and landing elements of the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3 Compliance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 7.4 Compliance with the Manufacturer's Weight Empty guarantee will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.5 The [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be used to demonstrate compliance with the guarantees of certification noise levels.
  - The A319 Aircraft will have a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- Data [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be adjusted as required [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] accordance with established aeronautical practices to show compliance with the Guarantees.
- 7.7 The Seller undertakes to furnish the Buyer with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] demonstrating compliance with the Guarantees [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 8 ADJUSTMENT OF GUARANTEES

- 8.1 In the event of any change to any law, governmental regulation or requirement or interpretation thereof ("**Rule Change**") by any governmental agency made subsequent to the date of the Agreement and such Rule Change affects the A319 Aircraft configuration or performance or both required to obtain certification the Guarantees will be appropriately modified to reflect the effect of any such change.
- 8.2 The Guarantees apply to the A319 Aircraft as described in Paragraph 1 above and may be adjusted in the event of:
  - a) Any further configuration change which is the subject of an SCN
  - b) Changes required to obtain certification which cause changes to the performance or weight of the A319 Aircraft.

#### 9 EXCLUSIVE GUARANTEES

The Guarantees are exclusive and are provided in lieu of any and all other performance and weight guarantees of any nature which may be stated, referenced or incorporated in the Specification or any other document.

### 10 UNDERTAKING REMEDIES

Should the A319 Aircraft fail to meet any of the Guarantees specified in this Letter Agreement the Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 10.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.3 The Seller's maximum liability in respect of deficiency in performance of any A319 Aircraft will be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 11 DUPLICATE REMEDIES

Except as provided in Paragraph 10, the remedies provided to the Buyer under this Guarantee are not cumulative of any other remedies provided to the Buyer under any other warranty or guarantee contained in the Agreement and the Buyer will not be entitled to duplicate remedies with respect to any single defect or costs incurred for any single defect.

#### 12 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

### 13 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

### 14 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execu Seller.	ate the original and one (1) copy hereof in the space provided below and return a copy to the
	Very truly yours,
	AIRBUS S.A.S.
	Ву:
	Its:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By:	
Its:	

## APPENDIX A TO LETTER AGREEMENT NO. 11A

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: A319 AIRCRAFT PERFORMANCE GUARANTEE – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 11B (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

#### 1 AIRCRAFT CONFIGURATION

The guarantees defined in Paragraphs 2, 3, 4 and 5 below (the "Guarantees") are applicable to the A319 Aircraft as described in the A319 Standard Specification [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] for:

- 1.1 Basic Configuration (not applicable to Paragraph 3.2)
  - i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - iii) the following design weights:
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 1.2 Optional Configuration

- i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iii) the following design weights:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

hereinafter referred to as the "Specification" without taking into account any further changes thereto as provided in the Agreement.

#### 2 GUARANTEED PERFORMANCE

#### 2.1 Take-Off

## 2.1.1 Take-Off Field Length

The FAR approved Take-Off Field Length at an A319 Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at the start of ground run at sea level pressure altitude in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2.1.2 Take-Off Weight

When the A319 Aircraft is operated in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

the FAR approved Take-Off Weight of the A319 Aircraft at the start of ground run shall be not less than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2.2 Landing Field Length

The FAR approved dry Landing Field Length at an A319 Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at sea level pressure altitude shall be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

-	
3	MISSION GUARANTEES
3.1	Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	The A319 Aircraft will be capable of carrying a Payload of not less than
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.1	The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	The destination airport conditions are such as to allow the required Landing Weight to be used without restriction [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.2	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.3	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.4	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.5	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.6	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR

3.1.7

CONFIDENTIAL TREATMENT]

- 3.1.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A319 Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.2.1 The departure airport conditions [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] are as follows:

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.2.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 3.3 Mission Block Fuel

In carrying a fixed payload of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the A319 Aircraft will be not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

when the A319 Aircraft is operated as defined here below.

- 3.3.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

  The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- The Mission Payload and Block Fuel guarantees are based on [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (see Appendix A).

## 4 MANUFACTURER'S WEIGHT EMPTY GUARANTEE

The Seller guarantees a Manufacturer's Weight Empty of the A319 Aircraft as defined in Paragraph 1 above of not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

This is the Manufacturer's Weight Empty of the A319 Aircraft as defined in Section 13-10 of the A319 Standard Specification as amended by the SCN's defined in Paragraph 1 above [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as defined in the Paragraph 8 below.

#### 5 NOISE GUARANTEE

5.1 Noise Certification

The A319 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Noise

The Seller shall provide the Buyer with a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5.3 Interior Noise in Flight

### 5.3.1 Cockpit

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 5.3.2 Cabir

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 5.4 Ramp Noise

The APU noise during ground operation of the A319 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### **6 GUARANTEE CONDITIONS**

- 6.1 The certification requirements for the A319 Aircraft, except where otherwise noted, will be as stated in Section 02 of the Specification.
- For the determination of FAR take-off and landing performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- When establishing take-off performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - Cruise performance at [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.4 Where applicable the Guarantees assume the use of an approved fuel having a density of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.5 Sound levels are to be specified in decibels (dB), at a reference pressure of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - The definition of dB(A) shall be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - The Speech Interference Level (SIL) is defined as the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 7 GUARANTEE COMPLIANCE

- 7.1 Compliance with the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2 Compliance with the take-off and landing elements of the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3 Compliance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.4 Compliance with the Manufacturer's Weight Empty guarantee will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

7.5 The [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be used to demonstrate compliance with the guarantees of certification noise levels.

The A319 Aircraft will have a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 7.6 Data [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be adjusted as required [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] accordance with established aeronautical practices to show compliance with the Guarantees.
- 7.7 The Seller undertakes to furnish the Buyer with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] demonstrating compliance with the Guarantees [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 8 ADJUSTMENT OF GUARANTEES

- 8.1 In the event of any change to any law, governmental regulation or requirement or interpretation thereof ("**Rule Change**") by any governmental agency made subsequent to the date of the Agreement and such Rule Change affects the A319 Aircraft configuration or performance or both required to obtain certification the Guarantees will be appropriately modified to reflect the effect of any such change.
- 8.2 The Guarantees apply to the A319 Aircraft as described in Paragraph 1 above and may be adjusted in the event of:
  - a) Any further configuration change which is the subject of an SCN
  - b) Changes required to obtain certification which cause changes to the performance or weight of the A319 Aircraft.

## 9 EXCLUSIVE GUARANTEES

The Guarantees are exclusive and are provided in lieu of any and all other performance and weight guarantees of any nature which may be stated, referenced or incorporated in the Specification or any other document.

#### 10 UNDERTAKING REMEDIES

Should the A319 Aircraft fail to meet any of the Guarantees specified in this Letter Agreement the Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 10.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.3 The Seller's maximum liability in respect of deficiency in performance of any A319 Aircraft will be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 11 DUPLICATE REMEDIES

Except as provided in Paragraph 10, the remedies provided to the Buyer under this Guarantee are not cumulative of any other remedies provided to the Buyer under any other warranty or guarantee contained in the Agreement and the Buyer will not be entitled to duplicate remedies with respect to any single defect or costs incurred for any single defect.

### 12 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

## 13 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

## 14 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

f the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the eller.		
	Very truly yours,	
	AIRBUS S.A.S.	
	By:	
	Its:	
Accepted and Agreed		
AMERICAN AIRLINES, INC.		
By:		
Its:		

# APPENDIX A TO LETTER AGREEMENT NO. 11B

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

### LETTER AGREEMENT NO. 11C

As of July 20, 2011

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

**Re: A319 NEO AIRCRAFT PERFORMANCE GUARANTEE** – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 11C (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

### 1 AIRCRAFT CONFIGURATION

The guarantees defined in Paragraphs 2, 3, 4 and 5 below (the "Guarantees") are applicable to the A319 NEO Aircraft as described in the A319 Standard Specification [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] for:

- 1.1 Basic Configuration (not applicable to Paragraph 3.2)
  - i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - iii) the following design weights:
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 1.2 Optional Configuration

- i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iii) the following design weights:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

hereinafter referred to as the "Specification" without taking into account any further changes thereto as provided in the Agreement.

#### 2 GUARANTEED PERFORMANCE

### 2.1 Take-Off

# 2.1.1 Take-Off Field Length

The FAR approved Take-Off Field Length at an A319 NEO Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at the start of ground run at sea level pressure altitude in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2.1.2 Take-Off Weight

When the A319 NEO Aircraft is operated in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

the FAR approved Take-Off Weight of the A319 NEO Aircraft at the start of ground run shall be not less than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2.2 Landing Field Length

The FAR approved dry Landing Field Length at an A319 NEO Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at sea level pressure altitude shall be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3	MISSION GUARANTEES
3.1	Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	The A319 NEO Aircraft will be capable of carrying a Payload of not less than
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.1	The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.2	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.3	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.4	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.5	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.1.6	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR

3.1.7

CONFIDENTIAL TREATMENT]

- 3.1.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A319 NEO Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.2.1 The departure airport conditions [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] are as follows:

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.2.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 3.3 Mission Block Fuel

In carrying a fixed payload of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the A319 NEO Aircraft will be not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

when the A319 NEO Aircraft is operated as defined here below.

- 3.3.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

  The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4 The Mission Payload and Block Fuel guarantees are based on [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (see Appendix A).

## 4 MANUFACTURER'S WEIGHT EMPTY GUARANTEE

 $The \ Seller \ guarantees \ a \ Manufacturer's \ Weight \ Empty \ of \ the \ A319 \ NEO \ Aircraft \ as \ defined \ in \ Paragraph \ 1 \ above \ of \ not \ more \ than$ 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

This is the Manufacturer's Weight Empty of the A319 NEO Aircraft as defined in Section 13-10 of the A319 Standard Specification as amended by the SCN's defined in Paragraph 1 above [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as defined in the Paragraph 8 below.

## 5 NOISE GUARANTEE

5.1 Noise Certification

The A319 NEO Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Noise

The Seller shall provide the Buyer with a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 5.3 Interior Noise in Flight

### 5.3.1 Cockpit

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 5.3.2 Cabir

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 5.4 Ramp Noise

The APU noise during ground operation of the A319 NEO Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### **6 GUARANTEE CONDITIONS**

- 6.1 The certification requirements for the A319 NEO Aircraft, except where otherwise noted, will be as stated in Section 02 of the Specification.
- 6.2 For the determination of FAR take-off and landing performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- When establishing take-off performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.4 Where applicable the Guarantees assume the use of an approved fuel having a density of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.5 Sound levels are to be specified in decibels (dB), at a reference pressure of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The definition of dB(A) shall be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The Speech Interference Level (SIL) is defined as the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 7 GUARANTEE COMPLIANCE

- 7.1 Compliance with the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2 Compliance with the take-off and landing elements of the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3 Compliance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.4 Compliance with the Manufacturer's Weight Empty guarantee will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.5 The [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be used to demonstrate compliance with the guarantees of certification noise levels.

The A319 NEO Aircraft will have a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 7.6 Data [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be adjusted as required [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] accordance with established aeronautical practices to show compliance with the Guarantees.
- 7.7 The Seller undertakes to furnish the Buyer with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] demonstrating compliance with the Guarantees [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 8 ADJUSTMENT OF GUARANTEES

- 8.1 In the event of any change to any law, governmental regulation or requirement or interpretation thereof ("**Rule Change**") by any governmental agency made subsequent to the date of the Agreement and such Rule Change affects the A319 NEO Aircraft configuration or performance or both required to obtain certification the Guarantees will be appropriately modified to reflect the effect of any such change.
- 8.2 The Guarantees apply to the A319 NEO Aircraft as described in Paragraph 1 above and may be adjusted in the event of:
  - a) Any further configuration change which is the subject of an SCN
  - b) Changes required to obtain certification which cause changes to the performance or weight of the A319 NEO Aircraft.

### 9 EXCLUSIVE GUARANTEES

The Guarantees are exclusive and are provided in lieu of any and all other performance and weight guarantees of any nature which may be stated, referenced or incorporated in the Specification or any other document.

#### 10 UNDERTAKING REMEDIES

Should the A319 NEO Aircraft fail to meet any of the Guarantees specified in this Letter Agreement the Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

- 10.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.3 The Seller's maximum liability in respect of deficiency in performance of any A319 NEO Aircraft will be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 11 DUPLICATE REMEDIES

Except as provided in Paragraph 10, the remedies provided to the Buyer under this Guarantee are not cumulative of any other remedies provided to the Buyer under any other warranty or guarantee contained in the Agreement and the Buyer will not be entitled to duplicate remedies with respect to any single defect or costs incurred for any single defect.

## 12 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

## 13 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

## 14 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the Seller.		
	Very truly yours,	
	AIRBUS S.A.S.	
	Ву:	
	Its:	
Accepted and Agreed		
AMERICAN AIRLINES, INC.		
By: Its:		

# APPENDIX A TO LETTER AGREEMENT NO. 11C

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### LETTER AGREEMENT NO. 11D

As of July 20, 2011

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

**Re: A319 NEO AIRCRAFT PERFORMANCE GUARANTEE** – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 11D (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

### 1 AIRCRAFT CONFIGURATION

The guarantees defined in Paragraphs 2, 3, 4 and 5 below (the "Guarantees") are applicable to the A319 NEO Aircraft as described in the A319 Standard Specification [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REOUEST FOR CONFIDENTIAL TREATMENT] for:

- 1.1 Basic Configuration (not applicable to paragraph 3.2)
  - i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - iii) the following design weights:
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 1.2 Optional Configuration

- i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iii) the following design weights:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

hereinafter referred to as the "Specification" without taking into account any further changes thereto as provided in the Agreement.

#### 2 GUARANTEED PERFORMANCE

### 2.1 Take-Off

# 2.1.1 Take-Off Field Length

The FAR approved Take-Off Field Length at an A319 NEO Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at the start of ground run at sea level pressure altitude in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2.1.2 Take-Off Weight

When the A319 NEO Aircraft is operated in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

the FAR approved Take-Off Weight of the A319 NEO Aircraft at the start of ground run shall be not less than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2.2 Landing Field Length

The FAR approved dry Landing Field Length at an A319 NEO Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at sea level pressure altitude shall be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 3 MISSION GUARANTEES Mission Pavload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST 3.1 FOR CONFIDENTIAL TREATMENT] The A319 NEO Aircraft will be capable of carrying a Payload of not less than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION 3.1.1 OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] An allowance of ICONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST 3.1.2 FOR CONFIDENTIAL TREATMENT] An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST 3.1.3 FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR 3.1.4 CONFIDENTIAL TREATMENT]. An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST 3.1.5 FOR CONFIDENTIAL TREATMENT] An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST 3.1.6 FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR

3.1.7

CONFIDENTIAL TREATMENT]

- 3.1.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A319 NEO Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.2.1 The departure airport [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] are as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.2.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 3.3 Mission Block Fuel

In carrying a fixed payload of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT of the A319 NEO Aircraft will be not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

when the A319 NEO Aircraft is operated as defined here below.

- 3.3.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

  The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - (CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4 The Mission Payload and Block Fuel guarantees are based on [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (see Appendix A).

## 4 MANUFACTURER'S WEIGHT EMPTY GUARANTEE

The Seller guarantees a Manufacturer's Weight Empty of the A319 NEO Aircraft as defined in Paragraph 1 above of not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

This is the Manufacturer's Weight Empty of the A319 NEO Aircraft as defined in Section 13-10 of the A319 Standard Specification as amended by the SCN's defined in Paragraph 1 above [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as defined in the Paragraph 8 below.

## 5 NOISE GUARANTEE

5.1 Noise Certification

The A319 NEO Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Noise

The Seller shall provide the Buyer with a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 5.3 Interior Noise in Flight

### 5.3.1 Cockpit

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 5.3.2 Cabir

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 5.4 Ramp Noise

The APU noise during ground operation of the A319 NEO Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 6 GUARANTEE CONDITIONS

- 6.1 The certification requirements for the A319 NEO Aircraft, except where otherwise noted, will be as stated in Section 02 of the Specification.
- 6.2 For the determination of FAR take-off and landing performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- When establishing take-off performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.4 Where applicable the Guarantees assume the use of an approved fuel having a density of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.5 Sound levels are to be specified in decibels (dB), at a reference pressure of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

The definition of dB(A) shall be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The Speech Interference Level (SIL) is defined as the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 7 GUARANTEE COMPLIANCE

- 7.1 Compliance with the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2 Compliance with the take-off and landing elements of the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3 Compliance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.4 Compliance with the Manufacturer's Weight Empty guarantee will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 7.5 The [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be used to demonstrate compliance with the guarantees of certification noise levels.

The A319 NEO Aircraft will have a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 7.6 Data [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be adjusted as required [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] accordance with established aeronautical practices to show compliance with the Guarantees.
- 7.7 The Seller undertakes to furnish the Buyer with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] demonstrating compliance with the Guarantees [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 8 ADJUSTMENT OF GUARANTEES

- In the event of any change to any law, governmental regulation or requirement or interpretation thereof ("**Rule Change**") by any governmental agency made subsequent to the date of the Agreement and such Rule Change affects the A319 NEO Aircraft configuration or performance or both required to obtain certification the Guarantees will be appropriately modified to reflect the effect of any such change.
- 8.2 The Guarantees apply to the A319 NEO Aircraft as described in Paragraph 1 above and may be adjusted in the event of:
  - a) Any further configuration change which is the subject of an SCN
  - b) Changes required to obtain certification which cause changes to the performance or weight of the A319 NEO Aircraft.

### 9 EXCLUSIVE GUARANTEES

The Guarantees are exclusive and are provided in lieu of any and all other performance and weight guarantees of any nature which may be stated, referenced or incorporated in the Specification or any other document.

#### 10 UNDERTAKING REMEDIES

Should the A319 NEO Aircraft fail to meet any of the Guarantees specified in this Letter Agreement the Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 10.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.3 The Seller's maximum liability in respect of deficiency in performance of any A319 NEO Aircraft will be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 11 DUPLICATE REMEDIES

Except as provided in Paragraph 10, the remedies provided to the Buyer under this Guarantee are not cumulative of any other remedies provided to the Buyer under any other warranty or guarantee contained in the Agreement and the Buyer will not be entitled to duplicate remedies with respect to any single defect or costs incurred for any single defect.

## 12 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

## 13 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

## 14 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

f the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the seller.		
	Very truly yours,	
	AIRBUS S.A.S.	
	By: Its:	
Accepted and Agreed		
AMERICAN AIRLINES, INC.		
By:		

# APPENDIX A TO LETTER AGREEMENT NO. 11D

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: A321 AIRCRAFT PERFORMANCE GUARANTEE – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 11I (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

### 1 AIRCRAFT CONFIGURATION

The guarantees defined in Paragraphs 2, 3, 4 and 5 below (the "Guarantees") are applicable to the A321 Aircraft as described in the A321 Standard Specification [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] for:

- i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH HE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iii) the following design weights:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

hereinafter referred to as the "Specification" without taking into account any further changes thereto as provided in the Agreement.

#### 2 GUARANTEED PERFORMANCE

- 2.1 Take-Off
- 2.1.1 Take-Off Field Length

The FAR approved Take-Off Field Length at an A321 Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at the start of ground run at sea level pressure altitude in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 2.1.2 Take-Off Weight

When the A321 Aircraft is operated in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

the FAR approved Take-Off Weight of the A321 Aircraft at the start of ground run shall be not less than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2.2 Landing Field Length

The FAR approved dry Landing Field Length at an A321 Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at sea level pressure altitude shall be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 3 MISSION GUARANTEES

3.1 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A321 Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The destination airport conditions are such as to allow the required Landing Weight to be used

without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO	) <i>P</i>
REQUEST FOR CONFIDENTIAL TREATMENT]	

- 3.1.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.2	Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	The A321 Aircraft will be capable of carrying a Payload of not less than
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.2.1	The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.2.2	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.2.3	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.2.4	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.2.5	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.2.6	An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
3.2.7	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.2.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 3.3 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A321 Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.3.1 The departure airport conditions [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] are as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.3.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.3.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 3.4 Mission Block Fuel

In carrying a fixed payload of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the A321 Aircraft will be not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

	•
3.4.1	The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

when the A321 Aircraft is operated as defined here below.

- 3.4.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.7 Stage distance is defined as the distance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Block fuel is defined as the fuel [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.5 The Mission Payload and Block Fuel guarantees are based on [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (see Appendix A).

## 4 MANUFACTURER'S WEIGHT EMPTY GUARANTEE

The Seller guarantees a Manufacturer's Weight Empty of the A321 Aircraft as defined in Paragraph 1 above of not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

This is the Manufacturer's Weight Empty of the A321 Aircraft as defined in Section 13-10 of the A321 Standard Specification as amended by the SCN's defined in Paragraph 1 above [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as defined in the Paragraph 8 below.

## 5 NOISE GUARANTEE

5.1 Noise Certification

The A321 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Noise

The Seller shall provide the Buyer with a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 5.3 Interior Noise in Flight

## 5.3.1 Cockpit

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 5.3.2 Cabir

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 5.4 Ramp Noise

The APU noise during ground operation of the A321 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## **6 GUARANTEE CONDITIONS**

- 6.1 The certification requirements for the A321 Aircraft, except where otherwise noted, will be as stated in Section 02 of the Specification.
- 6.2 For the determination of FAR take-off and landing performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- When establishing take-off performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - Cruise performance at [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.4 Where applicable the Guarantees assume the use of an approved fuel having a density of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.5 Sound levels are to be specified in decibels (dB), at a reference pressure of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - The definition of dB(A) shall be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - The Speech Interference Level (SIL) is defined as the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 7 GUARANTEE COMPLIANCE

- 7.1 Compliance with the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2 Compliance with the take-off and landing elements of the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3 Compliance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.4 Compliance with the Manufacturer's Weight Empty guarantee will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

7.5 The [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be used to demonstrate compliance with the guarantees of certification noise levels.

The A321 Aircraft will have a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- Data [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be adjusted as required [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] accordance with established aeronautical practices to show compliance with the Guarantees.
- 7.7 The Seller undertakes to furnish the Buyer with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] demonstrating compliance with the Guarantees [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 8 ADJUSTMENT OF GUARANTEES

- 8.1 In the event of any change to any law, governmental regulation or requirement or interpretation thereof ("**Rule Change**") by any governmental agency made subsequent to the date of the Agreement and such Rule Change affects the A321 Aircraft configuration or performance or both required to obtain certification the Guarantees will be appropriately modified to reflect the effect of any such change.
- 8.2 The Guarantees apply to the A321 Aircraft as described in Paragraph 1 above and may be adjusted in the event of:
  - a) Any further configuration change which is the subject of an SCN
  - b) Changes required to obtain certification which cause changes to the performance or weight of the A321 Aircraft.

#### 9 EXCLUSIVE GUARANTEES

The Guarantees are exclusive and are provided in lieu of any and all other performance and weight guarantees of any nature which may be stated, referenced or incorporated in the Specification or any other document.

#### 10 UNDERTAKING REMEDIES

Should the A321 Aircraft fail to meet any of the Guarantees specified in this Letter Agreement the Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 10.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- The Seller's maximum liability in respect of deficiency in performance of any A321 Aircraft will be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 11 DUPLICATE REMEDIES

Except as provided in Paragraph 10, the remedies provided to the Buyer under this Guarantee are not cumulative of any other remedies provided to the Buyer under any other warranty or guarantee contained in the Agreement and the Buyer will not be entitled to duplicate remedies with respect to any single defect or costs incurred for any single defect.

## 12 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

## 13 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

## 14 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

Seller.	
	Very truly yours,
	AIRBUS S.A.S.
	By:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By:	

If the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the

# APPENDIX A TO LETTER AGREEMENT NO. 11I

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: A321 AIRCRAFT PERFORMANCE GUARANTEE – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 11J (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

## 1 AIRCRAFT CONFIGURATION

The guarantees defined in Paragraphs 2, 3, 4 and 5 below (the "Guarantees") are applicable to the A321 Aircraft as described in the A321 Standard Specification [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] for:

- i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH HE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iii) the following design weights:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

hereinafter referred to as the "Specification" without taking into account any further changes thereto as provided in the Agreement.

#### 2 GUARANTEED PERFORMANCE

- 2.1 Take-Off
- 2.1.1 Take-Off Field Length

The FAR approved Take-Off Field Length at an A321 Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at the start of ground run at sea level pressure altitude in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2.1.2 Take-Off Weight

When the A321 Aircraft is operated in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

the FAR approved Take-Off Weight of the A321 Aircraft at the start of ground run shall be not less than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2.2 Landing Field Length

The FAR approved dry Landing Field Length at an A321 Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at sea level pressure altitude shall be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 3 MISSION GUARANTEES

3.1 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A321 Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL P	ORTION
OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREAT!	MENT]

- 3.1.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.2 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A321 Aircraft will be capable of carrying a Payload of not less than

- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

  The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.2.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A321 Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.3.1 The departure airport conditions [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] are as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.3.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 3.4 Mission Block Fuel

In carrying a fixed payload of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the A321 Aircraft will be not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

when the A321 Aircraft is operated as defined here below.

- 3.4.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

  The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.7 Stage distance is defined as the distance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Block fuel is defined as the fuel [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.4.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.5 The Mission Payload and Block Fuel guarantees are based on [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (see Appendix A).

## 4 MANUFACTURER'S WEIGHT EMPTY GUARANTEE

The Seller guarantees a Manufacturer's Weight Empty of the A321 Aircraft as defined in Paragraph 1 above of not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

This is the Manufacturer's Weight Empty of the A321 Aircraft as defined in Section 13-10 of the A321 Standard Specification as amended by the SCN's defined in Paragraph 1 above [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as defined in the Paragraph 8 below.

## 5 NOISE GUARANTEE

5.1 Noise Certification

The A321 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Noise

The Seller shall provide the Buyer with a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 5.3 Interior Noise in Flight

## 5.3.1 Cockpit

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5.3.2 Cabin

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5.4 Ramp Noise

The APU noise during ground operation of the A321 Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## **6 GUARANTEE CONDITIONS**

6.1 The certification requirements for the A321 Aircraft, except where otherwise noted, will be as stated in Section 02 of the Specification.

- 6.2 For the determination of FAR take-off and landing performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - When establishing take-off performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - Cruise performance at [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REOUEST FOR CONFIDENTIAL TREATMENT]
- 6.4 Where applicable the Guarantees assume the use of an approved fuel having a density of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.5 Sound levels are to be specified in decibels (dB), at a reference pressure of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - The definition of dB(A) shall be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - The Speech Interference Level (SIL) is defined as the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 7 GUARANTEE COMPLIANCE

- 7.1 Compliance with the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2 Compliance with the take-off and landing elements of the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3 Compliance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 7.4 Compliance with the Manufacturer's Weight Empty guarantee will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.5 The [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be used to demonstrate compliance with the guarantees of certification noise levels.
  - The A321 Aircraft will have a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.6 Data [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be adjusted as required using [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] accordance with established aeronautical practices to show compliance with the Guarantees.
- 7.7 The Seller undertakes to furnish the Buyer with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] demonstrating compliance with the
  - Guarantees [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 8 ADJUSTMENT OF GUARANTEES

- 8.1 In the event of any change to any law, governmental regulation or requirement or interpretation thereof ("**Rule Change**") by any governmental agency made subsequent to the date of the Agreement and such Rule Change affects the A321 Aircraft configuration or performance or both required to obtain certification the Guarantees will be appropriately modified to reflect the effect of any such change.
- 8.2 The Guarantees apply to the A321 Aircraft as described in Paragraph 1 above and may be adjusted in the event of:
  - a) Any further configuration change which is the subject of an SCN
  - b) Changes required to obtain certification which cause changes to the performance or weight of the A321 Aircraft.

#### 9 EXCLUSIVE GUARANTEES

The Guarantees are exclusive and are provided in lieu of any and all other performance and weight guarantees of any nature which may be stated, referenced or incorporated in the Specification or any other document.

## 10 UNDERTAKING REMEDIES

Should the A321 Aircraft fail to meet any of the Guarantees specified in this Letter Agreement the Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 10.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.3 The Seller's maximum liability in respect of deficiency in performance of any A321 Aircraft will be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 11 DUPLICATE REMEDIES

Except as provided in Paragraph 10, the remedies provided to the Buyer under this Guarantee are not cumulative of any other remedies provided to the Buyer under any other warranty or guarantee contained in the Agreement and the Buyer will not be entitled to duplicate remedies with respect to any single defect or costs incurred for any single defect.

## 12 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

## 13 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

# 14 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please ex Seller.	foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the r.		
	Very truly yours,		
	AIRBUS S.A.S.		
	By:		
Accepted and Agreed			
AMERICAN AIRLINES, INC.			
By:			

# APPENDIX A TO LETTER AGREEMENT NO. 11J

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: A321 NEO AIRCRAFT PERFORMANCE GUARANTEE – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 11K (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

## 1 AIRCRAFT CONFIGURATION

The guarantees defined in Paragraphs 2, 3, 4 and 5 below (the "Guarantees") are applicable to the A321 NEO Aircraft as described in the A321 Standard Specification [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] for:

- i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iii) the following design weights:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

hereinafter referred to as the "Specification" without taking into account any further changes thereto as provided in the Agreement.

#### 2 GUARANTEED PERFORMANCE

- 2.1 Take-Off
- 2.1.1 Take-Off Field Length

The FAR approved Take-Off Field Length at an A321 NEO Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at the start of ground run at sea level pressure altitude in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2.1.2 Take-Off Weight

When the A321 NEO Aircraft is operated in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

the FAR approved Take-Off Weight of the A321 NEO Aircraft at the start of ground run shall be not less than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2.2 Landing Field Length

The FAR approved dry Landing Field Length at an A321 NEO Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at sea level pressure altitude shall be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 3 MISSION GUARANTEES

3.1 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A321 NEO Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL P	ORTION
OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREAT!	MENT]

- 3.1.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.6 An allowance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR e) CONFIDENTIAL TREATMENT] Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST 3.2 FOR CONFIDENTIAL TREATMENT] The A321 NEO Aircraft will be capable of carrying a Payload of not less than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION 3.2.1 OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.2.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A321 NEO Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.3.1 The departure airport conditions [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] are as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.3.2 An allowance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 3.4 Mission Block Fuel

In carrying a fixed payload of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the A321 NEO Aircraft will be not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
when the A321 NEO Aircraft is operated as defined here below.

- 3.4.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

  The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.7 Stage distance is defined as the distance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Block fuel is defined as the fuel [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:

- a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.5 The Mission Payload and Block Fuel guarantees are based on [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (see Appendix A).

#### 4 MANUFACTURER'S WEIGHT EMPTY GUARANTEE

The Seller guarantees a Manufacturer's Weight Empty of the A321 NEO Aircraft as defined in Paragraph 1 above of not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

This is the Manufacturer's Weight Empty of the A321 NEO Aircraft as defined in Section 13-10 of the A321 Standard Specification as amended by the SCNs defined in Paragraph 1 above [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as defined in the Paragraph 8 below.

## 5 NOISE GUARANTEE

5.1 Noise Certification

The A321 NEO Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5.3 Interior Noise in Flight

## 5.3.1 Cockpit

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5.3.2 Cabin

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5.4 Ramp Noise

The APU noise during ground operation of the A321 NEO Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 6 GUARANTEE CONDITIONS

6.1 The certification requirements for the A321 NEO Aircraft, except where otherwise noted, will be as stated in Section 02 of the Specification.

- 6.2 For the determination of FAR take-off and landing performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - When establishing take-off performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.4 Where applicable the Guarantees assume the use of an approved fuel having a density of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.5 Sound levels are to be specified in decibels (dB), at a reference pressure of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - The definition of dB(A) shall be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The Speech Interference Level (SIL) is defined as the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 7 GUARANTEE COMPLIANCE

- 7.1 Compliance with the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2 Compliance with the take-off and landing elements of the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3 Compliance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.4 Compliance with the Manufacturer's Weight Empty guarantee will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 7.5 The [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be used to demonstrate compliance with the guarantees of certification noise levels.
  - The A321 NEO Aircraft will have a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 7.6 Data [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be adjusted as required [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] accordance with established aeronautical practices to show compliance with the Guarantees.
- 7.7 The Seller undertakes to furnish the Buyer with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] demonstrating compliance with the Guarantees [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 8 ADJUSTMENT OF GUARANTEES

- 8.1 In the event of any change to any law, governmental regulation or requirement or interpretation thereof ("**Rule Change**") by any governmental agency made subsequent to the date of the Agreement and such Rule Change affects the A321 NEO Aircraft configuration or performance or both required to obtain certification the Guarantees will be appropriately modified to reflect the effect of any such change.
- 8.2 The Guarantees apply to the A321 NEO Aircraft as described in Paragraph 1 above and may be adjusted in the event of:
  - a) Any further configuration change which is the subject of an SCN
  - b) Changes required to obtain certification which cause changes to the performance or weight of the A321 NEO Aircraft.

## 9 EXCLUSIVE GUARANTEES

The Guarantees are exclusive and are provided in lieu of any and all other performance and weight guarantees of any nature which may be stated, referenced or incorporated in the Specification or any other document with the exception of the A320 Family NEO Aircraft Performance Retention Guarantee.

#### 10 UNDERTAKING REMEDIES

Should the A321 NEO Aircraft fail to meet any of the Guarantees specified in this Letter Agreement the Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 10.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.3 The Seller's maximum liability in respect of deficiency in performance of any A321 NEO Aircraft will be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 11 DUPLICATE REMEDIES

Except as provided in Paragraph 10, the remedies provided to the Buyer under this Guarantee are not cumulative of any other remedies provided to the Buyer under any other warranty or guarantee contained in the Agreement and the Buyer will not be entitled to duplicate remedies with respect to any single defect or costs incurred for any single defect.

#### 12 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

## 13 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

# 14 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the Seller.	
	Very truly yours,
	AIRBUS S.A.S.
	By:Its:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By: Its:	

# APPENDIX A TO LETTER AGREEMENT NO. 11K

#### LETTER AGREEMENT NO. 11L

As of July 20, 2011

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: A321 NEO AIRCRAFT PERFORMANCE GUARANTEE – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 11L (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

#### 1 AIRCRAFT CONFIGURATION

The guarantees defined in Paragraphs 2, 3, 4 and 5 below (the "Guarantees") are applicable to the A321 NEO Aircraft as described in the A321 Standard Specification [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] for:

- i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iii) the following design weights:
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

hereinafter referred to as the "Specification" without taking into account any further changes thereto as provided in the Agreement.

#### 2 GUARANTEED PERFORMANCE

- 2.1 Take-Off
- 2.1.1 Take-Off Field Length

The FAR approved Take-Off Field Length at an A321 NEO Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at the start of ground run at sea level pressure altitude in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] be not more than:

#### 2.1.2 Take-Off Weight

When the A321 NEO Aircraft is operated in [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

the FAR approved Take-Off Weight of the A321 NEO Aircraft at the start of ground run shall be not less than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2.2 Landing Field Length

The FAR approved dry Landing Field Length at an A321 NEO Aircraft gross weight of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] at sea level pressure altitude shall be not more than:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 3 MISSION GUARANTEES

3.1 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A321 NEO Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.1.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL P	ORTION
OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREAT!	MENT]

- 3.1.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.1.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - The A321 NEO Aircraft will be capable of carrying a Payload of not less than
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

  The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT
- 3.2.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.2.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.2.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3 Mission Payload [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The A321 NEO Aircraft will be capable of carrying a Payload of not less than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3.3.1 The departure airport conditions [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] are as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.3.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.7 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 3.4 Mission Block Fuel

In carrying a fixed payload of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of the A321 NEO Aircraft will be not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

when the A321 NEO Aircraft is operated as defined here below.

- 3.4.1 The departure airport conditions are such as to allow the required Take-Off Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

  The destination airport conditions are such as to allow the required Landing Weight to be used without restriction. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.2 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.3 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT
- 3.4.5 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.6 An allowance of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4.7 Stage distance is defined as the distance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Block fuel is defined as the fuel [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.4.8 At the end of approach and landing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of fuel will remain in the tanks. This represents the estimated fuel required for:
  - a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.5 The Mission Payload and Block Fuel guarantees are based on [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (see Appendix A).

#### 4 MANUFACTURER'S WEIGHT EMPTY GUARANTEE

The Seller guarantees a Manufacturer's Weight Empty of the A321 NEO Aircraft as defined in Paragraph 1 above of not more than

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

This is the Manufacturer's Weight Empty of the A321 NEO Aircraft as defined in Section 13-10 of the A321 Standard Specification as amended by the SCNs defined in Paragraph 1 above [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT as defined in the Paragraph 8 below.

# 5 NOISE GUARANTEE

5.1 Noise Certification

The A321 NEO Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5.2 Noise

The Seller shall provide the Buyer with a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 5.3 Interior Noise in Flight

#### 5.3.1 Cockpit

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 5.3.2 Cabin

At a pressure altitude of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] shall be as follows:

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Noise levels shall be measured [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 5.4 Ramp Noise

The APU noise during ground operation of the A321 NEO Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## **6 GUARANTEE CONDITIONS**

- 6.1 The certification requirements for the A321 NEO Aircraft, except where otherwise noted, will be as stated in Section 02 of the Specification.
- 6.2 For the determination of FAR take-off and landing performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - When establishing take-off performance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.4 Where applicable the Guarantees assume the use of an approved fuel having a density of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.5 Sound levels are to be specified in decibels (dB), at a reference pressure of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The definition of dB(A) shall be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

The Speech Interference Level (SIL) is defined as the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 7 GUARANTEE COMPLIANCE

- 7.1 Compliance with the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.2 Compliance with the take-off and landing elements of the Guarantees will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.3 Compliance [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 7.4 Compliance with the Manufacturer's Weight Empty guarantee will be demonstrated [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7.5 The [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be used to demonstrate compliance with the guarantees of certification noise levels.
  - The A321 NEO Aircraft will have a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 7.6 Data [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will be adjusted as required [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] accordance with established aeronautical practices to show compliance with the Guarantees.
- 7.7 The Seller undertakes to furnish the Buyer with [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] demonstrating compliance with the Guarantees [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 8 ADJUSTMENT OF GUARANTEES

- 8.1 In the event of any change to any law, governmental regulation or requirement or interpretation thereof ("**Rule Change**") by any governmental agency made subsequent to the date of the Agreement and such Rule Change affects the A321 NEO Aircraft configuration or performance or both required to obtain certification the Guarantees will be appropriately modified to reflect the effect of any such change.
- 8.2 The Guarantees apply to the A321 NEO Aircraft as described in Paragraph 1 above and may be adjusted in the event of:
  - a) Any further configuration change which is the subject of an SCN
  - b) Changes required to obtain certification which cause changes to the performance or weight of the A321 NEO Aircraft.

#### 9 EXCLUSIVE GUARANTEES

The Guarantees are exclusive and are provided in lieu of any and all other performance and weight guarantees of any nature which may be stated, referenced or incorporated in the Specification or any other document with the exception of the A320 Family NEO Aircraft Performance Retention Guarantee.

#### 10 UNDERTAKING REMEDIES

Should the A321 NEO Aircraft fail to meet any of the Guarantees specified in this Letter Agreement the Seller will [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 10.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10.3 The Seller's maximum liability in respect of deficiency in performance of any A321 NEO Aircraft will be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 11 DUPLICATE REMEDIES

Except as provided in Paragraph 10, the remedies provided to the Buyer under this Guarantee are not cumulative of any other remedies provided to the Buyer under any other warranty or guarantee contained in the Agreement and the Buyer will not be entitled to duplicate remedies with respect to any single defect or costs incurred for any single defect.

#### 12 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

# 13 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

# 14 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execute the Geller.	original and one (1) copy hereof in the space provided below and return a copy to the
	Very truly yours,
	AIRBUS S.A.S. By:
	Its:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By:	

# APPENDIX A TO LETTER AGREEMENT NO. 11L

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: OTHER MATTERS RELATED TO [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an Airbus A320 Family Aircraft Purchase Agreement dated as of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement"), which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 14 (this "Letter Agreement") certain additional terms and conditions regarding the sale or lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement will have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

- 1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] AIRCRAFT
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] FAMILY AIRCRAFT
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] AIRCRAFT
- 3.1 Scope

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.2 <u>Applicability</u>
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3.5 Additional Terms and Conditions
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

(iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# 3.6 <u>Predelivery Payments</u>

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 4 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

## 5 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

## 6 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execute the original and one (1) Seller.	) copy hereof in the space provided below and return a copy to the
	Very truly yours,
	AIRBUS S.A.S.
	By:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By: Its:	

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: SELLER PARTS AND SELLER PARTS SERVICES

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement dated as of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement"), which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 15 (this "Letter Agreement") certain additional terms and conditions regarding the sale or lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement will have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement will constitute an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement will be governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

#### 1 SCOPE OF SPARES SUPPORT

For [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (the "**Term**"), the Seller will (i) maintain, or cause to be maintained, a stock of Seller Parts (as defined below) and make available to the Buyer materials management services (the "**Seller Services**"), [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and (ii) sell and deliver such Seller Parts and Seller Services or interchangeable Seller Parts (in each case, together with all necessary documentation and data) in accordance with the provisions of this Letter Agreement.

For the purposes of this Letter Agreement, the term "Seller Parts" means the Seller's proprietary parts bearing a part number of the Seller or for which the Seller has the exclusive sales rights.

## 2 DELIVERY

2.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Seller Parts

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.2 Emergency Services

During the Term, the Seller will maintain, or cause to be maintained, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] the lead-times for delivery of such qualified answer to the Buyer will not exceed:

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 3 PRICE
- 3.1 Price Condition

Price conditions for any Seller Parts or Seller Services provided hereunder will be in accordance with Article 5.1 of Exhibit H to the Agreement.

During the Term, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 3.3 Out of Production Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 4 WARRANTIES

Article 4 of Exhibit H to the Agreement will apply, mutatis mutandis, to all Seller Parts and Seller Services purchased by the Buyer pursuant to this Letter Agreement.

#### 5 TERMINATION

If the Buyer shall default in the performance of its obligations to the Seller under this Letter Agreement, the Seller will have the right to suspend deliveries of Seller Parts and Seller Services to the Buyer under this Letter Agreement and, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 6 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

#### 7 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

#### B COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the Seller.	
	Very truly yours,
	AIRBUS S.A.S.
	By:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By: Its:	

#### LETTER AGREEMENT NO. 16

As of July 20, 2011

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

## **Re: SPECIFICATION MATTERS**

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 16 (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement constitutes an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement is governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

# 1 MSCNs FOR [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 2 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

# 3 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

## 4 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execute the original and one (1) copy hereof in the space provided below and return a copy to the Seller.	
	Very truly yours,
	AIRBUS S.A.S.
	Ву:
	Its:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
Ву:	
Its:	

#### LETTER AGREEMENT NO. 17

As of July 20, 2011

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: TERMINATION

Dear Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented and amended by the other letter agreements, and as otherwise supplemented, amended or modified from time to time, the "Agreement") which covers, among other matters, the sale by the Seller and the purchase or lease by the Buyer of certain Aircraft, under the terms and conditions set forth in said Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 17 (this "Letter Agreement") certain additional terms and conditions regarding the sale and lease of the Aircraft. Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement constitutes an integral, nonseverable part of said Agreement, that the provisions of said Agreement are hereby incorporated herein by reference, and that this Letter Agreement is governed by the provisions of said Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

- 1.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] pursuant to this Paragraph 1.2.

#### 2 TERMINATION

Clause 20 of the Agreement is deleted in its entirety and replaced with Clause 20 attached hereto as Appendix 1.

### 3 ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

#### 4 CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

## 5 COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please execute the original and one $(1)$ c Seller.	opy hereof in the space provided below and return a copy to the
	Very truly yours,
	AIRBUS S.A.S.
	By:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By: Its:	

#### 20 TERMINATION

20.1

### **Termination Events**

Each of the following will constitute a "Termination Event"

- (1) The Buyer or any [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Affiliates commences in any jurisdiction any case, proceeding or other legal action with respect to the Buyer or any [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Affiliates or all or any substantial part of its assets relating to bankruptcy, insolvency, reorganization, winding-up, liquidation or dissolution.
- (2) A case, proceeding or other legal action is commenced in any jurisdiction seeking the appointment of a receiver, trustee, custodian or other similar official for the Buyer or any [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Affiliates or for all or any substantial part of their respective assets, and such action remains unstayed, undismissed or undischarged for [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Affiliates makes a general assignment for the benefit of its creditors.
- (3) The Buyer or any [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Affiliates (i) fails to make any payments [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] which are required to be made under this Agreement or any other [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] agreement between the Buyer or any [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Affiliates and the Seller or any [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Affiliates (after giving effect to any grace period applicable thereto) [CONFIDENTIAL

- PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (4) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (5) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (6) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 20.2 Remedies in Event of Termination

- 20.2.1 If [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Termination Event occurs, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] the Seller will have the right to resort to any remedy under applicable law or in equity, and may, without limitation, by written notice to the Buyer, immediately do any or all of the following:
  - A. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - B. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - C. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

20.2.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

20.2.3. The parties to this Agreement are commercially sophisticated parties acting within the same industry, and represented by competent counsel and the parties expressly agree and declare as follows:

A. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

B. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

C. the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] provision of this Clause 20 has been fully negotiated by sophisticated parties represented by counsel, is a material component of the consideration granted and, in the absence of such [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] provision, the consideration would have been materially different.

### 20.3 <u>Notice of Termination Event</u>

Within [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of becoming aware of the occurrence of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Termination Event, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will notify [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] of such occurrence in writing, provided, that any failure by [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION

PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to notify [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] will not prejudice [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] rights or remedies hereunder.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## 20.4 <u>Adequate Assurance</u>

#### LETTER AGREEMENT NO. 1

As of July 20, 2011

#### CONFIDENTIAL - SUBJECT TO RESTRICTIONS ON DISCLOSURE AS SET FORTH IN PARAGRAPH 12 BELOW

American Airlines, Inc. 4333 Amon Carter Boulevard Fort Worth, Texas 76155

#### Re: LEASING MATTERS

Ladies and Gentlemen,

American Airlines, Inc. (the "Buyer") and Airbus S.A.S. (the "Seller") have entered into an A320 Family Aircraft Purchase Agreement of even date herewith (as supplemented, amended or otherwise modified from time to time, the "Agreement"), which covers, among other matters, the sale by the Seller and the purchase by the Buyer of Purchase Aircraft and the lease of Leased Aircraft to the Buyer, in each case under the terms and subject to the conditions set forth in the Agreement. The Buyer and the Seller have agreed to set forth in this Letter Agreement No. 1 (this "Letter Agreement") certain additional terms and conditions regarding the Leased Aircraft and leasing transactions related thereto.

Attached to this Letter Agreement as Exhibits **A**, **B**, **C** and **D**, respectively, are the forms of the following documents, which have been fully negotiated by the Buyer and the Seller, with respect to the Leased Aircraft leasing transactions (collectively, the "Leasing Documentation"):

- (A) Lease Agreement, including schedules and exhibits thereto (the "Lease");
- (B) Trust Agreement, including schedules and exhibits thereto (the "Trust Agreement");
- (C) Participation Agreement, including schedules and exhibits thereto (the "Participation Agreement"); and
- (D) Definitions (the "Definitions Annex").

Capitalized terms used herein and not otherwise defined in this Letter Agreement have the meanings assigned thereto in the Agreement or the Definitions Annex, as applicable. The terms "herein," "hereof" and "hereunder" and words of similar import refer to this Letter Agreement.

Both parties agree that this Letter Agreement constitutes an integral, nonseverable part of the Agreement, that the provisions of the Agreement are hereby incorporated herein by reference, and that this Letter Agreement is governed by the provisions of the Agreement, except that if the Agreement and this Letter Agreement have specific provisions which are inconsistent, the specific provisions contained in this Letter Agreement will govern.

#### 1. THE LEASES

1.1 Forms

The documentation for each of the Leased Aircraft shall be in the form of the Leasing Documentation (subject to such changes as shall be agreed by the Buyer and the applicable Owner Participant, in their sole discretion) and shall be prepared by the Buyer.

1.2 Leases of Leased Aircraft

Subject to the terms and conditions set forth herein (including, without limitation, Paragraph 1.3 hereof) and in the Leasing Documentation and the Agreement, the Buyer shall lease from the Seller each of the Leased Aircraft, and the Seller shall lease to the Buyer each of the Leased Aircraft. The Buyer and the Seller agree that the Leasing Documentation with respect to each Leased Aircraft is to be entered into on the Delivery Date of such Leased Aircraft.

- 1.3. Certain Terms Applicable to Third Party Leasing Transactions
  - 1.3.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - 1.3.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - 1.3.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 1.3.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 1.4 Execution and Delivery of the Leasing Documentation of Leased Aircraft
  - 1.4.1 The obligation of the Buyer to execute and deliver the Leasing Documentation with respect to any Leased Aircraft on the respective Delivery Date thereof is subject to the satisfaction of or waiver by the Buyer of, the following conditions:
    - (a) such Leased Aircraft shall be Ready for Delivery and shall meet all requirements for Aircraft under the Agreement, and all of the Buyer's rights and the Seller's obligations pertaining thereto (including, but not limited to, aircraft specification, type and condition, delivery of an Export Certificate of Airworthiness without restriction or limitation, warranties and technical support, BFE arrangements and delivery and inspection procedures) shall have been satisfied in full accordance with the Agreement;
    - (b) the "Owner Trustee" of such Leased Aircraft shall be Wells Fargo Bank Northwest, National Association, or another U.S. bank or trust company reasonably acceptable to the Buyer;
    - (c) the "Owner Participant" of such Leased Aircraft shall be either: (1) Airbus Financial Services, (2) an Affiliate of Airbus Financial Services (Airbus Financial Services and such Affiliate, each an "AFS Entity") or (3) a Third Party Owner Participant;
    - (d) on the Delivery Date of such Leased Aircraft, the Seller shall have transferred title to such Leased Aircraft to the "Owner Trustee" and such "Owner Trustee" shall have title to such Leased Aircraft free and clear of Liens;
    - (e) in the case of a leasing transaction to be entered into with a Third Party Purchaser, (i) such Third Party Purchaser and the Buyer shall have executed and delivered a Leasing Letter with respect to such Leased Aircraft in accordance with Paragraph 1.3.4 hereof, (ii) such Leasing Letter shall not have been terminated (including as provided in such

Leasing Letter or Paragraph 5 hereof) and (iii) the Buyer shall not have provided written notice to the Seller that such Third Party Purchaser (or any Affiliate thereof) has breached any of its obligations set forth in such Leasing Letter or any other Leasing Letter executed by such Third Party Purchaser (or any Affiliate thereof) (in each case, including, if applicable, the obligation to lease any other Leased Aircraft to the Buyer in accordance with the terms of such Leasing Letter);

- (f) in the case of a leasing transaction to be entered into with a Third Party Purchaser, the Buyer shall not have provided written notice to the Seller that the Third Party Owner Participant (or any Affiliate thereof) related to such leasing transaction has breached any of its material obligations as "Owner Participant" under Leasing Documentation executed and delivered by it with respect to any other Leased Aircraft;
- (g) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (h) the "Basic Rent" and "Stipulated Loss Value" to be inserted in the Leasing Documentation with respect to such Leased Aircraft shall be calculated (x) in the case of a leasing transaction with an AFS Entity by the Buyer as provided in Exhibit G hereof and verified by such AFS Entity and (y) in the case of a leasing transaction with a Third Party Purchaser by the Buyer as provided in Exhibit F of the Leasing Letter and verified by such Third Party Purchaser; and
- (i) the Agreement shall not have been terminated, in whole, or with respect to such Leased Aircraft.

If any of the above conditions precedent or any condition precedent set forth in Section 4.3 of the Participation Agreement with respect to such Leased Aircraft are not satisfied (except if any such unsatisfied condition precedent relates solely to a Third Party Purchaser (or a Third Party Owner Participant) and the Seller has caused an AFS Entity to lease such Leased Aircraft to the Buyer in accordance with Paragraph 1.4.2 hereof), the Buyer shall have no obligation to lease such Leased Aircraft or to execute and deliver the Leasing Documentation relating to such Leased Aircraft, and the failure of such Leasing Documentation to be executed and delivered as a result thereof shall not be a breach of the Agreement by the Buyer or constitute or result in a Buyer Termination Event.

- 1.5 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - 1.5.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - 1.5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - (b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 2. LEASING DOCUMENTATION WITH THE SELLER

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2.1 Lease
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 3. BASIC RENT, STIPULATED LOSS VALUES & BASIC TERM

Basic Rent and Stipulated Loss Value for each Leased Aircraft will be calculated as provided in **Exhibit G**. The Basic Term of each Lease shall be [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

#### 4. DELIVERY

Inspection and technical acceptance of each Leased Aircraft will be conducted as set forth in the Agreement and resolution of any discrepancies in the condition of such Leased Aircraft at Delivery will be effected between the Buyer and the Seller in accordance therewith [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

#### 5. DELAY

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 6 SALE OF BEE

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### 7. LEASED AIRCRAFT PURCHASE OPTION

7.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

7.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

7.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

9. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

10. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

11. ASSIGNMENT

This Letter Agreement and the rights and obligations of the parties will be subject to the provisions of Clause 21 of the Agreement; provided, however, this Letter Agreement may not be assigned by the Buyer under either Clause 21.5 or 21.6 without the express written consent of the Seller, which the Seller may withhold in its sole discretion.

12. CONFIDENTIALITY

This Letter Agreement is subject to the terms and conditions of Clause 22.11 of the Agreement.

13. COUNTERPARTS

This Letter Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument.

If the foregoing correctly sets forth your understanding, please ex Seller.	xecute the original and one (1) copy hereof in the space provided below and return a copy to the
	Very truly yours,
	AIRBUS S.A.S.
	Ву:
	Name: Title:
Accepted and Agreed	
AMERICAN AIRLINES, INC.	
By: Name: Title:	

EXHIBIT A

FORM OF LEASE AGREEMENT

EXHIBIT B

FORM OF TRUST AGREEMENT

EXHIBIT C

FORM OF PARTICIPATION AGREEMENT

EXHIBIT D

FORM OF DEFINITIONS ANNEX

EXHIBIT E

FORM OF LEASING LETTER

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

4. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 1. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 1. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## SCHEDULE I

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**EXHIBIT G** 

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# PART A – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

I. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (A) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (B) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (A) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (B) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (C) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (D) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (E) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (F) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

PART B – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**PART C** – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**EXHIBIT H** 

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

EXHIBIT H
ANNEX A
TO

## [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### **EXHIBIT I**

- 2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 7. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### **EXHIBIT I**

- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 8. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 9. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 11. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 12. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 13. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 14. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## EXHIBIT I

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**EXHIBIT J** 

## [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## **EXHIBIT J**

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

\* \* \* \* \*

## [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Exhibit A

to

Exhibit J

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

**EXHIBIT K** 

## [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

CONFIDENTIAL: Annexes B and C of this Lease Are Subject to Restrictions on Dissemination Set Forth in Section 10.4 of the Participation Agreement (as defined herein)

## LEASE AGREEMENT ([YEAR] MSN [MSN])

dated as of

[Date]

between

### WELLS FARGO BANK NORTHWEST, NATIONAL ASSOCIATION,

not in its individual capacity, except as expressly provided herein, but solely as Owner Trustee, as Lessor

and

AMERICAN AIRLINES, INC., as Lessee

Covering One Airbus [Model] Aircraft (Generic Manufacturer and Model AIRBUS [Generic Model])

TO THE EXTENT, IF ANY, THAT THIS LEASE AGREEMENT CONSTITUTES CHATTEL PAPER (AS DEFINED IN THE UNIFORM COMMERCIAL CODE AS IN EFFECT IN ANY APPLICABLE JURISDICTION), NO SECURITY INTEREST IN THIS LEASE AGREEMENT MAY BE PERFECTED THROUGH DELIVERY OR POSSESSION OF ANY COUNTERPART OF THIS LEASE AGREEMENT OTHER THAN THE ORIGINAL COUNTERPART, WHICH SHALL BE THE COUNTERPART THAT CONTAINS THE RECEIPT EXECUTED BY LESSOR ON THE SIGNATURE PAGE THEREOF.

## Table of Contents

C + 1	Definitions.	Pag
Section 1.	Definitions	-
Section 2.	Leasing of Aircraft	-
Section 3.	Term and Rent Term Basic Rent Supplemental Rent Payments in General	
Section 4.	Lessor's Representations, Warranties and Covenants Disclaimer U.S. Citizenship Quiet Enjoyment Lien Lifting Warranties Lessor's Interest in Certain Engines Title Transfers by Lessor Vesting of Title	
Section 5.	Return of Aircraft	4
Section 6.	Liens	;
Section 7. (a) (b) (c)	Registration, Maintenance and Operation; Possession; Insignia Registration, Maintenance and Operation Possession Insignia	( ( ( 1)
Section 8.	Replacement and Pooling of Parts; Alterations, Modifications and Additions; Substitution of Engines Replacement of Parts Pooling of Parts Alterations, Modifications and Additions Substitution of Engines Excluded Equipment	1: 1: 1: 1: 1: 1: 1:
Section 9.	[Intentionally Left Blank]	15
Section 10.  (a) (b) (c) (d) (e) (f)	Loss, Destruction, Requisition, etc. Event of Loss with Respect to the Airframe Payments with Respect to Events of Loss Requisition for Use of the Airframe Not Constituting an Event of Loss Requisition for Use by a Government of an Engine Application of Payments During Existence of Event of Default Event of Loss with Respect to Engine	19 19 10 10 17 13
Section 11.	Insurance	17

(a) Aircraft Liability Insurance

## Table of Contents (continued)

(b)	Insurance Against Loss or Damage to Aircraft	Page 19
(c)	Self-Insurance	21
(d)	Application of Insurance Payments	21
(e)	Reports, Etc.	22
(f)	Salvage Rights; Other	22
Section 12.	Inspection	22
(a)	Annual Inspection of Aircraft	22
(b)	Marketing Inspection of Aircraft	23
(c)	Electronic Records	23
(d)	Confidentiality	24
(e)	Compliance	24
Section 13.	Assignment	24
Section 14.	Events of Default	24
Section 15.	Remedies	26
Section 16.	Further Assurances	29
Section 17.	Notices	29
Section 18.	No Set-Off, Counterclaim, etc.	30
Section 19.	Section 1110	31
Section 20.	Monies Received by Lessor	31
Section 21.	Renewal Options	31
Section 22.	Investment of Security Funds	31
Section 23.	Confidential Information	31
Section 24.	Lessor Right to Perform for Lessee	31
Section 25.	Lessee's Performance and Rights	32
Section 26.	Concerning Lessor	32
Section 27.	Successor Owner Trustee	32
Section 28.	Miscellaneous	33

## EXHIBITS, ANNEXES AND SCHEDULES

EXHIBIT A - FORM OF LEASE SUPPLEMENT
ANNEX A - DEFINITIONS
ANNEX B - RETURN CONDITIONS
ANNEX C - MID-TERM INSPECTION RECORDS LIST

### LEASE AGREEMENT ([YEAR] MSN [MSN])

This LEASE AGREEMENT ([YEAR] MSN [MSN]) (as amended, modified or supplemented from time to time, this "Lease"), dated as of [ ], [YEAR], between WELLS FARGO BANK NORTHWEST, NATIONAL ASSOCIATION, a national banking association, not in its individual capacity, except as expressly provided herein, but solely as Owner Trustee (herein in such capacity, together with its successors and permitted assigns, "Lessor" or "Owner Trustee", and in its individual capacity, together with its successors and permitted assigns, "Trust Company"), and AMERICAN AIRLINES, INC., a Delaware corporation (together with its successors and permitted assigns, "Lessee").

### **RECITALS:**

Lessee wishes to lease the Aircraft from Lessor, and Lessor wishes to lease the Aircraft to Lessee, on the terms and subject to the conditions provided herein.

- **NOW, THEREFORE**, in consideration of the premises and mutual agreements herein contained, the agreements contained in the other Operative Documents and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:
- **Section 1.** <u>Definitions</u>. Unless the context otherwise requires, all capitalized terms used herein and not otherwise defined herein shall have the meanings set forth, and shall be construed and interpreted in the manner described, in Annex A hereto for all purposes of this Lease.
- **Section 2.** <u>Leasing of Aircraft</u>. (a) Lessor hereby agrees (subject to satisfaction or waiver of the conditions set forth in <u>Sections 4.1</u> and <u>4.2</u> of the Participation Agreement) to lease to Lessee hereunder, and Lessee hereby agrees (subject to satisfaction or waiver of the conditions set forth in <u>Section 4.3</u> of the Participation Agreement) to lease from Lessor hereunder, the Aircraft, as evidenced by the execution by Lessor and Lessee of Lease Supplement No. 1 covering the Aircraft.
- (b) On the Delivery Date, subject to Lessee's acceptance of the Aircraft, Lessee will take possession of the Aircraft "AS-IS, WHERE-IS AND WITH ALL FAULTS."

### **Section 3.** Term and Rent.

- (a) <u>Term</u>. The Basic Term for the lease of the Aircraft hereunder shall commence on the Delivery Date and shall end on the Lease Expiry Date, or such earlier date on which this Lease is terminated in accordance with the provisions hereof.
- (b) <u>Basic Rent</u>. Lessee hereby agrees to pay to Lessor Basic Rent in advance for the Aircraft throughout the Term in installments, the first installment of which shall be due and payable on the Delivery Date, and the remaining installments of which shall be due and payable on the other Lease Period Dates, in the amounts computed as provided in <u>Schedule A</u> to Lease Supplement No. 1 for the Basic Term (Basic Rent payable for any Renewal Term, shall be as provided in <u>Section 21</u>). The installment of Basic Rent due and payable on the Delivery Date

shall be allocable to the Lease Period commencing on the Delivery Date and ending on the day immediately preceding the following Lease Period Date. Each other installment of Basic Rent is allocable to the Lease Period beginning on the Lease Period Date on which such installment is due and payable.

- (c) <u>Supplemental Rent</u>. Lessee also agrees to pay to Lessor, or to whomsoever shall be entitled thereto, any and all Supplemental Rent promptly as the same shall become due and owing, and in the event of any failure on the part of Lessee to pay any Supplemental Rent, Lessor shall, subject to <u>Section 15</u>, have all rights, powers and remedies provided for herein, in equity or law, as in the case of nonpayment of Basic Rent. In addition, Lessee will pay as Supplemental Rent, on demand, to the extent permitted by applicable Law, an amount equal to interest at the Overdue Rate on any part of any installment of Basic Rent not paid when due for any period for which the same shall be overdue and on any payment of Supplemental Rent not paid when due or demanded, as the case may be, for the period until the same shall be paid.
- (d) <u>Payments in General</u>. All payments of Rent shall be made in Dollars by wire transfer of immediately available funds not later than 1:00 p.m. (New York time) on the date of payment, to Lessor to the account set forth on <u>Annex B</u> to the Participation Agreement (or such other account in the United States of Lessor as Lessor directs by written notice to Lessee at least 10 Business Days prior to the date such payment of Rent is due, or, in the case of Supplemental Rent expressly payable to a Person other than Lessor, to the Person that shall be entitled thereto to such account in the United States as such Person directs by written notice to Lessee at least 10 Business Days prior to the date such payment of Rent is due). If any Rent is due on a day that is not a Business Day, such Rent shall be paid on the next succeeding Business Day with the same force and effect as if paid on the scheduled date of payment, and no interest shall accrue on the amount of such payment from and after such scheduled date to the time of payment on such next succeeding Business Day.

### Section 4. Lessor's Representations, Warranties and Covenants.

(a) <u>Disclaimer</u>. NONE OF OWNER TRUSTEE, TRUST COMPANY OR OWNER PARTICIPANT (IN EACH CASE, IN ITS CAPACITY AS SUCH) MAKES OR SHALL BE DEEMED TO HAVE MADE HEREIN ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, AS TO THE AIRWORTHINESS, VALUE, CONDITION, WORKMANSHIP, DESIGN, OPERATION, **MERCHANTABILITY OR FITNESS FOR USE OR FOR A PARTICULAR PURPOSE OF THE AIRCRAFT OR ANY ENGINE OR ANY PART THEREOF**, AS TO THE ABSENCE OF LATENT OR OTHER DEFECTS, WHETHER OR NOT DISCOVERABLE, AS TO THE ABSENCE OF ANY INFRINGEMENT OF ANY PATENT, TRADEMARK OR COPYRIGHT, OR AS TO THE ABSENCE OF OBLIGATIONS BASED ON STRICT LIABILITY IN TORT, OR ANY OTHER REPRESENTATION OR WARRANTY WHATSOEVER, EXPRESS OR IMPLIED, WITH RESPECT TO THE AIRCRAFT OR ANY ENGINE OR ANY PART THEREOF, except that nothing set forth in this <u>subsection (a)</u> shall (x) derogate from the representations and warranties made by Owner Trustee, Trust Company or Owner Participant in or pursuant to any Operative Document or (y) be construed as a waiver by Lessee of any warranty or other claim against any manufacturer, supplier, dealer, contractor, subcontractor or other Person.

- (b) <u>U.S. Citizenship</u>. Lessor at all times will be a Citizen of the United States to permit registration of the Aircraft with the FAA. Trust Company represents and warrants that it is a Citizen of the United States. Owner Trustee represents and warrants that it is a Citizen of the United States.
- (c) <u>Quiet Enjoyment</u>. Lessor covenants that, except as expressly permitted by <u>Section 15</u> following an Event of Default that has occurred and is continuing, notwithstanding anything herein or in any other Operative Document to the contrary, neither Lessor nor any Person claiming by, through or under Lessor shall (i) discharge the registration with the International Registry of the International Interests arising with respect to the Lease, (ii) transfer the right to discharge any of such International Interests to any other Person or cause any such right to be so transferred (except (x) in connection with a Transfer permitted by <u>Section 8.1</u> of the Participation Agreement, or [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] or (iii) take or cause to be taken any action inconsistent with Lessee's rights under this Lease and its right to quiet enjoyment of the Aircraft, the Airframe, any Engine or any Part, or otherwise in any way interfere with or interrupt the use, operation and continuing possession of the Aircraft, the Airframe, any Engine or any Part by Lessee or any sublessee, assignee or transferee under any sublease, assignment or transfer then in effect and permitted by the terms of this Lease.
- (d) Lien Lifting. Lessor agrees that (i) it shall promptly, at its own cost and expense, take such action as may be necessary duly to discharge and satisfy in full any Lessor's Lien attributable to it if the same shall arise at any time (by bonding or otherwise, so long as Lessee's operation and use of the Aircraft is not impaired); provided that Lessor may, for a period of not more than 60 days, contest any such Lessor's Lien diligently and in good faith by appropriate proceedings so long as such contest does not involve any material risk of the sale, forfeiture or loss of use of the Airframe or any Engine or any material risk of criminal penalties or material civil penalties being imposed on Lessee, and (ii) it shall indemnify and hold harmless Lessee from and against any loss, cost, expense or damage (including reasonable legal fees and expenses) that may be suffered or incurred by Lessee as a result of a failure by Lessor to promptly discharge or satisfy in full any such Lessor's Lien.
- (e) <u>Warranties</u>. Lessor agrees that, so long as no Event of Default shall have occurred and be continuing, Lessee shall have the benefit of and shall be entitled to enforce, either in its own name or in the name of Lessor for the use and benefit of Lessee, any and all warranties of any Person (whether express or implied) in respect of the Aircraft, the Airframe, any Engine or any Part, and Lessor agrees to execute and deliver such further documents and take such further action, as may be reasonably requested by Lessee and at Lessee's cost and expense, as may be necessary to enable Lessee to obtain such warranty service or the benefits of any such warranty as may be furnished for the Aircraft, Airframe, any Engine or any Part by such Person. Lessor hereby appoints and constitutes Lessee, except at such times as an Event of Default shall have occurred and be continuing, its agent and attorney-in-fact during the Term to assert and enforce, from time to time, in the name and for the account of Lessor and Lessee, as their interests may appear, but in all cases at the cost and expense of Lessee, whatever claims and rights Lessor may have against such Person.

- (f) <u>Lessor's Interest in Certain Engines</u>. Lessor hereby agrees, for the benefit of the lessor, conditional vendor or secured party of any airframe or any engine leased, purchased or owned by Lessee (or any Permitted Sublessee) subject to a lease, conditional sale or other security agreement, that Lessor will not acquire or claim, as against such lessor, conditional vendor or secured party, any right, title or interest in any engine or engines as the result of such engine or engines being installed on the Airframe at any time while such engine or engines are subject to such lease, conditional sale or other security agreement, <u>provided however</u>, that such agreement of Lessor shall not be for the benefit of any lessor, conditional vendor or secured party of any airframe or any engine leased, purchased or owned by Lessee (or any Permitted Sublessee) subject to a lease, conditional sale or other security agreement, unless such lessor, conditional vendor, or secured party has expressly agreed (which agreement may be contained in such lease, conditional sale or other security agreement) that neither it nor its successors or assigns will acquire or claim, as against Lessor, any right, title or interest in an Engine as a result of such Engine being installed on such airframe subject to such lease, conditional sale or security agreement.
- (g) <u>Title Transfers by Lessor</u>. If Lessor shall be required to transfer title to the Aircraft or any Engine to Lessee or its designee pursuant to this Lease, (i) Lessor will (A) transfer to Lessee or its designee, without recourse or warranty (except as to the absence of Lessor's Liens and Liens of the type described in <u>Section 6(h)</u>, all of Lessor's right, title and interest in and to such Aircraft or Engine, free and clear of all right, title and interest of Lessor and of Lessor's Liens and Liens of the type described in <u>Section 6(h)</u>, all in AS-IS WHERE-IS condition, (B) at Lessee's expense, execute and deliver such bills of sale (and any such bill of sale shall be in such form as will qualify as a "contract of sale" pursuant to Article V of the Aircraft Protocol) and other documents and instruments of transfer (including consents to appropriate registrations with the International Registry), all in form and substance reasonably satisfactory to Lessee, as Lessee shall reasonably request to evidence (on the public record or otherwise) such transfer and the vesting in Lessee or its designee of all of Lessor's right, title and interest in and to such Aircraft or Engine and (C) take such actions as may be required to be taken by Lessor so that the transfer of such Aircraft or Engine to Lessee or its designee shall be registered as a Sale on the International Registry, (ii) Lessor will assign (to the extent freely assignable) to Lessee or its designee, pursuant to an assignment agreement in form and substance reasonably satisfactory to Lessee, all of its right, title and interest in and to claims against third Persons relating to such Aircraft or Engine.
- (h) <u>Vesting of Title</u>. Lessor agrees that in each instance in which this Lease provides that title to the Aircraft, any Engine, engine, Part or Obsolete Part shall be transferred to or vest in Lessee, title to such Aircraft, Engine, engine, Part or Obsolete Part shall vest in Lessee free and clear of all right, title and interest of Lessor, Lessor's Liens and Liens of the type described in <u>Section 6(h)</u>, and Lessor shall do all acts necessary to discharge all such Liens and other rights held by it in such Aircraft, Engine, engine, Part or Obsolete Part.

**Section 5.** Return of Aircraft. Lessee hereby agrees to comply with the Return Conditions regarding return of the Aircraft to Lessor. In addition, Lessee agrees, in connection with any return of the Aircraft hereunder, to pay on the Return Date the amounts payable pursuant to Annex B, if any. All references in this Lease or elsewhere in any other Operative Document to this Section 5 shall be deemed to refer also to Annex B.

**Section 6.** <u>Liens.</u> Lessee will not, directly or indirectly, create, incur, assume or suffer to exist any Lien on or with respect to the Airframe or any Engine, title thereto or any interest therein or in this Lease except:

- (a) the respective rights of the parties to the Operative Documents as provided therein;
- (b) the rights of others under agreements or arrangements to the extent expressly permitted by this Lease;
- (c) Lessor's Liens;
- (d) Liens for Taxes that either are not yet overdue or are being contested in good faith by appropriate proceedings so long as such proceedings do not involve any material risk of the sale, forfeiture or loss of the Airframe or any Engine, title thereto or any interest therein or any material risk of criminal liability or material civil penalty against Lessor or Owner Participant;
- (e) materialmen's, mechanics', workers', landlord's, repairmen's, employees' or other like Liens arising in the ordinary course of business (including those arising under maintenance agreements entered into in the ordinary course of business) securing obligations that either are not yet overdue for a period of more than 60 days or are being contested in good faith by appropriate proceedings so long as such proceedings do not involve any material risk of the sale, forfeiture or loss of the Airframe or any Engine, title thereto or any interest therein or any material risk of criminal liability or material civil penalty against Lessor or Owner Participant;
- (f) Liens (other than Liens for Taxes) arising out of any judgment or award (i) for 60 days after the entry of such judgment or award, provided that during such 60-day period there is no material risk of the sale, forfeiture or loss of the Airframe or any Engine, title thereto or any interest therein or any material risk of criminal liability or material civil penalty against Lessor or Owner Participant, or (ii) during an appeal or other proceeding for review regarding such judgment or award with respect to which there shall have been secured a stay of execution pending such appeal or review;
  - (g) salvage or similar rights of insurers under insurance policies maintained pursuant to Section 11;
- (h) the respective rights of the financing parties under any financing arrangements entered into by Lessor or Owner Participant with respect to the Aircraft at any time, including, without limitation arrangements permitted by Section 8.3 of the Participation Agreement;
  - (i) Liens approved in writing by Lessor; and

(j) any other Lien with respect to which Lessee shall have provided cash collateral or other security adequate in the reasonable opinion of Lessor.

Liens described in <u>clauses (a)</u> through (j) above are referred to as "**Permitted Liens**". Lessee will promptly take (or cause to be taken) such action as may be necessary duly to discharge (by bonding or otherwise) any Lien not excepted above if the same shall arise at any time.

### Section 7. Registration, Maintenance and Operation; Possession; Insignia.

- (a) Registration, Maintenance and Operation. Lessee, at its expense, shall:
- (i) subject to the further provisions of this Section 7, cause the Aircraft to remain duly registered at the FAA in the name of Lessor, as owner, except:
  - (A) as otherwise required by the Transportation Code, or
- (B) to the extent that such registration cannot be maintained (x) because of the failure of Lessor or Owner Participant to comply with the citizenship or other eligibility requirements for registration of aircraft under the Transportation Code or with Section 6.3.1 or 6.4.4 of the Participation Agreement or (y) because of the failure by Lessor or Owner Participant to execute and deliver, upon request of Lessee, any documents required for the renewal of such registration;

provided that Lessor and Owner Participant shall execute and deliver all such documents as may be required by the FAA from time to time for the purpose of effecting and continuing such registration, and shall not register the Aircraft or permit the Aircraft to be registered under any laws other than the United States at any time except as provided in the following proviso; and provided, further, that Lessee may at any time, with the prior written consent of Owner Participant (such consent not to be unreasonably withheld), subject to satisfaction of the Re-registration Conditions or waiver of any thereof by Owner Participant, cause the Aircraft or permit the Aircraft to be registered under the applicable statutes of any country in which a Permitted Sublessee could be based, in the name of Lessor or, if required by applicable Law, in the name of any other Person, and Lessor and Owner Participant shall cooperate with Lessee's reasonable requests in effecting and continuing such foreign registration, and Lessee shall maintain such registration unless and until the Aircraft is re-registered in accordance with this Section 7;

(ii) cause the Aircraft to be maintained, serviced, repaired, reconditioned, overhauled, stored and tested in accordance with Lessee's maintenance program for aircraft of the same make and model, which shall be an FAA Part 121 approved program (the "Maintenance Program") (or, if the Aircraft is then registered in accordance with the terms of the Operative Documents in another country or shall be subleased to a Permitted Sublessee, in each case in accordance with the terms of this Lease, an Approved Program) and, except during any Sublease Period, in the same manner and with the same care used by Lessee with respect to comparable

[A319/320/321]¹ aircraft and engines owned or operated by Lessee and utilized in similar circumstances (and, during any Sublease Period, by a maintenance performer appropriately approved by the FAA or EASA of recognized standing, experience and facilities to perform the relevant work on aircraft of the same make and model as the Aircraft and in the same manner and with the same care used by the Permitted Sublessee with respect to comparable [A319/320/321]² aircraft and engines owned or operated by the Permitted Sublessee and utilized in similar circumstances) so as to keep the Aircraft in the same operating condition as when delivered to Lessee hereunder (ordinary wear and tear excepted);

- (iii) cause the Aircraft to be kept in such condition as may be necessary to enable an airworthiness certification of the Aircraft to be maintained in good standing at all times (other than during temporary periods of storage of not more than 90 calendar days in accordance with applicable regulations or during periods of grounding by applicable governmental authorities, except where such periods of grounding are the result of the failure by Lessee to maintain the Aircraft as otherwise required herein) under the Transportation Code or the applicable laws of any other jurisdiction in which the Aircraft may then be registered;
- (iv) cause all records, logs and other documentation with respect to the Aircraft to be maintained as required by the FAA or the applicable central authority of the jurisdiction where the Aircraft is registered to be maintained in respect of the Aircraft (all such records, logs and other documentation to be maintained in the English language);
- (v) maintain and update a [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] cross-reference table indicating for each of the Maintenance Program tasks the corresponding MPD reference task (if any); and
- (vi) cause to be furnished to Lessor (A) such information that is readily available without undue expense as may be reasonably requested by Lessor to enable Lessor to file any reports, filings or statements required to be filed by Lessor with the FAA (or the aeronautical authority of the country of registry of the Aircraft if the Aircraft is not registered under the laws of the United States) because of Lessor's interest in the Aircraft, and (B) such other information concerning the location, condition, use and operation of the Aircraft as Lessor may reasonably request.

Lessee agrees that it will comply with all mandatory airworthiness directives issued by the FAA (or the appropriate authorities in the jurisdiction where the Aircraft is registered) (each, an "AD" and collectively, "ADs") in respect of the Aircraft which require compliance no later

Specific aircraft type to be specified in each Lease.

Specific aircraft type to be specified in each Lease.

than the last day of the Term, as and to the extent required by such ADs and the Maintenance Program prior to such date. Lessee shall not be required to comply with any manufacturer service bulletins, except as and to the extent required by the Maintenance Program prior to the last day of the Term.

The Aircraft will not be maintained, used or operated in violation of any law, rule, regulation or order of any government or governmental authority having jurisdiction in any country in which the Aircraft is flown, or in violation of any AD, license or registration relating to the Aircraft issued by any such authority; provided that Lessee may in good faith contest the validity or application of any such law, rule, regulation, order, airworthiness certificate, license or registration or any AD referred to in the immediately preceding paragraph in any reasonable manner which does not materially adversely affect Lessor or Owner Participant or their respective interests in the Aircraft or any Operative Document, or involve any material risk of criminal liability or material civil penalty against Lessor or Owner Participant; and provided, further, that Lessee shall not be in default under this sentence if it is not possible for Lessee to comply with the laws of a jurisdiction other than the United States (or other than any jurisdiction in which the Aircraft is then registered) because of a conflict with the applicable laws of the United States (or such jurisdiction in which the Aircraft is then registered) in which event Lessee shall use its reasonable best efforts to cause the Aircraft to be removed, as soon as practicable, from the jurisdiction other than the United States (or other than the jurisdiction in which the Aircraft is then registered) creating the conflict or take such other reasonable action (including, if necessary, changing the registration of the Aircraft unless the Aircraft is then registered in the United States), as soon as practicable, as may be necessary to avoid the conflict.

Lessee may operate or allow the Aircraft to be operated anywhere in the world, except that Lessee agrees not to operate or locate the Aircraft, or suffer the Aircraft to be operated or located:

- (A) in any area excluded from coverage by any insurance required by the terms of <u>Section 11</u>, except in the case of a requisition for use by the U.S. government where Lessee obtains indemnity in lieu of such insurance from the U.S. government against the risks and in the amounts required by <u>Section 11</u> covering such area, or
- (B) in any war zone or recognized or, in Lessee's reasonable judgment, threatened area of hostilities unless covered by war risk insurance or unless the Aircraft is operated or used under contract with the U.S. government under which contract the U.S. government assumes liability for loss of, damage to, or loss of use of, the Aircraft and for injury to persons or damage to property of others.
- (b) <u>Possession</u>. Lessee will not, without the prior written consent of Lessor, sublease or otherwise in any manner deliver, transfer or relinquish possession of the Airframe or any Engine or install any Engine, or permit any Engine to be installed, on any airframe other than the Airframe; <u>provided</u> that Lessee or a Person permitted to be in possession of the Aircraft, the Airframe or any Engine may, without the prior consent of Lessor:

- (i) [Intentionally Left Blank.]
- (ii) deliver possession of the Airframe or any Engine to any Person for testing, service, repair, reconditioning, restoration, storage, maintenance, overhaul work or other similar purposes or for alterations, modifications or additions to the Airframe or such Engine to the extent required or permitted by the terms of this Lease:
- (iii) transfer possession of the Airframe or any Engine to the U.S. government pursuant to a sublease, contract or other instrument, a copy of which shall be furnished to Lessor; <u>provided</u> that the term of such sublease (including, without limitation, any option of the sublessee to renew or extend) or the term of possession under such contract or other instrument shall not continue beyond the end of the Basic Term or any Renewal Term then in effect or any Renewal Term that Lessee has irrevocably notified Lessor that it will exercise;
- (iv) subject the Airframe or any Engine to the CRAF Program or transfer possession of the Airframe or any Engine at any time to the U.S. government or any instrumentality or agency thereof in accordance with applicable laws, rulings, regulations or orders (including, without limitation, the CRAF Program); provided that Lessee (A) shall promptly notify Lessor upon transferring possession of the Airframe or any Engine pursuant to this clause (iv) and (B) in the case of a transfer of possession pursuant to the CRAF Program, within 60 days thereof, shall notify Lessor of the name, address and phone number of the responsible Contracting Office Representative for the Air Mobility Command of the U.S. Air Force or other appropriate Person to whom notices must be given with respect to such Airframe or Engine;
- (v) install an Engine on an airframe owned by Lessee free and clear of all Liens except Permitted Liens and those which apply only to the engines (other than Engines), appliances, parts, instruments, appurtenances, accessories, furnishings and other equipment (other than Parts) installed on such airframe (but not to the airframe as an entirety);
- (vi) install an Engine on an airframe leased, purchased or owned by Lessee subject to a lease, conditional sale or other security agreement; provided that (A) such airframe is free and clear of all Liens except (1) the rights of the parties to the lease or conditional sale or other security agreement covering such airframe and (2) Liens of the type permitted by clause (y) above and (B) either (1) there shall have been obtained from the lessor, conditional vendor or secured party of such airframe a written agreement (which may be the lease or conditional sale or other security agreement covering such airframe), in form and substance satisfactory to Lessor (it being understood that an agreement from such lessor, conditional vendor or secured party substantially in the form of Section 4(f) shall be deemed to be satisfactory to Lessor) whereby such lessor, conditional vendor or secured party expressly agrees that it will not acquire or claim any right, title or interest in any Engine by reason of such Engine being installed on such airframe at any time while such Engine is subject to this Lease or title thereto is held by Lessor or (2) such lease, conditional sale or other security agreement effectively provides that such Engine shall not become subject to the Lien of such lease, conditional sale or other security agreement at any time while such Engine is subject to this Lease or title thereto is held by Lessor, notwithstanding the installation thereof on such airframe;

(vii) install an Engine on an airframe owned by Lessee, leased to Lessee or owned by Lessee subject to a conditional sale or other security agreement under circumstances where neither <u>clause (v)</u> nor <u>clause (vi)</u> is applicable; <u>provided</u> that, if such installation shall divest Lessor's title to such Engine, such installation shall be deemed an Event of Loss with respect to such Engine and Lessee shall comply with <u>Section 8(d)</u> in respect thereof, it being understood that Lessor does not intend hereby to waive any right or interest it may have to or in such Engine under applicable law until compliance by Lessee with such <u>Section 8(d)</u>; and

(viii) sublease any Engine or the Airframe and Engines or engines then installed on the Airframe; <u>provided</u> that ( $\underline{A}$ ) such sublease will be to a Permitted Sublessee; ( $\underline{B}$ ) the sublessee is not then subject to any bankruptcy, insolvency, liquidation, reorganization, dissolution or similar proceeding and shall not have substantially all of its property in the possession of any liquidator, trustee, receiver or similar person; ( $\underline{C}$ ) the term of such sublease (including, without limitation, any option of the sublessee to renew or extend) shall not continue beyond the end of the Basic Term or any Renewal Term then in effect or any Renewal Term that Lessee has irrevocably notified Lessor that it will exercise; ( $\underline{D}$ ) such sublease shall require the sublessee to maintain such Engine or such Airframe and Engines, as the case may be (or cause such Engine or such Airframe and Engines, as the case may be, to be maintained) pursuant to an Approved Program and otherwise in compliance with the terms of this Lease; and ( $\underline{E}$ ) Lessor and Owner Participant shall have received assurances reasonably satisfactory to the Owner Participant to the effect that the insurance provisions of the Lease shall have been complied with after giving effect to such sublease;

provided that the rights of any transferee who receives possession by reason of a transfer permitted by this <u>subsection (b)</u> (other than the transfer of an Engine which is deemed an Event of Loss) shall be, during the period of such possession, subject and subordinate to, and any sublease permitted by this <u>subsection (b)</u> shall be made expressly subject and subordinate to, all the terms of this Lease, including, without limitation, Lessor's rights to repossession pursuant to <u>Section 15</u> and to avoid and terminate such sublease upon the occurrence of an Event of Default, and Lessee shall in all events remain primarily liable hereunder for the performance and observance of all the terms and conditions of this Lease (including, without limitation, the terms and conditions set forth in <u>Section 7(a)(ii)</u> and <u>Section 11</u>) to the same extent as if such sublease or transfer had not occurred, and that any such sublease shall provide that (except with respect to a sublease to a Permitted Sublessee described in clause (i) of the definition thereof) the sublessee may not further sub-sublease the Aircraft. No sublease or other relinquishment of possession of the Airframe or any Engine shall in any way discharge or diminish any of Lessee's obligations to Lessor hereunder. Lessee shall, prior to entering into a sublease of the Airframe or Engines, notify Lessor of the identity of the sublessee and the term of such sublease, and provide Lessor with a copy of such sublease; <u>provided</u> that the identity of the sublessee and terms of such sublease shall be Confidential Information and shall be held by Lessor in accordance with the provisions of <u>Section 23</u>. Any sublease having a term (including available renewal terms) in excess of 12 months shall be assigned to Lessor as additional security for the

obligations of Lessee hereunder (such assignment to be on such terms and subject to such conditions (including the making of registrations with the International Registry and filings and notifications with the FAA or other applicable governmental authority) as shall be reasonably satisfactory to Lessor and Lessee).

Lessor acknowledges that any "wet lease" or other similar arrangement under which Lessee maintains operational control of the Aircraft shall not constitute a delivery, transfer or relinquishment of possession for purposes of this <u>subsection (b)</u>. No "wet lease" will extend beyond the Basic Term or any Renewal Term then in effect or any Renewal Term that Lessee has irrevocably notified Lessor that it will exercise.

(c) <u>Insignia</u>. No later than 30 days following the Delivery Date, Lessee shall affix and shall thereafter during the Term maintain in the cockpit of the Airframe adjacent to the airworthiness certificate therein and (if not prevented by applicable law or regulations or by any governmental authority) on each Engine a metal nameplate bearing the legible inscription "TITLE TO THIS AIRCRAFT/ENGINE IS HELD BY WELLS FARGO BANK NORTHWEST, NATIONAL ASSOCIATION, NOT IN ITS INDIVIDUAL CAPACITY, BUT SOLELY AS OWNER TRUSTEE, AS LESSOR, WHICH HAS LEASED THIS AIRCRAFT/ENGINE TO AMERICAN AIRLINES, INC.", such nameplate to be replaced, if need be, with a nameplate reflecting the name of any successor Owner Trustee. Except as provided above, Lessee will not allow the name of any Person to be placed on the Airframe or on any Engine as a designation that constitutes a claim of ownership; <u>provided</u> that nothing herein contained shall prohibit Lessee from placing its customary colors and insignia (and those of any code-sharing partner or the **one**world global alliance or any member thereof) on such Airframe or Engine or displaying information concerning the registration of the Aircraft.

### Section 8. Replacement and Pooling of Parts; Alterations, Modifications and Additions; Substitution of Engines.

(a) Replacement of Parts. Lessee, at its own cost and expense, will promptly replace or cause to be replaced all Parts which may from time to time be incorporated or installed in or attached to the Airframe or any Engine and which may from time to time become worn out, lost, stolen, destroyed, seized, confiscated, damaged beyond repair or permanently rendered unfit for use for any reason whatsoever, except as otherwise provided in subsection (c) or if the Airframe or any Engine to which a Part relates has suffered an Event of Loss. In addition, Lessee may, at its own cost and expense, remove or cause to be removed in the ordinary course of maintenance, service, repair, overhaul or testing, any Parts, whether or not worn out, lost, stolen, destroyed, seized, confiscated, damaged beyond repair or permanently rendered unfit for use; provided that Lessee, except as otherwise provided in subsection (c), will, at its own cost and expense, replace or cause to be replaced such Parts as promptly as practicable. All replacement parts shall be free and clear of all Liens (except for Permitted Liens) and shall as good operating condition as, and shall have a value and utility at least equal to, the Parts replaced, assuming such replaced Parts were in the condition and state of repair required by the terms hereof. Title to all Parts at any time removed from the Airframe or any Engine shall remain vested in Lessor, no matter where located, until such time as such Parts shall be replaced by Parts which have been incorporated or installed in or attached to the Airframe or any Engine and which meet the requirements for replacement parts specified above. Immediately upon any

replacement part becoming incorporated or installed in or attached to the Airframe or any Engine as above provided, without further act, (i) title to the replaced Part shall thereupon vest in Lessee, free and clear of all right, title and interest of Lessor and of Lessor's Liens and Liens of the type described in Section 6(h), and shall no longer be deemed a Part hereunder, (ii) title to such replacement part shall thereupon vest in Lessor free and clear of all Liens (except Permitted Liens) and (iii) such replacement part shall become subject to this Lease and be deemed part of the Airframe or such Engine, and a Part, for all purposes to the same extent as the Parts originally incorporated or installed in or attached to the Airframe or such Engine.

(b) <u>Pooling of Parts</u>. Any Part removed from the Airframe or an Engine as provided in <u>subsection (a)</u> may be subjected by Lessee or a Person permitted hereunder to be in possession of the Aircraft to a pooling arrangement customary in the airline industry entered into in the ordinary course of Lessee's or such other Person's business; <u>provided</u> that a part replacing such removed Part shall be incorporated or installed in or attached to the Airframe or such Engine in accordance with <u>subsection (a)</u> as promptly as practicable after the removal of such removed Part, but in any case before the last day of the Term. In addition, any replacement Part when incorporated or installed in or attached to the Airframe or an Engine in accordance with <u>subsection (a)</u> may be owned by a third party subject to such a pooling arrangement; <u>provided</u> that Lessee, at its expense, as promptly thereafter as practicable, either (i) causes title to such replacement Part to vest in Lessor in accordance with <u>subsection (a)</u> by Lessee (or any such Person) acquiring title thereto for the benefit of, and transferring such title to, Lessor free and clear of all Liens (other than Permitted Liens) or (ii) replaces or causes to be replaced such replacement Part by incorporating or installing in or attaching to the Airframe or such Engine a further replacement Part owned by Lessee (or any such Person) free and clear of all Liens (other than Permitted Liens) and otherwise satisfying the requirements of <u>subsection (a)</u> above, and by causing title to such further replacement Part to vest in Lessor in accordance with <u>subsection (a)</u>.

(c) <u>Alterations, Modifications and Additions</u>. Lessee will make or cause to be made such alterations and modifications in and additions to the Airframe, the Engines and the Parts as may be required from time to time to meet the applicable standards of the FAA or other applicable regulatory agency or body of the foreign jurisdiction in which the Aircraft is then registered as permitted by <u>Section 7(a)</u>; <u>provided</u> that Lessee may in good faith contest the validity or application of any such standard in any reasonable manner which does not materially adversely affect Lessor, Owner Participant or their respective interests in the Aircraft or involve any material risk of criminal liability or material civil penalty against Lessor or Owner Participant. In addition, Lessee, at its own expense, may from time to time make or cause to be made such alterations and modifications in and additions to the Airframe, any Engine or any Part as Lessee may deem desirable in the proper conduct of its business, including without limitation, removal of Parts that Lessee deems to be obsolete or no longer suitable or appropriate for use on the Airframe or such Engine (such Parts, "Obsolete Parts"); <u>provided</u> that no such alteration, modification, addition or removal shall materially diminish the value (except as described in the last proviso of this sentence) or utility of the Airframe or such Engine, or impair the condition or airworthiness thereof, below the value, utility, condition and airworthiness thereof immediately prior to such alteration, modification, addition or removal assuming the Airframe or such Engine was then of the value and utility and in the condition and airworthiness required to be maintained by the terms of this Lease; <u>provided</u> that the value (but not the utility, condition or airworthiness) of the Aircraft may be reduced by the value of the Obsolete Parts which shall have been

removed, if the aggregate value of all such Obsolete Parts removed from the Aircraft and not replaced in accordance with the terms of this Section 8 shall not exceed the amount specified in Schedule A to the Participation Agreement. Title to all Parts incorporated or installed in or attached or added to the Airframe or any Engine or Part as the result of such alteration, modification or addition shall, without further act, vest in Lessor. Lessor shall not be required under any circumstances to pay or compensate Lessee for any such alteration, modification or addition. Notwithstanding the foregoing, Lessee may, at any time during the Term, remove any Part; provided that (i) such Part is in addition to, and not in replacement of or substitution for, any Part originally incorporated or installed in or attached (or which should have been incorporated or installed in or attached) to the Airframe or such Engine at the time of delivery thereof to Lessee on the Delivery Date or any Part in replacement of, or substitution for, any such Part, (ii) such Part is not required to be incorporated or installed in or attached or added to such Airframe or Engine pursuant to the first sentence of this subsection (c) and (iii) such Part can be removed from the Airframe or such Engine without materially diminishing or impairing the value, utility, condition or airworthiness required to be maintained by the terms of this Lease which the Airframe or such Engine would have had at such time had such Part never been installed on the Airframe or such Engine. Upon the removal by Lessee of any Part as provided in the immediately preceding sentence or the removal of any Obsolete Part permitted by this subsection (c), title thereto shall, without further act, vest in Lessee, free and clear of all right, title and interest of Lessor and of Lessor's Liens, and such Part shall no longer be deemed part of the Airframe or the Engine from which it was removed. Title to any Part not removed by Lessee as provided in such second preceding sentence pr

### (d) Substitution of Engines.

(i) Lessee shall have the right at its option at any time, so long as no Event of Default shall have occurred and be continuing, on at least 30 days' prior notice to Lessor, to terminate this Lease with respect to any Engine by substituting a Replacement Engine for such Engine (it being understood that the Return Conditions shall apply, in lieu of this Section 8(d), to any substitutions that occur pursuant to Section I of Annex B). In addition, if an Event of Loss shall have occurred or shall have been deemed to have occurred pursuant to Section 7(b) or Section 10(d) with respect to an Engine (other than an Event of Loss that also includes the Airframe, in which event Section 10(a) shall apply), Lessee shall within 60 days of the occurrence of such Event of Loss and on at least five days' prior notice to Lessor substitute a Replacement Engine for such Engine (any such Engine suffering such Event of Loss or being substituted pursuant to the first sentence of this paragraph, a "Replaced Engine (assuming that such Replaced Engine was of the condition and repair required by the terms hereof immediately prior to the occurrence of such Event of Loss); provided that, if any Replacement Engine is being substituted for a Replaced Engine pursuant to the first sentence of this paragraph, any such Replacement Engine will have value and utility at least equal to (taking into account the number of hours or cycles since new or overhaul, whichever is more recent) the Replaced Engine (assuming that such Replaced Engine was of the condition and repair required by the terms hereof immediately prior to such substitution). No Event of Loss with respect to an Engine shall result in any reduction in Basic Rent.

- (ii) Prior to or at the time of any such substitution, Lessee shall:
- (A) furnish Lessor with a warranty (as to title) bill of sale (which warranty shall except Permitted Liens) with respect to such Replacement Engine, which in the case of any such conveyance to which the Cape Town Treaty is applicable shall be in such form as will qualify as a "contract of sale" pursuant to Article V of the Aircraft Protocol;
- (B) if the seller of such Replacement Engine is "situated in" a country that has ratified the Cape Town Treaty, cause the sale of such Replacement Engine to Lessor to be registered on the International Registry as a Sale (or, if the seller of such Replacement Engine is not situated in a country that has ratified the Cape Town Treaty, use reasonable efforts to cause the seller to register the sale of such Replacement Engine on the International Registry);
- (C) cause a Lease Supplement substantially in the form of <a href="Exhibit A">Exhibit A</a>, subjecting such Replacement Engine to this Lease, and duly executed by Lessee, to be delivered to Lessor for execution (and Lessor shall promptly execute such Lease Supplement) and, upon such execution, to be filed for recordation pursuant to the Transportation Code or, if necessary, pursuant to the applicable laws of such jurisdiction other than the United States in which the Aircraft is registered, as the case may be;
- (D) cause the International Interest created pursuant to the Lease Supplement in favor of Lessor with respect to such Replacement Engine to be registered on the International Registry as an International Interest;
- (E) furnish Lessor with such evidence of compliance with the insurance provisions of <u>Section 11</u> with respect to such Replacement Engine as Lessor may reasonably request; and
- (F) (x) if such Replacement Engine is being substituted for a Replaced Engine pursuant to the first sentence of Section 8(d)(i), furnish Lessor with a certificate of an aircraft engineer or appraiser (who may be an employee of Lessee) certifying that such Replacement Engine has a value and utility (taking into account the number of hours or cycles since new or overhaul, whichever is more recent) at least equal to, and is in as good operating condition as, the Engine so replaced assuming such Engine was in the condition and repair required by the terms hereof; and (y) if such Replacement Engine is being substituted for a Replaced Engine pursuant to the second sentence of Section 8(d)(i), furnish Lessor with a certificate of an aircraft engineer or appraiser (who may be an employee of Lessee) certifying that such Replacement Engine has a value and utility (but in any event without regard to the number of hours or cycles) at least equal to, and is in as good operating condition as, the Engine so replaced assuming such Engine was in the condition and repair required by the terms hereof.

Promptly following the recordation of the Lease Supplement covering such Replacement Engine pursuant to the Transportation Code (or pursuant to the applicable laws of the jurisdiction in which the Aircraft is registered) described in <u>clause (C)</u> and the registrations on the International Registry described in <u>clauses (B)</u> and (D), Lessee will cause to be delivered to Lessor an opinion of Aviation Counsel as to such recordation and registration.

- (iii) Upon full compliance by Lessee with the terms of <u>subsection (ii)</u>, Lessor will transfer to Lessee or its designee the Replaced Engine in accordance with <u>Section 4(g)</u>. For all purposes hereof, each Replacement Engine shall, after delivery of the warranty (as to title) bill of sale with respect to such Replacement Engine to Lessor, be deemed part of the property leased hereunder, and be deemed an "Engine" as defined herein, and such Replaced Engine shall cease to be an Engine leased hereunder.
- (e) Excluded Equipment. Lessee may install in, and remove from, the Aircraft any Excluded Equipment, and in any such case, Lessor will not acquire or claim any right, title or interest in any such Excluded Equipment as a result of its installation on the Aircraft; provided that in connection with any removal of Excluded Equipment, Lessee shall repair any damage to the Aircraft caused by such removal and shall restore the applicable areas from which such Excluded Equipment was removed to a serviceable condition appropriate for commercial passenger service by Lessee.

**Section 9.** [Intentionally Left Blank].

Section 10. Loss, Destruction, Requisition, etc.

(a) Event of Loss with Respect to the Airframe. Upon the occurrence during the Term of an Event of Loss with respect to the Airframe, Lessee shall within 15 days after such occurrence give Lessor notice of such Event of Loss, and Lessee shall, on the Loss Payment Date, pay, or cause to be paid, (A) to Lessor, the Stipulated Loss Value for the Aircraft and (B) to the Persons entitled thereto, all Supplemental Rent other than Stipulated Loss Value due and owing on such Loss Payment Date; provided that (x) if the Loss Payment Date is a Lease Period Date, Lessee shall have no obligation to pay the installment of Basic Rent that would otherwise be due and payable on such Lease Period Date and (y) if the Loss Payment Date is not a Lease Period Date, Lessee shall be entitled to credit against its obligation to pay Stipulated Loss Value the portion of the installment of Basic Rent allocable to the period from (and including) such Loss Payment Date to (but not including) the next succeeding Lease Period Date, or if no Lease Period Date succeeds such Loss Payment Date, the last day of the Term.

The "Loss Payment Date" with respect to an Event of Loss means the 90th day following the date of the occurrence of such Event of Loss.

In the event of payment in full of the Stipulated Loss Value for the Aircraft and all amounts payable pursuant to this Section 10(a):

- (i) the obligation of Lessee to pay Basic Rent hereunder on any Lease Period Date occurring on or subsequent to the Loss Payment Date shall terminate:
- (ii) the obligation of Lessee to pay Supplemental Rent (other than payments of Supplemental Rent for indemnities surviving pursuant to Section 7.3.1 of the Participation Agreement or to be made by Lessee in respect of liabilities and obligations of Lessee which have accrued but not been paid or which are in dispute as of the date of such payment) shall terminate;
  - (iii) the Term shall end; and
  - (iv) Lessor shall transfer the Aircraft to Lessee or its designee in accordance with Section 4(g).
- (b) <u>Payments with Respect to Events of Loss</u>. Any payments (other than insurance proceeds, the application of which is provided for in <u>Section 11</u>) received at any time by Lessor or by Lessee from any governmental authority or other Person with respect to an Event of Loss to the Airframe or any Engine will be applied as follows:
  - (i) if such payments are received with respect to the Airframe (or the Airframe and the Engines or engines installed on the Airframe), (A) such payments shall, after reimbursement of Lessor for costs and expenses, be applied in reduction of Lessee's obligation to pay the Stipulated Loss Value and other amounts required to be paid by Lessee pursuant to <u>subsection (a)</u>, if not already paid by Lessee or, if already paid by Lessee, shall be applied to reimburse Lessee for its payment of Stipulated Loss Value and such other amounts, and (B) the balance, if any, of such payment remaining thereafter will be apportioned between Lessee (or its designee) and Lessor as their interests may appear; and
  - (ii) if such payments are received with respect to an Event of Loss with respect to an Engine under circumstances contemplated by <u>Section 8(d)</u>, such payments shall be paid over to, or retained by, Lessee or its designee; <u>provided</u> that, in the case of an Engine with respect to which an Event of Loss shall have occurred or shall have been deemed to have occurred pursuant to <u>Section 7(b)</u> or <u>Section 10(d)</u>, Lessee shall have fully performed the terms of <u>Section 8(d)</u> with respect to the Event of Loss for which such payments are made.
- (c) <u>Requisition for Use of the Airframe Not Constituting an Event of Loss</u>. In the event of the requisition for use by the U.S. government (including for this purpose any agency or instrumentality thereof), including, without limitation, pursuant to the CRAF Program, of the Airframe and the Engines or engines installed thereon during the Term not constituting an Event of Loss, Lessee shall promptly notify Lessor of such requisition, and all of Lessee's obligations under this Lease with respect to the Aircraft shall (to the extent feasible with respect to obligations other than payment obligations) continue to the same extent as if such requisition had not occurred.

All payments received by Lessor or Lessee from the U.S. government for the use of the Airframe and such Engines or engines during the Term shall be paid over to, or retained by,

Lessee or its designee; and all payments received by Lessor or Lessee from the U.S. government for the use of the Airframe and such Engines or engines after the Term shall be paid over to, or retained by, Lessor; <u>provided</u> that if such requisition constitutes an Event of Loss, then all such payments shall be applied as provided in Section 10(b).

- (d) <u>Requisition for Use by a Government of an Engine</u>. In the event of the requisition for use by the U.S. government (including for this purpose any agency or instrumentality thereof), for a period in excess of 60 days, of any Engine (but not the Airframe) during the Term not constituting an Event of Loss, Lessee will replace such Engine hereunder by substituting another engine for such Engine in accordance with the terms of <u>Section 8(d)</u> to the same extent as if an Event of Loss had occurred with respect to such Engine, and any payments received by Lessor or Lessee from the U.S. government with respect to such requisition shall be paid over to, or retained by, Lessee or its designee.
- (e) <u>Application of Payments During Existence of Event of Default</u>. Any amount referred to in <u>subsection (b)</u>, <u>subsection (c)</u> or <u>subsection (d)</u> which is payable to Lessee or its designee shall not be paid to Lessee or its designee (or, if it has been previously paid directly to Lessee, shall not be retained by Lessee), if at the time of such payment an Event of Default shall have occurred and be continuing, but shall be paid to and held by Lessor pursuant to <u>Section 22</u> as security for the obligations of Lessee under this Lease, and at such time as there shall not be continuing any such Event of Default such amount shall be paid to Lessee or its designee.
- (f) <u>Event of Loss with Respect to Engine</u>. Upon the occurrence during the Term of an Event of Loss with respect to an Engine (other than an Event of Loss that also includes the Airframe, in which event <u>Section 10(a)</u> shall apply), the parties shall comply with the terms of <u>Section 8(d)</u> with respect thereto.

## Section 11. Insurance.

- (a) Aircraft Liability Insurance.
- (i) Except as provided in <u>clause (ii)</u> of this <u>subsection (a)</u> and subject to self-insurance to the extent specified in <u>subsection (c)</u>, Lessee will carry, or cause to be carried at no expense to the Specified Persons, aircraft liability insurance (including, but not limited to, bodily injury, personal injury and property damage liability, exclusive of manufacturer's product liability insurance) and contractual liability insurance with respect to the Aircraft (x) in amounts per occurrence that are not less than the aircraft liability insurance applicable to similar aircraft and engines in Lessee's fleet on which Lessee carries insurance (<u>provided</u> that such liability insurance (including self-insurance specified in <u>subsection (c)</u>) shall not be less than the amount per occurrence certified in the insurance report delivered to Lessor on the Delivery Date)<sup>3</sup>; (y) of the type usually
- Amount to be certified shall be no less than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

carried by corporations engaged in the same or similar business, similarly situated with Lessee, and operating similar aircraft and engines and covering risks of the kind customarily insured against by Lessee; and (z) that is maintained in effect with insurers of recognized responsibility; provided that Lessee will carry, or cause to be carried, at no expense to the Specified Persons, aircraft liability war risk and allied perils insurance, if and only to the extent the same is maintained by Lessee with respect to other aircraft operated by Lessee on the same routes. Any policies of insurance carried in accordance with this subsection (a) and any policies taken out in substitution or replacement for any of such policies shall (A) name the Specified Persons as additional insureds; (B) subject to the conditions of <u>clause (C)</u> below, provide that, in respect of the interests of the Specified Persons in such policies, the insurance shall not be invalidated by any action or inaction of Lessee and shall insure the respective interests of the Specified Persons as they appear, regardless of any breach or violation of any warranty, declaration or condition contained in such policies by Lessee; (C) provide that, except to the extent not provided for by the war risk and allied perils insurance provider, if such insurance is canceled for any reason whatsoever, or if any change is made in the policy that reduces the amount of insurance or the coverage certified in the insurance report delivered to the Specified Persons on the Delivery Date or if such insurance is allowed to lapse for nonpayment of premium, such cancellation, change or lapse shall not be effective as to any Specified Person for 30 days (seven days, or such other period as is customarily available in the industry, in the case of any war risk or allied perils coverage) after receipt by such Specified Person of written notice from such insurers of such cancellation, change or lapse; (D) provide that the Specified Persons shall not have any obligation or liability for premiums, commissions, assessments or calls in connection with such insurance; (E) provide that the insurers shall waive any rights of (1) set-off, counterclaim or any other deduction, whether by attachment or otherwise, in respect of any liability of the Specified Persons to the extent of any moneys due to the Specified Persons and (2) subrogation against the Specified Persons to the extent that Lessee has waived its rights by its agreements to indemnify the Specified Persons pursuant hereto or in the other Operative Documents; (F) be primary without right of contribution from any other insurance that may be carried by any Specified Person; and (G) expressly provide that all of the provisions thereof, except the limits of liability, shall operate in the same manner as if there were a separate policy covering each insured. In the case of a sublease or contract with the U.S. government in respect of the Aircraft or any Engine, or in the case of any requisition for use of the Aircraft or any Engine by the U.S. government, a valid agreement by the U.S. government to indemnify Lessee, or an insurance policy issued by the U.S. government, against any risks that Lessee is required hereunder to insure against shall be considered adequate insurance for purposes of this subsection (a) to the extent of the risks (and in the amounts) that are the subject of such indemnification or insurance. The insurance provisions set forth above for the benefit of the Specified Persons shall only apply to the extent that Lessee has agreed to indemnify such Specified Person pursuant to the Operative Documents or a consent and acknowledgement referred to in Section 8.3.2(b) of the Participation Agreement and then only in such Specified Person's capacity as Lessor, Trust Company, Owner Participant or Back-Leveraging Indemnified Person, as applicable. To the extent that the war-risk and allied perils insurance provider does not provide for provision of direct notice to Specified Persons of cancellation,

change or lapse in the insurance required hereunder, Lessee hereby agrees that upon receipt of notice of any thereof from such insurance provider it shall give the Specified Persons immediate notice of each cancellation or lapse of, or material change to, such insurance.

(ii) During any period that the Airframe or an Engine, as the case may be, is on the ground and not in operation, Lessee may carry or cause to be carried as to such non-operating Airframe or Engine, in lieu of the insurance required by <u>clause (i)</u> above, and subject to self-insurance to the extent specified in <u>subsection (c)</u>, insurance otherwise conforming with the provisions of said <u>clause (i)</u> except that: (A) the amounts of coverage shall not be required to exceed the amounts of airline liability insurance from time to time applicable to airframes or engines owned or leased by Lessee of the same type as such non-operating Airframe or Engine and that are on the ground and not in operation and (B) the scope of the risks covered and the type of insurance shall be the same as from time to time applicable to airframes or engines owned or leased by Lessee of the same type as such non-operating Airframe or Engine and that are on the ground and not in operation.

### (b) Insurance Against Loss or Damage to Aircraft.

(i) Except as provided in <u>clause (ii)</u> of this <u>subsection (b)</u>, and subject to self-insurance to the extent specified in <u>subsection (c)</u>, Lessee shall maintain, or cause to be maintained, in effect with insurers of recognized responsibility, at no expense to the Specified Persons, all-risk aircraft hull insurance covering the Aircraft and all-risk coverage with respect to any Engines or Parts while removed from the Aircraft (including, without limitation, war risk and allied perils insurance if and to the extent the same is maintained by Lessee or, in the case of a sublease to a Permitted Sublessee, such Permitted Sublessee, with respect to other aircraft operated by Lessee or such Permitted Sublessee, as the case may be, on the same routes) that is of the type usually carried by corporations engaged in the same or similar business and similarly situated with Lessee; provided that (x) such insurance (including self-insurance specified in subsection (c)) will at all times while the Aircraft is subject to this Lease be for an amount not less than the Stipulated Loss Value for the Aircraft from time to time and (y) such insurance need not cover an Engine while attached to an airframe not owned, leased or operated by Lessee. Any policies carried in accordance with this subsection (b) and any policies taken out in substitution or replacement for any such policies shall (A) provide that any insurance proceeds up to an amount equal to the Stipulated Loss Value, payable for any loss or damage constituting an Event of Loss with respect to the Aircraft, and any insurance proceeds in excess of the Insurance Threshold Amount, up to the amount of the Stipulated Loss Value, for any loss or damage to the Aircraft (or Engines) not constituting an Event of Loss with respect to the Aircraft, shall be paid to the Loss Payee, and that all other amounts shall be payable to Lessee or its designee unless the insurer shall have received notice that an Event of Default exists, in which case all insurance proceeds for any loss or damage to the Aircraft (or Engines) up to the Stipulated Loss Value shall be payable to the Loss Payee; (B) subject to the conditions of <u>clause (C)</u> below, provide that, in respect of the interests of the Specified Persons in such policies, the insurance shall not be invalidated by any action or inaction of Lessee and shall insure

their respective interests as they appear, regardless of any breach or violation of any warranty, declaration or condition contained in such policies by Lessee; (C) provide that, except to the extent not provided by the war risk and allied perils insurance provider, if such insurance is canceled for any reason whatsoever, or if any change is made in the policy that reduces the amount of insurance or the coverage certified in the insurance report delivered to the Specified Persons on the Delivery Date or if such insurance is allowed to lapse for nonpayment of premium, such cancellation, change or lapse shall not be effective as to any Specified Person for 30 days (seven days, or such other period as is customarily available in the industry, in the case of any war risk or allied perils coverage) after receipt by such Specified Person of written notice from such insurers of such cancellation, change or lapse; (D) provide that the Specified Persons shall not have any obligation or liability for premiums, commissions, assessments or calls in connection with such insurance; (E) provide that the insurers shall waive any rights of (1) set-off, counterclaim or any other deduction, whether by attachment or otherwise, in respect of any liability of the Specified Persons to the extent of any moneys due to the Specified Persons and (2) subrogation against the Specified Persons to the extent that Lessee has waived its rights by its agreements to indemnify the Specified Persons pursuant hereto or in the other Operative Documents; and (F) be primary without right of contribution from any other insurance that may be carried by any Specified Person. In the case of a sublease or contract with the U.S. government in respect of the Aircraft or any Engine, or in the case of any requisition for use of the Aircraft or any Engine by the U.S. government, a valid agreement by the U.S. government to indemnify Lessee, or an insurance policy issued by the U.S. government, against any risks that Lessee is required hereunder to insure against shall be considered adequate insurance for purposes of this subsection (b) to the extent of the risks (and in the amounts) that are the subject of such indemnification or insurance. The insurance provisions set forth above for the benefit of the Specified Persons shall only apply to the extent that Lessee has agreed to indemnify such Specified Person pursuant to the Operative Documents or a consent and acknowledgement referred to in Section 8.3.2(b) of the Participation Agreement and then only in such Specified Person's capacity as Lessor, Trust Company, Owner Participant or Back-Leveraging Indemnified Person, as applicable. To the extent that the war-risk and allied perils insurance provider does not provide for provision of direct notice to Specified Persons of cancellation, change or lapse in the insurance required hereunder, Lessee hereby agrees that upon receipt of notice of any thereof from such insurance provider it shall give the Specified Persons immediate notice of each cancellation or lapse of, or material change to, such

(ii) During any period that the Airframe or Engine is on the ground and not in operation, Lessee may carry or cause to be carried as to such non-operating Airframe or Engine, in lieu of the insurance required by <u>clause (i)</u> above, and subject to self-insurance to the extent specified in <u>subsection (c)</u>, insurance otherwise conforming with the provisions of said <u>clause (i)</u> except that the scope of the risks covered and the type of insurance shall be the same as from time to time applicable to airframes or engines owned or leased by Lessee of the same type as such non-operating Airframe or Engine and that are on the ground and not in operation; <u>provided</u> that, subject to self-insurance to the extent permitted by <u>subsection (c)</u>, Lessee shall maintain insurance against risk of loss or damage to such non-operating Airframe in an amount at least equal to the Stipulated Loss Value during such period that such Airframe is on the ground and not in operation.

- (c) <u>Self-Insurance</u>. Lessee may from time to time self-insure, by way of deductible, self-insured retention, premium adjustment or franchise or otherwise (including, with respect to insurance maintained pursuant to <u>subsection (a)</u> or (b) above, insuring for a maximum amount that is less than the amounts required by <u>subsection (a)</u> or (b), the risks required to be insured against pursuant to <u>subsection (a)</u> or (b), but in no case shall such self-insurance with respect to all of the aircraft and engines in Lessee's fleet (including, without limitation, the Aircraft) exceed for any 12-month policy year 1% of the average aggregate insurable value (for the preceding policy year) of all aircraft (including, without limitation, the Aircraft) on which Lessee carries insurance unless Lessee's independent aircraft insurance broker certifies that the standard among major U.S. airlines is a higher level of self insurance, in which event Lessee may self insure the Aircraft to such higher level; <u>provided</u> that a deductible per occurrence that, in the case of the Aircraft, is not in excess of the amount customarily allowed as a deductible in the industry or is required to facilitate claims handling shall be permitted in addition to the above-mentioned self-insurance.
- (d) <u>Application of Insurance Payments</u>. All losses will be adjusted by Lessee with the insurers. All insurance payments received under policies required to be maintained by Lessee hereunder, exclusive of any payments received in excess of the Stipulated Loss Value for the Aircraft from such policies, as the result of the occurrence of an Event of Loss with respect to the Airframe or an Engine will be applied as follows:
  - (i) if such payments are received with respect to the Airframe (or the Airframe and the Engines installed on the Airframe), so much of such payments remaining after reimbursement of Lessor for its costs and expenses shall be applied (A) in reduction of Lessee's obligation to pay the Stipulated Loss Value and other amounts required to be paid by Lessee pursuant to Section 10(a), if not already paid by Lessee, or, if already paid by Lessee, will be applied to reimburse Lessee for its payment of such Stipulated Loss Value and such other amounts, and (B) the balance, if any, of such payment remaining thereafter will be paid over to, or retained by, Lessee or its designee; and
  - (ii) if such payments are received with respect to an Event of Loss with respect to an Engine under the circumstances contemplated by <u>Section 8(d)</u>, such payments shall be paid over to, or retained by, Lessee or its designee; <u>provided</u> that in the case of an Engine with respect to which an Event of Loss shall have occurred or shall have been deemed to have occurred pursuant to <u>Section 7(b)</u> or <u>Section 10(d)</u> Lessee shall have fully performed the terms of <u>Section 8(d)</u> with respect to the Event of Loss for which such payments are made.

In all events, the insurance payment for any loss or damage to the Aircraft in excess of the Stipulated Loss Value for the Aircraft shall be paid to Lessee or its designee.

The insurance payments for any loss or damage to the Airframe or an Engine not constituting an Event of Loss with respect to the Airframe or such Engine will be applied in

payment (or to reimburse Lessee) for repairs or for replacement property in accordance with the terms of Section 7 and Section 8, and any balance remaining after compliance with such Sections with respect to such loss or damage shall be paid to Lessee or its designee. Any amount referred to in the preceding sentence or in clause (i) or (ii) of the second preceding paragraph which is payable to Lessee or its designee shall not be paid to Lessee or its designee (or, if it has been previously paid directly to Lessee, shall not be retained by Lessee) if at the time of such payment an Event of Default shall have occurred and be continuing, but shall be paid to and held by Lessor pursuant to Section 22, as security for the obligations of Lessee under this Lease, and at such time as there shall not be continuing any such Event of Default, such amount shall be paid to Lessee or its designee.

- (e) Reports, Etc. On or before the Delivery Date, and annually upon renewal of Lessee's insurance coverage, Lessee will furnish to each Specified Person a report signed by a firm of independent aircraft insurance brokers appointed by Lessee (which firm may be in the regular employ of Lessee), stating the opinion of such firm that the commercial hull and liability insurance then carried and maintained on the Aircraft complies with the terms hereof; provided that all information contained in such report shall be Confidential Information and shall be treated as such by each of the Specified Persons and their respective officers, directors, agents and employees in accordance with the provisions of Section 23. Lessee will cause such firm to agree to advise each Specified Person in writing of any default in the payment of any premium or of any other act or omission on the part of Lessee of which such firm has knowledge and that might invalidate or render unenforceable, in whole or in part, any insurance on the Aircraft. Lessee will also cause such firm to advise each Specified Person in writing as promptly as practicable after such firm acquires knowledge that an interruption of any insurance carried and maintained on the Aircraft pursuant to this Section will occur.
- (f) <u>Salvage Rights</u>; <u>Other</u>. All salvage rights to the Airframe and each Engine shall remain with Lessee's insurers at all times. Nothing in this Section shall limit or prohibit each Specified Person or Lessee from obtaining insurance for its own account, and at its sole expense, with respect to the Airframe or any Engine, and any proceeds payable under such insurance shall be payable as provided in the insurance policy relating thereto; <u>provided</u> that no such insurance may be obtained which would limit or otherwise adversely affect the coverage or amounts payable under, or increase the premiums for, any insurance required to be maintained pursuant to this Section or any other insurance maintained by Lessee (or, in the case of a sublease to a Permitted Sublessee, such Permitted Sublessee) with respect to the Aircraft or any other aircraft in the fleet of Lessee (or such Permitted Sublessee).

#### Section 12. Inspection.

(a) <u>Annual Inspection of Aircraft</u>. At all reasonable times during the Term (but not more than once annually unless an Event of Default has occurred and is continuing, in which case there shall be no restriction on the number of inspections), upon at least 10 days' prior written notice to Lessee from Lessor, Lessor or its authorized representative (together with any representative of a potential financing party, lessee or transferee, if applicable, referred to in <u>Section 12(b)</u>, the "Inspecting Party") may at its own expense (other than following the occurrence and during the continuance of an Event of Default, in which case the reasonable expenses of one inspection, as designated by the Lessor, shall be at the expense of Lessee) and

risk (including, without limitation, any risk of personal injury), conduct a non-intrusive, visual walk-around inspection of the Aircraft and any Engine that may include going on board the Aircraft and examining the contents of any open panels, bays or other components of the Aircraft (but shall not include the opening of any unopened panels, bays or other components) and, subject to Section 12(c), may inspect the books and records of Lessee relating to the Aircraft specified in Annex C; provided that (i) the Inspecting Party shall provide, prior to conducting any such inspection, assurances reasonably satisfactory to Lessee that it is fully insured with respect to any risks incurred in connection with any such inspection and, if requested by Lessee, a written release satisfactory to Lessee with respect to such risks; (ii) any such inspection shall be subject to the safety, security and workplace rules applicable at the location where such inspection is conducted and to the requirements of any applicable law; and (iii) all such inspections shall be conducted so as not to interfere with Lessee's business or the operation or maintenance of the Aircraft, and, in the case of an inspection during a maintenance visit, such inspection shall not in any respect interfere with the normal conduct of such maintenance visit or extend the time required for such maintenance visit (as determined by Lessee in its sole discretion).

Lessor shall not have any duty to make any such inspection and shall not incur any liability or obligation by reason of not making any such inspection. No inspection pursuant to this Section shall relieve Lessee of any of its obligations under this Lease. Lessee will, upon the request of Lessor at any time, notify Lessor of the time and location of the next scheduled heavy maintenance visit to be conducted by Lessee in respect of the Aircraft during the Term; provided that Lessee shall have the right in its sole discretion to reschedule, or change the location of, any heavy maintenance visit of which it shall have notified Lessor pursuant to this sentence, Lessee hereby agreeing to use reasonable efforts to notify Lessor of any such rescheduling or change.

- (b) <u>Marketing Inspection of Aircraft</u>. In addition to the annual inspection described in <u>Section 12(a)</u>, but subject to the other conditions and requirements for inspections set forth in <u>Section 12(a)</u>, upon at least 10 days' prior written notice to Lessee and during times reasonably acceptable to Lessee, in connection with a proposed financing, lease or transfer of the Aircraft or of the Lease or Owner Participant's interest therein (including the Trust Estate), Lessor or Owner Participant or their respective authorized representatives, and up to two representatives of a potential financing party, lessee or transferee, if applicable, may inspect the Airframe and any Engines installed thereon and, unless Owner Participant has requested electronic records pursuant to <u>Section 12(c)</u>, the books and records of Lessee relating thereto specified in <u>Annex C</u> (any such inspection, a "**Marketing Inspection**"); <u>provided</u> that there shall be no more than two Marketing Inspections in any year. The identity of any potential financing party, lessee or transferee shall be held confidential by Lessee in manner consistent with the terms of <u>Section 10.4</u> of the Participation Agreement.
- (c) <u>Electronic Records</u>. In lieu of the annual physical inspection of the books and records referred to in <u>subsection (a)</u> or physical inspection of the books and records referred in <u>subsection (b)</u>, during the Term (but not more than three times annually) Lessor may request that Lessee provide to Lessor some or all of the books and records relating to the Aircraft that are available and are indicated in <u>Annex C</u> as being transmissible in electronic form, and Lessee shall provide such documents in electronic form within 30 days of such request to Lessor.

- (d) <u>Confidentiality</u>. All information obtained from Lessee in electronic form or in connection with any inspection shall be Confidential Information and shall be held by Lessor, Owner Participant and any Inspecting Party in accordance with the provisions of <u>Section 23</u>.
- (e) <u>Compliance</u>. Notwithstanding anything to the contrary in this Lease, in no event shall Lessee be required to permit Lessor, Owner Participant or any Inspecting Party to inspect any portion of the Aircraft or any Engine that Lessee would be prohibited from showing to such Person pursuant to the Export Administration Regulations or any other applicable law or to disclose to any such Person any information with respect to the Aircraft or any Engine that Lessee would be prohibited from disclosing to such Person pursuant to the Export Administration Regulations or any other applicable law.
- Section 13. <u>Assignment</u>. Except as expressly permitted by the Participation Agreement and this Lease, Lessee will not, without the prior written consent of Lessor, such consent not to be unreasonably withheld, Transfer any of its rights or obligations hereunder. Except as expressly permitted by the Participation Agreement, this Lease and Article IX of the Trust Agreement, Lessor will not, without the prior written consent of Lessee, Transfer any of its right, title and interest in and to this Lease or the Aircraft. The terms and provisions of this Lease shall be binding upon and inure to the benefit of Lessor and Lessee and their respective successors and permitted assigns.
- **Section 14.** Events of Default. The following events shall constitute Events of Default (whether any such event shall be voluntary or involuntary or come about or be effected by operation of law or pursuant to or in compliance with any judgment, decree or order of any court or any order, rule or regulation of any administrative or governmental body) and each such Event of Default shall be deemed to exist and continue so long as, but only so long as, it shall not have been remedied or waived:
- (a) Lessee shall fail to make any payment of Basic Rent (other than the payment of Basic Rent due pursuant to Section U of the Return Conditions) or Stipulated Loss Value within five Business Days after such payment shall be or become due; or
- (b) Lessee shall fail to make (i) any payment of Basic Rent due pursuant to Section U of the Return Conditions or (ii) any other payment of Supplemental Rent (including, without limitation, indemnity payments) hereunder (other than those described in <u>subsection (a)</u> above), in either case at the time required to be paid hereunder, and any such failure shall continue unremedied for a period of 10 Business Days after receipt by Lessee of written notice of such failure by Lessor; or
- (c) Lessee shall fail to carry and maintain insurance on or with respect to the Aircraft in accordance with the provisions of Section 11; provided that, in the case of insurance with respect to which cancellation, change or lapse for nonpayment of premium shall not be effective as to Lessor or Owner Participant for 30 days (seven days, or such other period as may from time to time be customarily obtainable in the industry, in the case of any war risk or allied perils coverage) after receipt of notice by Lessor or Owner Participant, as the case may be, of such cancellation, change or lapse, no such failure to carry and maintain insurance shall

constitute an Event of Default until the earlier of (i) the date such failure shall have continued unremedied for a period of 20 days (five days in the case of any war risk or allied perils coverage) after receipt by Lessor or Owner Participant, as the case may be, of the notice of cancellation, change or lapse referred to in Section 11(a)(i)(C) or Section 11(b)(i)(C) or (ii) the date on which such insurance is not in effect as to Lessor or Owner Participant; or

- (d) Lessee shall fail to perform or observe any other material covenant, condition or agreement not specified elsewhere in this Section 14 to be performed by it under any Operative Document to which Lessee is a party, and such failure in any such case shall continue unremedied for a period of 30 days after receipt by Lessee of written notice thereof by Lessor; provided that, if such failure is capable of being remedied, no such failure shall constitute an Event of Default hereunder for a period of 120 days (or, if such failure relates to the performance or observance of any such covenant, condition or agreement contained in Section 7(a), 8(a), 8(b) or 8(c), 180 days) after receipt of such notice so long as Lessee is diligently proceeding to remedy such failure; or
- (e) any representation or warranty made by Lessee in any Operative Document to which Lessee is a party or in any document or certificate furnished by Lessee to Lessor pursuant to the terms of any thereof shall prove to have been incorrect in any material respect at the time made, and such incorrectness shall continue to be material and shall continue to be unremedied for a period of 30 days after receipt by Lessee of written notice thereof by Lessor; or
- (f) Lessee shall consent to the appointment of a receiver, trustee or liquidator of itself or of a substantial part of its property or Lessee shall admit in writing its inability to pay its debts generally as they come due, or shall make a general assignment for the benefit of creditors; or
- (g) Lessee shall file a voluntary petition in bankruptcy or a voluntary petition or an answer seeking reorganization in a proceeding under any bankruptcy laws (as in effect at such time) or any answer admitting the material allegations of a petition filed against Lessee in any such proceeding, or Lessee shall, by voluntary petition or answer, consent to or seek relief under the provisions of any bankruptcy or other similar law (as in effect at such time) providing for the reorganization or winding-up of corporations, or providing for an agreement, composition, extension or adjustment with its creditors; or
- (h) an order, judgment, or decree shall be entered by any court of competent jurisdiction appointing, without the consent of Lessee, a receiver, trustee or liquidator of Lessee or of any substantial part of its property, or sequestering any substantial part of the property of Lessee, and any such order, judgment or decree of appointment or sequestration shall remain in force undismissed, unstayed or unvacated for a period of 90 days after the date of entry thereof; or
- (i) a petition against Lessee in a proceeding under the federal bankruptcy laws or other insolvency laws (as in effect at such time) shall be filed and shall not be withdrawn or dismissed within 90 days thereafter, or, under the provisions of any law providing for reorganization or winding-up of corporations which may apply to Lessee, any court of competent

jurisdiction shall assume jurisdiction, custody or control of Lessee or of any substantial part of its property and such jurisdiction, custody or control shall remain in force unrelinquished, unstayed or unterminated for a period of 90 days; or

(j) an "Event of Default" under a Related Lease, if any, shall have occurred and be continuing.

<u>provided</u> that, notwithstanding anything to the contrary contained in this Lease, any failure of Lessee to perform or observe any covenant, condition, or agreement herein shall not constitute an Event of Default if such observance is prevented solely by reason of an event referred to in the definition of Event of Loss so long as Lessee is continuing to comply with the applicable terms of <u>Section 10</u>.

**Section 15.** Remedies. Upon the occurrence of an Event of Default and at any time thereafter so long as the same shall be continuing, Lessor may, at its option, declare this Lease to be in default by a written notice to Lessee (provided that this Lease shall be deemed to have been declared in default without the necessity of such written notice upon the occurrence of any Event of Default described in Section 14(g), (h) or (i)); and at any time thereafter, so long as Lessee shall not have remedied all outstanding Events of Default, Lessor may do one or more of the following, as Lessor shall elect, to the extent permitted by, and subject to compliance with any mandatory requirements of, applicable law; provided that during any period the Aircraft is subject to the CRAF Program in accordance with the provisions of Section 7(b) and in the possession of the U.S. government or an instrumentality or agency thereof, Lessee shall not, on account of any Event of Default, be required to do any of the following in such manner as to limit Lessee's operational control under this Lease (or any sublessee's operational control under any sublease permitted by the terms of this Lease) of the Airframe or Engines, unless at least 60 days' (or such other period as may then be applicable under the Air Mobility Command Program of the U.S. government) prior notice of default hereunder shall have been given by Lessor by registered or certified mail to Lessee (or any sublessee) with a copy addressed to the Contracting Office Representative for the Air Mobility Command of the U.S. Air Force under any contract with Lessee (or any sublessee) relating to the Aircraft:

(a) cause Lessee, upon the written demand of Lessor and at Lessee's expense, to return promptly, and Lessee shall return promptly, all or such part of the Airframe and any Engines as Lessor may so demand, to Lessor in the manner and condition required by, and otherwise in accordance with all of the provisions of, Section 5 as if the Airframe or such Engines were being returned at the end of the Term; or Lessor, at its option, after Lessee shall have failed to so return the Aircraft after such demand, may enter upon the premises where the Airframe is or any or all Engines are located or reasonably believed to be located and, without breach of peace, take immediate possession of and remove such Airframe or Engines (together with any engine which is not an Engine but which is installed on the Airframe, subject to all of the rights of the owner, lessor, lienor or secured party of such engine; provided that, in the event that an engine (which is not an Engine) is installed on the Airframe, such engine shall be held for the account of any such owner, lessor, lienor or secured party or, if owned by Lessee, may, at the option of Lessee with the consent of Lessor (which consent shall not be unreasonably withheld), be exchanged with Lessee for an Engine in accordance with the Return Conditions), by summary proceedings or otherwise, all without liability to Lessor for or by reason of such entry or taking possession; or

- (b) sell all or any part of the Airframe and any Engine at public or private sale, whether or not Lessor shall at the time have possession thereof, as Lessor may determine, or otherwise dispose of, hold, use, operate, lease to others or keep idle all or any part of the Airframe or such Engine as Lessor, in its sole discretion, may determine, free and clear of any rights of Lessee; or
- (c) whether or not Lessor shall have exercised, or shall thereafter at any time exercise, any of its rights under <u>clause (a)</u> or <u>clause (b)</u> above with respect to all of any part of the Airframe or any Engine, Lessor, by written notice to Lessee, may cause Lessee to pay to Lessor, and Lessee shall pay to Lessor, on a payment date that is at least 15 days from the date of such written notice (such payment date, the "**Specified Payment Date**"), as liquidated damages for loss of a bargain and not as a penalty (in lieu of the Basic Rent due for Lease Period Dates occurring on and after the Specified Payment Date):
  - (i) any unpaid Basic Rent due on Lease Period Dates prior to the Specified Payment Date, provided that (x) if the Specified Payment Date is a Lease Payment Date, Lessee shall have no obligation to pay the installment of Basic Rent that would otherwise be due and payable on the Lease Period Date that is the Specified Payment Date and (y) if the Specified Payment Date is not a Lease Payment Date, Lessee shall be entitled to credit against its payment obligations in this subsection (c) the portion of the installment of Basic Rent allocable to the period from (and including) such Specified Payment Date to (but not including) the next succeeding Lease Period Date, or if no Lease Period Date succeeds such specified Payment Date, the last day of the Term; plus
  - (ii) an amount equal to the excess, if any, of the Stipulated Loss Value for the Aircraft computed as of the Reference Stipulated Loss Value Determination Date, over the amount determined as provided in <u>clause (A)</u> or <u>(B)</u> below, as applicable (whether to use the amount determined as provided in <u>clause (A)</u> or in <u>clause (B)</u> shall have been specified in such written notice by Lessor, in its sole discretion):
    - (A) the sum of (x) the Fair Market Rental Value of the Aircraft for the remainder of the useful life of the Aircraft, after discounting such Fair Market Rental Value to present value as of the Specified Payment Date at an annual rate equal to 4% and (y) the salvage value of the Aircraft at the end of its useful life (as such salvage value is determined by mutual written agreement between Lessor and Lessee or, in the absence of mutual written agreement, pursuant to an Independent Appraisal) after discounting such salvage value to the present value as of the Specified Payment Date at an annual rate equal to 4%, or
      - (B) the Fair Market Sales Value of the Aircraft determined as of the Specified Payment Date; or

- (d) in the event Lessor, pursuant to <u>clause (b)</u> above, shall have sold the Aircraft, Lessor, in lieu of exercising its rights under <u>clause (c)</u> above with respect to the Aircraft, by written notice to Lessee, may cause Lessee to pay to Lessor, and Lessee shall pay to Lessor, on the fifth day following the date of such sale (such fifth day, the "**Sale Date**"), as liquidated damages for loss of a bargain and not as a penalty (in lieu of the Basic Rent due on Lease Period Dates occurring on and after the Sale Date):
  - (i) any unpaid Basic Rent due on Lease Period Dates prior to the Sale Date; <u>provided</u> that (x) if the Sale Date is a Lease Period Date, Lessee shall have no obligation to pay the installment of Basic Rent that would otherwise be due and payable on the Lease Period Date that is the Sale Date and (y) if the Sale Date is not a Lease Period Date, Lessee shall be entitled to credit against its payment obligations in this <u>subsection (d)</u> the portion of the installment of Basic Rent allocable to the period from (and including) such Sale Date to (but not including) the next succeeding Lease Period Date, or if no Lease Period Date succeeds such Sale Date, the last day of the Term; plus
  - (ii) (A) if such sale is a public or private sale to a purchaser that is not an Affiliate of Owner Participant, the Stipulated Loss Value for the Aircraft, computed as of the Reference Stipulated Loss Value Determination Date, minus the net proceeds of such sale (after deduction of all actual and reasonable out-of-pocket costs of such sale) or (B) if such sale is a public or private sale to an Affiliate of Owner Participant, the Stipulated Loss Value for the Aircraft, computed as of the Reference Stipulated Loss Value Determination Date, minus the Fair Market Sales Value of the Aircraft, determined as of the Sale Date; or
- (e) rescind this Lease as to the Aircraft, or exercise any other right or remedy which may be available to it under applicable law or proceed by appropriate court action, either at law or in equity, to enforce the terms or to recover damages for the breach hereof.

In addition, to the extent permitted by applicable Law, Lessee shall be liable, except as otherwise provided above, and without duplication of amounts payable hereunder, for any and all unpaid Rent due hereunder before or during the exercise of any of the foregoing remedies and for all reasonable legal fees and other actual and reasonable costs and expenses incurred by Lessor or Owner Participant by reason of the occurrence of any Event of Default or the exercise of Lessor's remedies with respect thereto, including all costs and expenses incurred in connection with the return of the Airframe or any Engine in accordance with the Return Conditions or in placing such Airframe or Engine in the condition and airworthiness required by the Return Conditions (provided that, for the avoidance of doubt, Lessee shall not be liable for any amounts or obligations with respect to Return Conditions if Lessor exercises any remedy under subsection (c) or (d)). At any sale of the Airframe or an Engine or part thereof pursuant to this Section, Lessor or Owner Participant may bid for and purchase such property. Lessor agrees to give Lessee at least 30 days' prior written notice of the date fixed for any public sale of the Airframe or any Engine or of the date on or after which any private sale will be held and of any lease or other disposition of the Aircraft, which notice Lessee hereby agrees to the extent permitted by applicable law, no remedy referred to in this Section is intended to be exclusive, but each shall be cumulative and in addition to any other remedy referred to above or

otherwise available to Lessor at law or in equity; and, to the extent permitted by applicable law, the exercise or beginning of exercise by Lessor of any one or more of such remedies shall not preclude the simultaneous or later exercise by Lessor of any or all of such other remedies. To the extent permitted by applicable law, no express or implied waiver by Lessor of any Event of Default shall in any way be, or construed to be, a waiver of any future or subsequent Event of Default.

Notwithstanding anything to the contrary set forth herein or in any other Operative Document, but subject to the next sentence (i) as permitted by Article 15 of the Cape Town Convention, the provisions of Chapter III of the Cape Town Convention are hereby excluded and made inapplicable to this Lease and the other Operative Documents, except for those provisions of such Chapter III that cannot be derogated from and (ii) as permitted by Article IV(3) of the Aircraft Protocol, the provisions of Chapter II of the Aircraft Protocol are hereby excluded and made inapplicable to this Lease and the other Operative Documents, except for (x) Article XVI of the Aircraft Protocol and (y) those provisions of such Chapter II that cannot be derogated from. The parties agree that the exercise of remedies hereunder and the other Operative Documents is subject to other applicable law, including without limitation, the UCC (as in effect in the State of New York) and the Bankruptcy Code, and that nothing herein derogates from the rights of Lessor or Lessee under or pursuant to such other applicable law.

Section 16. Further Assurances. Forthwith upon the execution and delivery of each Lease Supplement from time to time required by the terms hereof, Lessee will cause such Lease Supplement (and, in the case of Lease Supplement No. 1, this Lease) to be duly filed and recorded in accordance with the Transportation Code or the applicable Laws of such jurisdiction other than the United States in which the Aircraft is registered, as the case may be. In addition, each of Lessor and Lessee will promptly and duly execute and deliver to the other such further documents and assurances and take such further action as may from time to time be reasonably requested in order more effectively to carry out the intent and purpose of this Lease including, without limitation, if requested by Lessee or Lessor, the execution and delivery of supplements or amendments hereto, each in recordable form, subjecting to this Lease any Replacement Engine and the recording or filing of counterparts thereof; provided that this sentence is not intended to impose upon Lessee any additional liabilities not otherwise contemplated by this Lease.

**Section 17.** <u>Notices</u>. Unless otherwise expressly specified or permitted by the terms hereof, all notices, requests, demands, authorizations, directions, consents or waivers required or permitted under the terms and provisions of this Lease shall be in English and in writing, and given by United States registered or certified mail, return receipt requested, postage prepaid, overnight courier service or facsimile, and any such notice shall be effective when received (or, if delivered by facsimile, upon completion of transmission and confirmation by the sender (by a telephone call to a representative of the recipient or by machine confirmation) that such transmission was received) and addressed as follows:

#### (a) if to Lessee:

American Airlines, Inc. 4333 Amon Carter Boulevard, MD 5662 Fort Worth, Texas 76155

Attention: Treasurer Facsimile: 817.967.4318 Telephone: 817.963.1234

#### (b) if to Lessor:

Wells Fargo Bank Northwest, National Association

MAC: U1228-120

299 South Main Street, 12th Floor

Salt Lake City, UT 84111

Attention: Corporate Trust Services

Facsimile: 801.246.5053 Telephone: 801.246.2755

#### (c) if to Owner Participant:

[Name of Owner Participant] [Address of Owner Participant]

Attention:

Facsimile: Telephone:

Any party, by notice to the other parties hereto, may designate different addresses for subsequent notices or communications. Whenever the words "notice" or "notify" or similar words are used herein, they mean the provision of formal notice as set forth in this <u>Section 17</u>.

Section 18. No Set-Off, Counterclaim, etc. This Lease is a net lease and to the full extent permitted by applicable law, Lessee's obligation to pay all Basic Rent and Stipulated Loss Value shall be absolute and unconditional and shall not be affected by any circumstance, including, without limitation:

- (a) any set-off, counterclaim, recoupment, defense or other right which Lessee may have against Lessor, Owner Participant or any other Person for any reason whatsoever;
  - (b) any defect in the title, airworthiness, condition, design, operation, or fitness for use of, or any damage to or loss or destruction of, the Aircraft;
  - (c) any insolvency, bankruptcy, reorganization or similar proceedings by or against Lessee or any Permitted Sublessee or any other Person; or
  - (d) any other circumstances, happening or event whatsoever, whether or not unforeseen or similar to any of the foregoing.

Lessee hereby waives, to the full extent permitted by applicable law, any and all rights which it may now have or which at any time hereafter may be conferred upon it, by statute or otherwise, to terminate, cancel, quit or surrender this Lease except in accordance with the express terms hereof. Nothing contained in this Section shall be construed to waive any claim which Lessee may have hereunder (including, without limitation, claims that Basic Rent, Stipulated Loss Value or any other payments demanded from or paid by Lessee are or were erroneous) or otherwise or to limit the right of Lessee to make any claim it may have against Lessor, Owner Participant or any other Person or to pursue any such claim in such manner as Lessee shall deem appropriate.

**Section 19.** Section 1110. It is the intention of the parties hereto that this Lease, to the fullest extent available under applicable law, entitles Lessor to the benefits of Section 1110 with respect to the Aircraft. In furtherance of the foregoing, Lessor and Lessee hereby confirm that this Lease is to be treated as a lease for U.S. federal income tax purposes. Nothing contained in this paragraph shall be construed to limit Lessee's use and operation of the Aircraft under this Lease or constitute a representation or warranty by Lessee as to tax consequences.

**Section 20.** <u>Monies Received by Lessor</u>. Except as otherwise provided herein, any monies received by Lessor in excess of the amounts to which Lessor is entitled pursuant to the terms hereof shall immediately be paid over by Lessor to Lessee.

Section 21. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Section 22. Investment of Security Funds. Any moneys which are (a) held by Lessor pursuant to the terms hereof, (b) required to be paid to or retained by Lessor and not required to be paid to Lessee pursuant to Section 10(e) or Section 11(d) solely because an Event of Default shall have occurred, or (c) required to be paid to Lessee pursuant to Section 10(b) or Section 11(d) after completion of a replacement to be made pursuant to Section 8(d) shall, until paid to Lessee as provided in Section 10 or Section 11, be invested in Permitted Investments by Lessor from time to time as directed in writing by Lessee. There shall, so long as no Event of Default shall have occurred and be continuing, be promptly remitted to Lessee any gain (including interest received) realized as the result of any such investment (net of any fees, commissions and other expenses, if any, incurred in connection with such investment), and Lessee will promptly pay to Lessor, on demand, the amount of any loss realized as the result of any such investment (together with any fees, commissions and other expenses, if any, incurred in connection with such investment).

**Section 23.** <u>Confidential Information</u>. All Confidential Information shall be held confidential by Lessor in accordance with <u>Section 10.4</u> of the Participation Agreement.

**Section 24.** <u>Lessor Right to Perform for Lessee</u>. If Lessee fails to make any payment of Rent required to be made by it hereunder, Lessor may, on behalf of Lessee and upon prior notice to Lessee, itself make such payment. The amount of any such payment and the amount of the reasonable expenses of Lessor incurred in connection with such payment shall be deemed Supplemental Rent immediately due and payable as of and when such payment is made by Lessor.

Section 25. Lessee's Performance and Rights. Any obligation imposed on Lessee in this Lease shall require only that Lessee perform or cause to be performed such obligation, even if stated herein as a direct obligation, and the performance of any such obligation by a permitted assignee, sublessee or transferee under an assignment, sublease or transfer agreement then in effect shall constitute performance by Lessee and to the extent of such performance in accordance with the terms of the applicable assignment, sublease or transfer agreement, discharge such obligation by Lessee. Except as otherwise expressly provided in this Lease, any right granted to Lessee in this Lease shall grant Lessee the right to exercise such right or permit such right to be exercised by such assignee, sublessee or transferee with the same force and effect as if such assignee, sublessee or transferee were named as "Lessee" herein. The inclusion of specific references to obligations or rights of any such assignee, sublessee or transferee in certain provisions of this Lease shall not in any way prevent or diminish the application of the provisions of the two sentences immediately preceding with respect to obligations or rights in respect of which specific reference to any such assignee, sublessee or transferee has not been made in this Lease.

Section 26. Concerning Lessor. Wells Fargo Bank Northwest, National Association is entering into the Operative Documents solely in its capacity as Owner Trustee under the Trust Agreement and not in its individual capacity (except as expressly provided in the Operative Documents) and in no case shall Wells Fargo Bank Northwest, National Association (or any entity acting as successor Owner Trustee under the Trust Agreement) be personally liable for or on account of any of the statements, representations, warranties, covenants or obligations stated to be those of Lessor under the Operative Documents; provided, however, that Wells Fargo Bank Northwest, National Association (or any such successor Owner Trustee) shall be personally liable under the Operative Documents for its own gross negligence, its own simple negligence in the handling of funds actually received by it in accordance with the terms of the Operative Documents, its willful misconduct and its breach of its covenants, representations and warranties in the Operative Documents, to the extent covenanted or made in its individual capacity or as otherwise expressly provided in the Operative Documents; provided, further, that nothing contained in this Section 26 shall be construed to limit the exercise and enforcement in accordance with the terms of the Operative Documents of rights and remedies against the Trust Estate.

**Section 27.** Successor Owner Trustee. Lessee agrees that, in the case of the appointment of any successor Owner Trustee pursuant to the terms of the Trust Agreement and Section 6.2.2 of the Participation Agreement, such successor Owner Trustee shall, upon written notice to Lessee by such successor Owner Trustee, succeed to all the rights, powers and title of Lessor hereunder and shall be deemed to be Lessor of the Aircraft for all purposes without in any way altering the terms of this Lease or Lessee's obligations hereunder. One such appointment and designation of a successor Owner Trustee shall not exhaust the right to appoint and designate further successor Owner Trustees pursuant to the Trust Agreement and Section 6.2.2 of the Participation Agreement, but such right may be exercised repeatedly as long as this Lease shall be in effect.

#### Section 28. Miscellaneous.

- (a) Any provision of this Lease which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof or affecting the validity or enforceability of such provision in any other jurisdiction.
- (b) No term or provision of this Lease may be amended, modified or supplemented orally, but only by an instrument in writing signed by the party against which the enforcement of the amendment, modification or supplement is sought.
- (c) This Lease and the other Operative Documents, and all certificates, instruments and other documents relating thereto delivered and to be delivered from time to time pursuant to the Operative Documents, supersede any and all representations, warranties and agreements (other than any Operative Document) prior to the date of this Lease, written or oral, between or among any of the parties hereto relating to the transactions contemplated hereby and thereby.
- (d) This Lease may be executed in any number of counterparts (and each of the parties hereto shall not be required to execute the same counterpart). Each counterpart of this Lease, including a signature page executed by each of the parties hereto shall be an original, but all of such counterparts together shall constitute one instrument. In the event that a security interest is granted in this Lease with respect to the issuance of debt by Lessor to the extent permitted by Section 8.3 of the Participation Agreement, and that this Lease constitutes chattel paper (as such term is defined in the UCC), no security interest in this Lease may be created through the transfer or possession of any counterpart hereof other than the original counterpart, which shall be identified as the counterpart containing the receipt therefor executed by Lessor on the signature page thereof.
- (e) The parties hereto do not intend any interest created by this Lease to be a perpetuity or to be subject to invalidation under any applicable perpetuities rule; however, if the rule is to be applied, then the perpetuities period shall be 21 years after the last to die of the currently living descendents of former United States President John F. Kennedy.
- (f) THIS LEASE SHALL IN ALL RESPECTS BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK, INCLUDING ALL MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE. THIS LEASE HAS BEEN DELIVERED IN THE STATE OF NEW YORK.

IN WITNESS WHEREOF, the parties have each caused this Lease to be duly executed as of the day and year first above written. [Receipt of the original counterpart of the foregoing Lease is hereby acknowledged on this day of WELLS FARGO BANK NORTHWEST, NATIONAL **ASSOCIATION,** not in its individual capacity but solely as Owner Trustee By: Name:

Title:

]4 For chattel paper copy only. WELLS FARGO BANK NORTHWEST, NATIONAL **ASSOCIATION,** not in its individual capacity (except as expressly provided herein) but solely as Owner Trustee

By: Jame: Vitle:				
AMERICAN AIRLINES, INC.				
sy:				
By:				
ïtle:				

### EXHIBIT A TO LEASE AGREEMENT ([YEAR] MSN [MSN])

LEASE SUPPLEMENT	NO. ([YEAR] MSN [MSN]), dated	, 20	, between WELLS FARGO BANK NORTHWEST, NATIONAL
ASSOCIATION, a national ban	king association, not in its individual capacity, t	but solely	as Owner Trustee under the Trust Agreement ([YEAR] MSN [MSN]),
dated as of [	], [YEAR], between the Owner Participant nam	ned therei	in and Wells Fargo Bank Northwest, National Association ("Lessor"), and
AMERICAN AIRLINES, INC.	, a Delaware corporation ("Lessee").		

#### **RECITALS:**

- 1. Lessor and Lessee have heretofore entered into that certain Lease Agreement ([YEAR] MSN [MSN]), dated as of [ ], [YEAR] (herein called, as at any time modified, supplemented or amended, the "Lease Agreement" and the defined terms in Annex A thereto being hereinafter used with the same meanings), providing for the execution and delivery from time to time of Lease Supplements, each substantially in the form hereof for the purpose of leasing specific aircraft and engines under the Lease Agreement as and when delivered by Lessor to Lessee in accordance with the terms thereof;
- 2. [The Lease Agreement relates to the aircraft and engines described below, and counterparts of the Lease Agreement are attached hereto and made a part hereof, and this Lease Supplement, together with such attachments, is being filed for recordation on the date hereof with the Federal Aviation Administration as one document.]<sup>1</sup>

[A counterpart of the Lease Agreement, attached to and made a part of Lease Supplement No. 1, dated [ ], 20 , to the Lease Agreement, has been recorded by the Federal Aviation Administration on , as one document and assigned Conveyance No.  $.]^2$ 

NOW, THEREFORE, in consideration of the premises and mutual agreements herein contained, the agreements contained in the other Operative Documents and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

1. Lessor hereby delivers and leases to Lessee under the Lease Agreement, and Lessee hereby accepts and leases from Lessor under the Lease Agreement, the following [described Aircraft, which Aircraft as of the date hereof consists of the following components]:<sup>3</sup>

Exhibit A

Page 1

<sup>&</sup>lt;sup>1</sup> This language for Lease Supplement No. 1.

This language for other Lease Supplements.

Only for Lease Supplement No. 1.

[(a) one Airbus [Model] (Generic Manufacturer and Model AIRBUS [Generic Model]) airframe: U.S. Registration Number ; Manufacturer's Serial No. ; and4

(b) two (2) [INSERT ENGINE INFO]) engines relating to such airframe and bearing, respectively, Manufacturer's Serial Nos. and , respectively (each of which engines has 550 or more rated takeoff horsepower or the equivalent of such horsepower and is a jet propulsion aircraft engine having at least 1750 pounds of thrust or the equivalent of such thrust).

The Basic Term for the lease of the Aircraft shall commence on the date of this Lease Supplement (the "**Delivery Date**") and shall end on (the "**Lease Expiry Date**"), or such earlier date on which the Lease is terminated in accordance with the provisions thereof.

The amount of Basic Rent for the Aircraft is set forth in  $\underline{\text{Schedule } A}$  hereto.

The Stipulated Loss Values for the Aircraft are set forth in Schedule B hereto.]

[Add description of Replacement Engine or Engines, if applicable].

- 2. All of the terms and provisions of the Lease Agreement are hereby incorporated by reference in this Lease Supplement to the same extent as if fully set forth herein.
- 3. This Lease Supplement may be executed in any number of counterparts (and each of the parties hereto shall not be required to execute the same counterpart). All of such counterparts together shall constitute one instrument.

TO THE EXTENT, IF ANY, THAT THIS LEASE SUPPLEMENT CONSTITUTES CHATTEL PAPER (AS DEFINED IN THE UNIFORM COMMERCIAL CODE AS IN EFFECT IN ANY APPLICABLE JURISDICTION), NO SECURITY INTEREST IN THIS LEASE SUPPLEMENT MAY BE PERFECTED THROUGH DELIVERY OR POSSESSION OF ANY COUNTERPART OF THIS LEASE SUPPLEMENT OTHER THAN THE ORIGINAL COUNTERPART, WHICH SHALL BE THE COUNTERPART THAT CONTAINS THE RECEIPT EXECUTED BY LESSOR ON THE SIGNATURE PAGE THEREOF.

Exhibit A Page 2

Only for Lease Supplement No. 1.

Insert tenth (10th) anniversary of Delivery Date.

<sup>6</sup> Language for other Lease Supplements.

	WELLS FARGO BANK NORTHWEST, NATIONAL ASSOCIATION, not in its individual capacity (except as expressly provided herein) but solely as Owner Trustee
	By:
	Name:
	Title:
	AMERICAN AIRLINES, INC.
	By:
	Name:
	Title:
[Receipt of the original counterpart of the foregoing lease is hereby acknowledged on this $\mbox{\ \ day of\ \ \ }$ .	
WELLS FARGO BANK NORTHWEST, NATIONAL ASSOCIATION, not in its individual capacity but solely as Owner Trustee	
By:	
Name:	
Title: ] <sup>7</sup>	
For chattel paper copy only.	

written.

IN WITNESS WHEREOF, Lessor and Lessee have each caused this Lease Supplement No. to be duly executed as of the day and year first above

Exhibit A Page 3

#### SCHEDULE A TO LEASE SUPPLEMENT NO. 1 ([YEAR] MSN [MSN])

#### **BASIC RENT**

Lease Period Dates during the Term:

The Delivery Date and the [ ]th day of each calendar month occurring after the Delivery Date during the Term

]

(but not including the last day of the Term if such day is the [ ] th day of a calendar month)

Basic Rent during the Basic Term:

] per month during the through months following the Delivery Date, and thereafter \$[

per month, in each case, payable in advance.

Basic Rent during any Renewal Term: An amount per month determined in accordance with <u>Section 21</u> of the Lease, payable in advance.

Schedule A to Lease Supplement No. 1

#### SCHEDULE A TO LEASE SUPPLEMENT NO. 1 ([YEAR] MSN [MSN])<sup>1</sup>

# INTENTIONALLY DELETED FROM THE VERSION OF THIS DOCUMENT FILED WITH THE FAA AS CONTAINING CONFIDENTIAL AND PROPRIETARY INFORMATION

Insert for FAA filing in lieu of Schedule A.

Schedule A to Lease Supplement No. 1

### SCHEDULE B TO LEASE SUPPLEMENT NO. 1 ([YEAR] MSN [MSN])

### STIPULATED LOSS VALUES

Stipulated Loss Value Determination Date

Stipulated Loss Value

Schedule B

#### SCHEDULE B TO LEASE SUPPLEMENT NO. 1 ([YEAR] MSN [MSN])<sup>1</sup>

# INTENTIONALLY DELETED FROM THE VERSION OF THIS DOCUMENT FILED WITH THE FAA AS CONTAINING CONFIDENTIAL AND PROPRIETARY INFORMATION

Insert for FAA filing in lieu of Schedule B.

Schedule B

# ANNEX A TO LEASE AGREEMENT ([YEAR] MSN [MSN])

### **DEFINITIONS**

Annex A

### ANNEX B TO LEASE AGREEMENT([YEAR] MSN [MSN])

#### RETURN CONDITIONS

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# B. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### D. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# E. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# H. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# I. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

J. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

K. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### M. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# N. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### O. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### P. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### R. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### S. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

U. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

V. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# ANNEX B TO LEASE AGREEMENT ([YEAR] MSN [MSN]) $^{\scriptscriptstyle 1}$

# INTENTIONALLY DELETED FROM THE VERSION OF THIS DOCUMENT FILED WITH THE FAA AS CONTAINING CONFIDENTIAL AND PROPRIETARY INFORMATION

Insert for FAA filing in lieu of Annex B.

#### ANNEX C

## TO LEASE AGREEMENT ([YEAR] MSN [MSN]

## MID-TERM INSPECTION RECORDS LIST

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Annex C

# ANNEX C TO LEASE AGREEMENT ([YEAR] MSN [MSN]) $^{\scriptscriptstyle 1}$

# INTENTIONALLY DELETED FROM THE VERSION OF THIS DOCUMENT FILED WITH THE FAA AS CONTAINING CONFIDENTIAL AND PROPRIETARY INFORMATION

Insert for FAA filing in lieu of Annex C.

Annex C

EXHIBIT B

FORM OF TRUST AGREEMENT

# TRUST AGREEMENT ([YEAR] MSN [MSN])

dated as of [Date]

between

## [NAME OF OWNER PARTICIPANT],

as Owner Participant

and

## WELLS FARGO BANK NORTHWEST, NATIONAL ASSOCIATION

Covering One Airbus [Model] Aircraft (Generic Manufacturer and Model AIRBUS [Generic Model])

### TABLE OF CONTENTS

ARTICLE I DEFINITIONS	1
Section 1.01 Certain Definitions	1
ARTICLE II AUTHORITY TO EXECUTE CERTAIN OPERATIVE DOCUMENTS; DECLARATION OF TRUST	1
Section 2.01 Authority to Execute Documents Section 2.02 Declaration of Trust	1 2
ARTICLE III ACCEPTANCE AND DELIVERY OF AIRCRAFT; ISSUANCE OF CERTIFICATES; LEASE OF AIRCRAFT; REPLACEMENT	2
Section 3.01 Authorization Section 3.02 Conditions Precedent Section 3.03 Replacement or Return of an Engine	2 3 3
ARTICLE IV RECEIPT, DISTRIBUTION AND APPLICATION OF INCOME FROM THE TRUST ESTATE	4
Section 4.01 Distribution of Payments Section 4.02 Method of Payments	4 5
ARTICLE V DUTIES OF OWNER TRUSTEE	5
Section 5.01 Certain Notices and Requests for Instructions; Related Actions Section 5.02 Action Upon Instructions Section 5.03 Indemnification Section 5.04 No Duties Except as Specified in Operative Documents or Instructions Section 5.05 No Action Except Under Specified Documents or Instructions Section 5.06 Limitations on Activities	5 6 6 7 7
ARTICLE VI OWNER TRUSTEE	8
Section 6.01 Acceptance of Trust and Duties Section 6.02 No Representations or Warranties as to Certain Matters Section 6.03 No Segregation of Monies Required; Investment Thereof Section 6.04 Reliance Upon Certificates; Counsel and Agents Section 6.05 Not Acting in Individual Capacity Section 6.06 Fees; Compensation Section 6.07 Books and Records; Tax Returns	8 8 8 9 9
ARTICLE VII INDEMNIFICATION OF OWNER TRUSTEE BY OWNER PARTICIPANT	10
Section 7.01 Owner Participant to Indemnify Trust Company	10
ARTICLE VIII TRANSFER OF OWNER PARTICIPANT'S INTEREST	11

i

# TABLE OF CONTENTS (continued)

Section 8.01	Transfer of Interest	Page 11
ARTICLE IX SU	JCCESSOR OWNER TRUSTEES	11
Section 9.01	Resignation of Owner Trustee; Appointment of Successor	11
ARTICLE X SU	JPPLEMENTS AND AMENDMENTS TO TRUST AGREEMENT AND OTHER DOCUMENTS	12
Section 10.01 Section 10.02 Section 10.03 Section 10.04 Section 10.05	Supplements and Amendments Discretion as to Execution of Documents Absence of Requirements as to Form Distribution of Documents No Request Needed as to Lease Supplements	12 13 13 13 13
ARTICLE XI M	ISCELLANEOUS	13
Section 11.01 Section 11.02 Section 11.03 Section 11.04 Section 11.05 Section 11.06	Termination of Trust Agreement Owner Participant Has No Legal Title in Trust Estate Assignment, Sale, etc. of Aircraft Third Party Beneficiary Notices Miscellaneous	13 14 14 14 14 14
ARTICLE XII C	ERTAIN LIMITATIONS ON CONTROL	15
Section 12.01 Section 12.02 Section 12.03 Section 12.04 Section 12.05	Limitations on Control Discretion, Actions and Payments of Owner Trustee General Purpose Adverse Effect of Citizenship on Registration	15 16 16 16 16
	. ~	

## ANNEXES

ANNEX A — DEFINITIONS

#### TRUST AGREEMENT ([YEAR] MSN [MSN])

THIS TRUST AGREEMENT ([YEAR] MSN [MSN]), dated as of [ ], [YEAR] (as amended, modified or supplemented from time to time, this "Trust Agreement"), is between [NAME OF OWNER PARTICIPANT], a [jurisdiction and organization] (together with its successors and permitted assigns, the "Owner Participant"), and WELLS FARGO BANK NORTHWEST, NATIONAL ASSOCIATION, a national banking association (as Owner Trustee, together with its successors and permitted assigns in such capacity, "Owner Trustee", and in its individual capacity, together with its successors and permitted assigns in such capacity, "Trust Company").

#### **RECITALS:**

- 1. On the Delivery Date, Owner Trustee will purchase the Aircraft from Manufacturer and immediately following Owner Trustee's purchase of the Aircraft, Lessee will lease the Aircraft from Owner Trustee pursuant to the Lease Agreement ([YEAR] MSN [MSN]) (such Lease together with Lease Supplement No. 1, the "Lease").
- 2. Owner Participant desires to create a trust for the purposes of the acquisition of the Aircraft by Owner Trustee and the leasing of it to Lessee in accordance with the Lease.
  - 3. Trust Company is willing to accept the duties and obligations imposed hereby on the terms and conditions hereinafter set forth.

**NOW, THEREFORE**, in consideration of the premises and mutual agreements herein contained, the agreements contained in the other Operative Documents and the acceptance by Owner Trustee of the trust hereby created, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

#### ARTICLE I

#### **DEFINITIONS**

**Section 1.01** Certain Definitions. Unless the context otherwise requires, all capitalized terms used herein and not otherwise defined herein shall have the respective meanings set forth, and shall be construed and interpreted in the manner described, in Annex A for all purposes of this Trust Agreement.

#### ARTICLE II

# AUTHORITY TO EXECUTE CERTAIN OPERATIVE DOCUMENTS; DECLARATION OF TRUST

Section 2.01 <u>Authority to Execute Documents</u>. Owner Participant hereby authorizes and directs Owner Trustee (<u>a</u>) to execute and deliver the Participation Agreement, the Lease, Lease Supplement No. 1 and any other agreements, instruments or documents in the respective forms thereof in which delivered from time to time by Owner Participant to Owner Trustee for execution and delivery, (<u>b</u>) to execute and deliver all other agreements, instruments and

certificates contemplated by the Operative Documents and (<u>c</u>) subject to the terms hereof, to exercise its rights (upon instructions received from Owner Participant) and perform its duties under the documents referred to in <u>clauses (a)</u> and (<u>b</u>) in accordance with the terms thereof.

**Section 2.02** <u>Declaration of Trust</u>. Trust Company hereby declares that it will hold, in its capacity as Owner Trustee, the Trust Estate upon the trust hereinafter set forth for the use and benefit of Owner Participant, subject, however, to the provisions of the Lease and the other Operative Documents. The name of the trust created hereby shall be "MSN [MSN] Trust" and such name may (but need not) be used in any correspondence and filings made by Owner Trustee in connection with the trust created hereby.

#### ARTICLE III

# ACCEPTANCE AND DELIVERY OF AIRCRAFT; ISSUANCE OF CERTIFICATES; LEASE OF AIRCRAFT; REPLACEMENT

**Section 3.01** <u>Authorization</u>. Owner Participant hereby authorizes and directs Owner Trustee to, and Owner Trustee agrees for the benefit of Owner Participant that, on or prior to the Delivery Date, it will, subject to due compliance with the terms of <u>Section 3.02</u>:

- (a) execute and deliver each of the Operative Documents to which it is a party;
- (b) purchase the Aircraft and accept from Manufacturer the Bills of Sale therefor;
- (c) authorize the financing statements contemplated by <u>Section 4.1.9</u> of the Participation Agreement;
- (d) make application to the FAA for registration of the Aircraft in the name of Owner Trustee by filing or causing to be filed (i) the FAA Bill of Sale, (ii) the Application for Aircraft Registration with the FAA (together with, without limitation, an affidavit from Owner Trustee stating that it is a Citizen of the United States) and (iii) this Trust Agreement;
  - (e) cause the Aircraft to be leased to Lessee under the Lease;
- (f) take such other action as may be required of Owner Trustee under the Operative Documents to effectuate the transactions contemplated thereby; and
- (g) execute and deliver all such other instruments, documents or certificates and take all such other actions as may be requested of Owner Trustee to effectuate the transactions contemplated under the Operative Documents, and take all other actions in accordance with the directions of Owner Participant as Owner Participant may deem necessary or advisable in connection with the transactions contemplated hereby, the taking of any such action by Owner Trustee in the presence (whether in person or pursuant to a conference call participated in by each of Owner Trustee and Owner Participant and/or its counsel) of Owner Participant or its counsel to evidence conclusively the direction of Owner Participant.

- **Section 3.02** Conditions Precedent. The right and obligation of Owner Trustee to take the actions required by Section 3.01 shall be subject to the following conditions precedent:
- (a) the terms and conditions of <u>Section 4.1</u> of the Participation Agreement shall have been waived or complied with in a manner satisfactory to Owner Participant; and
- (b) the terms and conditions of <u>Section 4.2</u> of the Participation Agreement shall have been waived or complied with in a manner satisfactory to Owner Trustee.

#### Section 3.03 Replacement or Return of an Engine.

- (a) Owner Participant hereby authorizes and directs Owner Trustee to, and Owner Trustee agrees for the benefit of Owner Participant that it will, in the event of any Replacement Engine being substituted pursuant to <u>Section 8(d)</u> of the Lease (and subject to compliance with the terms thereof and the satisfaction of the conditions thereunder), take the following actions:
  - (i) to the extent not previously accomplished by a prior authorization, authorize a representative or representatives of Owner Trustee (who shall be an employee or employees of Lessee) to accept delivery of such Replacement Engine, if the seller of such Replacement Engine is not Lessee;
  - (ii) accept from Lessee or other vendor of such Replacement Engine a bill of sale with respect to such Replacement Engine being furnished pursuant to <u>Section 8(d)</u> of the Lesse;
  - (iii) if the seller of such Replacement Engine is "situated in" a country that has ratified the Cape Town Treaty, cooperate with Lessee to cause the sale of such Replacement Engine to Lessor to be registered on the International Registry as a Sale (or, if the seller of such Replacement Engine is not situated in a country that has ratified the Cape Town Treaty, cooperate with Lessee's reasonable efforts to cause the seller to register the sale of such Replacement Engine on the International Registry);
  - (iv) execute and deliver a Lease Supplement covering such Replacement Engine, and cooperate with Lessee to cause such executed Lease Supplement to be filed for recordation pursuant to the Transportation Code or, if necessary, pursuant to the applicable laws of such jurisdiction other than the U.S. in which the Aircraft is registered, as the case may be;
  - (v) cooperate with Lessee to cause the International Interest created pursuant to the Lease Supplement in favor of Lessor with respect to such Replacement Engine to be registered on the International Registry as an International Interest;
    - (vi) transfer the Engine being replaced to Lessee or its designee in accordance with Section 4(g) of the Lease; and
    - (vii) take such further action as may be contemplated by the Operative Documents in connection with such replacement.

In the event of the substitution of a Replacement Engine for any Engine, all provisions of this Trust Agreement relating to such replaced Engine shall be applicable to such Replacement Engine with the same force and effect as if such Replacement Engine were the same engine as the Engine being replaced.

- (b) Owner Participant hereby authorizes and directs Owner Trustee to, and Owner Trustee agrees for the benefit of Owner Participant that it will, in the event of an engine being transferred to Owner Trustee pursuant to the Return Conditions (and subject to compliance with the terms of <u>Annex B</u> to the Lease and the satisfaction of the conditions thereunder applicable to such engine):
  - (i) accept from Lessee or other vendor of such engine the bill of sale with respect to such engine being furnished pursuant to the Return Conditions;
  - (ii) if the seller of such engine is "situated in" a country that has ratified the Cape Town Treaty, cooperate with Lessee to cause the sale of such engine to Lessor to be registered on the International Registry as a Sale (or, if the seller of such engine is not situated in a country that has ratified the Cape Town Treaty, cooperate with Lessee's reasonable efforts to cause the seller to register the sale of such engine on the International Registry);
    - (iii) transfer the Engine being replaced by such engine to Lessee or its designee in accordance with Section 4(g) of the Lease; and
    - (iv) take such further action as may be contemplated by the Operative Documents in connection with such replacement.

#### ARTICLE IV

# RECEIPT, DISTRIBUTION AND APPLICATION OF INCOME FROM THE TRUST ESTATE

#### Section 4.01 Distribution of Payments.

(a) <u>Payments to Owner Trustee</u>; <u>Other Parties</u>. Except as otherwise provided in <u>subsections</u> (b) and (c), all Basic Rent, Supplemental Rent, insurance proceeds and requisition, indemnity or other payments of any kind, in each case included in the Trust Estate and received by Owner Trustee, shall be distributed forthwith upon receipt by Owner Trustee in the following order of priority: <u>first</u>, so much of such payment as shall be required to pay or reimburse Owner Trustee for any fees or expenses not otherwise paid or reimbursed as to which Owner Trustee is entitled to be so paid or reimbursed pursuant to the provisions hereof or of the other Operative Documents shall be retained by Owner Trustee; <u>second</u>, so much of the remainder for which provision as to the holding, application or distribution thereof is contained in the Lease or any other Operative Document shall be held, applied or distributed in accordance with the terms of the Lease or such other Operative Document; and <u>third</u>, the balance, if any, shall be paid to Owner Participant. Nothing herein is intended to limit or restrict the payment of the Security Deposit to the Owner Participant.

- (b) <u>Certain Distributions to Lessee</u>. Any payment of the type referred to in <u>subsection (a)</u> received by Owner Trustee shall, if required by the terms of the Lease or any other Operative Document, be distributed to Lessee.
- (c) <u>Insurance Proceeds</u>. Any proceeds of any insurance for loss or damage to the Aircraft in excess of the Stipulated Loss Value for the Aircraft shall be paid to Lessee. Any proceeds of any insurance for loss or damage to the Aircraft not constituting an Event of Loss with respect to the Airframe, the Aircraft or any Engine received by Owner Trustee and not required by the terms of the Lease to be distributed to Lessee shall be applied as provided in <u>Section 11(d)</u> of the Lease.

Section 4.02 Method of Payments. Owner Trustee shall make distributions or cause distributions to be made to Owner Participant or Lessee, as applicable, pursuant to this Article IV by transferring by wire transfer in immediately available funds the amount to be distributed to such account or accounts of Owner Participant or Lessee, as applicable, as they respectively may designate from time to time by written notice to Owner Trustee (and Owner Trustee shall use reasonable efforts to cause such funds to be transferred by wire transfer on the same day as received, but in any case not later than the next Business Day); provided, however, that Owner Trustee shall use its reasonable best efforts to invest overnight, for the benefit of Owner Participant or Lessee, as applicable, in Permitted Investments (but only to the extent such investments are available and, if such investments are not available, then in such other investments available to Owner Trustee which, after consultation with Owner Participant or Lessee, as applicable, Owner Participant or Lessee, as applicable, shall direct), all funds not transferred by wire transfer on the same day as they were received. Notwithstanding the foregoing, Owner Trustee will, if so requested by Owner Participant or Lessee, as applicable, by written notice, pay any and all amounts payable by Owner Trustee hereunder to Owner Participant or Lessee, as applicable, either (a) by crediting such amount or amounts to an account or accounts maintained by Owner Participant or Lessee, as applicable, with Owner Trustee in immediately available funds or (b) by mailing an official bank check or checks in such amount or amounts payable to Owner Participant or Lessee, as applicable, at such address as Owner Participant or Lessee, as applicable, shall have designated in writing to Owner Trustee.

#### ARTICLE V

#### **DUTIES OF OWNER TRUSTEE**

Section 5.01 Certain Notices and Requests for Instructions; Related Actions. If Owner Trustee shall have knowledge of any Event of Default or Event of Loss, Owner Trustee shall give to Owner Participant prompt telephonic or facsimile notice thereof followed by prompt confirmation thereof by certified mail, postage prepaid. Subject to the terms of Sections 5.03 and 5.06 and Article XII, Owner Trustee shall (i) in the case of an Event of Default, take such action or shall refrain from taking such action, not inconsistent with the provisions of the Lease and the Participation Agreement, with respect to such Event of Default as Owner Trustee shall be directed in writing by Owner Participant, and (ii) in the case of an Event of Loss, take such action or refrain from taking such action as is provided in the Lease and the Participation Agreement. For all purposes of the Operative Documents, Owner Trustee shall not be deemed to have knowledge of an Event of Default or Event of Loss unless notified in writing thereof in the

manner and at the address set forth in <u>Section 11.05</u> or unless an officer in the corporate trust administration department of Owner Trustee who has responsibility for, or familiarity with, the transactions contemplated under the Operative Documents or any Vice President in such corporate trust administration department has actual knowledge thereof.

Section 5.02 Action Upon Instructions. Subject in all respects to the terms of Sections 5.01, 5.03 and 5.06 and Article XII and to the terms of the other Operative Documents, upon the written instructions at any time and from time to time of Owner Participant, Owner Trustee will take such of the following actions not inconsistent with the provisions of the Lease and Participation Agreement, as may be specified in such instructions: (a) give such notice or direction or exercise such right, remedy or power hereunder or under any Operative Document, or in respect of all or any part of the Trust Estate, as shall be specified in such instructions; (b) take such action to preserve or protect the Trust Estate (including the discharge of any Liens) as may be specified in such instructions; (c) approve as satisfactory to it all matters required by the terms of the Lease and the other Operative Documents to be satisfactory to Owner Trustee, it being understood that, without written instructions of Owner Participant, Owner Trustee shall not approve any such matter as satisfactory to it; (d) subject to the rights, if any, of Lessee under the Operative Documents, after the expiration or earlier termination of the Lease, convey all of Owner Trustee's right, title and interest in and to the Aircraft for such amount, on such terms and to such purchaser or purchasers as shall be designated in such instructions, or lease the Aircraft on such terms as shall be set forth in such instructions or deliver the Aircraft to the Person designated in such instructions in accordance with such instructions; and (e) take or refrain from taking such other action or actions as may be specified in such instructions. In the event that Owner Trustee is unsure of the application of any provision of this Trust Agreement or any other Operative Document, Owner Trustee may request and rely upon instructions of Owner Participant.

**Section 5.03** <u>Indemnification</u>. Owner Trustee shall not be required to take or refrain from taking any action under <u>Section 5.01</u> or <u>5.02</u> unless Owner Trustee shall have been indemnified by Owner Participant, in manner and form satisfactory to Owner Trustee, against any liability, cost or expense (including reasonable counsel fees and disbursements) which may be incurred in connection therewith, other than any such liability, cost or expense which results from the willful misconduct or gross negligence of Owner Trustee, or the failure of Owner Trustee to use ordinary care in the receipt and disbursement of funds, and, if Owner Participant shall have directed Owner Trustee to take or refrain from taking any such action, Owner Participant agrees to pay the reasonable fees and charges of Owner Trustee for the services performed or to be performed by it pursuant to such direction. Owner Trustee shall not be required to take any action under <u>Section 5.01</u> or <u>5.02</u> if Owner Trustee shall reasonably determine, or shall have been advised by counsel, that such action is contrary to the terms of any Operative Document or is contrary to Law.

Section 5.04 No Duties Except as Specified in Operative Documents or Instructions. Owner Trustee shall not have any duty or obligation to manage, control, use, sell, dispose of or otherwise deal with the Aircraft or any other part of the Trust Estate, or otherwise to take or refrain from taking any action under or in connection with the Operative Documents, except as expressly required by the terms of the Operative Documents or (to the extent not inconsistent with the provisions of the Lease and the Participation Agreement) in written instructions from

Owner Participant received pursuant to the terms of Section 5.01 or 5.02, and no implied duties or obligations shall be read into the Operative Documents against Owner Trustee. Without limiting the generality of the foregoing, Owner Trustee shall have no duty (i) to see to any registration of the Aircraft or any recording or filing of the Lease, this Trust Agreement or of any supplement to any thereof or to see to the maintenance of any such registration, rerecording or refiling, except that Owner Trustee shall comply with its obligations under Sections 6.3.1 and 6.4.4 of the Participation Agreement, (ii) to see to any insurance on the Aircraft or to effect or maintain any such insurance, whether or not Lessee shall be in default with respect thereto, other than to forward to Owner Participant copies of all reports and other information which Owner Trustee receives from Lessee pursuant to Section 11 of the Lease, to the extent not received by Owner Participant directly from Lessee, (iii) to see to the payment or discharge of any Tax or any Lien with respect to, assessed or levied against any part of the Trust Estate, except as provided by Section 6.07 hereof or Section 4(d) of the Lease, (iv) to confirm or verify any financial statements of Lessee or (v) to inspect the Aircraft or Lessee's books and records with respect to the Aircraft.

Section 5.05 No Action Except Under Specified Documents or Instructions. Owner Trustee shall have no power or authority to, and Owner Trustee agrees that it will not, manage, control, use, sell, dispose of or otherwise deal with the Aircraft or any other part of the Trust Estate except ( $\underline{a}$ ) as expressly required by the terms of any of the Operative Documents or ( $\underline{b}$ ) as expressly provided in written instructions from Owner Participant pursuant to Section 5.01 or 5.02 that are not inconsistent with the terms of the Operative Documents.

**Section 5.06** <u>Limitations on Activities</u>. Owner Participant and Trust Company agree to, and Owner Participant shall not direct Owner Trustee to take any action in contravention of, the following:

- (a) Owner Trustee shall not engage in any business or any other activity except as expressly permitted by the Operative Documents.
- (b) Except as expressly permitted by <u>Section 8.3</u> of the Participation Agreement, Owner Trustee shall not (<u>i</u>) create, incur or assume any indebtedness for money borrowed, (<u>ii</u>) assume or guarantee or become obligated for the debts of, or hold out the Trust Estate as being available to satisfy the obligations of, Owner Participant or any other Person or (<u>iii</u>) pledge any or all of the Trust Estate for the benefit of Owner Participant or any other Person.
- (c) Owner Trustee shall maintain bank accounts, financial statements, and other books and records for the trust created hereunder separate from those of Owner Participant or any other Person.
- (d) Owner Trustee shall hold the Trust Estate in its own name, as trustee, and shall conduct its activities as Owner Trustee in its own name, as trustee, or in the name of the trust specified in <u>Section 2.02</u>.

#### ARTICLE VI

#### OWNER TRUSTEE

**Section 6.01** Acceptance of Trust and Duties. Trust Company accepts the trust hereby created and agrees to perform the same but only upon the terms hereof applicable to it. Trust Company also agrees to receive and disburse all monies received by it constituting part of the Trust Estate upon the terms hereof. Trust Company, shall not be answerable or accountable under any circumstances, except for (a) its own willful misconduct or gross negligence, (b) its failure to use ordinary care in receiving or disbursing funds, (c) liabilities that may result from the inaccuracy of any representation or warranty of Trust Company (or from the failure by Trust Company to perform any covenant) in any Operative Document and (d) Taxes on or measured by any fees, commissions or other compensation received as compensation for services rendered as Owner Trustee; provided, however, that the failure to act or perform in the absence of instructions after Owner Trustee has requested instructions from Owner Participant pursuant to the last sentence of Section 5.02 shall not constitute willful misconduct or gross negligence for purposes of clause (a) of this Section.

Section 6.02 No Representations or Warranties as to Certain Matters. NEITHER OWNER TRUSTEE NOR TRUST COMPANY MAKES OR SHALL BE DEEMED TO HAVE MADE HEREIN ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, AS TO THE AIRWORTHINESS, VALUE, CONDITION, WORKMANSHIP, DESIGN, OPERATION, MERCHANTABILITY OR FITNESS FOR USE OR FOR A PARTICULAR PURPOSE OF THE AIRCRAFT OR ANY ENGINE OR ANY PART THEREOF, AS TO THE ABSENCE OF LATENT OR OTHER DEFECTS, WHETHER OR NOT DISCOVERABLE, AS TO THE ABSENCE OF ANY INFRINGEMENT OF ANY PATENT, TRADEMARK OR COPYRIGHT, OR AS TO THE ABSENCE OF OBLIGATIONS BASED ON STRICT LIABILITY IN TORT, OR ANY OTHER REPRESENTATION OR WARRANTY WHATSOEVER, EXPRESS OR IMPLIED, WITH RESPECT TO THE AIRCRAFT OR ANY ENGINE OR ANY PART THEREOF, except that nothing set forth in this sentence shall derogate from the representations and warranties made by Owner Trustee or Trust Company in or pursuant to any Operative Document.

**Section 6.03** No Segregation of Monies Required; Investment Thereof. Monies received by Owner Trustee hereunder need not be segregated in any manner, except to the extent required by Law, and may be deposited under such general conditions as may be prescribed by Law, and shall be invested as provided in Section 4.02 hereof or Section 22 of the Lease, as applicable; provided that such monies shall not be commingled with any funds or assets of Owner Participant.

**Section 6.04** Reliance Upon Certificates; Counsel and Agents. Owner Trustee shall incur no liability to anyone in acting in reliance upon any signature, instrument, notice, resolution, request, consent, order, certificate, report, opinion, bond or other document or paper reasonably believed by it to be genuine and reasonably believed by it to be signed by the proper party or parties. Unless other evidence in respect thereof is specifically prescribed herein, any request, direction, order or demand of Owner Participant or Lessee mentioned herein or in any of the other Operative Documents shall be sufficiently evidenced by written instruments signed by

a person purporting to be an officer of Owner Participant or Lessee, as the case may be. Owner Trustee may accept a copy of a resolution of the board of directors of Lessee or Owner Participant, as the case may be, certified by the Secretary or an Assistant Secretary of Lessee or Owner Participant, as the case may be, as conclusive evidence that such resolution has been duly adopted by said board of directors and that the same is in full force and effect. As to any fact or matter the manner of ascertainment of which is not specifically described herein, Owner Trustee may for all purposes hereof rely on a certificate signed by an officer of Lessee or Owner Participant, as the case may be, as to such fact or matter, and such certificate shall constitute full protection to Owner Trustee for any action taken or omitted to be taken by it in good faith in reliance thereon.

Section 6.05 Not Acting in Individual Capacity. Wells Fargo Bank Northwest, National Association is entering into the Operative Documents solely in its capacity as Owner Trustee under this Trust Agreement and not in its individual capacity (except as expressly provided in the Operative Documents) and in no case shall Wells Fargo Bank Northwest, National Association (or any entity acting as successor Owner Trustee under the Trust Agreement) be personally liable for or on account of any of the statements, representations, warranties, covenants or obligations stated to be those of Lessor or Owner Trustee under the Operative Documents; provided, however, that Wells Fargo Bank Northwest, National Association (or any such successor Owner Trustee) shall be personally liable under the Operative Documents for its own gross negligence, its own simple negligence in the handling of funds actually received by it in accordance with the terms of the Operative Documents, its willful misconduct and its breach of its covenants, representations and warranties in the Operative Documents, to the extent covenanted or made in its individual capacity or as otherwise expressly provided in the Operative Documents; provided, further, that nothing contained in this Section shall be construed to limit the exercise and enforcement in accordance with the terms of the Operative Documents of rights and remedies against the Trust Estate.

Section 6.06 Fees; Compensation. Lessee agrees to pay the fees and expenses of Owner Trustee as provided in Section 7.4 of the Participation Agreement.

Section 6.07 Books and Records; Tax Returns. Owner Trustee shall be responsible for keeping all appropriate books and records relating to the receipt and disbursement by it of all monies under this Trust Agreement or any agreement contemplated hereby. At the request of Owner Participant, Owner Trustee shall be responsible for causing to be prepared all income tax returns required to be filed with respect to the trust created hereby and shall execute and file such returns; provided that Owner Participant shall pay all costs and expenses incurred in connection therewith. In addition, Owner Trustee will file any withholding or other information returns required by the Code or the regulations thereunder (including, without limitation, IRS Forms 1042 and 1042-S or any similar or successor forms) with respect to payments received by it under the Operative Documents or distributed by it hereunder, and will withhold, and deposit with the relevant taxing authority, any required U.S. federal tax with respect thereto, in accordance with U.S. federal Tax Laws. Owner Participant shall furnish to Owner Trustee such duly completed and executed forms, statements or certificates, as may be reasonably requested by Owner Trustee, in order for Owner Trustee to file any such returns and to otherwise comply with any withholding or other requirements, and will promptly notify Owner Trustee if any such form, statement or certificate becomes obsolete or incorrect. Owner Participant shall be

responsible for causing to be prepared and filed, at its expense, all income tax returns required to be filed by Owner Participant. Each party hereto, upon request of the other party, will furnish any information in its possession or reasonably available to it as may be reasonably requested by the other party in connection with the preparation of such tax returns or to otherwise comply with the requirements of any taxing authority with respect to the transactions contemplated by the Operative Documents.

#### ARTICLE VII

# INDEMNIFICATION OF OWNER TRUSTEE BY OWNER PARTICIPANT

Section 7.01 Owner Participant to Indemnify Trust Company. Owner Participant hereby agrees, whether or not any of the transactions contemplated hereby shall be consummated, to assume liability for, and does hereby indemnify, protect, save and keep harmless Trust Company, and its successors, assigns, legal representatives, agents and servants, from and against any and all Claims and Taxes (excluding any Taxes payable by Trust Company on or measured by any fees, commissions or other compensation received for services rendered as Owner Trustee hereunder) of any kind and nature whatsoever which may be imposed on, incurred by or asserted against Trust Company in any way relating to or arising out of this Trust Agreement or any of the other Operative Documents or the enforcement of any of the terms of any thereof, or in any way relating to or arising out of the use, possession, operation, control, delivery, maintenance, repair, substitution, replacement, or other disposition of the Aircraft (including, without limitation, with respect thereto, any such Claim for patent, trademark or copyright infringement), or in any way relating to or arising out of the administration of the Trust Estate or the action or inaction of Owner Trustee or Trust Company hereunder; provided that such indemnification shall not extend to any of the foregoing resulting from (a) the willful misconduct or gross negligence on the part of Owner Trustee or Trust Company, (b) failure on the part of Owner Trustee or Trust Company to use ordinary care in receiving or disbursing funds, (c) the inaccuracy of any representation or warranty of Trust Company (or from the failure of Trust Company to perform any covenant) in any Operative Document or (d) a breach by Trust Company of its covenants set forth in Section 5.04 hereof and the first sentence of Section 5.01 hereof; provided, further, that (i) Owner Participant shall be liable under this Section only to the extent that Trust Company is indemnified by Lessee pursuant to Section 7 of the Participation Agreement and (ii) Trust Company shall not make any claim for indemnification or other payment from the Owner Participant pursuant to this Section 7.01 unless and until Trust Company shall have first made demand upon Lessee for such indemnification. The indemnities contained in this Section extend to Trust Company and shall not be construed as indemnities of the Trust Estate (except to the extent, if any, that the Trust Company has been reimbursed by the Trust Estate for amounts covered by the indemnities contained in this Section). The indemnities contained in this Section shall survive the termination of this Trust Agreement.

#### ARTICLE VIII

#### TRANSFER OF OWNER PARTICIPANT'S INTEREST

**Section 8.01** <u>Transfer of Interest</u>. All provisions of <u>Section 8.2</u> of the Participation Agreement shall (with the same force and effect as if set forth in full, *mutatis mutandis*, in this Section) be applicable to any direct or indirect Transfer by Owner Participant of any or all of its right, title or interest in and to this Trust Agreement or any of the other Operative Documents or the Trust Estate or any proceeds therefrom.

#### ARTICLE IX

#### SUCCESSOR OWNER TRUSTEES

Section 9.01 Resignation of Owner Trustee; Appointment of Successor.

- (a) Resignation or Removal. Owner Trustee (i) shall resign if required to do so pursuant to Section 9.3 of the Participation Agreement and (ii) may resign at any time without cause by giving at least 60 days prior written notice to Owner Participant and Lessee, such resignation to be effective upon the acceptance of appointment by the successor Owner Trustee under Section 9.01(b). In addition, subject to Article XII and subject to Section 6.2.2 of the Participation Agreement, Owner Participant may at any time remove Owner Trustee, only for cause (or, at any time when the Aircraft is registered in a non-United States jurisdiction, with or without cause), by a notice in writing delivered to Owner Trustee and Lessee, such removal to be effective upon the acceptance of appointment by the successor Owner Trustee under Section 9.01(b). In the case of the resignation or removal of Owner Trustee, subject to Article XII and subject to Section 6.2.2 of the Participation Agreement, Owner Participant may appoint a successor Owner Trustee by an instrument in writing signed by Owner Participant with the prior written consent of Lessee, such consent not to be unreasonably withheld; provided that, if an Event of Default shall have occurred and be continuing, then no such prior written consent of Lessee shall be so required. If a successor Owner Trustee shall not have been appointed within 30 days after such notice of resignation or removal, Owner Trustee or Lessee may apply to any court of competent jurisdiction to appoint a successor Owner Trustee to act until such time, if any, as a successor shall have been appointed as above provided. Any successor Owner Trustee so appointed by such court shall immediately and without further act be superseded by any successor Owner Trustee appointed as above provided within one year from the date of the appointment by such court.
- (b) Execution and Delivery of Documents, etc. Any successor Owner Trustee, however appointed, shall execute and deliver to the predecessor Owner Trustee an instrument accepting such appointment and shall give Owner Participant and Lessee written notice of such acceptance. Upon the execution and delivery of such instrument, such successor Owner Trustee, without further act, shall become vested with all the estates, properties, rights, powers, duties and trust of the predecessor Owner Trustee in the trust hereunder with like effect as if originally named Owner Trustee herein; but nevertheless, upon the written request of such successor Owner Trustee, such predecessor Owner Trustee shall execute and deliver an instrument transferring to such successor Owner Trustee, upon the trust herein expressed, all the

estates, properties, rights, powers and trust of such predecessor Owner Trustee, and such predecessor Owner Trustee shall duly assign, transfer, deliver and pay over to such successor Owner Trustee all monies or other property then held by such predecessor Owner Trustee as Owner Trustee upon the trust herein expressed, together with all the books and records maintained by such predecessor Owner Trustee with respect to such trust pursuant to Sections 5.04, 5.06 and 6.07. Upon the appointment of any successor Owner Trustee hereunder, the predecessor Owner Trustee will complete, execute and deliver to the successor Owner Trustee such documents as are necessary to cause registration of the Aircraft included in the Trust Estate to be transferred upon the records of the FAA, or the International Registry or other governmental authority having jurisdiction, into the name of the successor Owner Trustee.

- (c) <u>Qualification</u>. Any successor Owner Trustee, however appointed, shall be a Citizen of the United States and shall also be a bank or trust company organized under the laws of the United States or any state thereof having a combined capital and surplus of at least \$100,000,000 (or having a combined capital and surplus of at least \$25,000,000 and the obligations of which are guaranteed by a corporation or a bank or trust company having a combined capital and surplus of at least \$100,000,000), if there be such an institution willing, able and legally qualified to perform the duties of Owner Trustee hereunder upon reasonable and customary terms.
- (d) <u>Merger, etc.</u> Any corporation into which Trust Company may be merged or converted or with which it may be consolidated, or any corporation resulting from any merger, conversion or consolidation to which Trust Company shall be a party, or any corporation to which substantially all the corporate trust business of Trust Company may be transferred, shall, subject to the terms of <u>Section 9.01(c)</u>, be the institution acting as Owner Trustee hereunder without further act. Trust Company shall pay all costs and expenses associated with such merger, conversion or consolidation, without indemnification or reimbursement from either Lessee or Owner Participant, and shall obtain all necessary documentation properly to reflect such merger, conversion or consolidation.

#### ARTICLE X

# SUPPLEMENTS AND AMENDMENTS TO TRUST AGREEMENT AND OTHER DOCUMENTS

Section 10.01 Supplements and Amendments. Subject to Section 6.4.6(b) of the Participation Agreement, at any time and from time to time, upon the written request of Owner Participant, (a) Owner Trustee, together with Owner Participant, shall execute a supplement to this Trust Agreement for the purpose of adding provisions to, or changing or eliminating provisions of, this Trust Agreement as specified in such request, and (b) Owner Trustee shall enter into such written amendment of or supplement to any other Operative Document as Lessee may agree to and as may be specified in such request, or execute and deliver such written waiver or modification of or consent under the terms of any such Operative Document as Lessee may agree to and as may be specified in such request. Notwithstanding the foregoing, except to the extent permitted by Section 6.4.6(b) of the Participation Agreement, no supplement to this Trust Agreement or waiver or modification of the terms hereof shall be permitted.

**Section 10.02** <u>Discretion as to Execution of Documents</u>. If in the opinion of Owner Trustee any document required to be executed pursuant to the terms of <u>Section 10.01</u> adversely affects any right, duty, immunity or indemnity in favor of Trust Company or Owner Trustee hereunder or under any other Operative Document, Owner Trustee may in its discretion decline to execute such document.

**Section 10.03** <u>Absence of Requirements as to Form</u>. It shall not be necessary for any written request furnished pursuant to <u>Section 10.01</u> to specify the particular form of the proposed documents to be executed pursuant to such Section, but it shall be sufficient if such request shall indicate the substance thereof.

**Section 10.04** <u>Distribution of Documents</u>. Promptly after the execution by Owner Trustee of any document entered into pursuant to <u>Section 10.01</u>, Owner Trustee shall mail, by certified mail, postage prepaid, a conformed copy thereof to Owner Participant, but the failure of Owner Trustee to mail such conformed copy shall not impair or affect the validity of such document.

**Section 10.05** No Request Needed as to Lease Supplements. No written request pursuant to Section 10.01 shall be required to enable Owner Trustee to enter into any Lease Supplement with Lessee pursuant to Section 3.01 or Section 3.03(a).

#### ARTICLE XI

#### MISCELLANEOUS

Section 11.01 Termination of Trust Agreement. This Trust Agreement and the trust created hereby shall terminate and this Trust Agreement shall be of no further force or effect upon the earliest of (a) the later of (i) the sale or other final disposition by Owner Trustee of all property constituting part of the Trust Estate and the final distribution by Owner Trustee of all monies or other property or proceeds constituting part of the Trust Estate in accordance with Article IV, and (ii) the expiration or termination of the Lease in accordance with its terms; provided that at such time Lessee shall have fully complied with all of the terms of the Participation Agreement and the Lease or (b) 110 years less one day after the earlier execution of this Trust Agreement by either Trust Company or Owner Participant (or, without limiting the generality of the foregoing, if legislation shall become effective providing for the validity or permitting the effective grant of such rights, privileges and options for a period in gross, exceeding the period for which such rights, privileges and options are stated in this clause (b) to extend and be valid, then such rights, privileges or options shall not terminate as aforesaid in this clause (b) but shall extend to and continue in effect, but only if such non-termination and extension shall then be valid under applicable law, until such time as the same shall under applicable law cease to be valid), whereupon all monies or other property or proceeds constituting part of the Trust Estate shall be distributed in accordance with the terms of Article IV or (c) the election of Owner Participant by notice to Owner Trustee to revoke the trust created hereby; otherwise this Trust Agreement and the trust created hereby shall continue in full force and effect in accordance with the terms hereof. Notwithstanding the foregoing, the provisions of Section 6.4.6 of the Participation Agreement shall apply hereto.

**Section 11.02** Owner Participant Has No Legal Title in Trust Estate. Owner Participant does not have legal title to any part of the Trust Estate. No transfer, by operation of law or otherwise, of any right, title and interest of Owner Participant in and to the Trust Estate hereunder shall operate to terminate this Trust Agreement or the trust hereunder or entitle any successors or transferees of Owner Participant to an accounting or to the transfer of legal title to any part of the Trust Estate.

**Section 11.03** Assignment, Sale, etc. of Aircraft. Any Transfer of the Aircraft by Owner Trustee made pursuant to and in accordance with the terms hereof or of the Lease or the Participation Agreement shall bind Owner Participant and shall be effective to Transfer all right, title and interest of Owner Trustee and Owner Participant in and to the Aircraft. No assignee, purchaser, transferee or other grantee shall be required to inquire as to the authorization, necessity, expediency or regularity of such Transfer or as to the application of any sale or other proceeds with respect thereto by Owner Trustee.

**Section 11.04** <u>Third Party Beneficiary</u>. Lessee shall be an express third party beneficiary of this Trust Agreement to the extent the provisions of this Trust Agreement by their terms expressly confer upon Lessee any right or remedy.

Section 11.05 Notices. Unless otherwise expressly specified or permitted by the terms hereof, all notices, requests, demands, authorizations, directions, consents or waivers required or permitted under the terms and provisions of this Trust Agreement shall be in English and in writing, and given by United States registered or certified mail, return receipt requested, postage prepaid, overnight courier service or facsimile, and any such notice shall be effective when received (or, if delivered by facsimile, upon completion of transmission and confirmation by the sender (by a telephone call to a representative of the recipient or by machine confirmation) that such transmission was received) and addressed as follows: (a) if to Lessee, Owner Trustee or Trust Company, to the respective addresses set forth in Section 10.1 of the Participation Agreement, and (b) if to Owner Participant, to such address as it shall have furnished by notice to Owner Trustee, or, until an address is so furnished, to the respective address set forth in Section 10.1 of the Participation Agreement. Any party, by notice to the other parties hereto, may designate different addresses for subsequent notices or communications. Whenever the words "notice" or "notify" or similar words are used herein, they mean the provision of formal notice as set forth in this Section.

#### Section 11.06 Miscellaneous.

- (a) Any provision of this Trust Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof or affecting the validity or enforceability of such provision in any other jurisdiction.
- (b) No term or provision of this Trust Agreement may be changed, waived, discharged or terminated orally, but only by an instrument in writing entered into in compliance with the terms of Article X; and any waiver of the terms hereof shall be effective only in the specified instance and for the specific purpose given.

- (c) This Trust Agreement and the other Operative Documents, and all certificates, instruments and other documents relating thereto delivered and to be delivered from time to time pursuant to the Operative Documents, supersede any and all representations, warranties and agreements (other than any Operative Document) prior to the date of this Trust Agreement, written or oral, between or among any of the parties hereto relating to the transactions contemplated hereby and thereby.
- (d) This Trust Agreement may be executed in any number of counterparts (and each of the parties hereto shall not be required to execute the same counterpart). Each counterpart of this Trust Agreement, including a signature page executed by each of the parties hereto shall be an original, but all of such counterparts together shall constitute one instrument.
- (e) This Trust Agreement shall be binding upon and inure to the benefit of, Owner Participant and, subject to the provisions of Article VIII hereof, its successors and permitted assigns, Owner Trustee and its successors as Owner Trustee under this Trust Agreement and Trust Company and its successors and permitted assigns. Any request, notice, direction, consent, instruction, waiver or other instrument or action by Owner Participant shall bind its successors and permitted assigns.
- (f) THIS TRUST AGREEMENT SHALL IN ALL RESPECTS BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF UTAH, INCLUDING ALL MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE.

#### ARTICLE XII

#### CERTAIN LIMITATIONS ON CONTROL<sup>1</sup>

Section 12.01 <u>Limitations on Control</u>. Notwithstanding any other provision of this Trust Agreement, but subject to <u>Sections 12.02</u> and <u>12.04</u>, Owner Participant will have no rights or powers to direct, influence or control Owner Trustee in the performance of Owner Trustee's duties under this Trust Agreement in connection with any matters involving the ownership and operation of the Aircraft by Owner Trustee. In all such matters, Owner Trustee shall have absolute and complete discretion in connection therewith and shall be free of any kind of influence or control whatsoever by Owner Participant, and Owner Trustee shall exercise its duties under this Trust Agreement in connection with matters involving the ownership and operation of the Aircraft by Owner Trustee as it, in its discretion, shall deem necessary to protect the interests of the United States, notwithstanding any countervailing interest of any foreign power which, or whose citizens, may have a direct or indirect interest in Owner Participant, and any such action by Owner Trustee shall not be considered malfeasance or in breach of any obligation which Owner Trustee might otherwise have to Owner Participant; <u>provided, however</u>, that subject to the foregoing limitations, Owner Trustee shall exercise its discretion in all matters involving the ownership and operation of the Aircraft by Owner Trustee (<u>a</u>) with due regard for the interests of Owner Participant and (<u>b</u>) in a manner not inconsistent with the provisions of the

Provision subject to change pursuant to FAA regulations regarding NCT.

Operative Documents; <u>provided</u>, <u>further</u>, that Owner Participant may confer with Owner Trustee and/or Owner Trustee may consult with Owner Participant in connection with such matters involving the ownership and operation of the Aircraft (it being understood that any advice, opinion or suggestion obtained by Owner Trustee in the course of such conferring or consulting shall not be binding on Owner Trustee, but that Owner Trustee shall be free to follow or disregard such advice, opinion or suggestion in the exercise of its discretion). In addition, Owner Participant may not remove Owner Trustee or any successor Owner Trustee appointed hereunder, except for cause. Owner Trustee agrees to promptly notify Owner Participant of the exercise of its duties under this Trust Agreement in connection with matters involving the ownership and operation of the Aircraft by Owner Trustee.

Section 12.02 <u>Discretion, Actions and Payments of Owner Trustee</u>. Subject to the requirements of <u>Section 12.01</u>, Owner Trustee agrees that it will not, unless expressly required by the terms of this Trust Agreement, without the prior consent of Owner Participant, (<u>a</u>) sell, transfer, assign, lease, mortgage, pledge or otherwise dispose of the Aircraft or other assets held in the Trust Estate relating thereto or (<u>b</u>) amend or waive any rights under any Operative Document, or give any consents under any Operative Documents. Notwithstanding any other provision of this <u>Article XII</u>, the grant of the rights of Owner Trustee set forth in <u>Section 12.01</u> shall not extend to any other rights, powers or privileges in respect of the beneficial interest of Owner Participant in the Trust Estate, and Owner Participant (and not Owner Trustee) shall be entitled to receive from Owner Trustee or otherwise all payments of whatsoever kind and nature payable to Owner Participant pursuant to this Trust Agreement in the same manner as if the rights permitted to be exercised by Owner Trustee as described in <u>Section 12.01</u> had not been transferred to Owner Trustee and held in trust hereunder.

**Section 12.03** General. Owner Trustee and Owner Participant hereby agree with each other that if Persons who are neither Citizens of the United States nor resident aliens have the power to direct or remove Owner Trustee, either directly or indirectly through the control of another Person, those Persons together shall not have more than 25% of the aggregate power to direct or remove Owner Trustee.

Section 12.04 <u>Purpose</u>. The purpose of this <u>Article XII</u> is to give Owner Trustee the power to manage and control the Aircraft with respect to matters involving the ownership and operation of the Aircraft by Owner Trustee so as to ensure that (a) the Aircraft shall be controlled with respect to such matters by a Citizen of the United States, and (b) Owner Trustee shall be able to give the affidavit required by Section 47.7(c)(2)(iii) of the Federal Aviation Regulations, 14 C.F.R. §47.7(c)(2)(iii). This <u>Article XII</u> shall be construed in furtherance of the foregoing purposes; <u>provided, however</u>, that this <u>Article XII</u> shall be ignored and given no force or effect: (i) if Owner Participant determines that it meets the requirements for a Citizen of the United States and both Owner Participant and Owner Trustee file with the FAA the affidavits required by Section 47.7(c)(2)(ii) of the Federal Aviation Regulations, 14 C.F.R. §47.7(c)(2)(ii), or (ii) during periods when the Aircraft has been registered in a non-United States jurisdiction and a de-registration telex has been issued by the FAA in connection with the reregistration of the Aircraft in such non-United States jurisdiction.

Section 12.05 Adverse Effect of Citizenship on Registration. If the right (a) to exercise voting or similar rights hereunder by Owner Participant, or (b) (i) to direct, influence, or limit the

exercise of, or (<u>ii)</u> to prevent the direction or influence of, or (<u>iii)</u> place any limitation on the exercise of, Owner Trustee's authority, or (<u>c</u>) to remove Owner Trustee, would adversely affect the United States registration of the Aircraft, Owner Participant shall have no such right; <u>provided</u>, <u>however</u>, that this Section shall be of no force or effect during periods in which the Aircraft is registered in a non-United States jurisdiction.

[Remainder of Page Intentionally Left Blank; Signature Page Follows]

**IN WITNESS WHEREOF**, the parties hereto have caused this Trust Agreement to be duly executed by their respective officers thereunto duly authorized as of the day and year first above written.

	LLS FARGO BANK NORTHWEST, NATIONAL SOCIATION
Ву	
	Name:
	Title:
[NA	AME OF OWNER PARTICIPANT]
Dy	
	Name:

ANNEX A

## **DEFINITIONS**

ANNEX A Page 1

# PARTICIPATION AGREEMENT ([YEAR] MSN [MSN])

dated as of

[Date]

among

#### AMERICAN AIRLINES, INC.,

as Lessee

#### [NAME OF OWNER PARTICIPANT],

as Owner Participant

and

### WELLS FARGO BANK NORTHWEST, NATIONAL ASSOCIATION,

not in its individual capacity, except as expressly provided herein, but solely as Owner Trustee

Covering One Airbus [Model] Aircraft (Generic Manufacturer and Model AIRBUS [Generic Model])

### TABLE OF CONTENTS

Page

Section 1.	Definitions	1
Section 2.	Lease of Aircraft	1
Section 3.	Closing	1
Section 4.	Conditions Precedent	2
4.1	Conditions Precedent to Obligations of Owner Participant	2
4.2	Conditions Precedent to Obligations of Owner Trustee	5
4.3	Conditions Precedent to Obligations of Lessee	6
4.4	Aviation Counsel Opinions	7
Section 5.	Representations and Warranties	8
5.1	Representations and Warranties of Lessee	8
5.2	Representations and Warranties of Owner Participant	9
5.3	Representations and Warranties of Owner Trustee and Trust Company	11
Section 6.	Covenants and Agreements	13
6.1	Covenants of Lessee	13
6.2	Covenants of Owner Participant	16
6.3	Covenants of Owner Trustee and Trust Company	17
6.4	Other Covenants	18
6.5	Filings	20
Section 7.	Indemnification and Expenses	21
7.1	General Indemnity	21
7.2	General Tax Indemnity	24
7.3	Survival; Other	29
7.4	Expenses	31
Section 8.	Assignment or Transfer of Interests	31
8.1	Owner Trustee	31
8.2	Owner Participant	33
8.3	Back-Leverage	36
Section 9.	Change of Citizenship	39
9.1	Generally	39
9.2	Owner Participant	40
9.3	Owner Trustee	40
Section 10.	Miscellaneous	41
10.1	Notices	41
10.2	Late Payments; Business Days; Currency	41

### TABLE OF CONTENTS

(continued)

		Page
10.3	Concerning Owner Trustee	41
10.4	Confidential Information	42
10.5	Further Assurances	42
10.6	Third Party Beneficiary	43
10.7	Miscellaneous	43

## EXHIBITS, ANNEXES AND SCHEDULES

EXHIBIT A	FORM OF OPINION OF [ASSOCIATE] GENERAL COUNSEL FOR LESSEE
EXHIBIT B	FORM OF OPINION OF SPECIAL COUNSEL FOR OWNER TRUSTEE
EXHIBIT C	[INTENTIONALLY LEFT BLANK]
EXHIBIT D	FORM OF OPINION OF AVIATION COUNSEL
EXHIBIT E	FORM OF ASSIGNMENT AND ASSUMPTION AGREEMENT
EXHIBIT F	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR
	CONFIDENTIAL TREATMENT]
EXHIBIT G	FORM OF OWNER PARTICIPANT GUARANTEE
ANNEX A	DEFINITIONS
ANNEX B	PAYMENT INFORMATION
SCHEDULE A	CERTAIN TERMS
SCHEDIII E B	RE RECISTRATION CONDITIONS

#### PARTICIPATION AGREEMENT ([YEAR] MSN [MSN])

THIS PARTICIPATION AGREEMENT ([YEAR] MSN [MSN]), dated as of [ ] (as amended, modified or supplemented from time to time, this "Agreement"), among (i) AMERICAN AIRLINES, INC., a Delaware corporation (together with its successors and permitted assigns, "Lessee"), (ii) [NAME OF OWNER PARTICIPANT], a [jurisdiction] [type of entity] (together with its successors and permitted assigns, "Owner Participant"), and (iii) WELLS FARGO BANK NORTHWEST, NATIONAL ASSOCIATION, a national banking association, not in its individual capacity except as expressly provided herein, but solely as Owner Trustee (herein in such capacity, together with its successors and permitted assigns, "Owner Trustee", and in its individual capacity, together with its successors and permitted assigns, "Trust Company").

#### **RECITALS:**

- 1. Subject to the terms and conditions set forth herein, Owner Trustee is willing to lease the Aircraft to Lessee, and Lessee is willing to lease the Aircraft from Owner Trustee pursuant to the Lease.
- 2. On or prior to the date hereof, Owner Participant has entered into the Trust Agreement with Trust Company, pursuant to which Owner Trustee agrees, among other things, to hold the Trust Estate for the benefit of Owner Participant on the terms specified in such Trust Agreement.
- 3. Pursuant to the terms of the Trust Agreement, Owner Trustee is authorized and directed by Owner Participant to execute and deliver the Lease, pursuant to which, subject to the terms and conditions set forth therein, Owner Trustee agrees to lease to Lessee, and Lessee agrees to lease from Owner Trustee, the Aircraft on the Delivery Date, such lease to be evidenced by the execution and delivery of Lease Supplement No. 1.
- **NOW, THEREFORE**, in consideration of the premises and mutual agreements herein contained, the agreements contained in the other Operative Documents and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:
- **Section 1.** <u>Definitions</u>. Unless the context otherwise requires, all capitalized terms used herein and not otherwise defined herein shall have the meanings set forth, and shall be construed and interpreted in the manner described, in <u>Annex A</u> for all purposes of this Agreement.
- **Section 2.** <u>Lease of Aircraft</u>. Subject to the terms and conditions of this Agreement, on the Delivery Date, Owner Trustee agrees to lease the Aircraft to Lessee, and Lessee agrees to lease the Aircraft from Owner Trustee, pursuant to the Lease.
- Section 3. <u>Closing</u>. On the Delivery Date, subject to the terms and conditions of this Agreement and the Lease, Owner Trustee shall lease the Aircraft to Lessee and Lessee shall accept the Aircraft under the Lease by executing and delivering Lease Supplement No. 1. The closing (the "Closing") of the transactions contemplated hereby shall take place commencing at 9:00 a.m., Fort Worth, Texas time (or such later time as the parties may agree), on the Delivery Date at the offices of Lessee in Fort Worth, Texas.

#### Section 4. Conditions Precedent.

- 4.1 <u>Conditions Precedent to Obligations of Owner Participant</u>. The obligation of Owner Participant to take the actions required by this Agreement to be taken by it at the Closing is subject to the satisfaction or waiver by Owner Participant, prior to or at the Closing, of the conditions precedent set forth below in this <u>Section 4.1</u>; <u>provided</u> that it shall not be a condition precedent to the obligations of Owner Participant that any document be delivered or action be taken that is to be delivered or be taken by Owner Participant or by a Person within Owner Participant's control.
- 4.1.1 <u>Delivery of Documents</u>. This Agreement and the following documents shall have been duly authorized, executed and delivered by the respective party or parties thereto, shall be in form and substance reasonably satisfactory to Owner Participant, shall be in full force and effect and executed original counterparts or copies thereof shall have been delivered to Owner Participant or its special counsel unless the failure to receive such document is the result of any action or inaction by Owner Participant or by a Person within Owner Participant's control:
  - (a) the Trust Agreement and the Owner Participant Guarantee, if any;
- (b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (c) the Lease and Lease Supplement No. 1; provided that only Owner Trustee shall receive the sole executed chattel paper original of each thereof;
  - (d) [the Engine Warranty Agreement]1;
- (e) an insurance report of Lessee's independent insurance broker as to the due compliance with the terms of <u>Section 11</u> of the Lease relating to insurance with respect to the Aircraft and certificates of insurance;
- (f)(i) a copy of the resolutions of the board of directors (or executive committee) of Lessee, certified as of the Delivery Date by the Secretary or an Assistant Secretary of Lessee, duly authorizing the execution, delivery and performance by Lessee of the Operative Documents executed and to be executed by Lessee and each other document required to be executed and delivered by Lessee, in accordance with the provisions hereof; (ii) copies of the certificate of incorporation and by-laws of Lessee, certified as of the Delivery Date by the Secretary or an Assistant Secretary of Lessee, together with all amendments and supplements thereto; and (iii) an incumbency certificate of Lessee, dated as of the Delivery Date, as to the

Insert if applicable.

persons authorized to execute and deliver this Agreement, the other Operative Documents to which Lessee is or is to be a party and each other document executed or to be executed on behalf of Lessee in connection with the transactions contemplated hereby and the signatures of such person or persons;

- (g)(i) a copy of the resolutions of the board of directors (or executive committee) of Trust Company, certified as of the Delivery Date by the Secretary or an Assistant Secretary of Trust Company, duly authorizing the execution, delivery and performance by Trust Company and Owner Trustee of the Operative Documents executed and to be executed by each such party and each other document required to be executed and delivered by each such party in accordance with the provisions hereof; (ii) copies of the articles of association, by-laws and/or other constituent documents of Trust Company, certified as of the Delivery Date by the Secretary or an Assistant Secretary of Trust Company, together with all amendments and supplements thereto; and (iii) an incumbency certificate of Trust Company, dated as of the Delivery Date, as to the persons authorized to execute and deliver this Agreement, the other Operative Documents to which Trust Company or Owner Trustee is or is to be a party and each other document executed or to be executed on behalf of Trust Company or Owner Trustee in connection with the transactions contemplated hereby and the signatures of such person or persons;
- (h) officer's certificates, dated the Delivery Date, from (i) Lessee, certifying as to the correctness of each of the matters stated in Section 4.1.4 (insofar as the same relate to Lessee or the Aircraft); and (ii) each of Trust Company, Owner Trustee [and], Owner Participant [and Owner Participant Guarantor], certifying that no Lessor's Lien attributable to such party exists, and further certifying as to the correctness of each of the matters stated in Section 4.1.4 (insofar as the same relate to such Person); and
- (i) opinions, dated the Delivery Date, from (i) [David A. Allen], Esq., [Associate] General Counsel of Lessee, addressed to Owner Participant and Owner Trustee, in substantially the form attached hereto as <a href="Exhibit A">Exhibit A</a>, (ii) [Ray, Quinney & Nebeker P.C.], special counsel for Owner Trustee, addressed to Owner Participant and Lessee, in substantially the form attached hereto as <a href="Exhibit B">Exhibit B</a>, and (iii) opinion(s) of in-house/external counsel of Owner Participant [and inhouse/external counsel of Owner Participant Guarantor, in each case] addressed to Owner Trustee and Lessee, in form and substance satisfactory to Lessee.
- 4.1.2 No Violation. No change shall have occurred after the date of the execution and delivery of this Agreement in applicable law or regulations thereunder or interpretations thereof by appropriate governmental authorities or any court that would make it a violation of law or regulation for Lessee, Owner Participant or Owner Trustee to execute, deliver and perform its respective obligations under this Agreement or the other Operative Documents and any transactions contemplated by this Agreement or the other Operative Documents.
- 4.1.3 No Proceedings. No action or proceeding or governmental action shall have been instituted or threatened before any court or governmental authority, nor shall any order, judgment or decree have been issued or proposed to be issued by any court or governmental authority, at the time of the Delivery Date to set aside, restrain, enjoin or prevent the completion and consummation of this Agreement or the other Operative Documents or the transactions contemplated hereby and thereby.

- 4.1.4 <u>Representations</u>, <u>Warranties and Covenants</u>. On the Delivery Date, the representations and warranties of each of Lessee and Trust Company made herein and in the other Operative Documents shall be correct and accurate in all material respects, in each case as though made on and as of such date, or if such representations and warranties relate solely to an earlier date, as of such earlier date, and each of Lessee and Trust Company shall have performed and observed, in all material respects, all of its covenants, obligations and agreements in this Agreement and in the other Operative Documents to which it is party to be observed and performed by it as of the Delivery Date.
- 4.1.5 <u>Governmental Authority</u>. All appropriate actions required to have been taken prior to the Delivery Date in connection with the transactions contemplated by this Agreement and the other Operative Documents by any governmental authority shall have been taken, and all orders, permits, waivers, exemptions, authorizations and approvals of any governmental authority required to be in effect on the Delivery Date in connection with the transactions contemplated by this Agreement and the other Operative Documents (other than the filings and registrations referred to in <u>Section 5.1.7</u>) shall have been issued, and all such orders, permits, waivers, exemptions, authorizations and approvals shall be in full force and effect on the Delivery Date.
  - 4.1.6 No Event of Default. On the Delivery Date, no event has occurred and is continuing that constitutes an Event of Default.
- 4.1.7 <u>Aircraft Status</u>. The Aircraft shall have been duly certified by the FAA as to type and shall otherwise meet the conditions specified in the American/Airbus Purchase Agreement.
- 4.1.8 <u>Sales Tax</u>. Lessee shall have provided such exemption certificates for sales, use, value added, goods and services, transfer, stamp or similar Tax purposes with respect to the delivery and lease of the Aircraft as Owner Participant may reasonably request.
- 4.1.9 Filings. On the Delivery Date, (a) the Lease and Lease Supplement No. 1 shall have been duly filed for recordation (or shall be in the process of being so duly filed for recordation) with the FAA pursuant to the Transportation Code and (b) a precautionary UCC financing statement covering the Lease shall have been duly filed in the State of Delaware (or arrangements shall have been made for filing promptly after the Delivery Date).
- 4.1.10 <u>Aircraft Registration</u>. Counsel to the FAA or Aviation Counsel shall have confirmed in writing that the Aircraft is registered or is eligible to be registered in the name of Owner Trustee.
  - 4.1.11 No Event of Loss. On the Delivery Date, no event has occurred and is continuing that constitutes an Event of Loss with respect to the Aircraft.
- 4.1.12 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 4.2 <u>Conditions Precedent to Obligations of Owner Trustee</u>. The obligation of Owner Trustee to take the actions required by this Agreement to be taken by it at the Closing is subject to the satisfaction or waiver by Owner Trustee, prior to or at the Closing, of the conditions precedent set forth below in this <u>Section 4.2</u>; <u>provided</u> that it shall not be a condition precedent to the obligations of Owner Trustee that any document be delivered or action be taken that is to be delivered or be taken by Owner Trustee or by a Person within Owner Trustee's control.
- 4.2.1 <u>Delivery of Documents</u>. Executed original counterparts or copies of the following documents shall have been received by Owner Trustee or its special counsel, unless the failure to receive such document is the result of any action or inaction by Owner Trustee or by a Person within Owner Trustee's control:
  - (a) the documents described in <u>Section 4.1.1</u>, except as specifically provided therein;
- (b)(i) a copy of the resolutions of the board of directors (or executive committee) of Owner Participant, certified as of the Delivery Date by the Secretary or an Assistant Secretary of Owner Participant, duly authorizing the execution, delivery and performance by Owner Participant of the Operative Documents executed and to be executed by Owner Participant and each other document required to be executed and delivered by Owner Participant, in accordance with the provisions hereof; (ii) copies of the certificate of incorporation, by-laws and/or other constituent documents of Owner Participant, certified as of the Delivery Date by the Secretary or an Assistant Secretary of Owner Participant, together with all amendments and supplements thereto; and (iii) an incumbency certificate of Owner Participant, dated as of the Delivery Date, as to the persons authorized to execute and deliver this Agreement, the other Operative Documents to which Owner Participant is or is to be a party and each other document executed or to be executed on behalf of Owner Participant in connection with the transactions contemplated hereby and thereby and the signatures of such person or persons;

(c)(i) a copy of the resolutions of the board of directors (or executive committee) of Owner Participant Guarantor, certified as of the Delivery Date by the Secretary or an Assistant Secretary of Owner Participant Guarantor, duly authorizing the execution, delivery and performance by Owner Participant Guarantor of the Operative Documents executed and to be executed by Owner Participant Guarantor and each other document required to be executed and delivered by Owner Participant Guarantor, in accordance with the provisions hereof; (ii) copies of the certificate of incorporation, by-laws and/or other constituent documents of Owner Participant Guarantor, certified as of the Delivery Date by the Secretary or an Assistant Secretary of Owner Participant Guarantor, together with all amendments and supplements thereto; and (iii) an incumbency certificate of Owner Participant Guarantor, dated as of the Delivery Date, as to the persons authorized to execute and deliver the Operative Documents to which Owner Participant Guarantor is or is to be a party and each other document executed or to be executed on behalf of Owner Participant Guarantor in connection with the transactions contemplated hereby and the signatures of such person or persons;] and

- (d) [evidence of Owner Participant's appointment of a process agent as provided in <u>Section 10.7.7</u> and [of Owner Participant Guarantor's appointment of a process agent as provided in the Owner Participant Guarantee and]<sup>2</sup> [each] of such process agent's acceptance of such appointment.]<sup>3</sup>
- 4.2.2 <u>Other Conditions Precedent</u>. Each of the conditions set forth in <u>Sections 4.1.2</u>, <u>4.1.3</u>, <u>4.1.4</u>, <u>4.1.5</u>, <u>4.1.6</u>, <u>4.1.7</u>, <u>4.1.8</u>, <u>4.1.9</u>, <u>4.1.10</u> and <u>4.1.11</u> shall have been satisfied or waived by Owner Trustee, unless the failure of any such condition to be satisfied is the result of any action or inaction by Owner Trustee or by a Person within Owner Trustee's control.
- 4.3 <u>Conditions Precedent to Obligations of Lessee</u>. The obligation of Lessee to take the actions required by this Agreement to be taken by it at the Closing is subject to the satisfaction or waiver by Lessee, prior to or at the Closing, of the conditions precedent set forth below in this <u>Section 4.3</u>; <u>provided</u> that it shall not be a condition precedent to the obligations of Lessee that any document be delivered or action be taken that is to be delivered or be taken by Lessee or by a Person within Lessee's control.
- 4.3.1 <u>Delivery of Documents</u>. Executed original counterparts or copies of the following documents shall have been received by Lessee or its counsel, unless the failure to receive such document is the result of any action or inaction by Lessee or by a Person within Lessee's control:
  - (a) the documents described in Sections 4.1.1, [and] 4.2.1(b), [and 4.2.1(c)], [and 4.2.1(d)]<sup>4</sup> except as specifically provided therein; and
  - (b) the Owner Participant Guarantee, if any.
- 4.3.2 <u>Sales Tax</u>. Owner Trustee and Owner Participant shall have provided such exemption certificates for sales, use, value added, goods and services, transfer, stamp or similar Tax purposes with respect to the delivery and lease of the Aircraft as Lessee may reasonably request, and Lessee shall be reasonably satisfied that no such Tax is payable with respect to such delivery and lease.
- 4.3.3 <u>Tax Forms</u>. Lessee shall have received from Owner Trustee a duly completed and executed original IRS Form W-9, and each of Lessee and Owner Trustee shall have received from Owner Participant a duly completed and executed original IRS Form [W-9]<sup>5</sup> (and/or other applicable IRS Form(s)), in each case, establishing a complete exemption from U.S. federal withholding Taxes with respect to all payments of Rent or other amounts to or for the benefit of Owner Trustee or Owner Participant under the Operative Documents.
- <sup>2</sup> Include if Owner Participant Guarantor is foreign.
- Include if foreign OP.
- Include if foreign OP or foreign OP guarantor.
- If foreign OP, replace W-9 with W-8BEN or other relevant IRS form(s).

- 4.3.4 <u>Representations, Warranties and Covenants</u>. On the Delivery Date, the representations and warranties of each of Lessor, Trust Company, Owner Participant and Owner Participant Guarantor (if any) made herein and in the other Operative Documents shall be correct and accurate in all material respects, in each case as though made on and as of such date, or if such representations and warranties relate solely to an earlier date, as of such earlier date, and each of Lessor, Trust Company, Owner Participant and Owner Participant Guarantor (if any) shall have performed and observed, in all material respects, all of its covenants, obligations and agreements in this Agreement and in the other Operative Documents to which it is party to be observed and performed by it as of the Delivery Date.
- 4.3.5 <u>Title</u>. Title to the Aircraft shall have been conveyed to Owner Trustee (subject to the recordation of the FAA Bill of Sale with the FAA pursuant to <u>Section 6.5.1</u> and the registration on the International Registry of the Sale of the Airframe and Engines from Manufacturer to Owner Trustee pursuant to <u>Section 6.5.2</u>), free and clear of Liens other than (<u>a</u>) the rights and interests of Owner Trustee and Lessee under the Lease and Lease Supplement No. 1 covering the Aircraft and (<u>b</u>) the beneficial interest of Owner Participant created by the Trust Agreement.
- 4.3.6 Filings. On the Delivery Date, the FAA Bill of Sale shall have been duly filed for recordation (or shall be in the process of being so duly filed for recordation) with the FAA pursuant to the Transportation Code.
- 4.3.7 <u>Application for Registration</u>. Counsel to the FAA or Aviation Counsel shall have confirmed in writing that the Aircraft is registered or is eligible to be registered in the name of Owner Trustee, the Aircraft shall be registered with the FAA in the name of the Owner Trustee (or application for registration of the Aircraft in the name of Owner Trustee shall have been duly made with the FAA) and Lessee has temporary or permanent authority to operate the Aircraft.
- 4.3.8 Other Conditions Precedent. Each of the conditions set forth in Sections 4.1.2, 4.1.3, 4.1.5, 4.1.7, 4.1.8, 4.1.9, 4.1.10 and 4.1.11 shall have been satisfied or waived by Lessee, unless the failure of any such condition to be satisfied is the result of any action or inaction by Lessee or by a Person within Lessee's control.

# 4.4 Aviation Counsel Opinions.

- 4.4.1 <u>Filing Opinion</u>. Promptly following the filings and registrations described in <u>Sections 6.5.1</u> and <u>6.5.2</u>, Lessee, Owner Trustee and Owner Participant shall receive an opinion addressed to each of them from Aviation Counsel, substantially in the form of <u>Exhibit D</u>.
- 4.4.2 <u>Recordation Opinion</u>. Promptly following the registration of the Aircraft, the recording of the FAA Bill of Sale, the Lease and Lease Supplement No. 1 pursuant to the Transportation Code, and the receipt of appropriate and correct recording information from the FAA, Lessee, Owner Trustee and Owner Participant shall receive an opinion addressed to each of them from Aviation Counsel, as to the due registration of the Aircraft, the due recording of such instruments and the lack of filing of any intervening documents with respect to the Aircraft.

#### **Section 5.** Representations and Warranties.

- 5.1 Representations and Warranties of Lessee. Lessee hereby represents and warrants that as of the date hereof:
- 5.1.1 <u>Organization</u>. Lessee is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Delaware and has the corporate power and authority to own or hold under lease its properties and to enter into and perform its obligations under the Operative Documents to which it is or will be a party.
- 5.1.2 <u>Corporate Authorization; No Violation</u>. The execution, delivery and performance by Lessee of this Agreement and the other Operative Documents to which it is or will be a party have been duly authorized by all necessary corporate action on the part of Lessee, do not require any stockholder approval or consent of any trustee or holder of indebtedness or obligations of Lessee, except such as have been duly obtained, and do not and will not violate the certificate of incorporation or by-laws of Lessee or any current law, governmental rule, regulation, judgment or order binding on Lessee or result in a breach of, or constitute a default under, or result in the creation of any Lien (other than as permitted under the Operative Documents) upon the property of Lessee under, any indenture, mortgage, contract or other agreement to which Lessee is a party or by which Lessee or its properties is or are bound or affected.
- 5.1.3 <u>Approvals</u>. Neither the execution and delivery by Lessee of, nor the performance by Lessee of its obligations under, nor the consummation by Lessee of the transactions contemplated in, this Agreement and the other Operative Documents to which Lessee is or will be a party, requires the consent or approval of, or the giving of notice to, or the registration with, or the taking of any other action in respect of, the Department of Transportation, the FAA or any other United States federal or state governmental authority having jurisdiction, or the International Registry, except for (a) the filings and registrations referred to in <u>Section 5.1.7</u>, (b) notices, filings, recordings and other actions required to be given, made or performed after the Delivery Date and (c) such action, as a result of any act or omission by Owner Trustee, Owner Participant or any Affiliate of any thereof, as may be required under the United States federal securities laws or the securities or other laws of any state thereof or other jurisdiction applicable to sales of securities.
- 5.1.4 <u>Valid and Binding Agreements</u>. This Agreement has been duly executed and delivered by Lessee and constitutes, and each other Operative Document to which Lessee will be a party will be duly executed and delivered by Lessee and, when executed and delivered, will constitute, the legal, valid and binding obligation of Lessee enforceable against Lessee in accordance with its terms, except as may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the rights of creditors generally and general principles of equity and except, in the case of the Lease, as limited by applicable laws that may affect the remedies provided in the Lease, which laws, however, do not make the remedies provided in the Lease inadequate for the practical realization of the rights and benefits intended to be provided thereby.

- 5.1.5 <u>Litigation</u>. Except for such matters disclosed in press releases issued by AMR Corporation or Lessee or in public filings made with the Securities and Exchange Commission under the Exchange Act by AMR Corporation or Lessee, there are no pending or, to Lessee's knowledge, threatened actions or proceedings before any court, arbitrator or administrative agency which would materially adversely affect the ability of Lessee to perform its obligations under this Agreement or any of the other Operative Documents to which Lessee is or will be a party.
- 5.1.6 <u>Securities Law</u>. Neither Lessee nor any Person authorized to act on its behalf has directly or indirectly offered any interest in the Trust Estate or the Trust Agreement or any similar security to, or solicited any offer to acquire any of the same from, any Person in violation of the registration requirements of the Securities Act or any applicable securities law.
- 5.1.7 Registration and Recordation. Except for (a) the registration of the Aircraft with the FAA pursuant to the Transportation Code and periodic renewals of such registration as may be necessary under the FAA regulations governing U.S. registration of aircraft, (b) the filing for recordation with the FAA pursuant to the Transportation Code of the FAA Bill of Sale, the Lease and Lease Supplement No. 1 and (c) the registration on the International Registry of (i) the Sale of the Airframe and Engines from Manufacturer to Owner Trustee and (ii) the International Interests created under the Lease (as supplemented by Lease Supplement No. 1), no further filing or recording of any document is necessary or advisable in order to establish and perfect Owner Trustee's interests in the Aircraft as against Lessee and any third parties in any jurisdiction within the United States, other than the filing of a precautionary financing statement in respect thereof under Article 9 of the UCC as in effect in the State of Delaware and the filing of continuation statements with respect thereto required to be filed at periodic intervals under such UCC.
  - 5.1.8 Certificated Air Carrier. Lessee is a Certificated Air Carrier.
  - 5.2 Representations and Warranties of Owner Participant. Owner Participant hereby represents and warrants that as of the date hereof:
- 5.2.1 <u>Organization</u>. Owner Participant is [type of entity] duly organized, validly existing and in good standing under the laws of [jurisdiction of organization] and has the [corporate] power and authority to own or hold under lease its properties, to carry on its business and operations and to enter into and perform its obligations under the Operative Documents to which it is or will be a party.

- 5.2.2 <u>Corporate Authorization; No Violation</u>. The execution, delivery and performance by Owner Participant of this Agreement and the other Operative Documents to which it is or will be party have been duly authorized by all necessary [corporate] action on the part of Owner Participant, do not require any [stockholder] approval or approval or consent of any trustee or holder of indebtedness or obligations of Owner Participant, except such as have been duly obtained, or violate or result in a breach of, or constitute a default under, or result in the creation of any Lien (other than as permitted under the Operative Documents) upon the property of Owner Participant under, any indenture, mortgage, contract or other agreement to which Owner Participant is a party or by which Owner Participant or its properties is or are bound or affected. The execution, delivery and performance by Owner Participant of this Agreement and the other Operative Documents to which it is or will be party and the acquisition by Owner Participant of its interest in the Trust Estate (and the rights related thereto) do not and will not violate the [organizational documents] of Owner Participant or any current law, governmental rule, regulation, judgment or order binding on Owner Participant (including, without limitation, any such law, rule, regulation, judgment or order relating to money-laundering, anti-corruption or export control or imposing economic sanctions).
- 5.2.3 <u>Approvals</u>. Neither the execution and delivery by Owner Participant of, nor the performance by Owner Participant of its obligations under, nor the consummation by Owner Participant of the transactions contemplated in, this Agreement and the other Operative Documents to which Owner Participant is or will be a party, requires the consent or approval of, or the giving of notice to, or the registration with, or the taking of any other action in respect of any [jurisdiction of organization] governmental authority having jurisdiction.
- 5.2.4 <u>Valid and Binding Agreements</u>. This Agreement has been duly executed and delivered by Owner Participant and constitutes, and each other Operative Document to which Owner Participant will be a party will be duly executed and delivered by Owner Participant and, when executed and delivered, will constitute, the legal, valid and binding obligation of Owner Participant enforceable against Owner Participant in accordance with its terms, except as may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the rights of creditors generally and general principles of equity.
- 5.2.5 <u>Litigation</u>. There are no pending or, to Owner Participant's knowledge, threatened actions or proceedings before any court, arbitrator or administrative agency which would materially adversely affect the ability of Owner Participant to perform its obligations under this Agreement or any of the other Operative Documents to which Owner Participant is or will be a party.
- 5.2.6 <u>Securities Law</u>. Neither Owner Participant nor any Person authorized to act on its behalf has directly or indirectly offered any interest in the Trust Estate or the Trust Agreement or any similar security to, or solicited any offer to acquire any of the same from, any Person in violation of the registration requirements of the Securities Act or any other applicable securities law.

- 5.2.7 No Liens. On the Delivery Date, there are no Lessor's Liens attributable to it.
- 5.2.8 <u>Citizenship</u>. Either (a) Owner Participant is a Citizen of the United States or (b) the Trust Agreement is in a form that permits the Aircraft to be registered with the FAA in the name of Owner Trustee (without regard to any provision of applicable law that permits FAA registration of an aircraft by limiting its location and usage but with regard to voting trust provisions and provisions delegating certain control rights to Owner Trustee), notwithstanding the failure of Owner Participant to be a Citizen of the United States.
- 5.2.9 <u>ERISA</u>. Either (a) no part of the funds to be used by Owner Participant to make and hold its investment pursuant to this Agreement directly or indirectly constitutes assets of any "employee benefit plan" (as defined in Section 3(3) of ERISA) or of any "plan" (as defined in Section 4975(e) of the Code) or (b) its purchase and holding of its interest in the Trust Estate and its investment pursuant to this Agreement are exempt from the prohibited transaction restrictions of ERISA and the Code pursuant to one or more prohibited transaction statutory or administrative exemptions.
  - 5.2.10 Qualifying Institution. Owner Participant shall be a Qualifying Institution (as defined in Section 8.2(a)(ii)) as of the Delivery Date.
  - 5.2.11 Tax Status. [Owner Participant is a domestic [corporation][partnership] for U.S. federal income tax purposes.]
- 5.3 <u>Representations and Warranties of Owner Trustee and Trust Company</u>. Wells Fargo Bank Northwest, National Association, as Trust Company (except with respect to <u>Sections 5.3.4(b)</u>, <u>5.3.5(b)</u> and <u>5.3.7(b)</u>) and as Owner Trustee, hereby represents and warrants that as of the date hereof:
- 5.3.1 <u>Organization</u>. Trust Company is a national banking association duly organized and validly existing and in good standing under the laws of the United States and has the corporate power, authority and legal right under the laws of the United States pertaining to its banking, trust and fiduciary powers to enter into and perform its obligations under the Operative Documents to which it is or will be a party.

- 5.3.2 <u>Corporate Authorization</u>. The execution, delivery and performance by each of Trust Company and Owner Trustee of this Agreement and the other Operative Documents to which it is or will be party have been duly authorized by all necessary corporate action on the part of Trust Company or Owner Trustee, as the case may be, do not require any stockholder approval or approval or consent of any trustee or holder of indebtedness or obligations of Trust Company or Owner Trustee, except as such as have been duly obtained, and do not and will not violate the certificate of incorporation or by-laws of Trust Company or any current law, governmental rule, regulation, judgment or order binding on Trust Company or Owner Trustee pertaining to its banking, trust or fiduciary powers or violate or result in a breach of, or constitute a default under, or result in the creation of any Lien (other than as permitted under the Operative Documents) upon the property of Trust Company or Owner Trustee under, any indenture, mortgage, contract or other agreement to which Trust Company or Owner Trustee is a party or by which Trust Company or Owner Trustee or its properties is or are bound or affected.
- 5.3.3 <u>Approvals</u>. Neither the execution and delivery by Trust Company or Owner Trustee of, nor the performance by Trust Company or Owner Trustee of its obligations under, nor the consummation by Trust Company or Owner Trustee of the transactions contemplated in, this Agreement and the other Operative Documents to which Trust Company or Owner Trustee is or will be a party, requires the consent or approval of, or the giving of notice to, or the registration with, or the taking of any other action in respect of, any United States federal or Utah state governmental authority having jurisdiction over its banking, trust or fiduciary powers.
- 5.3.4 <u>Valid and Binding Agreements</u>. (a) This Agreement has been duly executed and delivered by Trust Company and constitutes, and each other Operative Document to which Trust Company will be a party will be duly executed and delivered by Trust Company and, when executed and delivered, will constitute, the legal, valid and binding obligation of Trust Company enforceable against Trust Company in accordance with its terms, except as may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the rights of creditors generally and general principles of equity.
- (b) This Agreement has been duly executed and delivered by Owner Trustee and constitutes, and each other Operative Document to which Owner Trustee will be a party will be duly executed and delivered by Owner Trustee and, when executed and delivered, will constitute, the legal, valid and binding obligation of Owner Trustee enforceable against Owner Trustee in accordance with its terms, except as may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the rights of creditors generally and general principles of equity.
- 5.3.5 <u>Litigation</u>. (a) There are no pending or, to Trust Company's knowledge, threatened actions or proceedings before any court, arbitrator or administrative agency which would materially adversely affect the financial condition of Trust Company or the ability of Trust Company to perform its obligations under this Agreement or any of the other Operative Documents to which Trust Company is or will be a party.

- (b) There are no pending or, to Owner Trustee's knowledge, threatened actions or proceedings before any court, arbitrator or administrative agency which would materially adversely affect the financial condition of Owner Trustee or the ability of Owner Trustee to perform its obligations under this Agreement or any of the other Operative Documents to which Owner Trustee is or will be a party.
- 5.3.6 Securities Law. Neither Owner Trustee nor Trust Company nor any Person authorized to act on their respective behalf has directly or indirectly offered any interest in the Trust Estate or the Trust Agreement or any similar security for sale to, or solicited any offer to acquire any of the same from, any Person in violation of the registration requirements of the Securities Act or any applicable securities law.
  - 5.3.7 No Liens; Title. (a) On the Delivery Date, there are no Lessor's Liens attributable to Trust Company.
  - (b) On the Delivery Date, there are no Lessor's Liens attributable to Owner Trustee.
  - (c) On the Delivery Date, Owner Trustee shall have received whatever title to the Aircraft was conveyed to it by Manufacturer.
- 5.3.8 <u>Citizenship</u>. Each of Owner Trustee and Trust Company is a Citizen of the United States (without making use of a voting trust agreement or a voting powers agreement).
- 5.3.9 No Taxes. There are no Taxes imposed by the State of Utah or any political subdivision thereof in connection with the execution and delivery by Trust Company or Owner Trustee of this Agreement or the other Operative Documents to which it is a party or the acquisition by Owner Trustee of its interest in the Aircraft. There are no Taxes imposed by the State of Utah or any political subdivision thereof on Owner Trustee or Trust Company (other than franchise or other taxes based on or measured by any fees or compensation received for services rendered as Owner Trustee) in connection with the ownership and leasing of the Aircraft under, or the performance by Trust Company or Owner Trustee of, the Lease or the other Operative Documents to which it is a party, which Taxes would not have been imposed if the Trust Agreement were not governed by Utah law and neither Trust Company nor Owner Trustee had its principal place of business in, held the Trust Estate in or performed its duties under the Trust Agreement and the other Operative Documents in the State of Utah.

## Section 6. Covenants and Agreements.

# 6.1 Covenants of Lessee.

6.1.1 <u>Corporate Existence</u>; <u>Certificated Air Carrier</u>. Lessee shall at all times maintain its corporate existence (except as permitted by <u>Section 6.1.3</u>) and shall do or cause to be done all things necessary to preserve and keep in full force and effect its rights (charter and

statutory) and franchises to the extent deemed necessary in the good faith judgment of Lessee in the ordinary course of business except for any right or franchise that Lessee determines is no longer necessary or desirable in the conduct of its business. Lessee shall, for as long as and to the extent required under Section 1110 in order that Lessor be entitled to the benefits of Section 1110 with respect to the Aircraft (if any), remain a Certificated Air Carrier.

- 6.1.2 Financial and Other Information. Lessee agrees to furnish Owner Trustee and Owner Participant:
- (a) within 60 days after the end of each of the first three quarterly periods in each fiscal year of Lessee during the Term, either (i) a consolidated balance sheet of Lessee and its consolidated subsidiaries as of the close of such period, together with the related consolidated statements of income for such period, or (ii) a report of Lessee on Form 10-Q in respect of such period in the form filed with the Securities and Exchange Commission;
- (b) within 120 days after the close of each fiscal year of Lessee during the Term, either (i) a consolidated balance sheet of Lessee and its consolidated subsidiaries as of the close of such fiscal year, together with the related consolidated statements of income for such fiscal year, as certified by independent public accountants, or (ii) a report of Lessee on Form 10-K in respect of such year in the form filed with the Securities and Exchange Commission;
- (c) within 120 days after the close of each fiscal year of Lessee during the Term, a certificate of Lessee signed by a Responsible Officer of Lessee, to the effect that the signer has reviewed the relevant terms of the Lease and has made, or caused to be made under his or her supervision, a review of the transactions and condition of Lessee during the accounting period covered by the financial statements referred to in <a href="clause(b)">clause(b)</a> above, and that such review has not disclosed the existence during such accounting period, nor does the signer have knowledge of the existence as of the date of such certificate, of any Event of Default or, if any such Event of Default exists or existed, specifying the nature and period of existence thereof and what action Lessee has taken or is taking or proposes to take with respect thereto; and
  - (d) such other non-confidential information readily available to Lessee without undue expense as Owner Trustee shall reasonably request.

The items required to be furnished pursuant to <u>clause (a)</u> and <u>clause (b)</u> above shall be deemed to have been furnished on the date on which such item is posted on the Securities and Exchange Commission's website at <u>www.sec.gov</u>, and such posting shall be deemed to satisfy the requirements of <u>clause (a)</u> and <u>clause (b)</u>; <u>provided</u> that Lessee shall deliver a paper copy of any item referred to in <u>clause (a)</u> and <u>clause (b)</u> above to Owner Trustee and Owner Participant if Owner Participant so requests.

- 6.1.3 <u>Merger</u>. Lessee shall not consolidate with or merge into any other Person, or convey, transfer or lease all or substantially all of its assets as an entirety to any Person, unless:
- (a) the Person formed by such consolidation or into which Lessee is merged or the Person that acquires by conveyance, transfer or lease all or substantially all of the assets of Lessee as an entirety (the "Successor"):

- (i) if and to the extent required under Section 1110 in order that Lessor continues to be entitled to any benefits of Section 1110 with respect to the Aircraft, shall be a Certificated Air Carrier; and
- (ii) shall execute and deliver to Owner Trustee and Owner Participant an agreement in form reasonably satisfactory to Owner Participant containing an assumption by such Successor of the due and punctual performance and observance of each covenant and condition to be performed or observed by Lessee of each of the Operative Documents to which Lessee is a party;
  - (b) immediately after giving effect to such transaction, no Event of Default caused by such transaction shall have occurred and be continuing; and
- (c) Lessee shall have delivered to Owner Trustee and Owner Participant an officer's certificate and an opinion of counsel (which may be Lessee's General Counsel), each stating that such consolidation, merger, conveyance, transfer or lease and the assumption agreement described in <a href="clause(a)">clause(a)</a> above comply with this <a href="Section 6.1.3">Section 6.1.3</a> and that all conditions precedent herein provided for relating to such transaction have been complied with (except that such opinion need not cover the matters referred to in <a href="clause(b)">clause(b)</a> above and may rely, as to factual matters, on an officer's certificate of Lessee) and, in the case of such opinion, that such assumption agreement has been duly authorized, executed and delivered by the Successor, constitutes its legal, valid and binding obligation and is enforceable against the Successor in accordance with its terms, except as the same may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the rights of creditors generally and by general principles of equity.

Upon any consolidation or merger, or any conveyance, transfer or lease of all or substantially all of the assets of Lessee as an entirety in accordance with this Section 6.1.3, the Successor shall succeed to, be substituted for, and may exercise every right and power of, and shall assume every obligation and liability of, Lessee under the Operative Documents with the same effect as if the Successor had been named as Lessee herein; and thereafter, Lessee shall be released and discharged from all obligations and covenants under the Operative Documents.

- 6.1.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (e) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - 6.2 Covenants of Owner Participant.
- 6.2.1 <u>Owner Trustee Obligations</u>. Owner Participant agrees that it will not direct Owner Trustee to take any action in violation of any agreement or undertaking of Owner Trustee in any of the Operative Documents.
- 6.2.2 <u>Replacement of Owner Trustee</u>. Owner Participant agrees not to remove the institution acting as Owner Trustee, and not to replace the institution acting as Owner Trustee in the event that such institution resigns as Owner Trustee, without in either case having obtained Lessee's prior written consent (such consent not to be unreasonably withheld); <u>provided</u> that Lessee's consent shall not be required if any Event of Default shall have occurred and be continuing. In any such event, a new Owner Trustee selected by Owner Participant which is a Citizen of the United States and, unless an Event of Default shall have occurred and be continuing, is acceptable to Lessee, shall be substituted for Owner Trustee; <u>provided</u> that Owner Participant shall not choose a replacement Owner Trustee which, in the good faith opinion of Lessee, may (or, if an Event of Default shall have occurred and be continuing, Owner Participant shall use its best efforts to select a replacement Owner Trustee which will not) result in additional liability to Lessee pursuant to <u>Section 7.1.</u> or <u>7.2.</u> except in the case of a mandatory or voluntary resignation of Owner Trustee where Lessee has not proposed an alternative Owner Trustee that is reasonably satisfactory to Owner Participant.

- 6.2.3 <u>Certain Payments</u>. Owner Participant agrees to pay or cause Owner Trustee to pay all or any portion of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] pursuant to <u>Section 6.1.4(e)</u> as and when specified in such Section.
- 6.2.4 <u>Replacement Engines</u>. Owner Participant agrees that, in the case of any Replacement Engine substituted pursuant to <u>Section 8(d)</u> of the Lease or any engine substituted pursuant to Section I of the Return Conditions, Owner Trustee is hereby authorized and directed to take the actions specified in <u>Section 8(d)</u> of the Lease or <u>Section I</u> of the Return Conditions, as applicable, with respect to such Replacement Engine or engine, as applicable.

#### 6.3 Covenants of Owner Trustee and Trust Company.

- 6.3.1 <u>FAA and International Registry Correspondence</u>. Owner Trustee agrees to furnish to Lessee copies of (a) all periodic reports sent by it to the FAA (or to the aeronautical authority of the country of registry of the Aircraft if the Aircraft is not registered under the laws of the United States) or the International Registry relating to the Aircraft, (b) all notices, certificates of aircraft registration and other documents and correspondence received by it from the FAA (or from the aeronautical authority of the country of registry of the Aircraft if the Aircraft is not registered under the laws of the United States) or the International Registry relating to the Aircraft and (c) any other notices, assessments, affidavits, instruments or other documents relating to the Aircraft, the Trust Estate or Owner Trustee's ownership thereof in its possession after the date hereof.
- 6.3.2 <u>Distribution of Funds</u>. With respect to any amount stated in the Lease or any other Operative Document to be distributable by Trust Company or Owner Trustee to Lessee or Owner Participant, each of Trust Company and Owner Trustee, upon receipt thereof, agrees to distribute such amount (or cause such amount to be distributed) to Lessee or Owner Participant, as applicable, in accordance with the terms of the Lease or such other Operative Document.
- 6.3.3 <u>Indebtedness and Other Business</u>. Each of Trust Company and Owner Trustee agrees that Owner Trustee will not create, incur or assume any indebtedness for money borrowed, or enter into any business or other activity, except to the extent expressly provided in this Agreement or the other Operative Documents.
- 6.3.4 <u>Trust Administration</u>. Each of Owner Trustee and Trust Company agrees that it will perform all of its administrative duties under this Agreement and the other Operative Documents solely in the State of Utah or in such other location to which the situs of the Trust is moved in accordance with <u>Section 6.4.9</u>, except to the extent necessary to exercise any of its rights or remedies to the extent permitted by applicable laws in connection with an Event of Default or an event that with notice or lapse of time or both would become an Event of Default.

- 6.3.5 Banking Law Filing. Trust Company agrees to make any filing required to be made under Section 131.3 of the New York State Banking Law.
- 6.3.6 <u>Funds Transfer Fees</u>. Each of Owner Trustee and Trust Company agrees that it will not impose, directly or indirectly, any lifting charge, cable charge, remittance charge or any other charge or fee on any transfer by Lessee of funds to, through or by Owner Trustee and Trust Company pursuant to any Operative Document, except as may be otherwise agreed to in writing by Lessee (in which case such agreed charge or fee shall be for Lessee's account).

#### 6.4 Other Covenants.

- 6.4.1 <u>Lessor's Liens</u>. Each of Owner Participant and Trust Company agrees that (a) it shall promptly, at its own cost and expense, take such action as may be necessary duly to discharge and satisfy in full any Lessor's Lien attributable to it if the same shall arise at any time (by bonding or otherwise, so long as Lessee's operation and use of the Aircraft is not impaired); <u>provided</u> that Owner Participant may, for a period of not more than 60 days, contest any such Lessor's Lien diligently and in good faith by appropriate proceedings so long as such contest does not involve any material risk of the sale, forfeiture or loss of use of the Airframe or any Engine or any material risk of criminal penalties or material civil penalties being imposed on Lessee, and (b) it shall indemnify and hold harmless the other parties hereto from and against any loss, cost, Tax, expense or damage (including reasonable legal fees and expenses) that may be suffered or incurred by any of them as a result of its failure to promptly discharge or satisfy in full any such Lessor's Lien.
- 6.4.2 <u>Vesting of Title</u>. Each of Owner Participant, Owner Trustee and Trust Company agrees that in each instance in which the Lease provides that title to the Aircraft, any Engine, engine, Part or Obsolete Part shall be transferred to or vest in Lessee, title to such Aircraft, Engine, engine, Part or Obsolete Part shall vest in Lessee, free and clear of all right, title and interest of such party, Lessor's Liens and Liens of the type described in <u>Section 6(h)</u> of the Lease, and each of Owner Participant, Owner Trustee and Trust Company shall do all acts necessary to discharge all such Liens and other rights held by it in such Aircraft, Engine, engine, Part or Obsolete Part.
- 6.4.3 Quiet Enjoyment. Each of Owner Participant, Owner Trustee and Trust Company agrees that, except as expressly permitted by Section 15 of the Lease following an Event of Default that has occurred and is continuing, notwithstanding anything herein or in any other Operative Document to the contrary, neither it nor any Person claiming by, through or under it shall (a) discharge the registration with the International Registry of the International Interests arising with respect to the Lease, (b) transfer the right to discharge any of such International Interests to any other Person or cause any such right to be so transferred or (c) take or cause to be taken any action inconsistent with Lessee's rights under the Lease and its right to quiet enjoyment of the Aircraft, the Airframe, any Engine or any Part, or otherwise in any way interfere with or interrupt the use, operation and continuing possession of the Aircraft, the Airframe, any Engine or any Part by Lessee or any sublessee, assignee or transferee under any sublease, assignment or transfer then in effect and permitted by the terms of the Lease.

- 6.4.4 <u>Aircraft Registration</u>. Each of Owner Participant, Owner Trustee and Trust Company agrees (a) to execute and deliver all documents and instruments required by the FAA from time to time or as Lessee reasonably requests for the purpose of effecting and continuing the United States registration of the Aircraft pursuant to <u>Section 7(a)</u> of the Lease, (b) to cooperate with Lessee in effecting and continuing any foreign re-registration of the Aircraft pursuant to <u>Section 7(a)</u> of the Lease and (c) to perform all action necessary or appropriate in order for Lessee to have temporary or permanent authority under applicable United States federal law to operate the Aircraft as contemplated by the Lease.
- 6.4.5 <u>Interest in Certain Engines</u>. Each of Owner Participant, Owner Trustee and Trust Company agrees, for the benefit of the lessor, conditional vendor or secured party of any airframe or any engine leased, purchased or owned by Lessee subject to a lease, conditional sale or other security agreement, that it will not acquire or claim, as against such lessor, conditional vendor or secured party, any right, title or interest in any engine or engines as the result of such engine or engines being installed on the Airframe at any time while such engine or engines are subject to such lease, conditional sale or other security agreement.
- 6.4.6 <u>Compliance with Trust Agreement</u>. Each of Owner Participant, Owner Trustee and Trust Company agrees (a) to comply with all of the terms of the Trust Agreement (as the same may hereafter be amended or supplemented from time to time in accordance with the terms thereof and <u>clause (b)</u> of this <u>Section 6.4.6</u>) applicable to it, (b) not to amend, supplement or otherwise modify any provision of the Trust Agreement without Lessee's prior written consent (such consent not to be unreasonably withheld) and (c) notwithstanding anything to the contrary contained in the Trust Agreement, not to terminate or revoke the Trust Agreement or the trusts created by the Trust Agreement without Lessee's prior written consent (such consent not to be unreasonably withheld).
- 6.4.7 Warranties. Each of Owner Participant, Owner Trustee and Trust Company agrees that, so long as no Event of Default shall have occurred and be continuing, Lessee shall, throughout the Term, have the benefit of and shall be entitled to enforce, either in its own name or in the name of Owner Trustee for the use and benefit of Lessee, any and all warranties of any Person (whether express or implied) in respect of the Aircraft, the Airframe, any Engine or any Part, and each of Owner Participant, Owner Trustee and Trust Company agrees to execute and deliver such further documents and take such further action, as may be reasonably requested by Lessee and at Lessee's cost and expense, as may be necessary to enable Lessee to obtain such warranty service or the benefits of any such warranty as may be furnished for the Aircraft, Airframe, any Engine or any Part by such Person. Each of Owner Participant, Owner Trustee and Trust Company hereby appoints and constitutes Lessee, for the duration of the Term except at such times as an Event of Default shall have occurred and be continuing, its agent and attorney-in-fact during the Term to assert and enforce, from time to time, in the name and for the account of Owner Trustee and Lessee, as their interests may appear, but in all cases at the cost and expense of Lessee, whatever claims and rights any of them may have against such Person.
- 6.4.8 <u>Bankruptcy of Trust Estate</u>. Each of Owner Participant, Trust Company and Owner Trustee agrees that it shall not commence or join in any action to subject the Trust Estate or the trust established by the Trust Agreement, as debtor, to the reorganization or liquidation provisions of the Bankruptcy Code or any other applicable bankruptcy or insolvency statute.

- 6.4.9 <u>Change in Situs of Trust</u>. Each of Owner Participant, Trust Company and Owner Trustee agrees that if, at any time, Lessee (a) certifies that Lessee has, or in its good faith opinion will, become obligated to pay an amount pursuant to <u>Section 7.2</u>, and such amount would be reduced or eliminated if the situs of the Trust were changed and (b) requests that the situs of the Trust be moved to another state in the United States from the state in which it is then located, Owner Participant shall direct such change in situs as may be specified in writing by Lessee, and Owner Participant and Owner Trustee shall take whatever action as may be reasonably necessary to accomplish such change. All reasonable out-of-pocket fees and expenses of Owner Participant, Trust Company and Owner Trustee incurred in connection with such a change in situs shall be borne by Lessee. Notwithstanding anything to the contrary contained herein or in any other Operative Document, Owner Participant and Owner Trustee will not consent to or direct a change in the situs of the Trust without the prior written consent of Lessee.
- 6.4.10 <u>Insurance</u>. Each of Owner Participant, Owner Trustee and Trust Company agrees not to obtain or maintain insurance for its own account if such insurance would limit or otherwise adversely affect the coverage or amounts payable under, or increase the premiums for, any insurance required to be maintained pursuant to <u>Section 11</u> of the Lease or any other insurance maintained by Lessee.
- 6.4.11 <u>Stamp Tax</u>. Owner Participant shall pay any stamp, documentation or similar Tax imposed or levied upon or in respect of its execution or delivery of this Agreement or any other Operative Document by any jurisdiction outside the United States in which it (a) is organized, (b) has its principal office or an office through which it is acting hereunder or (c) executes or delivers any such document.

## 6.5 Filings.

- 6.5.1 <u>FAA Filings</u>. On the Delivery Date, Lessee and Owner Trustee will cause the FAA Bill of Sale, the Application for Aircraft Registration, the Lease and Lease Supplement No. 1 to be promptly filed and recorded, or filed for recording, with the FAA to the extent permitted under the Transportation Code or required under any other applicable United States law, in the following order: <u>first</u>, the FAA Bill of Sale; <u>second</u>, the Application for Aircraft Registration; and <u>third</u>, the Lease, to be effected by so filing the Lease with Lease Supplement No. 1 attached thereto.
- 6.5.2 <u>International Registry Filings</u>. On or promptly after the Delivery Date, Lessee and Owner Trustee will cause the registration of the following to be effected on the International Registry in accordance with the Cape Town Treaty in the following order: <u>first</u>, the Sale of the Airframe and Engines from Manufacturer to Owner Trustee; and <u>second</u>, the International Interests created under the Lease (as supplemented by Lease Supplement No. 1). Lessee and Owner Trustee each shall also, as and to the extent applicable, consent to such registrations upon the issuance of a request for such consent by the International Registry.

#### **Section 7.** Indemnification and Expenses.

#### 7.1 General Indemnity.

- 7.1.1 Claims Indemnified. Subject to Section 7.1.2, if the Closing occurs, Lessee agrees to indemnify and hold harmless each Indemnified Person on an After-Tax Basis against any and all Claims imposed on, incurred by or asserted against such Indemnified Person resulting from or arising out of (a) Lessee's use, possession and operation of the Aircraft, including the control, delivery, redelivery, location, pooling, maintenance, repair, substitution, replacement, registration, re-registration, sublease, storage, modification, alteration, return, transfer or other disposition of the Aircraft, the Airframe, any Engine or any Part (including, without limitation, with respect thereto, any such Claim for any death or injury to passengers or others, any such Claim for any damage to the environment, and any such Claim for patent, trademark or copyright infringement) and (b) any incorrectness of any representations or warranties of Lessee contained in any Operative Document, or any failure by Lessee to perform or observe any covenant, agreement or other obligation to be performed or observed by Lessee under the Lease and the other Operative Documents.
- 7.1.2 <u>Claims Excluded</u>. Lessee shall have no obligation to indemnify and hold harmless any Indemnified Person under <u>Section 7.1</u> (or otherwise under the Operative Documents) with respect to Claims described in any one or more of the following subsections:
- (a) Any Claim to the extent attributable to acts or events occurring after the earlier of (i) the return of the Aircraft to Lessor pursuant to Section 5 of the Lease and (ii) the expiration or earlier termination of the Lease except to the extent such Claim arises pursuant to the exercise of remedies in accordance with Section 15 of the Lease in connection with an Event of Default that shall have occurred and be continuing;
- (b) Any Claim that is or is attributable to a Tax (including any Tax benefits), whether or not Lessee is required to indemnify therefor under <u>Section 7.2</u>, it being agreed that <u>Section 7.2</u> sets forth Lessee's entire liability with respect to Taxes, other than Taxes taken into account in order to make an indemnity payment under this <u>Section 7.1</u> on an After-Tax Basis;
- (c) Any Claim to the extent attributable to the gross negligence or willful misconduct of any Indemnified Person (other than the gross negligence or willful misconduct imputed as a matter of law to any Indemnified Person solely by reason of its interest in the Aircraft);
- (d) Any Claim to the extent attributable to the failure by any Indemnified Person to perform or observe any covenant, agreement or other obligation to be performed or observed by it under, or any incorrectness of any representations or warranties of any Indemnified Person contained in, the Lease or any other Operative Document or any agreement relating hereto or thereto to which any such Indemnified Person is a party;
  - (e) Any Claim that constitutes a Permitted Lien;
- (f) Any Claim to the extent attributable to the Transfer (voluntary or involuntary) (i) by any Indemnified Person of any interest in the Aircraft, the Airframe, any

Engine, any Part, the Trust Estate, Rent or any interest arising under any Operative Document, or any similar interest or security, in each case other than such a Transfer pursuant to the Return Conditions or Section 8 or 10 of the Lease or pursuant to the exercise of remedies in accordance with Section 15 of the Lease in connection with an Event of Default that shall have occurred and be continuing, or (ii) of any interest (direct or indirect) in any Indemnified Person;

- (g) Any Claim to the extent attributable to a failure on the part of Owner Trustee to distribute in accordance with the Operative Documents any amounts received and distributable by it thereunder;
- (h) Any Claim to the extent relating to any cost, fee, expense or other payment obligation (i) that is payable or borne by (A) Lessee pursuant to any expense, indemnification, compensation or reimbursement provision of any Operative Document other than this Section 7.1 or (B) a Person other than Lessee pursuant to any provision of any Operative Document or (ii) that such Indemnified Person expressly agrees shall not be payable or borne by Lessee;
  - (i) Any Claim to the extent that it is an ordinary and usual operating or overhead expense;
  - (j) Any Claim resulting from a violation of ERISA or a "prohibited transaction" under Section 4975 of the Code;
- (k) Any Claim that would not have arisen but for the authorization, giving or withholding of any future amendments, supplements, waivers or consents with respect to any Operative Document, other than such (i) as are requested in writing by Lessee, or (ii) that occur as a result of the exercise of remedies in accordance with Section 15 of the Lease in connection with an Event of Default that shall have occurred and be continuing;
- (l) Any Claim that would not have arisen but for any indebtedness, head lease, swap, hedge, or other financing (other than the Lease) arrangements of any Indemnified Person relating to the Aircraft, the Airframe, any Engine, any Part, Rent or any Operative Document; provided that, in the case of any such Claim against any Back-Leveraging Indemnified Person, such Claim shall only be excluded by this subsection (l) to the extent that the nature of such Claim is different than it would have been had such Back-Leveraging Indemnified Person (i) been Lessor or Owner Participant or an officer, director, servant, agent, successor and permitted assign of Lessor or Owner Participant, as applicable, and (ii) entered into the Operative Documents but not the other documents relating to the applicable Back-Leveraging Transaction;
- (m) Any Claim that would not have arisen but for the failure of Trust Company, Owner Trustee or Owner Participant to be a Citizen of the United States: and
- (n) Any Claim that is attributable to or relates to any broker's fee, commission or finder's fee in connection with any transaction contemplated by the Operative Documents (other than such fees of Lessee's lease advisor, SkyWorks Capital, LLC).

- 7.1.3 <u>Indemnified Person</u>. All rights (including, without limitation, the right to receive any indemnity payment under this <u>Section 7.1</u>) of an Indemnified Person and any member of such Indemnified Person's Related Indemnitee Group shall be exercised solely by an Indemnified Person who is a party to this Agreement. If any Indemnified Person fails to comply with this <u>Section 7.1</u>, such Indemnified Person shall not be entitled to indemnity under this <u>Section 7.1</u> with respect to any Claim to the extent (but only to the extent) that Lessee shall have been prejudiced by such failure and that such failure is not the result of or otherwise attributable to the failure of Lessee to comply with any of its duties or obligations under this <u>Section 7.1</u>.
- 7.1.4 <u>Insured Claims</u>. If any Claim indemnified by Lessee is covered by a policy of insurance maintained by Lessee pursuant to <u>Section 11</u> of the Lease, each Indemnified Person agrees to cooperate, at Lessee's expense, with the insurers in the exercise of their rights to investigate, defend or compromise such Claim as may be required to retain the benefits of such insurance with respect to such Claim.
- 7.1.5 <u>Claims Procedure</u>. An Indemnified Person shall promptly notify Lessee of any Claim as to which indemnification is sought. Any amount payable by Lessee to any Indemnified Person pursuant to this <u>Section 7.1</u> shall be paid within 30 days after receipt of a written demand therefor from such Indemnified Person accompanied by a written statement describing in reasonable detail the Claim which is the subject of and basis for such indemnity and the computation of the amount so payable.

Subject to the rights of insurers under policies of insurance maintained by Lessee, Lessee shall have the right to investigate, and the right in its sole discretion to defend or compromise, any Claim for which indemnification is sought under this Section 7.1, and each Indemnified Person shall cooperate with all reasonable requests of Lessee in connection therewith; provided that Lessee shall reimburse such Indemnified Person for all reasonable costs and expenses incurred by it in connection therewith. No Indemnified Person shall enter into a settlement or other compromise with respect to any Claim without the prior written consent of Lessee, unless such Indemnified Person waives its right and the rights of its Related Indemnitee Group to be indemnified with respect to such Claim. Where Lessee or the insurers under a policy of insurance maintained by Lessee undertake the defense of an Indemnified Person with respect to a Claim, no additional legal fees or expenses of such Indemnified Person in connection with the defense of such Claim shall be indemnified hereunder unless such fees or expenses were incurred at the request of Lessee or such insurers; provided, however, that, if in the written opinion (a "Conflict Opinion") of counsel to such Indemnified Person an actual or potential material conflict of interest exists where it is advisable for such Indemnified Person to be represented by separate counsel, the reasonable fees and expenses of such separate counsel shall be borne by Lessee. Subject to the requirements of any policy of insurance maintained by Lessee, an Indemnified Person may participate at its own expense in any judicial proceeding controlled by Lessee pursuant to the precedings, interfere with such party's participation shall not constitute a waiver of the indemnification provided in this Section 7.1. Notwithstanding anything to the contrary contained herein, Lessee shall not under any circumstances be liable for the fees and expenses of more than one counsel for all Indemnified Persons except in

7.1.6 <u>Subrogation</u>. To the extent that a Claim indemnified by Lessee under this <u>Section 7.1</u> is in fact paid in full by Lessee and/or an insurer under a policy of insurance maintained by Lessee, Lessee and/or such insurer, as the case may be, shall, without any further action, be subrogated to the rights and remedies of the Indemnified Person on whose behalf such Claim was paid (other than rights and remedies of such Indemnified Person under insurance policies maintained at its own expense) with respect to the transaction or event giving rise to such Claim. Such Indemnified Person shall give such further assurances or agreements and shall cooperate with Lessee or such insurer, as the case may be, to permit Lessee or such insurer to pursue such rights and remedies, if any, to the extent reasonably requested by Lessee and at Lessee's expense. Should an Indemnified Person receive any payment from any party other than Lessee or its insurers, in whole or in part, with respect to any Claim paid in full by Lessee or its insurers hereunder, such Indemnified Person shall promptly pay the amount so received (but not an amount in excess of the amount Lessee or any of its insurers has paid in respect of such Claim) over to Lessee.

7.1.7 No Guaranty. Notwithstanding anything to the contrary contained in the Lease or in any other Operative Document, Lessee shall not have any responsibility for, or incur any liabilities as a result of, any residual value guaranty, deficiency guaranty or similar agreement in connection with the Aircraft, the Airframe, any Engine or any Part. In addition, nothing set forth in this Section 7.1 shall constitute a guarantee by Lessee that the Aircraft shall have any particular value, useful life or residual value.

## 7.2 General Tax Indemnity

7.2.1 Taxes Indemnified. Subject to Section 7.2.2, if the Closing occurs, Lessee agrees to indemnify and hold harmless each Tax Indemnitee on an After-Tax Basis against any and all Taxes imposed on any Tax Indemnitee, Lessee, the Aircraft, the Airframe, any Engine or any Part upon or with respect to (a) the Aircraft, the Airframe, any Engine or any Part, (b) the lease, possession, operation, use, non-use, control, purchase, sale, delivery, redelivery, location, pooling, maintenance, repair, substitution, replacement, registration, re-registration, purchase, sale, sublease, storage, modification, alteration, return, transfer or other disposition of the Aircraft, the Airframe, any Engine or any Part, (c) any Basic Rent or Supplemental Rent payable by or on behalf of Lessee, (d) any incorrectness of any representations or warranties of Lessee contained in any Operative Document, or any failure of Lessee to perform or observe any covenant, agreement or other obligation to be performed or observed by Lessee, under the Lease or any Operative Document or (e) the exercise of remedies in accordance with Section 15 of the Lease in connection with an Event of Default that shall have occurred and be continuing.

7.2.2 <u>Taxes Excluded from Indemnity</u>. Lessee shall have no obligation to indemnify and hold harmless any Tax Indemnitee under <u>Section 7.2</u> (or otherwise under the Operative Documents) with respect to Taxes described in any one or more of the following subsections; <u>provided</u> that <u>subsections (a)</u> and (e) below shall not apply in determining the additional amount necessary to make any payment on an After-Tax Basis:

- (a) Taxes imposed by any government or taxing authority on, based on, measured by or with respect to capital, net worth, retained earnings, gross or net income or gross or net receipts or proceeds or that are doing business, franchise, minimum or withholding Taxes; provided that this subsection (a) shall not apply to (i) any such Taxes imposed by any government or taxing authority located outside the United States to the extent such Taxes would have been imposed had the sole connection between the Tax Indemnitee and such government or taxing authority been (A) the location, use, operation or presence of the Aircraft, the Airframe, any Engine or any Part in such jurisdiction, (B) the presence or activity of Lessee or any Permitted Sublessee or any Affiliate of either in such jurisdiction or (C) Lessee's (or another Person on its behalf) making a payment from or through such jurisdiction or (ii) any sales, use, goods and services, license, value added or property Taxes, or Taxes of a similar nature, imposed by any government or taxing authority;
- (b) Taxes that would not have been imposed but for (i) any Lessor's Lien, (ii) the gross negligence or willful misconduct of any Tax Indemnitee (other than gross negligence or willful misconduct imputed as a matter of law to such Tax Indemnitee solely by reason of its interest in the Aircraft), (iii) the breach or inaccuracy of any representation, warranty or covenant of any Tax Indemnitee contained in any Operative Document (unless attributable to the breach by Lessee of any representation, warranty or covenant of Lessee contained in any Operative Document), or (iv) a failure of any Tax Indemnitee to comply with any certification, information, documentation, reporting or other similar requirement, if such compliance is necessary or appropriate to claim any relief from such Taxes for which such Tax Indemnitee was eligible, unless such failure is due to the failure of Lessee to comply with its obligations under Section 7.2.5 below;
- (c) Taxes imposed on or with respect to a Transfer (voluntary or involuntary) (i) by a Tax Indemnitee of any interest in the Aircraft, the Airframe, any Engine, any Part, the Trust Estate, Rent or any interest arising under any Operative Document or (ii) of any interest (direct or indirect) in a Tax Indemnitee, in each case other than a Transfer pursuant to the Return Conditions or Section 8 or 10 of the Lease or pursuant to the exercise of remedies in accordance with Section 15 of the Lease in connection with an Event of Default that shall have occurred and be continuing;
- (d) Taxes to the extent imposed with respect to any period commencing after the earlier of (i) the return of the Aircraft to Lessor pursuant to  $\underline{Section\ 5}$  of the Lease and (ii) the expiration or earlier termination of the Lease;  $\underline{provided}$  that there shall not be excluded by this  $\underline{subsection\ (d)}$  any Taxes to the extent (x) attributable to events occurring or matters arising prior to or simultaneously with the earlier of such times, (y) imposed with respect to any payment by Lessee under the Operative Documents after such date or (z) arising pursuant to the exercise of remedies in accordance with  $\underline{Section\ 15}$  of the Lease in connection with an Event of Default that shall have occurred and be continuing;
- (e) Taxes imposed by any government or taxing authority to the extent such Taxes would not have been imposed but for a connection between any Tax Indemnitee or any Affiliate thereof and such government or taxing authority unrelated to the transactions contemplated by the Operative Documents;

- (f) Taxes to the extent such Taxes would not have been imposed but for an amendment or waiver with respect to any Operative Document, unless such amendment or waiver is (i) requested in writing by Lessee or (ii) made as a result of the exercise of remedies in accordance with Section 15 of the Lease in connection with an Event of Default that shall have occurred and be continuing;
- (g) value added Taxes imposed in lieu of a net income Tax by the United States or any state or local government or taxing authority thereof or therein;
  - (h) Taxes resulting from a violation of ERISA or a "prohibited transaction" under Section 4975 of the Code;
  - (i) Taxes on, based on, measured by or with respect to any consideration payable for services rendered by Trust Company as owner trustee;
- (j) Taxes that would not have been imposed but for any indebtedness, head lease, swap, hedge or other financing (other than the Lease) arrangements of any Tax Indemnitee relating to the Aircraft, the Airframe, any Engine, any Part, Rent or any Operative Document; provided that, for the avoidance of doubt, Taxes imposed on a Back-Leveraging Indemnified Person, if any, that has been added as a Tax Indemnitee in a consent and acknowledgement described in Section 8.3.2(b) shall not be treated as described in this clause (j) to the extent that such Taxes would have been imposed on another Tax Indemnitee and would have been subject to indemnification by Lessee under this Section 7 had there been no such indebtedness, head lease, swap, hedge or other financing (other than the Lease) arrangements;
- (k) Taxes in excess of the Taxes that would have been imposed and indemnified against by Lessee hereunder had there not been a Transfer (voluntary or involuntary) (i) by a Tax Indemnitee of any interest in the Aircraft, the Airframe, any Engine, any Part, the Trust Estate, Rent or any interest arising under any Operative Document or (ii) of any interest (direct or indirect) in a Tax Indemnitee, in each case other than a Transfer pursuant to the Return Conditions or Section 8 or 10 of the Lease or pursuant to the exercise of remedies in accordance with Section 15 of the Lease in connection with an Event of Default that shall have occurred and be continuing;
- (l) withholding Taxes imposed by the U.S. federal government that would not have been imposed but for a Tax Indemnitee or any Person holding a direct or indirect interest in the Tax Indemnitee being other than a "United States person" within the meaning of Section 7701(a)(30) of the Code; or
  - (m) Taxes payable by Owner Participant under Section 6.4.11 of this Agreement.
- 7.2.3 <u>Payment</u>. Lessee shall pay any Tax for which it is liable pursuant to this <u>Section 7.2</u> directly to the appropriate taxing authority, if allowable, or, if not so allowable, directly to the relevant Tax Indemnitee. Any amount payable directly to any Tax Indemnitee pursuant to this <u>Section 7.2</u> shall be paid to such Tax Indemnitee on or prior to the later of (<u>a</u>) 30 days after receipt by Lessee of a written demand therefor from such Tax Indemnitee accompanied by a written statement describing in reasonable detail the Taxes that are the subject

of such indemnity and the computation of the amount so payable, (b) one Business Day prior to the due date for the payment of such Taxes (including all extensions) or (c) in the case of amounts that are being contested in accordance with Section 7.2.4, the time such contest (including all appeals, if any) is finally resolved; provided that Lessee shall pay any amounts due pursuant to Section 7.2.4 at the time or times required by such Section. If requested by a Tax Indemnitee in writing, Lessee shall furnish to such Tax Indemnitee the original or a certified copy of a receipt (if any is reasonably available to Lessee) for Lessee's payment of any Tax directly to a taxing authority pursuant to this Section 7.2 or such other evidence of such payment by Lessee as is reasonably acceptable to such Tax Indemnitee and reasonably available to Lessee. If, for any reason, Lessee makes any payment with respect to any Taxes of any Tax Indemnitee that are not the responsibility of Lessee with respect to such Tax Indemnitee under this Section 7, such Tax Indemnitee shall pay to Lessee, within 30 days of Lessee's demand therefor, an amount equal to the amount paid by Lessee with respect to such Taxes.

#### 7.2.4 Contests; Refunds.

If a written claim is made against any Tax Indemnitee for any Tax for which Lessee may be obligated pursuant to this Section 7.2, such Tax Indemnitee shall promptly notify Lessee in writing of such claim. If requested by Lessee in writing, Lessee shall, subject to the conditions set forth in the next paragraph, be entitled at its sole expense to contest such Tax in the name of the relevant Tax Indemnitee or of Lessee through appropriate administrative and judicial proceedings (including pursuing all judicial appeals); provided that (a) no Event of Default under Section 14(a), (b), (f), (g), (h) or (i) of the Lease shall have occurred and be continuing, (b) if such contest shall be conducted in a manner requiring the payment of the Tax, Lessee shall advance to or for the benefit of such Tax Indemnitee (on an interest-free basis) the amount of such payment and shall agree to indemnify such Tax Indemnitee against any adverse tax consequences to such Tax Indemnitee resulting from such interest-free loan and (c) the action to be taken will not result in any material danger of forfeiture, sale or loss of the Aircraft, the Airframe or any Engine (unless Lessee shall have provided to Owner Trustee a bond or other sufficient protection against such risk reasonably acceptable to Owner Trustee) or any material risk of the imposition of criminal penalties. In any contest under this Section 7.2.4 conducted by Lessee, Lessee shall determine the forum and manner in which such contest shall be conducted and, upon the written request of the relevant Tax Indemnitee, shall advise such Tax Indemnitee of the status of such contest, and each Tax Indemnitee shall take reasonable steps to cooperate with Lessee, at Lessee's request and expense, in connection with such contest.

Lessee shall not be permitted to conduct such a contest in its name or in the name of the relevant Tax Indemnitee (and instead a Tax Indemnitee, at Lessee's request, shall contest in its own name as provided in the next paragraph) if  $(\underline{x})$  an Event of Default under Section 14(a),  $(\underline{b})$ ,  $(\underline{f})$ ,  $(\underline{g})$ ,  $(\underline{h})$  or  $(\underline{i})$  of the Lease shall have occurred and be continuing or  $(\underline{y})$  such contest involves issues for which Lessee is not obligated under this Section 7.2 that can not be severed by reasonable efforts of the Tax Indemnitee from all issues for which Lessee might be so obligated.

If requested by Lessee in writing, and if Lessee is not itself contesting a claim under this <u>Section 7.2.4</u>, the relevant Tax Indemnitee shall contest, diligently and in good faith, in the name of such Tax Indemnitee the validity, applicability and amount of the relevant Tax by (<u>I</u>) resisting payment thereof, (<u>II</u>) not paying the same except under protest, if protest be necessary or proper,

or (III) if payment be made, using reasonable efforts to promptly obtain a refund thereof in appropriate administrative and judicial proceedings; provided that (1) Lessee shall have agreed to pay such Tax Indemnitee on demand all reasonable out-of-pocket costs and expenses which such Tax Indemnitee may incur in connection with contesting such claim, including, without limitation, all reasonable legal, accountants' and investigatory fees and disbursements, (2) if such contest shall be conducted in a manner requiring the payment of the Tax, Lessee shall advance to such Tax Indemnitee (on an interest-free basis) the amount of such payment and shall agree to indemnify such Tax Indemnitee against any adverse tax consequences to such Tax Indemnitee resulting from such interest-free loan, (3) the action to be taken will not result in any material danger of forfeiture, sale or loss of the Aircraft, the Airframe or any Engine or Part (unless Lessee shall have provided to Owner Trustee a bond or other sufficient protection against such risk reasonably acceptable to Owner Trustee) and (4) if an Event of Default shall have occurred and be continuing, Lessee shall have provided security for its related tax indemnity obligation reasonably acceptable to such Tax Indemnitee. In any contest under this Section 7.2.4 conducted by a Tax Indemnitee, such Tax Indemnitee shall determine the forum for such contest and the manner in which it shall be conducted; provided that such Tax Indemnitee shall consult in good faith with Lessee and its counsel, and provide to Lessee and its counsel any communications to or from the relevant taxing authority or administrative or judicial body, with respect to the issues for which Lessee may be obligated under this Section 7.2.

If a refund (whether in cash or in any other form) shall be obtained by or for any Tax Indemnitee of all or part of any Tax paid by Lessee or for which Lessee shall have made an advance to, or reimbursed, such Tax Indemnitee, such Tax Indemnitee shall promptly pay Lessee an amount equal to the amount of such refund (which shall reduce the amount of any interest-free loan previously made by Lessee under this Section 7.2.4), together with any interest received on such refund attributable to such Tax that is properly attributable to the period subsequent to such payment or reimbursement by Lessee, reduced by any Taxes payable by such Tax Indemnitee as a result of the receipt or accrual of such refund and interest, and increased by any Tax benefit realized by such Tax Indemnitee as a result of any payment by such Tax Indemnitee pursuant to this sentence; provided that the subsequent loss of a refund for which payment has been made to Lessee under this paragraph shall be treated as an indemnifiable Tax hereunder without regard to the exclusions set forth in Section 7.2.2.

If, without the consent of Lessee, a Tax Indemnitee elects not to, or fails to, contest or cooperate in the contest of any Tax as required in accordance with this Section 7.2.4, or elects to settle, compromise or otherwise terminate any such contest, such election or failure shall constitute a waiver by each Tax Indemnitee of any right to any amount that might otherwise be payable by Lessee pursuant to this Section 7.2 with respect to such Tax (and any other Tax for which a successful contest is materially adversely affected because of such election or failure), other than any expenses of the contest, and, if Lessee has theretofore provided such Tax Indemnitee with an interest-free loan to pay such amount, such Tax Indemnitee shall promptly repay an amount which, after subtraction of any further net savings of Taxes actually realized by such Tax Indemnitee as a result of such repayment, shall be equal to the amount of such interest-free loan, together with interest on the amount of such loan from the date such loan was made to the date of repayment pursuant to this sentence at the rate that would have been paid by the relevant taxing authority had such contest resulted in a refund.

7.2.5 Reports and Returns; Information; Forms. If any report or return is required to be filed with respect to a Tax subject to indemnification by Lessee under this Section 7.2, Lessee shall timely file such report or return in its own name if it is permitted by applicable law to do so (unless Lessee has been notified by the relevant Tax Indemnitee that such Tax Indemnitee intends to file such report or return), showing ownership of the Aircraft in Owner Trustee. If requested by the relevant Tax Indemnitee, Lessee shall send a copy of such report or return to such Tax Indemnitee. If Lessee is not permitted by applicable Law to file any such report or return in its own name, or has insufficient information to do so, Lessee shall, upon obtaining actual knowledge of such requirement, promptly notify the relevant Tax Indemnitee of such requirement and, to the extent it is able to do so, prepare and deliver to such Tax Indemnitee a proposed form of such report or return. Lessee shall furnish to each Tax Indemnitee, and each Tax Indemnitee shall furnish to Lessee, upon the written request of such Tax Indemnitee or Lessee, as the case may be, such data in its possession or otherwise reasonably available to it as may be reasonably requested to enable Lessee or such Tax Indemnitee, as the case may be, as is reasonably necessary to file any such returns or reports and to otherwise comply with the requirements of any taxing authority with respect to the transactions contemplated by the Lease.

Each Tax Indemnitee agrees to furnish from time to time to or as directed by Lessee, upon Lessee's written request and at Lessee's expense, such duly executed and properly completed forms, statements or certificates as may be necessary or appropriate in order to claim any available reduction of any Tax for which Lessee may be obligated under this Section 7.2 or to comply with the requirements of any taxing authority with respect to the transactions contemplated by the Lease; provided that Lessee shall have furnished such Tax Indemnitee with any information necessary to complete such form, statement or certificate that is not otherwise reasonably available to such Tax Indemnitee. If any form, statement or certificate provided by Owner Participant or another Tax Indemnitee to Lessee pursuant to any Operative Document becomes obsolete or incorrect, such Person shall promptly notify Lessee.

#### 7.3 Survival; Other.

7.3.1 <u>Survival</u>. The indemnities and other obligations of Lessee (subject to Sections 7.1.2(a) and 7.2.2(d)), and the obligations of each Indemnified Person and Tax Indemnitee, under <u>Sections 7.1, 7.2</u> and <u>7.3</u> shall survive the expiration or other termination of the Operative Documents.

7.3.2 Tax Savings. If, by reason of any Claims or Taxes paid or indemnified against by Lessee pursuant to Section 7.1 or 7.2, any Indemnified Person or Tax Indemnitee at any time realizes a net reduction in any Taxes not indemnified against by Lessee and not taken into account previously in computing the amount of any indemnity payable by Lessee under Section 7.1 or 7.2, such Indemnified Person or Tax Indemnitee shall promptly pay to Lessee an amount that, after subtraction of any further Tax savings such Indemnified Person or Tax Indemnitee realizes as a result of the payment thereof, is equal to the amount of such net Tax reduction; provided that any subsequent loss of a Tax benefit for which a payment has been made to Lessee under this Section 7.3.2 (or which was taken into account previously in computing an amount payable by Lessee under Section 7.2) shall be treated as an indemnifiable Tax hereunder without regard to the exclusions set forth in Section 7.2.2. Each Indemnified Person and each Tax Indemnitee shall in good faith use diligence in filing tax returns and in dealing with taxing authorities to seek and claim any Tax benefit that would result in such a reduction in Taxes and to minimize the Taxes indemnifiable by Lessee hereunder.

- 7.3.3 <u>Non-Parties</u>. If an Indemnified Person or Tax Indemnitiee is not a party to this Agreement, Lessee may require such Indemnified Person or Tax Indemnitiee to agree in writing, in a form reasonably acceptable to Lessee, to the terms of <u>Section 7.1</u> or <u>7.2</u>, as the case may be, and this <u>Section 7.3</u>, prior to making any payments to such Indemnified Person or Tax Indemnitiee under <u>Section 7.1</u> or <u>7.2</u>, as the case may be.
- 7.3.4 <u>Application of Payments During Event of Default</u>. If, at the time an amount would otherwise be payable to Lessee under <u>Section 7.1, 7.2</u> or <u>7.3.2</u>, an Event of Default shall have occurred and be continuing, such amount shall be held by the relevant Indemnified Person or Tax Indemnitee as security for the obligations of Lessee under the Operative Documents. At such time as no Event of Default is continuing, such amount shall be paid to Lessee.
- 7.3.5 <u>Verification</u>. At the request of Lessee, any computation by an Indemnified Person or a Tax Indemnitee of any amount payable by or to Lessee pursuant to <u>Section 7.1</u>, 7.2 or 7.3.2 shall be verified and certified by a nationally recognized firm of independent accountants selected by the Indemnified Person or Tax Indemnitee, as the case may be, and reasonably acceptable to Lessee. In the event such accounting firm shall determine that the computation of any such amount is incorrect, it shall determine what it believes to be the correct amount, and, absent *prima facie* error, such determination shall be binding upon the parties. Such Indemnified Person or Tax Indemnitee, as the case may be, shall cooperate with such accounting firm and provide it with such information as is reasonably necessary for such verification and certification; <u>provided</u> that such accounting firm shall have entered into a confidentiality agreement reasonably satisfactory to such Indemnified Person or Tax Indemnitee. If Lessee or such Indemnified Person or Tax Indemnitee, as the case may be, has paid any amount under <u>Section 7.1</u>, 7.2 or 7.3.2 prior to such accounting firm's completion of its review, appropriate adjustments will be made promptly after such completion to take into account the determination by such firm. The costs of any such verification and certification shall be borne by Lessee unless such accounting firm determines that any amount payable (a) by Lessee to such Indemnified Person or Tax Indemnitee, as the case may be, is less than 95% of the amount determined by such Indemnified Person or Tax Indemnitiee to be so payable or (b) by such Indemnified Person or Tax Indemnitee, as the case may be, to Lessee is greater than the amount determined by such Indemnified Person or Tax Indemnitiee to be so payable or (b) by such Indemnified Person or Tax Indemnitiee, as the case the cost of such verification and certification shall be paid by such Indemnified Person or Tax Indemnitee. Notwithstanding anything to the contrary i

7.3.6 <u>Withholding Agent</u>. Owner Trustee hereby agrees to act as the U.S. federal withholding Tax agent in respect of Rent and all other amounts payable to it, or distributable by it for or on account of Owner Participant under the Operative Documents, and to be responsible for preparing and filing IRS Forms 1042 and 1042-S (or any similar or successor forms), as well as any other governmental filings and information requirements in connection therewith, and making deposits of U.S. federal withholding Taxes (if any), in accordance with U.S. federal Tax Laws.

7.4 Expenses. Except as otherwise provided in this Section 7.4, each of Lessee and Owner Participant will be responsible for its own costs and expenses incurred in connection with the negotiation, preparation, execution and delivery of the Operative Documents. Lessee agrees promptly to pay (a) all the reasonable out-of-pocket costs and expenses incurred by Trust Company in connection with the negotiation, preparation, execution and delivery of the Operative Documents (including, without limitation, the reasonable fees, expenses and disbursements of [Ray, Quinney & Nebeker P.C.,] special counsel for Trust Company); and (b) the reasonable fees, expenses and disbursements of Aviation Counsel in connection with the negotiation, preparation, execution and delivery of the Operative Documents. Lessee also agrees to pay all costs and expenses imposed by the FAA, the International Registry and the State of Delaware in connection with the registrations and filings described in Section 5.1.7. Lessee agrees to pay the initial and on-going fees of Trust Company in connection with the transactions contemplated hereby during the Term of the Lease.

# Section 8. Assignment or Transfer of Interests.

- 8.1 <u>Owner Trustee</u>. Except as expressly provided in the Operative Documents, Owner Trustee shall not, directly or indirectly, Transfer any of its right, title or interest in and to the Aircraft, any of the Operative Documents, the Trust Estate or any proceeds therefrom without the prior written consent of Lessee; <u>provided</u> that such consent shall not be required for a Transfer pursuant to the exercise of remedies by Owner Trustee under and in accordance with <u>Section 15</u> of the Lease in connection with an Event of Default that shall have occurred and be continuing, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (vi) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (vii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (viii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ix) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (x) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (xi) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (xii) CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### 8.2 Owner Participant.

- (a) <u>Owner Participant Transfer Requirements</u>. Owner Participant shall not directly or indirectly Transfer any of its right, title or interest in and to all or any part of this Agreement, any of the other Operative Documents or the Trust Estate, except that Owner Participant may Transfer all (but not less than all) of its right, title and interest therein to a single bank, lending institution, leasing company, other financial institution, corporation, limited partnership, statutory trust, limited liability company or special purpose entity if, as preconditions to such Transfer:
  - (i) (A) the Person to whom such Transfer is made (the "**Transferee**") either is a Citizen of the United States or qualifies as a Citizen of the United States through a voting trust agreement, voting powers agreement or similar arrangement (including, without limitation, provisions delegating certain control rights to the Owner Trustee) by the Transferee or any Affiliate thereof, but in each case without reliance on any rule that would restrict in any way the use and operation of the Aircraft, and has the requisite power, authority and legal right to enter into and carry out the transactions contemplated by the Operative Documents; (B) unless Lessee consents, the Transferee is not an airline or other commercial operator of aircraft, freight forwarder, or any other company directly or indirectly engaged in the business of passenger, cargo, freight or parcel transportation or any Affiliate of any thereof; (C)(1) the Transfer does not violate the Transportation Code, the Securities Act or any other Law (including, without limitation, ERISA, any laws or regulations imposing U.S. economic sanctions measures or any orders or licenses issued thereunder), or create a relationship that would be in violation thereof, (2) the Transfer does not result in a "prohibited transaction" under Section 4975 of the Code, (3) the Transfer does not adversely affect the registration of the Aircraft in the name of Owner Trustee with the FAA (or the aeronautical authority of the

country of registry of the Aircraft if the Aircraft is not registered under the laws of the United States), (4) the Transfer will not subject Lessee to any additional regulation under, or require Lessee to give any notice to, register with, make any filings with or take any other action in respect of, any governmental authority or agency of any jurisdiction, (5) the Transfer does not require registration under the Securities Act or any foreign securities laws, require qualification of an indenture under the Trust Indenture Act, or require Lessee to sign any registration statement, (6) unless Lessee consents, the Transfer does not involve a Rule 144A, Regulation S or other capital markets or equity syndication transaction not described in the immediately preceding clause (5), and (7) the Transfer does not result in, or involve, incurrence by Lessee of any indebtedness for accounting purposes (it being understood that, if any change in the lease accounting standards applicable to Lessee requires that Lessee, independently of the Transfer, capitalize its leases, including the Lease, in Lessee's books, such capitalization of the Lease is not intended to constitute, and shall not be construed as, incurrence by Lessee of any indebtedness for accounting purposes within the meaning of this clause (7)); (D) Owner Participant and the Transferee shall have entered into an agreement in the form attached hereto as Exhibit E (the "Assumption Agreement") or in such other form as shall be acceptable to Lessee; (E) Owner Participant shall have delivered to Owner Trustee and Lessee an opinion or opinions of counsel (which shall either be the in-house counsel of the Transferee or other counsel reasonably satisfactory to Lessee) to the effect that the Assumption Agreement has been duly authorized, executed and delivered by the Transferee and is enforceable against the Transferee in accordance with its terms, except as the same may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the rights of creditors generally or by general principles of equity and to the effect that (subject to customary exceptions, qualifications and exclusions) such Transfer complies with clause (A) (except as to citizenship) and clause (C) (with respect to the Transportation Code and the Securities Act and no violation of Law) above (provided that, in determining observance with all factual matters contained in this Section, such counsel may rely on representations of the Transferee); and (F) the Transferee shall have provided to each of Lessee and Owner Trustee a duly completed and executed original IRS Form W-9 (and/or other applicable IRS Form(s)) establishing a complete exemption from U.S. federal withholding taxes with respect to all payments of Rent or other amounts to or for the benefit of Owner Trustee or Owner Participant under the Operative Documents; and

(ii) except with the consent of Lessee, either [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] (IV) is otherwise approved in writing by Lessee, such approval not to be unreasonably withheld or delayed (any Transferee meeting any of the requirements of (I), (II), (III) or (IV) above being hereinafter referred to as a "Qualifying Institution"), or (B) if the Transferee is not itself a Qualifying Institution, a parent corporation of the Transferee which qualifies as a Qualifying Institution shall have executed and delivered to Owner Trustee and Lessee an absolute and unconditional guaranty, substantially in the form of Exhibit G or otherwise in form and substance reasonably satisfactory to Lessee ([the guaranty by

[Name of initial Owner Participant Guarantor] or any other<sup>7</sup>] [any] such guaranty, an "Owner Participant Guarantee") and [Name of initial Owner Participant Guarantor or] such parent the "Owner Participant Guarantor"), with respect to the obligations undertaken by the Transferee under the Assumption Agreement referred to above, together with an opinion of counsel (which may be the in-house counsel of the Qualifying Institution providing such guaranty or other counsel reasonably satisfactory to Lessee) to the effect that such guaranty is enforceable against the guarantor in accordance with its terms. The Transferee shall, at the time of Transfer, deliver to Owner Trustee and Lessee a certificate of a duly authorized officer of the Transferee or its guarantor evidencing satisfaction of the requirements of (I), (II) or (III), as applicable, set forth in this clause (ii).

It shall be a further condition to any such Transfer, and the parties hereby agree, that: (x) as determined at the time of such Transfer, none of Lessee's obligations, responsibilities, liabilities, costs and risks in the use and operation of the Aircraft or under, relating to or in respect of the Operative Documents or otherwise, including, without limitation, under or in respect of any of Lessee's payment or indemnity obligations, shall be increased or altered, and none of Lessee's rights and benefits under any Operative Document shall be diminished, as a result of or in connection with such Transfer or any aspect thereof or any other transaction relating thereto (it being acknowledged that an increase in the number of Indemnified Persons or Tax Indemnitees shall not, of itself, constitute an increase in Lessee's obligations under the Operative Documents); and (y) Lessee shall have no obligation, responsibility or liability of any kind under, relating to or in respect of such Transfer or any aspect thereof or any other transaction relating thereto, except acknowledging acceptance of the Assumption Agreement.

The transferor Owner Participant will pay or cause the Transferee to pay any fees, costs, charges and expenses incurred by Owner Trustee, Trust Company, Lessee or any other party in connection with any such Transfer (including, without limitation the reasonable out-of-pocket expenses of Lessee and its legal fees and expenses) whether or not such Transfer is consummated, and in no case will Lessee be responsible for (and Owner Participant will hold Lessee harmless from) any such fees, charges or expenses or for any fees, charges or expenses incurred by any party to a Back-Leveraging Transaction in connection with such Transfer.

Insert if applicable.

(b) Effect of Transfer. Upon any such Transfer by Owner Participant to a Transferee permitted by this Section 8.2, the Transferee shall be deemed the "Owner Participant" for all purposes hereof (unless the context is inappropriate) and each reference herein or in any other Operative Document to "Owner Participant" shall thereafter be deemed a reference to the Transferee as Owner Participant (unless the context is inappropriate). Upon any such Transfer by Owner Participant to a Transferee permitted by this Section 8.2, the transferor Owner Participant shall be relieved of all of its duties, liabilities and obligations hereunder and under the Trust Agreement that have been expressly assumed by such Transferee; provided that in no event will any such Transfer release the transferor Owner Participant from any duty, liability or obligation (i) arising or relating to any event occurring prior to the effective time of such Transfer, (ii) on account of any breach by the transferor Owner Participant of any of its representations, warranties, covenants or obligations contained herein or in any other Operative Document or any Assumption Agreement, or for any fraudulent or willful misconduct engaged in by the transferor Owner Participant, (iii) that relates to any indemnity claimed by the transferor Owner Participant or (iv) relating to or arising out of any Lessor's Lien attributable to the transferor Owner Participant. If Owner Participant proposes to Transfer its interests pursuant to this Section 8.2, it shall give at least 10 days prior written notice thereof to Owner Trustee and Lessee, specifying the name and address of the Transferee and the facts necessary to determine whether the conditions of this Section 8.2 have been or will be satisfied.

8.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Section 9. Change of Citizenship.

9.1 <u>Generally</u>. Without prejudice to the representations, warranties or covenants regarding the status of any party hereto as a Citizen of the United States, each of Owner Participant, Owner Trustee and Trust Company agrees that, during the Term, in the event its status is to change or has changed as a Citizen of the United States or it makes public disclosure of circumstances as a result of which it believes that such status is likely to change, it will notify all the other parties to this Agreement of (a) such change in status promptly after obtaining actual knowledge thereof or (b) such belief as soon as practicable after such public disclosure but in any event within 10 Business Days after such public disclosure.

9.2 Owner Participant. Owner Participant covenants that if, at any time during the Term when the Aircraft is registered in the United States, Owner Participant is not or ceases to be a Citizen of the United States and the Aircraft would thereupon become ineligible for registration in the name of Owner Trustee under the Transportation Code as in effect at such time and the regulations then applicable thereunder (without regard to any "based and primarily used" provision, or other provision that in any way could restrict the use and operation of the Aircraft by Lessee but with regard to voting trust provisions and provisions delegating certain control rights to the Owner Trustee), then Owner Participant at its own expense shall promptly (and, in any event, within a period of 30 days) either transfer, pursuant to Article VIII of the Trust Agreement and Section 8.2 hereof, its right, title and interest in and to the Trust Agreement, the Trust Estate and this Agreement, or take such other action, as may be necessary to prevent any deregistration of the Aircraft or to make possible its registration in the United States (without regard to any "based and primarily used" provision, or other provision that in any way could restrict the use and operation of the Aircraft by Lessee but with regard to voting trust provisions and provisions delegating certain control rights to the Owner Trustee), as the case may be. Each party hereto agrees to take such steps, at Owner Participant's expense, as Owner Participant shall reasonably request in order to assist Owner Participant in complying with its obligations under this Section 9.2. Owner Participant agrees to indemnify and hold harmless the other parties hereto for any and all losses, liabilities, costs and expenses incurred by such parties arising from the failure of the Aircraft to be eligible for registration in the name of Owner Trustee attributable to Owner Participant's failure to be a Citizen of the United States at any time during the Term.

9.3 <u>Owner Trustee</u>. Trust Company covenants that, if at any time when the Aircraft is registered in the United States Trust Company is not or ceases to be a Citizen of the United States and the Aircraft would thereupon become ineligible for registration in the name of Owner Trustee under the Transportation Code as in effect at such time and the regulations then applicable thereunder (without regard to any "based and primarily used" provision, or other provision that in any way could restrict the use and operation of the Aircraft by Lessee), Trust Company shall resign immediately as Owner Trustee in accordance with <u>Section 9.01</u> of the Trust Agreement. Trust Company agrees to indemnify and hold harmless the other parties hereto for any and all losses, liabilities, costs and expenses incurred by such parties arising from Trust Company's failure to be a Citizen of the United States at any time during the Term.

#### Section 10. Miscellaneous.

10.1 Notices. Unless otherwise expressly specified or permitted by the terms hereof, all notices, requests, demands, authorizations, directions, consents or waivers required or permitted under the terms and provisions of this Agreement shall be in English and in writing, and given by United States registered or certified mail, return receipt requested, postage prepaid, overnight courier service or facsimile, and any such notice shall be effective when received (or, if delivered by facsimile, upon completion of transmission and confirmation by the sender (by a telephone call to a representative of the recipient or by machine confirmation) that such transmission was received) and addressed as follows:

#### (a) if to Lessee:

American Airlines, Inc. 4333 Amon Carter Boulevard, MD 5662 Fort Worth, Texas 76155

Attention: Treasurer Facsimile: 817.967.4318 Telephone: 817.963.1234

## (b) if to Owner Participant:

[Name of Owner Participant]
[Address of Owner Participant]
Attention:
Facsimile:
Telephone:

## (c) if to Owner Trustee:

Wells Fargo Bank Northwest, National Association

MAC: U1228-120

299 South Main Street, 12th Floor

Salt Lake City, UT 84111

Attention: Corporate Trust Services

Facsimile: 801.246.5053 Telephone: 801.246.2755

With a copy to the Owner Participant.

Any party, by notice to the other parties hereto, may designate different addresses for subsequent notices or communications. Whenever the words "notice" or "notify" or similar words are used herein, they mean the provision of formal notice as set forth in this <u>Section 10.1</u>.

10.2 <u>Late Payments; Business Days; Currency</u>. In the event that any amounts required to be paid hereunder are not paid when due, such amounts shall bear interest, to the extent permitted by applicable law, from the due date thereof to, but not including, the date such amount is paid, at the Overdue Rate. If any amount required to be paid hereunder is due on a day that is not a Business Day, such amount shall be paid on the next succeeding Business Day with the same force and effect as if paid on the scheduled date of payment, and no interest shall accrue on the amount of such payment from and after such scheduled date to the time of payment on such next succeeding Business Day. All payment obligations by the parties hereto under the Operative Documents shall be payable in U.S. Dollars.

10.3 <u>Concerning Owner Trustee</u>. Wells Fargo Bank Northwest, National Association is entering into the Operative Documents solely in its capacity as Owner Trustee under the Trust Agreement and not in its individual capacity (except as expressly provided in the Operative Documents), and in no case shall Wells Fargo Bank Northwest, National Association (or any entity acting as successor Owner Trustee under the Trust Agreement) be personally liable for or

on account of any of the statements, representations, warranties, covenants or obligations stated to be those of Lessor or Owner Trustee under the Operative Documents; provided, however, that Wells Fargo Bank Northwest, National Association (or any successor Owner Trustee) shall be personally liable under the Operative Documents for its own gross negligence, its own simple negligence in the handling of funds actually received by it in accordance with the terms of the Operative Documents, its willful misconduct and its breach of its covenants, representations and warranties in the Operative Documents, to the extent covenanted or made in its individual capacity or as otherwise expressly provided in the Operative Documents; provided, further, that nothing contained in this Section 10.3 shall be construed to limit the exercise and enforcement in accordance with the terms of the Operative Documents of rights and remedies against the Trust Estate.

10.4 Confidential Information. All Confidential Information shall be held confidential by each of Owner Trustee, Trust Company and Owner Participant and shall not, without the prior written consent of Lessee, be furnished or disclosed to anyone other than (a) such party's bank examiners, auditors, accountants, agents and legal counsel, each with an absolute need to know such information; (b) any Person with whom such party is in good faith conducting negotiations relating to the possible Back-Leveraging Transaction or permitted transfer, sale or other disposition of its rights and obligations under this Agreement, the Lease and the other Operative Documents, if such Person shall have entered into an agreement for the express benefit of Lessee to hold such Confidential Information confidential in accordance with the provisions of this Section 10.4; (c) except as may be required by an order of any court or administrative agency or by any statute, rule, regulation or order of any governmental authority; and (d) except to the extent such Confidential Information becomes publicly available or becomes available on a non-confidential basis from a source other than any party to the Operative Documents or any Affiliate thereof. Notwithstanding anything to the contrary in the Operative Documents, except as reasonably necessary to comply with applicable securities law, the parties to the Operative Documents (and their respective employees, representatives and agents) may disclose to any and all persons, without limitation of any kind, the United States federal or state income tax treatment and tax structure of the transaction contemplated thereby and all materials of any kind provided to them relating to such tax treatment and tax structure. For this purpose, "tax structure" means any facts relevant to the United States federal or state income tax treatment of such transaction, but (unless otherwise required by applicable Law) does not include information relating to the identity of the parties. The obligations set forth

10.5 <u>Further Assurances</u>. Each party hereto shall execute, acknowledge and deliver or shall cause to be executed, acknowledged and delivered, all such further agreements, instruments, certificates or documents, and shall do and cause to be done such further acts and things, including, without limitation, making or consenting to registrations on the International Registry with respect to the Lease contemplated by <u>Section 2</u> and appointing Aviation Counsel as its "professional user entity" (as defined in the Cape Town Treaty) to make or consent to any registrations or discharges on the International Registry with respect to the Airframe or any Engine, in any case, as any other party hereto shall reasonably request in connection with the administration of, or to carry out more effectively the purposes of, or to better assure and confirm into such other party the rights and benefits to be provided under this Agreement; <u>provided</u> that this sentence is not intended to impose upon Lessee any additional liabilities not contemplated by this Agreement.

10.6 <u>Third Party Beneficiary</u>. Except for Indemnified Persons and Tax Indemnitees not a party hereto (each of which shall be deemed to be express third party beneficiaries with respect to the provisions of <u>Section 7.1</u> or <u>7.2</u>, as the case may be, subject to <u>Section 7.3.3</u>), this Agreement is not intended to, and shall not, provide any Person not a party hereto with any rights of any nature whatsoever against any of the parties hereto and no Person not a party hereto shall have any right, power or privilege in respect of any party hereto, or have any benefit or interest, arising out of this Agreement.

# 10.7 Miscellaneous.

- 10.7.1 <u>Severability</u>. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof or affecting the validity or enforceability of such provision in any other jurisdiction.
- 10.7.2 <u>Amendments</u>. No term or provision of this Agreement may be amended, modified or supplemented orally, but only by an instrument in writing signed by the party against which the enforcement of the amendment, modification or supplement is sought.
- 10.7.3 <u>Prior Agreements</u>. This Agreement and the other Operative Documents, and all certificates, instruments and other documents relating thereto delivered and to be delivered from time to time pursuant to the Operative Documents, supersede any and all representations, warranties and agreements (other than any Operative Document) prior to the date of this Agreement, written or oral, between or among any of the parties hereto relating to the transactions contemplated hereby and thereby.
- 10.7.4 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts (and each of the parties hereto shall not be required to execute the same counterpart). Each counterpart of this Agreement, including a signature page executed by each of the parties hereto shall be an original, but all of such counterparts together shall constitute one instrument.
- 10.7.5 <u>Binding Agreement</u>. This Agreement shall be binding upon and inure to the benefit of, Owner Participant and, subject to the provisions of <u>Section 8.2</u>, its successors and permitted assigns, Owner Trustee and its successors as Owner Trustee under the Trust Agreement, Trust Company and its successors and permitted assigns, and Lessee and, subject to the terms of <u>Section 6.1.3</u>, its successors and permitted assigns.
- 10.7.6 No Waiver. No failure on the part of Owner Participant, Owner Trustee, Trust Company or Lessee to exercise, and no delay in exercising, any right, power or privilege under this Agreement or any other Operative Document shall operate as a waiver thereof; nor shall any single or partial exercise of any right, power or privilege hereunder or thereunder preclude any other or further exercise thereof or the exercise of any other right, power or privilege hereunder and thereunder. Except as may be expressly limited herein or by any other Operative Document, the remedies herein provided are cumulative and not exclusive of any remedies provided by Law.

10.7.7 Governing Law; Jurisdiction. THIS AGREEMENT SHALL IN ALL RESPECTS BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK, INCLUDING ALL MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE. THIS AGREEMENT HAS BEEN DELIVERED IN THE STATE OF NEW YORK. In relation to any legal action or proceeding arising out of or in connection with this Agreement or any other Operative Document, each of Owner Participant, Trust Company, Owner Trustee and Lessee (a) irrevocably submits to the nonexclusive jurisdiction of each of the Supreme Court of the State of New York, New York County and the United States District Court for the Southern District of New York, and other courts with jurisdiction to hear appeals from such courts, and (b) to the maximum extent permitted by applicable Law, waives, and agrees not to assert, by way of motion, as a defense, or otherwise, in any such action or proceeding, that the action or proceeding is brought in an inconvenient forum, that the venue of the action or proceeding is improper or that this Agreement or any other Operative Document or the subject matter hereof or thereof or any of the transactions contemplated hereby or thereby may not be enforced in or by such courts. [Owner Participant irrevocably designates and appoints [name of process agent] as process agent to receive for it and on its behalf service of process in any proceedings arising hereunder or under any other Operative Document to which it is a party. Nothing in this Agreement shall affect the right to serve process in any other manner permitted by law.]<sup>8</sup>

10.7.8 Section 1110. It is the intention of the parties hereto that the Lease, to the fullest extent available under applicable law, entitles Lessor to the benefits of Section 1110 with respect to the Aircraft. In the furtherance of the forgoing, the parties hereby confirm that the Lease is to be treated as a lease for U.S. federal income tax purposes. Nothing contained in this paragraph shall be construed to limit Lessee's use and operation of the Aircraft under the Lease or constitute a representation or warranty by any party as to tax consequences.

10.7.9 <u>Waiver of Immunity</u>. To the extent that Owner Participant or any of its properties has or may hereafter acquire any right of immunity, whether characterized as sovereign immunity or otherwise, and whether under the United States Foreign Sovereign Immunities Act of 1976 (or any successor legislation) or otherwise, from any legal proceedings, whether in the United States or elsewhere, to enforce or collect upon any Operative Documents to which it is a party, including, without limitation, immunity from suit or service of process, immunity from jurisdiction or judgment of any court or tribunal or execution of a judgment, or immunity of any of its property from attachment prior to any entry of judgment, or from attachment in aid of execution upon a judgment, Owner Participant hereby irrevocably and expressly waives any such immunity, and agrees not to assert any such right or claim in any such proceeding, whether in the United States or elsewhere.

[Remainder of Page Intentionally Left Blank; Signature Page Follows]

Include only if there is a foreign Owner Participant.

**IN WITNESS WHEREOF**, the parties hereto have caused this Participation Agreement to be duly executed as of the day and year first above written.

By:	
Name:	
Title:	
[NAME	OF OWNER PARTICIPANT]
By:	
Name:	
Title:	
	S FARGO BANK NORTHWEST, NATIONAI
WELLS	
	-
ASSOC	<b>IATION</b> , not in its individual capacity (except a y provided herein) but solely as Owner Trustee
ASSOC	<b>IATION</b> , not in its individual capacity (except a
ASSOC	<b>IATION</b> , not in its individual capacity (except a
ASSOC expressl	<b>IATION</b> , not in its individual capacity (except a

AMERICAN AIRLINES, INC.

# FORM OF OPINION OF [ASSOCIATE] GENERAL COUNSEL FOR LESSEE

[Letterhead of American Airlines, Inc.]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# FORM OF OPINION OF [ASSOCIATE] GENERAL COUNSEL FOR LESSEE

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- $1. \ [\text{CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]\\$
- 2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

EXHIBIT A

# FORM OF OPINION OF [ASSOCIATE] GENERAL COUNSEL FOR LESSEE

- 7. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 8. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# FORM OF OPINION OF [ASSOCIATE] GENERAL COUNSEL FOR LESSEE

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# FORM OF OPINION OF [ASSOCIATE] GENERAL COUNSEL FOR LESSEE

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# SCHEDULE I

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## FORM OF OPINION OF SPECIAL COUNSEL FOR OWNER TRUSTEE

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**EXHIBIT B** 

# FORM OF OPINION OF SPECIAL COUNSEL FOR OWNER TRUSTEE

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

EXHIBIT B

# FORM OF OPINION OF SPECIAL COUNSEL FOR OWNER TRUSTEE

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

7. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- A. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- B. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- C. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- D. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- E. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

# FORM OF OPINION OF SPECIAL COUNSEL FOR OWNER TRUSTEE

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- F. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- G. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- H. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- I. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- J. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**EXHIBIT B** 

# SCHEDULE A

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**EXHIBIT C** 

# [INTENTIONALLY LEFT BLANK]

## **EXHIBIT D**

## FORM OF OPINION OF AVIATION COUNSEL

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (b) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

#### EXHIBIT D

## FORM OF OPINION OF AVIATION COUNSEL

- (c) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (d) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 1. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

EXHIBIT D

#### **EXHIBIT D**

## FORM OF OPINION OF AVIATION COUNSEL

- 7. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 8. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 9. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

N /filop

## SCHEDULE 1

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

EXHIBIT D

# SCHEDULE 2

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (vi) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (vii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (viii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ix) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

EXHIBIT D

- (x) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (xi) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (xii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (xiii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (xiv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (xv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

(i) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

EXHIBIT D

- (ii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (iv) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (v) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (vi) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (vii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (viii) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (ix) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

EXHIBIT D

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

#### **EXHIBIT E**

# FORM OF ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS ASSIGNMENT AND ASSUMPTION AGREEMENT ([YEAR] MSN [MSN]), dated as of [ ] (as amended, modified or supplemented from time to time, this "Agreement)", among (i) [NAME OF ASSIGNOR], a [jurisdiction] [type of entity] (together with its successors and permitted assigns, "Assignor"), (ii) [NAME OF ASSIGNEE], a [jurisdiction] [type of entity] (together with its successors and permitted assigns, "Assignee") and (iii) AMERICAN AIRLINES, INC., a Delaware corporation (together with its successors and permitted assigns, "Lessee").

#### **RECITALS:**

- 1. Reference is made to one Airbus [Model] aircraft bearing the manufacturer's serial number [MSN] and U.S. Registration No. [Reg. No.] (as more fully described in the Participation Agreement referred to below, the "Aircraft").
- 2. Assignor and Assignee desire to effect (a) the transfer by Assignor to Assignee of all of the right, title and interest of Assignor (except as reserved below) in and to (i) the Operative Documents, (ii) the Trust Estate and (iii) the proceeds from any of the foregoing and (b) the assumption by Assignee of the obligations of Assignor accruing under the Operative Documents (such transfer and assumption, the "Assignment and Assumption").
- 3. The Participation Agreement ([YEAR] MSN [MSN]), dated as of [ ], among Lessee, Assignor, as Owner Participant, and Wells Fargo Bank Northwest, National Association, a national banking association, not in its individual capacity except as expressly provided therein, but solely as Owner Trustee (as amended, modified or supplemented from time to time, the "Participation Agreement") permits such Assignment and Assumption upon satisfaction of certain conditions heretofore or concurrently herewith being complied with.

**NOW, THEREFORE**, in consideration of the premises and mutual agreements herein contained, the agreements contained in the Operative Documents and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- 1. <u>Definitions</u>. Unless the context otherwise requires, all capitalized terms used herein and not otherwise defined herein shall have the meanings set forth, and shall be construed and interpreted in the manner described, in the Participation Agreement.
  - 2. Conditions to Effectiveness; Effective Time.
    - (a) Prior to the Effective Time or, in the case of <u>subclause [(y)] [(yi)] [(yii)]</u>, at the Effective Time:
  - (i) Assignor or Assignee shall have paid or reimbursed Owner Trustee, Trust Company, Lessee or any other party for any fees, charges or expenses incurred by Owner Trustee, Trust Company, Lessee or any such party in connection with the Assignment and Assumption (including, without limitation the reasonable out-of-pocket expenses of Lessee and its legal fees and expenses);

EXHIBIT E

(ii) Assignee shall have provided to each of Lessee and Owner Trustee a duly completed and executed original IRS Form [	] <sup>9</sup> establishing a
complete exemption from U.S. federal withholding Taxes with respect to all payments of Rent or other amounts to or for the benefit of	Owner Trustee o
Owner Participant under the Operative Documents;	

- (iii) Assignee shall have delivered to each of Lessee and Owner Trustee a legal opinion of [ ] in accordance with <u>Section 8.2(a)</u> of the Participation Agreement;
- (iv) Assignee shall have delivered to each of Lessee and Owner Trustee a certificate of a duly authorized officer of [Assignee/Owner Participant Guarantor] in accordance with <u>Section 8.2(a)</u> of the Participation Agreement;
- (v) [Owner Participant Guarantor shall have delivered to each of Lessee and Owner Trustee [describe the Owner Participant Guarantee] in accordance with <u>Section 8.2(a)</u> of the Participation Agreement;]
- (vi) [Assignee shall have provided to each of Lessee and Owner Trustee evidence of Assignee's appointment of the process agent as provided in Section 11(c) and of such process agent's acceptance of such appointment; [10]; and
- (vii) the representations and warranties of Assignor and Assignee made herein shall be correct and accurate in all material respects, in each case as though made on and as of such date, or if such representations and warranties relate solely to an earlier date, as of such earlier date.
- (b) Subject to the satisfaction or waiver of the conditions set forth in <u>subsection (a)</u> by the parties hereto, this Agreement shall become effective at [a.m.][p.m.]] on [ ] (the "**Effective Time**").
- 3. <u>Assignment</u>. Assignor has transferred, and does hereby transfer unto Assignee, as of the Effective Time, all of its present and future right, title and interest in and to the Operative Documents, the Trust Estate, and any proceeds from the foregoing, except such rights of Assignor as have arisen or accrued to Assignor prior to the Effective Time (including specifically, but without limitation, the right to receive any amounts due or accrued to Assignor under the Operative Documents as of a time prior to the Effective Time and the right to receive any indemnity payment pursuant to the Participation Agreement with respect to events occurring prior to such time), in each case subject to the rights of Lessee thereunder.

Insert the applicable IRS form number(s).

[

EXHIBIT E

Include if foreign OP; otherwise, insert "[intentionally omitted]".

- 4. <u>Assumption</u>. Assignee hereby accepts the Transfer set forth in <u>Section 3</u> and assumes and undertakes all of the duties and obligations of Assignor whenever accrued (other than duties and obligations of Assignor required to be performed by Assignor prior to the Effective Time under the Operative Documents) pursuant to the Operative Documents, including without limitation, any obligations it may have under any Operative Document with regard to Lessee or Owner Trustee, in each case subject to Lessee's rights thereunder. Assignee hereby confirms that from and after the Effective Time it (a) shall be deemed a party to the Participation Agreement and the Trust Agreement, (b) shall be deemed the party named as the "Owner Participant" for all purposes of the Operative Documents and (c) shall be bound by, and shall perform and observe, all of the terms of each Operative Document (including the agreements and obligations of Assignor set forth therein) as if therein named the Owner Participant. Assignor hereby assumes the risk of any adverse tax or other adverse consequences of the Assignment and Assumption to any party to, or any Indemnified Person or Tax Indemnitee under, any of the Operative Documents (other than Assignee). Based on the terms and conditions of this Agreement and the representations, warranties and covenants of Assignor and Assignee contained herein, Lessee agrees that from and after the Effective Time Assignee shall be deemed the party named as the "Owner Participant" for all purposes of the Operative Documents.
- 5. <u>Release of Assignor</u>. Assignor will remain liable for the duties, obligations and liabilities of the "Owner Participant" under the Operative Documents except for the duties, obligations and liabilities expressly assumed by Assignee under <u>Section 4</u>. Except as provided in the preceding sentence, Assignor shall be relieved of all of its duties, obligations and liabilities under the Operative Documents; <u>provided</u> that Assignor shall in no event be released from any such duty, obligation or liability (i) arising or relating to any event occurring prior to the Effective Time, (ii) on account of any breach by Assignor of any of its representations, warranties, covenants or obligations contained herein or in any Operative Document or any other Assumption Agreement, or for any fraudulent or willful misconduct engaged in by Assignor, (iii) that relates to any indemnity claimed by Assignor or (iv) relating to or arising out of any Lessor's Lien attributable to Assignor.
- 6. <u>Appointment as Attorney-in-Fact</u>. In furtherance of the assignment set forth in <u>Section 3</u>, Assignor hereby constitutes and appoints Assignee the true and lawful attorney of Assignor, with full power of substitution, in the name of Assignee or in the name of Assignor but on behalf of and for the benefit of and at the expense of Assignee, to collect for the account of Assignee all items sold, transferred or assigned to Assignee pursuant hereto; to institute and prosecute, in the name of Assignor or otherwise, but at the expense of Assignee, all proceedings that Assignee may deem proper in order to collect, assert or enforce any claim, right or title of any kind in or to the items sold, transferred or assigned; to defend and compromise at the expense of Assignee any and all actions, suits or proceedings as to title to or interest in any of the property acquired by Assignee; and to do all such acts and things in relation thereto at the expense of Assignee as Assignee shall reasonably deem advisable. Assignor hereby acknowledges that this appointment is coupled with an interest and is irrevocable by Assignor in any manner or for any reason.
- 7. <u>Payments</u>. Assignor hereby covenants and agrees to pay over to Assignee, if and when received following the Effective Time, any amounts (including any sums payable as interest in respect thereof) paid to or for the benefit of Assignor that, under <u>Section 3</u>, belong to

Assignee, and Assignee hereby covenants and agrees to pay over to Assignor, if and when received following the Effective Time, any amounts (including any sums payable as interest in respect thereof) paid to or for the benefit of Assignee that, under Section 3, belong to Assignor.

- 8. Representations and Warranties of Assignor. Assignor represents and warrants that:
- (a) Assignor is [type of entity] duly organized, validly existing and in good standing under the laws of [jurisdiction of organization] and has the [corporate] power and authority to own or hold under lease its properties, to carry on its business and operations and to enter into and perform its obligations under this Agreement.
- (b) The execution, delivery and performance by Assignor of this Agreement have been duly authorized by all necessary [corporate] action on the part of Assignor, do not require any [stockholder] approval or approval or consent of any trustee or holder of indebtedness or obligations of Assignor, except such as have been duly obtained, or violate or result in a breach of, or constitute a default under, or result in the creation of any Lien (other than as permitted under the Operative Documents) upon the property of Assignor under, any indenture, mortgage, contract or other agreement to which Assignor is a party or by which Assignor or its properties is or are bound or affected. The execution, delivery and performance by Assignor of this Agreement do not and will not violate the [organizational documents] of Assignor or any current law, governmental rule, regulation, judgment or order binding on Assignor (including, without limitation, any such law, rule, regulation, judgment or order relating to money-laundering, anti-corruption or export control or imposing economic sanctions).
- (c) Neither the execution and delivery by Assignor of, nor the performance by Assignor of its obligations under, nor the consummation by Assignor of the transactions contemplated in, this Agreement requires the consent or approval of, or the giving of notice to, or the registration with, or the taking of any other action in respect of any [jurisdiction of organization] governmental authority having jurisdiction.
- (d) This Agreement has been duly executed and delivered by Assignor and constitutes the legal, valid and binding obligation of Assignor enforceable against Assignor in accordance with its terms, except as may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the rights of creditors generally and general principles of equity.
- (e) There are no pending or, to Assignor's knowledge, threatened actions or proceedings before any court, arbitrator or administrative agency which would materially adversely affect the ability of Assignor to perform its obligations under this Agreement or the Operative Documents.
- (f) Neither Assignor nor any Person authorized to act on its behalf has directly or indirectly offered any interest in the Trust Estate or the Trust Agreement or any similar security to, or solicited any offer to acquire any of the same from, any Person in violation of the registration requirements of the Securities Act or any other applicable securities law.

- (g) At the Effective Time, there are no Lessor's Liens attributable to Assignor.
- (h) No Person acting on behalf of Assignor is or will be entitled to any broker's fee, commission or finder's fee in connection with any transaction contemplated by this Agreement or the Operative Documents.
- (i) The Assignment and Assumption (1) does not violate the Transportation Code, the Securities Act or any other Law (including, without limitation, ERISA, any laws or regulations imposing U.S. economic sanctions measures or any orders or licenses issued thereunder), or create a relationship that would be in violation thereof, (2) does not result in a "prohibited transaction" under Section 4975 of the Code, (3) does not adversely affect the registration of the Aircraft in the name of Owner Trustee with the FAA (or the aeronautical authority of the country of registry of the Aircraft if the Aircraft is not registered under the laws of the United States), (4) will not subject Lessee to any additional regulation under, or require Lessee to give any notice to, register with, make any filings with or take any other action in respect of, any governmental authority or agency of any jurisdiction, (5) does not require registration under the Securities Act or any foreign securities laws, require qualification of an indenture under the Trust Indenture Act, or require Lessee to sign any registration statement, (6) unless Lessee consents, the Transfer contemplated hereby does not involved a Rule 144A, Regulation S or other capital markets or equity syndication transaction not described in the immediately preceding clause (5), and (7) does not result in, or involve, incurrence by Lessee of any indebtedness for accounting purposes (it being understood that, if any change in the lease accounting standards applicable to Lessee requires that Lessee, independently of the Transfer contemplated hereby, capitalize its leases, including the Lease, in Lessee's books, such capitalization of the Lease is not intended to constitute, and shall not be construed as, incurrence by Lessee of any indebtedness for accounting purposes within the meaning of this clause (7)).
- (j) Assignor has fully performed all of its obligations under the Operative Documents, which obligations by their terms are required to be satisfied or performed prior to the Effective Time or prior to the consummation of the transactions contemplated hereby.
  - 9. Representations and Warranties of Assignee. Assignee represents and warrants that:
- (a) Assignee is [type of entity] duly organized, validly existing and in good standing under the laws of [jurisdiction of organization] and has the [corporate] power and authority to own or hold under lease its properties, to carry on its business and operations, to enter into and perform its obligations under this Agreement and to perform its obligations under the Operative Documents to which it is or will be a party.
- (b) The execution, delivery and performance by Assignee of this Agreement, and the performance by Assignee of the Operative Documents to which it is or will be party, have been duly authorized by all necessary [corporate] action on the part of Assignee, do not require any [stockholder] approval or approval or consent of any trustee or holder of indebtedness or obligations of Assignee, except such as have been duly obtained, or violate or result in a breach of, or constitute a default under, or result in the creation of any Lien (other than

as permitted under the Operative Documents) upon the property of Assignee under, any indenture, mortgage, contract or other agreement to which Assignee is a party or by which Assignee or its properties is or are bound or affected. The execution, delivery and performance by Assignee of this Agreement and the performance by Assignee of the Operative Documents to which it is or will be party and the acquisition by Assignee of its interest in the Trust Estate (and the rights related thereto) do not and will not violate the [organizational documents] of Assignee or any current law, governmental rule, regulation, judgment or order binding on Assignee (including, without limitation, any such law, rule, regulation, judgment or order relating to money-laundering, anti-corruption or export control or imposing economic sanctions).

- (c) Neither the execution and delivery by Assignee of this Agreement, nor the performance by Assignee of its obligations under, nor the consummation by Assignee of the transactions contemplated in, this Agreement and the Operative Documents to which Assignee is or will be a party, requires the consent or approval of, or the giving of notice to, or the registration with, or the taking of any other action in respect of any [jurisdiction of organization] governmental authority having jurisdiction.
- (d) This Agreement has been duly executed and delivered by Assignee and constitutes, and each Operative Document to which Assignee will be a party will constitute, the legal, valid and binding obligation of Assignee enforceable against Assignee in accordance with its terms, except as may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the rights of creditors generally and general principles of equity.
- (e) There are no pending or, to Assignee's knowledge, threatened actions or proceedings before any court, arbitrator or administrative agency which would materially adversely affect the ability of Assignee to perform its obligations under this Agreement or any Operative Document to which it is or will be a party.
- (f) Neither Assignee nor any Person authorized to act on its behalf has directly or indirectly offered any interest in the Trust Estate or the Trust Agreement or any similar security to, or solicited any offer to acquire any of the same from, any Person in violation of the registration requirements of the Securities Act or any other applicable securities law.
- (g) At the Effective Time, there are no Lessor's Liens attributable to Assignee, and the execution, delivery and performance of this Agreement will not result in any Lessor's Lien attributable to Assignee.
- (h) Either (i) Assignee is a Citizen of the United States or (ii) the Trust Agreement is in a form that permits the Aircraft to be registered with the FAA in the name of Owner Trustee (without regard to any provision of applicable law that permits FAA registration of an aircraft by limiting its location and usage but with regard to voting trust provisions and provisions delegating certain control rights to the Owner Trustee), notwithstanding the failure of Assignee to be a Citizen of the United States.

- (i) Assignee is not an airline or other commercial operator of aircraft, freight forwarder, or any other company directly or indirectly engaged in the business of passenger, cargo, freight or parcel transportation, or any Affiliate thereof.
- (j) Either (a) no part of the funds to be used by Assignee to make and hold its investment pursuant to this Agreement directly or indirectly constitutes assets of any "employee benefit plan" (as defined in Section 3(3) of ERISA) or of any "plan" (as defined in Section 4975(e) of the Code) or (b) its purchase and holding of its interest in the Trust Estate and its investment pursuant to this Agreement are exempt from the prohibited transaction restrictions of ERISA and the Code pursuant to one or more prohibited transaction statutory or administrative exemptions.
- (k) Assignee is a Qualifying Institution (as such term is defined in <u>Section 8.2(a)(ii)</u> of the Lease) (or a parent corporation of the Assignee which qualifies as a Qualifying Institution shall have executed and delivered to Lessee a guaranty substantially in the form of <u>Exhibit G</u> to the Participation Agreement or otherwise in form and substance reasonably satisfactory to Lessee).
- (l) The Assignment and Assumption (1) does not violate the Transportation Code, the Securities Act or any other Law (including, without limitation, ERISA, any laws or regulations imposing U.S. economic sanctions measures or any orders or licenses issued thereunder), or create a relationship that would be in violation thereof, (2) does not result in a "prohibited transaction" under Section 4975 of the Code, (3) does not adversely affect the registration of the Aircraft in the name of Owner Trustee with the FAA (or the aeronautical authority of the country of registry of the Aircraft if the Aircraft is not registered under the laws of the United States), (4) will not subject Lessee to any additional regulation under, or require Lessee to give any notice to, register with, make any filings with or take any other action in respect of, any governmental authority or agency of any jurisdiction, (5) does not require registration under the Securities Act or any foreign securities laws, require qualification of an indenture under the Trust Indenture Act, or require Lessee to sign any registration statement, (6) unless Lessee consents, the Transfer contemplated hereby does not involved a Rule 144A, Regulation S or other capital markets or equity syndication transaction not described in the immediately preceding clause (5), and (7) does not result in, or involve, incurrence by Lessee of any indebtedness for accounting purposes (it being understood that, if any change in the lease accounting standards applicable to Lessee requires that Lessee, independently of the Transfer contemplated hereby, capitalize its leases, including the Lease, in Lessee's books, such capitalization of the Lease is not intended to constitute, and shall not be construed as, incurrence by Lessee of any indebtedness for accounting purposes within the meaning of this clause (7)).
  - (m) [Assignee is a domestic [corporation][partnership] for U.S. federal income tax purposes.]11

#### 10. Certain Agreements.

- (a) Assignee agrees that, except as expressly permitted by Section 15 of the Lease in connection with an Event of Default that shall have occurred and be continuing, notwithstanding anything herein or in any Operative Document to the contrary, neither it nor any Person claiming by, through or under it shall take or cause to be taken any action inconsistent with Lessee's rights under the Lease and Lessee's right to quiet enjoyment of the Aircraft, the Airframe, any Engine or any Part, or otherwise in any way interfere with or interrupt the use, operation and continuing possession of the Aircraft, the Airframe, any Engine or any Part by Lessee or any sublessee, assignee or transferee under any sublease, assignment or transfer then in effect and permitted by the terms of the Lease.
- (b) Notwithstanding anything to the contrary contained herein or in the Operative Documents, each of Assignor and Assignee hereby agrees, for the benefit of Lessee, that as determined at the Effective Time none of Lessee's obligations, responsibilities, liabilities, costs and risks in the use and operation of the Aircraft or under, relating to or in respect of the Operative Documents or otherwise, including, without limitation, under or in respect of any of Lessee's payment or indemnity obligations, shall be increased or altered, and none of Lessee's rights and benefits under any Operative Document shall be diminished, as a result of or in connection with the Assignment and Assumption or any aspect thereof or any other transaction relating thereto (it being acknowledged that an increase in the number of indemnitees shall not, of itself, constitute an increase in Lessee's obligations under the Operative Documents).

### 11. Miscellaneous.

- (a) Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof or affecting the validity or enforceability of such provision in any other jurisdiction.
- (b) This Agreement may be executed in any number of counterparts (and each of the parties hereto shall not be required to execute the same counterpart). Each counterpart of this Agreement, including a signature page executed by each of the parties hereto shall be an original, but all of such counterparts together shall constitute one instrument.

EXHIBIT E Page 8 (c) THIS AGREEMENT SHALL IN ALL RESPECTS BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK, INCLUDING ALL MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE. THIS AGREEMENT HAS BEEN DELIVERED IN THE STATE OF NEW YORK. [Assignee irrevocably designates and appoints [name of process agent] as process agent to receive for it and on its behalf service of process in any proceedings arising hereunder or under any other Operative Document to which it is a party. Nothing in this Agreement shall affect the right to serve process in any other manner permitted by law.]<sup>12</sup>

(d) To the extent that Assignee or any of its properties has or may hereafter acquire any right of immunity, whether characterized as sovereign immunity or otherwise, and whether under the United States Foreign Sovereign Immunities Act of 1976 (or any successor legislation) or otherwise, from any legal proceedings, whether in the United States or elsewhere, to enforce or collect upon any Operative Documents to which it is a party, including, without limitation, immunity from suit or service of process, immunity from jurisdiction or judgment of any court or tribunal or execution of a judgment, or immunity of any of its property from attachment prior to any entry of judgment, or from attachment in aid of execution upon a judgment, Assignee hereby irrevocably and expressly waives any such immunity, and agrees not to assert any such right or claim in any such proceeding, whether in the United States or elsewhere.

[Remainder of Page Intentionally Left Blank; Signature Page Follows]

Include only if Assignee is a foreign entity.

EXHIBIT E Page 9

		[NAME OF ASSIGNOR]	
		Ву:	
		Name: Title:	
		[NAME OF ASSIGNEE]	
		Ву:	
		Name: Title:	
Accepted and Agreed:			
AMERICAN AIRLINES, INC.			
Ву:			
Name: Citle:			
	EXHIBIT E Page 10		

IN WITNESS WHEREOF, the parties hereto have caused this Assignment and Assumption Agreement to be duly executed as of the day and year first

above written.

## **EXHIBIT F**

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

EXHIBIT F Page 1 **EXHIBIT F** 

ANNEX A TO BFE BILL OF SALE

EXHIBIT F Page 2

## **EXHIBIT G**

# FORM OF OWNER PARTICIPANT GUARANTEE

EXHIBIT G Page 1 ANNEX A

# **DEFINITIONS**

ANNEX A Page 1

### ANNEX B

## **PAYMENT INFORMATION**

Payments to Owner	Trustee/Lessor: Payments made to Owne	er Trustee pursuant to the terms of the	Operative Documents shall	be made to the following
account of Lessor:				

[ ]

or to such other account of Lessor in the United States as may be specified in a notice delivered by Owner Trustee to Lessee and Owner Participant in accordance with <u>Section 10.1</u> at least 10 Business Days prior to the due date after which payments are to be made pursuant to such notice.

**Payments to Owner Participant**: Payments made to Owner Participant pursuant to the terms of the Operative Documents shall be made to the following account of Owner Participant:

[

or to such other account of Owner Participant in the United States as may be specified in a notice delivered by Owner Participant to Owner Trustee and Lessee in accordance with Section 10.1 at least 10 Business Days prior to the due date after which payments are to be made pursuant to such notice.

Payments to Lessee: Payments made to Lessee pursuant to the terms of the Operative Documents shall be made to the following account of Lessee:

Į.

or to such other account of Lessee in the United States as may be specified in a notice delivered by Lessee to Owner Trustee and/or Owner Participant in accordance with Section 10.1 at least 10 Business Days prior to the due date after which payments are to be made pursuant to such notice.

ANNEX B

Page 1

### **SCHEDULE A**

## **CERTAIN TERMS**

Insurance Threshold Amount: [CONFIDENTIAL

PORTION OMITTED

AND FILED

SEPARATELY WITH THE

COMMISSION
PURSUANT TO A
REQUEST FOR
CONFIDENTIAL
TREATMENT]

Obsolete Parts cap (for purposes of Section 8(c) of the Lease)

[CONFIDENTIAL PORTION OMITTED

AND FILED

SEPARATELY WITH THE

COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## PERMITTED COUNTRIES

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

SCHEDULE A

Page 1

Insert value for applicable aircraft type.

#### SCHEDULE B

### **RE-REGISTRATION CONDITIONS**

Lessee's right to cause or permit the Aircraft to be re-registered pursuant to <u>Section 7(a)</u> of the Lease is subject to the satisfaction of the conditions below or waiver thereof by Lessor and Owner Participant, as applicable:

- (a) no Event of Default shall have occurred and be continuing at the date of such request or at the effective date of the change in registration; provided that it shall not be necessary to comply with this condition if the change in registration involves the registration of the Aircraft under the laws of the United States;
- (b) each of Lessor and Owner Participant shall have received a legal opinion addressed to it from counsel to Lessee admitted to practice in the jurisdiction of registration (which counsel shall be reasonably satisfactory to Owner Participant) (i) to the effect that (A) after giving effect to such change in registration, all filing, recording or other action necessary to perfect and protect Lessor's rights and interests in and to the Aircraft and the Lease has been accomplished (or if such opinion cannot be given at the time by which Lessor has been requested to consent to a change in registration, (x) the opinion shall detail what filing, recording or other action is necessary and (y) Lessor and Owner Participant shall have received a certificate from Lessee that all possible preparations to accomplish such filing, recording and other action shall have been done, and such filing, recording and other action shall be accomplished and a supplemental opinion to that effect shall be delivered to Lessor and Owner Participant on or prior to the effective date of such change in registration), (B) the terms of the Lease are legal, valid and binding and enforceable against Lessee in such jurisdiction (subject to bankruptcy and equitable remedies exceptions and other customary exceptions), and (C) it is not necessary for Lessor or Owner Participant to qualify to do business in such jurisdiction or otherwise satisfy any other applicable law, rule or regulation existing at the date of such request (or if such opinion cannot be given, the opinion shall detail what other existing law, rule or regulation must be satisfied by Lessor or Owner Participant, as the case may be) solely as a result of the proposed re-registration, and (ii) if such re-registration is in connection with a sublease to a Permitted Sublessee and such country is not, at the time of reregistration, the United States or a Permitted Country, to the effect that there exist no possessory rights in favor of Permitted Sublessee under the laws of such country that would, assuming at such time Permitted Sublessee is not insolvent or bankrupt, prevent the return of the Aircraft in accordance with and when permitted by the terms of <u>Sections 14</u> and <u>15</u> of the Lease upon the exercise of remedies by Lessor of its remedies in accordance with <u>Section 15</u> of the Lease in connection with an Event of Default that shall have occurred and be continuing;
- (c) Lessor and Owner Participant shall have received assurances reasonably satisfactory to Owner Participant to the effect that the insurance provisions of the Lease shall have been complied with after giving effect to such change in registration;

SCHEDULE B Page 1

- (d) such re-registration will not result in the imposition by such country of any Taxes on Lessor or Owner Participant for which Lessee is not required to indemnify Lessor or Owner Participant, as the case may be, unless Lessee agrees to indemnify Lessor or Owner Participant, as the case may be, for any Taxes imposed by such country in connection with or relating to the transactions contemplated by the Lease that would not have been imposed but for such re-registration; provided that it shall not be necessary to comply with the conditions contained in this clause (d) if such change in registration results in the re-registration of the Aircraft under the laws of the United States, except to the extent that the provisions of the tax indemnification provisions relating to Lessor or Owner Participant, as the case may be, were amended in effecting a previous foreign registration;
  - (e) such re-registration will not divest Lessor of title to the Aircraft; and
- (f) Lessee shall have paid or made provision for the payment of all reasonable out-of-pocket expenses (including reasonable attorneys' fees) of Lessor and Owner Participant in connection with such change in registration;

provided, further, that Lessee shall not cause the Aircraft to be registered under the laws of any foreign jurisdiction without the prior written consent of Owner Participant if (1) the civil aviation laws of such foreign jurisdiction impose unusual requirements on lessors of civil aircraft, and (2) Lessor or Owner Participant, as the case may be, would be required to comply with such unusual requirements upon the registration of the Aircraft in such foreign jurisdiction, and compliance therewith by Lessor or Owner Participant, as the case may be, would result in a material burden on the business activities of Lessor or Owner Participant, as the case may be.

SCHEDULE B Page 2

EXHIBIT D

FORM OF DEFINITIONS ANNEX

FORM OF LEASING LETTER

GUARANTEE ([YEAR] MSN [MSN])		
dated as of —, 20—		
by		
[NAME OF OWNER PARTICIPANT GUARANTOR]		

## TABLE OF CONTENTS

Defined Terms	1
Affirmation of Representations and Warranties under OP Documents	1
Guarantee	1
Guarantee of Obligations under OP Documents	1
Unconditional Obligations	2
Guarantor's Obligations Not Affected	3
Waiver	4
Waiver of Rights of Subrogation and Contribution	4
Payments	4
Representations, Warranties and Covenants of Guarantor	5
Representations and Warranties of Guarantor	5
Covenants of Guarantor	5
Costs and Expenses	5
Survival of Representations, Warranties and Agreements	5
Notices, etc	6
Amendments and Waivers	6
Severability of this Guarantee	6
Miscellaneous	6
[Agent for Service of Process	7
Time of the Essence	7
Waiver of Immunity	7
	Affirmation of Representations and Warranties under OP Documents Guarantee Guarantee of Obligations under OP Documents Unconditional Obligations Guarantor's Obligations Not Affected Waiver Waiver of Rights of Subrogation and Contribution Payments Representations, Warranties and Covenants of Guarantor Representations and Warranties of Guarantor Covenants of Guarantor Costs and Expenses Survival of Representations, Warranties and Agreements Notices, etc Amendments and Waivers Severability of this Guarantee Miscellaneous [Agent for Service of Process Time of the Essence

EXHIBIT A - Beneficiaries

### **GUARANTEE**

**GUARANTEE** dated as of [—], 20[—] by **[OWNER PARTICIPANT GUARANTOR]**, a **[FORM OF ENTITY]** organized and existing under the laws of **[JURISDICTION]** (together with its successors and assigns, "**Guarantor**"), for the benefit the parties listed in Exhibit A attached hereto (together with their respective successors and assigns, the "**Beneficiaries**").

**WHEREAS**, [OWNER PARTICIPANT], a [FORM OF ENTITY] organized and existing under the laws of [JURISDICTION] (together with its successors and assigns, the "Owner Participant") is majority-owned subsidiary of Guarantor;

WHEREAS, Guarantor derives substantial benefit from the Owner Participant entering into the transactions contemplated by the OP Documents;

WHEREAS, American Airlines, Inc. ("American"), as lessee, the Owner Participant and Wells Fargo Bank Northwest, National Association, a national banking association, not in its individual capacity except as expressly provided herein, but solely as owner trustee, as lessor ("Lessor"), entered into a Participation Agreement ([YEAR] MSN [MSN]), dated as of the date hereof (as may be amended, supplemented or otherwise modified from time to time in accordance with its terms, the "Participation Agreement"):

**WHEREAS**, American and Lessor entered into a Lease Agreement ([YEAR] MSN [MSN]) dated as of the date hereof (as may be amended, supplemented or otherwise modified from time to time in accordance with its terms, the "Lease");

WHEREAS, it is a condition to the Lessee's obligation to enter into the transactions contemplated by the Participation Agreement that Guarantor agrees to guarantee the obligations of the Owner Participant in each of the Operative Documents executed or to be executed by the Owner Participant or by which the Owner Participant is bound (such Operative Documents, in each case, as amended, modified or supplemented from time to time, being referred to herein as the "OP Documents");

**NOW, THEREFORE**, in consideration of the mutual agreements herein contained, the parties hereto agree as follows:

Section 1. Defined Terms. Capitalized terms used herein and not otherwise defined herein shall have the meanings given them in Annex A to the Lease.

**Section 2.** <u>Affirmation of Representations and Warranties under OP Documents</u>. Guarantor hereby represents and warrants to the Beneficiaries that all of the representations and warranties of the Owner Participant contained in the OP Documents are true and correct as of the date hereof.

### Section 3. Guarantee.

3.1. <u>Guarantee of Obligations under OP Documents</u>. Guarantor irrevocably and unconditionally guarantees to the Beneficiaries the due and punctual performance of and compliance with all covenants, agreements, terms and conditions of each of the OP Documents

required to be performed or complied with by the Owner Participant (including, without limitation, in the case of a Back-Leveraging Transaction, the Owner Participant's representations, warranties and covenants described in Section 8.3.1(b) of the Participation Agreement and compliance of the applicable Back-Leveraging Transaction with the terms of Section 8.3 of the Participation Agreement) (all such payment obligations and other covenants, agreements, terms and conditions, being referred to herein as the "Obligations"). In case the Owner Participant shall fail to perform or comply with any Obligation, Guarantor will forthwith perform and comply with such Obligation or cause the same forthwith to be performed or complied with, and, in case the Owner Participant shall fail to pay or perform duly and punctually any Obligation required to be made or performed by the Owner Participant under any OP Document when and as the same shall be due and payable, or required to be performed, as the case may be, in accordance with the terms of such OP Document, Guarantor will immediately pay or perform, as the case may be, the same to the Person entitled thereto and in addition, pay such further amount, if any, as shall be sufficient to cover all reasonable costs and expenses (including, without limitation, all reasonable fees and disbursements of counsel) that may be paid or incurred by the Beneficiaries in enforcing, or obtaining advice of counsel in respect of, any rights with respect to, or collecting any or all of the Obligations.

3.2. <u>Unconditional Obligations</u>. The guarantee by Guarantor contained in <u>Section 3.1</u> hereof is a primary obligation of Guarantor and is an unconditional, absolute, present and continuing obligation and is not conditioned in any way upon the institution of suit or the taking of any other action with respect to the representations and warranties of the Owner Participant contained in any OP Document or any attempt to enforce performance of or compliance with the Obligations (including, without limitation, any payment obligations). To the extent that performance or compliance with the guarantee by Guarantor contained in <u>Section 3.1</u> hereof requires the payment of money, such guarantee is an absolute, unconditional, present and continuing guarantee of payment and not of collectability and is in no way conditioned or contingent upon the validity, or enforceability of any OP Document or any of the Obligations or any collateral security, other guarantee, if any, or credit support therefor or any attempt to collect from the Owner Participant or any other entity or to perfect or enforce any security or upon any other condition or contingency or upon any other action, occurrence or circumstance whatsoever. Such guarantee shall continue to be effective, or be reinstated, as the case may be, if at any time payment, in whole or in part, of any of the sums due to any of the Beneficiaries pursuant to the terms of any OP Document is rescinded or must otherwise be restored or returned upon the bankruptcy, insolvency, reorganization, arrangement, adjustment, composition, dissolution, liquidation, or the like, of the Owner Participant or Guarantor, or upon or as a result of, the appointment of a custodian, receiver, trustee or other officer with similar powers with respect to the Owner Participant or Guarantor or any substantial part of their respective property, or otherwise, all as though such payment had not been made notwithstanding any termination of this Guarantee or any OP Document. Guarantor shall not commence against

- 3.3. <u>Guarantor's Obligations Not Affected</u>. The obligations of Guarantor under this Guarantee shall remain in full force and effect without regard to, and shall not be impaired or affected by:
  - (a) any extension, indulgence or renewal in respect of the payment of any amount payable, or the performance of any Obligation; or
- (b) any amendment or modification of or addition or supplement to or deletion from any of the terms of any OP Document, or any other agreement (including, without limitation, any collateral security, other guarantee, if any, or other credit support or right of offset with respect thereto) which may be made relating to any OP Document or any Obligation; or
- (c) any compromise, waiver, release or consent or other action or inaction in respect of any of the terms of any OP Document, or any other agreement (including, without limitation, any collateral security, other guarantee, if any, or other credit support or right of offset with respect thereto) which may be made relating to any OP Document or any Obligation; or
- (d) any exercise or non-exercise by any of the Beneficiaries of any right, power, privilege or remedy under or in respect of this Guarantee or any OP Document, or any waiver of any such right, power, privilege or remedy or of any default in respect of this Guarantee or any OP Document or any guarantee or other agreement executed pursuant hereto, or any receipt of any security or any release of any security; or
- (e) any bankruptcy, insolvency, reorganization, arrangement, adjustment, composition, dissolution, liquidation, or the like, of the Owner Participant, Guarantor or any other Person; or
- (f) any limitation of the liability of the Owner Participant under the terms of any OP Document which may now or hereafter be imposed by any statute, regulation or rule of law; or
- (g) any merger or consolidation of the Owner Participant or Guarantor into or with any other person or entity, or any sale, lease or transfer of any or all of the assets of the Owner Participant or Guarantor to any other person or entity; or
  - (h) any indebtedness of the Owner Participant to any person or entity, including Guarantor; or
- (i) any claim, set-off, deduction or defense Guarantor or the Owner Participant may have against any of the Beneficiaries, whether hereunder or under any OP Document or independent of or unrelated to the transactions contemplated by the OP Documents; or
  - (i) any change in law; or
- (k) absence of any notice to, or knowledge by, Guarantor of the existence or occurrence of any of the matters or events set forth in the foregoing subdivisions (a) through (j); or

- (l) any sale, transfer or other disposition by the Owner Participant of any right, title or interest in and to any OP Document or the Aircraft; or
- (m) any other circumstance whatsoever, whether similar or dissimilar to the foregoing, whether foreseen or unforeseen, that might otherwise constitute a legal or equitable defense or discharge of the liabilities of a guarantor or surety or that might otherwise limit recourse against Guarantor. No obligations of the Owner Participant are affected hereby.
- 3.4. Waiver. Guarantor unconditionally waives, to the fullest extent permitted by Law, (a) notices of the creation of any Obligation under the OP Documents or any of the matters referred to in Section 3.3 hereof or any notice of or proof of reliance by any of the Beneficiaries upon this Guarantee or acceptance of this Guarantee (the Obligations shall conclusively be deemed to have been created, contracted, incurred or renewed, extended, amended or waived in reliance upon this Guarantee and all dealings between the Owner Participant or Guarantor and any Beneficiary shall be conclusively presumed to have been had or consummated in reliance upon this Guarantee), (b) all notices which may be required by statute, rule of law or otherwise, now or hereafter in effect, to preserve intact any rights of any of the Beneficiaries against Guarantor, including, without limitation, any demand, presentment and protest, proof of notice of non-payment under any OP Document, and notice of default or any failure on the part of the Owner Participant to perform and comply with any Obligation, (c) any right to the enforcement, assertion or exercise by any of the Beneficiaries of any right, power, privilege or remedy conferred herein or in any OP Document or otherwise, (d) any requirement of promptness or diligence on the part of any of the Beneficiaries, or (e) any other circumstance whatsoever which might otherwise constitute a legal or equitable discharge, release or defense of a guarantor or surety, or which might otherwise limit recourse against Guarantor.
- 3.5. <u>Waiver of Rights of Subrogation and Contribution</u>. Guarantor will not assert any right to which it may become entitled, whether by subrogation, contribution or otherwise, against the Owner Participant or any of its properties, by reason of the performance by Guarantor of its obligations under this Agreement, nor shall Guarantor seek or be entitled to seek any reimbursement from the Owner Participant in respect of payments made by Guarantor until such time as all of the Obligations of the Owner Participant under the OP Documents shall be duly and fully performed.
- 3.6. <u>Payments</u>. Guarantor hereby guarantees that all payments hereunder shall be paid without set-off, counterclaim, deduction or withholding (or, if there is any such deduction or withholding for Taxes, Guarantor hereby agrees to pay additional amounts such that Guarantor bears such Taxes), and shall be made in U.S. Dollars; <u>provided</u>, that such Beneficiary shall have provided Guarantor with any withholding form, certificates or documents that such Beneficiary is legally entitled to provide if necessary or advisable to reduce or eliminate such withholding taxes, provided, however, that no Beneficiary shall be required to deliver such form, certificates or documents to reduce or eliminate any withholding taxes imposed by any non U.S. jurisdiction as a result of payments being made from, or Guarantor's or Owner Participant's connection with, such jurisdiction unless (x) Guarantor shall have provided to such Beneficiary timely notice of the requirement for such documentation, (y) such Beneficiary determines in good faith that it would suffer no risk of adverse consequences by providing the applicable form, and (z) Guarantor has agreed to pay, and does pay after demand therefor, on an After-Tax Basis, all

costs and expenses incurred by such Beneficiary in providing the applicable form. If any payment of Guarantor hereunder is converted into a claim, proof, judgment or order in a currency other than Dollars, Guarantor will indemnify the Beneficiaries as an independent obligation against any loss arising out of or as a result of such receipt or conversion.

# Section 4. Representations, Warranties and Covenants of Guarantor.

- 4.1. Representations and Warranties of Guarantor. As of the date hereof, Guarantor hereby represents and warrants that: (a) it is a [FORM OF ENTITY] duly organized and validly existing under the laws of [JURISDICTION] and has the corporate power and authority to carry on its present business and operations, to own or hold under lease its properties and to enter into and perform its obligations under this Guarantee, and this Guarantee has been duly authorized, executed and delivered by it and is legal, valid and binding on it and is enforceable against it in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the rights of creditors generally and by general principles of equity; (b) the execution and delivery by Guarantor of this Guarantee and compliance by it with all of the provisions hereof do not and will not contravene any Law or any order of any court or governmental authority or agency applicable to or binding on it or contravene the provisions of, or constitute a default under, [its certificate of incorporation or by-laws] or any indenture, mortgage, contract or any agreement or instrument to which it is a party or by which it or any of its property may be bound or affected; (c) no authorization or approval or other action by, and no notice to or filing with, any [jurisdiction of organization] governmental authority having authority over Guarantor or its assets is required for the due execution, delivery or performance by it of this Guarantee; (d) there are no pending or threatened actions or proceedings before any court or administrative agency which would materially adversely affect its ability to perform its obligations under this Guarantee; (e) the Owner Participant is a majority-owned subsidiary of Guarantor; and (f) Guarantor is a Qualifying Institution.
  - 4.2. Covenants of Guarantor. For so long as the Owner Participant is a party to the OP Documents:
    - (a) Guarantor agrees that it will not impair Owner Participant's ability to perform its obligations under the OP Documents; and
- (b) Guarantor agrees to comply with the terms and conditions of Section 10.4 of the Participation Agreement with respect to any Confidential Information.
- **Section 5.** <u>Costs and Expenses</u>. Guarantor will pay all reasonable costs and expenses (including, without limitation, reasonable legal fees and expenses) incurred by or on behalf of any of the Beneficiaries in connection with the enforcement of Guarantor's obligations under this Guarantee.
- **Section 6.** <u>Survival of Representations, Warranties and Agreements</u>. The representations, warranties and agreements of Guarantor contained herein shall survive the execution and delivery of this Guarantee and the consummation of the transactions contemplated hereby and by the Operative Documents.

Section 7. Notices, etc. Unless otherwise expressly specified or permitted by the terms hereof, all notices, requests, demands, authorizations, directions, consents or waivers required or permitted under the terms and provisions of this Guarantee shall be in English and in writing, and given by United States registered or certified mail, return receipt requested, postage prepaid, overnight courier service or facsimile, and any such notice shall be effective when received (or, if delivered by facsimile, upon completion of transmission and confirmation by the sender (by a telephone call to a representative of the recipient or by machine confirmation) that such transmission was received) and addressed, (a) if to Guarantor, at — or at such other address as Guarantor shall from time to time designate in writing to American, (b) if to American, at 4333 Amon Carter Boulevard, Mail Drop 5662, Ft. Worth, Texas 76155, Attention: Treasurer, Fax: (817) 967-4318, Tel: (817) 963-1234 or at such other address as American shall from time to time designate in writing to Guarantor, or (c) if to the Lessor, at — or at such other address as the Lessor shall from time to time designate in writing to Guarantor.

**Section 8.** <u>Amendments and Waivers</u>. Neither this Guarantee nor any of the terms hereof may be terminated, amended, supplemented, waived or modified orally, but only by an instrument in writing signed by Guarantor and the Beneficiaries.

**Section 9.** <u>Severability of this Guarantee</u>. Any provision of this Guarantee which is prohibited and unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

Section 10. Miscellaneous. This Guarantee shall remain in full force and effect until payment in full of all sums payable hereunder under the OP Documents, and performance in full of all obligations of Guarantor hereunder, it being understood that upon the transfer of the Owner Participant's interest pursuant to the terms of the OP Documents, subject to the payment in full of all sums due and payable hereunder and performance in full of all obligations of Guarantor hereunder and subject to such transfer complying with Section 8.2 of the Participation Agreement, this Guarantee shall terminate. This Guarantee constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof. The index preceding this Guarantee and the headings of the various Sections of this Guarantee are for convenience of reference only and shall not modify, define, expand or limit any of the terms or provisions hereof. The terms of this Guarantee shall be binding upon the successors of Guarantor, and shall inure to the benefit of the Beneficiaries and their successors and permitted assigns. Guarantor hereby irrevocably submits to the nonexclusive jurisdiction of each of the Supreme Court of the State of New York, New York County and the United States District Court for the Southern District of New York, and other courts with jurisdiction to hear appeals from such court for the purposes of any suit, action or other proceeding arising out of this Guarantee, the subject matter hereof or any of the transactions contemplated hereby. THIS GUARANTEE SHALL IN ALL RESPECTS BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK, INCLUDING ALL MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE. THIS GUARANTEE IS BEING DELIVERED IN THE STATE OF NEW YORK.

**Section 11.** [Agent for Service of Process. Guarantor agrees that its designated agent for service of process relating to any proceedings arising out of or connected with this Guarantee is —. Guarantor agrees that service of process in any action or proceeding described in Section 10 may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to such agent for service of process at its address referred to in the first sentence of this Section 11. Guarantor, by notice to the Beneficiaries, may designate a different agent and address for subsequent service of process; provided that Guarantor will take all action, including the filing of any and all documents and instruments, as may be necessary so that it shall at all times have an agent for service of process for the above purposes in the County of New York, State of New York. Nothing in this Agreement shall affect the right to serve process in any other manner permitted by law.]<sup>1</sup>

**Section 12.** <u>Time of the Essence</u>. The time stipulated in this Guarantee for all payments by Guarantor to any of the Beneficiaries and for prompt, punctual performance of Guarantor's obligations under this Guarantee shall be of the essence for this Guarantee.

Section 13. Waiver of Immunity. To the extent that Owner Participant or any of its properties has or may hereafter acquire any right of immunity, whether characterized as sovereign immunity or otherwise, and whether under the United States Foreign Sovereign Immunities Act of 1976 (or any successor legislation) or otherwise, from any legal proceedings, whether in the United States or elsewhere, to enforce or collect upon any Operative Documents to which it is a party, including, without limitation, immunity from suit or service of process, immunity from jurisdiction or judgment of any court or tribunal or execution of a judgment, or immunity of any of its property from attachment prior to any entry of judgment, or from attachment in aid of execution upon a judgment, Owner Participant hereby irrevocably and expressly waives any such immunity, and agrees not to assert any such right or claim in any such proceeding, whether in the United States or elsewhere.

Include if Guarantor is foreign.

IN WITNESS WHEREOF, the undersigned has caused this Guarantee to be duly executed by its duly authorized officer as of the day and year first a	above
written.	

[GUARANTOR]		
P.v.		

Title:

# EXHIBIT A

# BENEFICIARIES

American Airlines, Inc., as Lessee (and its successors and permitted assigns)

Wells Fargo Bank Northwest, National Association, Owner Trustee and Lessor (and its successors and permitted assigns)

#### ANNEX A

### **DEFINITIONS**

#### General Provisions

- (a) In each Operative Document (as defined below), unless otherwise expressly provided, a reference to:
- (i) each of "Lessee," "Lessor," "Owner Trustee," "Owner Participant" or any other Person includes, without prejudice to the provisions of any Operative Document, any successor in interest to it and any permitted transferee, permitted purchaser or permitted assignee of it;
  - (ii) words importing the plural include the singular and words importing the singular include the plural;
- (iii) any agreement, instrument or document, or any annex, schedule or exhibit thereto, or any other part thereof, includes, without prejudice to the provisions of any Operative Document, that agreement, instrument or document, or annex, schedule or exhibit, or part, respectively, as amended, modified or supplemented from time to time in accordance with its terms, and any agreement, instrument or document entered into in substitution or replacement therefor;
- (iv) any provision of any law includes any such provision as amended, modified, supplemented, substituted, reissued or reenacted prior to the Delivery Date (as defined below in this <u>Annex A</u>), and thereafter from time to time;
  - (v) the word "government" includes any instrumentality or agency thereof;
- (vi) the words "Agreement," "this Agreement," "hereby," "herein," "hereto," "hereof" and "hereunder" and words of similar import when used in any Operative Document refer to such Operative Document as a whole and not to any particular provision of such Operative Document;
- (vii) the words "including," "including, without limitation," "including, but not limited to," and terms or phrases of similar import when used in any Operative Document, with respect to any matter or thing, mean including, without limitation, such matter or thing; and
- (viii) a "Section," a "subsection," an "Exhibit," an "Annex" or a "Schedule" in any Operative Document, or in any annex thereto, is a reference to a section or a subsection of, or an exhibit, an annex or a schedule to, such Operative Document or such annex, respectively.

.

- (b) Each attachment, appendix, exhibit, annex, supplement and schedule to each Operative Document is incorporated in, and shall be deemed to be a part of, such Operative Document.
- (c) Headings and tables of contents used in any Operative Document are for convenience only and shall not in any way affect the construction of, or be taken into consideration in interpreting, such Operative Document.

### **Defined Terms**

- "AD" has the meaning set forth in Section 7(a) of the Lease.
- "Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under common control with such Person. For the purposes of this definition, "control" (including "controlled by" and "under common control with") shall mean the power, directly or indirectly, to direct or cause the direction of the management and policies of such Person whether through the ownership of voting securities or by contract or otherwise. In no event shall Trust Company be deemed to be an Affiliate of any of Owner Trustee, Lessor or Owner Participant or vice versa.
- "After-Tax Basis", in the context of determining the amount of a payment to be made on such basis, means the payment of an amount which, after subtraction of the net increase, if any, in U.S. federal, state and local income tax liability incurred by the Indemnified Person or Tax Indemnitee to whom the payment is made as a result of the receipt or accrual of such payment (taking into account any current Tax benefits realized by such Indemnified Person or Tax Indemnitee as a result of the event or circumstances giving rise to such payment), shall equal the amount that would have been payable if no net increase in such tax liability had been incurred.
- "Aircraft" means the Airframe together with the two Engines described in Lease Supplement No. 1 (or any Replacement Engine substituted for any of such Engines under, and pursuant to the terms of, the Lease) (except in each case for any Excluded Equipment), whether or not any of such initial or substituted Engines may from time to time be installed on such Airframe or may be installed on any other airframe or on any other aircraft.
- "Aircraft Protocol" means the official English language text of the Protocol to the Convention on International Interests in Mobile Equipment on Matters Specific to Aircraft Equipment, adopted on November 16, 2001, at a diplomatic conference in Cape Town, South Africa, and all amendments, supplements, and revisions thereto, as in effect in the United States.
- "Airframe" means (i) the aircraft described in Lease Supplement No. 1 (except (x) Engines or engines from time to time installed thereon and any and all Parts related to such Engines or engines and (y) any Excluded Equipment) to be leased under the Lease by Lessor to Lessee and (ii) any and all Parts so long as the same shall be incorporated or installed in or attached to such aircraft, or so long as title thereto shall remain vested in Lessor in accordance with the terms of Section 8 of the Lease after removal from such aircraft.

- "American/Airbus Purchase Agreement" means the A320 Family Aircraft Purchase Agreement, dated July 19, 2011, between Lessee and the Manufacturer, as amended, supplemented or otherwise modified from time to time.
- "Application for Aircraft Registration" means the application for registration on Federal Aviation Administration AC Form 8050-1 with respect to the Aircraft in the name of Owner Trustee.
- "Approved Program" means a maintenance program for aircraft of the same make and model as the Aircraft which shall be (i) the Maintenance Program, (ii) the MPD, or (iii) such other maintenance program approved by Lessor (such approval not to be unreasonably withheld); provided that for purposes of this clause (iii), such approval right shall, in connection with the re-registration or subleasing of the Aircraft, Airframe or any Engine, only be required at the time of the initial re-registration or at the commencement of such sublease, as applicable. For purposes of the foregoing sentence, with respect to the flight/hours/cycles/calendar time limitations of Parts and inspections, references to the MPD mean the most restrictive applicable limitation set forth therein.
  - "Assumption Agreement" has the meaning set forth in  $\underline{Section 8.2(a)(i)(\underline{D})}$  of the Participation Agreement.
- "<u>Aviation Counsel</u>" means Daugherty, Fowler, Peregrin, Haught & Jenson, a Professional Corporation, or such other nationally recognized special aviation counsel located in Oklahoma City, Oklahoma as is designated by Lessee.
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - "Bankruptcy Code" means the United States Bankruptcy Code, 11 United States Code §§ 101 et seq.

- "Basic Rent" means the basic rent payable to Lessor for the Aircraft pursuant to Section 3(b) of the Lease in the amounts and payable at the times as provided therein.
- "Basic Term" means the term for which the Aircraft is leased under the Lease pursuant to Section 3(a) thereof commencing on the Delivery Date and ending on the Lease Expiry Date, or such earlier date on which the Lease is terminated in accordance with the provisions thereof.
  - "Bills of Sale" means the FAA Bill of Sale and the Warranty Bill of Sale collectively.
- "<u>Business Day</u>" means any day other than a Saturday, Sunday or a day on which commercial banks are authorized or required by law, regulation or executive order to be closed in Fort Worth, Texas or New York, New York or the city and state in which the principal corporate trust office of Owner Trustee is located.
- "Cape Town Convention" means the official English language text of the Convention on International Interests in Mobile Equipment, adopted on November 16, 2001, at a diplomatic conference in Cape Town, South Africa, and all amendments, supplements, and revisions thereto, as in effect in the United States.
- "Cape Town Treaty" means, collectively, the official English language text of (i) the Cape Town Convention, (ii) the Aircraft Protocol, (iii) all rules and regulations adopted pursuant thereto and as in effect in the United States and (iv) with respect to each of the foregoing described in clauses (i) through (iii), all amendments, supplements and revisions thereto as in effect in the United States.
- "<u>Certificated Air Carrier</u>" means an air carrier holding an air carrier operating certificate issued by the Secretary of Transportation pursuant to Chapter 447 of Title 49, United States Code, for aircraft capable of carrying ten or more individuals or 6,000 pounds or more of cargo or that otherwise is certified or registered to the extent required to fall within the purview of Section 1110.
- "Citizen of the United States" has the meaning specified for such term in Section 40102(a)(15) of Title 49 of the United States Code or any similar legislation of the U.S. enacted in substitution or replacement therefor.
- "Claims" or "Claims" means any and all liabilities, obligations, losses, damages, penalties, claims, costs, actions or suits of whatsoever kind and nature (whether or not on the basis of negligence, strict or absolute liability or liability in tort) and, except as otherwise expressly provided, shall include all reasonable costs, disbursements and expenses (including reasonable legal fees and expenses) in connection therewith or related thereto.
  - "Closing" has the meaning set forth in Section 3 of the Participation Agreement.
  - "Code" means the U.S. Internal Revenue Code of 1986, as currently in effect or hereafter amended.

"Confidential Information" means the provisions of, and all matters relating to, the Lease (other than any portions of the Lease recorded with the FAA and available for public inspection), the Participation Agreement and the other Operative Documents including, without limitation, (i) the existence and terms of any sublease of the Airframe or Engines pursuant to Section 7(b) of the Lease and the identity of the Permitted Sublessee thereunder (other than any portions of any sublease recorded with the FAA and available for public inspection); (ii) all information obtained in connection with any inspection conducted pursuant to Section 12(a) or 12(b) of the Lease or obtained from Lessee in electronic form pursuant to Section 11(a) and Section 11(b) of the Lease; and (iv) all information contained in each report furnished pursuant to Section 11(e) of the Lease.

"Conflict Opinion" has the meaning set forth in Section 7.1.5 of the Participation Agreement.

"CRAF Program" means the Civil Reserve Air Fleet Program authorized under 10 U.S.C. §9511 et seq. or any similar or substitute program under the laws of the United States.

"Delivery Date" has the meaning specified in Lease Supplement No. 1.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

"Dollars" and "\$" mean the lawful currency of the United States of America.

"EASA" means the European Aviation Safety Agency of the European Union and any successor agency.

"Engine" means (i) each of the engines listed by manufacturer's serial numbers in Lease Supplement No. 1, whether or not from time to time installed on the Airframe or installed on any other airframe or on any other aircraft; (ii) any Replacement Engine which may from time to time be substituted, pursuant to the Return Conditions or Sections 8(d) or 10(d) of the Lease, for an Engine leased under the Lease; and (iii) any and all Parts incorporated or installed in or attached to such Engine or Replacement Engine or any and all Parts removed from such Engine or Replacement Engine so long as title thereto shall remain vested in Lessor in accordance with the terms of Section 8 of the Lease after removal from such Engine or Replacement Engine, but in each case, except any Excluded Equipment. Except as otherwise set forth in the Lease, at such time as a Replacement Engine shall be so substituted, the replaced Engine shall cease to be an Engine. The term "Engines" also means, as of any date of determination, all Engines then leased under the Lease.

"Engine Manufacturer" means [NAME OF ENGINE MANUFACTURER].

["Engine Warranty Agreement" means that certain Engine Warranty Agreement, dated as of the Delivery Date, between Lessee, Lessor and Engine Manufacturer, in form and substance satisfactory to the parties thereto.]<sup>1</sup>

"ERISA" means the Employee Retirement Income Security Act of 1974 and the regulations promulgated and rulings issued thereunder.

"Event of Default" has the meaning set forth in Section 14 of the Lease.

"Event of Loss" with respect to any property means any of the following events with respect to such property:

- (i) loss of such property or the use thereof due to theft, disappearance, destruction, damage beyond repair or rendition of such property permanently unfit for normal use for any reason whatsoever;
- (ii) any damage to such property that results in an insurance settlement with respect to such property on the basis of a total loss or a compromised or constructive total loss;
- (iii) the condemnation, confiscation or seizure of, or requisition of title to or use of, such property (other than a requisition for use of the Aircraft, the Airframe or any Engine by the U.S. government or any agency or instrumentality thereof which shall not have resulted in loss of possession of such property for a period continuing beyond the end of the Term) which, in the case of any event referred to in this clause (other than requisition of title), shall have resulted in the loss of possession of such property by Lessee (or any Permitted Sublessee) for a period in excess of 120 consecutive days or a shorter period that ends on or after the last day of the Term (in which event the Event of Loss pursuant to this <u>clause (iii)</u> shall be deemed to have occurred on the last day of the Term);
- (iv) as a result of any rule, regulation, order or other action by the FAA, the Department of Transportation or other governmental body of the U.S. or other country of registry of the Aircraft having jurisdiction, the use of such property in the normal course of passenger air transportation shall have been prohibited for a period of six consecutive months, unless Lessee (or any Permitted Sublessee), prior to the expiration of such six-month period, shall have undertaken and shall be diligently carrying forward all steps which in its judgment are necessary or desirable to permit the normal use of such property by Lessee (or any Permitted Sublessee) or, in any event, if such use shall have been prohibited for a period of twelve consecutive months; or
- (v) the operation or location of the Aircraft, while under requisition for use by the U.S. government, in any area excluded from coverage by any insurance policy in

Include if applicable.

effect with respect to the Aircraft required by the terms of <u>Section 11</u> of the Lease, if Lessee shall be unable to obtain indemnity or insurance in lieu thereof from the U.S. government;

<u>provided</u> that if such property shall be returned to Lessee in usable condition after the occurrence of an event described in <u>clause (i)</u>, (iii) or ( $\underline{v}$ ) above but prior to the date on which Stipulated Loss Value would be payable pursuant to <u>Section 10(a)</u> of the Lease, then such event shall, at the option of Lessee, not constitute an Event of Loss.

"Exchange Act" means the Securities Exchange Act of 1934, and the rules and regulations promulgated thereunder.

"<u>Excluded Equipment</u>" means (<u>i</u>) defibrillators, enhanced emergency medical kits and other medical and emergency equipment, (<u>ii</u>) airphones and other components or systems installed on or affixed to the Airframe that are used to provide individual telecommunications or electronic entertainment or services to passengers aboard the Aircraft, (<u>iii</u>) branded passenger convenience or service items, and (<u>iv</u>) cargo containers.

"Export Administration Regulations" means the United States Export Administration Regulations, 15 C.F.R. §§ 730-774, as amended, modified or supplemented from time to time, and any successor thereto.

"FAA" or "Federal Aviation Administration" means the Federal Aviation Administration of the U.S. and any successor governmental authority.

"FAA Bill of Sale" means the bill of sale for the Aircraft on Federal Aviation Administration AC Form 8050-2 executed by the Manufacturer in favor of Owner Trustee.

"Fair Market Rental Value" means the rental value which could be obtained in an arm's-length transaction between an informed and willing lessee under no compulsion to lease and an informed and willing lessor in possession under no compulsion to lease, assuming that the Aircraft is unencumbered by the Lease and is in the condition required thereby; provided that, in determining such value under Section 15 of the Lease, the Aircraft shall be valued on an "as-is, where-is" basis, taking into account customary brokerage and other reasonable costs and out-of-pocket expenses that would be typically incurred in connection with the re-letting of equipment such as the Airframe, Engines or any Part thereof. Such value shall be determined by mutual written agreement between Lessor and Lessee or, in the absence of mutual written agreement, pursuant to an Independent Appraisal, except in determining such value under Section 15 of the Lease, such value shall be determined by Independent Appraisal as provided therein.

"Fair Market Sales Value" means the sales value which could be obtained in an arm's-length transaction between an informed and willing purchaser under no compulsion to purchase and an informed and willing seller in possession under no compulsion to sell, assuming that the Aircraft is unencumbered by the Lease and is in the condition required thereby; provided that, in determining such value under Section 15 of the Lease, the Aircraft shall be valued on an "as-is, where-is" basis, taking into account customary brokerage and other reasonable costs and

out-of-pocket expenses that would be typically incurred in connection with the sale of equipment such as the Airframe, Engines or any Part thereof. Such value shall be determined by mutual written agreement between Lessor and Lessee or, in the absence of mutual written agreement, pursuant to an Independent Appraisal, except in determining such value under Section 15 of the Lease, such value shall be determined by Independent Appraisal as provided therein.

"Indemnified Person" means Owner Trustee, Trust Company and Owner Participant (including, for this purpose, a Person identified in writing to Lessee by Owner Participant who manages or services Owner Participant's interest in the Trust Estate) and each Back-Leveraging Indemnified Person that has been added as an "Indemnified Person" in a consent and acknowledgment described in <a href="Section 8.3.2.(b)">Section 8.3.2.(b)</a> of the Participation Agreement and their respective officers, directors, servants, agents, successors and permitted assigns, but excluding any such Person in its capacity as the manufacturer, supplier or subcontractor of the Aircraft, Airframe or Engines or any Part and any officer, director, servant, agent, successor or permitted assign of such Person in such capacity.

"Independent Appraisal" means an appraisal mutually agreed to by two nationally recognized independent aircraft appraisers, one of which appraisers shall be chosen by Lessor and one by Lessee, or, if such appraisers cannot agree on such appraisal, an appraisal arrived at by a third nationally recognized independent aircraft appraiser chosen by the mutual consent of such two appraisers; provided that if either party shall fail to appoint a nationally recognized independent aircraft appraiser within 15 days after a written request to do so by the other party, the "Independent Appraisal" shall be the appraisal rendered by the appraiser that has been appointed; provided, further, that if both Lessor and Lessee appoint nationally recognized independent aircraft appraisers but such appraisers cannot agree on an appraisal and fail to appoint a third nationally recognized independent aircraft appraiser within 20 days after the date of the appointment of the second of such appraisers, then either party may apply to the American Arbitration Association to make such appointment. In the event such third independent appraiser shall be chosen to provide such appraisal, unless the parties agree otherwise, such appraisal shall be required to be made within 20 days of such appointment. Notwithstanding the foregoing, if an Independent Appraisal is used to determine the Fair Market Rental Value for the purposes of Section 21 of the Lease, the time periods set forth in the two preceding sentences shall be shortened to the extent necessary to allow the Fair Market Rental Value to be determined within 30 days after Lessee provides its revocable notice of its intent to renew the Lease pursuant to Section 21 of the Lease. Notwithstanding the foregoing, if an Independent Appraisal is used to determine the Fair Market Rental Value and/or Fair Market Sales Value for the purposes of Section 15 of the Lease, an Independent Appraisal shall be an appraisal prepared by Ascend Worldwide Limited, BK Associates, Inc., AVITAS, Inc. or another nationally recognized independent aircraft appraiser chosen in good faith by Owner Participant. The fees and expenses of appraisers for an Independent Appraisal, whenever undertaken pursuant to the Lease, shall be borne equally by Lessor and Lessee, and Lessor and Lessee each shall separately bear any fees, costs and expenses of its respective attorneys and experts (other than the appraisers referred to above) incurred in connection with such Independent Appraisal, except that the costs of an Independent Appraisal undertaken pursuant to Section 15 of the Lease shall be for the account of Lessee.

- "Inspecting Party" has the meaning set forth in Section 12(a) of the Lease.
- "Insurance Threshold Amount" has the meaning set forth in Schedule A to the Participation Agreement.
- "International Interest" has the meaning ascribed to the defined term "international interest" under the Cape Town Treaty.
- "International Registry" means the international registry established pursuant to the Cape Town Treaty.
- "Investment Company Act" means the Investment Company Act of 1940, and the rules and regulations promulgated thereunder.
- "<u>Law</u>" means and includes (<u>a</u>) any statute, decree, constitution, regulation, order, judgment or other directive of any governmental authority; (<u>b</u>) any treaty, pact, compact or other agreement to which any governmental authority is a signatory or party; (<u>c</u>) any judicial or administrative interpretation or application of any Law described in (a) or (b) above; and (<u>d</u>) any amendment or revision of any Law described in (a), (b) or (c) above.
- "Lease" or "Lease Agreement" means that certain Lease Agreement ([YEAR] MSN [MSN]), dated as of [ ], [YEAR], between Lessor and Lessee.
- "Lease Expiry Date" means the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] anniversary of the Delivery Date.
- "<u>Lease Period</u>" means the period commencing on and including each Lease Period Date and ending on and including (<u>i</u>) the day immediately preceding the next subsequent Lease Period Date, (<u>ii</u>) the Lease Expiry Date, in the case of final Lease Period during the Basic Term, or (<u>iii</u>) the last day of any Renewal Term, in the case of the final Lease Period during such Renewal Term.
  - "Lease Period Date" means, during the Basic Term or any Renewal Term, each date specified in Schedule A to Lease Supplement No. 1.
- "Lease Supplement" means (i) Lease Supplement No. 1 and (ii) any other supplement to the Lease Agreement from time to time executed and delivered in connection with one or more Replacement Engines.
- "<u>Lease Supplement No. 1</u>" means a lease supplement, substantially in the form of Exhibit A to the Lease, entered into between Lessor and Lessee on the Delivery Date for the purpose of subjecting the Aircraft to the Lease.
  - "Lessee" means American Airlines, Inc., a Delaware corporation.
  - "Lessor" means Owner Trustee as lessor under the Lease.

"<u>Lessor's Liens</u>" means any Lien on or relating to or affecting the Aircraft, the Airframe, any Engine or any Part, title thereto or any interest therein, the Lease or the Trust Estate arising as a result of:

- (i) Claims against or affecting Trust Company, Owner Trustee, Owner Participant or any Owner Participant Guarantor, as applicable, not related to ownership of the Aircraft or the transactions contemplated by the Lease and the other Operative Documents;
- (ii) acts or omissions of Trust Company, Owner Trustee, Owner Participant or any Owner Participant Guarantor, as applicable, not related to the transactions contemplated by, or not expressly provided for under the terms of, the Lease and the other Operative Documents;
- (iii) Taxes or Claims imposed against Trust Company, Owner Trustee, Owner Participant or any Owner Participant Guarantor, as applicable, which are not indemnified against by Lessee pursuant to Section 7.1, 7.2 or 7.3 of the Participation Agreement; or
- (iv) Taxes or Claims imposed against Trust Company, Owner Trustee, Owner Participant or any Owner Participant Guarantor, as applicable, arising out of any voluntary or involuntary Transfer (other than pursuant to the Return Conditions or Section 8 or 10 of the Lease or pursuant to the exercise of remedies in accordance with Section 15 of the Lease in connection with an Event of Default that shall have occurred and be continuing) by Trust Company, Owner Trustee, Owner Participant or any Owner Participant Guarantor, as applicable, of its respective interest in the Aircraft, the Airframe, any Engine, any Part, any Obsolete Part, the Trust Estate, Rent or any interest arising under any Operative Document, including, without limitation, by means of granting a security interest therein;

<u>provided</u> that an arrangement expressly permitted by <u>Section 8.3</u> of the Participation Agreement shall not constitute a Lessor's Lien so long as such arrangement remains compliant with <u>Section 8.3</u> of the Participation Agreement.

"LIBOR" means, with respect to any Rent payment not paid when due, the rate for deposits in Dollars for a period of one month which appears on the Reuters Screen LIBOR01 Page as of 11:00 A.M., London time, two London Banking Days prior to the date such Rent payment came due. If such rate does not appear on the Reuters Screen LIBOR01 Page, the rate will be determined on the basis of the rates at which deposits in Dollars are offered by the principal London offices of the Reference Banks at approximately 11:00 A.M., London time, on the date two London Banking Days prior to the date such Rent payment came due to prime banks in the London interbank market for a period of one month commencing on such date and in an amount of \$10,000,000. Lessor will request the principal London office of each of the Reference Banks to provide a quotation of its rate. If at least two such quotations are provided, the rate for such overdue Rent payment will be the arithmetic mean of the quotations. If fewer than two quotations are provided as requested, the rate for such overdue Rent payment will be the arithmetic mean of the rates quoted by major banks in New York City, selected by Lessor, at

approximately 11:00 A.M., New York City time, on the date two London Banking Days prior to the date such Rent payment came due for loans in Dollars to leading European banks for a period of one month commencing on such date and in an amount of \$10,000,000.

- "Lien" means any mortgage, pledge, lien, charge, encumbrance, lease, conditional sale or security interest.
- "London Banking Day" means any day on which commercial banks are not authorized or required to close in London, England and which is also a day on which dealings in U.S. Dollar deposits are carried out in the London Interbank market.
  - "Loss Payee" means Lessor, except as otherwise provided in a consent and acknowledgment described in Section 8.3.2(b) of the Participation Agreement.
  - "Loss Payment Date" has the meaning specified in Section 10(a) of the Lease.
  - "Maintenance Program" has the meaning specified in Section 7(a)(ii) of the Lease.
  - "Manufacturer" means Airbus S.A.S.
  - "Marketing Inspection" has the meaning specified in Section 12(b) of the Lease.
  - "Moody's" means Moody's Investors Services, Inc. (or any successor thereto that is a nationally recognized statistical rating organization).
  - "MPD" means the maintenance planning document published by the Manufacturer and applicable to the Aircraft, as revised from time to time.
  - "New Owner Participant" has the meaning set forth in Section 8.1 of the Participation Agreement.
  - "New Owner Trustee" has the meaning set forth in Section 8.1 of the Participation Agreement.
  - "Obsolete Parts" has the meaning specified in Section 8(c) of the Lease.
- "Operative Documents" means the Lease, the Participation Agreement, the Trust Agreement[, the Engine Warranty Agreement]<sup>2</sup> and any Owner Participant Guarantee.
- "<u>Overdue Rate</u>" means, as at any date of determination, an interest rate equal to [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] per annum plus the applicable LIBOR, calculated on the basis of a 360-day year and the number of actual days elapsed.

<sup>&</sup>lt;sup>2</sup> Include if applicable.

"Owner Participant" means [Name of Owner Participant], a [jurisdiction] [type of entity].

"<u>Owner Participant Guarantee</u>" means [(<u>i</u>) for so long as [Name of Owner Participant] is the Owner Participant, the Guarantee ([YEAR] MSN [MSN]), dated as of [ ], by [Name of Owner Participant Guarantor], and (<u>ii</u>) otherwise,]<sup>3</sup> an absolute and unconditional guarantee by the applicable Owner Participant Guarantor, substantially in the form of <u>Exhibit G</u> to the Participation Agreement or, otherwise, in form and substance reasonably satisfactory to Lessee, delivered pursuant to <u>Section 8.2(a)(ii)</u> of the Participation Agreement.

"Owner Participant Guarantor" [(i) with respect to the period during which [Name of Owner Participant] is Owner Participant under the Operative Documents, means [Name of Owner Participant Guarantor] and  $(\underline{ii})$  with respect to the period during which any Transferee is Owner Participant under the Operative Documents, has the meaning set forth in Section 8.2.(a)(ii) of the Participation Agreement.

"Owner Trustee" means (i) Wells Fargo Bank Northwest, National Association, a national banking association, not in its individual capacity, but solely in its capacity as owner trustee under the Trust Agreement, or (ii) if Wells Fargo Bank Northwest, National Association is not then serving as Owner Trustee under the Trust Agreement, the successor Owner Trustee under the Trust Agreement.

"Participation Agreement" means that certain Participation Agreement ([YEAR] MSN [MSN]), dated as of [ ], [YEAR], among Lessee, Owner Trustee, Trust Company and Owner Participant.

"<u>Parts</u>" means any and all appliances, parts, instruments, appurtenances, accessories, furnishings and other equipment of whatever nature (other than (i) complete Engines or engines and (ii) any Excluded Equipment), so long as the same are incorporated or installed in or attached to the Airframe or any Engine or so long as title thereto remains vested in Lessor in accordance with the terms of <u>Section 8</u> of the Lease after removal from the Airframe or any Engine.

"Permitted Country" means each of the countries listed in Schedule A to the Participation Agreement.

"Permitted Investment" means each of the following:

- (i) direct obligations of the U.S. and agencies thereof;
- (ii) obligations fully guaranteed by the U.S.;
- Include if there will be an Owner Participant Guarantee for the initial Owner Participant.
- Include if there will be an Owner Participant Guarantee for the initial Owner Participant.

- (iii) certificates of deposit issued by, or bankers acceptances of, or time deposits with, any bank, trust company or national banking association incorporated or doing business under the laws of the U.S. or one of the states thereof having combined capital and surplus and retained earnings of at least \$100,000,000, and having a rating of A, its equivalent or better by Moody's or S&P (or if neither such organization shall rate such institution at any time, by any nationally recognized rating organization in the United States);
  - (iv) commercial paper of any holding company of a bank, trust company or national banking association described in clause (iii);
- (v) bearer note deposits with, or certificates of deposit issued by, or promissory notes of, any subsidiary incorporated under the laws of Canada (or any province thereof) of any bank, trust company or national banking association described in <u>clause (iii)</u>, (viii) or (ix);
- (vi) commercial paper of companies having a rating assigned to such commercial paper by Moody's or S&P (or, if neither such organization shall rate such commercial paper at any time, by any nationally recognized rating organization in the United States) equal to either of the two highest ratings assigned by such organization;
- (vii) U.S. dollar-denominated certificates of deposit issued by, or time deposits with, the European subsidiaries of ( $\underline{A}$ ) any bank, trust company or national banking association described in <u>clause (iii)</u> or ( $\underline{B}$ ) any other bank described in <u>clause (viii)</u> or ( $\underline{ix}$ );
- (viii) U.S.-issued Yankee certificates of deposit issued by, or bankers acceptances of, or commercial paper issued by, any bank having combined capital and surplus and retained earnings of at least \$100,000,000 and headquartered in Canada, Japan, the United Kingdom, France, Germany, Switzerland or The Netherlands;
- (ix) U.S. dollar-denominated time deposits with any Canadian bank having a combined capital and surplus and retained earnings of at least \$100,000,000 and having a rating of A, its equivalent or better by Moody's or S&P (or, if neither such organization shall rate such institution at any time, by any nationally recognized rating organization in the United States);
  - (x) Canadian Treasury Bills fully hedged to U.S. dollars;
- (xi) repurchase agreements with any financial institution having combined capital and surplus and retained earnings of at least \$100,000,000 collateralized by transfer of possession of any of the obligations described in clauses (i) through (x) above; and
- (xii) money market mutual funds that are registered with the Security and Exchange Commission under the Investment Company Act and operated in accordance with Rule 2a-7 and that at the time of such investment are rated "Aaa" by Moody's and/or "AAA" by S&P.

"**Permitted Lien**" has the meaning specified in Section 6 of the Lease.

### "Permitted Sublessee" means:

- (i) the Manufacturer or Engine Manufacturer (or any Affiliate of either thereof);
- (ii) any Certificated Air Carrier;
- (iii) any foreign air carrier that is principally based in and a domiciliary of a Permitted Country, if, at the time Lessee enters into a sublease with such foreign air carrier, Lessor receives an opinion from counsel to Lessee (which counsel shall be reasonably satisfactory to Lessor) to the effect that:
  - (A) all filing, recording and other action necessary to perfect and protect Lessor's rights and interests in and to the Aircraft and the Lease has been accomplished;
  - (B) there exist no possessory rights in favor of such sublessee under the laws of such sublessee's country which would, upon bankruptcy or insolvency of or other default by Lessee and assuming that at the time of such bankruptcy, insolvency or other default by Lessee, such sublessee is not insolvent or bankrupt, prevent the return of an Engine or the Airframe and each Engine or engine subject to such sublease to Lessor in accordance with and when permitted by the terms of Sections 14 and 15 of the Lease upon the exercise by Lessor of its remedies under Section 15 of the Lease; and
  - (C) the terms of the Lease are legal, valid, binding and enforceable in the country in which such foreign air carrier is principally based (subject to customary exceptions); or
  - (iv) any foreign air carrier not described in clause (iii) above consented to in writing by Lessor (such consent not to be unreasonably withheld);

<u>provided</u> that in the case of any such foreign air carrier referred to in <u>clause (iii)</u> or (<u>iv</u>) above (other than a foreign air carrier principally based in Taiwan), the U.S. maintains full diplomatic relations with the country in which such foreign air carrier is principally based at the time such sublease is entered into.

"Person" means any individual person, corporation, limited liability company, partnership, joint venture, association, joint-stock company, trust, trustee, unincorporated organization or government.

- "Qualifying Institution" has the meaning specified in Section 8.2(a)(ii) of the Participation Agreement.
- "Reference Banks" means Citibank, JP Morgan Chase Bank, Deutsche Bank, and such other or additional banking institutions as may be designated from time to time by mutual agreement of Lessee and Lessor.
- "Reference Stipulated Loss Value Determination Date" means (i) with respect to Section 10 of the Lease, the Stipulated Loss Value Determination Date on or immediately preceding the Loss Payment Date, (ii) with respect to Section 15(c) of the Lease, the Stipulated Loss Value Determination Date on or immediately preceding the Specified Payment Date and (iii) with respect to Section 15(d) of the Lease, the Stipulated Loss Value Determination Date on or immediately preceding the Sale Date.
  - "Related Aircraft" means any aircraft that is a "Leased Aircraft" (as defined in the American/Airbus Purchase Agreement) other than the Aircraft.
- "Related Lease" means, as of any date of determination, any aircraft lease agreement in substantially the form of the Lease Agreement with respect to any Related Aircraft between Lessee, as lessee, and Wells Fargo Bank Northwest, National Association, as owner trustee (or a successor trust company or any other bank or trust company pursuant to a "Trust Transfer" with respect to such aircraft lease agreement), as lessor, provided that, as of such date, the "Owner Participant" with respect to such aircraft lease agreement are identical or are Affiliates and the further conditions in both of the following clauses (A) and (B) are satisfied: (A) with respect to such aircraft lease agreement, either (x) Owner Participant is and has been the "Owner Participant" since the inception of such aircraft lease agreement or [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] and (B) with respect to the Lease Agreement, Owner Participant is and has been [name of initial Owner Participant] since the inception of the Lease Agreement.
- "Related Indemnitee Group" with respect to any Indemnified Person, subject to Section 8.3.2(a) of the Participation Agreement, means each of such Indemnified Person's officers, directors, servants, agents, successors and permitted assigns.
  - "Renewal Term" has the meaning set forth in Section 21 of the Lease.
  - "Rent" means Basic Rent and Supplemental Rent.
  - "Replaced Engine" has the meaning set forth in Section 8(d)(i) of the Lease.

"Replacement Engine" means an engine of the same make and model as the Replaced Engine (or engine of the same or another manufacturer of a comparable or an improved model and suitable for installation and use on the Airframe and compatible with the other Engine) which shall have been substituted under the Lease pursuant to the Return Conditions or Sections 8(d) or 10(d) of the Lease, together with all Parts relating to such engine, but in each case excluding any Excluded Equipment.

"Re-registration Conditions" means the terms and conditions set forth in Schedule B to the Participation Agreement.

"Responsible Officer" means, with respect to Lessee, its Chairman of the Board, its President, any Executive Vice President, any Senior Vice President, the Chief Financial Officer, any Vice President, the Treasurer, the Secretary or any other management employee (i) whose power to take the action in question has been authorized, directly or indirectly, by the Board of Directors of Lessee, (ii) working under the supervision of any such Chairman of the Board, President, Executive Vice President, Senior Vice President, Chief Financial Officer, Vice President, Treasurer or Secretary and (iii) whose responsibilities include the administration of the transactions and agreements contemplated by the Lease and other Operative Documents.

"Return Conditions" means the return conditions set forth in Annex B to the Lease.

"Return Date" has the meaning set forth in Annex B to the Lease.

"<u>S&P</u>" means Standard & Poor's Ratings Services, a Standard & Poor's Financial Services LLC business (or any successor thereto that is a nationally recognized statistical rating organization).

"Sale" has the meaning ascribed to the term "sale" in the Cape Town Treaty.

"Sale Date" has the meaning specified in Section 15(d) of the Lease.

"Section 1110" means Section 1110 of the United States Bankruptcy Code of 1978 (11 U.S.C. § 1110).

"Securities Act" means the Securities Act of 1933, and the rules and regulations promulgated thereunder.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

"Specified Payment Date" has the meaning set forth in Section 15(c) of the Lease.

"<u>Specified Persons</u>" means Owner Trustee, Trust Company, Owner Participant and, to the extent provided in <u>Section 8.3</u> of the Participation Agreement, each Back-Leveraging Indemnified Person. that has been added as an "Indemnified Person" in a consent and acknowledgment described in <u>Section 8.3.2(b)</u> of the Participation Agreement.

"Stipulated Loss Value" with respect to the Aircraft means (i) during the Basic Term, the amount set forth in Schedule B to Lease Supplement No. 1 opposite the Stipulated Loss Value Determination Date that is the Reference Stipulated Loss Value Determination Date and (ii) during any Renewal Term, the amount determined as provided in Section 21 of the Lease applicable to the Stipulated Loss Value Determination Date that is the Reference Stipulated Loss Value Determination Date.

"<u>Stipulated Loss Value Determination Date</u>" means (<u>i</u>) during the Basic Term, each date specified in <u>Schedule B</u> to Lease Supplement No. 1 and (<u>ii</u>) during any Renewal Term, each Lease Period Date occurring during such Renewal Term.

"Sublease Period" means any period during which a sublease permitted by the terms of the Lease is in effect.

"Successor" has the meaning set forth in Section 6.1.3(a) of the Participation Agreement.

"<u>Supplemental Rent</u>" means all amounts (other than Basic Rent) which Lessee agrees to pay to Lessor, Owner Participant or any other Indemnified Person or Tax Indemnitee pursuant to any Operative Document, including without limitation payments of Stipulated Loss Value and indemnities payable under <u>Sections 7.1</u> and <u>7.2</u> of the Participation Agreement.

"<u>Taxe</u>" or "<u>Taxes</u>" means all governmental or quasi-governmental fees (including, without limitation, license, filing and registration fees) and all taxes (including, without limitation, franchise, excise, stamp, value added, income, gross receipts, sales, use, property, personal and real, tangible and intangible taxes), withholdings, assessments, levies, imposts, duties or charges, of any nature whatsoever, together with any penalties, fines, additions to tax or interest thereon or other additions thereto imposed, levied or assessed by any country, taxing authority or governmental subdivision thereof or therein or by any international authority, including any taxes imposed on any Person as a result of such Person being required to collect and pay over withholding taxes.

"<u>Tax Indemnitee</u>" means Lessor, Owner Trustee, Trust Company, Owner Participant and each Back-Leveraging Indemnified Person that has been added as a "Tax Indemnitee" in a consent and acknowledgment described in <u>Section 8.3.2(b)</u> of the Participation Agreement (including any security trustee that has been so added) and their respective officers, directors, servants, agents, successors and permitted assigns, and, with respect to any Tax imposed on a consolidated or combined group of companies of which Lessor, Owner Trustee, Trust Company, Owner Participant or any such Back-Leveraging Indemnified Person is a member, such group and any member thereof, but excluding any such Person in its capacity as the manufacturer, supplier or subcontractor of the Aircraft, Airframe or Engines or any Part and any officer, director, servant, agent, successor or permitted assign of such Person in such capacity.

"Term" means the Basic Term and, if actually entered into, any Renewal Term.

- "Transfer" means an offer, sale, assignment, transfer, participation, conveyance or other disposition.
- "Transferee" has the meaning set forth in Section 8.2(a)(i)(A) of the Participation Agreement.
- "<u>Transportation Code</u>" means that portion of Title 49 of the United States Code comprising those provisions formerly referred to as the Federal Aviation Act of 1958, as amended.
  - "Trust" means the trust created under the Trust Agreement.
- "<u>Trust Agreement</u>" means that certain Trust Agreement ([YEAR] MSN [MSN]), dated as of [ ], [YEAR], between Owner Participant and Trust Company.
- "<u>Trust Company</u>" means (<u>i</u>) Wells Fargo Bank Northwest, National Association, a national banking association, in its individual capacity, or (<u>ii</u>) if such Wells Fargo Bank Northwest, National Association is not then serving as Owner Trustee under the Trust Agreement, the entity serving as Owner Trustee, in its individual capacity.
- "Trust Estate" means all estate, right, title and interest of Owner Trustee in and to the Aircraft, the Participation Agreement, the Lease and the other Operative Documents, including, without limitation, all amounts of Basic Rent, Supplemental Rent, insurance proceeds (other than any insurance proceeds payable under liability policies to or for the benefit of Trust Company, for its own account or in its individual capacity, or to Owner Participant) and requisition, indemnity or other payments of any kind for or with respect to the Aircraft including, without limitation, any and all payments and proceeds received by Owner Trustee after the termination of the Lease with respect to the Aircraft resulting from the sale, lease or other disposition thereof.
  - "Trust Indenture Act" means the Trust Indenture Act of 1939.
  - "Trust Transfer" has the meaning set forth in Section 8.1 of the Participation Agreement.
  - "UCC" means the Uniform Commercial Code, as in effect in any applicable jurisdiction.
  - "U.S." or "United States" means the United States of America.
  - "Warranty Bill of Sale" means the warranty (as to title) bill of sale with respect to the Aircraft executed by the Manufacturer in favor of Owner Trustee.

### [FORM OF LEASING LETTER]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

1. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 5. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 7. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 8. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 9. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 10. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 11. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

EXHIBIT B

FORM OF LEASE AGREEMENT

EXHIBIT C

FORM OF TRUST AGREEMENT

EXHIBIT D

FORM OF PARTICIPATION AGREEMENT

FORM OF DEFINITIONS ANNEX

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

 $1. \ [\text{CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]\\$ 

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

4. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- 1. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### SCHEDULE I

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**EXHIBIT G** 

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

## PART A – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

I. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (A) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (B) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- II. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (A) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (B) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (C) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (D) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
  - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
    - [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

- (E) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- (F) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

PART B – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

PART C – [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**EXHIBIT H** 

ANNEX A
TO

## [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

**EXHIBIT J** 

## [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

(a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### **EXHIBIT J**

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

\* \* \* \* \*

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

(a) [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Exhibit A

to

Exhibit J

 $[CONFIDENTIAL\ PORTION\ OMITTED\ AND\ FILED\ SEPARATELY\ WITH\ THE\ COMMISSION\ PURSUANT\ TO\ A\ REQUEST\ FOR\ CONFIDENTIAL\ TREATMENT]$ 

EXHIBIT K

# [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

### AMERICAN AIRLINES, INC Computation of Ratio of Earnings to Fixed Charges (in millions)

	Three Months Ended September 30,		Nine Months Ended September 30,	
	2011	2010	2011	2010
Loss:				
Loss before income taxes	\$ (153)	\$ 129	\$ (868)	\$ (367)
Add: Total fixed charges (per below)	433	401	1,286	1,202
Less: Interest capitalized	11	7	27	23
Total earnings (loss) before income taxes	\$ 269	\$ 523	\$ 391	\$ 812
Fixed charges:				
Interest	\$ 173	\$ 165	\$ 512	\$ 503
Portion of rental expense representative of the interest factor	255	232	758	683
Amortization of debt expense	5	4	16	16
Total fixed charges	\$ 433	\$ 401	\$1,286	\$1,202
Ratio of earnings to fixed charges		1.30		
Coverage deficiency	\$ 164	<u>\$                                    </u>	\$ 895	\$ 390

### I, Gerard J. Arpey, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of American Airlines, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 19, 2011 /s/ Gerard J. Arpey

Gerard J. Arpey

Chairman and Chief Executive Officer

#### I, Isabella D. Goren, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of American Airlines, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 19, 2011 /s/ Isabella D. Goren

Isabella D. Goren

Senior Vice President and Chief Financial Officer

### American Airlines, Inc. Certification

# Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code)

Pursuant to section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code), each of the undersigned officers of American Airlines, Inc., a Delaware corporation (the Company), does hereby certify, to such officer's knowledge, that:

The Quarterly Report on Form 10-Q for the quarter ended September 30, 2011 (the Form 10-Q) of the Company fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934 and information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: October 19, 2011 /s/ Gerard J. Arpey

Gerard J. Arpey

Chairman and Chief Executive Officer

Date: October 19, 2011 /s/ Isabella D. Goren

Isabella D. Goren

Senior Vice President and Chief Financial Officer

The foregoing certification is being furnished solely pursuant to section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code) and is not being filed as part of the Form 10-Q or as a separate disclosure document.